# **Technical Report on Mirror Bacteria: Feasibility and Risks**

December, 2024

## Abstract

This report describes the technical feasibility of creating mirror bacteria and the potentially serious and wide-ranging risks that they could pose to humans, other animals, plants, and the environment. It accompanies the *Science* Policy Forum article titled "Confronting risks of mirror life", published in December 2024.

In a mirror bacterium, all of the chiral molecules of existing bacteria—proteins, nucleic acids, and metabolites—are replaced by their mirror images. Mirror bacteria could not evolve from existing life, but their creation will become increasingly feasible as science advances. Interactions between organisms often depend on chirality, and so interactions between natural organisms and mirror bacteria would be profoundly different from those between natural organisms. Most importantly, immune defenses and predation typically rely on interactions between chiral molecules that could often fail to detect or kill mirror bacteria due to their reversed chirality. It therefore appears plausible, even likely, that sufficiently robust mirror bacteria could spread through the environment unchecked by natural biological controls and act as dangerous opportunistic pathogens in an unprecedentedly wide range of other multicellular organisms, including humans.

This report draws on expertise from synthetic biology, immunology, ecology, and related fields to provide the first comprehensive assessment of the risks from mirror bacteria. It consists of eight chapters and starts with a general introduction, followed by an examination of the initial creation of mirror bacteria, their further engineering, as well as biosecurity and biosafety implications. The remaining five chapters cover risks to human health, medical countermeasures, risks to other animals, risks to plants, and the potential ecological consequences of their introduction into the environment.

## **Report Authors**

In alphabetical order: Katarzyna P. Adamala, University of Minnesota Deepa Agashe, National Centre for Biological Sciences, Bengaluru Damon J. Binder, Open Philanthropy Yizhi Cai, University of Manchester Vaughn S. Cooper, University of Pittsburgh Ryan K. Duncombe, Mirror Biology Dialogues Fund Kevin M. Esvelt, Massachusetts Institute of Technology John I. Glass, J. Craig Venter Institute Timothy W. Hand, University of Pittsburgh Thomas V. Inglesby, Johns Hopkins Center for Health Security Farren J. Isaacs, Yale University Jonathan D. G. Jones, Sainsbury Laboratory, University of East Anglia Richard E. Lenski, Michigan State University Gregory Lewis, Massachusetts Institute of Technology Ruslan Medzhitov, Yale University Matthew L. Nicotra, Johns Hopkins Center for Health Security Sebastian B. Oehm, University of Cambridge Jaspreet Pannu, Johns Hopkins Center for Health Security and Stanford University David A. Relman, Stanford University Hiroaki Suga, University of Tokyo Johanna M. Sweere, Mirror Biology Dialogues Fund Jack W. Szostak, University of Chicago Nicholas J. Talbot, Sainsbury Laboratory, University of East Anglia Brian Wang, Panoplia Laboratories

#### Contact

The authors welcome comments and feedback. Please contact us via technical-report@mbdialogues.org.

# Review

Prior to release, feedback on scope, completeness, accuracy, and presentation of the analysis was solicited from scientific experts in each of the areas covered by the report. Experts were invited to comment on one or several chapters of the report. The authors of each chapter retained final editorial authority over the content and integration of expert feedback. Providing review comments does not imply endorsement of the report's overall findings or conclusions. Experts were given the option to be publicly acknowledged in this report. Reviewers who agreed to be acknowledged were (in alphabetical order):

Rocco Casagrande, Gryphon Scientific Matthew Wook Chang, National University of Singapore George M. Church, Harvard University Gitta L. Coaker, University of California, Davis Jeffery L. Dangl, University of North Carolina at Chapel Hill Drew Endy, Stanford University Jeff Gore, Massachusetts Institute of Technology Chuan He, University of Chicago Sheng Yang He, Duke University Kenya Honda, Keio University Wilhelm T. S. Huck, Radboud University Jonathan C. Kagan, Harvard University Michael S. Kay, University of Utah Simon A. Levin, Princeton University Robert D. Miller, University of New Mexico John H. Rex, AMR.Solutions Daniel Simberloff, University of Tennessee, Knoxville James M. Tiedje, Michigan State University Paul E. Turner, Yale University J. Craig Venter, J. Craig Venter Institute

# **About this Report**

This technical report analyzes the feasibility of creating mirror bacteria, evaluates the possible consequences of their release into the environment, and outlines some potential countermeasures.

Many authors of this report are co-authors of the article "Confronting risks of mirror life", published in *Science* in December 2024. Both that article and this technical report emerged from the activities of an international working group of scientific experts from a range of disciplines. This technical report informed the findings and recommendations of the *Science* article; while that article contains policy recommendations, this report focuses solely on technical analysis and risk assessment. The views expressed in this report reflect those of the authors and not necessarily those of their affiliated institutions.

This work is licensed under a Creative Commons Attribution-NonCommercial-ShareAlike 4.0 International License (CC BY-NC-SA 4.0).

### **Rationale for Public Release**

Releasing this report inevitably draws attention to a potentially destructive scientific development. We do not believe that drawing attention to threats is always the best approach for mitigating them. However, in this instance we believe that public disclosure and open scientific discussion are necessary to mitigate the risks from mirror bacteria. We have two primary reasons to believe disclosure is necessary:

#### 1) To prevent accidents and well-intentioned development

If no serious concerns are raised, the default course of well-intentioned scientific and technological development would likely result in the eventual creation of mirror bacteria. Creating mirror life has been a long-term aspiration of many academic investigators, and efforts toward this have been supported by multiple scientific funders.<sup>1</sup> While creating mirror bacteria is not yet possible or imminent, advances in enabling technologies are expected to make it achievable within the coming decades. It does not appear possible to develop these technologies safely (or deliberately choose to forgo them) without widespread awareness of the risks, as well as deliberate planning to mitigate them.

This concern is compounded by the possibility that mirror bacteria could accidentally cause irreversible harm even without intentional misuse. Without awareness of the threat, some of the most

<sup>&</sup>lt;sup>1</sup> For instance, in 2019 the U.S. National Science Foundation awarded \$4 million to a group of investigators seeking "to design, construct, and safely deploy synthetic mirror cells in which all of the key molecules—nucleic acids, proteins, carbohydrates, and lipids—exist in chiral states opposite to their natural forms" (National Science Foundation Award #1935120, #1935372). The National Natural Science Foundation of China has similarly supported work toward the creation of a mirror-image central dogma (National Natural Science Foundation of China grants 32050178 and 21925702), and the European Commission ERA-Net MirrorBio consortium, with similar goals, was supported by the German Federal Ministry of Education and Research (Bundesministerium für Bildung und Forschung grant 031A461).

dangerous modifications would likely be made for well-intentioned reasons, such as endowing mirror bacteria with the ability to metabolize D-glucose to allow growth in standard media.

#### 2) To build guardrails that could reliably prevent misuse

There are currently substantial technical barriers to creating mirror bacteria. Success within a decade would require efforts akin to those of the Human Genome Project or other major scientific endeavors: a substantial number of skilled scientists collaborating for many years, with a large budget and unimpeded access to specialized goods and services. Without these resources, entities reckless enough to disregard the risks or intent upon misuse would have difficulty creating mirror bacteria on their own. Disclosure therefore greatly reduces the probability that well-intentioned funders and scientists would unwittingly aid such an effort while providing very little actionable information to those who may seek to cause harm in the near term.

Crucially, maintaining this high technical barrier in the longer term also appears achievable with a sustained effort. If well-intentioned scientists avoid developing certain critical components, such as methods relevant to assembling a mirror genome or key components of the mirror proteome, these challenges would continue to present significant barriers to malicious or reckless actors. Closely monitoring critical materials and reagents such as mirror nucleic acids would create additional obstacles. These protective measures could likely be implemented without impeding the vast majority of beneficial research, although decisions about regulatory boundaries would require broad discussion amongst the scientific community and other stakeholders, including policymakers and the public. Since ongoing advances will naturally erode technical barriers, disclosure is necessary in order to begin discussions while those barriers remain formidable.

### **Content Decisions**

An understanding of both the major technical barriers and how they could be eroded by technological progress over the coming decades is necessary for grounded scientific deliberation and the development of effective guardrails. For this reason, Chapters 2 and 3 provide a high-level assessment of the potential pathways by which mirror bacteria might be created and then transformed into other species. They do not provide explicit, detailed technical instructions for the creation of a mirror bacterium (and given the immaturity of the precursor technologies, doing so would be impossible at this stage).

As a matter of prudence, our discussion of risks focuses on mirrored versions of existing bacteria, particularly those that are commonly studied and engineered, as well as traits that could arise naturally through mutation. We also discuss certain modifications to mirror bacteria that legitimate researchers would likely pursue if unaware of the risks described in this report. We intentionally do not discuss possible modifications that would serve only destructive purposes.<sup>2</sup>

 $<sup>^{2}</sup>$  The closest the report comes to this is in Chapter 3, where we describe at a high level how a malicious or reckless actor could remove added safety mechanisms. We believe such analysis is essential for understanding which, if any, biocontainment strategies are promising.

# **Table of Contents**

Abstract	i
Report Authors	ii
Review	iii
About this Report	iv
Table of Contents	vi
Boxes, Figures, and Tables	viii
Summary	1
Chapter 1: Introduction	5
Chapter 2: Pathways to Mirror Life	26
2.1 Advances in chemistry permit the synthesis of mirror biomolecules with diverse applications	28
2.2 Progress in synthetic biology could allow the assembly of a mirror bacterium from non-living mirror components	33
2.3 A natural-chirality bacterium might be converted into a mirror bacterium in a stepwise fashion	41
2.4 Other approaches to creating mirror bacteria are plausible	50
2.5 The feasibility of mirror life will increase as related technologies advance	51
Chapter 3: Engineering, Biosafety, and Biosecurity of Mirror Bacteria	54
<ul><li>3.1 The creation of any mirror bacterium could enable the generation of diverse mirror bacterial strains and species and their modification by routine genetic engineering</li><li>3.2 Biocontainment approaches might reduce accident risk, but they would face challenges</li></ul>	55 60
3.3 Creating robustly biosecure mirror bacteria is not feasible	62
Chapter 4: Risks to Human Health	65
4.1 Innate immune detection of mirror bacteria could be significantly impaired	67
4.2 Mirror bacteria would likely be resistant to most innate immune responses	72
4.3 Adaptive immunity to mirror bacteria would likely be impaired	81
4.4 Mirror bacteria could plausibly pass barrier surfaces and translocate into the bloodstream and tissues	91
4.5 Mirror bacteria could plausibly replicate in blood and cause lethal systemic infection	98

Chapter 5: Medical Countermeasures	107
5.1 New antimirror compounds could be developed to target mirror bacteria,	
but most existing antibiotics would not function	108
5.2 Conjugate vaccines could plausibly be developed against mirror bacteria	113
5.3 The efficacy of other countermeasures against mirror bacterial infection is unclear	116
Chapter 6: Animal Infection	119
6.1 Vertebrate susceptibility to mirror bacterial infection would likely be similar to that of	
humans	120
6.2 Many invertebrates would likely be susceptible to mirror bacteria	123
Chapter 7: Plant Infection	134
7.1 Mirror bacteria are likely to evade plant innate immunity	135
7.2 Mirror bacteria could plausibly establish chronic local infections within leaves and roots	138
7.3 The extent to which mirror bacteria could spread through and colonize	
plant vascular tissue is unclear	145
7.4 Countermeasures for agricultural plants	151
Chapter 8: Environmental Survival and Spread	156
8.1 Mirror bacteria would be inherently resistant to many biological controls	157
8.2 Mirror bacteria could colonize natural environments outside of multicellular hosts	165
8.3 Invasive mirror bacteria could rapidly disperse through the environment	175
8.4 Invasive mirror bacteria could rapidly evolve and diversify	178
8.5 Invasive mirror bacteria could cause irreversible ecological harm	181
8.6 Countermeasures to invasive mirror bacteria might lessen but would not halt	
the ecological damage	187
Contributions and Acknowledgments	192
References	195

# **Boxes, Figures, and Tables**

## Boxes

1.1: Violations of chiral symmetry are negligible in biological systems	13
1.2: Could natural-chirality molecules be toxic to mirror life?	18
1.3: Why is mirror life absent from nature?	22
2.1: Applications of mirror peptides and proteins	29
2.2: Applications of mirror nucleic acids	32
3.1: Some potential applications of mirror bacteria would require high-risk strains	56
6.1: Mirror bacterial infection would likely be fatal to <i>D. melanogaster</i> due to defective	
peptidoglycan recognition	124
8.1: Dictyostelium discoideum may not effectively phagocytose and kill mirror bacteria	160
8.2: Escherichia coli can survive and replicate under many environmental conditions	170

## Figures

1.1: Chiral molecules have non-superimposable mirror images	6
1.2: Living organisms are built from both chiral and achiral molecules	7
1.3: Mirror-image amino acids compose mirror-image peptides that fold into mirror-image prote	ins 8
1.4: Mirror-image enzymes catalyze mirror-image reactions	9
1.5: A mirror image of the central dogma	10
2.1: Chemical synthesis of mirror proteins	30
2.2: Synthesis of kilobase-length natural-chirality and mirror DNA and RNA	33
2.3: Components of ribosomes and the PURE system	38
2.4: Milestones towards the creation of a mirror bacterium	40
2.5: Genetic code reprogramming allows the genetically encoded synthesis of non-canonical	
biopolymers in cells	45
2.6: Natural, crossover, and mirror-image translation	46
2.7: Stepwise conversion of a natural-chirality bacterium into a mirror bacterium	49
4.1: The human health risks of mirror bacteria	67
4.2: Innate immune recognition of mirror bacteria could be severely impaired	69
4.3: Complement activity against mirror bacteria would likely be impaired	75
4.4: Mirror bacteria could evade phagocytosis and killing by macrophages	80
4.5: T cell activation is likely to be severely impaired in response to mirror bacterial infection	84
5.1: Conjugate vaccines may improve adaptive immune recognition of mirror bacteria	115

6.1: Mirror bacteria would likely evade <i>D. melanogaster</i> immune mechanisms due to defective	
peptidoglycan recognition	126
6.2: The number of PRR genes varies widely across animal species	132
7.1: Mirror bacteria are likely to evade detection by plant immunity	136
7.2: Mirror bacteria might enter leaves through open stomata and grow within the leaf apoplast	140
7.3: The anatomy of a root could facilitate mirror bacterial colonization	144
7.4: Plants transport water and nutrients via xylem and phloem	146
8.1: Mirror bacteria could be resistant to protists, a major cause of bacterial mortality	162
8.2: Mirror phototrophic bacteria could evade predation, thereby colonizing large fractions of	
the ocean with severe consequences for marine ecosystems	173

### Tables

1.1: Achiral organic molecules that can be utilized by wild-type or mutant <i>E. coli</i> K-12	15
1.2: Utilization of L- and D-amino acids by E. coli	16
1.3: Utilization of the enantiomers of common monosaccharides by E. coli	17
5.1: Classes of existing antibiotics and characteristics relevant to their applicability for treatment of mirror bacterial infections	110
8.1: Naturally occurring antibiotics that could retain efficacy against mirror bacteria	163
8.2: Efficacy of broad-spectrum bacteriocins against mirror bacteria	164
8.3: Other examples of bacterial weapon systems	166
8.4: Rates of spread of animals and novel pathogens during historical invasions and epidemics	176

## **Summary**

Many of the key molecules common to all living organisms are chiral: their mirror images cannot be superimposed upon one another. In nature, nearly all amino acids and sugars are predominantly present as only one of their two possible mirror-image forms. Synthetic mirror polypeptides and nucleic acids fold into perfect mirror images of their natural counterparts and interact identically with the mirror images of their natural substrates. The same principle applies to other biological structures. Scaling this logic, a mirror cell in which all molecules were replaced with their mirror-images would function identically to a natural-chirality cell, but with everything precisely mirrored.

Scientists have long considered the possibility of creating mirror cells, including mirror bacteria. In isolation, they would behave identically to natural-chirality bacteria—mirror bacteria would grow under the same physical conditions, and would grow at the same rates on achiral nutrients. But because natural life is chiral, interactions between natural organisms and mirror bacteria would be profoundly altered. This report describes the technical feasibility of creating mirror bacteria, and the potentially serious risks that mirror life could pose to humans, other animals, plants, and the environment.

# Robust mirror bacteria could be developed in the coming years and would be vulnerable to accidental or deliberate misuse

It is not yet possible to create a living cell from non-living precursors. Despite major technical challenges, it appears plausible that this feat could be achieved within as little as a decade given sufficient resources. Once methods of creating natural-chirality cells have been established, a similar approach could be used to create mirror cells from mirror components. A small group of researchers and funders have begun to explicitly work toward the creation of mirror life, with a focus on a "bottom-up" approach, in which a mirror genome as well as transcription and translation machinery are used to build other mirror cellular components. Modern methods for protein synthesis are in principle sufficient to synthesize most of the mirror proteins that would be needed to construct a mirror cell, though currently at prohibitive costs. Mirror-image enzymes have recently been used to construct kilobase-length mirror RNA and DNA, and research is progressing toward building a mirror ribosome.

Other pathways to mirror bacteria appear possible. For example, a "top-down" approach, in which a natural-chirality bacterium is stepwise converted into a mirror bacterium, may become technically feasible as translational engineering advances.

The first mirror bacterium would likely be a fragile microbe exhibiting metabolic defects, which would limit its growth and durability outside the laboratory. Once created, however, mirror bacteria could be readily engineered to become more robust by using standard techniques to deliver mirror versions of existing bacterial genes. This could confer new capabilities or even transform them into

#### Summary

mirrored versions of robust existing bacteria. Adding mirrored versions of known metabolic pathways would allow mirror bacteria to utilize natural-chirality glucose and other common sugars.

Biocontainment measures could ensure that mirror bacteria are incapable of survival outside of a lab. However, these measures would be vulnerable to human error or deliberate misuse. Achieving robust biocontainment that could not be removed by a moderately skilled actor with access to mirror DNA does not appear feasible.

# Mirror bacteria could evade many aspects of human immunity and potentially cause life-threatening infection

Most immunological mechanisms rely on precise stereospecific interactions between host and pathogen macromolecules. The mirror-image macromolecules of mirror bacteria would likely not properly bind to host receptors, enzymes, or other host effectors, as they would have the "wrong" chirality. This could grant mirror bacteria a degree of intrinsic immune evasion well beyond any known natural pathogen.

The immune system could be compromised in three key ways. First, the innate immune response relies upon initial detection of conserved microbial biomolecules, such as bacterial lipopolysaccharides and peptidoglycans, by host pattern recognition receptors. Because these molecules are almost exclusively chiral, immune recognition of mirror bacteria could be substantially impaired. Second, many innate immune mechanisms of pathogen control could be directly compromised; for example, phagocytosis, antimicrobial enzymes, and several complement system pathways rely on stereospecific protein interactions. Finally, mirror proteins would resist degradation and other stereospecific mechanisms necessary for antigen processing and presentation by innate immune cells, which would impair the activation of adaptive T and B immune cells and antibody production.

Human exposure to intrinsically immune-evasive mirror organisms may lead to infection and possibly death. While mirror bacteria would lack functional virulence factors that facilitate invasion, transient bacteremias caused by environmental bacteria are common, suggesting that exposure to mirror bacteria through inhalation, ingestion, microabrasions, or wounds could result in passive translocation across epithelial barriers.

Once inside, mirror versions of common bacteria such as *Escherichia coli* would be able to grow on achiral nutrients such as glycerol and, with suitable engineering, on glucose and other common chiral nutrients. Impaired immunity would likely permit extensive replication within the bloodstream. The exact clinical presentation of a mirror bacterial infection is unclear, but absent an effective immune response, a lethal outcome appears highly plausible.

Treatment options would be limited. Most antibiotics interact stereospecifically with their microbial targets, so existing stocks would be restricted to a few achiral or racemic antibiotics. As an emerging infectious disease, there would be no pre-existing vaccine. It should be possible to develop novel antimirror compounds and conjugate vaccines; nevertheless, as with a new pandemic, the practical challenges to developing such measures quickly and at scale would be considerable.

#### Summary

Mirror bacteria could potentially infect many other animals and plants and colonize some external environments, causing irreversible ecological disruption

Other multicellular organisms may be similarly vulnerable to mirror bacterial infection. Vertebrates share broadly similar immune systems and would likely suffer from equivalent defects to humans. Invertebrate immune systems are more variable and less thoroughly characterized, but may be similarly ineffective. For example, the model insect *Drosophila melanogaster* relies on peptidoglycan recognition to initiate antibacterial immune mechanisms, and so is unlikely to recognize mirror bacteria. The model nematode *Caenorhabditis elegans* does not appear to rely on the recognition of pathogen-associated molecules, but instead on pathogen avoidance and the recognition of host damage; the latter may or may not permit nematodes to survive a mirror bacterial infection. The consumption of tissues from an infected organism by a predator or scavenger could lead to infection, causing mirror bacteria to spread through the food web.

Plant immune systems also rely on the detection of conserved pathogen biomolecules. Almost all known immune receptors in plants recognize chiral ligands, and thus would be unlikely to recognize mirror bacteria. Although mirror bacteria would not by default possess specific adaptations to invade and colonize plants, both plant roots and leaf stomata could allow for passive translocation of mirror bacteria from the environment. Whether mirror bacteria could spread through vasculature is unclear and may vary for different types of plants, but if systemic infections did occur they might be fatal. Key crops could probably be engineered to be resistant, but protecting wild plants and their associated ecosystems would be infeasible.

In addition to functioning as a dangerous "accidental pathogen" to a wide range of natural-chirality species, mirror bacteria could persist within and potentially colonize external environments. Unlike their natural chirality counterparts, mirror bacteria would be completely resistant to all bacteriophages, partially evasive of and largely indigestible to predators, and largely resistant to antimicrobial compounds released by competing microbial species. These potentially decisive competitive advantages could allow sufficiently robust mirror bacteria to successfully invade many ecological niches despite lacking specific adaptations for them. Because predators would not be able to digest most mirror macromolecules, a growing mirror bacterial population would not be controlled by any commensurate increase in predation, which could allow populations to reach high abundance.

Once released, invasive mirror bacteria could be spread rapidly by infected animals and humans, and by the transport of contaminated goods and produce. While a released mirror bacterium would likely have little pre-existing genetic variation and no ability for gene transfer with natural-chirality bacteria, evolutionary diversification and adaptation to expand the environmental and host range of mirror bacteria would ensue.

Mirror bacteria may directly drive vulnerable plant and animal species to extinction, and the loss of vulnerable "keystone species" could indirectly trigger severe ecological disruptions. Very large mirror bacterial populations, especially autotrophic mirror bacteria, may disrupt nutrient cycling in many ecosystems, and could impact the global carbon cycle. Ecological countermeasures such as the synthesis and release of mirror phages that target the invasive mirror bacteria could reduce their maximum population size, but would have little realistic prospect of averting these large and irreversible harms.

Katarzyna P. Adamala<sup>1</sup>, Damon J. Binder<sup>2</sup>, Kevin M. Esvelt<sup>3</sup>, John I. Glass<sup>4</sup>, Richard E. Lenski<sup>5,6</sup>, Jaspreet Pannu<sup>7,8,9</sup>, David A. Relman<sup>9,10,11</sup>

All known life on Earth shares a common set of chiral molecules. Proteins, sugars, lipids, and nucleic acids can all physically exist in either of two mirror-image configurations, but living organisms overwhelmingly use only one of these two configurations. By convention, one member of the pair is designated to be the L, or "left-handed" form, while the other is designated the D, or "right-handed" form. Virtually all proteins are constructed solely from the 20 canonical amino acids, of which all but achiral glycine are L-amino acids (Figure 1.1). Their mirror images are D-amino acids, which play only marginal roles in existing life. The ribose in RNA is always in the form of D-ribose; its mirror image L-ribose is essentially absent from nature (Figure 1.2).

Pairs of molecules that are mirror images of each other are known as enantiomers, which have almost exactly mirrored physical and chemical properties (see Box 1.1). Their crystalline structures are mirror images of each other and melt at the same temperatures. They have identical solubilities in water and identical acid-base dissociation constants, but they rotate polarized light in opposite directions.

Proteins are large, chiral molecules. Just as a protein with a specific sequence of L-amino acids will fold into a specific configuration, a mirror-image protein with the same specific sequence of D-amino acids folds into a configuration precisely mirroring that of the natural-chirality protein (Figure 1.3). If a protein naturally binds to some small molecule, an exact mirror-image of that protein will bind equally well to the enantiomer of that small molecule. This exact mirroring extends to catalytic activity: mirror-image enzymes will catalyze reactions involving the enantiomers of the products and substrates of the natural enzyme (Figure 1.4). Despite being exactly as versatile as natural-chirality proteins, mirror proteins cannot be made by natural living organisms and do not exist in nature.

Other mirror-image macromolecules also precisely mirror natural-chirality macromolecules. Mirror DNA, made with L-deoxyribose rather than D-deoxyribose, forms an opposite-handed double helix

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup> Department of Genetics, Cell Biology and Development, University of Minnesota, Minneapolis, Minnesota, USA; <sup>2</sup> Open Philanthropy, San Francisco, California, USA; <sup>3</sup> Media Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, USA; <sup>4</sup> J. Craig Venter Institute, La Jolla, California, USA; <sup>5</sup> Department of Microbiology, Genetics, and Immunology, Michigan State University, East Lansing, Michigan, USA; <sup>6</sup> Program in Ecology, Evolution, and Behavior, Michigan State University, East Lansing, Michigan, USA; <sup>7</sup> Center for Health Security, Bloomberg School of Public Health, Johns Hopkins University, Baltimore, Maryland, USA; <sup>8</sup> Department of Health Policy, Stanford School of Medicine, Stanford University, Stanford, California, USA; <sup>9</sup> Department of Medicine, Stanford University, Stanford, California, USA; <sup>10</sup> Department of

Microbiology and Immunology, Stanford School of Medicine, Stanford, California, USA; <sup>11</sup> Infectious Diseases Section, Veterans Affairs Palo Alto Health Care System, Palo Alto, California, USA



#### Figure 1.1: Chiral molecules have non-superimposable mirror images

A. The amino acid glycine is an achiral molecule, as it can be superimposed on its mirror image. **B.** All other canonical amino acids are chiral molecules, with two non-superimposable mirror images. Alanine is illustrated as an example. Even if D-alanine is rotated 180° so its amino and carboxyl groups are in the same orientation as L-alanine, as shown in the figure, the two cannot be superimposed because the methyl group in L-alanine is projecting into the page but the methyl group in D-alanine is projecting out of the page.

that precisely mirrors the double helix of natural-chirality DNA. Mirror nucleic acid polymerases could make new mirror DNA and mirror RNA from mirror nucleotides, and mirror ribosomes could synthesize mirror proteins from mirror amino acids using mirror mRNA templates (Figure 1.5).

Given the symmetry between enantiomers, it should in theory be possible to construct an organism entirely out of mirror biomolecules. Replacing every single molecule in a bacterium with its enantiomer would result in a "mirror bacterium" that would behave identically to a natural-chirality bacterium except for being mirrored. Such a mirror bacterium would natively consume "opposite" chirality nutrients relative to natural bacteria; for example, it would utilize L-glucose rather than D-glucose. But alone in an achiral medium, such as minimal media with achiral glycerol for a carbon source, a mirror bacterium should behave indistinguishably from its mirror-image natural-chirality bacterium.

It is well known that protein binding is stereospecific and therefore generally only one member of an enantiomeric pair is recognized by its binding partner. For example, the enantiomers of chiral drugs typically bind different receptors and consequently have different biological effects (Shen *et al.*, 2013; Smith, 2009). Most famously, thalidomide was originally approved as a treatment for morning

sickness, but it was manufactured as a racemic mixture of both enantiomers. One enantiomer acts as a mild sedative, but the other is harmful to dividing cells and can cause severe birth defects (Tokunaga *et al.*, 2018). Similarly, enantiomers often have different odors because they bind to different olfactory receptors; one enantiomer of the oil carvone smells like caraway, while the other smells like spearmint (Bentley, 2006). Larger mirror macromolecules are also generally not recognized by natural enzymes. Proteases, for instance, are generally unable to cleave p-peptides (Miller *et al.*, 1995), which therefore have much longer half-lives in the body, making them potentially useful therapeutics (Lander *et al.*, 2023).



#### Figure 1.2: Living organisms are built from both chiral and achiral molecules

**A.** Glycerol and simple fatty acids are common in animals and plants; they are also achiral molecules that can be superimposed on their mirror images. **B.** Sugars are among the most abundant biological molecules on Earth, and are almost always chiral. For example, both glucose and ribose have two enantiomers (L- and D-) which are non-superimposable mirror images of each other. Living organisms use only one of the available enantiomers of chiral molecules: all RNA in all organisms on Earth is composed of D-ribose, while L-ribose is essentially absent from nature.



**Figure 1.3: Mirror-image amino acids compose mirror-image peptides that fold into mirror-image proteins** All natural amino acids except glycine are chiral, and only the L-enantiomers are used in natural proteins. Substituting the L-amino acids for their mirror image D-amino acid enantiomers would result in a mirror-image polypeptide chain. The interactions between the side chains for amino acids in the D-peptide are the same as the L-peptide, but with the opposite orientation. Thus, the mirror-image peptide chain folds into a protein that is the mirror image of its natural chirality counterpart. (Structure image for example protein, carbonic anhydrase, produced by Pymol 3.0.0 from Protein Data Bank accession number 1CA2.)

A mirror lifeform would be made entirely of mirror-image molecules, and as a consequence its interactions with natural biological organisms would be profoundly different. Just as a right-handed glove cannot fit a left hand, the molecules within a mirror bacterium would not "fit" the opposite-handed molecules found in nature. The natural-chirality receptors and enzymes of natural organisms would not properly bind to the mirror metabolites and macromolecules produced by a mirror bacterium, nor would the mirror-image receptors and enzymes in a mirror bacterium properly bind to natural-chirality metabolites and macromolecules.

Scientists, including several authors of this report, have long considered the possibility of creating mirror cells. One motivation comes from advances in synthetic chemistry, which have permitted the creation of progressively larger mirror nucleic acids and proteins. These mirror macromolecules retain the flexible programmability and functionality of natural biopolymers, but they resist degradation by nucleases and proteases and are much less likely to trigger undesired immune reactions—both properties that are desirable in therapeutic molecules. It is currently laborious to create mirror macromolecules synthetically, but increasingly accessible mirror nucleic acids and mirror enzymes will enable the creation of larger and more complex mirror molecular systems.

Independently, many synthetic biologists seek to understand how cells could be created from their constituent molecules. This is motivated both by a desire to better understand how life first arose, and

to understand what other kinds of life might be possible. If a natural-chirality cell can be created from lifeless molecules, then a mirror-image cell could be created from mirror-image molecules using the same methods.



#### Figure 1.4: Mirror-image enzymes catalyze mirror-image reactions

Enzyme catalysis requires precise (and thus stereospecific) binding of substrate molecules to the active site. **A.** If all substrates and products of an enzyme are achiral, such as for carbonic anhydrase, the natural-chirality and mirror-image enzymes would behave identically. **B.** When the products of an enzymatic reaction are chiral, the natural-chirality enzyme exclusively produces one enantiomer and its mirror-image enzyme would produce the other. For example, both fumarase and mirror-image fumarase would produce malic acid from fumaric acid and water, but natural-chirality fumarase exclusively produces L-malic acid (the enantiomer seen in nature), while mirror-image fumarase would exclusively produce D-malic acid. **C.** When the substrate of an enzyme is chiral, catalysis is typically specific to that enantiomer. For example, natural-chirality phosphoglucomutase can convert D-glucose 1-phosphate to D-glucose 6-phosphate. In contrast, mirror-image phosphoglucomutase, fumarase, and phosphoglucomutase produced by Pymol 3.0.0 from Protein Data Bank accession numbers 1CA2, 3E04, and 3PMG respectively.)

Chapter 1: Introduction



#### Figure 1.5: A mirror image of the central dogma

**A.** The central dogma of molecular biology. Genetic information in the form of D-DNA is transcribed by natural-chirality RNA to D-messenger RNA (D-mRNA), which is in turn translated into natural-chirality polypeptides and proteins. All of these stages exclusively use one of the two possible chiralities: natural chirality RNA polymerase transcribes D-DNA into D-mRNA, which the natural-chirality ribosome and transfer RNA (tRNA) carrying L-amino acids then translate into L-proteins. **B.** The central dogma could be chirally inverted: L-DNA would be transcribed by a mirror-image RNA polymerase into L-mRNA, which would be translated into mirror-image D-proteins. (Structure images for RNA polymerase, the ribosome, tRNA, and example protein (phosphoglucomutase) produced by Pymol 3.0.0 from Protein Data Bank accession numbers 4YG2, 5AFI, 6UGG, and 3PMG respectively.)

Once created, mirror cells could be further engineered using many of the same methods as natural-chirality organisms. Genes from natural-chirality organisms could be recreated with mirror-image DNA and engineered into mirror organisms, and the functionality of these genes would exactly mirror that of their natural counterparts. Diverse mirror lifeforms could thus be created by copying natural life. By perfectly mirroring existing life in isolation, but exhibiting greatly altered

interactions with natural organisms, mirror life could present an unprecedented and difficult to predict mixture of familiar and entirely novel behavior.

The authors of this report have come to the conclusion that mirror bacteria could present unique and severe risks. Their altered interactions with natural organisms could lead to profound and highly consequential effects on those natural organisms, as will be detailed in the subsequent chapters of this technical report.

The immune systems of humans, other animals, and plants primarily detect pathogens through the recognition of conserved chiral molecules. Recognition is stereospecific—it depends on the precise spatial arrangement of atoms in a molecule—and so the mirrored chiral molecules of mirror bacteria may not be properly recognized. Nor would antimicrobial enzymes or chiral antibiotics necessarily harm mirror bacteria, since they generally operate through chiral mechanisms. Protists and other predators could similarly have difficulty recognizing, killing, and digesting mirror bacteria. Mirror bacteria would also be entirely resistant to all natural bacteriophages. As a result, many if not most of the biological mechanisms that protect multicellular organisms from bacterial infection and limit bacterial populations in natural ecosystems would not function properly against mirror bacteria.

Like natural bacteria, mirror bacteria could be able to survive and replicate under diverse conditions. Some mirror bacteria may be fragile or lack the ability to synthesize required metabolites, just as is the case for many natural and engineered bacteria. But once the first mirror cells exist, creating more robust and diverse mirror bacteria should be straightforward and, absent concerns about potential risks, likely to be pursued by scientists seeking to advance the science and its practical applications. Such mirror bacteria could grow on achiral nutrients that are commonly found in many environments. They could also be engineered to consume D-glucose and other common chiral nutrients, and though there could be practical benefits to such engineering, this would also greatly facilitate growth in the environment.

If mirror bacteria could replicate within humans or other animals without being effectively cleared by the host immune system, they could establish a systemic infection with severe and plausibly fatal consequences. If that host or its carcass was consumed by another animal, the infection could spread, leading to transmission through the food chain. If mirror bacteria could replicate in soils and natural waters, but were not kept in check by natural enemies such as predators and phages, they would become invasive, severely disrupting natural ecosystems.

This chapter provides an overview of the entire report. Chapter 2 describes potential pathways scientists could pursue to create the first mirror bacterium. Chapter 3 then describes how mirror bacteria could be engineered to improve robustness and metabolic capabilities, which would raise serious biosafety and biosecurity challenges. Together these two chapters explain why, were it not for safety and security concerns, we think scientific progress over the next one to few decades could lead to the creation of mirror bacteria.

The remaining five chapters describe in detail the risks from mirror bacteria to humans, other animals, plants, and the environment more broadly. Chapter 4 describes the potential consequences of mirror bacteria for human health, detailing first the likely failures of both innate and adaptive immunity to recognize and respond to mirror bacteria before discussing the plausible course and

consequences of a systemic mirror bacterial infection. Chapter 5 then gives a brief overview of potential medical countermeasures that could protect humans from mirror bacteria, and the difficulties that developing and deploying such countermeasures would likely face.

Chapters 6 and 7 discuss other animals and plants, respectively. As with humans, serious immunological impairments could plausibly enable mirror bacteria to establish systemic infections within many animals and plants. Chapter 8 describes the survival and spread of mirror bacteria in the environment, both through and external to multicellular hosts, and closes by describing the irreversible ecological damage that an invasive mirror bacterium could cause.

Many aspects of our conclusions are necessarily tentative and uncertain. Natural organisms and ecosystems are complex and highly diverse, and any assessment of risk must extrapolate from limited information and a handful of examples subjected to detailed study. This single report, written at a single point in time with access to limited information on mirror biology, cannot be considered definite. We hope that others will build upon our initial analyses to examine these interdisciplinary risks in greater detail, potentially unearthing key considerations that we may have overlooked.

#### Mirror bacteria could plausibly be created in the next one to few decades

Many laboratories currently seek to achieve synthetic abiogenesis: the creation of life from its chemical building-blocks. The synthetic cell community is working to encapsulate self-replicating nucleic acids within a membrane and initiate metabolism under conditions that will trigger replication. Other biologists aim to simplify and then build synthetic versions of existing bacteria (Rothschild *et al.*, 2024). As the smallest known bacterial genome size continues to decline—currently at half a million base pairs for engineered strains with reduced genomes (Hutchison *et al.*, 2016)—the feasibility of creating a cell from scratch will grow.

Work toward mirror life is a natural outgrowth of work on synthetic cells. If scientists could construct a living bacterium from natural-chirality molecular precursors, the same methods could be used to construct a mirror bacterium from opposite-chirality precursors. While normal bioengineering often fails because the engineered alterations are incompatible with the rest of the organism, the processes for creating the first mirror cells could closely follow efforts to create natural-chirality cells. The first mirror cell could be an exact mirror-image of a pre-existing (albeit synthetic) natural-chirality cell.

Research on mirror-image proteins and nucleic acids is primarily motivated by potential near-term applications (see Boxes 2.1 and 2.2 in Chapter 2). Creating mirror life is a longer-term aspiration of multiple laboratories and major funders in an effort to better understand life and potentially aid in therapeutic development. For example, the U.S. National Science Foundation awarded \$4 million to a group of investigators aiming "to design, construct, and safely deploy synthetic mirror cells in which all of the key molecules—nucleic acids, proteins, carbohydrates, and lipids—exist in chiral states opposite to their natural forms" (National Science Foundation, 2019a; National Science Foundation, 2019b). The National Natural Science Foundation of China has likewise supported work on the creation of a mirror-image central dogma (National Natural Science Foundation of China, 2022), and the European Commission's ERA-NET MirrorBio consortium, with similar goals, was

#### Box 1.1: Violations of chiral symmetry are negligible in biological systems

Throughout this report we assume that mirror life would perfectly mirror natural-chirality life. If enantiomeric molecules perfectly mirrored one another, then it follows that the behavior of mirror bacteria would perfectly mirror that of natural-chirality bacteria. Conversely, if mirror bacteria did not perfectly mirror natural-chirality bacteria, this would imply the existence of some physical process that violates chiral symmetry between enantiomers.

Of the four fundamental forces in physics, only the weak force violates chiral symmetry. Any violation of chiral symmetry would result in a "parity-violation energy difference", or PVED, between enantiomers. Theoretical calculations for the weak force estimate a PVED of about  $10^{-11}$  J/mol (Horný & Quack, 2015), which is about  $10^{14}$  times smaller than thermal noise at room temperature, and  $10^{16}$  times smaller than the strength of a typical chemical bond. It is therefore extremely difficult to see how such negligible effects could have any role in biology.

Attempts to experimentally observe differences between enantiomeric molecules have not yet succeeded (Quack *et al.*, 2022; Sallembien *et al.*, 2022). The most precise measurements to date, in CFHClBr, found no PVED at a sensitivity of ~10<sup>-8</sup> J/mol (Daussy *et al.*, 1999). While experimental methods cannot yet detect the PVED from the weak force, these precision measurements, along with others in particle, atomic, and molecular physics (Chupp *et al.*, 2019; Safronova *et al.*, 2018) provide compelling evidence for the absence of other, as yet unknown, processes that could violate chiral symmetry at molecular scales.

Precision measurements focus on individual atoms and small molecules because they are the easiest to study; the possibility that parity violation would somehow become visible only in larger systems is remote. Studies of mirror-image proteins (Milton *et al.*, 1992; Zawadzke & Berg, 1993) and mirror-image nucleic acids (Xu & Zhu, 2022) have not revealed any deviations from the expected mirror symmetry, nor have measurements of optical rotation angles or crystallographic structures of organic compounds provided any evidence of violations of chiral symmetry. It is therefore implausible that any hitherto unknown violation of chiral symmetry could play a role in biology, and so it is almost certain that a correctly constructed mirror life would essentially be a perfect mirror of natural life.

supported by the German Federal Ministry of Education and Research<sup>3</sup> and the U.S. National Science Foundation (National Science Foundation, 2014). We expect interest in mirror bacteria to grow as work on both synthetic cells and mirror-image macromolecules advances, which would make the task of building a mirror cell increasingly feasible.

<sup>&</sup>lt;sup>3</sup> Bundesministerium für Bildung und Forschung grant 031A461

Estimating timelines is difficult. It is likely that normal-chirality synthetic cells are still years away, and that creating a synthetic mirror bacterium would take longer. Still, as detailed in Chapter 2, the foundational technologies are rapidly progressing, and there are multiple pathways by which the first mirror bacterium could be created.

#### The creation of any mirror bacterium could enable the creation of robust mirror bacteria

While creating the first mirror bacterium would be extremely challenging, the subsequent engineering of an existing mirror bacterium should be comparatively straightforward. Any bioengineering technique that can be accomplished by engineering a natural-chirality bacterium with normal DNA would apply to an equivalent mirror bacterium and mirror DNA. Even if the first mirror cells were fragile and catabolically limited—which is plausible but not certain—a logical next step would be to further engineer mirror bacteria in order to functionally mirror existing bacterial strains. A minimal and fragile mirror *Escherichia coli*, for instance, could be converted into the exact mirrors of widely used strains such as *E. coli* K-12, BL21, and Nissle 1917, and plausibly into more diverse bacterial species. Chapter 3 discusses the potential engineering of mirror bacteria in more detail.

Absent the severe risks discussed in this report, the mirror bacteria of greatest interest and utility would likely be mirrored versions of commonly studied heterotrophs such as *E. coli, Bacillus subtilis*, and *Pseudomonas aeruginosa*. These bacteria are useful, in part, because they are robust and can grow under diverse conditions. Because chirality would have little impact on robustness to abiotic stressors such as temperature, acidity, and oxidative stress, mirror versions of these heterotrophs would be similarly robust under diverse conditions. Given the likelihood that the first mirror bacteria would be a mirrored strain of *E. coli* or another species commonly used in the lab, a "generalist" *E. coli* is frequently used in this report as an illustrative example.

#### Mirror bacteria could acquire nutrients in natural environments

Sugars, amino acids, and other common nutrients are typically chiral. It might therefore appear that a mirror bacterium could not survive in the natural world: it would be unable to eat the available organic compounds and would thus starve. But this is not the case. Many common nutrients would be accessible to the mirrored versions of common heterotrophic bacteria, and with suitable engineering, additional nutrients like p-glucose could be made accessible.

As a concrete illustration, consider the example of an exact mirror *E. coli* K-12. The preferred carbon source for natural-chirality *E. coli* is D-glucose, so a mirror *E. coli* would prefer L-glucose, which is thought to be essentially absent from nature. However, natural *E. coli* can grow on a wide range of achiral substrates as sole carbon sources (see Table 1.1), including a number of central metabolites, fatty acids, and fermentation byproducts that are common in both multicellular hosts and in many external environments. A mirror *E. coli* should therefore grow as well as natural *E. coli* on these compounds.

There is nothing particularly exceptional about *E*. coli K-12 in this regard. The major metabolic pathways of life include a number of achiral metabolites, which are relatively abundant in nature and can be acquired and utilized relatively easily as nutrient sources. Some common achiral metabolites

Sole carbon source	28	References	
Central Metabolites	<i>Citrate,</i> fumarate, glycolate, glyoxylate, a-ketoglutarate, pyruvate, succinate	Clark & Cronan, 1996; Lin, 1996; Tong <i>et al.</i> , 2020	
Fatty Acids	Acetate, acetoacetate, <i>butyrate</i> , propionate, <i>valerate</i> Medium-chain ( $C_6-C_{10}$ ) fatty acids Long-chain ( $\geq C_{12}$ ) fatty acids	Clark & Cronan, 1996; Clark & Rod, 1987	
Alcohols	Butanol, ethanol, propanol	Clark & Cronan, 1996; Clark & Rod, 1987	
Sugars and polyols	Dihydroxyacetone, <i>ethylene glycol</i> , galactitol, glycerol, mucate	Lin, 1996; Tong et al., 2020	
Aromatic Acids	<i>Benzoate</i> , m-coumarate, <i>2-furoate</i> , <i>3-hydroxyphenylacetate</i> , phenylacetate, phenylpropionate, phenylethylamine	Abdulrashid & Clark, 1987; Díaz <i>et al.</i> , 2001	
Amines	y-aminobutyrate, putrescine	Dover & Halpern, 1972; Prieto <i>et al.</i> , 1987	
Miscellaneous	$\gamma$ -hydroxybutyric acid, methyl pyruvate, m-tartaric acid	Tong et al., 2020	
Sole nitrogen sour	ces		
Amino Acids	Glycine	Reitzer, 2005	
Amines	Agmatine, $\gamma$ -aminobutyrate, dopamine, phenylethylamine, putrescine, spermidine, tyramine	Díaz et al., 2001; Reitzer, 2005	
Nucleobases	Adenine, cytidine, thymine, uracil Allantoin*, urate*	Cusa <i>et al.</i> , 1999; Iwadate & Kato, 2019; Tyler, 1978; Xi <i>et al.</i> , 2000	

**Table 1.1:** Achiral organic molecules that can be utilized by wild-type or mutant *E. coli* K-12

 *E. coli* K-12 can use a number of achiral organic compounds as either sole carbon or nitrogen sources. Entries in *italics* indicate use by mutants. \*Can only be utilized as a nitrogen source under anaerobic conditions.

cannot be utilized as a sole carbon source by natural *E. coli* K-12, but can be used by other bacteria. These include ethanol, glycine, inositol, methanol, and ribitol, although in all five cases strains of *E. coli* K-12 have been engineered to grow on these compounds as sole carbon sources (Cao *et al.*, 2020; Fung *et al.*, 2023; Keller *et al.*, 2022; Scangos & Reiner, 1978; Shiue & Prather, 2014).

A number of chiral nutrients could also be utilized by mirror *E. coli* K-12 to varying extents. Table 1.2 compares the catabolic utilization of L- and D-amino acids by natural-chirality *E. coli* K-12. Although amino acids of the wrong chirality cannot be incorporated directly into proteins by the ribosome, they can be degraded to provide energy and useful nutrients. A mirror *E. coli* could utilize seven of the canonical amino acids to some extent, and the ability to utilize a further eight could quickly arise through mutations (see Section 8.4 for further discussion of mirror bacterial evolution). Most notably, natural-chirality *E. coli* catabolizes L-alanine through D-alanine and can use either as a sole carbon and nitrogen source (Reitzer, 2005). A mirror *E. coli* would therefore have no difficulty catabolizing L-alanine, which is among the most abundant amino acids. Other common chiral

nutrients that could be utilized by mirror *E. coli* include lactic, malic, and tartaric acid, as natural *E. coli* can catabolize both enantiomers of these chiral compounds (Dong *et al.*, 1993; Lukas *et al.*, 2010; Unden *et al.*, 2016).

An exactly mirrored *E. coli* could not catabolize D-glucose, but engineering mirror *E. coli* to catabolize D-glucose should be feasible. Genes for L-glucose catabolism have been identified from  $\alpha$ -proteobacteria *Paracoccus laeviglucosivorans* (Shimizu *et al.*, 2012); while this ability has not yet been engineered into a natural-chirality *E. coli*, there appears to be no inherent barrier to doing so. Plausible catabolic pathways for other monosaccharides also exist (see Table 1.3). Mirror *E. coli* would also exhibit chemotaxis toward common sugars, though less effectively than natural-chirality *E. coli* (Adler *et al.*, 1973).

Amino Acid	L-isomer	d-isomer	References
Alanine	Sole C and N source	Sole C and N source	
Arginine	C source, sole N source	Utilize	Celis, 1977
Asparagine	Sole C and N source		
Aspartate	Sole C and N source		
Cysteine	C source, sole N and S source	C and N source, sole S source	Soutourina et al., 2001
Glutamine	C source, sole N source		
Glutamate	C source, sole N source	Utilize in cell wall	Dougherty et al., 1993
Histidine	Utilize	Utilize	Kuhn & Somerville, 1971
Isoleucine	Utilize	Utilize	Kuhn & Somerville, 1971
Leucine	Utilize	Utilize	Kuhn & Somerville, 1971
Lysine	Utilize		
Methionine	Utilize	Utilize	Kadner, 1977
Phenylalanine	Sole C and N source	Utilize	Kuhn & Somerville, 1971
Proline	Sole C and N source		
Serine	Sole C and N source	C source, sole N source	
Threonine	C source, sole N source	Utilize to synthesize isoleucine	Umbarger & Brown, 1956
Tryptophan	C source, sole N source	Utilize	Kuhn & Somerville, 1971
Tyrosine	Utilize	Utilize	Kuhn & Somerville, 1971
Valine	Utilize	Utilize	Kuhn & Somerville, 1971

Table 1.2: Utilization of L- and D-amino acids by *E. coli* 

Entries in *italics* indicate use by mutants. Citation to (Kuhn & Somerville, 1971; Reitzer, 2005) unless otherwise indicated. For some p-amino acids we were unable to find studies of their utilization by *E. coli*, though p-asparagine, p-aspartate, and p-lysine cannot be used as a sole nitrogen source (Bochner *et al.*, 2001). Glycine, which is achiral, is absent from the table; it can be used as a carbon source and as a sole nitrogen source.

Monosaccharide	Utilization by <i>E. coli</i>	Potential pathways	References
Hexoses			
L-fructose	No	Through isomerase to L-glucose	Park et al., 2007
L-galactose	By mutation	L-fucose regulatory mutation and selection	Zhu & Lin, 1986
L-glucose	No	Through L-gluconate to pyruvate and GAP	Shimizu et al., 2012
L-mannose	No	Through isomerase to L-fructose	Park et al., 2007
Pentoses			
D-arabinose	By mutation	L-fucose regulatory mutation and selection	LeBlanc & Mortlock, 1971
L-ribose	By mutation	Through mutation and selection	Trimbur & Mortlock, 1991
L-xylose	No	Through isomerase to L-xylulose	Usvalampi et al., 2012
L-xylulose	Yes	Native ability	Ibañez et al., 2000
Trioses			
L-glyceraldehyde	By mutation	Through glycerol-3-phosphate Through glycerol	Baldomà & Aguilar, 1987; Kalyananda <i>et al.</i> , 1987; Zhu & Lin, 1987

Table 1.3: Utilization of the enantiomers of common monosaccharides by *E. coli* 

*E. coli* is able to utilize the enantiomers of a number of common sugars. Note that unlike most sugars, L-arabinose is the form most commonly found in nature.

Catabolizing larger sugars such as L-sucrose or L-trehalose might be more challenging, as it is unclear whether suitable transporters or hydrolytic enzymes exist in nature. Nevertheless, there is no underlying principle that would prohibit engineering a mirror bacterium with rich and varied catabolic capabilities; absent concerns about the risks, such work could proceed rapidly.

In addition to organic carbon, heterotrophic bacteria also require inorganic nutrients, but these are almost always available in achiral forms. Like most bacteria, *E. coli* can acquire nitrogen from inorganic ammonium, nitrate, and nitrite, and from achiral organic compounds such as glycine and urea. Phosphorus and sulfur are likewise available as phosphate and sulfate, while chlorine and metals are available as monatomic ions. Indeed, an organic carbon source like D-glucose is often the only chiral nutrient used in minimal culture media.

It appears that many environments could have suitable abiotic conditions and sufficient nutrients to support growth of a mirror *E. coli* or other similarly robust mirror bacterium; we discuss specific environments in more detail in Chapters 4, 6, 7, and 8. It is conceivable that some common chiral compounds could be toxic to mirror bacteria, but as discussed in Box 1.2, this does not appear to be a significant impediment. Thus, there do not appear to be any fundamental barriers preventing the survival and replication of mirror bacteria in many natural environments.

#### Box 1.2: Could natural-chirality molecules be toxic to mirror life?

The organic compounds found in natural environments are generally not toxic to bacteria. But for a mirror bacteria containing mirror enzymes and other mirror macromolecules, it is conceivable that toxicities could arise. This possibility can be addressed by considering the equivalent "mirrored" question: are the enantiomers of common organic compounds toxic to natural bacteria?

D-amino acids are the most widely studied class of mirror metabolites. Studies have found that some can inhibit bacterial growth, but they typically require 1–10 mM concentrations for even mildly inhibitory effects (Kao *et al.*, 2017; Rumbo *et al.*, 2016; Sarkar & Pires, 2015; Soutourina *et al.*, 1999; Soutourina *et al.*, 2000). L-amino acid concentrations in most natural environments are typically much lower, and therefore would be unlikely to inhibit mirror bacterial growth.

A few D-amino acids are particularly toxic. D-tryptophan has been reported to mildly suppress *E. coli* growth at low concentrations (Soutourina *et al.*, 2000). However, given that *E. coli*, *Acinetobacter baumannii*, and *P. aeruginosa* can all sustain growth even at 5 mM (Rumbo *et al.*, 2016; Soutourina *et al.*, 2000), naturally occurring concentrations of L-tryptophan are unlikely to significantly inhibit mirror bacterial growth. Toxicity is likely caused by the accidental incorporation of D-tryptophan into proteins. D-amino acid tRNA deacylases, which are present in nearly all living organisms (Wydau *et al.*, 2009), protect against misincorporation toxicity by eliminating mischarged tRNAs. Simply upregulating this enzyme would likely provide mirror bacteria with additional protection for a mirror bacterium.

D-cysteine can likewise mildly suppress *E. coli* growth at micromolar concentrations in minimal media, although it has no inhibitory effects at even 10 mM in LB broth (Soutourina *et al.*, 2001). D-cysteine and D-cystine are actively transported into the cytosol through transporters for L-cysteine and L-cystine (Chonoles Imlay *et al.*, 2015; Zhou & Imlay, 2022). Toxicity appears to be primarily driven by inhibition of threonine deaminase, needed for the biosynthesis of isoleucine, leucine, and valine, as addition of these amino acids to minimal media reduces (though does not entirely alleviate) growth inhibition by D-cysteine (Soutourina *et al.*, 2001); L-cysteine is toxic through a similar mechanism (Sørensen & Pedersen, 1991). D-cysteine is detoxified by D-cysteine desulfhydrase, which decomposes it into pyruvate, ammonium, and hydrogen sulfide; upregulation of this enzyme increases the resistance of *E. coli* to D-cysteine toxicity (Soutourina *et al.*, 2001).

The extent to which L-cysteine would pose a barrier to growth of an unadapted mirror E. *coli* within living organisms is unclear. D-isoleucine, D-leucine, and D-valine would not be present in natural environments, so mirror-image E coli would likely be reliant on its mirror-image threonine deaminase for biosynthesis, and so plausibly could be susceptible to

growth inhibition by L-cysteine. Resistance to L-cysteine could rapidly emerge, for instance through mutations to the transporters or by upregulation of D-cysteine desulfhydrase. Utilization of naturally occurring amino acids could also obviate the need for threonine deaminase (see Table 1.3), and *E. coli* strains lacking threonine deaminase may still produce isoleucine through other metabolic pathways (Cotton *et al.*, 2020). It is plausible that L-cysteine would have similar inhibitory effects on other mirror bacterial strains, though not universally; *P. aeruginosa*, for instance, is uninhibited at millimolar concentrations of D-cysteine (Rumbo *et al.*, 2016).

To our knowledge, the enantiomers of common sugars are generally not toxic to bacteria. For *E. coli* the only exception we have found is L-glyceraldehyde, which can mildly inhibit *E. coli* growth at millimolar concentrations (Tang *et al.*, 1977). It appears, however, that toxicity is due to the oxidation of L-glyceraldehyde to L-glycerate, as *E. coli* mutants lacking aldehyde dehydrogenase can grow on L-glyceraldehyde (Zhu & Lin, 1987). The mechanism by which L-glycerate is toxic to *E. coli* is not clear, but glyceric acid is not particularly abundant in nature.

We have not found other plausible chiral molecules that would preclude mirror *E. coli* growth in natural environments. Not all enantiomers of common molecules have been tested and reported in the literature, and it is possible we have missed other known toxicities. Nevertheless, it appears unlikely that any natural-chirality organic compounds would prove unusually toxic to most mirror bacteria at the very low concentrations typically present in most environments.

#### Mirror bacteria could be innately resistant to many immunological mechanisms

Biochemically, the interiors of multicellular organisms are generally nutrient-rich and growth-permissive for many natural bacteria. Sufficiently robust mirror bacteria would similarly be capable of replicating within multicellular organisms by utilizing achiral nutrients and, with suitable engineering, common sugars such as D-glucose. Plants and animals have evolved sophisticated immune systems to protect themselves from bacterial infection. But both the innate immune receptors and the immune effectors that cause harm to bacteria are generally chiral, having evolved to target conserved chiral molecules specific to natural-chirality bacteria. Therefore, it is likely that many immunological mechanisms would not function effectively against mirror bacteria because they could not recognize the enantiomers of their targets.

Plant and animal immune systems, for example, possess pattern-recognition receptors (PRRs) that detect the presence of microbe-associated molecular patterns (MAMPs). While the specific PRRs vary between species, they typically recognize highly conserved MAMPs, such as lipopolysaccharides and peptidoglycans that are present in many or all bacteria. Almost all such MAMPs are chiral. Because interactions between PRRs and MAMPs are stereospecific, PRRs would not typically recognize mirror lipopolysaccharides, mirror peptidoglycan, or the other

opposite-chirality macromolecules produced by mirror cells. There could be exceptions: at least one achiral MAMP is known to be recognized by a human PRR (Uldrich *et al.*, 2020) and another in plants (Kakkar *et al.*, 2015), though both appear to have only marginal importance. Some PRRs might primarily recognize achiral regions of their targets and retain weak affinity for the enantiomer, and unexpected cross-reactivities could also trigger PPRs. Moreover, innate immune systems detect damage-associated molecular patterns (DAMPs) caused by cell death, but these are unlikely to be triggered until late in an infection and appear unlikely to help the immune system target and eliminate infecting mirror bacteria. Failures of innate immune recognition could seriously impair immunological responses to mirror bacterial infection, as discussed in Chapters 4, 6, and 7.

Humans and other vertebrates also possess an adaptive immune system, which is theoretically capable of generating antibodies against any novel pathogen and could in principle produce antibodies that would bind to mirror-image macromolecules. But as discussed in Section 4.3, many of the steps required to process and present antigens for T cell activation are stereospecific and would likely fail, severely limiting B cell activation and the production of antibodies. An MHC-II immune deficiency is perhaps the closest clinical analog, and in such patients the antigen-specific T and B cell response is profoundly impaired.

Impairments in immunological recognition would be further compounded by the inability of some, though not all, immune effectors to harm bacteria. Antimicrobial enzymes such as lysozyme, phospholipases, and proteases target peptidoglycan, phospholipids, and proteins, all of which are chiral; these enzymes are unlikely to retain significant activity against their mirror counterparts. Other antimicrobial peptides cause harm by disrupting membranes and, in at least some cases, would retain efficacy against mirror bacteria. Within vertebrates, the alternative complement pathway may still function, but other complement activation pathways could be more significantly impaired. The extent to which professional phagocytes could successfully internalize and kill mirror bacteria is unclear.

Chapters 4, 6, and 7 discuss these immunological failures, and the consequent possibility of mirror bacterial infection in humans, animals, and plants in more detail. In all cases, physical barriers are imperfect and allow for at least occasional bacterial translocation into vulnerable parts of the organism. Given the extent to which mirror bacteria would innately evade many immunological mechanisms, it appears plausible that a sufficiently robust mirror bacterium would be capable of entering and harming many multicellular hosts, including humans.

While mirror bacteria would lack the toxins or other virulence factors typically produced by natural pathogens, uncontrolled mirror bacterial growth within a multicellular host could prove dangerous. In humans, for instance, a systemic mirror bacterial infection might prove fatal through a sepsis-like severe immunological dysregulation (see Section 4.5). Even if this were avoided, a bloodstream mirror bacterial infection could likely deplete nutrients, modify blood chemistry, and cause physical disruptions to blood vessels and organs, all of which could be lethal.

#### Mirror bacteria could be innately resistant to most forms of predation

Chapter 8 discusses potential ecological risks from mirror bacteria. In the natural environment, bacterial populations are often kept in check by protists, phages, and other predators. Mirror bacteria would be completely resistant to natural phages, which could neither adsorb to mirror receptors nor replicate their genomes with mirror polymerase. Mirror bacteria may also avoid most predation from protists, which recognize bacterial prey through receptors that rely on stereospecific interactions. While the mechanisms of prey recognition are not well understood, plausible failures of protist receptors to recognize mirror metabolites and macromolecules would impair both tracking and engulfment of mirror bacteria. Even if ingested, it is not clear how effectively protists would kill mirror bacteria. It is clear, however, that the mirror macromolecules would not be digested by the natural-chirality hydrolases produced by protists. Protists would gain few, if any, nutrients from mirror bacteria and thus would be unlikely to evolve to consume them more effectively. Other bacterial predators such as nematodes and filter feeders could likely face similar challenges.

Although a robust mirror bacterium could potentially replicate in many environments, any given mirror strain would be unlikely to replicate as rapidly as natural bacteria in the same environment because it would not be able to utilize the full range of chiral nutrients available to natural organisms. Moreover, any single mirror strain would initially be poorly adapted to most of the environments it might encounter.

However, for a mirror strain to become established within a natural environment merely requires the mirror bacteria to replicate faster than they die. Replication rates of mirror bacteria relative to natural bacteria could be lower, but so could mortality rates, as mirror bacteria are likely to evade many forms of predation. Slowly growing mirror bacteria might therefore be able to invade some ecosystems.

An expanding mirror bacterial population would face no commensurate expansion in phages or predation because phages would not infect mirror bacteria and predators would gain few, if any, nutrients from consuming them. As a result, mirror bacterial populations could become large in successfully colonized environments, with growth ceasing only once the available nutrients became sufficiently depleted.

# Invasive mirror bacteria could spread and rapidly adapt to new environments, causing severe ecological harms

A mirror bacterium capable of infecting multicellular organisms could spread through varied means: predation of infected animals and scavenging of infected corpses, insect bites and similar vectors, consumption of infected plants, and exposure to contaminated soil or water. It is likely that a mirror bacterium capable of infecting many multicellular species would undergo sustained transmission through the environment.

Natural pathogens can spread many kilometers each day if transported by insects, birds, or other rapidly dispersing hosts. If mirror bacteria were able to infect such animals, they would likewise

#### Box 1.3: Why is mirror life absent from nature?

The risks from mirror bacteria arise because reversed-chirality confers significant advantages through immune evasion and predation resistance. If reversed-chirality could be so advantageous, why does mirror life appear to be absent from nature?

Evolution is a stepwise process involving the gradual accumulation of advantageous mutations. If it was possible to invert the chirality of one surface molecule at a time, each of which would confer some resistance, then we might see such changes in natural organisms. But inverted chirality appears to be all-or-nothing: either a cell can make mirror proteins, and therefore other mirror macromolecules, or it cannot. There is simply no plausible series of stepwise changes through which a natural-chirality cell could build functional mirror proteins without severely damaging its ability to make normal proteins. Even small alterations to the genetic code alter the structure of all existing proteins, likely causing numerous fatal changes (Crick, 1968). As a result, the genetic code has remained remarkably conserved across living organisms (Ambrogelly *et al.*, 2007; Koonin & Novozhilov, 2017). Modifying an existing cell into a mirror cell would require many steps of similar or greater complexity to the incorporation of a novel amino acid, and therefore appears to be essentially impossible.

In principle, mirror life could arise independently of existing life, but there is also no compelling reason to think that mirror life would have arisen independently of ordinary life. Opinion varies as to whether the origin of life was an extremely rare event, almost inevitable, or something in between; the very limited evidence available does not allow strong conclusions to be drawn on this matter (Spiegel & Turner, 2012). If abiogenesis is rare, it would have been very unlikely for life to have arisen twice, and so mirror life probably never arose on Earth. Even if abiogenesis happened more than once on the early Earth, it is not clear that other living organisms would have had a close resemblance to modern life, mirrored or otherwise.

It thus appears most likely that nothing similar to a mirror bacterium ever existed on Earth. The absence of mirror life from nature provides no cause for reassurance with respect to synthetic mirror life.

spread at a similar speed (see Section 8.3). Human activity could spread invasive mirror bacteria even faster via airplanes, cargo ships, and road vehicles.

Mirror bacteria that escape containment would likely start with essentially no standing genetic variation. They would have no ability to acquire genes by horizontal transfer from existing bacteria, because the mirrored molecular machinery of a mirror bacterium could not read natural-chirality DNA. As they spread, they would encounter diverse environments they are poorly suited for, but mutation and natural selection would allow them to adapt (see Section 8.4). Invasive mirror bacteria could thus diversify into multiple lineages, specializing and evolving to colonize previously

inaccessible environments and infect novel hosts. Even if an escaped mirror bacterium could initially survive only under restricted conditions, it could subsequently evolve increased fitness and an expanded range.

An invasive mirror bacterium able to infect many multicellular hosts could plausibly drive most susceptible animal and plant species to extinction owing to repeated infections from abundant mirror bacterial populations in the external environment. The loss of many species simultaneously would be highly disruptive to natural ecosystems, particularly if ecologically critical species are susceptible. A mirror photoautotroph would be particularly disruptive to marine ecosystems because it would not require any chiral nutrients to thrive.

Creating biocontained mirror bacteria may be feasible, but securing them against misuse appears exceptionally challenging

If mirror bacteria are created, it would be relatively straightforward to engineer more robust and diverse mirror strains and species, which could pose serious risks to both human health and the environment. The creation of any mirror bacterium would thus create serious biosafety and biosecurity challenges. The technical feasibility of containment strategies to mitigate risks are discussed in Chapter 3.

Creating a "biocontained" mirror bacterium—that is, one incapable of replicating outside of a laboratory—should be possible. For instance, a metabolic auxotroph that lacked the biosynthetic pathways for pantothenate and biotin would be able to grow only if provided mirror pantothenate and mirror biotin, both of which are likely absent from nature. Given the potentially severe consequences of an escape, any biocontainment methods used with mirror bacteria would have to be extremely robust to all possible paths to escape, including failures arising from human error.

Most concerningly, even if robust biocontainment were possible, such measures could always be undone by malicious or reckless actors. Given the relative ease with which mirror bacteria could be engineered if created, engineering versions that could not be abused by malicious actors appears to be extraordinarily difficult.

Countermeasures could mitigate some harms, but could not prevent irreversible environmental damage

Chapter 5 describes potential medical countermeasures that could be used to protect humans from mirror bacteria. While most existing antibiotics are chiral and interact stereospecifically with their microbial targets, those antibiotics that are achiral would still prove harmful to mirror bacteria; novel "antimirror" compounds could also be identified and produced. The clinical efficacy of such antimirror compounds would be unclear until tested, and scaling production to sufficiently protect all humans would be a challenging, and perhaps impossible, undertaking. Other therapeutics such as conjugate vaccines, antibody therapy, and phage therapy appear to be technically feasible, but the practical challenges to developing such measures quickly and at scale would be considerable.

Protecting agricultural production against invasive mirror bacteria could require genetically engineering resistant crops or cultivating resistant microbes or simple plants as alternative food sources (see Section 7.4). Protecting most wild plants or animals through similar engineering would not be feasible. Biological countermeasures such as mirror phages, discussed in Section 8.6, could in principle reduce mirror bacterial populations after they become established. Like existing biocontrol measures, however, they would be unable to completely eliminate mirror bacteria from ecosystems, and so could not protect plants and animals from repeated exposure and infection. If an invasive mirror bacterium were inadvertently or maliciously released into the environment, there does not appear to be any realistic prospect for averting irreversible and widespread environmental harm.
## **Chapter 2: Pathways to Mirror Life**

Katarzyna P. Adamala<sup>1</sup>, Kevin M. Esvelt<sup>2</sup>, John I. Glass<sup>3</sup>, Sebastian B. Oehm<sup>3,4</sup>, Hiroaki Suga<sup>5</sup>, Jack W. Szostak<sup>6</sup>

Ever since Louis Pasteur recognized the chiral nature of life, scientists have speculated about the possibility of mirror-image life forms. Mirror-image organisms would be expected to share the properties of their natural-chirality counterparts, akin to the similarities between left and right hands. However, they could not emerge by natural evolution from existing natural-chirality life: this would require the simultaneous inversion of the chirality of all mirror biomolecules, which is incompatible with the gradual nature of evolution (see Box 1.3). The construction of entirely new forms of life in the laboratory has long seemed like a distant prospect, but advances in synthetic chemistry and synthetic biology make the human-led creation of mirror-image organisms increasingly realistic in the coming decades.

In this chapter, we discuss key enabling technologies and potential paths through which a mirror bacterium might be created based on our current understanding. We do not expect that the creation of a mirror bacterium, if further pursued, would follow exactly one of the outlined strategies, but the analysis here may be illustrative to understand the overall feasibility, as well as the challenges and strategies that would be involved in a future effort. As we will show, plausible paths to the creation of a mirror bacterium exist, but numerous technical hurdles remain to be overcome. Overall, the creation of a mirror bacterium will require extensive research. We estimate that if substantial resources were invested in a concerted effort, the creation of a mirror bacterium might still be 10 years away; and if research continues on its current trajectory, mirror bacteria might be created in the next 15 to 30 years.

Mirror bacteria could present novel and severe biosafety and biosecurity challenges, as outlined in the other chapters of this report. By default, many of the bottlenecks toward the creation of mirror cells will likely be addressed by scientists attempting to solve related problems, even if these researchers have no explicit interest in creating mirror bacteria. This chapter aims to assess the feasibility of the creation of mirror bacteria in the wider context of ongoing scientific advances and highlight the technical hurdles, which might serve as an indicator of progress toward mirror life. Discussion of governance options is out of the scope of this report.

*Authors are listed in alphabetical order. Affiliations*: <sup>1</sup>Department of Genetics, Cell Biology and Development, University of Minnesota, Minneapolis, Minnesota, USA; <sup>2</sup>Media Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, USA; <sup>3</sup>J. Craig Venter Institute, La Jolla, California, USA; <sup>4</sup>Trinity College Cambridge, University of Cambridge, UK; <sup>5</sup>Department of Chemistry, Graduate School of Science, The University of Tokyo, Tokyo, Japan; <sup>6</sup>Howard Hughes Medical Institute, Department of Chemistry, The University of Chicago, Chicago, Illinois, USA

Section 2.1 provides an overview of recent advances in the chemical synthesis of mirror biomolecules. Chemical synthesis, unlike biological synthesis, is equally well suited to create natural-chirality or mirror-image biomolecules such as nucleic acids and proteins. Advances in chemical synthesis have recently enabled the synthesis of mirror-image DNA and RNA at the kilobase-scale, and the synthesis of large mirror-image proteins up to 100 kDa; in principle, this is sufficient to create most components of a mirror bacterium. Ongoing research on improving these methods is expected to lower the cost and complexity further, making mirror biomolecule synthesis more accessible.

Section 2.2 considers how, leveraging the advances in chemical synthesis, a mirror bacterium could be created "bottom-up". While it is not yet possible to create living cells from non-living components of any chirality, bottom-up synthetic cell research is rapidly advancing, and a subset of the methods could be adapted to create mirror cells in the future. A plausible strategy to create a mirror bacterium bottom-up would involve creating a mirror genome and mirror transcription and translation machinery by chemical synthesis and then encapsulating them in a synthetic mirror membrane. Under the right conditions, the mirror transcription and translation machinery might then be used to express a mirror transcriptome and proteome from the mirror genome, to "boot up" a self-replicating mirror cell.

Section 2.3 considers an alternative path to the creation of a mirror cell, the stepwise conversion of a natural-chirality bacterium into a mirror bacterium. Extensive work in the field of genetic code expansion and reprogramming is enabling researchers to make proteins *in vivo* with non-canonical amino acids beyond the canonical twenty. Building on this work, it might be possible to create a "crossover" translation pathway in which natural-chirality bacteria are able to produce mirror proteins using natural-chirality machinery. The crossover translation system could be used to create a self-sustaining mirror central dogma within the cell. Finally, a synthetic mirror genome could be delivered to this mixed-chirality cell and the natural-chirality genome removed, to create a fully mirrored bacterium. While this pathway might appear more challenging than the approach discussed in Section 2.2, it might be differentially accelerated by advances in synthetic biology and translational engineering over the coming decades.

Section 2.4 discusses synergies between these two conceptual approaches and additionally considers more speculative pathways. Scientific progress is hard to predict, and as different areas of science advance, alternative pathways might become increasingly plausible. For example, the development of abiological artificial cells could provide the basis for mirror cell research in the future. Advances in computational or experimental methods that would greatly improve our ability to create designer enzymes could enable alternative pathways to mirror life.

Section 2.5 synthesizes the major bottlenecks along each pathway and discusses the overall feasibility of creating mirror bacteria.

## 2.1 Advances in chemistry permit the synthesis of mirror biomolecules with diverse applications

Historically, most biological macromolecules such as proteins or nucleic acids were derived from living organisms. As a result, only their natural-chirality versions have been accessible for applications in basic science and biotechnology. Advances in chemical synthesis permit the creation of proteins and nucleic acids of either chirality, enabling new applications of these mirror biomolecules.

## Progress in chemical peptide and protein synthesis is making mirror-image peptides and proteins more accessible

The creation of a mirror bacterium would require the synthesis of a large number of mirror-image proteins. Peptides of both chiralities are routinely made by solid-phase peptide synthesis (SPPS) (Behrendt *et al.*, 2016). In this process, the starting amino acid is covalently attached to a solid polymer support through its carboxyl group and additional amino acids are added in a stepwise fashion. This works equally well for natural-chirality and mirror-image polypeptides and typically permits the synthesis of polypeptides of 30–60 amino acids, although longer polypeptides can sometimes be made under optimized conditions. Building on this method, the first synthesis of a mirror-image protein was achieved by Kent and colleagues, who in 1992 created a 99-amino-acid mirror-image HIV-1 protease entirely by SPPS and demonstrated that—as expected—the mirror-image protease was capable of cleaving a mirror-image peptide substrate but not the natural-chirality substrate (Milton *et al.*, 1992). Modern SPPS can be automated, and improving methods enable the synthesis of increasingly challenging and long polypeptides. For example, the recent development of fast-flow automated peptide synthesis devices has permitted the production of small proteins of up to 164 amino acids, including several small mirror-image proteins, within a few days (Callahan *et al.*, 2024; Hartrampf *et al.*, 2020).

The development of native chemical ligation has substantially increased the size and complexity of proteins that can be made by total chemical synthesis (Kulkarni *et al.*, 2018). This approach involves synthesizing several peptide fragments of 30–60 amino acid length by SPPS and then ligating these peptides together to form the full-length protein (X. Lin *et al.*, 2023). Successes include the total synthesis of the mirror image of the 312-amino acid 4-hydroxy-tetrahydrodipicolinate synthase from *Escherichia coli* (DapA) (Weinstock *et al.*, 2014) and the 352-amino acid DNA polymerase IV from *Sulfolobus solfataricus* (Dpo4) (Pech *et al.*, 2017; W. Xu *et al.*, 2017).

Recently, the chemical synthesis of the mirror image of the 775-amino acid DNA polymerase from *Pyrococcus furiosus (Pfu* DNA Pol) (Fan *et al.*, 2021), and the 883-amino acid T7 RNA polymerase (T7 RNAP) (Y. Xu & Zhu, 2022) have also been accomplished. Since these proteins are too large to be made even with native chemical ligation, the authors performed rational engineering of the enzymes to generate split proteins. The synthesis of *Pfu* DNA Pol was achieved by splitting the enzyme into a 467-amino acid and a 308-amino acid subunit. T7 RNAP was made by splitting the enzyme into a 363-amino acid, a 238-amino acid, and a 282-amino acid subunit. The authors additionally introduced point mutations to remove hydrophobic residues to increase the solubility of

### Box 2.1: Applications of mirror peptides and proteins

Peptides are promising **therapeutics** due to their ability to bind drug targets with high affinity. However, natural-chirality peptides suffer from comparatively low biostability as they can be degraded by proteases (enzymes that degrade peptides and proteins) that are abundant in most living organisms and play important roles in the immune system (Muttenthaler *et al.*, 2021). Scientists are therefore exploring the use of additional building blocks, including D-amino acids, to modify peptides in order to create biostable peptide-like drugs (Muttenthaler *et al.*, 2021). In a classical approach, a peptide made of natural-chirality L-amino acids is first developed using standard drug discovery techniques, and then modified to include non-canonical building blocks, including D-amino acids, to improve its stability while retaining its biological activity (Imanishi *et al.*, 2021; Muttenthaler *et al.*, 2021). For example, etelcalcetide, a drug for the treatment of secondary hyperparathyroidism, consists of a 7-amino acid D-peptide linked to a single L-amino acid (Blair, 2016).

Mirror-image proteins might also be used in drug discovery to aid the development of stable fully mirrored peptide drug candidates. Scientists can use standard drug discovery techniques to find natural-chirality peptides that bind to **mirror-image versions of a drug target**. By chiral symmetry, the mirror-image version of the peptide will then bind to the natural-chirality version of the target in patients, but is expected to have a significantly improved half-life because it would be resistant to protease degradation and immune recognition (Callahan *et al.*, 2024; Harrison *et al.*, 2023; Muttenthaler *et al.*, 2021; Welch *et al.*, 2010).

Several other applications of mirror-image proteins have been explored to date. For example, mirror-image proteins have been used to enable **protein structure determination** by X-ray crystallography. Chiral proteins can only form crystals in chiral space groups, whereas racemic mixtures can form centrosymmetric crystals, which significantly simplifies phase determination and thereby aids resolution of the protein structure (Mackay, 1989). This approach has been successfully used to determine a number of protein structures (Avital-Shmilovici *et al.*, 2013; Banigan *et al.*, 2010; Dang *et al.*, 2016; Harrison *et al.*, 2023; Huang *et al.*, 2016; Hung *et al.*, 1999; Luisier *et al.*, 2010; Mandal, Pentelute, Tereshko, Kossiakoff *et al.*, 2009; Mandal, Pentelute, Tereshko, Thammavongsa *et al.*, 2009; Payne *et al.*, 2021; Pentelute *et al.*, 2010; Teng *et al.*, 2021; Yeung *et al.*, 2016).

### Chapter 2: Pathways to Mirror Life



#### Figure 2.1: Chemical synthesis of mirror proteins

**A.** Solid-phase peptide synthesis can use L- or D-amino acid building blocks to produce L- or D-peptides. The process can be automated, and typically yields peptides of 30–60 amino acids in length, although longer chains can be achieved under optimized conditions. The current record is 164 amino acids (Hartrampf *et al.*, 2020). **B.** Native chemical ligation connects peptides irrespective of chirality to synthesize longer polypeptide chains. Applying this method in a hierarchical fashion, small to medium-sized proteins up to 400 amino acids can be produced. **C.** Protein engineering can facilitate the total chemical synthesis of functional equivalents of large proteins. In this approach, a target protein is split into multiple subunits, and mutations are introduced that facilitate synthesis while not affecting protein function. The subunits can then be made by a combination of solid-phase peptide synthesis and native chemical ligation, and the full protein yielded by co-folding the subunits. The current record is a triply split 883-amino acid protein (Y. Xu & Zhu, 2022).

the peptide segments produced in SPPS, and to introduce additional ligation sites to facilitate synthesis (Figure 2.1). The functionality of both split proteins was demonstrated, which enabled the transcription of kilobase mirror-image RNA, an important step toward the creation of mirror life.

As we will discuss in Section 2.2, the creation of a mirror bacterium through a bottom-up pathway would likely require the synthesis of at least 100 mirror proteins to create mirror ribosomes and other mirror transcription and translation machinery. Modern chemical protein synthesis technologies are in principle sufficiently advanced to synthesize most of the mirror proteins required. However, the cost and effort involved in synthesizing the large number of proteins required to create a cell bottom-up would be large. Scientists are continuously advancing methods for total protein synthesis that can be applied for mirror-image protein synthesis, including automated fast-flow peptide synthesis devices (Callahan *et al.*, 2024) or improved ligation methods (Agouridas *et al.*, 2019). Advances in protein engineering might likewise increase the ease and accessibility of split protein and mutagenesis approaches for chemical protein synthesis in the future (Dolberg *et al.*, 2021; Freschlin *et al.*, 2022; Lovelock *et al.*, 2022).

Alternatively, mirror protein synthesis might in the future be carried out using biological machinery, for example using in vitro ribosomal protein synthesis systems (Katoh & Suga, 2022). Unlike chemical approaches, biological machinery is inherently chirality-specific, so the production of mirror proteins using natural-chirality biological machinery is not feasible with current technology. Extensive academic research has explored re-engineering natural-chirality ribosomal protein synthesis systems to use building blocks other than canonical L-amino acids, including mirror p-amino acids (Katoh & Suga, 2022). A "crossover" translation system would use natural-chirality biological machinery to translate natural-chirality mRNAs into mirror-image proteins by polymerizing D-amino acids instead of natural L-amino acids. Notably, tRNAs charged with all 20 p-amino acids are available through in vitro flexizyme technology (Achenbach et al., 2015; Fujino et al., 2013). This crossover translation system is not necessary for the bottom-up creation of mirror cells if all the required components can be made by chemical synthesis. However, if an efficient and scalable biological mirror protein synthesis system were created in the future, it would be possible to rapidly synthesize most mirror proteins of interest. There is great interest in re-engineered protein translation systems for medical applications, hence progress in this area is expected even absent explicit interest in building mirror proteins (Katoh & Suga, 2022).

### Mirror-image nucleic acids can be made chemically using established methods

Mirror-image DNA and RNA would be key components of a mirror cell. Current research, however, mainly focuses on potential applications as therapeutics or in biotechnology (see Box 2.2). DNA and RNA are chiral because the p-deoxyribose or p-ribose sugar backbone contains several chiral centers; the nucleobases are achiral. Methods to chemically synthesize the mirror-image L-ribose and corresponding mirror-image L-nucleotide building blocks of DNA and RNA have been available since the 1960s. Natural-chirality p-DNA oligonucleotides of up to 200 nucleotides and p-RNA oligos of up to 70 nucleotides are routinely produced by chemical solid-phase synthesis, a process that is highly automated and performed at scale by commercial DNA synthesis providers, and longer DNA or RNA oligos can often be produced, although at lower yields (Hoose *et al.*, 2023). Adapting this chemical method for mirror DNA or RNA synthesis is straightforward. Several companies offer mirror DNA oligos commercially, although at higher prices due to the cost of mirror nucleotide building blocks (Biomers, n.d.; Glen Research, n.d.). Laboratories with DNA synthesis machines can also purchase mirror nucleotides directly and use them on existing machines to synthesize mirror oligonucleotides.

DNA oligos are commonly assembled into kilobase-length DNA pieces using enzymes such as DNA polymerase and DNA ligase (Hoose *et al.*, 2023). RNA is made at kilobase length by transcription from a DNA template using RNA polymerase (Figure 2.2). The advances in chemical protein synthesis described in the previous subsection have recently enabled researchers to create mirror versions of these enzymes (Fan *et al.*, 2021; Weidmann *et al.*, 2019), permitting the assembly and

#### Box 2.2: Applications of mirror nucleic acids

DNA is commonly synthesized for recombinant DNA technology, but nucleic acids have numerous additional applications in basic science, biotechnology, and medicine. A drawback of DNA and RNA in these applications is their limited biostability, as they are recognized and degraded by biological processes. Mirror DNA and RNA retain properties of DNA and RNA, including the ability to form predictable structures determined by base-pairing, but are more stable as they are not susceptible to degradation by nucleases (enzymes which break down DNA and RNA) which are highly abundant in the environment.

These results led to interest in using mirror oligonucleotides as aptamer therapeutics and in other applications. **Aptamers** are oligonucleotides that can bind with high affinity to some biological molecule, e.g., a drug target. Aptamers can be discovered rapidly through directed evolution (Tuerk & Gold, 1990). However, similar to natural-chirality peptides, natural-chirality oligonucleotides are generally too unstable in physiological environments to be useful as aptamer drugs. In contrast, mirror-image aptamers—termed Spiegelmers by TME Pharma, the company developing them—could retain the binding properties but exhibit greater biostability and thus become useful drugs. Two Spiegelmers have reached early clinical testing as of 2024 (Kaur *et al.*, 2018).

Mirror oligonucleotides might have diverse applications beyond aptamer therapeutics. For example, **molecular beacon** or **riboswitch** nucleic acids have been explored as biosensors (Seeman & Sleiman, 2017), and mirror-image versions of these can recapitulate their key properties while being more stable. Mirror image DNA nanostructures are being explored as drug delivery systems due to their greater resistance to serum host nucleases (Kim *et al.*, 2016). They could also serve as interesting materials in **DNA nanotechnology** due to their ability to form predictable shapes with other mirror nucleic acids without interfering with natural-chirality nucleic acid assemblies (C. Lin *et al.*, 2009).

accurate replication of kilobase-length mirror DNA, and the synthesis of kilobase-length mirror RNA from a corresponding mirror DNA template (Y. Xu & Zhu, 2022).

While synthetic mirror enzymes are currently available in only a handful of laboratories, current methods could in principle generate the mirror-image DNA genes required to assemble a mirror bacterial genome, each of which could be transcribed into mirror mRNAs, tRNAs, and rRNAs. However, assembling these genes into complete mirror chromosomes would be a major challenge, which will be discussed in Section 2.2.



### Figure 2.2: Synthesis of kilobase-length natural-chirality and mirror DNA and RNA

**A.** Synthesis of oligonucleotides from nucleotide building blocks by solid-phase synthesis can use D- or L-nucleotide building blocks to produce D- or L-oligonucleotides. This process is usually highly automated and typically yields DNA oligonucleotides up to 200 bp in length or RNA oligonucleotides up to 70 nucleotides in length. **B.** *In vitro* enzymatic reactions such as polymerase chain reaction and related methods can be used to assemble D-nucleotides into kilobase-length, double-stranded D-DNA constructs such as plasmids or gene fragments. Natural-chirality L-enzymes selectively act on natural-chirality D-DNA oligonucleotides, but these methods can be adapted for mirror L-DNA by using mirror-image D-proteins. **C.** *In vitro* transcription with a natural-chirality RNA polymerase can be used to use a mirror-image RNA polymerase to produce kilobase-length L-RNA from a kilobase-length L-DNA template.

## **2.2 Progress in synthetic biology could allow the assembly of a mirror bacterium from non-living mirror components**

The ability to create a mirror bacterial cell could arise from the convergence of two distinct ongoing research programs. The first is the ability to synthesize the mirror-image versions of the necessary components of a bacterial cell, as described in Section 2.1. The second is ongoing research to reconstitute a living natural-chirality bacterium from synthetic components, which we will discuss in this section. Once the ability to do this is refined using normal-chirality molecules, it would become possible to repeat this process with mirror image components to create a living mirror bacterium.

This "bottom-up" pathway could involve the following steps:

- 1. Synthesis of a mirror genome in vitro.
- 2. Creation of mirror ribosomes and a high-efficiency *in vitro* mirror protein synthesis system from mirror components.
- 3. Encapsulation of the mirror genome and mirror protein synthesis system into a membrane and "booting" of the mirror bacterium.

We begin by providing a brief overview of existing synthetic cell research, then consider how each of the above steps could be accomplished with mirror-image components in the future.

Research is progressing toward the creation of self-replicating cells from non-living components

Bottom-up synthetic biology research attempts to build lifelike self-replicating cells from simple, non-living components. These components could be similar to those used in natural life—that is, be made of biomolecules like DNA, lipids, and proteins—or could instead use alternative chemistries to recapitulate the key processes of living cells.

A key feature of all cells is the presence of a lipid cell membrane that separates the cell interior from the environment. Synthetic lipid-bound compartments or vesicles were first created in the 1960s (Bangham *et al.*, 1965; Chang, 1964), and diverse methods for the creation of vesicles have been developed since (Has & Sunthar, 2020). The lipid bilayer of natural cells consists in large part of phospholipids, and these can be used to create synthetic vesicles, but other lipid molecules can be used as well.

Gene expression, i.e., the synthesis of RNAs and proteins, is another key feature of living cells. Gene expression can be recapitulated in a test tube. For example, cell lysate maintains some transcription and translation activity if suitably prepared, and its productivity can be enhanced by providing an energy regeneration system. These so-called cell-free protein synthesis systems have found diverse applications for rapid prototyping, cell-free devices, rapid small-scale biomanufacturing, and beyond (Silverman *et al.*, 2020).

Transcription and translation can also be accomplished using a fully defined system, known as "protein synthesis using recombinant elements" (PURE), in which the RNAs and proteins that make up the transcription and translation machinery (e.g., the RNA polymerase, ribosome, aminoacyl-tRNA synthetases, tRNAs, and translation factors) have been individually prepared and purified (Shimizu *et al.*, 2001). This system has the advantage that its composition, unlike that of cell extract-based systems, is fully known and can be precisely controlled by including or omitting specific proteins or RNAs of interest. The drawback is that the PURE system is more expensive to prepare than lysate-based systems as it requires the preparation of a large number of recombinant proteins. Current PURE preparations are also generally less efficient at protein synthesis than extract-derived systems (Gregorio *et al.*, 2019; Li *et al.*, 2017).

DNA and transcription-translation systems can be encapsulated in lipid vesicles, creating artificial cells that resemble natural cells in that they are membrane-bound compartments capable of gene expression. Interestingly, membrane encapsulation can improve the protein synthesis efficiency of lysate-based systems (Garenne *et al.*, 2021; Noireaux & Libchaber, 2004). Encapsulated transcription-translation systems can be used to produce diverse proteins, including metabolic enzymes or membrane proteins that are inserted into the vesicle membrane (Kuruma & Ueda, 2015).

The full central dogma—DNA replication, transcription, and translation—has recently been reconstituted within an artificial compartment (van Nies *et al.*, 2018). Researchers expressed the DNA replication machinery from  $\Phi$ 29 phage within PURE-containing lipid vesicles and showed that the encoding DNA could subsequently be replicated. The advantage of a  $\Phi$ 29 phage replication system is its comparative simplicity; but recapitulating the DNA replication machinery from bacteria such as *E. coli* in a lipid vesicle would likely be of interest to researchers in the future, as it might

permit the replication of synthetic genomes greater than the ~150 kb size limit of the  $\Phi$ 29 replication machinery.

Ongoing research aims to encode the PURE components within the encapsulated genome to create a self-sustaining central dogma. In one attempt, a 116 kb genome encompassing the full set of *E. coli* central dogma machinery was encapsulated in an artificial compartment; however, it was not possible to achieve PURE synthesis at self-sustaining levels, or ribosome biogenesis (Libicher *et al.*, 2020). Future work on enhancing PURE synthesis efficiency, and work aimed at improving *in vitro* ribosome biogenesis, might allow scientists to accomplish this milestone in the future (Hagino & Ichihashi, 2023; Lavickova *et al.*, 2020; Wei & Endy, 2021).

Alternatively, researchers might attempt to encapsulate a full bacterial genome within a vesicle containing transcription and translation machinery in order to "boot" a live bacterium. In one attempt, researchers encapsulated the *E. coli* genome and a lysate-derived transcription-translation solution in a synthetic lipid vesicle and observed the expression of a reporter gene from the genome (Deyama *et al.*, 2021). However, achieving a full "boot", i.e., the creation of a self-replicating bacterium, might require better transcription-translation efficiencies, encapsulation methods that ensure the genome is not damaged by shearing forces, and the use of native phospholipid membrane (unlike the artificial lipid membrane used in Deyama *et al.*). As noted above, improved transcription-translation systems now produce almost as much protein as the input protein required to create the transcription-translation solution (Garenne *et al.*, 2021), and such high-efficiency transcription-translation systems have enabled the boot of a 40 kb T7 phage genome within a lipid vesicle to generate functional T7 phage (Garenne *et al.*, 2021; Shin *et al.*, 2012). The boot of a live bacterium from a cell-derived genome, transcription-translation system, and membrane thus appears increasingly feasible—several of the authors are pursuing optimized strategies and are aware of colleagues doing the same.

Bottom-up synthetic cell research is progressing rapidly, and ever more complex behaviors of cells, including movement, sensing and response to external stimuli, cell-cell communication, synthetic organelles, and multicellular assemblies have been reconstituted with synthetic cells (Rothschild *et al.*, 2024). This research is already leading to interesting applications (Adamala *et al.*, 2024). Liposomes play a key role in the delivery of mRNA therapeutics and gene therapies, and scientists are exploring artificial liposomes as therapeutic delivery devices. While most hallmarks of life have been reconstituted in artificial cell-like structures in isolation (Gaut & Adamala, 2021), a frontier of current research is to integrate the different systems in a single self-replicating cell to create an artificial compartment that exhibits all features of life—that is, to create life in the test tube (Kriebisch *et al.*, 2024).

The synthesis of a mirror genome *in vitro* might be attainable by building on existing technology

To create a mirror bacterium, researchers would need to synthesize a mirror DNA genome. Chemical DNA oligo synthesis and enzymatic assembly, as discussed in Section 2.1, permits the synthesis of mirror DNA up to several kilobases in length. The synthesis of a mirror genome would require

large-scale synthesis of the genome in kilobase-length pieces by these methods, and their subsequent assembly into a megabase-length genome. The exact length would depend on the specific mirror bacterium to be created: the synthesis of a minimized *Mycoplasma mycoides* genome, the smallest genome known to be viable, would require a 0.53 megabase (Mb) genome (Hutchison *et al.*, 2016) whereas the genome of *E. coli* strain K-12 MDS42, a common laboratory strain, would require 3.98 Mb (Pósfai *et al.*, 2006).

Scientists have successfully synthesized a small number of natural-chirality bacterial genomes, but each synthesis still involves major effort and the *in vivo* replication of increasingly large pieces within bacteria or yeast (Fredens *et al.*, 2019; Gibson *et al.*, 2010). The first challenge in adapting these methods for the synthesis of a mirror genome is the cost of synthesizing the required DNA in kilobase-length pieces, which can be up to \$1 million USD from commercial DNA synthesis providers for a large genome; the cost to synthesize an equivalent amount of mirror DNA could be orders of magnitude higher.

The second challenge is the assembly of the kilobase-length pieces into a megabase-scale genome. Larger-scale DNA assembly into 50–100 kb pieces, and their subsequent assembly into Mb-length pieces, is typically done in living yeast or *E. coli* (Fredens *et al.*, 2019; Gibson *et al.*, 2010). Such methods could not be adapted to assemble mirror DNA unless mirror organisms were already available. However, DNA pieces up to 900 kb in length can also be assembled *in vitro* using synthetic enzymes (Gibson *et al.*, 2009). It appears likely that extensions of this or other methods could be used to assemble Mb-scale mirror DNA using mirror enzymes, although fidelity and yield may need to be optimized. Alternatively, it may be possible to boot synthetic cells with genomes composed of multiple small chromosomes similar to some existing bacterial strains (diCenzo & Finan, 2017; Itaya & Tanaka, 1997; K. Wang *et al.*, 2019; Yoneji *et al.*, 2021). Including fewer replication and assembly steps could reduce the opportunity for errors that could disrupt essential genes and potentially ease the process of encapsulating the DNA within a membrane.

Current *in vitro* DNA synthesis and assembly methods exhibit a low but non-negligible error rate, and genome synthesis projects typically use clonal amplification *in vivo* to propagate and verify assembly intermediates. However, methods such as digital PCR (Vogelstein & Kinzler, 1999) or clonal amplification in synthetic vesicles (Abil *et al.*, 2023) could serve a similar purpose for an *in vitro* effort, and gene synthesis researchers have substantially improved *in vitro* error correction methods (Hoose *et al.*, 2023; Sidore *et al.*, 2020; J. Zhang *et al.*, 2020).

Overall, while the creation of a synthetic genome entirely *in vitro* would require the adaptation and improvement of a number of existing assembly and error-correction methods to work with mirror DNA, the synthesis of which would be extremely expensive at current prices, this step does not appear to be an insurmountable bottleneck to the bottom-up creation of a mirror bacterium.

### Research is progressing toward mirror ribosomes and a mirror protein synthesis system

Bacteria contain thousands of RNAs and proteins. However, it may not be necessary to create all cellular components from scratch to assemble a living cell. RNA synthesis can be accomplished by a single protein: RNA polymerase. In the PURE system mentioned above (Shimizu *et al.*, 2001),

protein synthesis can be accomplished with as few as 36 proteins, ribosomes (comprising several RNAs and about 50 proteins), tRNAs, and a defined chemical energy mix.

If scientists were to succeed at creating sufficiently efficient mirror transcription and translation systems from a handful of mirror-image RNAs and mirror proteins, and combine them with a mirror genome, they could synthesize directly from the mirror genome the thousands of proteins and RNAs required for life. The resulting mirror proteins could then synthesize mirror lipids, carbohydrates, a cell wall, and metabolites. Hence, to generate mirror proteins outside of a cell, researchers would at least need to synthesize the mirror components of the minimal PURE translation system: mirror aminoacyl-tRNA synthetases, mirror translation factors, mirror tRNAs, and a mirror ribosome (Shimizu *et al.*, 2001).

Ribosomes are complex structures consisting of several large rRNAs and many ribosomal proteins (r-proteins). For example, the *E. coli* ribosome consists of three rRNAs and 54 r-proteins (Figure 2.3). In the natural-chirality PURE system, ribosomes are typically purified from cells, which would not be possible for mirror ribosomes. Instead, mirror-image r-proteins and unmodified mirror rRNAs would need to be created chemically, and subsequently assembled into functional mirror-image ribosomes *in vitro*. It would be important to assemble intact ribosomes with productivities similar to native ribosomes, as the efficiency of the translation system is expected to be critical to successfully booting up a bacterium.

The assembly of small ribosomal subunits entirely from *in vitro*-generated rRNA and protein has not yet been reported. However, small ribosomal subunits made from *in vitro*-generated rRNA and *in vivo*-derived r-proteins are functional (Krzyzosiak *et al.*, 1987), suggesting that small subunit rRNA modifications are not essential for function. Making mirror small ribosomal subunits therefore seems feasible after some optimization. The construction of large ribosomal subunits from *in vitro*-generated rRNA and protein has likewise not been achieved. Functional ribosomal large subunits can be made from *in vitro*-generated rRNA and *in vivo*-derived purified r-proteins, but at substantially reduced activity (Aoyama *et al.*, 2022; Semrad & Green, 2002). Ribosomal large subunits with greater activity can be generated in extract-based systems, which may contain enzymes that supply at least some of the post-transcriptional modifications and other proteins that aid ribosome biogenesis (Jewett *et al.*, 2013). Identifying these rRNA modification enzymes and ribosome biogenesis factors might therefore enable the *in vitro* synthesis of large ribosomal subunits. Making mirror large ribosomal subunits that support protein synthesis with high efficiency thus seems like a difficult but tractable technical step.

Native tRNAs, like rRNAs, are extensively modified by post-transcriptional modification enzymes in bacteria like *E. coli*, and *in vitro*-produced tRNAs and rRNAs lack such modifications. Experimental work shows that these modifications are essential to achieve high translation activity for the tRNAs corresponding to 3 out of 20 amino acids (Cui *et al.*, 2015) so creating mirror-image versions of these tRNA-modifying enzymes might be necessary to achieve high mirror PURE productivity. However, it has been shown that a complete reconstitution of *in vitro*-produced tRNAs lacking chemical modifications could function in the PURE system when their concentrations are optimized. Therefore, synthetic mirror tRNAs could function in the mirror PURE system (Iwane *et al.*, 2016; Iwane *et al.*, 2018). As noted in a previous section, the PURE system can produce most proteins,

### Chapter 2: Pathways to Mirror Life



#### Figure 2.3: Components of ribosomes and the PURE system

**A.** Cryo-EM structure of the *E. coli* ribosome (PDB 5AFI). The *E. coli* ribosome consists of two major subunits, the 50S large subunit and the 30S small subunit. The 50S subunit consists of two rRNAs (23S, light blue, and 5S, yellow) and 34 r-proteins (blue and dark blue). The small subunit consists of one rRNA (16S, orange), and 21 r-proteins (red and dark red). **B.** Components of the PURE system. DNA encoding a target protein is added to the PURE system, which consists of all proteins, amino acids, nucleotides, and non-coding RNAs necessary to transcribe it into mRNA and translate the mRNA into the desired polypeptide.

including membrane proteins (Kuruma & Ueda, 2015), but PURE synthesis generally yields lower amounts of protein product than a comparable amount of cellular protein synthesis machinery, and below self-sustaining levels. Therefore, the synthesis capacity of current PURE systems is unlikely to be sufficient to produce a cellular proteome as part of an attempt to "boot up" a living cell from a synthetic genome.

Several research groups are actively working to improve the PURE system's efficiency, with the aim of developing a PURE system with the capacity for self-regeneration (Hagino & Ichihashi, 2023;

Lavickova *et al.*, 2020; Wei & Endy, 2021). Challenges for the original PURE formulation might include lack of important translation factors, chaperones and supporting proteins, lack of homeostasis, and non-native protein concentrations (about 10-fold lower than in the cytosol). Research approaches to increase PURE's productivity therefore include improved systems to provide energy during the reaction and remove byproducts, altering the concentration of the components, and adding additional protein factors (de Maddalena *et al.*, 2016; Kazuta *et al.*, 2014; Lavickova *et al.*, 2020; Li *et al.*, 2017). Progress in this field would likely be required to permit the booting experiments described in the following subsection. However, it is difficult to determine precisely what level of PURE efficiency would be required. A fully self-regenerating PURE system might not be necessary; during the boot, the PURE system might produce the missing translation factors, chaperones, and other supportive proteins.

## Encapsulating the mirror genome and mirror protein synthesis system in a membrane could "boot up" the mirror bacterium

If mirror proteins and nucleic acids were successfully used to create a mirror genome and a mirror protein synthesis system *in vitro*, the final step to creating a mirror bacterium would involve combining those components in a lipid compartment and attempting to boot up a living, self-replicating cell.

The lipids in the bacterial membrane are chiral. Bacterial membranes consist primarily of phospholipids with a chiral L-glycerol-3-phosphate backbone. Notably, archaeal membranes consist primarily of phospholipids with a D-glycerol-3-phosphate (more commonly known as L-glycerol-1-phosphate) backbone alongside other differences, and experiments have shown that *E. coli* can grow without fitness defect with 20–30% archaeal phospholipid content in its membrane (Caforio *et al.*, 2018). In any case, mirror lipids can be synthesized chemically, and so are unlikely to be a bottleneck for mirror bacterium assembly.

The bacterial cell wall is likewise chiral, and the chemical construction of its mirror-image appears difficult. However, the construction of a mirror cell wall may not be necessary. Some bacterial species, most notably *Mycoplasma*, naturally lack a cell wall. *E. coli* has a cell wall, but experiments have shown that *E. coli* can replicate without a cell wall and subsequently regenerate it (Petrovic Fabijan *et al.*, 2022; Tabata *et al.*, 2019). Therefore, scientists could plausibly create mirror bacteria without chemically constructing a cell wall.

Scientists have not yet created a living cell of any chirality in a test tube from non-living components. As discussed above, a first milestone might be the assembly of a living natural-chirality cell in a test tube from cell-derived components (Figure 2.4A). In this experiment, the genome, cell lysate/cytosol, and cell membrane would first be harvested from living bacteria, and then recombined to create a living cell. This experiment could establish the core methods for combining a genome, protein synthesis system, and membrane in a way suitable to boot a live bacterium. It is plausible that such an experiment will succeed in the next few years with moderate improvements to existing experimental protocols, for instance by improving the efficiency of extract-based transcription-translation systems and gentle encapsulation methods that would prevent shearing of



#### Figure 2.4: Milestones towards the creation of a mirror bacterium

**A.** Schematic depicting the boot of a living natural-chirality bacterium from non-living, cell-derived components. In this experiment, the bacterial genome, lysate, and membrane would be harvested from an existing bacterium and then recombined to create a new living bacterium. **B.** Schematic depicting the boot of a living natural-chirality bacterium from non-living, fully synthetic components. In this experiment, the genome, lysate, and membrane from A would be replaced by synthetic components made without the help of existing cells, such as a synthetic genome, the PURE system, and synthetic lipids. **C.** Schematic depicting the boot of a mirror bacterium from non-living, fully synthetic mirror components. In this experiment, mirror versions of the components in B would be made and then combined using the protocol established in B to boot a mirror bacterium.

the genome. However, it may be necessary to extensively optimize the preparation of the individual components, the genome, and the booting-up protocol to create a living cell. The difficulty of this step is hard to predict, but the necessary technologies for such an attempt are available to scientists today.

Each of the three components in the cell reconstitution—genome, cell lysate, and membrane—would need to be replaced with a fully synthetic version (not one purified from other cells) to subsequently enable the creation of a mirror bacterium using the same protocol (Figure 2.4B).

Ideally, the starting PURE system would have sufficient synthesis capacity to generate the full proteome from the genome and thereby boot a self-replicating cell in a single step. As noted in the

previous section, the required synthesis efficiency is hard to determine, but it might be less than the synthesis capacity needed for self-regeneration, as there might be positive feedback in the expression of the genome due to the production of translation factors, chaperones, and other auxiliary systems that might help to facilitate further genome expression. However, should this fail, it may be possible to do a "partial" boot, recover the newly synthesized partial proteome (e.g., by removing the "used-up" PURE system using affinity tags and concentrating the new proteins), and add the new components to another booting attempt. Iterations of this process might generate a large fraction of the proteome, enhancing the PURE synthesis capacity until it crosses the threshold for positive feedback to make the entire proteome.

Positive feedback is not guaranteed, however. Some intermediate booting states could be toxic to the cell or the booting process (e.g., translation of proteases, nucleases, or membrane channels before their regulators). If the source of failure were found, it might be addressed by supplementing specific proteins to the original PURE mixture or making changes to the booting genome; the viability of the altered genome could be confirmed in separate experiments. Advances in creating cells with minimal genomes might be informative and minimal cells might be easier to boot, as their genomes are less complex (Haimovich *et al.*, 2015; Hutchison *et al.*, 2016). Alternatively, advanced genome design or powerful whole-cell models could be used to simplify or otherwise optimize the target genome and hence reduce the complexity of the booting task.

Overall, a boot from fully synthetic components would be a difficult technical step in the bottom-up pathway. However, as research in bottom-up synthetic biology progresses, the enabling methods will continue to improve. The exact timelines and challenges are hard to forecast with precision, but as synthetic cell research progresses rapidly, the natural-chirality version of this booting experiment might well succeed within the coming decade (Adamala *et al.*, 2024). Once this has been accomplished, it might "simply" be a matter of investing the resources required to produce mirror versions of all the components, as outlined in the previous sections, and then applying the booting protocol to produce a synthetic mirror bacterium (Figure 2.4C).

## **2.3 A natural-chirality bacterium might be converted into a mirror bacterium in a stepwise fashion**

As an alternative to the "bottom-up" path outlined in Section 2.2, a mirror bacterium could be created by a "top-down" pathway in which a living natural-chirality bacterium is converted into a mirror bacterium. This pathway might require more extensive technology development compared to the "bottom-up" pathway in Section 2.2, but these technologies are likely to benefit from greater investments for reasons unrelated to mirror life. A "top-down" strategy to create mirror bacteria might involve three steps:

- 1. Production of mirror-image proteins *in vivo* by creating a crossover pathway made of natural-chirality components.
- 2. Production of mirror-image proteins *in vivo* by creating an entirely mirror-image central dogma.

3. Delivery or assembly of a full mirror-image DNA genome *in vivo*, and removal of the natural-chirality genome, to create a mirror bacterium.

Such work would be able to build on existing research on genetic code reprogramming. We begin by outlining these existing approaches, and then consider each step in turn.

Genetic code reprogramming allows for the production of proteins with non-canonical amino acids

Research in genetic code expansion and reprogramming aims to create living cells that produce proteins made of an expanded set of amino acids. These non-canonical amino acids (ncAAs) are chemically diverse and thereby confer new properties to proteins. This includes a plethora of tools to study protein function and create enhanced proteins, such as ncAAs that crosslink to protein interaction partners, bioorthogonal reaction handles that allow site-specific protein modification, and mimics of post-translational modifications (Chin, 2017; Young & Schultz, 2018). More recent work has seen re-engineering of translation to create polymers made entirely of ncAAs, creating new classes of biopolymers that could serve as future materials or therapeutics (de la Torre & Chin, 2021).

The incorporation of an ncAA can be achieved by expressing a suitably engineered aminoacyl-tRNA synthetase/tRNA (aaRS/tRNA) pair in a cell (Mukai *et al.*, 2017; Young & Schultz, 2018). The aaRS charges the ncAA onto its cognate tRNA, which then can be used by the ribosome to incorporate the ncAA. In many experiments, the tRNA is engineered to decode the TAG stop codon, as TAG-decoding tRNAs compete efficiently with release factor 1 to permit site-specific incorporation of the ncAA at the TAG codon (Mukai *et al.*, 2017).

An important requirement is that the aaRS/tRNA pair needs to be orthogonal to the host translation machinery; that is, the additional aaRS must charge only its cognate tRNA and none of the host tRNAs, and the additional tRNA likewise must not be charged by any of the host aaRSs. Moreover, the aaRS active site must only charge the ncAA of interest onto its tRNA and not any of the canonical amino acids or other small molecules present in the cell.

Building on these advances, in recent years researchers have been exploring ways of using multiple ncAAs in parallel in the same cell, toward creating a translation pathway for the synthesis of genetically encoded non-canonical biopolymers (de la Torre & Chin, 2021). There are three principal requirements for such a pathway:

- 1. Blank codons that can be used to encode additional ncAAs need to be created.
- 2. Mutually orthogonal aaRS/tRNA pairs that can work in parallel with the host aaRS/tRNA pairs and each other and can charge multiple distinct ncAAs need to be found.
- 3. If ncAAs with complex side-chains or altered backbone structures are to be used, the ribosome and other translation factors might need to be engineered to enable their efficient polymerization.

Researchers have been progressing rapidly on these three challenges in recent years (de la Torre & Chin, 2021). For example, there are multiple pathways that might create blank codons in cells. The

genetic code consists of 64 three-letter codons encoding 21 signals, the 20 naturally occurring amino acids and a "stop" or termination signal. Many codons are synonymous; they encode the same signal. Synonymous recoding of a genome involves choosing a particular codon, and substituting all instances of that codon with a synonymous codon throughout the genome. The recoded codon no longer appears in the genome. Its cognate tRNA or release factor may be deleted, or reassigned to a new, potentially non-canonical, amino acid.

The falling cost of DNA synthesis and progress in synthetic genomics has recently enabled the synthesis of codon-compressed genomes of this kind. In 2013, the first such codon-compressed genome was developed by multisite editing (Lajoie *et al.*, 2013; H. H. Wang *et al.*, 2009). The TAG stop codon was replaced by synonymous alternatives genome-wide, which allowed the deletion of release factor 1 and hence reassignment of TAG to ncAAs at enhanced efficiency (Lajoie *et al.*, 2013). Subsequently, researchers created an *E. coli* with a fully synthetic genome containing just 61 codons, in which TAG as well as TCG and TCA, two of the six serine codons, were removed and substituted with synonymous alternatives (Fredens *et al.*, 2019). This organism, dubbed Syn61, contains three blank codons that can be reassigned to three non-canonical amino acids and used to synthesize short non-canonical polymers (Robertson *et al.*, 2021).

Genome recoding is not necessary if blank codons can be created by a different pathway. In one strategy, scientists engineered a ribosome to read quadruplet (4-base) codons rather than canonical triplet (3-base) codons (Neumann *et al.*, 2010). In principle this provides up to 256 blank codons, and researchers have shown that up to four non-canonical amino acids can be used in parallel in response to four quadruplet codons (Dunkelmann *et al.*, 2021). However, quadruplet decoding suffers from competition with triplet decoding in these experiments, limiting the efficiency of ncAA incorporation. Combining recoding with quadruplet decoding can address this challenge (Chatterjee *et al.*, 2014). Codon context is another strategy that has been used to optimize quadruplet decoding (Costello *et al.*, 2024).

In an alternative approach, researchers have created an organism that uses an additional nucleotide pair alongside the A:T and C:G pair that makes up canonical DNA. The additional pair (dTPT3:dNaM) can be replicated as part of the DNA, transcribed into RNA, and used to encode protein synthesis in cells (Malyshev *et al.*, 2014; Y. Zhang *et al.*, 2017). Future progress in creating an expanded genetic alphabet that can be used efficiently to code for non-canonical amino acids might likewise provide a diverse range of blank codons in the future. In *in vitro* experiments, researchers have also considered the possibility of creating parallel genetic codes, where one type of ribosome translates the proteome using the canonical genetic code, and a different type of ribosome uses a separate tRNA pool with a different assignment between triplet codons and amino acids (Terasaka *et al.*, 2014).

The development of mutually orthogonal aaRS/tRNA pairs is challenging due to the numerous orthogonality requirements. Much work in the field has been driven by the discovery that *Methanocaldococcus jannaschii* tyrosine-tRNA synthetase (TyrRS)/tRNA pair and the pyrrolysyl-tRNA synthetase (PylRS)/tRNA pairs from *Methanosarcina mazei* and *Methanosarcina barkeri* fulfill these orthogonality requirements *in vivo* and that their active sites can be engineered to charge a diversity of ncAAs with large hydrophobic and aromatic side chains (Mukai *et al.*, 2017).

Nonetheless, several additional pairs beyond the *M. jannaschii* TyrRS/*M. mazei* and *M. barkeri*/PylRS pairs have recently been found by exploring the natural sequence diversity of synthetases across the kingdoms of life. Up to five of these pairs have been found to be mutually orthogonal (Beattie *et al.*, 2023). Most pairs are derived from tyrosine and pyrrolysine synthetases, and hence available ncAAs mostly have large hydrophobic or aromatic side chains and a canonical  $\alpha$ -L-amino acid backbone. However, further effort at discovering more diverse aaRS/tRNA pairs and developing improved methods for their engineering will likely increase the number and diversity of ncAAs that can be used in coming years.

Most approaches in genetic code expansion rely on the promiscuity of the ribosome. However, while the ribosome is generally permissive to different side chains, the efficiency of ncAAs with altered backbone structures (such as N-alkyl, D-, or  $\beta$ -amino acids) is generally reduced (de la Torre & Chin, 2021). Ribosome engineering is challenging due to the central role of ribosomes in metabolism, and many active site mutations are lethal. However, researchers have laid the foundation for ribosome engineering by creating so-called orthogonal ribosomes (Orelle *et al.*, 2015; Rackham & Chin, 2005; Schmied *et al.*, 2018). These are copies of the ribosome that are isolated from host translation and translate their own set of orthogonal mRNAs, which are not read by the host ribosome (Figure 2.5). Initial experiments indicate that such ribosomes can be engineered to use some challenging amino acids with increased efficiency (Schmied *et al.*, 2018).

## Extensions of genetic code reprogramming might permit the creation of a crossover pathway for *in vivo* mirror protein synthesis made of natural-chirality components

Researchers studying genetic code expansion and reprogramming are re-engineering protein translation to use non-canonical amino acids, including D-amino acids (Katoh, Iwane *et al.*, 2017), in addition to the canonical amino acids. Extensions of this approach could in principle enable the creation of a crossover translation path for the production of mirror proteins inside natural-chirality bacteria (Figure 2.6A–B). Advancing this technology to enable *in vivo* mirror protein synthesis would likely require:

- 1. Generating 19 blank codons in the *E. coli* genome, one for each *D*-amino acid needed.
- 2. Generating 19 orthogonal aaRS/tRNA pairs that charge the 19 D-amino acids onto orthogonal tRNAs decoding the blank codons.
- 3. Engineering a crossover ribosome that can efficiently polymerize D-amino acids, and perhaps engineering other supporting translation factors.

As discussed in the previous section, many methods for generating blank codons are being studied, but synonymous codon compression is one possible approach. Synthesizing a viable genome that uses only 45 codons could free up 19 blank codons. A challenge in creating codon-compressed genomes is that synonymous codons can have coding functions in addition to specifying the order of amino acids, such that synonymous codon replacements can have fitness costs to the bacterium or be lethal. The codon replacement scheme in the 61-codon Syn61 genome was determined by empirical testing, and the final organism grew 60% slower than its parent 64-codon organism, although some



## Figure 2.5: Genetic code reprogramming allows the genetically encoded synthesis of non-canonical biopolymers in cells

**A.** Natural translation. Aminoacyl-tRNA synthetases acylate cognate tRNAs with amino acids, which are then used by the ribosome to make a protein. **B.** Orthogonal translation. The genetic code is reprogrammed to enable engineered orthogonal aminoacyl-tRNA synthetases to acylate orthogonal tRNAs with non-canonical amino acids (ncAAs). The resulting aminoacyl-tRNAs can then be used to decode orthogonal codons, potentially by an evolved orthogonal ribosome, to produce non-canonical biopolymers.

of that fitness loss could be recovered by subsequent laboratory evolution (Fredens *et al.*, 2019; Robertson *et al.*, 2021).

A 45-codon organism would still have substantial redundancy in its genome; a hypothetical "minimal-code" genome would only use 21 codons (for the 20 canonical amino acids and stop codon). As our understanding of synonymous codon choices and predictive and empirical methods to design recoded genomes advances, the creation of such a codon-compressed genome might become increasingly feasible.

Codon compression is just one strategy to generate the required blank codons. Alternatively, if the efficiency of quadruplet decoding or of an expanded genetic alphabet could be improved, *p*-amino acids could be encoded by these newly generated codons. Likewise, if a parallel translation pathway in which a separate ribosome selectively uses its own pool of tRNAs *in vivo* could be constructed, codons could be selectively reassigned in this parallel translation pathway without interfering with cellular translation. While substantial progress will be required to generate 19 blank codons by any of



#### Figure 2.6: Natural, crossover, and mirror-image translation

**A.** Natural translation. L-Amino acids are charged by L-aminoacyl-tRNA synthetases onto D-tRNAs, which read codons. L-Amino acids are polymerized by a natural-chirality ribosome to produce L-proteins. **B.** Crossover translation. D-Amino acids are charged by orthogonal L-aminoacyl-tRNA synthetases onto orthogonal D-tRNAs, which read orthogonal codons. D-Amino acids are polymerized by an engineered orthogonal natural-chirality ribosome to produce D-proteins. **C.** Mirror-image translation. D-Amino acids are charged by D-aminoacyl-tRNA synthetases onto L-tRNAs which read codons on a D-mRNA. D-Amino acids are polymerized by a mirror ribosome to produce D-proteins.

these approaches, and further technical hurdles such as wobble decoding would need to be addressed, it seems likely that this will eventually be possible.

The *in vivo* incorporation of D-amino acids would require the generation of 19 aaRS/tRNA pairs that could load the 19 D-amino acids onto tRNAs to be used for *in vivo* mirror protein synthesis. Each aaRS/tRNA pair would need to be orthogonal in its aaRS/tRNA recognition to other aaRS/tRNA pairs used for D-amino acid incorporation and the host aaRS/tRNA pairs that use the canonical L-amino acids, exclusively and efficiently charge their cognate D-amino acid and not other D- or L-amino acids, and exclusively and efficiently decode their target codon.

While no aaRS/tRNA pairs for D-amino acid incorporation *in vivo* have yet been reported, and creating pairs that fulfill these requirements would be a major protein engineering effort, most of the necessary methods have been described in principle. This includes approaches for discovering

additional orthogonal aaRS/tRNA pairs (Cervettini *et al.*, 2020), improving their efficiency (Amiram *et al.*, 2015), and methods for engineering synthetases to charge non-canonical amino acids with altered backbones (Dunkelmann *et al.*, 2024). Further advances in this field or in related areas, e.g., computational methods for protein design, would lower the expertise and resource requirements for this task further.

Genetic code expansion has so far mostly relied on the promiscuity of the ribosome toward non-canonical amino acids. However, while the ribosome is generally permissive toward non-canonical amino acids with altered side chains, amino acids with altered backbones such as p-amino acids are polymerized less efficiently. In vitro, under highly optimized conditions, up to ten p-amino acids can be polymerized consecutively by the ribosome, but at significantly reduced efficiency compared to L-amino acids (Katoh, Tajima, & Suga, 2017). Key translation factors such as EF-Tu, which delivers aminoacyl-tRNAs to the ribosome, are also specific to the L-amino acid backbone (Arranz-Gibert et al., 2018). Hence, it is likely that it would be necessary to engineer the ribosome and translation factors to support more efficient polymerization of p-amino acids in vivo to enable the cellular synthesis of mirror proteins. It is an open question to what extent ribosomal mutations can be found that enhance p-amino acid incorporation in vivo, and it is hard to predict how efficient the crossover ribosomes would be. However, as tools for ribosome engineering continue to be developed (d'Aquino et al., 2018; de la Torre & Chin, 2021) and general protein engineering and directed evolution methods improve (Simon et al., 2019; Yang et al., 2019), this challenge is becoming increasingly tractable. For example, a ribosome engineered to incorporate p-amino acids and not L-amino acids, with a Shine-Dalgarno sequence that is orthogonal to the existing ribosome, would minimize the risk of unwanted crosstalk between normal and mirror protein creation (Dunkelmann et al., 2021; Rackham & Chin, 2005).

The levels of p-amino acid required for efficient mirror protein translation from a crossover ribosome might be toxic to existing cells (Aliashkevich et al., 2018). Many p-amino acids are already present in existing organisms, though at lower concentrations than L-amino acids. Extracellular concentrations in the millimolar range of certain amino acids, especially of p-Trp, can inhibit growth due to misincorporation by the ribosome (Leiman et al., 2013; Rumbo et al., 2016; Soutourina et al., 2000; see also Box 1.2); intracellular concentrations of the corresponding L-amino acid enantiomers are only a few-fold lower. That is, the current canonical tRNA/aaRS pairs for existing L-amino acids may misincorporate *D*-amino acids too frequently to tolerate the higher concentrations that would be needed to support a crossover ribosome. The enzyme p-aminoacyl tRNA deacylase (DTD) functions to hydrolyze tRNAs mischarged with p-amino acids. DTD is highly promiscuous, functioning on a wide range of p-amino acids and an equally broad range of tRNA structures, which may make it difficult to produce orthogonal tRNAs intended to be charged with p-amino acids by their own aaRS enzymes. Deleting DTD would resolve this problem, but doing so greatly increases the toxicity of p-amino acids to bacteria (Leiman et al., 2013). In principle, DTD and the new tRNAs might be engineered to be orthogonal to one another. Alternatively, artificial organelles might be used to sequester the mirror components, including p-amino acids, away from the rest of the cell; this possibility is discussed further in Section 2.4.

A crossover pathway could be used to create a self-sustaining mirror central dogma *in vivo* 

Once mirror proteins could be made *in vivo* using the crossover pathway, the next step would involve creating pathways for the production of mirror DNA and RNA *in vivo* (Figure 2.6C). Mirror DNA could be replicated *in vivo* by making mirror versions of the enzymes that replicate DNA, i.e., initiator proteins, a DNA polymerase, helicase, ligase, topoisomerase, single-strand binding protein, and other accessory proteins that make up the replisome (Oakley, 2019). Additionally, it would be necessary to provide a pathway for the synthesis of mirror-image nucleotide triphosphate (NTP) building blocks. Once these mirror proteins were produced inside the cell, it might be possible to transform an *in vitro*-made mirror DNA plasmid into the cell, which would then be propagated by the mirror replisome. Once mirror DNA could be propagated within a bacterium, RNA could be produced by expressing a mirror RNA polymerase.

In the crossover pathway, a natural-chirality protein synthesis system is used to create mirror proteins (Figure 2.6B). However, once both mirror proteins and mirror RNA can be produced *in vivo*, scientists might be able to construct a mirror version of translation (Figure 2.6C). Unlike the crossover pathway, a mirror version of translation would use mirror translation machinery made of mirror proteins and mirror RNA to create mirror proteins. Once this is established, it could be self-sustaining: mirror replication, transcription, and translation machinery could be encoded in mirror DNA, which is then transcribed and translated by the mirror machinery. At this point, the crossover translation system could be removed. This would yield a mixed chirality bacterium that simultaneously carried a natural-chirality genome and a natural-chirality proteome similar to natural life, but additionally carried a mirror DNA plasmid and mirror central dogma machinery capable of creating mirror DNA, RNA, and proteins (Figure 2.6C).

Establishing a mirror central dogma might face some technical challenges. For example, the synthesis of mirror ribosomal RNA and mirror ribosomal proteins might have to be coordinated for *in vivo* mirror ribosome biogenesis, and suitable mirror RNA-modifying enzymes would additionally need to be present. The many components of the mirror translation system might interfere with the natural-chirality host cellular machinery in hard-to-predict ways. Nonetheless, while troubleshooting this system at this stage would require skillful experimentation, it would not necessarily require the large resource investments that are involved in creating 19 blank codons or engineering 19 mutually orthogonal aaRS/tRNA pairs, for example, or the synthesis of a large number of mirror proteins like in the bottom-up approach.

## A mirror bacterium could potentially be created by delivering a mirror DNA genome and removing the natural-chirality genome

Finally, researchers might expand the mirror DNA plasmid to encompass a complete mirror genome and remove the natural-chirality genome from the cell, to create a fully mirrored bacterium (Figure 2.7). One strategy could be to build a mirror genome *in vitro* and deliver it to the mixed-chirality cell in a single step. If the mirror genome is taken up, the mirror transcription and translation machinery could begin gene expression from the mirror genome, producing all the mirror proteins and RNAs



## **Figure 2.7: Stepwise conversion of a natural-chirality bacterium into a mirror bacterium A.** Stage 1: A natural-chirality bacterium with a natural-chirality central dogma. **B.** Stage II: A natural-chirality bacterium that additionally contains a crossover translation system made of natural-chirality components that is able

bacterium that additionally contains a crossover translation system made of natural-chirality components that is able to produce mirror proteins *in vivo*. **C.** Stage III: A mixed-chirality bacterium that contains a natural-chirality central dogma and a mirror central dogma. **D.** Stage IV: A fully mirrored bacterium that contains a mirror genome and a mirror central dogma and no natural-chirality components.

that make up a mirror bacterium. At the same time, enzymes that degrade the natural-chirality genome could be expressed, thereby stopping gene expression from the natural-chirality genome. Hence, after some turnover, the mirror genome, transcriptome, and proteome would be expressed whereas the natural-chirality genome, transcriptome, and proteome would be degraded. If the transition can be successfully managed, the result would be a viable mirror cell. This is conceptually similar to the "booting" experiment in the bottom-up pathway.

This chirality conversion step could be made less drastic, improving the chance of success, by expanding the DNA episome in a stepwise fashion to build a mirror genome. A challenge in this approach is that the co-expression of some natural-chirality and mirror genes could cause toxicity as some mirror proteins would likely interfere with regular cellular processes. For example, mirror bacterial cell division proteins might interfere with cell division, or mirror metabolic enzymes with metabolism. To mitigate this, researchers could try to replace sets of proteins governing natural-chirality functional modules, such as cell division or energy metabolism, entirely with mirror components. This would result in a step-by-step conversion of natural-chirality to mirror bacteria in which some functional modules would be natural-chirality, and some mirror, during the conversion

stages. Like the previous step, the cell conversion step would be a technically challenging but tractable path to create a mirror bacterium.

### 2.4 Other approaches to creating mirror bacteria are plausible

Science rarely proceeds in linear fashion, and progress is unlikely to precisely follow the pathways outlined in the above sections. In this section, we turn to consider some pathways that appear more speculative based on our current understanding of the challenges, but could become more plausible in the coming decades as related scientific fields advance.

Research along the bottom-up and top-down pathways would not proceed in isolation, and would be able to draw on wider progress in the synthetic cell, genetic code expansion, and bioproduction communities. For example, an *in vivo* system for the synthesis of mirror proteins could provide mirror proteins for a bottom-up approach. Similarly, *in vitro* methods for the synthesis of a mirror DNA genome would facilitate the introduction of a mirror DNA genome in a natural-chirality cell. Insights as to the difficulties of booting up a genome in the bottom-up approach might inform the boot in a top-down path, and vice versa.

As discussed in Section 2.2, there is great academic interest in generating synthetic cells from non-living components. The first synthetic cell could be a replica of an existing organism and use DNA, proteins, carbohydrates, and other macromolecules familiar from natural life. However, there is also work aimed at creating synthetic cells from abiological components, for example using compartments other than phospholipid vesicles (Discher *et al.*, 1999; Holowka *et al.*, 2005; Joesaar *et al.*, 2019; Shang *et al.*, 2022; van Swaay *et al.*, 2015) or atypical or abiological macromolecules to perform key cellular functions (Jiang *et al.*, 2022; Zhan *et al.*, 2022). Similarly, researchers are interested in synthesizing plausible early forms of life to study the origin of life, for example RNA-only protocells (Joyce & Szostak, 2018). Most artificial cells of this kind are unlikely to be viable outside the laboratory. However, the creation of such artificial cells may facilitate the creation of a mirror bacterium by stepwise conversion of the abiological artificial cell into a mirror bacterium.

Establishing a crossover and a mirror translation system inside a living bacterium, as discussed in Section 2.3, would be challenging and labor-intensive. An alternative approach could be to temporarily and fully switch the protein synthesis machinery from natural-chirality protein production to mirror protein production. If this could be done, the natural-chirality DNA genome could be temporarily translated into a mirror proteome. With the mirror proteome in place, if a mirror DNA genome could be delivered at this stage and the natural-chirality DNA genome degraded, it might be possible to rapidly convert a natural-chirality bacterium into a mirror bacterium. This strategy is attractive as it might be easier to temporarily remodel the translation machinery for mirror protein synthesis using the natural-chirality genome than to stably co-express mirror protein synthesis systems alongside the natural-chirality protein synthesis that keeps the natural-chirality bacterium alive. However, there would be fewer opportunities for troubleshooting compared to the path outlined in Section 2.3.

Another modification of the approach outlined in Section 2.3 would employ a natural organelle like the mitochondrion (which is non-essential for yeast replication) or the creation of an artificial

organelle inside a natural-chirality organism to avoid the need for creating blank codons and to sequester the potentially toxic mirror components. Researchers have successfully constructed artificial organelles made of proteins and of lipid membranes in bacteria using a variety of methods and used them to isolate metabolic pathways from the rest of the cell (Oerlemans et al., 2021). For example, introducing peroxisomal proteins from eukarva into bacteria has generated membrane-encapsulated artificial peroxisomes (Cross et al., 2017). The same peptide tags used to transport peroxisomal enzymes in eukaryotes (Baker et al., 2024) can direct tagged proteins and associated biomolecules into the bacterial peroxisomes, allowing them to perform biochemical reactions in isolation. Bacterial organelles harboring independently replicating DNA exist in nature: the bacterium Candidatus Thiomargarita magnifica features numerous membrane-bound organelles called pepins that contain copies of the genome and/or active ribosomes (Volland et al., 2022). It appears likely that the organelles themselves replicate independently, much like mitochondria in eukaryotic cells. Such an organelle could in principle harbor an independent episome encoding the desired mirror proteins, the polymerase required to transcribe them, and the crossover translational machinery-including p-amino acids at high concentrations-capable of producing them in vivo. As artificial organelle as well as mitochondrial and chloroplast engineering advance, it appears plausible that the technologies required to sequester a mirror central dogma within a natural-chirality cell could be developed.

Future advances in enzyme engineering could enable the creation of enzymes using energy to drive the conversion of L-amino acids to D-amino acids within a polypeptide chain. A "protein handedness converter", if it could be created, could provide a shortcut to create mirror life: it could be used to produce mirror-image proteins *in vitro* for the bottom-up pathway, or *in vivo* for a top-down pathway, or enable the creation of a transient crossover translation system as discussed above.

In recent years, progress in artificial intelligence (AI)-based scientific methods have had a transformative impact in the life sciences. It is plausible that further advances in AI might accelerate solutions for many of the technical hurdles in the approaches outlined above. This could include designing enzymes to manipulate mirror biomolecules, strategies to create key components such as synthetic ribosomes, engineered translational machinery to facilitate *in vivo* mirror protein production, or improved genome design approaches that might facilitate booting or cell conversion experiments, among others. AI might also help to accelerate experimentation, for example by facilitating laboratory automation.

## **2.5** The feasibility of mirror life will increase as related technologies advance

In the previous sections, we found that multiple plausible pathways to the creation of mirror bacteria exist, although major technological hurdles remain.

In the bottom-up pathway outlined in Section 2.2, the most notable bottlenecks are the development of recipes for the assembly of self-replicating cells from a protein synthesis system, a membrane, and a genome, alongside strategies for creating entirely synthetic ribosomes. Moreover, advances that would reduce the cost and effort of mirror protein and nucleic acid synthesis would greatly facilitate

any effort toward the creation of a mirror bacterium. Finally, existing genome synthesis approaches would need to be adapted to enable the synthesis of a mirror genome without *in vivo* manipulations.

In the cell conversion pathway described in Section 2.3, a major bottleneck is the development of a crossover translation pathway that would allow the synthesis of mirror proteins entirely with natural-chirality cellular machinery. This, in turn, would require the creation of 19+ blank codons that could be assigned to D-amino acids or an organelle system for sequestration, alongside the development of suitable aaRS/tRNA pairs that could use the 19 D-amino acids *in vivo*, and a crossover ribosome capable of polymerizing these D-amino acids efficiently. Moreover, extensive cellular engineering may be required to maintain a mirror central dogma within a natural-chirality cell, and to facilitate the final cell conversion step to create a fully mirrored bacterium.

Overall, our analysis shows that the creation of a mirror bacterium is not possible with current technology absent extensive investment of time and resources to overcome several scientifically challenging and labor-intensive technical hurdles. However, we found no technical hurdle that would pose a fundamental obstacle to the creation of a mirror cell, and we expect that dedicated research programs could eventually resolve each bottleneck.

While few laboratories explicitly focus on the creation of mirror life, many of the enabling technologies are actively being developed by a large community of academic and industrial researchers for reasons unrelated to mirror life. For example, a large community of researchers is pursuing synthetic cell research for applications in basic science and medicine without the aim of creating a mirror bacterium; likewise, researchers in genetic code reprogramming are advancing methods toward the *in vivo* production of non-canonical biopolymers without an explicit interest in mirror bacteria. For these reasons, we expect that over time, with scientific progress, many of the technical hurdles will be overcome even in the absence of explicit efforts to create mirror life. This will increasingly open up pathways to the creation of mirror bacteria. While forecasting technological progress is highly uncertain, absent governance of the enabling technologies, we estimate that mirror bacteria could be created within the next 15–30 years; or sooner, if substantial resources were invested in a focused effort.

## Chapter 3: Engineering, Biosafety, and Biosecurity of Mirror Bacteria

Katarzyna P. Adamala<sup>1</sup>, Yizhi Cai<sup>2</sup>, Kevin M. Esvelt<sup>3</sup>, John I. Glass<sup>4</sup>, Farren J. Isaacs<sup>5,6,7</sup>, Sebastian B. Oehm<sup>4,8</sup>, Jaspreet Pannu<sup>9,10,11</sup>, Jack W. Szostak<sup>12</sup>

As outlined in subsequent chapters, due to their opposite chirality, sufficiently robust mirror bacteria could plausibly evade immune responses and resist predation, allowing them to infect an unusually wide range of multicellular organisms and invade diverse environments. Depending on the path through which they are created, the first mirror cells might not be very robust—perhaps more akin to a synthetic minimal cell than a natural bacterium. However, it would likely be relatively straightforward to convert a fragile mirror bacterium into the mirror image of a robustly growing natural-chirality bacterium, yielding a robust mirror bacterium. This distinguishes mirror bacteria from other types of highly engineered bacteria or synthetic cells, where the creation of fast- and robust-growing strains is often challenging. Though biocontainment methods could reduce risks from mirror bacteria, these safeguards are susceptible to human error or could be undone through genetic engineering. Because fragile and biocontained mirror bacteria could be converted into more dangerous versions by a nefarious or reckless actor, the creation of any mirror bacterium would create a major security risk.

In Section 3.1, we discuss how any mirror bacterium could be used as a starting point to generate additional mirror bacterial strains and species, and how these mirror bacteria could be augmented using routine genetic engineering. While creating the first mirror bacterium would be a major undertaking, subsequent modifications could be accomplished using standard genetic engineering techniques. Using these methods, a fragile mirror bacterium could be converted into a strain that grows robustly and further enhanced—for example, by engineering its ability to metabolize natural-chirality nutrients.

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup> Department of Genetics, Cell Biology and Development, University of Minnesota, Minneapolis, Minnesota, USA; <sup>2</sup> Manchester Institute of Biotechnology, University of Manchester, Manchester, UK; <sup>3</sup> Media Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, USA; <sup>4</sup> J. Craig Venter Institute, La Jolla, California, USA; <sup>5</sup> Department of Molecular, Cellular, and Developmental Biology, Yale University, New Haven, Connecticut, USA; <sup>6</sup> Systems Biology Institute, Yale University, West Haven, Connecticut, USA; <sup>7</sup> Department of Biomedical Engineering, Yale University, New Haven, Connecticut, USA; <sup>8</sup> Trinity College Cambridge, University of Cambridge, Cambridge, UK; <sup>9</sup> Center for Health Security, Bloomberg School of Public Health, Johns Hopkins University, Baltimore, Maryland, USA; <sup>10</sup> Department of Health Policy, Stanford School of Medicine, Stanford University, Stanford, California, USA; <sup>12</sup> Howard Hughes Medical Institute, Department of Chemistry, The University of Chicago, Chicago, Illinois, USA

In Section 3.2, we discuss how physical and biological biocontainment approaches might reduce the risk of an inadvertent release of mirror bacteria. Making the growth and survival of mirror bacteria dependent upon multiple molecules not found in nature would reduce the chance of an accidental release. Given the serious potential harms of such a release, however, the safety standards for mirror bacteria would need to be substantially greater than those applied to natural-chirality organisms.

In Section 3.3, we consider the biosecurity implications of mirror bacteria. We find that, while the initial creation of a mirror bacterium would be technically challenging, the conversion of a fragile or biocontained mirror bacterium into a potentially dangerous mirror bacterium could be accomplished with relative ease, posing serious security concerns.

# 3.1 The creation of any mirror bacterium could enable the generation of diverse mirror bacterial strains and species and their modification by routine genetic engineering

As outlined in Chapter 2, the creation of a mirror bacterium would be a highly challenging undertaking. Depending on its path to construction, the first mirror bacterium might be intrinsically fragile, metabolically limited, and hampered by genetic artifacts of the assembly process. Such an organism would likely have insufficient robustness and fitness for invasion of the environment and might require numerous mirror-image nutrients that are rare or absent in nature. However, compared with the initial creation of a mirror bacterium, the transformation of such a bacterium into more robust mirror bacterial strains or species would be notably less challenging. In addition, mirror bacteria could be augmented with new capabilities through genetic engineering, including the ability to digest natural-chirality sugars.

Theoretical applications of mirror bacteria in industry and medicine (Box 3.1) would require robustly growing mirror bacterial strains and would benefit from strains capable of digesting natural-chirality nutrients. Absent concerns regarding potential safety and security risks, these applications would create incentives for the development of such mirror bacteria.

### Existing techniques for engineering bacteria could be adapted to mirror bacteria

Because mirror biology and natural-chirality biology would behave symmetrically, most tools and techniques developed for genetic engineering of natural-chirality bacteria could be adapted for mirror bacteria without further development or requirements beyond access to synthetic mirror DNA, the mirror bacterium of interest, and a small set of mirror protein reagents for more complex manipulations. While accessibility decreases as the required changes become more numerous and complex, simple changes—such as providing antibiotic resistance and engineering a mirror bacterium to feed on natural-chirality nutrients like p-glucose—would be straightforward for individuals with basic molecular biology training.

Scientists have been genetically engineering bacteria for decades and have developed powerful tools for doing so. Notable successes include the production of recombinant proteins such as insulin (Goeddel *et al.*, 1979; Overton, 2014), the production of valuable small molecules by fermentation

## Box 3.1: Some potential applications of mirror bacteria would require high-risk strains

The initial development of mirror bacteria may be driven by academic interest rather than interest in any specific application. Nevertheless, mirror bacteria could have potential applications in both industry and medicine if not precluded by the serious biosafety and biosecurity risks outlined in this report.

Mirror bacteria could be used to **manufacture mirror biomolecules** such as mirror-image nucleic acids or mirror-image proteins. These mirror biomolecules can have applications in drug discovery and beyond, as discussed in Chapter 2. Mirror bacteria would provide a straightforward route to manufacturing these molecules at scale, similar to the production of recombinant natural-chirality DNA or natural-chirality proteins in natural-chirality bacteria. This would require the creation of a mirror bacterium (like *E. coli* or *B. subtilis*) that grows robustly and is fit enough to pay the metabolic costs of biomolecule production. Such strains would benefit from being able to metabolize natural-chirality nutrients such as p-glucose, as growth on natural-chirality media would be substantially cheaper than creating synthetic media with mirror nutrients. Hence, even if biocontained through other means, such bioproduction strains would be closely related to mirror bacterial species that could potentially act as pathogens and spread uncontrollably in the environment.

There are several alternatives to mirror bacterium-based manufacture of mirror biomolecules. Mirror biomolecules can be made at small to medium scale using existing chemical synthesis approaches (Harrison *et al.*, 2023), and future research might improve the accessibility, scale, and cost-effectiveness of these methods (Tan *et al.*, 2020). In the future, scientists could develop methods for mirror biomolecule synthesis using *in vitro* enzymatic approaches (using natural-chirality or mirror enzymes) that do not rely on self-replicating cells (Weidmann *et al.*, 2019). Scientists might also succeed in engineering natural-chirality bacteria to produce mirror biomolecules (de la Torre & Chin, 2021).

More speculatively, mirror bacteria might be engineered to act as **immune-invisible live cell therapeutics.** The increasing ability to genetically engineer bacteria has led to increased interest in using bacteria as live 'smart' therapeutics that could seek out a specific environment within a human patient, sense the disease state, compute an appropriate response, and produce effector molecules correspondingly to treat the disease (Riglar & Silver, 2018). Mirror bacteria might present an interesting chassis for live cell therapeutics, as they are expected to have greatly reduced recognition by mammalian immune systems (see Chapter 4). If growth of the mirror bacteria could be made dependent on one or several unnatural metabolites—i.e., a synthetic auxotrophy—their growth within a human host could be precisely controlled (Mandell *et al.*, 2015). However, therapeutic applications are generally incompatible with physical containment or access control. A therapeutic mirror bacterial strain would need to be capable of robust growth within a human host, and hence

be close to highly dangerous forms. Natural-chirality microbes that exhibit reduced immune reactions could represent alternative chassis for live-cell therapeutic applications that pose much lower biosafety and biosecurity concerns.

(Nielsen & Keasling, 2016; Ro *et al.*, 2006), the detection of specific chemical compounds (Chiang & Hasty, 2023; Riglar & Silver, 2018), and the *in situ* production of therapeutics to ameliorate disease (Riglar & Silver, 2018).

Some bacteria are most readily engineered by introducing plasmids—extrachromosomal circular DNA that can replicate independently from the bacterial genome—while others readily integrate introduced DNA sequences into their genomes. Researchers can rapidly generate suitable genes through chemical DNA synthesis and *in vitro* assembly methods, which can then be transformed into bacterial hosts within a few days. The process is routinely practiced by biology undergraduates and, in some cases, high school students (Gronvall, 2016; Jackson *et al.*, 2019). It can also be done with minimal laboratory equipment by citizen scientists and biohackers (Gronvall, 2016).

The synthesis of mirror genes and the transformation of mirror bacteria could be relatively straightforward for anyone with access to synthetic mirror DNA and—for larger constructs requiring assembly—a small set of mirror enzymes. Mirror DNA is already commercially available, although it is more expensive than normal DNA. While mirror enzymes are currently only available in a small handful of laboratories, they would likely be much more accessible by the time the first mirror cells are created. Alternatively, advances in protein engineering could allow natural-chirality enzymes to perform the required functions in the future. Transformation methods such as electroporation or heat shock are not sensitive to chirality and could be used to introduce synthetic mirror DNA into a mirror bacterium in a straightforward manner.

Some applications in synthetic biology require more extensive engineering of the bacterial genome. In model organisms such as *Escherichia coli*, numerous methods permitting the rapid and facile editing of bacterial genomes have been developed using tools such as recombineering (Sharan *et al.*, 2009; Wang *et al.*, 2009) or CRISPR-Cas9 (Teng *et al.*, 2024). In extreme cases involving thousands of changes, it can be more economical to synthesize the entire genome from scratch, as discussed in the following section. Many of these techniques require only the introduction of specific plasmids or synthetic DNA and hence could be adapted to mirror bacteria in a straightforward manner with access to synthetic mirror DNA. However, more complex manipulations may require skilled practitioners, greater quantities of synthetic DNA, or additional enzymes.

Finally, bacterial fitness can be improved and bacteria can be adapted to desired environments by adaptive laboratory evolution, which would work equally well for natural-chirality or mirror bacteria. In this process, scientists continuously passage a bacterial strain under selective conditions of their choice, such as a specific growth medium containing unusual nutrients or abiotic stresses (Sandberg *et al.*, 2019). This simple process can yield substantial increases in fitness for strains that are

otherwise fragile and grow slowly due to large-scale genome modifications such as recoding or genome reduction (Hemez *et al.*, 2024; Moger-Reischer *et al.*, 2023).

Adaptive evolution can also optimize novel metabolic pathways, conferring the ability to utilize otherwise inaccessible nutrients. For example, *E. coli* strains have been evolved to grow on suboptimal or even non-native carbon sources (Fong *et al.*, 2003; Herring *et al.*, 2006; Lee & Palsson, 2010). Evolution also occurs in the absence of intentional human effort: as described in Section 8.4, evolution within the natural environment could exacerbate the risks from mirror bacteria if they escape confinement.

### Any mirror bacterium would facilitate the creation of diverse mirror bacterial strains

The first mirror bacterium, if created, could be similar to a mirror *Mycoplasma* or *E. coli*, perhaps with a reduced or minimal genome or other genetic changes to facilitate the booting process, as discussed in Chapter 2. Because strains with highly modified or reduced genomes can be fragile and difficult to study, a logical next step would involve restoring missing genes and removing any artifacts of the boot process. A minimal *E. coli*, for instance, could be converted into mirror analogs of more robust and useful strains by introducing genes that mirror those found in other *E. coli*. For instance, researchers might wish to create mirror versions of *E. coli* K-12, BL21, and Nissle 1917, strains that are widely used for science, bioproduction, and medicine.

Creating a diverse range of mirror bacterial species would be a logical further step: mirror *Bacillus subtilis* or mirror cyanobacterium *Synechococcus elongatus* would offer different properties and enable different applications compared to mirror *Mycoplasma* or *E. coli*. While it is plausible that such mirror bacteria could be created from scratch using analogous techniques to that of the original mirror bacterium, it is also possible that such mirror bacteria could be created by converting existing mirror bacteria.

In 2010, researchers demonstrated that it is possible to convert *Mycoplasma capricolum* into the closely related Mycoplasma mycoides by transplanting a chemically synthesized M. mycoides genome into a *M. capricolum* recipient cell (Gibson *et al.*, 2010). Genomic transplants between *M*. *capricolum* and more distantly related members of the *Spiroplasma* genus have also been performed, though the likelihood of success appears to be inversely related to phylogenetic distance (Labroussaa et al., 2016). Large-scale hybrids of E. coli and Salmonella enterica have also been created, in some cases without any significant reduction in growth fitness (Bartke et al., 2021), ultimately replacing over 90% of the genome and effectively transforming the initial bacterium into a species from a different genus without the need for genomic transplantation. Conversion between more distant species remains poorly explored, but the creation of a *Bacillus subtilis* bacterium containing the entire genome of the cyanobacterium Synechocystis PCC6803 suggests a potential path via switching the 16S ribosome to the new strain (Itaya et al., 2005). Ongoing advances in DNA synthesis (Hughes & Ellington, 2017) and synthetic genomics (Schindler et al., 2018) will likely make the process easier in the coming decades. While much remains unknown about the versatility and robustness of interspecies conversion methods, the successful creation of the first mirror bacterial species could enable the creation of additional mirror bacterial species even absent independent "boots".

### Fitness-enhancing capabilities can be ported from existing organisms

Natural organisms contain a diverse array of genes and functional capabilities that bioengineers could draw upon to create novel and diverse mirror bacterial strains. A natural-chirality gene can be mirrored by creating mirror DNA with an identical genetic sequence. By chiral symmetry, mirror genes in a mirror organism would function as mirror images of their natural-chirality counterparts in a natural-chirality organism. For example, genes allowing the catabolism of D-lactose or D-mannitol would, if encoded into mirror DNA, allow a mirror bacterium to catabolize L-lactose or L-mannitol.

As a result, creating metabolic pathways for the catabolism of natural-chirality molecules within mirror bacteria is not straightforward. Conferring the ability to catabolize natural-chirality D-glucose would require a metabolic pathway that would, in natural-chirality bacteria, allow the catabolism of mirror-image L-glucose. This ability is not found in common lab strains (including *E. coli*). It is, however, possible to isolate bacteria with this capacity from soil samples (Shimizu *et al.*, 2012; Yachida & Nakamura, 2024), and the genes comprising one such utilization pathway in *Paracoccus laeviglucosivorans* have been characterized *in vitro*. By encoding genes for this pathway into mirror DNA along with a suitable transporter from an L-glucose catabolizer, D-glucose catabolism could be engineered into mirror bacteria, then evolved or engineered for higher metabolic efficiency. Given that D-glucose is far cheaper and more abundant than L-glucose, engineering D-glucose catabolizing strains would be a logical aim for nearly any potential application—but this modification would greatly increase the anticipated fitness of the mirror bacterium in the environment by enabling it to utilize the most abundant natural sugar.

Pathways exist to catabolize the mirror versions of other common sugars, including D-arabinose (LeBlanc & Mortlock, 1971), L-fructose (Park *et al.*, 2007), L-galactose (Zhu & Lin, 1987), L-glyceraldehyde (Zhu & Lin, 1987), L-ribose (Trimbur & Mortlock, 1991), and L-xylose (Usvalampi *et al.*, 2012; see also Table 1.3). The ability to catabolize mirror-image D-amino acids is also common in natural bacteria (see Table 1.2 for D-amino acid catabolism in *E. coli*), often proceeding through amino acid racemases that interconvert L- and D-amino acids (Miyamoto & Homma, 2021). Such racemases could similarly allow mirror bacteria to catabolize the ubiquitous canonical L-amino acids.

Abilities that require chiral interactions between mirror macromolecules and natural macromolecules would be more challenging to engineer. For instance, mirror proteases will not digest normal-chirality proteins (as discussed in Chapter 1), mirror antibodies will not bind to the same ligands as their natural-chirality counterparts, and a mirror AB toxin would be unlikely to function against natural cells. The *de novo* design of D-peptides to bind to specific natural-chirality peptide and protein targets has recently been demonstrated (Sun *et al.*, 2024). As protein design tools improve, it should be possible to create mirror proteins with more complex activities and specificities, though such mirror proteins would not necessarily share high sequence and structure similarity with any existing natural proteins.

## **3.2** Biocontainment approaches might reduce accident risk, but they would face challenges

Given that (1) mirror bacteria might become synthetically accessible within the next few decades, (2) once created, engineering more robust and diverse mirror bacteria would be comparatively straightforward, and (3) a sufficiently robust and metabolically flexible mirror bacterium released into the environment could have severe consequences, mirror bacteria pose serious potential biosafety and biosecurity challenges. Biocontainment techniques might reduce accident risk from mirror bacteria, but securing them against misuse appears more challenging.

## Physical containment would likely be insufficiently robust to prevent leaks of dangerous mirror bacteria

Biologists have long worked with dangerous pathogens. Such work can provide substantial benefits, enabling the behavior of pathogens to be better understood, and in some cases, the development of vaccines and other countermeasures. However, such work can pose risks to both the scientists performing the research and to others in the community who could be inadvertently exposed to the pathogen. Over time, biosafety practices have developed in order to best mitigate risks while still allowing vital research to be conducted.

When work on dangerous pathogens is required, strict physical containment measures within special biosafety laboratories are used to minimize risks. Nevertheless, even highly safe laboratories (e.g., with the BSL-4 designation) are susceptible to accident risk, and a number of such cases have been reported (Byers & Harding, 2016; Gillum *et al.*, 2016; Manheim & Lewis, 2022). Therefore, physical containment alone is typically considered insufficient to permit experiments with biological agents that could have catastrophic consequences. For example, the scientific community has abstained from work involving the recreation or modification of smallpox (World Health Organization, 1980, 1996, 2010), which was eradicated in 1977.

Most of the highest-risk biological research is conducted on obligate pathogens, where protecting lab workers from infection is the highest priority. While BSL-3 and 4 laboratories do include measures to sterilize waste material and prevent the escape of aerosols, it is difficult to know how often such measures fail because most accidental releases do not result in detectable consequences. There are some well-known exceptions in which accidental leaks did result in visible infections. For instance, the last known death from smallpox, that of Janet Parker in 1978, was almost certainly due to a laboratory leak (UK Department of Health and Social Security, 1980). The 2007 UK outbreak of foot and mouth disease is also likely to have originated from a lab, and probably entered the environment through leaky pipes (Spratt, 2007).

A sufficiently robust mirror bacterium, unlike a virus, could potentially replicate independently within the environment, even absent a suitable host (see Chapter 8). Therefore, any failure of ventilation or waste sterilization systems could potentially be catastrophic. Given the rate of known laboratory-acquired infections, which likely reflects a higher rate of unknown accidental leaks of

biological material with or without infection, current physical containment measures do not appear sufficiently robust to permit work with potentially dangerous mirror bacteria.

Biological containment approaches might reduce the likelihood of environmental escape but would not alone be sufficient to achieve effective biosafety or biosecurity

Synthetic biologists have long been concerned with the escape of bioengineered natural-chirality microbes and have begun developing techniques to contain such organisms. Unlike physical containment, biological containment involves modifying the organism itself so that it is incapable of replicating in the external environment without the addition of specific chemical compounds. Such measures could allow the development of mirror bacteria with a reduced likelihood of environmental escape, providing greater robustness compared to physical containment (though implementing physical containment measures would still provide an additional layer of protection).

A common technique for the biocontainment of engineered bacteria is to make their growth and survival dependent on the supply of a specific metabolite—a metabolic auxotrophy. For example, genes for the biosynthesis of key metabolites like nucleosides and amino acids can be disrupted, rendering the bacterium dependent on the supply of a missing metabolite that it cannot produce itself (Bahey-El-Din *et al.*, 2010; Ronchel & Ramos, 2001; Steidler *et al.*, 2003). This system is straightforward to implement, but suffers from several drawbacks in natural-chirality bacteria: the auxotroph might find the missing nutrients in some natural environments or regain the ability to produce them by horizontal gene transfer. In a mirror bacterium, these drawbacks can be largely eliminated, but care would be required to create systems that are robust to human error.

Many mirror molecules are thought to be absent from the natural environment, which should make it possible to create auxotrophs that definitely could not survive in nature. Careful experimentation with natural-chirality bacteria could be used to verify that these auxotrophs could not escape using unexpected metabolic pathways. For instance, though p-amino acids are not as common as L-amino acids, they do occur naturally—and a subset is known to fulfill specific biological functions (Genchi, 2017). Furthermore, as discussed in Chapter 1, many bacteria can utilize both amino acid enantiomers due to promiscuous transporters and enzymes (see Table 1.2). With careful design, it should be possible to identify metabolic auxotrophies that would avoid these difficulties and minimally disrupt cell functioning. L-pantothenate, for instance, might be absent in nature. Since some natural-chirality bacteria already lack p-pantothenate biosynthesis and instead obtain p-pantothenate from the environment (Gerdes *et al.*, 2002), the pathway is dispensable. Lipoic acid and biotin auxotrophy are other plausible options. Multiple distinct auxotrophies could be used simultaneously to increase robustness.

Though mirror bacteria cannot obtain genetic material from natural-chirality organisms, gene transfer between mirror bacterial strains is possible. In particular, if mirror bacteria with different metabolic auxotrophies were created, they may inadvertently come into contact and transfer genetic material, creating a mirror bacterium that is no longer auxotrophic. Technical interventions to prevent horizontal gene transfer could help to mitigate this risk. Additionally, human error during attempts to
switch an auxotrophy or introduce additional biosynthetic capabilities could also lead to accidental escape.

While metabolic auxotrophy is a straightforward way to create a mirror bacterium with reduced escape risk, other auxotrophic biocontainment methods have been developed. One approach involves re-engineering multiple essential cellular proteins to require the incorporation of an unnatural non-canonical amino acid in order to function, creating a form of 'synthetic auxotrophy' (Mandell *et al.*, 2015; Rovner *et al.*, 2015). If properly implemented, escape from synthetic auxotrophy through horizontal gene transfer or other evolutionary mechanisms would be very unlikely. Such biocontainment would, however, be more technically challenging to implement. While non-auxotrophic containment methods exist, they are unlikely to be suitable for mirror bacteria because they are vulnerable to the gradual acquisition of individual mutations, even taking into account efforts to stack multiple non-auxotrophic methods or implement interventions to reduce the rate of evolutionary drift (Brophy & Voigt, 2014; Calles *et al.*, 2019; Gallagher *et al.*, 2015). Future advances could yield superior biocontainment strategies.

Biological approaches to biocontainment have been developed for microbes used in industrial or medical applications. In these cases, environmental escape is undesirable, but generally not expected to cause major consequences for humans or ecosystems. In contrast, it appears that the accidental release of mirror bacteria could have potentially severe consequences for humans and the environment (see subsequent chapters). While effective biocontainment using the approaches discussed above might reduce the escape risk by many orders of magnitude, a low accident risk due to human error or unexpected evolutionary events would persist even if highly advanced biocontainment approaches were implemented. If mirror bacteria prove highly dangerous, this residual risk may still outweigh the benefits of their scientific and technological applications.

An effective biosafety regime would require not only methods to robustly biocontain mirror bacteria, but also strict provisions to prevent these biocontainment measures from being disabled by accident or recklessness. Whether governance mechanisms exist that could lower the risk to an acceptable level, and whether there are realistic implementations of such measures that could be effectively implemented across the many jurisdictions where work on mirror bacteria might occur, lies beyond the scope of this report. We will, however, briefly discuss the biosecurity challenges raised by mirror bacteria, which appear particularly severe.

### 3.3 Creating robustly biosecure mirror bacteria is not feasible

Terrorist groups, including Al-Qaeda (Parachini & Gunaratna, 2022), the Islamic State (Parachini & Gunaratna, 2022), and Aum Shinrikyo (Danzig *et al*, 2011), are known to have pursued biological weapons. Offensive biological weapon development is banned by the 1972 Biological Weapons Convention (United Nations, 1972), but several states have in the past nevertheless clandestinely pursued such weapons illegally, including Iraq under Saddam Hussein (United Nations Monitoring, Verification and Inspection Commission, 2007), apartheid South Africa (Gould & Folb, 2002), and the Soviet Union (Leitenberg & Zilinskas, 2012). The possibility that similar actors might pursue mirror bacteria as potential weapons is deeply concerning. Though any danger posed by mirror

bacteria would likely cause indiscriminate harm to humans and natural ecosystems, they might still be pursued for blackmail by rogue governments or by non-state groups that see such mass harm as intrinsically desirable. Given that the misuse of mirror bacteria could result in especially grave consequences for humans and many ecosystems, mirror biogenesis would require robust biosecurity protection. Unfortunately, devising a form of biocontainment that could resist deliberate jailbreaking attempts by skilled individuals with access to mirror DNA appears exceptionally difficult.

Any bacterial auxotrophy resulting from missing or conditionally functional genes can be removed by adding easily identified functional versions of those genes through bacterial transformation. Therefore, a person with moderate molecular biology training could release bacteria from most forms of auxotrophy. Given access to mirror DNA—which could be generated from commercially available mirror phosphoramidites using a standard nucleic acid synthesizer—they could achieve the same outcome for a mirror bacterium. Depending on the mirror bacterial species, the type of biocontainment implemented, and the skill of the practitioner, it might take weeks or even months to remove multiple auxotrophies, especially if many genome modifications are required. Still, since the skill and resource requirements are limited, many actors could remove biocontainment measures with relative ease. More elaborate biocontainment strategies, such as synthetic auxotrophy for an unnatural amino acid in a recoded mirror bacterium would be more complex to reverse, requiring the replacement of every essential gene rendered dependent upon a non-canonical amino acid, but could still be disabled by a sufficiently resourced actor. Alternatively, if a known biosynthetic pathway for the non-canonical amino acid is available or newly discovered, it could be engineered into the mirror bacterium (Butler *et al.*, 2023).

Even if robust biocontainment strategies could make jailbreaking mirror bacteria difficult, several important asymmetries between attack and defense make achieving security through such measures challenging. Defenders must block every possible pathway by which extant mirror bacteria could be weaponized, while malicious actors simply need a single viable path to weaponization. A single mistake could create an unpatchable hole. Moreover, defenders need to create mirror cells that remain safe indefinitely, regardless of how other biotechnologies develop. Feats that seem impossible now might become routine in coming years and decades, so mirror bacteria that appear robustly biosecure today may not remain so forever—especially given the potential for malicious actors to directly generate weaponized mirror bacteria by following established protocols for mirror biogenesis.

Given that even fragile or rigorously biocontained mirror bacteria could be weaponized by malicious actors, preventing their access to mirror bacteria would be imperative. Existing access controls for dangerous pathogens are imperfect, and given the potential for far more severe consequences from an intentional mirror bacterial release, any access controls on mirror bacteria would need to be very restrictive. Further analysis of potential access control measures and whether they could be feasibly implemented across jurisdictions where mirror bacterial research might occur, is beyond the scope of this technical report.

### **Chapter 4: Risks to Human Health**

Ryan K. Duncombe<sup>1</sup>, Timothy W. Hand<sup>2</sup>, Thomas V. Inglesby<sup>3,4,5</sup>, Gregory Lewis<sup>6</sup>, Ruslan Medzhitov<sup>7,8,9,10</sup>, Matthew L. Nicotra<sup>3</sup>, Jaspreet Pannu<sup>3,11,12</sup>, David A. Relman<sup>12,13,14</sup>, Johanna M. Sweere<sup>1</sup>

The previous chapters described the potential pathways by which mirror bacteria could be created and modified. The remainder of the report focuses on the potential risks that such mirror bacteria could pose to humans, animals, plants, and the environment. This chapter focuses on risks to human health, while the next chapter discusses potential medical countermeasures that could protect humans from mirror bacteria.

Humans have evolved complex immune mechanisms to combat bacterial pathogens; many, though not all, of these mechanisms appear unlikely to function properly against mirror bacteria. If a generalist mirror bacterium as described in Chapters 2 and 3 entered the human body, the initial infection would likely be extracellular because it would lack the adaptations typically needed to survive and proliferate as an intracellular bacterial pathogen. For this reason, we focus on immune mechanisms relevant to extracellular bacteria throughout this chapter.

Section 4.1 describes pathogen recognition, which generally relies on stereospecific interactions between immune receptors and specific microbial macromolecules. The macromolecules within mirror bacteria would likely be unable to properly trigger these receptors, and this could greatly impair the immune response.

Section 4.2 describes the innate immune effectors that target pathogens. Mirror bacteria would likely be partially or fully resistant to many of these effectors, including antimicrobial enzymes, professional phagocytes, and the complement system.

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup>Mirror Bacteria Dialogues Fund, New York, New York, USA; <sup>2</sup>Pediatrics Department, Infectious Disease Section, UPMC Children's Hospital of Pittsburgh, University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, USA; <sup>3</sup>Center for Health Security, Bloomberg School of Public Health, Johns Hopkins University, Baltimore, Maryland, USA; <sup>4</sup>Environmental Health and Engineering, Johns Hopkins Bloomberg School of Public Health, Baltimore, Maryland, USA; <sup>5</sup>Johns Hopkins School of Medicine, Baltimore, Maryland, USA; <sup>6</sup>Media Laboratory, Massachusetts Institute of Technology; Cambridge, Massachusetts, USA. <sup>7</sup>Department of Immunobiology, Yale University School of Medicine, New Haven, Connecticut, USA; <sup>8</sup>Howard Hughes Medical Institute, Chevy Chase, Maryland, USA; <sup>9</sup>Center for Infection and Immunity, Yale School of Medicine, New Haven, Connecticut, USA; <sup>10</sup>Tananbaum Center for Theoretical and Analytical Human Biology, Yale School of Medicine, New Haven, Connecticut, USA; <sup>11</sup>Department of Health Policy, Stanford School of Medicine, Stanford University, Stanford, California, USA; <sup>12</sup>Department of Medicine, Stanford School of Medicine, Stanford, California, USA; <sup>13</sup>Department of Microbiology and Immunology, Stanford University School of Medicine, Stanford, California, USA; <sup>14</sup>Infectious Diseases Section, Veterans Affairs Palo Alto Health Care System, Palo Alto, California, USA Section 4.3 describes the adaptive immune response, which could also be significantly weakened. In particular, it is plausible that deficiencies in dendritic cell activation and antigen presentation would result in a failure to activate T cells, which would inhibit both cellular and humoral (antibody) immune responses.

As discussed throughout Sections 4.1–4.3, humans with even partial immunodeficiencies are often at greatly heightened risk of serious disease or death from bacterial infection, including from bacteria that are not usually pathogenic for healthy adults. Similarly, natural specialized pathogens commonly exhibit high degrees of immune evasion in order to establish infection. The unique biology of mirror bacteria could allow them to evade immune mechanisms to a possibly unparalleled extent. Natural pathogens typically utilize virulence factors—bacterial proteins that interact with host ligands to facilitate bacterial adhesion or invasion, or that can directly induce damage. Mirror bacteria would not, absent deliberate engineering, possess virulence factors, and most virulence factors copied from natural-chirality pathogens would not function due to chirality incompatibilities with host ligands. But even in the absence of functional pathogenic virulence factors, their capacity for immune evasion raises the possibility that a mirror *Escherichia coli* or similar mirror generalist bacteria could act as a potentially lethal human pathogen.

Section 4.4 describes the mechanisms by which mirror bacteria could enter the human body without the need for functional virulence factors. Human barrier defenses are imperfect and commensal bacteria can be passively translocated across physical barriers into the bloodstream in healthy individuals. Similar translocations would likely occur with sufficient exposure to mirror bacteria, although it is very uncertain how many mirror bacteria would be needed to establish an infection. Mirror bacteria would lack specialized adhesion mechanisms or other specialized mechanisms that facilitate invasion. This could impair but not necessarily preclude colonization of human barrier surfaces like the lung, and adhesion does not appear necessary for infection or translocation into other bodily regions.

Section 4.5 describes the potential consequences of mirror bacteria within the bloodstream. Given that nutrients and abiotic conditions within blood are plausibly growth-permissive to a generalist mirror bacterium and that the normal immunological clearance mechanisms are likely significantly impaired, it is plausible that mirror bacteria within the blood could grow exponentially. Both the reduced immunological recognition and absence of virulence factors could result in a disease progression that is difficult to predict but likely highly unusual. Nevertheless, it is plausible that at high mirror bacterial concentrations, residual immunological recognition would lead to a lethal sepsis-like hyperinflammatory response. Even if this is avoided, other consequences of infection could potentially be fatal or otherwise highly detrimental to human health.

In summary, sufficient exposure to mirror bacteria may be a life-threatening event, even if the mirror bacteria have no particular adaptations or virulence factors for human infection (Figure 4.1). Deliberate misuse could increase this danger further.



#### Figure 4.1: The human health risks of mirror bacteria

Mirror bacteria that encounter humans may be able to survive and potentially colonize boundary surfaces like the skin and gut, aided by their reversed chirality granting resistance to defenses such as secreted antimicrobial peptides (AMPs), antimicrobial enzymes, and immunoglobulin A (IgA). Mirror bacteria at these surfaces may translocate into the body through transient breaches in physical barriers, which occur frequently from a variety of causes. The reversed chirality of mirror bacteria may prevent the innate immune system from detecting mirror microbe-associated molecular patterns (MAMPs) and the adaptive immune system from processing and presenting mirror peptide antigens via major histocompatibility complex (MHC) molecules. Downstream defenses, such as complement, phagocytosis, and antibody production would also be impaired. Since mirror bacteria may not be effectively cleared by the immune system, this could allow entry into the bloodstream and unchecked proliferation therein.

# 4.1 Innate immune detection of mirror bacteria could be significantly impaired

### Most pattern recognition receptors could have severely impaired ability to recognize mirror versions of their ligands

Pathogen recognition is the first step required for an effective immune response. Bacterial pathogens are typically first detected by macrophages and dendritic cells present in submucosal tissues, such as alveolar macrophages in the lung or Langerhans and interstitial dendritic cells in the skin<sup>4</sup>. These immune cells express pattern recognition receptors (PRRs) that bind to microbe-associated molecular

<sup>&</sup>lt;sup>4</sup> There are many other immune cell subsets within major lineages, often resident in specific tissues, that are too numerous to cover in this report. Therefore, we have limited discussion to responses across major cellular lineages, with the exception of certain cell subsets that appear particularly important for understanding immunological responses to mirror bacteria.

patterns (MAMPs). MAMPs are molecules that are highly conserved across a broad range of microbes but otherwise absent in animals, making them reliable indicators of potential pathogens<sup>5</sup>. The binding of MAMPs by PRRs is a critical step in the activation of both the innate and the adaptive immune systems. PRR engagement activates macrophages, causing them to become highly phagocytic and secrete cytokines that induce inflammation, leading to an influx of neutrophils to the infection site. Dendritic cells are also activated when their PRRs engage MAMPs. Activation causes these professional phagocytes to increase antigen uptake and migrate to lymphoid tissues, where they present antigens to T cells to initiate adaptive immunity.

Almost all known bacterial MAMPs detected by human PRRs are chiral. Standard PRRs such as Toll-like receptors (TLRs), for example, bind to chiral molecules such as lipopolysaccharide (LPS), proteins, peptidoglycan, various lipoproteins, and CpG DNA (Kawai & Akira, 2010). Other key PRRs, including members of the nucleotide-binding domain and leucine-rich repeat containing (NLR) family, bind to flagellin, proteins from bacterial secretion systems, and peptidoglycan degradation products (Muñoz-Wolf & Lavelle, 2016), which are also chiral. Formyl peptide receptors, which recognize chiral f-met peptides from bacteria<sup>6</sup>, enable phagocytes—especially neutrophils—to home to the site of infection (He & Ye, 2017). In addition, chiral molecules play a dominant role in non-canonical immunosurveillance mechanisms. For example, the semi-invariant T cell receptor (TCR) in mucosal-associated invariant T cells recognizes transient chiral intermediates of bacterial riboflavin metabolism (Corbett *et al.*, 2014). Because protein interactions are stereospecific (see Chapter 1), we would expect binding between many chiral PRRs and the mirror-images of these MAMPs to be significantly impaired (Figure 4.2).

There might be some PRRs that could plausibly bind mirror MAMPs, particularly at high MAMP concentration, or for repetitive ligands where avidity effects could amplify weak residual interactions. For instance, high concentrations of mirror dsRNA appear to activate the PRRs TLR3 and TLR7 (Yu & Sczepanski, 2023), though dsRNA is typically a viral MAMP and so its potential relevance for a mirror bacterial infection is unclear. In addition, scavenger receptors on macrophages can bind to many different kinds of substances, including abiotic ones (Alquraini & El Khoury, 2020), and may be capable of recognizing certain mirror macromolecules.

Lipid antigens may potentially be less sensitive to changes in chirality than most of the MAMPs discussed above. For example, TLR2 is unusually promiscuous in its binding of diacylated and triacylated lipopeptides found in bacterial membranes. Most contacts between TLR2 and its ligands form with the achiral acyl chains of the lipopeptides, though some bonds are dependent on the chiral glycerol backbone carbon and the chiral amino acids in the peptide chain (M. S. Jin *et al.*, 2007). Lipopeptides with reversed chirality at some of these carbons are able to activate TLR2, albeit with greatly reduced potency. It remains unclear if fully mirrored lipopeptides could activate TLR2

<sup>&</sup>lt;sup>5</sup> The molecules sensed by PRRs are often referred to as "pathogen associated molecular patterns" or PAMPs (Janeway, 1989). In this report, we use MAMP to acknowledge that these molecules are found in pathogenic and non-pathogenic microbes alike.

<sup>&</sup>lt;sup>6</sup> Bacterial proteins start with formyl methionine (f-met), but eukaryotic proteins do not. This makes f-met an effective MAMP.



#### Figure 4.2: Innate immune recognition of mirror bacteria could be severely impaired

**A.** Natural-chirality bacteria produce MAMPs, which are conserved molecules that are recognized by various PRRs, including TLRs and NLRs. These receptors are present on the cell surface, in endosomes, and in the cytosol. PRR activation triggers the upregulation of innate immune response genes, which results in increased phagocytosis by immune cells, increased cytokine and chemokine expression, and many other responses. **B.** Mirror bacteria would produce mirrored versions of these MAMPs. PRR-mediated detection of mirror MAMPs is expected to be severely impaired because almost all known bacterial MAMPs detected by human PRRs are chiral. Impaired PRR activation would likely result in reduced or failed induction of innate immune response genes.

signaling (Metzger *et al.*, 1991; Takeuchi *et al.*, 2000). Similarly, LPS is detected by TLR4 and MD2 via interactions with the acyl chains and phosphate groups of the lipid A component. Lipid A is chiral, but its long, flexible acyl chains that mediate a majority of the contacts with MD2 and TLR4 are achiral (B. S. Park *et al.*, 2009). Experimental data with enantiomeric LPS is lacking, so it is ultimately unclear how sensitive LPS binding to TLR4/MD2 would be to changes in chirality. Thus, while most PRRs would likely have severely impaired ability to recognize mirror MAMPs, uncertainty remains around how much residual PRR activation could be triggered by a mirror

bacterial infection, particularly at high concentrations of MAMPs. Empirical evidence from laboratory experiments using mirror MAMPs or computational modeling could help address these uncertainties.

While critical PRR pathways would likely be compromised by mirror biology, we are aware of at least one achiral MAMP: the bacterial molecule (E)-4-Hydroxy-3-methyl-but-2-enyl pyrophosphate (HMBPP). HMBPP is typically produced as intermediate in bacterial metabolism (Boucher & Doolittle, 2000), and some bacteria use an alternative metabolic pathway which does not produce HMBPP (Heuston *et al.*, 2012; C. Wang *et al.*, 2017; R. Yoshida *et al.*, 2020). When released by intracellular bacteria, HMBPP binds to the cytoplasmic domain of a membrane protein called butyrophilin, inducing a conformational change that allows it to dimerize with a second butyrophilin protein and be detected by  $V\gamma9/V\delta2$  T cells, a class of cells which bridge innate and adaptive immunity (Uldrich *et al.*, 2020). It is possible that other achiral MAMPs may be discovered as the human PRR repertoire is further characterized. It seems unlikely, however, that HMBPP recognition or other, unknown, achiral MAMPs would themselves suffice to produce an effective innate immune response.

Alongside MAMPs, elements of the innate immune system can be triggered when PRRs and other receptors detect damage-associated molecular patterns (DAMPs) generated by host-cell damage and tissue disruption (Gong *et al.*, 2020). As DAMPs arise from the natural-chirality host rather than from pathogen particles, DAMP detection would function normally. Given mirror bacteria would generally lack virulence factors that could directly damage host tissue, it seems unlikely that significant host-cell damage would arise in the early stages of a mirror bacterial infection unless damage-causing virulence factors were intentionally engineered into the bacterium. At the relatively late stages of infection where tissue damage would probably occur, it appears unlikely that DAMP-mediated inflammation alone would impede bacterial proliferation. DAMPs are not sufficient to combat bacterial infections in PRR-deficient animal models or patients, as described below. As such, DAMP-induced inflammation may instead exacerbate harms to the patient (see Section 4.5).

## Compromised immune detection of mirror bacteria could result in vulnerability to infection

The expected inability of much of the human PRR repertoire to respond to mirror ligands suggests that the innate immune system's ability to detect mirror bacteria would likely be markedly impaired. Evidence from experiments, analogous human immunodeficiencies, and the clinical consequences of PRR masking observed in natural pathogens suggests this loss would result in substantial vulnerability to mirror bacterial infection.

Various natural-chirality bacterial pathogens have evolved mechanisms to shield surface MAMPs from detection by their corresponding PRRs. For example, certain *Helicobacter pylori* strains express tri- or tetra-acylated LPS instead of hexa-acylated LPS, which are less immunostimulatory (Chmiela *et al.*, 2014; Tran *et al.*, 2005). *Salmonella* and *Yersinia pestis* are both able to alter their LPS composition in order to make them less stimulatory during infection (Kawasaki *et al.*, 2004; Telepnev *et al.*, 2009). Various *Proteobacteria*, including some strains of *Campylobacter jejuni* and *H. pylori*,

express flagellin that is significantly less able to stimulate TLR5 signaling than flagellin from other bacterial species (Andersen-Nissen *et al.*, 2005). Many commensal bacteria also express flagellins that weakly stimulate TRL5, which facilitates their persistence in the human gut (Clasen *et al.*, 2023). These modifications, and others, have been found to enhance the ability of bacteria to evade the immune system and subsequently survive (Arpaia & Barton, 2013).

A range of mouse immunity studies show that mice with deletions in multiple PRRs are much more likely to die from a bacterial infection than healthy controls. For example, mice unable to express any TLRs succumb to bacterial infections that wild-type mice can clear (Sivick *et al.*, 2014). Similarly, mice lacking the NLRs NOD1 and NOD2 (which recognize bacterial peptidoglycans) also die from *E. coli* infections that wild-type mice can survive (Y.-G. Kim *et al.*, 2008). Finally, mice lacking formyl peptide receptors are more susceptible to bacterial infections than their wild-type counterparts (Gao *et al.*, 1999; Liu *et al.*, 2012).

Additionally, studies of mice with deficiencies in individual PRRs show the impact even one receptor failure can have on health. Mice lacking individual TLRs that recognize bacterial MAMPs carry increased bacterial burdens and are less likely to survive infection (Bhan *et al.*, 2008; Feng *et al.*, 2003; Sjölinder *et al.*, 2008; Takeuchi *et al.*, 2000; Torres *et al.*, 2004). Similarly, mice lacking a single Nod-like receptor, NOD2, are much more vulnerable to bacterial infections than their wild-type counterparts (Deshmukh *et al.*, 2009; Divangahi *et al.*, 2008; K. Shimada *et al.*, 2009).

The closest human analog to such a deficit in PRR function is the severe inherited defects across PRR signaling in those with IRAK-4 or MyD88 deficiency. Both IRAK-4 and MyD88 are downstream of and essential for most TLR signaling, and thus their deficiency disables much of the human PRR repertoire (Picard *et al.*, 2003; von Bernuth *et al.*, 2008). Individuals with either condition are extremely susceptible to pyogenic (pus-producing) bacteria, and often experience multiple life-threatening invasive infections during the first years of life (Picard *et al.*, 2010). Roughly 40% of known cases die before reaching adulthood despite treatment for invasive infections and intensive use of vaccination, antibiotic prophylaxis, and immunoglobulin infusions (Picard *et al.*, 2010). Survival dramatically improves from the second decade onwards, and both the frequency and severity of infections fade. This is hypothesized to be due to the maturation of other innate and adaptive immune mechanisms which can compensate for the lack of PRR signaling (Ku *et al.*, 2007). However, as discussed throughout this chapter, these compensatory immune mechanisms are also unlikely to properly function against mirror bacteria.

IRAK-4 and MyD88 deficiencies do not perfectly mimic the mechanistic deficits that would be expected in response to mirror bacteria. For example, these deficiencies also impair immune signaling pathways other than TLR signaling, while leaving intact pathways that may be impaired against mirror bacteria (Picard *et al.*, 2011). Nonetheless, evidence from both human immunodeficiencies and mouse models suggests that healthy individuals could be vulnerable to mirror bacterial challenge due to the limited ability of the innate immune system to detect mirror bacteria.

# 4.2 Mirror bacteria would likely be resistant to most innate immune responses

Detection of an infection by the immune system triggers a variety of responses that act to eliminate the invading microbes. In humans, these mechanisms include the secretion of antimicrobial proteins and peptides, the complement system, and phagocytosis. Activation of many of these mechanisms would likely be severely impaired because they are downstream of host receptors. Moreover, the molecular mechanisms of these responses, much like those for detection, often rely on stereospecific interactions between host and pathogen. Therefore, the reversed chirality of mirror bacteria would likely confer a high degree of resistance to these immune responses.

Some antimicrobial peptides could retain the ability to kill mirror bacteria, but their release would likely be impaired by failures of innate immune recognition

Antimicrobial peptides (AMPs) are a crucial component of the human innate immune system, serving as endogenous antibiotics that provide a first line of defense against invading pathogens. Humans express more than 100 antimicrobial peptides, many of which have not been characterized in great detail (G. Wang, 2014), and a detailed review of their functions is beyond the scope of this report.

Most AMPs can disrupt bacterial cell membranes via electrostatic and amphipathic interactions, which either destroy osmotic potential, create pores, or act in a detergent-like manner to break apart the phospholipid bilayer (Brogden, 2005). Such mechanisms are insensitive to chirality and should be able to target mirror bacterial membranes; this has previously been confirmed in experiments. For instance, the D-enantiomer of defensin HNP4 has similar antimicrobial properties to its natural counterpart (Wei *et al.*, 2009), while the D-enantiomer of cathelicidin LL-37 has similar or somewhat reduced efficacy (Dean *et al.*, 2011a, 2011b), suggesting both should retain potency against mirror bacteria.

Many AMPs, however, are pleiotropic in effect, acting via both generic membrane disruption and specific chiral targets. For instance, the human defensins HNP1 and hBD-3 both block the synthesis of peptidoglycan in the bacterial cell wall by binding to a chiral precursor called lipid II (de Leeuw *et al.*, 2010; Sass *et al.*, 2010). The D-enantiomer of HNP1 is significantly less bactericidal than natural HNP1 against *Staphylococcus aureus*, but not *E. coli* (Wei *et al.*, 2009), which implies that HNP1 would retain efficacy against a mirror *E. coli* but not a mirror *S. aureus*. By testing D-enantiomers of other AMPs against common bacteria, it should be possible to ascertain which AMPs will retain efficacy against mirror bacteria.

However, given the likely defects in innate immune detection, it is plausible that AMPs would not be present in sufficient concentrations to be effective against mirror bacteria in most extracellular environments. Because many AMPs have toxic effects on mammalian membranes, their expression and secretion are generally tightly controlled and often rely on MAMPs, inflammatory cytokines, or other immunological signals (Gallo & Hooper, 2012). Impaired innate immune detection, described in Section 4.1, and defects in professional phagocyte and T cell activation, described below, could

therefore significantly reduce AMP concentrations in extracellular spaces. While some constitutive production and secretion does occur, particularly at mucosal surfaces (Laube *et al.*, 2006; Muniz *et al.*, 2012; Yarbrough *et al.*, 2015), the fact that constitutively expressed AMPs do not appear to compensate for the loss of PRR signaling in humans with MyD88/IRAK4 or other immunodeficiencies suggests impaired innate immune recognition could substantially limit the overall impact of AMPs against mirror bacteria.

# Antibacterial enzymes would likely have greatly reduced activity against mirrored substrates

Some antimicrobial proteins are enzymes that cleave bacterial molecules. Because enzyme catalysis is stereospecific, it is generally anticipated that many of these enzymes would have little activity against mirror molecules (Uppalapati *et al.*, 2016). For example, experimental evidence indicates that p-peptides are partially or completely resistant to degradation by representatives from several of the major classes of proteases, including carboxypeptidase A, papain, pepsin, trypsin, elastase, and chymotrypsin (Guichard *et al.*, 1994; Miller *et al.*, 1995; Vaissière *et al.*, 2017). Limited evidence also suggests that cathepsins are unable to efficiently cleave p-peptides (Chu *et al.*, 2012; Meldal *et al.*, 1998). This protease resistance has led to the development of p-peptides as a therapeutic drug class due to their increased half-life (Lander *et al.*, 2023; Uppalapati *et al.*, 2016).

The phospholipase  $A_2$  (PLA<sub>2</sub>) family is another class of antibacterial enzymes. These enzymes hydrolyze the sn-2 position of bacterial phospholipids present in cell walls, resulting in bacterial cell death (Murakami *et al.*, 2016). Two secreted PLA<sub>2</sub> (sPLA<sub>2</sub>) enzymes, PLA<sub>2</sub>G2A and PLA<sub>2</sub>G5, are upregulated and expressed in phagocytes and other cells after pathogen stimulation (Dabral & van den Bogaart, 2021). In addition to killing bacteria, the fatty acids and lysophospholipids they produce also trigger innate immunity in another positive feedback loop (Murakami *et al.*, 2016). Given that sPLA<sub>2</sub> interacts with phospholipids in a chiral manner, mirror bacteria would likely be resistant to this mechanism: porcine and bee venom PLA<sub>2</sub> do not hydrolyze mirror-image phospholipids (Gudmand *et al.*, 2010; Singh *et al.*, 2010).

Lysozyme is an important antibacterial enzyme that cleaves peptidoglycan, the major structural component of the cell wall that protects bacteria from osmotic and environmental stressors (Callewaert & Michiels, 2010). Lysozyme is expressed at high concentrations in the mucosal epithelia, where it kills extracellular bacteria, and in the phagolysosomes of macrophages and neutrophils (Ragland & Criss, 2017). The peptidoglycan cleaved by lysozyme triggers innate immune responses through activation of extracellular and intracellular PRRs, initiating a positive feedback loop.

Lysozyme would likely fail to cleave mirror peptidoglycan. Although this has not been assessed experimentally, lysozyme's catalytic mechanism is well studied (Callewaert & Michiels, 2010; Ragland & Criss, 2017), and it seems unlikely that its active site could accommodate mirror peptidoglycan in a configuration that would permit hydrolysis. However, catalytically-disabled lysozyme still retains bactericidal activity through disrupting membranes (Derde *et al.*, 2013; Ibrahim *et al.*, 2001; Nash *et al.*, 2006), similar to the action of many AMPs described previously. It is

therefore plausible that this bactericidal activity would function against mirror bacteria, though lysozyme has a strong affinity for lipopolysaccharide (Derde *et al.*, 2015) and it is not clear whether lysozyme would be as effective at disrupting the outer membranes of a mirror Gram-negative bacteria. To our knowledge, mirror-image lysozyme—which could be used to definitively answer these questions—has not been chemically synthesized.

Knockout mice lacking lysozyme-M (one of two murine lysozyme proteins) have respiratory tracts with dramatically elevated levels of resident bacteria<sup>7</sup> and show defects in killing and eliminating bacterial infections (Cole *et al.*, 2005; Davis *et al.*, 2011; Ganz *et al.*, 2003; Markart *et al.*, 2004; Nash *et al.*, 2006; J. Shimada *et al.*, 2008). Many human commensals and pathogens have also evolved to evade lysozyme's catalytic activity, either by modifying peptidoglycan to reduce binding (Ragland & Criss, 2017; Vollmer & Tomasz, 2000) or by producing proteinaceous inhibitors (Clarke *et al.*, 2010). This further underscores the importance of this catalytic activity as an immune defense.

Neutrophil granules also contain a number of other enzymes that act directly on bacterial components. These include serine proteases, cysteine proteases, neutrophil gelatinase-associated lipocalin, proteinase 3, cathepsin G, neutrophil elastase, CAP37, and NSP4 (Cowland & Borregaard, 2016). These would not be expected to properly function against mirror bacteria, which could markedly decrease the effectiveness of neutrophil-mediated bacterial killing.

#### Complement activity versus mirror bacteria could be partially impaired

The complement system is a set of blood proteins that, upon activation, produce proteins that can stimulate immune cells or attach to (opsonize) pathogens to aid their consumption by phagocytes. It can also directly harm Gram-negative bacteria by self-assembling into the membrane attack complex (MAC), which creates holes in the bacterial cell membrane, causing the cell to burst (Dunkelberger & Song, 2010). Complement system proteins are consistently present in the blood, and *in vitro* studies suggest complement can target, opsonize, and kill bacteria—either via the MAC or phagocytosis—in a matter of minutes to hours (Heesterbeek *et al.*, 2018; Verbrugh *et al.*, 1979; Zewde *et al.*, 2016). Activation of the complement system occurs through three distinct pathways: the classical, alternative, and lectin pathways. In the context of mirror bacterial challenge, it appears the lectin pathway may be significantly impaired, the classical pathway could retain some activity through natural antibody binding, and the alternative pathway may remain largely intact (Figure 4.3).

The lectin pathway of complement activation is initiated when ficolins or collectins, such as mannose-binding lectin (MBL), recognize and bind to specific carbohydrate patterns on the surface of pathogens (Mastellos *et al.*, 2024). This binding activates MBL-associated serine proteases (MASPs), which then activate the complement system. The specificity and affinity of lectin-carbohydrate bonds differs across lectins and can be very sensitive to the structure of the carbohydrate (Raposo *et al.*, 2021). Therefore, it seems unlikely that human lectins would bind strongly to mirror carbohydrates. As a result, the lectin pathway would likely fail to activate.

<sup>&</sup>lt;sup>7</sup> Markart *et al.* (2004) were able to culture 100–5 000 colony forming units of lactobacilli from lung homogenates of lysozyme-deficient mice, but nothing grew from lung homogenates of co-housed wildtype mice.



#### Figure 4.3: Complement activity against mirror bacteria would likely be impaired

**A.** Natural bacteria trigger three distinct pathways of complement activation. The classical pathway is triggered by circulating antibodies that bind to the surface of a bacterium and are then bound by the complement protein C1q. The lectin pathway is initiated when lectins, such as mannose-binding lectin (MBL), bind to specific carbohydrate patterns on the surfaces of bacteria. The alternative pathway is activated by the spontaneous hydrolysis of C3 in the blood, generating C3b that binds to bacterial surface molecules. All three pathways activate the complement cascade, a series of protein interactions that results in increased inflammation, innate immune activation, opsonization, and formation of the membrane attack complex (MAC). **B.** Mirror bacteria would trigger fewer pathways of complement activation. The classical pathway could be impaired due to lack of circulating IgG, but natural IgM antibodies or phosphocholine could activate C1q if present. The lectin pathway is unlikely to be activated because it depends on the recognition of chiral carbohydrates. The alternative pathway could remain functional because it involves no stereospecific interactions. Impaired complement activation might result in reduced inflammation, innate immune cell activation, opsonization, and formation of MACs.

In the classical pathway, circulating IgM or IgG antibodies bind to the surface of a pathogen and are subsequently bound by a protein complex called C1 (Mastellos et al., 2024). C1 (specifically the component protein C1q) then activates the complement cascade. Production of affinity-matured anti-mirror antibodies that could potently activate the classical pathway is expected to be absent or greatly reduced, but predominantly IgM antigen-specific antibodies might still be elicited through T cell-independent mechanisms (see Section 4.3). Natural IgM antibodies, which are present in circulation prior to antigen encounter and have not undergone affinity maturation to any specific antigen, typically have broad specificity and are able to weakly bind to a range of different antigens (Coutinho et al., 1995; Z.-H. Zhou et al., 2007). When bound to bacterial antigen, natural IgM is able to activate the classical complement pathway, albeit less potently than affinity-matured antibodies (Boes et al., 1998; Z.-H. Zhou et al., 2007). As such, natural IgM could plausibly play a role during early stages of mirror bacterial infection. Complement activation through natural IgM has been shown to confer resistance to bacterial infection in mice (Islam et al., 2023; Mold et al., 2002). However, it is uncertain, whether natural IgM would be able to bind mirror antigens, as some murine studies suggest that natural IgM polyreactivity is restricted to phylogenetically conserved microbial structures (Hardy & Hayakawa, 1994; Kantor & Herzenberg, 1993). If natural IgM antibodies indeed predominantly bind conserved epitopes, it would be less likely they would bind mirror versions of these epitopes. Without anti-mirror antibodies, the classical pathway would be largely unable to function.

The classical pathway can also be activated by C reactive protein (CRP) if it binds to phosphocholine in cell membranes and is subsequently bound by C1q (Thompson et al., 1999). However, although phosphocholine is achiral, it is absent from an estimated 85% of bacteria, including E. coli (Geiger et al., 2013; Kleetz et al., 2021). Moreover, even if phosphocholine were present on the surfaces of mirror bacteria, it is unclear whether CRP would be present at sufficiently high concentrations to alter the course of a mirror bacterial infection. Although CRP is constitutively produced by the liver, its concentration in the blood increases a thousand-fold during infection (from  $1 \mu g/mL$  to 1000 µg/mL) in response to the inflammatory cytokines IL-6, IL16, and TNF (Del Giudice & Gangestad, 2018; Ji et al., 2023). Both IL1ß and TNF are primarily produced by activated macrophages (Ott et al., 2007), and as discussed in the previous subsection, it appears unlikely that significant macrophage activation would occur in response to mirror bacteria. IL-6, meanwhile, can be produced by epithelial cells in response to both MAMPs and DAMPs (Krueger et al., 1991), so it is possible that CRP concentrations could increase given sufficient inflammation during later stages of infection. If this occurred, a substantial amount of CRP might bind to phosphocholine (if present on mirror bacteria), and this could potentially preserve some functionality of the classical pathway. However, the upstream failures of pathogen recognition appear to be a significant obstacle to enhanced CRP production, and it is unclear whether low constitutive CRP levels would be sufficient to activate the classical pathway such that it would contribute to anti-mirror immunity in a meaningful way.

The alternative complement pathway could function against mirror bacteria because it is initiated by the spontaneous production of C3b, which is highly reactive and will covalently bond with any nucleophilic group (Law & Dodds, 1997). Factor B then binds to C3b and is cleaved to form a

C3bBb complex, which in turn can cleave C3 to form more C3b, so that the alternative complement activation can self-amplify (Merle *et al.*, 2015). The C3bBb complex is short-lived, however, unless stabilized by the protein properdin, which is constitutively produced in blood plasma by leukocytes but also stored and released by neutrophils in response to inflammatory agonists (J. Y. Chen *et al.*, 2018).

To summarize, complement activation would likely be impaired due to disruptions to both the lectin and classical activation pathway, but the alternative pathway could remain functional and this could allow some activation to occur. While many pathogens have evolved diverse strategies to avoid complement activation, such as attracting host complement regulators, enzymatically cleaving complement components, or mimicking host surfaces (Serruto *et al.*, 2010), mirror bacteria would not by default have functional versions of these mechanisms. Therefore, one would expect C3b to opsonize mirror bacteria for phagocytosis, and for C3a and C5a (which are produced downstream of C3b) to be released and induce inflammation.

In the last stages of complement activation, C5b interacts with additional complement proteins to form a membrane attack complex (MAC) which is capable of rupturing Gram-negative cell envelopes (Doorduijn *et al.*, 2019). This process is not known to depend on any specific surface receptors, so it is likely that the MAC could correctly assemble and target mirror Gram-negative bacteria. Gram-positive bacteria are resistant to lysis by MAC due to their thick peptidoglycan layer (Berends *et al.*, 2013; Joiner *et al.*, 1984).

It is difficult to estimate how the impairments in complement functioning described above would ultimately impact the ability of mirror bacteria to establish infection, especially in the context of other expected failures of the immune response. There are a number of inherited diseases of the human complement system that could be informative. If we assume that the alternative pathway would function during a mirror bacterial infection, but the lectin pathway and possibly the classical pathway would be impaired, then the most analogous human deficiency would be that caused by deficiencies in C2 or C4, which are shared by the latter two pathways (but not the alternative pathway). Both of these deficiencies are rare and incompletely characterized, but their primary clinical features are an increased susceptibility to bacterial infections, particularly to encapsulated bacteria. These features are broadly shared with other complement deficiencies, albeit with variations in autoimmune manifestations and to which bacterial species individuals are most susceptible (Pettigrew *et al.*, 2009). Quantifying the degree of increased susceptibility to bacterial infection in C2 and C4 deficiencies is difficult given data scarcity.

#### Phagocytosis of mirror bacteria would likely be compromised

Phagocytosis is the process by which innate immune cells capture, kill, and clear bacteria (Gordon, 2016). The first phagocytic cells to encounter mirror bacteria would likely be tissue resident macrophages and dendritic cells. At rest, these cells exhibit a low level of phagocytosis, but they become highly phagocytic upon activation (Mosser & Edwards, 2008). As discussed above, macrophages and dendritic cells are unlikely to be activated by mirror MAMPs, thereby reducing the number of mirror bacteria these cells would attempt to phagocytose.

#### Chapter 4: Risks to Human Health

If macrophages fail to activate, this could also limit the number of neutrophils and monocytes that respond to mirror bacteria. Neutrophils, which play an essential role in the clearance of many bacteria as the most phagocytic innate immune cells, respond to inflammation by entering the tissues and then following gradients of cytokines, chemokines, and MAMPs to the infection site (Burn *et al.*, 2021). Without these signals, many of which are produced by activated macrophages, neutrophils may face great difficulties in finding mirror bacteria. Like neutrophils, monocytes would also likely fail to respond to a mirror bacterial infection due to reduced inflammation and impaired ability to sense MAMPs. This could hinder both entry into infected tissues and differentiation into new macrophages and dendritic cells (Jakubzick *et al.*, 2017).

Overall, we expect few phagocytes to respond to mirror bacteria at the beginning of an infection. This could change at later points if the release of DAMPs from necrotic cells combined with non-specific inflammation is sufficient to activate macrophages and dendritic cells and recruit neutrophils and monocytes to the infection site. Accumulation of the soluble complement factor C5a, which can induce inflammation, activate phagocytes, and promote neutrophil chemotaxis, may also improve the likelihood of a phagocyte response (Guo & Ward, 2005), although it is unclear how these functions would be affected by the lack of PRR signaling.

Regardless of their activation status, the reversed chirality of mirror bacteria would likely make it difficult for phagocytes to internalize them. Phagocytosis is an active, receptor-mediated process. Phagocytic receptors bind either directly to bacteria or indirectly via opsonizing complement proteins or antibodies bound to bacteria. These receptors enable phagocytes to grasp the bacterial cells and, importantly, activate signaling pathways that induce the cytoskeletal changes required for internalization (Uribe-Querol & Rosales, 2020). Most phagocytosis receptors are not expected to bind mirror bacteria due to the reversed chirality of their ligands. Even if some receptors did bind, this would not necessarily enable phagocytosis if PRRs or other signals are absent. Scavenger receptors, for instance, have broad specificities and may in theory bind some mirror molecules. But they generally partner with PRRs or Fc receptors to signal (Alguraini & El Khoury, 2020; Fu & Harrison, 2021), and therefore may be unable to facilitate phagocytosis by themselves. Complement receptors would also still bind complement-opsonized mirror bacteria without additional assistance. which might allow some mirror bacteria to be internalized. Again, however, this process would plausibly be less efficient because the affinity of complement receptors for their ligands is typically enhanced by signaling from other phagocytic receptors and PRRs (Uribe-Querol & Rosales, 2020). Macrophages can internalize inanimate particles like latex beads (Collin-Faure et al., 2023) or asbestos (Ishida et al., 2019), suggesting some nonspecific uptake of mirror bacteria may also occur. Still, the overall reduced functionality of phagocytic receptors suggests mirror bacteria could be internalized at lower rates than comparable non-mirror bacteria.

It is also unclear whether phagocytosis would kill those mirror bacteria that are internalized. Once engulfed, mirror bacteria would be enveloped in a new vesicle called the phagosome, which would eventually mature via fusion with lysosomes to form a bactericidal organelle called the phagolysosome (Sivaloganathan & Brynildsen, 2021). Phagosomes containing synthetic beads can mature into phagolysosomes even in resting phagocytes (Alloatti *et al.*, 2015; Dill *et al.*, 2015; Hoffmann *et al.*, 2012; Sanjuan *et al.*, 2007), suggesting internalized mirror bacteria would also

arrive in phagolysosomes, although the absence of MAMPs or inflammatory cytokines might speed or slow phagosome maturation, depending on the context (Pauwels *et al.*, 2017).

Normally, bacteria are killed in phagolysosomes by oxidants that permeabilize the cell membrane, AMPs, or activity of antimicrobial enzymes whose function is enhanced by low phagolysosomal pH (~4.5) (Brown, 2024). As discussed earlier, AMPs that disrupt membranes via electrostatic and amphipathic interactions would likely be unaffected by the reversed chirality of mirror bacteria. Because many of them are constitutively expressed in the lysosomes of phagocytes (M. S. Park *et al.*, 2018), they may also be present at concentrations high enough to damage mirror bacteria. Antimicrobial enzymes, however, would likely not function against mirror bacteria due to incompatible chirality. In addition, many kinds of mirror bacteria, including mirror *E. coli*, could potentially survive the phagolysosomal acidity, given that many natural bacteria can survive in acidic conditions with pH < 4.5 (Baka *et al.*, 2013; Gorden & Small, 1993; Ross *et al.*, 2003).

On the other hand, oxidants like reactive oxygen species (ROS) would remain harmful to mirror bacteria. Although the exact mechanism by which ROS contributes to antibacterial immunity is controversial (Herb & Schramm, 2021), there is some evidence that ROS are sufficient to kill bacteria in phagolysosomes. A recent study shows that the ROS that accumulate in phagocytic vesicles immediately after internalization can kill *Pseudomonas aeruginosa* cells without the aid of antimicrobial enzymes in murine alveolar macrophages (Riazanski *et al.*, 2020). It is not known whether this can be generalized to other phagocytic cells or to other species of bacteria, or whether the oxidant levels in the phagolysosomes of unactivated phagocytes would be sufficient to kill mirror bacteria. Moreover, mirror bacteria could plausibly be protected from low levels of ROS (which are achiral) by the same catalases that naturally protect *E. coli* from oxidative stress (Khademian & Imlay, 2021; Sultana *et al.*, 2022). Thus, it is difficult to determine how effective ROS would be at killing internalized mirror bacteria by themselves. If ROS levels were low enough, it is plausible that many mirror bacteria would survive.

To summarize, the available evidence suggests phagocytosis of mirror bacteria would be less effective compared to phagocytosis of natural bacteria (Figure 4.4). Phagocytes may have difficulty finding mirror bacteria, engulfing them, and killing them. Some mirror bacteria may survive inside phagocytes. Furthermore, even those that are killed would not be properly digested due to incompatibilities between lysosomal enzymes and mirror macromolecules. In either case, phagosome resolution—the process by which phagocytes digest and recycle the contents of phagolysosomes (Mylvaganam & Freeman, 2023)—would probably be disrupted.

While the persistence of live mirror bacteria and mirror bacterial debris inside phagocytes seems likely to impede their function, the implications are unclear. During infections with natural-chirality bacteria, the presence of live intracellular bacteria or intracellular MAMPs can trigger programmed cell death, often in the form of apoptosis, pyroptosis, or necroptosis, which enclose the pathogen in cellular bodies that are processed by phagocytes (Lacey & Miao, 2020). Pyroptosis and necroptosis additionally induce broad-spectrum inflammatory responses that can aid in clearing bacterial



#### Figure 4.4: Mirror bacteria could evade phagocytosis and killing by macrophages

**A.** Natural bacteria trigger a number of processes that enable macrophages to phagocytose and kill them. Macrophages detect microbe associated molecular patterns (MAMPs), including formyl-peptides (Nfor-MLF) and chemotax toward bacteria. Phagocytosis receptors bind bacteria and initiate their internalization. After internalization, phagolysosomal reactive oxygen species (ROS), antimicrobial peptides (AMPs), and enzymes kill phagocytosed bacteria, which are then digested, their nutrients being recycled and waste products exocytosed. **B.** Mirror bacteria would not be expected to induce chemotaxis, binding, or substantial internalization by macrophages because each of these processes relies on macrophage receptors binding chiral proteins or peptides. If a mirror bacterium is internalized, it may or may not be killed in the phagolysosome, as some mechanisms of cell lysis are stereospecific (antimicrobial enzymes, some AMPs) but others (ROS, potentially other AMPs) are not. Digestion of mirror bacteria and nutrient recycling are unlikely to function properly, potentially resulting in macrophage death or decreased function.

infections, but they can also harm the host by continuously releasing DAMPs if not properly regulated (Booty & Bryant, 2022). However, intracellular mirror bacteria might not trigger programmed cell death. Pyroptosis and necroptosis are often initiated through MAMP detection by intracellular PRRs—e.g., TLRs or NLRs (Jorgensen *et al.*, 2017). Necroptosis can also be triggered directly by inflammatory cytokines or bacterial virulence factors (X. Yu *et al.*, 2024), whereas pyroptosis can be initiated by specialized host proteins that detect disruptions in cellular pathways (Lacey & Miao, 2020). Mirror bacteria would likely be resistant to PRR binding, would not have functional virulence factors to directly trigger cell death, and it is unclear whether cellular pathways would be sufficiently disrupted by the persistence of mirror bacteria in phagocytic vesicles. Taken together, it is plausible that mirror bacteria would avoid triggering programmed cell death unless persistence of live bacteria or debris severely interferes with cellular pathways. Ultimately, even if triggered, programmed cell death would likely not facilitate clearance, as the phagocytes taking up the cellular bodies containing mirror bacteria would in turn face difficulties lysing them.

#### The role of innate lymphoid cells in mirror bacterial infection is unclear

Innate lymphoid cells (ILCs)—specifically group three ILCs (ILC3s)—are one additional innate immune cell type known to be activated during bacterial infections. ILC3s are found in lymphoid tissues and near barrier surfaces such as the gut, respiratory tract, and skin (Sonnenberg & Hepworth, 2019) and are an early source of effector cytokines that elicit and support antibacterial immune responses in helper T cells and other cell types (Vivier et al., 2018; von Burg et al., 2014). Most pathways of ILC3 activation seem likely to be impaired in the case of mirror bacteria. The primary drivers of ILC3 activation are inflammatory cytokines secreted by activated phagocytes (W. Zhou & Sonnenberg, 2020), but this pathway may be hindered by the impaired activation of these cells during mirror bacterial infection. Human ILC3s also express TLRs and can be directly activated by MAMPs, but, as discussed earlier, TLR signaling is likely to be significantly impaired in the case of most mirror MAMPs. Although ILC3s can be directly activated by bacterial tryptophan metabolites and components of bacterial cell walls (W. Zhou & Sonnenberg, 2020), these are all chiral and unlikely to be sensed in their mirror form. ILC3s can also directly sense short chain fatty acids (W. Zhou & Sonnenberg, 2020), some of which are achiral, such as acetic acid, propionic acid, and butyric acid, and would likely be sensed if they were secreted by mirror bacteria. The overall impact of ILC3s sensing mirror bacteria via short chain fatty acids is unclear because experiments show that this pathway can either promote or suppress the production of effector cytokines depending on the specific location and receptors involved (Chun et al., 2019; S.-H. Kim et al., 2017). Moreover, short chain fatty acids are produced in abundance by commensal bacteria (Fusco et al., 2023), so it is not obvious that their production by mirror bacteria would alter ILC3 function. Thus, although ILC3s can promote antibacterial responses, whether they would be relevant during a mirror bacterial infection is unclear.

### 4.3 Adaptive immunity to mirror bacteria would likely be impaired

In addition to the innate immune system, humans also possess an adaptive immune system. A key difference between the innate and adaptive systems is that while the innate immune system relies on

invariant PRRs for pathogen detection, the adaptive immune system uses a process of genetic recombination to generate a large and diverse repertoire of variable receptors, each capable of binding to different pathogen-derived molecules. This vast repertoire of adaptive immune receptors can bind a wide array of biological macromolecules (Chiu *et al.*, 2019; Marks & Deane, 2020) and could bind mirror macromolecules. However, B or T cell receptor binding is only one of the components necessary for B and T cell activation. As we describe in the following section, T and B cell activation also depends on a diversity of signals from the innate immune system and, in the majority of cases, the successful presentation of peptide antigens. We expect these to be impaired, and thus we do not expect humans to mount a strong adaptive immune response to mirror bacteria. The likelihood of a weak adaptive immune response would mean the formation of protective memory responses may also be impaired.

#### T cell activation would likely be severely impaired during mirror bacterial infection

The activation of conventional T cells requires a number of steps, most of which could have serious defects in response to a mirror bacterial infection. Adaptive immunity is initiated by dendritic cells from the innate immune system (Iwasaki & Medzhitov, 2015). In a bacterial infection, dendritic cells that have been activated via PRR stimulation increase their uptake of bacteria and then migrate to lymph nodes. There, the dendritic cells present peptides from the bacteria to naive T cells. If there are T cells with TCRs that can recognize the bacterial peptides, they become activated and proliferate, effectively initiating the adaptive immune response. Activated T cells either migrate to the infected tissues, where they kill infected cells and secrete proinflammatory cytokines, or they remain in the lymph node and activate B cells, enabling the latter to make antibodies.

The inability of the innate immune system to detect most, if not all, mirror MAMPs would likely prevent dendritic cells from being activated. This would result in few, if any, dendritic cells trafficking to the lymph nodes and being available to activate T cells—presenting a substantial barrier to mounting an adaptive immune response.

Even if dendritic cells were activated in large numbers, able to phagocytose mirror bacteria, and subsequently migrate to the lymph node, it appears unlikely that they would be able to efficiently process and present mirror peptide antigens to T cells. Antigen presentation involves multiple stereospecific steps (Figure 4.5, left-hand side). In a normal immune response to extracellular bacteria, bacteria-derived proteins taken up by dendritic cells are first cleaved by lysosomal proteases into short (13-25 amino acid) peptides before binding to major histocompatibility complex class II molecules (MHC-II)<sup>8</sup> (Chicz *et al.*, 1992). Peptide:MHC-II complexes on the surface of dendritic cells are presented to the TCRs of naive T cells expressing CD4, a co-receptor for MHC-II. Each CD4<sup>+</sup> T cell bears a unique TCR sequence, endowing it with a distinct structure and the ability to bind specifically to a narrow set of peptide:MHC-II complexes. If this binding takes place with high affinity (and proper co-stimulation through other, non-pathogen-specific receptors expressed on both

<sup>&</sup>lt;sup>8</sup> We focus on MHC-II because we expect mirror bacterial infection to be largely extracellular. Although antigen presentation by MHC-I would suffer analogous defects to MHC-II, this would likely be less relevant during mirror bacterial infection because MHC-I molecules typically present peptides derived from intracellular pathogens.

cells), the T cell is activated, allowing it to proliferate. Each stage of this process would likely face substantial challenges during a mirror bacterial infection.

As discussed in Section 4.2, mirror proteins and mirror peptides resist protease degradation. With respect to antigen processing, an inability to degrade mirror proteins has been demonstrated for two lysosomal proteases that process peptides for MHC-II molecules (Chu *et al.*, 2012; Meldal *et al.*, 1998). Given the available evidence, it appears unlikely that antigen-presenting cells would successfully cleave the vast majority of mirror peptides derived from mirror bacteria, generating few, if any, mirror peptides as candidates for binding to MHC molecules (Figure 4.5, Step 1). This would substantially reduce the ability of dendritic cells to activate T cells. However, it is possible that small peptides could be derived from lysed mirror bacteria directly, as mirror bacterial proteases would be able to cleave mirror proteins.

Even if mirror peptides were available to bind MHC molecules, it seems likely that this step would be relatively inefficient. Mirror peptides of appropriate size and amino acid composition (*vis a vis* regular peptides) tend to bind MHC molecules only weakly, if at all (Bartnes *et al.*, 1997; Hervé *et al.*, 1997; Hill *et al.*, 1994; Lombardi *et al.*, 2020; Mézière *et al.*, 1997). Thus, during a mirror bacterial infection, only a fraction of the available mirror peptides may be expected to bind MHC molecules with sufficient affinity to be presented on the surfaces of dendritic cells in peptide:MHC complexes (Figure 4.5, Steps 2 and 3).

Were mirror-bacteria-derived, high-affinity mirror peptide:MHC complexes to be presented to naive T cells, the extent to which T cells would be activated is unclear. TCR binding to a mirror peptide:MHC complex can involve interactions between the TCR, MHC molecule and both the side chains and backbone of the peptide being presented (Rossjohn *et al.*, 2015) (Figure 4.5, Step 4 and 5). The enormous diversity of TCR receptors and MHC allotypes, with a genetic recombination process in each cell that is capable generating one of 10<sup>18</sup> potential TCRs (Sethna *et al.*, 2019) and with more than 40 000 MHC alleles in the human population (encoding over 26 000 distinct protein sequences) (EBI Web Services, n.d.), provides many potential combinations that could result in functional TCR-peptide-MHC interactions (Heijmans *et al.*, 2020). The available experimental evidence shows that T cells can respond to mirror-peptide:MHC complexes, but the response is generally weaker than to similar peptide:MHC complexes (Bartnes *et al.*, 1997; Nair *et al.*, 2003). Very few peptides have been tested, however, so it is not clear if this is because p-peptides are less able to activate T cells or to be efficiently presented by MHC, or if this is simply an artifact of the handful of examples studied in the literature.

In addition to engagement of the TCR with an MHC-peptide complex, T cells need additional signals to become fully activated, including binding of dendritic cell co-stimulatory molecules CD80 or CD86 to T cell molecule CD28, and activating soluble factors like IL-6, IL-12, or IL-23. These activation factors are largely regulated by PRR signaling, and plausibly would not be significantly upregulated during a mirror bacterial infection (Figure 4.5, Step 6). As such, T cells that can recognize mirror bacterial peptides might not receive sufficient costimulatory signals to survive and differentiate into effector T cells.



#### Figure 4.5: T cell activation is likely to be severely impaired in response to mirror bacterial infection

**A.** T cell activation in response to natural-chirality bacteria is a multi-step process. Activated dendritic cells phagocytose bacteria and degrade their proteins into peptides, many of which are loaded onto major histocompatibility complex (MHC) molecules, forming peptide:MHC complexes that translocate to the cell surface. Peptide:MHC complexes are recognized by T cell receptors (TCRs), resulting in T cell activation. **B.** Many steps of T cell activation could be impaired in responses to mirror bacteria. Mirror bacteria would likely evade phagocytosis by dendritic cells because this action is mediated by stereospecific protein receptors. If mirror bacteria were phagocytosed, the proteases that mediate protein degradation and peptide formation are stereospecific as well, so few mirror peptides would likely be produced (Step 1). If mirror peptides were formed, MHC molecules would likely bind them weakly, if at all (Step 2) and few mirror peptide:MHC complexes would be presented on the surface of the dendritic cell (Step 3). Existing data, although limited, suggests that TCRs might bind mirror peptide:MHC complexes weakly (Step 4). Failure of these steps could weaken TCR signaling (Step 5). Impaired innate immune activation could also result in insufficient co-stimulation and cytokine signaling necessary for T cell proliferation (Step 6). Taken together, T cell activation in response to mirror bacteria would likely be severely impaired.

The combined effect of impaired innate immune activation, protease resistance, weak MHC binding of most mirror peptides, and (potentially) weak T cell recognition is likely to severely impair T cell activation (Figure 4.5). Although it is hard to rule out the possibility that a small subset of mirror peptides could pass through the aforementioned hurdles—the space of mirror peptides is very large—this seems significantly less likely to occur during a mirror bacterial infection compared to a natural bacterial infection. It therefore appears that a mirror bacterial infection could result in a diminished or ineffective T cell response.

Antibacterial responses in peripheral tissues are enhanced by the functions of various types of T cells and other lymphoid cells. Therefore, impaired activation of these cells could weaken the overall immune response to mirror bacteria. Activated helper T cells-particularly Th17 cells-are important for driving a potent antimicrobial response. During infections with natural extracellular bacteria, activated dendritic cells secrete cytokines that cause many newly activated T cells to differentiate into Th17 and other helper T cells (Korn et al., 2009). These T cells then migrate to the site of infection and secrete cytokines that greatly enhance neutrophil recruitment and cause epithelial cells to proliferate, increase the formation and maintenance of tight junctions, and secrete antimicrobial peptides (Lewinsohn et al., 2011; Mills, 2023; Schnell et al., 2023; Wan & Flavell, 2009). Impaired generation of helper T cells during a mirror bacterial infection could diminish these antibacterial mechanisms, including AMP production, mucus secretion, enhancement of tight junctions, and transport of IgM and IgA across epithelial barriers (Mills, 2023), all of which might otherwise be effective against mirror bacteria. On the other hand, any damage caused by a mirror bacterial infection could expose innate immune cells to MAMPs from natural bacteria (especially at mucosal surfaces like the gut), causing them to produce inflammatory mediators that would stimulate tissue resident helper T cells to produce inflammatory cytokines and enhance the local innate immune response.

### CD1- and MR1-restricted T cells might be able to recognize some mirror ligands

CD1-restricted and MR1-restricted T cells secrete potent inflammatory cytokines upon activation, which can be protective against many bacterial species (Mori *et al.*, 2016). It is possible that both cell types would be able to detect some mirror bacterial ligands and activate during a mirror bacterial infection.

CD1-restricted T cells detect microbial lipids from a variety of bacterial species (e.g., Bacteroides and Mycobacteria) presented by antigen presenting cells via the nonclassical MHC molecules CD1a, CD1b, CD1c, and CD1d (Van Rhijn *et al.*, 2015). Although CD1 molecules have limited sequence variation (Salio *et al.*, 2014), the mechanism by which they bind their ligands does allow for some flexibility in the lipid antigens they present. CD1 molecules have deep, hydrophobic pockets that bind the largely achiral alkyl chains of microbial lipids, which allow the chiral head regions of the lipids to project up out of the binding pocket and interact with TCRs (Gras *et al.*, 2018). Although the loading of glycolipids into CD1 molecules can require chirality-dependent processing by host lipases and glycosidases, smaller lipids can be loaded without the need for accessory proteins (Salio *et al.*, 2014). This mechanism allows CD1 molecules to bind a limited diversity of lipid antigens and might also permit CD1 molecules to bind and present some mirror lipids. Moreover, although the TCRs of

CD1-restricted T cells are less variable than those of conventional T cells, they maintain some variation, which might enable them to bind to the achiral regions of mirror lipids. Together, these characteristics suggest that some CD1-restricted T cells might be activated during a mirror bacterial infection.

MR1-restricted T cells share many characteristics of CD1-restricted T cells, but recognize small molecule metabolites instead of lipids. These metabolites are released by live bacteria and bind to MR1, a nonclassical MHC molecule expressed by a broad range of cell types. Although MR1 displays limited sequence variation in humans (Rozemuller *et al.*, 2021) and all known stimulatory MR1 ligands are chiral, the molecule has a flexible binding pocket that might accommodate mirror bacterial metabolites (McWilliam & Villadangos, 2024). Further study may reveal additional, achiral ligands. In addition, MR1 ligands are not derived from larger precursor molecules, and therefore could plausibly be available in the phagolysosome without the need for enzymatic processing (Mori *et al.*, 2016). Although the TCRs of MR1-restricted T cells are semi-invariant, the presence of some sequence variation makes it difficult to rule out the recognition of mirror metabolites presented by MR1. Thus, similar to the case for CD1-restricted T cells, it seems possible that some MR1-restricted T cells could be activated by mirror bacteria.

Ultimately, the extent to which CD1- and MR1-restricted T cell activation would enhance immune responses to mirror bacteria is unclear. Most bacterial infection studies rely on mouse models, but mice make poor models to study these cells: they have much lower percentages of MR1-restricted T cells than humans, and they lack CD1a, CD1b, and CD1c (Mori *et al.*, 2016). In addition, not all bacteria express (known) CD1 or MR1 ligands. As such, it is difficult to predict to what extent CD1- and MR1- restricted T cells would contribute to protection against mirror bacteria.

#### Impaired T cell activation would impair antibody-mediated immunity to mirror bacteria

Antibodies typically play an important role in immune responses to extracellular bacteria by blocking bacterial binding to host surfaces, opsonizing bacterial cells for phagocytosis, and neutralizing bacterial toxins (Lu et al., 2018). Antibodies against protein antigens are normally generated after B cells in secondary lymphoid organs capture the proteins via their B cell receptors (BCRs), internalize them, and subsequently process and present peptide:MHC-II complexes to activated helper T cells. The T cells that recognize the peptide:MHC-II complex then deliver activating signals to that B cell, enabling it to proliferate and produce soluble antibodies. While B cells are expected to have BCRs that could recognize mirror proteins, their ability to process and present mirror peptides would be impaired. Impaired ability to process and present mirror peptides would severely limit B cell survival, differentiation potential, and the potential for these BCRs to undergo affinity maturation in secondary lymphoid organs because all of these processes rely on (sometimes repeated) T cell help (Victora & Nussenzweig, 2022). Even if some mirror peptides were presented, the previously described deficit of activated T cells could leave the corresponding B cells without T cell help. Instead of proliferating and maturing, these B cells would then become anergic or undergo apoptosis (Turner et al., 2020; Y. Wang et al., 2020; Yarkoni et al., 2010). The lack of T cell help could also reduce the B cell response to non-protein mirror antigens. Non-protein antigens from bacteria are often associated with proteins as part of intact cells or cell fragments (Kappler & Hennet, 2020). B

cells with BCRs that bind non-protein antigens thus internalize proteins as well, and these proteins are subsequently presented via peptide:MHC-II complexes to helper T cells, which, in turn, activate the B cells to produce antibodies against the non-protein antigen. Impaired T cell-dependent B cell activation could therefore result in a severe reduction in the overall antibody response to mirror bacteria.

B cells can also be activated without helper T cells, but it is unclear how effective this would be in response to mirror bacteria. A major activation mechanism requires the crosslinking of a sufficient number of BCRs by highly repetitive molecules, such as the capsular polysaccharides found on the surfaces of encapsulated bacteria (Vos *et al.*, 2000). Such crosslinking could plausibly be triggered by mirror bacteria, particularly by mirror encapsulated bacteria. While crosslinking of BCRs alone can lead to B cell proliferation (Snapper, 2016), T cell-independent antibody secretion generally requires co-stimulation with TLR ligands, although other co-stimulation routes have been documented, especially those involving IL-2 and IL-5 (Snapper, 2016). The extent of co-stimulation that would occur in response to mirror bacteria remains unclear. In particular, we expect TLR stimulation by most mirror MAMPs to be impaired, although it is possible that yet unseen co-stimulation routes would emerge. Thus, it is possible, though uncertain, that plasmablasts secreting IgM antibodies against mirror bacteria may arise, which would offer an avenue for the activation of the classical complement pathway (see Section 4.2).

It is unclear to what extent T cell-independent B cell activation would be able to elicit IgG antibodies directed toward mirror bacteria. Although class switch recombination and AID activity have been observed in T cell independent mechanisms, researchers had previously observed that a second signal provided via pathways other than the BCR, such as TLR co-stimulation, is typically necessary for these processes (Pone *et al.*, 2012; Snapper, 2016). A reduction in IgG antibodies would, at a minimum, result in a functionally altered response; IgG antibodies are typically the most abundant antibodies in serum, and can activate innate immune cells via Fc receptors and initiate the complement cascade via the classical pathway (Lu *et al.*, 2018). Additionally, IgGs can penetrate extravascular sites and provide signaling through Fc receptors that enhance the phagocytic efficiency of phagocytes (Uribe-Querol & Rosales, 2020).

Although we expect the production of mirror-antigen-specific IgG antibodies would be impaired, B cells in the gut may be able to produce IgA antibodies upon encountering mirror bacteria in the gut. Chronic immune stimulation from the intestinal microbiota creates an environment that promotes the activation of B cells whose BCRs bind to T cell independent antigens (e.g., polysaccharides), allowing them to undergo isotype switching and differentiate into plasma cells that produce IgA in gut associated lymphoid tissues (GALT) (Bemark *et al.*, 2024; Fagarasan *et al.*, 2010). IgA antibodies are the dominant isotype in the gut and are particularly effective at trapping bacteria in the mucus layer and keeping them from adhering to epithelial cells (T. Chen *et al.*, 2020). Research has also suggested that IgA antibodies against T cell independent antigens can acquire mutations through mechanisms that are not well-defined, allowing them to undergo affinity maturation (Bemark *et al.*, 2024). Together, this suggests that GALT-resident B cells responding to mirror bacteria in the gut might generate IgA antibodies against mirror antigens. Furthermore, if  $\gamma\delta$  T cells in the GALT become activated by DAMPs, cytokines, and/or the achiral MAMP HMBPP, this may increase the

likelihood of anti-mirror IgA antibodies production, as  $\gamma\delta$  T cells have been proposed to directly interact with B cells in the GALT to activate them (Rampoldi *et al.*, 2020). Even if these mechanisms do not enable GALT-associated B cells to produce IgA antibodies specific for mirror antigens, high concentrations of IgA antibodies are typically present in the gut, and many bind to specific gut bacterial species (C. Yang *et al.*, 2022). Therefore, it is plausible that some may have cross-reactivity with mirror bacterial epitopes.

Some individuals might fortuitously have pre-existing antibodies that would cross-react with mirror ligands. These could be natural IgM antibodies which, as discussed in Section 4.2, may have broader specificities than other antibody types. Alternatively, they could be antibodies previously produced in response to other pathogenic infections that happen to cross-react with mirror antigens. Natural IgM antibodies would be able to initiate the classical complement pathway, but would not be able to directly opsonize mirror bacteria because the Fc receptor for IgM (FcµR) is not expressed on phagocytic cells (Kubagawa *et al.*, 2009). For cross-reactivity with antibodies from previous infections, mirror antigens able to crosslink the BCRs of memory B cells would likely be sufficient to induce an antibody response. However, as discussed above, impaired T cell responses may hamper the strengthening of this humoral response via B cell differentiation or the affinity maturation of mirror-specific antibodies (Snapper, 2016).

Experimental evidence shows that mice immunized with full-length mirror proteins do not produce detectable IgG, despite producing robust IgG responses to non-mirror versions of the same proteins (King *et al.*, 1994; Marinec *et al.*, 2021; Uppalapati *et al.*, 2016). Although co-stimulation with an adjuvant containing natural-chirality MAMPs did result in some IgG production against a mirror protein in one study (Dintzis *et al.*, 1993), this mechanism is unlikely to be relevant for mirror bacterial infections due to the lack of natural-chirality MAMPs. We are not aware of any studies examining IgM or IgA responses to mirror proteins. As discussed above, failure to produce antibodies against mirror proteins is likely due to their resistance to proteolytic processing into peptides suitable for MHC loading. Since a majority of protein antigens derived from mirror bacteria are likely to exist as full-length folded proteins, the results of these studies are consistent with a dramatically impaired antibody response to mirror bacteria.

In contrast to full-length mirror proteins, immunization with short (7–26 amino acid) mirror peptides has been shown to produce weak or altered IgG responses compared to non-mirror peptide controls in laboratory animals (Arranz-Gibert *et al.*, 2018; Chong *et al.*, 1996). However, these mirror peptide studies may be less relevant to mirror bacterial infection because they involved the injection of concentrated doses of a single peptide species along with adjuvants containing MAMPs, rather than intact proteins in the absence of MAMPs.

Several studies published between 1960 and 1993 have been interpreted as evidence that mirror peptides are immunogenic (Sela & Zisman, 1997; Van Regenmortel & Muller, 1998), but the relevance of these studies to mirror bacteria is unclear for multiple reasons. In three studies, animals were immunized with long, highly repetitive mirror peptides (Gill *et al.*, 1964, 1967; Janeway & Sela, 1967) that are unlike the peptides expected to exist in a mirror bacterium. Furthermore, in two of these two studies, the peptides also contained 1–4% L-amino acids, which could have contributed to the observed immune response (Jung, 1992; Stupp & Sela, 1967). Several other contemporaneous

studies with similar peptides failed to detect antibody responses (Borek *et al.*, 1965; Gill *et al.*, 1963; Maurer, 1963, 1965). Other studies employed mirrored versions of branched multichain peptides—which are not found in nature—as immunogens (Mozes *et al.*, 1974; Sela *et al.*, 1972; Simonian *et al.*, 1968; Zisman *et al.*, 1993). In another two studies cited as evidence for the immunogenicity of mirror peptides, animals were immunized with mirror peptides conjugated to liposomes filled with MPLA, a potent TLR agonist that may have elicited T-independent antibody responses (Benkirane *et al.*, 1993; Guichard *et al.*, 1994). Moreover, some of these studies indicated mirror peptides were only immunogenic when administered at low doses (Gill *et al.*, 1964; Janeway & Sela, 1967; Mozes *et al.*, 1974). The experimental design of these early studies and their lack of replication since publication more than 50 years ago highlight the urgent need for further research in this area.

In summary, secretion of mirror bacteria-specific antibodies would likely rely almost exclusively on T cell-independent routes. Although T cell-independent routes have been known to elicit antibody secretion and even durable antibody memory—for example, the Pneumovax-23 vaccine (Grabenstein & Manoff, 2012)—they typically require co-stimulation from other immune pathways, notably those involving TLRs. Given the expected impaired activation of the innate immune system, it is less clear where co-stimulating signals may come from. Moreover, additional signals are often critical in determining the functional form of an antibody response, including the isotypes of antibodies that eventually arise. The unusual features of mirror bacterial infection make it very difficult to anticipate the functional features of antibody responses against mirror bacteria.

Immune memory responses formed after an initial infection enable faster, more effective responses against future exposures to the same pathogen and are less dependent on MAMP and DAMP signaling (Chandok & Farber, 2004; Lam *et al.*, 2024; Palm & Henry, 2019). However, durable memory T cell and B cell responses are dependent on a robust adaptive immune response to the initial infection (Palm & Henry, 2019; Pennock *et al.*, 2013). Therefore, the impaired initial immune response described here would likely preclude the formation of protective memory responses to mirror bacteria as well.

#### Mirror bacteria evasion of adaptive immunity might mimic MHC-II immune deficiency

MHC-II deficiency is likely the closest clinical analog to the severe deficit in antigen presentation expected during mirror bacterial infections. MHC-II deficiency, also known as bare lymphocyte syndrome type 2, is a rare autosomal recessive primary immunodeficiency where loss-of-function in transcription factors for MHC-II results in reduced or absent expression of MHC class II molecules on the surface of immune cells (Reith *et al.*, 2013). This condition results in a profound impairment of antigen-specific T and B cell responses because of MHC-II's essential role in antigen presentation.

Without stem cell transplantation—the only curative treatment—the prognosis for MHC-II deficiency is bleak. Despite supportive care, antimicrobial prophylaxis, and immunoglobulin administration, the typical clinical course involves multiple severe acute and chronic infections, and death typically occurs within the first decade of life (Hanna & Etzioni, 2014; Reith *et al.*, 2013).

Survival into adulthood is rarely documented, and the factors that explain these more favorable outcomes are poorly understood.

Although a close mechanistic analog, MHC-II deficiency would not exactly match the severe deficits in antigen presentation expected for healthy individuals challenged with mirror bacteria. Complete absence of response to foreign antigens is not the only immunological feature of MHC-II deficiency: most patients have a substantial CD4+ T-cell cytopenia, likely due to the requirement of MHC-II for T-cell maturation, and 20% of patients have autoimmune manifestations. Although the deficit in antigen presentation is likely the primary driver of the immunodeficiency—successful stem cell transplant normalizes antigen responses while CD4+ T cell counts remain low (Ouederni *et al.*, 2011), and autoimmune manifestations do not correlate with prognosis (Hanna & Etzioni, 2014)—these additional immunological defects associated with MHC-II deficiency may contribute to a greater overall functional deficit in immunity than would occur in a healthy immune system challenged with mirror bacteria.

The poor outcomes of MHC-II deficiency are attributable to a broad and pervasive vulnerability to infection from many organisms, not just bacteria. Alongside bacterial infections and bacterial sepsis, viral meningitis, liver failure secondary to infection by protist Cryptosporidium parvum, and pneumonia caused by the fungus *Pneumocystis jiroveci* are life-threatening complications frequently reported in MHC-II deficiency case series (Ben-Mustapha et al., 2013; Ouederni et al., 2011). Nevertheless, MHC-II deficiency demonstrates a mechanistically similar deficit to that expected against mirror bacteria and is sufficient to confer extreme vulnerability to bacterial infection. In one case series of 35 patients with MHC-II deficiency (Ouederni et al., 2011), 11 (31%) had documented episodes of bacterial sepsis, 10 (28%) bacterial pneumonia, and 5 (14%) bacterial pyelonephritis, numbers which are greatly elevated compared to the general population (e.g., the global incidence of pediatric and neonatal sepsis is estimated to be 48 cases per 100 000 person years (Fleischmann-Struzek et al., 2018). This vulnerability to bacterial disease may still be underestimated in the MHC-II clinical data due to extensive anti-bacterial prophylaxis (which may not be easily available for infections with mirror bacteria, see Chapter 5), incomplete or unclear ascertainment of the causative organism for infections, and competing causes of death (i.e., more individuals may have developed serious bacterial infections had they survived earlier non-bacterial infections).

# Profound immune evasion by mirror bacteria could plausibly render humans vulnerable to infection

As discussed throughout Sections 4.1–4.3, it is plausible that the human immune system would be severely impaired if it were to encounter mirror bacteria. Significant uncertainty remains regarding which elements of bacterial immunity would remain functional, and whether these remaining functionalities would be sufficient to stave off mirror bacterial infection. The immune system is extremely complex, with a wide range of microbial sensors and antimicrobial effector mechanisms providing great coverage, diversity, and redundancy that ensure robustness of defense. Furthermore, these different host defense pathways can cooperate, complement, and compensate each other, making it challenging to establish a clear hierarchy of importance during any given infection (Nish & Medzhitov, 2011). Compounding the difficulty of this assessment is the possible diversity in the types

of mirror bacteria that could be created (e.g., Gram-negative or Gram-positive, encapsulated or non-encapsulated), and the natural heterogeneity of the human immune system across individuals. There is also a dearth of experimental evidence and lack of a clear analog for mirror bacteria: they are unlike natural-chirality pathogens in lacking functional virulence factors, yet they are also unlike commensals due to their suspected profound ability to evade immunity. They are also unlike inanimate particles, as they can actively spread and multiply. Simply put, mirror bacteria would represent a unique and unprecedented challenge to the human immune system.

However, several pieces of evidence suggest that the plausible extent of immune evasion exhibited by mirror bacteria could render humans highly susceptible to infection, with potentially harmful outcomes. First, immunodeficiencies affecting even one microbial sensor or effector pathway can already render patients (or animal models) more vulnerable to bacterial infection, as is seen with deficiencies in IRAK-4 or MyD88 (Picard *et al.*, 2010), C2 or C4 (Pettigrew *et al.*, 2009), or MHC-II (Ouederni *et al.*, 2011). Second, individuals who are immunocompromised in a manner that affects multiple cell types or effector pathways—e.g., due to chemotherapy, bone marrow transplants, or immunosuppressive therapy—are often at significant risk of life-threatening infection (Bock *et al.*, 2013; Roberts & Fishman, 2021; Safdar *et al.*, 2011). Third, profound immune evasion in the form of MAMP masking, phagocytosis inhibition, and antigenic variation, among other strategies, enables various natural-chirality pathogens to block immune clearance, as seen with e.g., *Mycobacterium tuberculosis, Salmonella typhi, Y. pestis*, and *Neisseria gonorrhoeae*. This can result in long-term, persistent infections that do not resolve until medical intervention (Chandra *et al.*, 2022; Demeure *et al.*, 2019; Lovett & Duncan, 2018; Monack *et al.*, 2004).

These individual pieces of evidence do not make perfect analogies for mirror bacterial infections due to caveats outlined elsewhere in this chapter. However, when taken together, they paint a concerning picture: even if some immune defenses remain functional, mirror bacteria could pose a severe infection risk to humans. In Section 4.4, we discuss what a mirror bacterial invasion could look like if humans were sufficiently exposed to mirror bacteria in the environment.

# 4.4 Mirror bacteria could plausibly pass barrier surfaces and translocate into the bloodstream and tissues

As described in Sections 4.1–4.3, mirror bacteria would likely evade or impair many elements of the immune system. Consequently, if mirror bacteria were to reach internal tissues or the blood, it is plausible that they would not be efficiently removed by normal immune processes and might instead cause infection (see Section 4.5). Unlike natural-chirality pathogens, however, mirror bacteria would likely not, by default, have functional virulence factors that would facilitate invasion or survival. Furthermore, any virulence factors present could have incompatible chirality with their host ligands.

In this section, we consider whether mirror bacteria would be able to pass barrier surfaces and enter human tissues in the absence of typical virulence factors that normally mediate such processes. As we will discuss, human physical barriers are often leaky due to trauma, environmental compounds, and inflammatory diseases, yielding opportunities for bacteria to translocate. It is difficult to predict how frequently mirror bacteria would bypass these defenses in practice, as that would depend on the environmental prevalence of mirror bacteria, the efficacy of non-pharmaceutical measures like sterilization techniques or hand-washing, and the robustness of the mirror bacteria to the environmental stressors that limit growth and survival on skin and mucosal surfaces. Nevertheless, commensal bacteria lacking specialized virulence factors for invasion are able to translocate through natural barriers and can be found in the bloodstream (Pat *et al.*, 2024), which makes it plausible that mirror bacteria would be able to do the same if present in sufficient numbers and given the opportunity.

While many pathogens invade deeper tissues only after adhesion to—and subsequent colonization of—barrier surfaces like the lungs, skin, or gut, mirror bacteria would plausibly face great difficulty establishing sessile colonies due to a lack of functional adhesion receptors that confer strong adhesion capability. However, adhesion is not an absolute requirement for infection, and comparison with encapsulated bacteria suggests that a non-adhesive phenotype might even predispose mirror bacteria to enter the bloodstream. It is also not clear that absence of adhesion factors would preclude colonization of the lung and other barrier surfaces in all cases, particularly given other unusual properties of mirror bacteria.

Physical barriers are imperfect and can be crossed passively, which could plausibly allow mirror bacteria to invade

Humans encounter environmental bacteria on a continuous basis on body surfaces that are directly exposed to the outside world. These surfaces include the skin, eyes, and mucosa of the respiratory, genitourinary, and gastrointestinal tracts, which are protected by both physical and chemical barriers that keep the vast majority of commensal and pathogenic bacteria from entering the underlying tissues (Marshall *et al.*, 2018). Barrier epithelial cells are structurally connected like a fence that selectively allows gasses and nutrients to pass but holds microbes at bay. The structural integrity of the barrier relies on intercellular junctions that connect epithelial cells: protein complexes including tight junctions, adherens junctions, and desmosomes (Matter & Balda, 2003).

Damage to epithelial barriers can allow bacteria to enter the body. Scratches, abrasions, lacerations, burns, and insect bites routinely break the skin and provide portals of entry for microbial infection. Humans also deliberately induce breaches through surgical wounds, needles, and medical and dental implants, and even through less invasive daily actions like flossing and shaving (Brodell & Rosenthal, 2008).

Epithelial damage that is sufficient to allow bacterial translocation can also occur in the absence of trauma. Air pollution-related compounds like smoke and ozone, as well as numerous chemical irritants found in common cleaning agents, have damaging effects on the skin and respiratory tract in case of prolonged exposure, which can result in increased bacterial translocation (Pat *et al.*, 2024). Ingested alcohol, fatty acids, and various food additives can disrupt the integrity of the gut epithelial barrier (Portincasa *et al.*, 2022), as can many pharmaceutical drugs and medical treatments (Fine *et al.*, 2020). Respiratory viral infections disrupt functioning of lung epithelia (Gawrysiak *et al.*, 2021; Lalbiaktluangi *et al.*, 2023) and can predispose patients to secondary bacterial infections in which bacteria benefit from virus-induced barrier damage to invade deeper underlying tissues

#### Chapter 4: Risks to Human Health

(Lalbiaktluangi *et al.*, 2023; Morris *et al.*, 2017). Some bacterial pathogens directly cause damage to epithelial cells or tight junctions in order to facilitate invasion (Eichner *et al.*, 2017; M. Kim *et al.*, 2010). Finally, various non-infectious diseases, such as inflammatory bowel disease, atopic dermatitis, asthma, obesity, and non-alcoholic fatty liver disease, are associated with epithelial barrier dysfunction (Hellings & Steelant, 2020; B. E. Kim & Leung, 2018; König *et al.*, 2016; Portincasa *et al.*, 2022). Disruptions to epithelial layers allow the translocation of (non-invasive) commensal microbiota (Pat *et al.*, 2024). Therefore, it is plausible that mirror bacteria could be similarly translocated as a consequence of such disruptions.

Even in the absence of epithelial or endothelial damage, bacteria can pass physical barriers through natural antigen sampling processes. M cells—specialized epithelial cells found in parts of the gastrointestinal tract and in certain other mucosal-associated lymphoid tissues—have been shown in various mammals to transport particulates like resin and latex beads (Gebert *et al.*, 1996; Payne *et al.*, 1960) and several bacterial pathogens, including *Salmonella typhimurium*, *Yersinia enterocolitica*, and *Vibrio cholerae*, across the epithelial barrier through both receptor-mediated and endocytic pathways (Ribet & Cossart, 2015). While many of these bacterial pathogens have specialized mechanisms by which to target M cells—mechanisms which would not work even if present in mirror bacteria due to incompatible chirality—given that inert particles can be transported by M cells, it appears at least plausible that mirror bacteria could be transferred across epithelial barriers by M cells. Mouse studies have shown that interstitial dendritic cells also sample commensal bacteria to mesenteric lymph nodes (Niess *et al.*, 2005). To our knowledge, the presence of antigen-sampling dendritic cells has not been demonstrated in humans, making it unclear whether mirror bacteria could benefit from this or any additional antigen-sampling mechanisms.

### Many chemical mechanisms of epithelial barriers would likely be impaired against mirror bacteria

In addition to the physical protection afforded by the epithelium itself, epithelial barriers are protected by a number of chemical mechanisms that mirror bacteria would need to overcome to establish infection. Membranes of the gastrointestinal, genitourinary, and respiratory tracts are protected by a layer of mucus. This layer is primarily composed of glycoproteins called mucins, which form a slimy network with two protective functions. The top layers act like a physical barrier that can bind and trap pathogenic bacteria, whereas the layers closer to the epithelial cells sequester high concentrations of protective AMPs, proteases, and IgA antibodies (Sheng & Hasnain, 2022). Mirror bacteria are expected to have decreased susceptibility to various chemical defenses, which could plausibly allow them to stay close enough to physical barriers to increase the basal translocation rate and aid with potential colonization.

Mucus hinders bacterial invasion through steric hindrance—i.e., providing a physical block. It also adheres to bacteria through both nonspecific mechanisms, such as electrostatic and hydrophobic interactions with bacterial pili and fimbriae, and specific interactions between glycosaminoglycan mucins and bacterial adhesin receptors (McGuckin *et al.*, 2011). Steric hindrance and the non-specific interactions are likely insensitive to chirality, but mirror adhesins would not be able to

tightly bind their natural-chirality mucin ligands, plausibly reducing the efficacy of mucus at trapping mirror bacteria. Even so, migration of mirror bacteria in the mucus layer may still be constrained because mirror versions of the mucin-degrading proteases that facilitate movement of natural bacteria are unlikely to degrade natural mucins. On the other hand, even without such proteases, bacteria capable of flagella-based motility can push through the mucin network (Guerry, 2007; Ottemann & Lowenthal, 2002). As such, flagellated mirror bacteria may be able to move through these mucin networks with reduced opposition due to weaker binding forces.

A bacterium that penetrates the top layers of the mucus network will encounter the antimicrobial defenses present in the bottom layer, such as IgA, a-defensins, lysozymes, lactoperoxidases, siderophore inhibitors, and many others (Ribet & Cossart, 2015). AMPs and proteases produced by keratinocytes also comprise the primary chemical defenses of the human skin (Gallo & Hooper, 2012). As discussed in Section 4.2, mirror bacteria would likely be less sensitive to lysozyme and many other antimicrobial enzymes that would be unable to cleave their respective mirror ligands. This could increase the likelihood of mirror bacterial infection because natural-chirality gastrointestinal bacteria like *H. pylori* and *Enterococcus faecalis* that are resistant to lysozyme are more capable of establishing infections in mouse models (Benachour *et al.*, 2012; G. Wang *et al.*, 2010). Siderophore inhibitors such as lipocalin-2 would likely remain effective, however, as their binding to diverse siderophores is not very stereoselective (Goetz *et al.*, 2002) and has been shown to bind both mirror-image and natural-chirality enterobactin (the common siderophore of *E. coli*) with similar affinity (Abergel *et al.*, 2006).

It is possible that mirror-specific IgA antibodies would be present at mucosal surfaces (see Section 4.3). IgA deficiency moderately predisposes patients to recurrent respiratory and gastrointestinal infections, suggesting IgA does confer protection against infection. However, this phenotype might partially be due to general disruptions of the microbiome (Swain *et al.*, 2019), and it is difficult to determine how mirror bacteria specifically would be affected by the absence or presence of IgA binding.

AMPs that are constitutively expressed at barrier defenses may have some efficacy against mirror bacteria because, as mentioned earlier, many are expected to retain their bactericidal properties. In the intestine, for example, some AMPs are expressed and secreted by intestinal epithelial cells during homeostasis as a means to control the microbiota (Muniz *et al.*, 2012; Ostaff *et al.*, 2013). Breaches in the mucosal barrier typically result in PRR-mediated increases in the concentrations of these and other, more tightly regulated AMPs (Muniz *et al.*, 2012; Ostaff *et al.*, 2013). As discussed in Section 4.2, the absence of inflammatory signaling during a mirror bacteria infection might be insufficient to induce additional AMP production. However, the presence of natural commensals alongside mirror bacteria may still cause localized inflammation and, in some cases, might therefore stimulate AMP production and other defenses.

Overall, mirror bacteria would plausibly be more resistant to chemical barriers than natural bacteria. IgA antibodies, AMPs, antimicrobial enzymes, and the physical properties of the mucus layer itself would likely be less effective against mirror bacteria than natural bacteria. The magnitude and significance of this advantage, however, is difficult to determine, a challenge further compounded by the influence of other microbiota on localized inflammatory signaling.

Evidence from transient commensal bacteremias suggests translocated mirror bacteria could end up in the bloodstream

Bacteria that pass the epithelial barriers reach the interstitial spaces, which house tissue-resident immune cells, non-immune cells, extracellular matrix, lymphatic vessels, and blood capillaries. These tissue-resident immune cells typically induce a local inflammatory response upon detection of translocated bacteria, which results in clearance of most, if not all, of these invaders (Walker, 2017). Bacteria that are not locally cleared can be taken up by the lymphatic system and subsequently killed in draining lymph nodes (Siggins & Sriskandan, 2021). Given the immunological impairments described in Sections 4.1–4.3, we expect significantly reduced immunological clearance of mirror bacteria in the interstitium and lymph nodes, potentially creating opportunities for mirror bacteria to disseminate through the body and enter the bloodstream if they were to pass through a breach in the epithelial barrier.

Various human studies demonstrate that bacteria of the indigenous microbiota translocate into deep tissues and the bloodstream in healthy humans on a regular basis without the use of specialized invasion factors. For example, culture-based studies of blood collected from subjects after they undertook "daily oral activities" show that the prevalence of transient bacteremia can be as high as 24% after brushing, 40% after flossing, and 17% after chewing on hard candy (Cobe, 1954; Tomás *et al.*, 2012). Furthermore, metagenomic sequencing of whole blood and cell-free DNA in blood from healthy subjects strongly suggests routine bacterial translocation in healthy adults. In the most compelling of these studies, metagenomic analysis of whole blood from 9 770 healthy individuals found that 11% of adults had credible evidence of DNA from presumptive translocating commensal bacterial species (Tan *et al.*, 2023). Finally, upward of a quarter of bacterial bloodstream infections lack a known source or focus of infection (Renaud *et al.*, 2001; Weinstein *et al.*, 1997), suggesting that transient translocation is a possible cause of a significant proportion of bacteremias.

There is additional evidence to suggest that human lymph nodes can be a reservoir for some bacteria, and that bacteria disseminate to the bloodstream through the lymphatic system (Siggins & Sriskandan, 2021). Culture-based studies of patients who undergo a laparotomy for a wide variety of surgical indications showed that live bacteria could be recovered from mesenteric lymph nodes in 5–10% of patients at the time of initial entry into the abdominal cavity (Deitch, 2002; S. Jin *et al.*, 2022; Sedman *et al.*, 1994). Live bacteria have also been recovered from pelvic lymph nodes of patients with gynecologic tumors (Wells *et al.*, 1990) and from cervical lymph nodes in patients with oral cancer (Sakamoto *et al.*, 1999). In a study of 1 762 lymph node biopsies from patients with suspected infectious lymphadenitis, bacterial DNA from a variety of species was detected in 30% of biopsies through real-time PCR; when 351 lymph nodes were further cultured on agar plates, 22% of samples yielded live bacteria (Prudent *et al.*, 2018).

Another line of evidence suggesting mirror bacteria could passively translocate into the body comes from the observation that inert microplastics of similar size to bacteria have been detected in various human body compartments, including the blood and the heart (Guan *et al.*, 2023; Leslie *et al.*, 2022; Marfella *et al.*, 2024). Although the exact mechanisms of how microplastics enter the bloodstream

are unknown, their initial entry point is thought to be through the epithelial layers of the lungs and the gut (Khan & Jia, 2023).

Because these studies are only snapshots in time, they do not establish how frequently bacterial translocations to the bloodstream occur, although the >5% frequencies across studies suggest yearly incidence is high. The number of bacteria typically translocated in these events, and whether some species have a higher propensity for translocation than others, is unknown. However, given the unusual degree of immune evasion intrinsic to mirror bacteria, it is plausible that mirror bacteria that have passively translocated across epithelial barriers could escape immune clearance and eventually find their way to the bloodstream.

#### By default, mirror bacteria would likely be weak adherents

Bacterial adhesion to host barrier surfaces is a very common first step in bacterial colonization and invasion for commensals and pathogens alike. Adhesion provides resistance to mechanical removal and facilitates manipulation of host cells through effector systems and toxins (Stones & Krachler, 2016). Additionally, bacterial adhesion to host cells or other surfaces is the first event in the formation of biofilms (Achinas *et al.*, 2019). These structured communities of bacterial cells provide protection for their inhabitants from environmental stressors like bacteriophages, immune cells, antibiotics, antibodies, and surfactants (Dunne, 2002).

Adhesion and biofilm-forming capacity vary considerably between individual bacterial species (Malhotra *et al.*, 2019; Tu *et al.*, 2022) and even between different strains of the same bacterial species, which often differ from each other in the expression of key regulators (Abdulkadieva *et al.*, 2022; Martínez-Meléndez *et al.*, 2022; Y. Yang *et al.*, 2024). As such, the extent to which mirror bacteria would be able to form sessile colonies or biofilms within the human body will highly depend on the specific properties of the mirror bacterial strain. Specialized adhesion mechanisms that confer tight binding are generally stereospecific, and as a result, mirrored versions of natural bacteria would probably not adhere well to specific sites unless they were intentionally engineered to do so.

For concreteness, consider a mirror *E. coli*, which is a plausible model for the first robust mirror bacteria. *E. coli* can form biofilms on abiotic surfaces (Beloin *et al.*, 2008), so a mirror *E. coli* could be capable of biofilm formation in some environments. In these cases, adhesion is mediated by non-specific physical-chemical interactions between bacterial appendages and host surfaces that are relatively weak (Dufrêne, 2015; Kreve & Dos Reis, 2021). *E. coli* generally employs a host of different adhesins that confer specific binding. The α-D-mannose-binding fimbrial adhesin FimH seems particularly important for biofilm formation within the human body, where mucosal surfaces are constantly washed by fluids. Point mutations in FimH prevent strong stationary attachment under fluid shear force, so that mutated bacteria instead roll across the surface (B. N. Anderson *et al.*, 2007). Because mirror FimH would not be able to bind α-D-mannose (Firon *et al.*, 1983; Gade *et al.*, 2017), it is likely that an unmodified, generalist mirror *E. coli* would have reduced adhesion and biofilm-forming capacity compared to natural commensal and pathogenic *E. coli* strains.

While a mirror *E. coli* would likely adhere poorly to bodily surfaces, there is no fundamental reason why mirror bacteria could not evolve or be engineered to be better adherents. As discussed in Section

8.4, mirror bacteria could evolve rapidly. A mirror *E. coli* would already possess nonspecific adhesion mechanisms and be capable of biofilm formation, and evolutionary experiments have shown that adhesion and biofilm formation on abiotic surfaces can improve upon positive natural selection (M. Yoshida *et al.*, 2022). If improved adhesion would facilitate vertebrate colonization and infection—say, by reducing the exposure to mirror bacteria needed to cause infection—then it could plausibly evolve very rapidly.

## Weak adhesion may impair colonization of barrier surfaces, but not preclude systemic infection

Since the gut, urinary tract, and other internal barrier surfaces are subject to aggressive mechanical clearing mechanisms, a weakly adhesive mirror *E. coli* might have difficulty colonizing these areas. Biofilms also protect bacteria from dry, acidic environments, so any impediments in biofilm formation could make skin colonization challenging.

While a reduction in adhesive capacity might hinder colonization of barrier surfaces, it is not a requirement for translocation or infection. Indeed, a weakly adhesive phenotype can confer certain benefits such as improved nutrient access and increased proliferation (Achinas *et al.*, 2019), as well as reduced hindrance from the extracellular matrix during translocation (McGuckin *et al.*, 2011). FimH-mutated *E. coli* strains that rolled over surfaces under shear stress ultimately colonized larger surface areas than non-mutated *E. coli* (B. N. Anderson *et al.*, 2007). Furthermore, a major function of biofilms is to protect bacteria from immune cells, antibiodies, antibiotics, and bacteriophages. Since mirror bacteria would already be resistant to most such agents, the need for biofilms as a protective measure would be plausibly diminished.

An analogy for the clinical consequences of a weakly adhesive phenotype can be seen in infections with encapsulated bacteria, which produce exopolysaccharides that increase electrostatic repulsion to the extracellular matrix, mask adhesion molecules, and inhibit phagocytic clearance (Absolom, 1988; Hollands *et al.*, 2010; Schembri *et al.*, 2004). Encapsulated or mucoid bacterial strains are inherently less tissue adhesive (but more immune evasive), and are more commonly associated with bloodstream infections than with localized biofilm infections (Moxon & Kroll, 1990; Siggins *et al.*, 2020). For example, non-encapsulated strains of *Haemophilus influenzae* are more commonly found as colonies in the nasopharyngeal carriage than encapsulated strains; however, the encapsulated strains are strongly linked to invasive disease and bacteremia (Crowe, 1987; Moxon & Kroll, 1990; Oliver *et al.*, 2023). Similarly, in a study with 1 100 patients with *Streptococcus pyogenes* infection, encapsulated strains were strongly associated with invasive disease syndromes such as bacteremia, toxic shock syndrome, and rheumatoid fever, but very weakly associated with non-invasive disease like pharyngitis (Johnson *et al.*, 1992).

A weakly adhesive or planktonic mirror *E. coli* (or similar generalist mirror bacterium) might be able to colonize niches in the human body so long as there are sufficient nutrients to sustain their proliferation at a rate that outpaces mechanical or immunological clearance (Stones & Krachler, 2016). Colonization of the lungs would be particularly concerning. *Yersinia pseudotuberculosis* can colonize the lungs of mice in the absence of specialized adhesion factors (Paczosa *et al.*, 2014), and
*P. aeruginosa* can establish acute lung infections in planktonic (free-living, non-biofilm) form in human patients (Valentini *et al.*, 2018). As such, mirror bacteria might be able to colonize the lungs even without functioning adhesins. The gut would be another concerning site for long-term colonization, but the role of bacterial adhesion in gut colonization is less clear (Lin *et al.*, 2024). Whether initial colonization would result in long-term, persistent colonies would likely depend on a wide range of factors, including the ability to access and process available nutrients. Mirror bacteria would have to compete with native microbiota for nutrient access at these sites, but their resistance to many bodily defenses, phages, and most forms of bacterial warfare (see Section 8.1 for a more detailed discussion) could provide significant advantages that might, in some cases, be enough to compensate for reduced access to nutrients and other disadvantages that would result from reversed chirality.

The potential clinical consequences of such an infection are hard to predict, but if mirror bacteria were able to successfully establish a persistent population within some bodily niche, this could provide them with a place to proliferate and disseminate from, increasing the chances of eventual translocation into the bloodstream. It is also possible that successful colonization could cause adverse clinical pathology directly—for instance, by causing gut dysbiosis or pulmonary disease. A more detailed assessment of the potential consequences of mirror bacterial colonization of luminal or exterior spaces would be necessarily speculative, and is beyond the scope of this report.

# 4.5 Mirror bacteria could plausibly replicate in blood and cause lethal systemic infection

Infection with natural-chirality bacteria spans a vast range of clinical manifestations. The course of infectious disease pathology is very challenging to infer from pathophysiological first principles due to the inherent complexities and heterogeneity of host-pathogen interactions. Assessing general patterns for the clinical course of any mirror bacterial infection would be highly uncertain and speculative. Nonetheless, given the extent to which mirror bacteria may evade immune defenses (Section 4.1–4.3), it appears plausible that if mirror bacteria were to access internal compartments like the blood, this could result in a systemic infection. When uncontrolled, systemic infections from natural bacteria are extremely dangerous, regardless of the properties or severity of any localized infections caused by the same species. The possibility that exposure to mirror bacteria could lead to similar systemic infections is therefore concerning and warrants further assessment.

As discussed in Section 4.4, commensal bacteria frequently translocate across barrier surfaces. Human blood has suitable abiotic conditions and sufficient nutrients to permit the replication of many commensals and other common bacteria. However, bacteria within blood must also face immunological clearance mechanisms, including the complement system, antibodies, and professional phagocytes (particularly within the spleen and liver). So long as these mechanisms remove bacteria more rapidly than they can replicate, any translocated bacteria will be removed before they can establish a persistent bacteremia. Thus, bacteremias associated with passive translocation are typically transient, subclinical, and self-limiting. Many immunological responses to natural bacteria appear unlikely to function effectively against mirror bacteria (Section 4.1–4.3). In addition, as discussed below, conditions within the blood could be growth-permissive for many mirror bacteria, including mirror strains of generalists such as *E. coli*. It is therefore plausible that mirror bacteria within the bloodstream could replicate without being checked by normal immunological responses. Absent medical intervention, the result would likely be fatal, whether due to a delayed sepsis-like response or other changes to the blood.

### Generalist mirror bacteria could plausibly grow in human blood

The blood is a broadly temperate environment for bacterial growth. With a temperature of roughly 37°C, a pH of 7.4, an oxygen tension of 75–100 mmHg, and a water activity of 0.99, abiotic conditions within blood would be growth-permissive for many common bacteria. This is supported by the fact that many bacterial species can proliferate in blood: specialist pathogens, commensals of skin and gut, and environmental bacteria have all caused transfusion-transmitted bacterial infections (Brecher & Hay, 2005).

A mirror bacterium would likely be at a nutritional disadvantage relative to natural-chirality bacteria. Nevertheless, nutrient availability does not appear to be limiting for a sufficiently robust mirror bacterium. Human fasting blood glucose is between 3.9 and 5.6 mM (Wilkinson *et al.*, 2024), while growth in *E. coli* strains has been demonstrated at glucose concentrations as low as 0.1  $\mu$ M (Füchslin *et al.*, 2012; Shehata & Marr, 1971). If engineered to catabolize natural-chirality glucose, a mirror *E. coli* or similar generalist bacterium would have an ample carbon source. As discussed in Chapter 1, a number of other nutrients could also be used as a sole carbon source by a mirror *E. coli* K-12 even absent D-glucose catabolism, many of which are present in blood. These include glycerol, L-alanine, and L-lactate, which are all found at concentrations of a few hundred micromolar, and acetate, acetoacetate, pyruvate, and succinate, present in the tens of micromolar (Psychogios *et al.*, 2011). A mirror *E. coli* could likely already utilize, or quickly adapt to utilize, most other L-amino acids (Table 1.2).

Another nutritional challenge would be access to trace metals essential for growth. The most important (and best characterized) of these is iron, although the picture for copper, zinc, and manganese appears broadly similar (Coverdale *et al.*, 2019; Davidsson *et al.*, 1989; Kirsipuu *et al.*, 2020). Iron is poorly soluble, and free iron is particularly scarce in blood due to dedicated iron-binding proteins such as transferrin (G. J. Anderson & Frazer, 2017; Raymond *et al.*, 2003). Bacteria can scavenge iron in many environments by producing siderophores, small organic compounds that chelate free iron with extremely high affinity and are subsequently reabsorbed by the bacteria. These remain effective in blood as they have a higher affinity for iron than host iron-binding proteins (Miethke & Marahiel, 2007; Wilson *et al.*, 2016). Lipocalin-2, a human acute phase protein, targets these siderophores to restrict pathogen access to iron (Flo *et al.*, 2004). Specialized pathogens can have further adaptations to counteract this in turn, including producing siderophores that evade lipocalin-2 (Fischbach *et al.*, 2006) or possessing alternative means of iron acquisition, including scavenging host heme (Contreras *et al.*, 2014).

#### Chapter 4: Risks to Human Health

The reversed chirality of mirror bacteria would likely impair their iron acquisition, although this impairment is of uncertain severity and unlikely to apply universally. Mirror-image and natural-chirality siderophores should chelate ferric iron equally well, although the diminished ability to form biofilms would likely deprive mirror bacteria of the benefits of cooperative production between individual mirror bacteria. Unlike many other immune mechanisms, the broad electrostatic/cation- $\pi$  binding of lipocalin-2 to a wide variety of siderophores (Clifton *et al.*, 2019) appears to be broadly non-stereospecific, and similar binding of lipocalin-2 to natural-chirality and mirror-image enantiomers of one common siderophore, enterobactin, has been experimentally demonstrated (Abergel *et al.*, 2006). However, lipocalin-2 production is induced in infection by recognition of MAMPs by TLRs. Since most TLRs are unlikely to recognize mirror MAMPs, it is uncertain whether constitutive lipocalin-2 expression would be sufficient during a mirror bacterial infection to inhibit growth. The alternative mechanisms of iron acquisition (e.g., scavenging heme) have chiral targets, thus their mirror images are unlikely to remain effective.

High concentrations of certain D-amino acids can be toxic to natural-chirality bacteria, as discussed in Box 1.2. Most D-amino acid toxicities require concentrations much greater than found in blood, but D-cysteine is an exception that can cause growth inhibition of *E. coli* at micromolar concentrations in some environments (Soutourina *et al.*, 2000). L-cysteine and L-cystine concentrations within serum total around 200 uM (Psychogios *et al.*, 2011), and though in blood the majority of cysteine is likely bound to proteins through disulfide bonds (Chawla *et al.*, 1984), it is possible that L-cysteine could slow or even entirely preclude growth of an unadapted mirror *E. coli*. However, even if this were the case, it is plausible that resistance to cysteine would arise rapidly (see Box 1.2). The extent to which other mirror bacterial strains would be sensitive to L-cysteine is unclear; *P. aeruginosa*, for instance, shows no inhibition by D-cysteine even at millimolar concentrations when grown in minimal medium. It is conceivable that other compounds in blood could prove toxic to a mirror *E. coli* or other mirror bacteria due to their reversed chirality, but this appears unlikely given both the low concentrations of most metabolites and the direct evidence that, for many metabolites, their enantiomer is not toxic to natural-chirality *E. coli*.

## Absent an effective immune response, mirror bacteria could replicate unchecked within the bloodstream

The typical fate of bacteria within the blood is clearance rather than proliferation, as natural pathogens are usually rapidly removed by the immune system (Levin & Antia, 2001). However, mirror bacteria would likely evade many of the immunological defenses that clear natural bacteria from the blood. As discussed in Sections 4.1–4.3, impaired innate immune recognition would considerably reduce the efficacy of professional phagocytes, the complement system, anti-microbial enzymes and peptides, and antibodies. If mirror bacteria are able to replicate faster than they are removed by any remaining immunological defenses, their population within the blood will grow exponentially.

It is plausible that some clearance processes would retain at least partial efficacy, such as the alternative activation pathway of the complement system against Gram-negative bacteria. If these processes are able to remove the mirror bacteria faster than they replicate, this may be protective

against some kinds of mirror bacteria, particularly those suffering from nutritional or other major defects that limit their growth rate. However, given that a wide variety of species can cause transfusion-transmitted bacterial infections (Brecher & Hay, 2005), and that people with relatively mild immunodeficiencies or immunosuppression are at much greater risk of systemic infection (Chaudhary *et al.*, 2017; Kristinsson *et al.*, 2014), it is plausible that sufficiently robust mirror bacteria could establish infection even within healthy adults.

If the replication rate for a mirror bacterial strain within blood is greater than immune clearance, the presence of even a single bacterium within the blood could lead to infection. In practice, any given mirror bacterium that reaches the blood might die before it replicates, either due to immune mechanisms that would still function against mirror bacteria (e.g., complement) or other processes (e.g., a fatal mutation). If more than a handful of mirror bacteria enter the blood, however, the probability of all of them dying before replicating would be lower, and infection would become more likely. Similar behavior is seen in many natural systemic infections, in which all bacteria present may have descended from as few as a single bacterium (Gerlini *et al.*, 2014).

## Systemic mirror bacterial infection could plausibly be fatal

If able to establish themselves within the blood, mirror bacterial concentrations would likely grow exponentially as long as they have access to sufficient concentrations of appropriate nutrients. In natural infections, the escalating immune response acts as a negative control on bacterial populations, but this is unlikely to function efficiently in a mirror bacterial infection for the reasons outlined in Sections 4.1–4.3. Instead, the mirror bacterial population could plausibly grow until it severely disrupts vital bodily processes, resulting in host death. In typical natural bacterial infections, death is caused by sepsis, which is a severe immunological dysregulation. An analogous severe immune dysregulation appears to be the most plausible cause of death in a mirror bacterial blood infection. Nevertheless, even if a sepsis-like response is avoided, mirror bacteria in high concentrations would likely deplete nutrients, modify blood chemistry, and cause physical disruptions to blood vessels and organs. Each effect is credibly lethal, and the prospect that an uncontrolled mirror bacterial infection in the blood would not have severe health consequences appears remote.

The usual cause of death from natural systemic bacterial infections is sepsis, a life-threatening condition that arises when the host response to infection injures its own tissues and organs (Singer *et al.*, 2016). While the immune system is usually able to protect humans from bacterial infections, if it is unable to do so, mounting pathogen load and tissue damage drive an extreme, dysregulated, and self-reinforcing hyperinflammation response (Jarczak *et al.*, 2021).

The pathophysiology of sepsis is incompletely understood. In natural infections, the immune system is activated by many pathways, and it is challenging to determine which mechanisms drive sepsis-associated pathology (van der Poll *et al.*, 2017). Stimulation of the immune system by MAMPs or DAMPs appears to play an important role in driving the hyperinflammatory state (Delano & Ward, 2016; van der Poll *et al.*, 2017; Wiersinga *et al.*, 2014), as does failure of regulatory pathways to limit the induced inflammation (Zhang & Ning, 2021). However, no specific MAMP or DAMP signals appear to be strictly necessary for the condition to develop. For example,

inflammation induced by LPS endotoxin can drive sepsis pathophysiology during Gram-negative bacterial infections (Kellum & Ronco, 2023), but LPS is not present in Gram-positive bacteria and therefore cannot play a role in causing sepsis in those types of infections. A variety of similar hyperinflammatory conditions—grouped under several overlapping diagnostic labels such as Systemic Inflammatory Response Syndrome, Cytokine Storm Syndrome, and hypercytokinemia—can occur after widespread tissue injury, such as those resulting from burns, trauma, or ischemia (Chakraborty & Burns, 2024). This suggests that DAMP production and recognition alone may be sufficient to drive life-threatening hyperinflammation, and that DAMPs may play important roles in mediating sepsis during infection (Cicchinelli *et al.*, 2024).

Innate immune activation would be expected to be less potent in the context of a mirror bacterial infection. Compared to natural-chirality pathogens, mirror bacteria would not natively produce immunogenic endotoxin nor virulence factors that would directly damage host tissue; as discussed earlier (Sections 4.1–4.3), most immunological mechanisms of recognition and response to mirror bacteria are likely to be substantially impaired.

Nonetheless, we think sepsis-like immune dysregulation would likely be the most plausible ultimate outcome of an uncontrolled mirror bacterial infection. Although immune recognition of mirror bacteria is expected to be seriously compromised, mirror bacteria remain likely to trigger some immune responses. The complement system, for example, could still be activated through the alternative (and possibly classical) pathway, resulting in the release of the anaphylatoxins C3a and C5a. Both promote inflammation and activate professional phagocytes; C5a in particular plays a crucial role in causing many of the life-threatening effects of sepsis (Ward, 2010). Cross-reactivity of PRRs with mirror macromolecules, which would likely be very weak individually, could plausibly also trigger an immune response if mirror bacterial concentrations expand and the number of those interactions multiplies. Tissue damage caused either indirectly by the immune system or directly due to other consequences of mirror bacterial infection (e.g., by local nutrient depletion, phagocyte cell death, or toxic metabolites) could also trigger PRRs through DAMPs, which could nonspecifically activate immune cells and further drive the sepsis-like response. This residual immunological function trigger dby DAMPs may not be competent to clear a mirror bacterial infection, yet still be sufficient to trigger sepsis once this infection becomes overwhelming.

This observation is consistent with the clinical experience of infection in immunocompromised patients: despite the hyperinflammatory nature of the condition, these patients are at an increased risk of developing sepsis (Greenberg *et al.*, 2018; Kumar *et al.*, 2015; Tolsma *et al.*, 2014). Sepsis can occur in serious infection across bacterial species, and does not appear to require any specific pathogen factor. Indeed, to the best of our knowledge, overwhelming systemic bacterial infection essentially always results in sepsis, regardless of the patient's immune status or bacterial species, and as noted above, similar hyperinflammatory conditions also arise in response to widespread tissue damage even absent infection. While the exact processes that would initiate a sepsis-like immunological dysfunction during mirror bacterial infection are unclear, it appears plausible that such dysfunction would ultimately be triggered, and the result could resemble sepsis seen in natural bacterial infection.

Sepsis is lethal through multiple distinct and overlapping mechanisms. Generalized inflammation disrupts and damages most organ systems, resulting in progressive multiple organ failure (Gotts & Matthay, 2016) and hemodynamic collapse ("septic shock"). Perhaps paradoxically, late-stage sepsis also causes immunosuppression, which can lead to the acquisition of secondary hospital-acquired infections (Hotchkiss *et al.*, 2013). With intensive supportive care and urgent administration of antimicrobial therapy, sepsis mortality is 25–50% (Cecconi *et al.*, 2018); without treatment, sepsis is expected to be fatal within hours or days.

Although a sepsis-like disease appears to be the most plausible outcome of mirror bacterial infection, it is difficult to be certain given the expected properties of mirror bacteria. Another possibility is that the combination of immune evasiveness and limited ability to damage host tissue could result in an unprecedented pathophysiology: uncontrolled infection in internal compartments which does not provoke a severe systemic inflammatory response. If proliferation is not interrupted by host death, mirror bacterial populations could grow to much higher concentrations than typically seen in human or animal models of bacterial disease.

Extremely high mirror bacterial titers in blood are unlikely to be survivable, though the exact pathologies that would first manifest are unclear. A large mass of metabolically active mirror bacteria would consume the nutrients and oxygen carried by the blood. If able to catabolize D-glucose, for instance, mirror bacteria would likely grow until the patient died either from hypoglycemia or hypoxemia, depending on whether D-glucose or oxygen was depleted first. Notably, per unit mass, the metabolic rate of rapidly growing *E. coli* is about a thousand times higher than that of a resting human (Milo & Phillips, 2015); since blood is about 10% of body mass, this suggests that death could occur through this mechanism once mirror bacteria reach a few percent of blood by mass (or  $\sim 10^{10}$  cells/mL for a typical bacterium). Even absent D-glucose catabolism, growth on achiral nutrients like glycerol or uric acid, or on other accessible nutrients like L-alanine and other L-amino acids, could eventually lead to mirror bacterial populations large enough to cause hypoxemia, among other problems. Waste products from bacterial metabolism could also cause fatal disruptions to host clinical chemistry, such as acidosis secondary to lactic or acetic acid production.

Very high concentrations of mirror bacteria could also cause physical disruptions. Because mirror bacteria would adhere at most very weakly to the blood vessel endothelium and (in most cases) are unlikely to form biofilms, such physical disruptions would be unlikely to occur until mirror bacterial concentrations become very high. At that point, mirror bacteria could thicken blood and disrupt its flow, leading to a hyperviscosity syndrome (Kwaan & Bongu, 1999). The coagulation cascade is finely controlled, so alterations in viscosity, dilution of coagulation proteins, and other gross changes secondary to high bacterial concentrations could predispose patients to hemorrhage, clotting, or both simultaneously. Mirror bacteria may also diffuse from blood and lymphatics into the end-organs they supply, disrupting their tissues and impairing their functions.

It is not inconceivable that a systemic mirror bacterial infection could stabilize if key nutrients are depleted, although this appears unlikely. Significant depletion of oxygen or glucose from the blood is likely to kill the patient. For a mirror bacterium that cannot consume D-glucose, depletion of achiral and other available nutrients within blood would still plausibly be harmful. Even if growth were nutrient-limited, the infection would only be benign if mirror bacterial concentrations were kept

sufficiently low to avoid an adverse immune response. Even if such stabilization occurred, the equilibrium could be brittle over longer time periods, with later physiological disturbances (e.g., age, intercurrent illness, or even long-term harms caused by a stable mirror bacterial population) pushing the host back into the uncontrolled growth regime.

To summarize, it appears plausible that unchecked mirror bacterial replication could result in a severe and dysregulated inflammatory response similar to sepsis. Even if this was avoided, mirror bacterial infection would probably cause severe harm through a number of other mechanisms. For example, low but chronic levels of inflammation, even if they do not eventually amount to hyperinflammation, are associated with a plethora of health issues ranging from cardiovascular disease to autoimmune disorders and cancer (Furman et al., 2019). The human body is complex, and accurately predicting the consequences of a putative mirror bacterial infection is impossible. Nevertheless, if infection is established and effective treatment is not available (as discussed in Chapter 5), the prognosis would likely be bleak.

## The unusual properties of mirror bacteria could create significant potential for misuse

As described throughout this chapter, mirror bacteria would have very unusual properties as potential human pathogens. A mirror bacterium would have unprecedented immune evasion capabilities because of incompatibilities between its mirror macromolecules and natural-chirality host defenses. On the other hand, mirror bacteria would, by default, lack specific adhesion mechanisms and other virulence factors. Even if present, these would mostly be dysfunctional, as they rely on chiral interactions with host proteins. For any given mirror bacterium, it is difficult to know whether immune evasion would suffice to overcome other disadvantages. Nevertheless, it appears plausible that sufficiently robust mirror bacteria, such as mirror *E. coli* strains, could prove to be virulent pathogens without any additional engineering.

If robust biocontainment measures are engineered into mirror bacteria, it should be possible to create specific strains that pose minimal risks to human health (see Section 3.3). However, the fact that, absent robust biocontainment, a generalist mirror bacterium might be a novel and highly dangerous pathogen is very concerning from both a biosafety and, particularly, a biosecurity standpoint. Even if the immune system or other barriers provide some level of protection against lethal mirror bacterial infection, invasive mirror bacteria would rapidly evolve (see Section 8.4) and could therefore become increasingly dangerous to humans.

A malicious actor could intentionally seek to increase the harms caused by a mirror bacterium by providing it with abilities that are unlikely to evolve through natural selection. Many methods for increasing virulence might be technically straightforward if mirror bacteria were created and became generally accessible. Describing potential "weaponization" steps in more detail (beyond merely disabling biocontainment measures) is outside the scope of this report. Nevertheless, it appears plausible that even if some biological barriers make mirror bacterial infection by, for example, a mirror *E. coli* much less likely than anticipated here, these deficiencies could be circumvented by a sufficiently skilled malicious actor. With continued progress in biotechnology in general, and mirror

biology more specifically, it would become progressively easier for a malicious actor to bridge the gap between a "safe" mirror bacterium and a highly pathogenic one.

## **Chapter 5: Medical Countermeasures**

Thomas V. Inglesby<sup>1,2,3</sup>, Gregory Lewis<sup>4</sup>, David A. Relman<sup>5,6,7</sup>, Johanna M. Sweere<sup>8</sup>, Brian Wang<sup>9</sup>

Public health and medicine have a large suite of interventions to manage infectious disease. Many of these ultimately rely on stereospecific interactions, and thus the reverse chirality of mirror bacteria reduces the breadth and efficacy of potential responses. Adapting interventions (or pioneering new ones) that could be effective versus mirror bacteria would likely be challenging.

This chapter focuses on possible pharmaceutical countermeasures to mirror bacterial infection. Non-pharmaceutical interventions, such as PPE, handwashing, and sterilization techniques, are generally insensitive to chirality and could be used to reduce human exposure to mirror bacteria. Akin to natural-chirality pathogens, it appears unlikely these measures alone would be sufficient to guarantee protection. If a sufficiently invasive mirror bacterium were to become common in many environments, a possibility explored in Chapter 8, then these non-pharmaceutical interventions to reduce exposure would carry significant costs and may not be sufficient to reliably prevent infection, though this would depend on abundances in the environment.

Chapter 4 detailed how mirror bacteria could pose serious risks to human health upon exposure and subsequent infection. Pharmaceutical interventions such as antibiotics and vaccines could potentially offer robust means to prevent infection if administered prophylactically or to cure infections once diagnosed. This chapter considers the prospects of developing antibiotics and vaccines against mirror bacteria, and touches upon other potential pharmaceuticals such as antibody or phage therapies that are used in clinical practice to treat bacterial infections. In principle, adapting these modalities to target mirror bacteria appears feasible; whether they could substantially reduce the plausible danger posed by mirror bacteria to humans is very uncertain, for a number of reasons.

First, antimirror pharmaceuticals that are potent *in vitro* may not prove clinically effective *in vivo*. While mirror bacterial infections in otherwise healthy individuals would be importantly different from natural pathogen infections in immunodeficient individuals, the deficits in the immune system's response to mirror bacteria may resemble severe immunocompromise (see Chapter 4 for further discussion). Infection with natural-chirality bacteria in a severely immunocompromised patient is often fatal despite maximal medical intervention, and prophylactic antibiotic coverage and extensive immunization are not sufficient to ensure a normal life expectancy for those with severe

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup> Center for Health Security, Bloomberg School of Public Health, Johns Hopkins University, Baltimore, Maryland, USA; <sup>2</sup> Environmental Health and Engineering, Johns Hopkins Bloomberg School of Public Health, Baltimore, Maryland, USA; <sup>3</sup> Johns Hopkins Bloomberg School of Medicine, Baltimore, Maryland, USA; <sup>4</sup> Media Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, USA. <sup>5</sup> Department of Medicine, Stanford School of Medicine, Stanford University, Stanford, California, USA; <sup>6</sup> Department of Microbiology and Immunology, Stanford University School of Medicine, Stanford, California, USA; <sup>7</sup> Infectious Diseases Section, Veterans Affairs Palo Alto Health Care System, Palo Alto, California, USA; <sup>8</sup> Mirror Bacteria Dialogues Fund, New York, New York, USA; <sup>9</sup> Panoplia Laboratories, Cambridge, Massachusetts, USA immunodeficiencies (Hanna & Etzioni, 2014; Picard *et al.*, 2010). It is thus uncertain whether pharmaceutical interventions—or a combination of interventions—could reliably treat or prevent life-threatening mirror bacterial infection.

Second, even for natural-chirality pathogens, discovery and development of a clinically effective pharmaceutical is not easy. Doing so rapidly in the midst of an emerging disease outbreak is typically harder still. Mirror bacteria may pose additional challenges for countermeasure development versus a newly emerging natural-chirality pathogen, including the likely necessity for live organisms to be handled in high-containment facilities, the need for mirror-image reagents and chemistry for experimental manipulation, and little relevant pre-existing expertise on these organisms within the scientific community. Scientific progress over the coming decades may lessen these challenges, but it seems likely they will remain formidable.

Third, the successful development of a highly effective medical countermeasure does not guarantee those who need it will receive it. Large-scale manufacture of a newly developed antimirror countermeasure may be difficult, especially if it demands extensive modification of existing manufacturing capacity, or new facilities to be brought online. A large proportion of the world population lives in low-income countries, where preventable deaths from a lack of access to relatively inexpensive medicines are commonplace (Adenowo *et al.*, 2015; UNICEF, 2024): access for these populations to antimirror medical countermeasures, which would probably be substantially more expensive, is likely to be limited. The possible combination of environmental prevalence, frequent human exposure, and limited durability of protection to a mirror bacteria could, in a worst-case scenario, make protection of their population difficult even for the wealthiest nations. Mirror bacteria pose further challenges to other aspects of outbreak responses (e.g., diagnostics, sample management) similar to those noted for countermeasure development, but these are beyond the scope of this chapter.

Last, while in this chapter we focus on protecting humans, mirror bacteria could potentially also infect animals and plants, and invade natural ecosystems (see Chapters 6, 7, and 8). Countermeasures to protect agriculture and the natural environment are discussed in Chapters 7 and 8. As the prospects of preventing or remediating ecological damage caused by mirror bacteria appear to be very limited (see Section 8.6), environmentally invasive mirror bacteria could cause severe harms even if humans could themselves be protected from direct infection through medical countermeasures.

## 5.1 New antimirror compounds could be developed to target mirror bacteria, but most existing antibiotics would not function

Antibiotics are small molecules that target mechanisms of survival and replication that are absent in humans but present in many or all bacteria. They constitute one of the most important and effective methods to combat bacterial infections. Some existing antibiotics could be repurposed to combat mirror bacterial infection, mirror enantiomers of existing antibiotics could be developed (with varying degrees of difficulty), and antibiotics that exploit the reversed chirality of mirror bacteria to be selectively toxic toward them could possibly be developed.

### Some existing antibiotics could be repurposed to combat mirror bacterial infection

Most existing antibiotics function by inhibiting chiral enzymes or targeting other specific chiral biomolecules within bacteria (Kohanski *et al.*, 2010). The enantiomers of chiral antibiotics are generally not expected to retain bactericidal activity, as they are unlikely to bind tightly to the biomolecules targeted by their enantiomer. This has been confirmed experimentally in a number of cases, including for chloramphenicol (Maxwell & Nickel, 1954), fosfomycin (Hirota *et al.*, 1998), and polymyxin B (Slingerland *et al.*, 2022), as well as for enantiomers of specific cephalosporins (Cesta, 2021), oxazolidinones (Brickner, 1996; Fortuna *et al.*, 2014), penicillins (Cesta, 2021), and tetracyclines (Korst *et al.*, 1968; Muxfeldt *et al.*, 1979). The fluoroquinolones, built around a core achiral structure, provide a partial exception with enantiomers sometimes having similar activity (Chu *et al.*, 1991; Liu *et al.*, 2005), though the enantiomer of levofloxacin (sometimes used racemically as ofloxacin) has much lower activity (Morrissey *et al.*, 1996).

Just as the enantiomers of chiral antibiotics usually have little or no antibacterial activity, the chiral antibiotics themselves would usually have little or no antibacterial activity against mirror bacteria. As a result, the utility of antibiotics for the treatment of mirror bacterial infections can be informed by their structures. Chiral antibiotics would be expected to lose activity against mirror bacteria, as can be verified by testing their enantiomers against natural-chirality bacteria. Achiral and racemic antibiotics—those consisting of a mixture of both enantiomers—should by contrast remain active. In Table 5.1 we classify the major structural classes of antibiotics included in the WHO Essential Medicines List (World Health Organization, 2023) by whether they are expected to be stereospecific or not and, if so, the likely ease of producing mirror-image versions to combat mirror bacteria.

Chirality would not be not the sole determinant of susceptibility: in the same way that a particular antibiotic may only be potent against some normal-chirality bacterial species, a given mirror bacteria may not be susceptible to all antibiotics, even if they were of the "correct" chirality. Similarly, susceptibility is not the sole determinant of clinical effectiveness: for example, the pharmacokinetic properties of an antibiotic may make it incapable of reaching a particular site of infection at a sufficient concentration, or may preclude prophylactic use. It is therefore uncertain whether an antibiotic that would be expected to retain its potency across various mirror bacterial species would in fact effectively treat infections caused by a particular mirror bacterial strain.

Achiral and racemic antibiotics—which are expected to be as potent against mirror bacteria as they are against natural-chirality bacteria—are therefore promising therapeutic candidates against susceptible mirror bacterial infections, as they could be straightforwardly repurposed with little further development or clinical safety testing. Of the major antibiotic classes, only (fluoro)quinolones, sulfonamides, and trimethoprim derivatives consist mainly of achiral or racemic members, alongside the miscellaneous antibiotics metronidazole and nitrofurantoin. Co-trimoxazole, an inexpensive broad-spectrum formulation that combines trimethoprim and sulfamethoxazole, is widely used, including in developing nations (Church *et al.*, 2015), and quinolones are also among the most commonly used antibiotics globally (Browne *et al.*, 2021).

Chapter 5	5:	Medical	Countermeasures
-----------	----	---------	-----------------

Antibiotic Class	Examples on 2023 WHO Essential Medicines List	Target	Activity vs. mirror	References
Achiral or racemic	molecules—plausibly usable i	mmediately		
Fluoroquinolones†	Ciprofloxacin Ofloxacin	Topoisomerases Topoisomerases	Yes Racemate	Morrissey et al., 1996
Sulfonamides	Sulfadiazine, sulfamethoxazole	Dihydropteroate synthase	Yes	
Miscellaneous	Metronidazole Nitrofurantoin Trimethoprim	DNA DNA Dihydrofolate reductase	Yes Yes Yes	
Chiral molecules, p	roduced synthetically—enant	tiomers plausibly synthetical	ly accessibl	e, but safety unclear
Carbapenems	Meropenem	Penicillin-binding proteins	No	
Oxazolidinones	Linezolid	50S ribosomal subunit	No*	Brickner, 1996
Miscellaneous	Chloramphenicol Fosfomycin	50S ribosomal subunit MurA enzyme	No* No*	Maxwell & Nickel, 1954 Hirota et al., 1998
Chiral molecules, p	roduced through biosynthesi	s—enantiomers unlikely to b	e easily syn	nthesized
Aminocyclitol	Amikacin, gentamicin, plazomicin, spectinomycin, streptomycin	30S ribosomal subunit	No	
Cephalosporins	Cefalexin, cefazolin, cefiderocol, cefixime, cefotaxime, ceftazidime, ceftriaxone, ceftolozane, cefuroxime	Penicillin-binding proteins	No*	Cesta, 2021
Lincosamides	Clindamycin	50S ribosomal subunit	No	
Glycopeptides	Vancomycin	Lipid II	No	
Macrolides	Azithromycin, clarithromycin, erythromycin	50S ribosomal subunit	No	
Penicillins	Amoxicillin, ampicillin, benzylpenicillin, cloxacillin, phenoxymethylpenicillin, piperacillin	Penicillin-binding proteins	No*	Cesta, 2021
Polymyxins	Colistin, polymyxin B	Lipid A	No*	Slingerland et al., 2022
Tetracyclines	Doxycycline, tetracycline	30S ribosomal subunit	No*	Korst <i>et al.</i> , 1968; Muxfeldt <i>et al.</i> , 1979

## Table 5.1: Classes of existing antibiotics and characteristics relevant to their applicability for treatment of mirror bacterial infections

All antibiotics in the 2023 WHO Essential Medicines List are listed, excluding antileprosy and antituberculosis medicines, which tend to be very specialized. Biomolecular targets and mechanisms of action are described in (R. Anderson *et al.*, 2012; Kohanski *et al.*, 2010) unless otherwise noted. Activity against mirror bacteria is inferred from chirality; for those with an asterisk (\*), enantiomer compounds from the class have been tested and found to lack antibacterial activity. †The basic fluoroquinolone scaffold is achiral, but some antibiotics in this class are chiral.

Enantiomers of existing antibiotics could be produced, although this appears to be difficult for most classes

The enantiomers of chiral antibiotics may also be effective against mirror bacterial infections, as they should interact with chiral targets within mirror bacteria identically to how existing antibiotics interact with their targets in natural bacteria. However, unlike achiral and racemic antibiotics, the enantiomers of chiral antibiotics may possess a different pharmacokinetic and side-effect profile than their natural-chirality counterparts, and consequently would require testing and separate regulatory authorization (Center for Drug Evaluation and Research, 1992; European Medicines Agency, 1993). However, this process could be accelerated or abbreviated in the context of a major outbreak.

Large-scale production of a mirror-image antibiotic may be relatively feasible for those chiral antibiotics that are produced synthetically (Table 5.1). These include the carbapenems, monobactams, oxazolidinones, and the miscellaneous antibiotics chloramphenicol and fosfomycin (Brewer & Hellinger, 1991; Cao *et al.*, 2019; Foti *et al.*, 2021; Hahn, 1967; Papp-Wallace *et al.*, 2011). Synthesis would require replacing the natural chirality reagents with their enantiomers, but it is plausible that the enantiomers of these antibiotics could be straightforwardly manufactured.

Most classes of existing chiral antibiotics, however, are predominantly produced in whole or in part via biosynthesis, and the biosynthetic steps cannot be easily adapted to generate the enantiomer of their current product (Fedorenko *et al.*, 2015). Given the complexity of these antibiotics, the re-engineering of biosynthetic pathways or the execution of total synthetic routes to produce enantiomeric versions of these antibiotics (Table 5.1) on an industrial scale could present major scientific and practical challenges. It is conceivable that strains of mirror bacteria could be cultivated for these biosynthetic tasks, although this would share many of the same scientific and practical challenges and in addition would raise considerable safety and security concerns (see Chapter 3).

Antibiotics with pore-forming or membrane-disrupting activity could prove an exception to the principle that chiral antibiotics would not be effective against mirror bacteria, as their mechanism of action is often not sensitive to the chirality of the lipid membrane (examples of such antibiotics are listed in Table 8.1 of Chapter 8). In many cases, the enantiomers of these antibiotics have even been shown to be effective at killing bacteria, so these antibiotics would be expected to remain effective against mirror bacteria. Not all membrane-disrupting antibiotics would retain efficacy, however, as some bind to chiral targets in the membrane. Daptomycin, for example, binds to the chiral molecule phosphatidylglycerol, and its enantiomer is much less effective at harming bacteria (Moreira & Taylor, 2022).

Given their lack of specificity, membrane-disrupting antibiotics that would retain efficacy against mirror bacteria are often toxic to humans, which can limit applicability. Gramicidin and bacitracin are the only examples that see occasional clinical use, and they are all but exclusively used for topical infections due to their cytotoxicity (Nguyen *et al.*, 2024; Pavithrra & Rajasekaran, 2020); salinomycin and other ionophores see use only in veterinary medicine (Zhou *et al.*, 2013). It is therefore doubtful that such antibiotics would prove particularly promising drugs to combat mirror bacterial infections.

Novel classes of antimirror antibiotics could be developed to target mirror bacterial infection

Current antibiotic therapies exploit the comparatively minor biochemical differences between prokaryotes and eukaryotes in order to selectively kill microbes while causing few side effects in human patients. Biochemically, all metabolic processes in mirror bacteria will be very different from those in natural-chirality organisms due to the reversed chirality of their enzymes, providing a much wider range of potential antibiotic targets. For example, metabolic pathways such as glycolysis cannot be targeted by antibiotics because they are highly conserved in all living organisms, yet small molecules that selectively inhibit mirror-image glycolytic enzymes would be credible antimirror antibiotic candidates. This abundance of targets could reduce the cost of discovering effective, safe, readily manufactured antimirror therapeutics and prophylactics.

Existing research may be informative. If a compound X is highly toxic to all existing life, but its enantiomer Y is well-tolerated in humans, then Y may be a potent antimirror compound. For example, the enantiomers of the most toxic known nucleoside analog, including but not limited to those with potent anticancer, antiviral, and antibacterial activities (De Jonghe & Herdewijn, 2022; Niu & Tan, 2015), may be promising antimirror compound candidates. Toxicity studies might assess the comparative safety of candidate antimirror compounds and the minimum inhibitory concentrations of their enantiomers against existing bacteria, with promising candidates entering drug development.

On the other hand, achieving *in vitro* efficacy is typically easy for antibiotic discovery campaigns, and toxicity is typically mediated via highly unpredictable off-target interactions (largely driven by physicochemical properties). Developing new antibiotics from discovery to clinical testing and approval takes years, and historically the interval between discovery and clinical introduction has averaged around a decade (Gigante *et al.*, 2022; Hutchings *et al.*, 2019), though some of this disappointing track record may be due to limited economic incentives for rapid commercial antibiotic development alongside the scientific challenges (M. Anderson *et al.*, 2023). It is unclear whether a dedicated antimirror antibiotic development effort could be expected to outperform this historical average during the urgency of a major outbreak.

### The clinical efficacy and overall utility of antimirror compounds is unclear

Even if existing antibiotics or novel antimirror compounds could kill mirror bacteria, they would not necessarily be clinically effective. As discussed in Chapter 4, human immunity is anticipated to be impaired versus mirror bacterial challenge, and infection is frequently life-threatening in roughly analogous immunodeficiencies despite available options for antibiotic treatment. For antimirror compounds to be useful therapeutically, infections would need to be diagnosed and treatment begun before irreversible harm had occurred. It is therefore not clear that post-infection antibiotics would be sufficient to cure a mirror bacterial infection, or to mitigate serious health effects even if they could prevent fatality.

If post-infection antimirror treatments are ineffective or insufficiently reliable, the only available strategy would involve the prophylactic use of antimirror compounds (vaccination may be another

means to prevent infection, discussed below). Prophylactic antibiotics are commonly prescribed for the immunosuppressed or immunocompromised (Picard *et al.*, 2011; Seddon & Bhagani, 2011; Taplitz *et al.*, 2018), although they are not always sufficient to prevent infection depending on the magnitude of the deficiency. Given the important disanalogies between mirror bacterial infections and natural bacterial infections, it is difficult to know whether antimirror compounds taken prophylactically would be effective. Comprehensive prophylaxis would require antimirror compounds that were safe to use indefinitely, ideally with minimal side effects. Comprehensive prophylaxis for all living humans would likely require antimirrors to be produced at a scale exceeding total current worldwide antibiotic production.

Antimicrobial resistance could also evolve in mirror bacteria in a similar fashion to their natural-chirality counterparts. Mirror bacteria are expected to have a comparable capacity for *de novo* mutation, but would by default have limited ability for horizontal gene transfer (see Section 8.4). As such, resistance mechanisms that depend on point mutations in antimirror targets would be much more accessible to mirror bacterial populations than those reliant on a novel protein (e.g., a hypothetical mirror-image  $\beta$ -lactamase). It is also possible a malicious actor could intentionally engineer antimicrobial resistance to counter existing achiral antibiotics, resist specific known antimirrors, or boost overall resilience to antimirror compounds.

# **5.2** Conjugate vaccines could plausibly be developed against mirror bacteria

In addition to antibiotics, bacterial vaccines have historically played an important role in protection against bacterial infection (Osterloh, 2022). All existing bacterial vaccines appear to elicit immunity against pathogen-specific chiral antigens, and so pre-existing vaccines would be unlikely to provide effective protection against mirror bacteria. Even if the mirror bacteria share an antigen that is targeted by an existing vaccine, immunological memory is unlikely to extend to the mirrored antigen as B cell receptors, antibodies, and T cell receptors are all chiral molecules (see Section 4.3). A new vaccine specific to the mirror bacterial species would need to be developed.

Bacterial vaccines harness a range of different technologies: inactivated, live attenuated, recombinant protein, protein-polysaccharide conjugate, and outer membrane vesicle vaccines have all been licensed. Newer technologies such as plasmid DNA and mRNA are currently being explored for the development of novel bacterial vaccines (Frost *et al.*, 2023; Mba *et al.*, 2023; Osterloh, 2022). However, just as mirror bacterial infections are unlikely to stimulate a protective immune response due to the expected inability of the human immune system to process mirror bacterial protein antigens, the same is likely to be true of vaccination strategies that rely on immunogens composed solely of mirror protein antigens. As a result, inactivated, live attenuated, recombinant protein, and outer membrane vesicle vaccines would not be expected to prove effective. Nucleic acid vaccines would also not be effective against mirror bacteria because they cannot encode mirror protein antigens.

On the other hand, conjugate vaccines may be a promising approach against mirror bacteria. Conjugate vaccines stimulate an immune response to a normally non-immunogenic antigen by conjugating an immunogenic carrier to it (Rappuoli *et al.*, 2019). Glycoconjugate vaccines—which consist of a carrier protein or peptide conjugated to a polysaccharide—direct antibody responses against polysaccharides that are prevalent on the surface of bacteria, ultimately resulting in bacterial cell death via opsonophagocytosis or complement-mediated lysis (L. L. Lu *et al.*, 2018; Romero-Steiner *et al.*, 2006). These vaccines have seen success in the prevention of infections by encapsulated bacteria, as conjugate vaccines against *Streptococcus pneumoniae*, *Haemophilus influenzae* type B (Hib), and *Neisseria meningitidis* have dramatically reduced incidences of infection with these pathogens in children over the past two decades (McHugh, 2020; Pollard *et al.*, 2009; Wasserman *et al.*, 2021). Peptide-protein conjugates use a similar strategy to direct antibody responses to peptides that are poor immunogens. Although no peptide vaccines are in common use, a number are in development for infectious diseases (Eskandari *et al.*, 2021; Ou *et al.*, 2020) and cancer therapy (Abd-Aziz & Poh, 2022; Malonis *et al.*, 2020). They may be particularly promising for vaccines against mirror bacteria as the reversed chirality of mirror bacterial proteins is expected to render them poorly immunogenic (see Section 4.3).

Conjugate vaccines against mirror bacteria could be developed by conjugating a non-mirror carrier protein or peptide to a mirror polysaccharide or peptide. The non-mirror carrier protein would be expected to stimulate a T cell response that could then assist activation of B cells that recognize the conjugated mirror antigen. Such vaccines would be expected to generate antibodies that bind bacteria bearing the mirror antigen, potentially preventing infection (Figure 5.1).

Although promising, the sufficiency of conjugate vaccine-induced immunity for protection against mirror bacterial infections remains uncertain. While existing glycoconjugate vaccines in clinical use are known to induce immunological memory, a memory response may take four or more days to become established upon re-encountering the antigen, which may not be rapid enough to prevent colonization by bacteria with short incubation periods (Pollard *et al.*, 2009). In the context of infection by a rapidly invasive mirror bacterium largely unimpeded by a natural innate immune response, the induction of a memory response may be too slow or simply inadequate. Conjugate vaccines are also unlikely to generate a strong CD8<sup>+</sup> T cell response, which could impair efficacy against mirror bacteria capable of surviving opsonophagocytosis or otherwise entering and growing inside cells (Osterloh, 2022; Shepherd & McLaren, 2020).

In the context of substantial impairment of native B cell and CD8<sup>+</sup> T cell responses, long-term conjugate vaccine-induced protection would largely rely on the maintenance of high serum anti-polysaccharide antibody titers. Glycoconjugate vaccines can (though do not always) induce persistent antibody titers; for example, antibody titers are maintained at protective levels five years after a single dose of a meningococcal glycoconjugate vaccine in adolescents (Quiambao *et al.*, 2017). However, given the failure of key components of the immune system outlined in Chapter 4, serum antimirror antibodies alone may not be sufficient to prevent mirror bacterial infection. If mirror conjugate vaccines are achievable, antibody levels needed for protection are expected to be higher than the usual dosage, and frequent re-boosting to maintain protection may be necessary.

Manufacturing challenges may hinder the timely development and distribution of glycoconjugate vaccines against mirror bacteria due to the need for large-scale mirror polysaccharide production. Current manufacturing of glycoconjugate vaccines typically involves the large-scale fermentation of



#### Figure 5.1: Conjugate vaccines may improve adaptive immune recognition of mirror bacteria

**A.** Expected B cell response against a vaccine composed of mirror bacterial polysaccharides. Follicular dendritic cells and B cell receptors should bind mirror polysaccharides, and their long, repeating structure may activate receptor signaling. However, without the ability to present peptide antigens to and receive signals from T follicular helper (Tfh) cells, these B cells would be unlikely to mount an IgG antibody response or develop immune memory and are likely to undergo apoptosis instead (Section 4.3). **B.** Expected B cell response against a vaccine composed of mirror polysaccharides conjugated to natural-chirality carrier proteins. Upon binding follicular dendritic cells, glycoconjugates are presented to B cells and bind and activate B cell receptors. This results in internalization of the glycoconjugate and proteolysis of the protein carrier into peptides that can be presented to Tfh cells. Peptide presentation results in T cell help and a functional IgG antibody response and immune memory against the target, in this case the mirror polysaccharide. Conjugate vaccines using mirror peptides could also be feasible.

pathogenic bacteria, purification of the desired polysaccharide component, and chemical conjugation to the carrier (Kay *et al.*, 2019; Stark *et al.*, 2021), although direct synthesis of the polysaccharide component has been successfully used in the Quimi-Hib vaccine licensed in Cuba (Verez-Bencomo *et al.*, 2004). Manufacture of antimirror peptide-protein conjugates by chemical synthesis would be more straightforward, though this approach would rely on identifying appropriate peptide targets on the mirror bacterium.

Developing such vaccines as a preparedness measure—that is, in advance of a potential outbreak—would be challenging given the diversity of bacterial antigens a future mirror bacterial species could possess. Developing such a vaccine amidst a mirror bacteria outbreak could be even more challenging, especially if the outbreak is spreading rapidly. Standard timelines for vaccine development span 5–10 years (Pronker *et al.*, 2013), although vaccine development and large-scale deployment in a major public health emergency could be much faster, as was seen in the COVID-19 pandemic.

# **5.3** The efficacy of other countermeasures against mirror bacterial infection is unclear

Although antibiotics and vaccines are the most important medical countermeasures against bacterial infections, there is a mix of rarely used and early-stage therapeutic modalities that could be deployed against mirror bacteria as alternative or adjunct therapies. Compared to antibiotics and vaccines, these other countermeasures tend to have limited (or expensive) production, a much more limited evidence base, or require further technical development, limiting their promise. Other hypothetical countermeasures to mirror bacteria could be enabled by future advances in science and technology, but guessing at their promise and feasibility is beyond the scope of this chapter.

It may be possible to develop antibody therapy against mirror bacteria, but scaling would be difficult

Antibodies can be administered to prevent or treat infection. A broad repertoire of antibodies can be harvested from healthy donors and used as replacement therapy for patients with antibody-related immunodeficiencies (Orange *et al.*, 2006). Antibodies with specific activity against a given pathogen can be harvested from humans or animals that have been exposed to the pathogen and form a common element in post-exposure prophylaxis for rabies and tetanus (Forrat *et al.*, 1998; Haradanhalli *et al.*, 2022). Although more commonly used in non-infectious disease, monoclonal antibodies can be raised to target a specific pathogen target, such as palivizumab for respiratory syncytial virus (Fenton *et al.*, 2004).

As a novel pathogen, the chance of any pre-existing antibody therapy being efficacious against mirror bacteria is remote. Most existing monoclonal antibody treatments were developed by the immunization of transgenic mice (R.-M. Lu *et al.*, 2020), and given that mirror bacteria are unlikely to provoke an adaptive immune response in vertebrates (see Section 4.3), this strategy would not straightforwardly work. Fully *in vitro* methods such as phage display should, however, remain viable. Even if neutralizing antibodies can be generated, production is expensive (especially if monoclonal antibodies are necessary), and prolonged administration to large vulnerable populations is expected to be infeasible.

#### Chapter 5: Medical Countermeasures

Mirror phages could allow for phage therapies to be developed against mirror bacteria

First pioneered in the 19th century, the use of phages to eliminate bacterial infections has seen renewed interest given the mounting challenge of antimicrobial resistance. Besides offering an alternative treatment modality, combination treatments of phages and antimicrobials could have synergistic effects in cases where phage-resistance adaptations in bacteria increase their antimicrobial sensitivity (Hitchcock *et al.*, 2023; Lin *et al.*, 2017; Oromí-Bosch *et al.*, 2023). Except for a few countries, however, phage therapy is largely restricted to experimental or compassionate use (Hitchcock *et al.*, 2023; McCallin *et al.*, 2019).

Reversed chirality renders mirror bacteria completely resistant to all existing bacteriophages, so mirror phages would need to be created to target mirror bacterial species. Creating such mirror phages is not yet possible, but would likely be easier than creating mirror bacteria; given an outbreak of mirror bacteria it should be possible to create mirror phages by creating mirror-image genomes copied from natural phages. Analogous procedures have already been used to create (natural-chirality) phiX174 and T7 bacteriophage genomes from synthetic oligonucleotides (Chan *et al.*, 2005; Smith *et al.*, 2003). Numerous synthetic phages have also been generated from cell-free systems, including for bacteria exhibiting transformation efficiencies too low for *in vivo* production (Mitsunaka *et al.*, 2022).

Mirror phage therapy may have some advantages compared to non-mirror phage therapy. Mirror phages are likely more resistant to clearance by the immune system, and because the phages would be exclusive to mirror species, there would be no risk of horizontal gene transfer with natural organisms. Mirror phages, however, could also present some additional risks. As noted in Section 8.4, mirror bacteria would by default lack efficient mechanisms for horizontal gene transfer. While *de novo* mutations would accumulate and cause the initial strain to diversify, adaptations could not spread between strains. By providing a mechanism for horizontal gene transfer, mirror phages could potentially accelerate the evolutionary processes.

Even aside from the more unusual challenges associated with a mirror phage, phage therapy remains nascent. The clinical efficacy of a hypothetical "mirror phage therapy", especially in the context of deficient host responses to mirror bacteria, is unknown. It is also unknown whether mirror bacteria would evolve resistance to clinical mirror phage therapies. There are likely significant practical challenges in designing an effective mirror phage and deploying it at a large scale. The prospects of using mirror phages as a countermeasure to mirror bacteria in the environment are considered further in Chapter 8.

## **Chapter 6: Animal Infection**

Deepa Agashe<sup>1</sup>, Damon J. Binder<sup>2</sup>, Timothy W. Hand<sup>3</sup>, Matthew L. Nicotra<sup>4</sup>, David A. Relman<sup>5,6,7</sup>

Chapter 4 describes the reasons why a mirror bacterium could be expected to evade many of the immunological defenses found in humans. A sufficiently robust mirror bacterium, if translocated past the body's epithelial barriers, could plausibly cause a life-threatening systemic infection.

In this chapter, we consider whether mirror bacteria could pose similar risks to other animals. Many aspects of the innate immune system are conserved between vertebrates and invertebrates (Buchmann, 2014; Hoffmann *et al.*, 1999; Janeway & Medzhitov, 2002), including the use of specific pattern recognition receptors (PRRs) that recognize chiral ligands (Brennan & Gilmore, 2018) and the killing of bacteria through phagocytic cells (Hartenstein, 2006) and antimicrobial peptides (Zasloff, 2002). As is the case in humans, these mechanisms may be relatively ineffective against mirror bacteria. However, given the largely underexplored diversity of animal species (at least immunologically), it is plausible that some animals would be able to mount a robust defense. Variation in susceptibility could also arise due to diversity in physical barriers to bacterial infection, as well as physiological and behavioral differences.

Section 6.1 discusses other vertebrates, which are immunologically similar to humans and possess an adaptive immune system (Flajnik, 2018). The expected defects in the human immune system—which are described in detail in Chapter 4—will likely be shared by most other vertebrates, suggesting that many vertebrates would be similarly susceptible to infection. Transmission of mirror bacteria between infected vertebrate hosts could facilitate their spread through the environment and, ultimately, transmission from the environment to humans.

Section 6.2 discusses invertebrates, focusing on insects and nematodes. In the best studied insect model, the fruit fly *Drosophila melanogaster*, antibacterial immunity primarily relies on recognition of peptidoglycan, which is chiral. Like the mirror microbe-associated molecular patterns (MAMPs) discussed in Chapter 4, mirror peptidoglycan is unlikely to be recognized by insect PRRs. It therefore appears plausible that many insects would also be vulnerable to mirror bacterial infection. Not only are insects ecologically crucial, they also play important roles in disease transmission, including

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup>National Centre for Biological Sciences, Tata Institute of Fundamental Research, Bengaluru, India; <sup>2</sup> Open Philanthropy, San Francisco, California, USA; <sup>3</sup> Pediatrics Department, Infectious Disease Section, UPMC Children's Hospital of Pittsburgh, University of Pittsburgh School of Medicine, Pittsburgh, Pennsylvania, USA; <sup>4</sup> Center for Health Security, Bloomberg School of Public Health, Johns Hopkins University, Baltimore, Maryland, USA; <sup>5</sup> Department of Medicine, Stanford School of Medicine, Stanford University, Stanford, California, USA; <sup>6</sup> Department of Microbiology and Immunology, Stanford University School of Medicine, Stanford, California, USA; <sup>7</sup> Infectious Diseases Section, Veterans Affairs Palo Alto Health Care System, Palo Alto, California, USA

between non-human vertebrates and between non-human vertebrates and humans. As a result, insects could provide another potential route for mirror bacteria to spread.

By contrast, the model nematode *Caenorhabditis elegans* is bacterivorous and protects itself by not ingesting pathogenic bacteria and by recognizing specialized pathogen signals within the intestines. Mirror bacteria would likely be recognized as prey but not as pathogens, and may not be killed by normal digestive processes involving lytic enzymes. This might allow mirror bacteria to persist within the intestinal tract and cause potentially fatal infection, though it is also plausible that mirror bacteria would simply be excreted along with other waste. Compared to insects and nematodes, the molecular details of pathogen recognition in other invertebrate immune systems are relatively poorly studied, making it difficult to draw more general conclusions about their susceptibility to mirror bacterial infections.

# 6.1 Vertebrate susceptibility to mirror bacterial infection would likely be similar to that of humans

All vertebrates (with the partial exception of jawless vertebrates<sup>9</sup>) possess an innate and adaptive immune system closely analogous to that of humans (Flajnik, 2018; Riera-Romo *et al.*, 2016). Many of the expected deficits in the human immune response to mirror bacteria described in Chapter 4 should therefore be shared by mammals (Tizard, 2023), birds (Kaspers *et al.*, 2021), reptiles (Zimmerman, 2020), amphibians (Ruiz & Robert, 2023), and jawed fish (Rauta *et al.*, 2012). Physiological differences among vertebrate species, particularly differences in immune function, would result in variations in susceptibility to mirror cell infection. For example, Atlantic cod have lost the genes essential for processing and presenting antigens via major histocompatibility complex class II (MHC-II) molecules and have expanded repertoires of MHC class I (MHC-I) and Toll-like receptor (TLR) genes (Solbakken *et al.*, 2016; Star *et al.*, 2011). Similarly, zebrafish have roughly ten times as many nucleotide-binding domain and leucine-rich repeat-containing (NLR) genes as humans or mice (Howe *et al.*, 2016). Most of these immune receptors remain uncharacterized, and some may bind achiral MAMPs, allowing at least partial recognition of mirror bacteria in these species. Nevertheless, it appears likely that a mirror bacterium capable of infecting humans could also infect many species of jawed vertebrates.

Most vertebrate immune mechanisms are chirality-specific and thus would likely be defective

The immune systems of jawed vertebrates are highly conserved. Vertebrate innate immunity relies on chiral-specific PRRs, such as TLRs and NLRs, to detect and respond to conserved features of microbial pathogens such as peptidoglycan and lipopolysaccharide (Buchmann 2014; Magnadóttir,

<sup>&</sup>lt;sup>9</sup> Lamprey and hagfish have a form of adaptive immunity that does not rely on antigen presentation by MHC molecules (Boehm *et al.*, 2018). It is unclear whether these alternative adaptive immune systems rely on the processing and presentation of peptide antigens, making it difficult to assess whether they would be sensitive to mirror bacteria. Lamprey and hagfish innate immune systems are expected to be broadly similar to that of other vertebrates.

2006; Neerukonda & Katneni, 2020; Riera-Romo *et al.*, 2016). Vertebrates also use the complement system, professional macrophages, and lysozyme to target and kill foreign microorganisms (Riera-Romo *et al.*, 2016). Adaptive immunity based on the use of MHC molecules that present pathogen-derived peptides to elicit adaptive cellular and humoral immune responses is conserved across jawed vertebrates (Flajnik, 2018).

As discussed in Chapter 4, each of the aforementioned mechanisms will likely be at least somewhat defective against mirror bacteria in humans. Given that immunological differences between human and other jawed vertebrates are relatively slight, the immunity of jawed vertebrates could be similarly impaired.

## A mirror bacterium capable of infecting humans could potentially infect many vertebrates

Like humans, other vertebrates are susceptible to infection by a diverse range of both obligate and opportunistic bacterial pathogens (Prescott *et al.*, 2022). Some are specialists, while others are capable of infecting a broader range of hosts (Shaw *et al.*, 2020). Common bacterial pathogens, including *Escherichia coli*, *Bacillus anthracis*, *Staphylococcus aureus*, *Salmonella enterica*, and *Pseudomonas aeruginosa*, are all capable of infecting a broad range of mammalian and avian hosts, while the last two are also capable of infecting some reptiles and amphibians as well as plants (Herczeg *et al.*, 2021; Mermin *et al.*, 2004; Pasmans *et al.*, 2020; Schikora *et al.*, 2012). All but *B. anthracis* have been found, at least occasionally, within fish, though only *P. aeruginosa* is known to be pathogenic within fish (Gauthier, 2015; Ziarati *et al.*, 2022). Many other bacterial species can function as human pathogens if given the opportunity—a recent literature survey reported 1 513 known bacterial pathogens (Bartlett *et al.*, 2022)—and there is little reason to expect other vertebrates to be more resistant to bacterial pathogens than humans. A mirror bacterium capable of evading human immunity may therefore be able to infect many vertebrate species.

Temperature strongly influences the growth rates of bacteria, and physiological variation in vertebrate body temperatures could influence their susceptibility to infection. Birds and mammals, for instance, typically maintain constant body temperatures between 35–42°C (Clarke & Rothery, 2008), which is optimal for the growth of *E. coli* and many other human pathogens. Reptiles, amphibians, and most species of fish, though, are generally ectothermic. However, most environments are still warm enough to support *E. coli* growth—which has a minimum temperature requirement of about 8°C (Baka *et al.*, 2013; T. Ross *et al.*, 2003) with some variation between strains—for at least part of the year, albeit at lower growth rates. Temperatures in the deep ocean, and in the polar waters are around -1°C (Kirchman *et al.*, 2009; Yasuhara & Danovaro, 2016), so ectothermic fish living in these environments would not be susceptible to mirror *E. coli* infection.

Gastric acidity also varies between vertebrates: in mammals and birds, gastric pH varies from a low of 0.7 in the common buzzard (Herpol, 1967) to 6.7 or higher in sloths (Denis *et al.*, 1966). With a fasting pH of around 1.5 (Dressman *et al.*, 1990; Russell *et al.*, 1993), the human stomach is much more acidic than those of most other mammals, but similar to those of many avian species (Beasley *et al.*, 2015). This suggests substantial variability in infection risk from the foodborne transmission of mirror bacteria. Given that *E. coli* frequently pass through mammalian stomachs unharmed and can

maintain growth at any pH above 4 in the intestines (Xu *et al.*, 2020), a mirror *E. coli* could presumably do the same. Other physiological and behavioral differences among vertebrates, including variation in skin and mucosa anatomy (Akat *et al.*, 2022; A. A. Ross *et al.*, 2019), dietary habits, and gut microbiota (G. Huang *et al.*, 2022), could affect susceptibility to mirror bacterial infection.

Like humans (discussed in Section 4.4), other vertebrates would likely suffer from leaky barrier defenses. Bacterial entry can occur through open wounds and oral microabrasions. Bacterial translocation also occurs in both the lungs and intestines, though outside humans, the only experimental studies of translocation frequency are limited to mice, which exhibit rates similar to those in humans (Berg, 1999; Ohsugi *et al.*, 1992; Wells *et al.*, 1988). It therefore seems plausible that a mirror bacterium capable of infecting humans would also be capable of infecting many other vertebrates.

## Mirror bacterial proliferation during infections could be fatal and may promote environmental spread

In most vertebrates, the consequences of a mirror bacterial infection are likely to be similar to those described for humans in Chapter 4. Systemic bacterial infections are usually rapidly fatal due to sepsis, a severe immunological dysregulation. Because mirror bacteria would evade most immunological recognition and would, by default, lack toxins or other tissue-damaging virulence factors, it seems possible that mirror bacterial populations within a vertebrate host could become large prior to adverse effects on the host. In humans, it nevertheless appears plausible that very high mirror bacterial concentrations would eventually result in potential fatal sepsis-like immunological dysregulation (see Section 4.5). Even if this were somehow avoided, mirror bacteria within the bloodstream could cause lethal harm through other mechanisms, such as by depleting critical nutrients or by modifying the flow and coagulation of blood.

Unlike humans, the cadavers of non-human vertebrates (particularly wild animals) are usually left to decompose in the environment, which could promote the spread of mirror bacteria. After death, the immune system ceases to function, and mirror bacteria would face competition from endogenous gut-associated bacteria and, after corpse rupture occurs, from environmental bacteria as well (Metcalf *et al.*, 2013). Changes to the internal environment within the cadaver, including the loss of body heat and rapid depletion of internal oxygen (Carter *et al.*, 2007), may also disfavor continued mirror bacterial growth after death. Nevertheless, a large mirror cell population would likely be present in the cadaver for many days after death (the longer-term fate of mirror bacteria within the environment is considered in Chapter 8). Indeed, mirror bacterial numbers upon death might be higher than in typical bacterial infections due to the expected immune deficiencies. This larger mirror bacteria population could increase the length of time that live mirror bacteria persist in the cadaver and, therefore, the overall probability of transmitting mirror bacteria to new hosts. Transmission to new vertebrate hosts might occur either directly through scavenging, or indirectly through insects or other vectors.

Prior to death, an infected vertebrate might transmit mirror bacteria to other vertebrates through a variety of mechanisms including bites and scratches, fecal contamination of food, water, or soil, consumption of infected animals by susceptible predators, and via arthropod vectors such as mosquitoes, ticks, fleas, and lice. Transmission could occur between wild animals, spreading mirror bacteria through the environment (see Section 8.4), from animals to humans (Rahman *et al.*, 2020), or from humans to humans. Unlike most zoonotic bacterial infections, which are limited to a subset of animal species, mirror bacteria could exhibit an extremely broad host range, with transmission also possible between livestock and pet species.

## 6.2 Many invertebrates would likely be susceptible to mirror bacteria

Invertebrates lack adaptive immunity, but many basic features of innate immunity are shared with vertebrates (Pradeu *et al.*, 2024). Vulnerabilities analogous to those of the vertebrate innate immune system would therefore likely be present in many invertebrates. Since most invertebrate immune systems are poorly characterized compared to vertebrates, we have focused our assessment on the best-studied examples.

## Insect immune systems may fail to recognize mirror bacteria

In Box 6.1, we discuss the immune system of the model organism *D. melanogaster*, the best-studied invertebrate (Buchon *et al.*, 2014). Essentially all antibacterial immunity in *D. melanogaster* is downstream of peptidoglycan recognition. As discussed in Chapter 4, mirror versions of chiral MAMPs such as peptidoglycan would likely not be recognized by vertebrate PRRs. For similar reasons, insect PRRs are also unlikely to recognize mirror MAMPs, rendering an effective immune response unlikely (Figure 6.1).

Mosquitoes and bees also rely on peptidoglycan recognition to trigger antibacterial responses (Evans *et al.*, 2006; Liu *et al.*, 2019; Liu *et al.*, 2020; Rodgers *et al.*, 2020; S. Wang & Beerntsen, 2015; Q. Wang *et al.*, 2019). However, bees and fruit flies are not particularly closely related—both are within the superorder Holometabola, a group that also contains butterflies, ants, and beetles (Misof *et al.*, 2014). It is therefore plausible that reliance on peptidoglycan recognition is conserved across many insects, although too little is known to be certain.

Other aspects of immunity that rely on chiral interactions may be similarly defective. Antimicrobial enzymes are expected to have chiral mechanisms of action and are unlikely to retain potency against mirror bacteria. Lysozymes, for example, cleave peptidoglycan and are important components of early innate immune responses in a range of insects, especially against Gram-positive bacteria (Eleftherianos, Zhang *et al.*, 2021). This enzymatic activity would be unlikely to function against mirror bacteria because the lysozymes would likely fail to bind to mirror peptidoglycan. However, many insect genomes encode multiple lysozymes (Hillyer, 2016; Van Herreweghe & Michiels, 2012), some of which lack muramidase activity and might function via achiral mechanisms, akin to some antimicrobial peptides (AMPs).

## Box 6.1: Mirror bacterial infection would likely be fatal to *D. melanogaster* due to defective peptidoglycan recognition

Antibacterial immunity in *D. melanogaster* is triggered almost exclusively by the presence of peptidoglycan and its breakdown products (Buchon *et al.*, 2014). Peptidoglycan, a chiral polymer present in bacterial cell walls (Pazos & Peters, 2019), is recognized by a protein called Gram-negative binding protein (GNBP) and members of the peptidoglycan recognition protein (PGRP) family (Cherry & Buchon, 2022; Liegeois & Ferrandon, 2022). Recognition activates the Toll and Imd signaling pathways, which regulate the production of antimicrobial peptides that constitute the main defense against systemic infection (Cherry & Buchon, 2022; Liegeois & Ferrandon, 2022). Other immune responses to bacteria, including phagocytosis, melanization<sup>10</sup>, nodulation, autophagy, and immune priming are also downstream of innate immune sensing (Hillyer, 2016) and would be unlikely to function normally against mirror bacteria.

PGRPs would also be unlikely to bind or be activated by mirror peptidoglycan. Structural studies have revealed binding interfaces between PGRPs and peptidoglycan that span chiral centers (Chang *et al.*, 2004; Chang *et al.*, 2006; Leone *et al.*, 2008; Lim *et al.*, 2006; Reiser *et al.*, 2004). Although the structural basis of peptidoglycan binding by GNBP is unknown, it might be driven by similar interactions that would be disrupted upon interaction with mirrored peptidoglycan. GNBP appears to process peptidoglycan to enhance detection via PGRPs and does not trigger the antibacterial pathways by itself (Filipe *et al.*, 2005; Gobert *et al.*, 2003; L. Wang *et al.*, 2006). Therefore, the expected inability of mirror peptidoglycan to detect and respond to a mirror bacterial infection.

Adult flies deficient in PGRPs or GNBP fail to express antimicrobial peptides when challenged with a range of Gram-positive and Gram-negative bacteria and succumb to bacterial infections at much higher rates than wild-type flies (Bischoff *et al.*, 2004; Gobert *et al.*, 2003; Gottar *et al.*, 2002; Iatsenko *et al.*, 2016; Michel *et al.*, 2001; Rämet *et al.*, 2002; Takehana *et al.*, 2004). It is not uncommon for 100% of PGRP or GNBP-deficient flies to die from bacterial infections that most wild-type flies survive, including infections by bacteria such as *E. coli* that are not typically *D. melanogaster* pathogens. Knockout of the downstream Imd signaling pathway also results in susceptibility to lethal *E. coli* infection (Rutschmann *et al.*, 2000). Given that otherwise benign infections can kill flies

<sup>&</sup>lt;sup>10</sup> In mosquitoes and wax moths, melanization reactions of varying intensity can be triggered by the injection of beads composed of dextran or coated with chiral materials, but not glass beads (Barreaux *et al.*, 2017; Schwartz & Koella, 2004; Wiesner, 1992). Although these responses might be mediated by a form of immune recognition, the precise mechanisms have not been identified, and it remains unclear whether mirror bacteria would trigger similar responses. We have been unable to locate similar studies in *D. melanogaster*.

unable to detect peptidoglycan, it seems likely that mirror bacteria could be fatal to adult flies. $^{11}$ 

Unlike other forms of antibacterial immunity in *D. melanogaster*, production of reactive oxygen species (ROS) within the gut is triggered by uracil (Lee *et al.*, 2013; Lee *et al.*, 2015), which is achiral. While uracil is present in all living cells, it is not usually released by bacteria except under conditions of stress (Rinas *et al.*, 1995), and symbionts within the *D. melanogaster* gut do not elicit ROS generation (S.-H. Kim & Lee, 2014). Uracil instead appears to be a specific signature of the catabolism of uridine obtained from the gut by enteric pathogens (E.-K. Kim *et al.*, 2020). Since mirror bacteria would not possess the necessary enzymes to catabolize this exogenous, chiral molecule, they would be unlikely to trigger ROS production in this way.

Insects produce a diverse range of AMPs (Vilcinskas, 2013; Wu *et al.*, 2018). As in vertebrates (see Section 4.2), many of these AMPs function by disrupting bacterial cell membranes through mechanisms that are insensitive to chirality, and should therefore remain harmful to mirror bacteria. The enantiomers of two cecropins and of melittin, for example, are both bactericidal (Bland *et al.*, 2001; Wade *et al.*, 1990), which implies that these AMPs should retain efficacy against mirror bacteria. Other AMPs, however, have chiral targets. Enantiomers of apidaecin, droscin, pyrrhocoricin, and thanatin (found in honeybees, *D. melanogaster*, European firebugs, and the spined soldier bug) appear to be inactive (Bulet *et al.*, 1996; Casteels & Tempst, 1994; Fehlbaum *et al.*, 1996; Kragol *et al.*, 2001), while enantiomers of two peptides produced by green bottle flies show markedly reduced activities against bacteria (Hirsch *et al.*, 2019). AMPs are often released in response to pathogen recognition, so even those that would retain activity against mirror bacteria may not be produced and deployed at high concentrations, given the anticipated defects in immunological detection.

Hemocytes are analogs of the professional phagocytes in vertebrates (Browne *et al.*, 2013) and would be expected to suffer from similar defects to those described in Chapter 4. Hemocytes recognize bacteria directly via transmembrane phagocytic receptors or indirectly via opsonins (Browne *et al.*, 2013; Lin *et al.*, 2020; Melcarne *et al.*, 2019). In *Drosophila*, for example, the phagocytic receptor Draper binds to lipoteichoic acid on *S. aureus* (Hashimoto *et al.*, 2009). Although the molecular ligands for most insect phagocytic receptors are not well characterized (Eleftherianos, Heryanto *et al.*, 2021), many are likely similar to the vertebrate phagocytic receptors discussed in Chapter 4 and probably have chiral ligands. Recognition of mirror bacteria is therefore likely to be impaired. The processes of particle engulfment, phagosome maturation, and bacterial killing are also broadly similar between insects and vertebrates (Eleftherianos, Heryanto *et al.*, 2021); thus, if mirror bacteria were internalized, they would likely resist the killing activity of antibacterial enzymes and

<sup>&</sup>lt;sup>11</sup> There appear to be no studies of immune responses in PGRP knockout *D. melanogaster* larvae. Larval exposure to bacteria may be even higher than adult fruit flies, given that larvae eat dead and decaying matter.

Chapter 6: Animal Infection



## Figure 6.1: Mirror bacteria would likely evade *D. melanogaster* immune mechanisms due to defective peptidoglycan recognition

**A.** Natural-chirality bacteria trigger innate immunity in *D. melanogaster* primarily through the recognition of peptidoglycan and its breakdown products by multiple innate immune receptors. Membrane-bound peptidoglycan recognition protein (PGRP) receptors bind peptidoglycan and signal through the Imd pathway. Soluble PGRP molecules bind to peptidoglycan, which indirectly activates the Toll pathway via membrane-bound Toll receptors. Both pathways induce antimicrobial peptides, phagocytosis, autophagy, and other mechanisms of anti-bacterial immunity. **B.** Mirror bacteria are unlikely to trigger innate immunity in *D. melanogaster*. Because peptidoglycan is a chiral molecule, PGRPs are unlikely to recognize it, leading to absent or severely impaired antibacterial immune responses. Although uracil, which is achiral, can trigger production of reactive oxygen species (ROS) within the gut (A), it appears to be released only by enteric pathogens when they catabolize exogenous uridine, which is chiral. Mirror bacteria would be unlikely to catabolize by natural-chirality uridine and would therefore be unlikely to produce uracil.

antimicrobial peptides with chiral mechanisms. Even if mirror bacteria were killed by pore-forming peptides or oxidative damage within the phagosome, hemocytes may be unable to digest the remains efficiently.

## Mirrored generalist bacteria could plausibly cause lethal infections in insects

Despite the substantial physiological differences between mammals and insects, some bacteria are capable of causing opportunistic infections in both. Both D. melanogaster and the moth larvae Galleria mellonella, for example, are used as infection models for a wide range of human pathogens (Apidianakis & Rahme, 2009; Tsai et al., 2016). In most experiments, high inocula are used to kill the insects because immune defenses are otherwise able to clear the infection. Nevertheless, some strains of *P. aeruginosa* and *Burkholderia* are virulent pathogens in humans as well as *D.* melanogaster and/or G. mellonella (Jander et al., 2000; Mulcahy et al., 2011; Schell et al., 2008). Other species that can cause opportunistic infection in both humans and insects include Bacillus thuringiensis (Ghelardi et al., 2007), Serratia marcescens (Hejazi & Falkiner, 1997; Raymann et al., 2018), Enterococcus faecalis (Cabrera et al., 2023), and the plant-pathogen Agrobacterium tumefaciens (Gottar et al., 2002; Southern, 1996). The ease with which many bacterial pathogens can cause infection in many different hosts, if given the opportunity, suggests that a sufficiently robust mirror bacterium might likewise be capable of infecting a broad range of hosts. While the infection of immunocompromised animals by natural-chirality bacteria is not a perfect analogy for a mirror bacterial infection, the fact that both immunocompromised D. melanogaster (see Box 6.1) and immunocompromised humans (Bartlett et al., 2022) are susceptible to a broad and overlapping set of opportunistic pathogens is concerning.

To illustrate this more concretely, consider E. coli, which is not only a mammalian commensal but also the most commonly studied bacterial species in microbiology and biotechnology. While not traditionally considered an insect pathogen, E. coli can survive and replicate within insect hosts (Ciesielczuk et al., 2015; Kobayashi et al., 1999; Koga et al., 2022; Rochon et al., 2004; Van den Bergh, 2022) and, if injected into adult D. melanogaster, can persist for at least a week without causing obvious harm (Duneau et al., 2017; Troha et al., 2018). Most relevantly, infection by E. coli is typically lethal in immunocompromised D. melanogaster that lack either a functional peptidoglycan-sensing pattern recognition receptor (Rämet et al., 2002) or a functional Imd signaling pathway (Rutschmann et al., 2000). While most experiments deliver bacteria directly into the fly by needle, an experiment examining oral exposure found that knockout of the Imd pathway caused D. melanogaster to be susceptible to oral infection by ROS-resistant E. coli (Ryu et al., 2006), demonstrating survival of E. coli within the gut, translocation into the hemocoel by unknown mechanisms, and subsequent establishment of infection within the hemocoel with ROS as the main potential barrier. Because ROS production within the gut is elicited as a response to uridine catabolism (E.-K. Kim et al., 2020), it is probably not relevant for mirror bacterial infections. Together, these experiments suggest that an immune-evasive generalist like mirror E. coli might be able to cause lethal infection in *D. melanogaster* and, potentially, other insect species.

Insects are constantly exposed to environmental bacteria and often ingest them incidentally while consuming other food. Mirror bacteria would be resistant to lytic enzymes, which would facilitate their survival upon ingestion. Insect guts are often colonized by diverse microorganisms (Engel & Moran, 2013; Shao *et al.*, 2024). Though the molecular mechanisms regulating the insect gut microbiota are highly varied (Schmidt & Engel, 2021), resistance to lytic enzymes and evasion of innate immune detection would be major advantages that might facilitate mirror bacterial

colonization of the gut in some species. Insect guts often have alkaline regions (Terra & Ferreira, 1994), reaching a pH of 9.5 in *D. melanogaster* (Overend *et al.*, 2016) and 12 in some termites (Brune & Kühl, 1996), which could present a substantial barrier to mirrored *E. coli* or other neutrophils. Rapid transit times through the gut may also preclude colonization in some cases (Hammer *et al.*, 2017), especially if mirror bacteria prove to be poor adherents (see Section 4.4). Mirror bacterial colonization of the insect gut could disrupt the natural microbiome and cause harm to the insect, as is known to occur during other forms of dysbiosis (Buchon *et al.*, 2013; Raymann & Moran, 2018).

An even more concerning possibility is that mirror bacteria might cross the gut epithelium and enter the insect hemolymph. This appears to have occurred, for instance, for *E. coli* in *D. melanogaster* lacking a functional Imd pathway (Ryu *et al.*, 2006). It is unclear how frequently natural-chirality commensal bacteria are translocated through the gut epithelium of insects; specialized pathogens such as *S. marcescens* cross the gut epithelium to cause systemic infection (Nehme *et al.*, 2007), but likely have specialized adaptations to do so efficiently. Nevertheless, these barriers are likely imperfect and could allow at least low rates of translocation. Translocation may occur even if mirror bacteria do not successfully colonize the gut, though colonization would greatly increase the rate at which it occurs. Mirror bacteria could also enter insects directly through wounds (Subasi *et al.*, 2024).

Insect hemolymph is also somewhat different in composition to vertebrate blood. The primary sugar is typically trehalose, rather than glucose, though glucose usually remains present at lower concentrations (Simpson, 2003). Amino acid concentrations are generally high, with both alanine and glycine present at millimolar concentrations (Piyankarage *et al.*, 2008; Uchida *et al.*, 1990; Zanotto *et al.*, 1996). As described in Chapter 1, *E. coli* can use both L- and D-alanine as sole carbon and nitrogen sources, and mutants that can utilize most of the enantiomers of the canonical amino acids have been observed; a strain that can utilize glycine, which is achiral, has also been engineered (Fung *et al.*, 2023). It therefore appears plausible that mirrored *E. coli* could replicate within insect hemolymph even without p-glucose catabolism.

As discussed in Box 1.2, D-amino acids can be toxic to natural-chirality bacteria, so L-amino acids could prove similarly toxic to mirror bacteria. Insect hemolymph has higher concentrations of free amino acids than internal compartments in most other animals (e.g., human blood), although these concentrations appear to vary widely both within and between species. However, these concentrations are likely too low to cause growth inhibition of a mirror *E. coli* and other similar mirror bacteria. L-Cysteine may be an exception, as it can inhibit *E. coli* growth at micromolar concentrations in some media (Soutourina *et al.*, 2001); it appears plausible, however, that mirror bacteria could quickly adapt to the concentrations of L-cysteine found in most insects (see Box 1.2). Toxicities to other chiral molecules in hemolymph also appear unlikely to prove a significant barrier to infection, given the lack of evidence that enantiomers of common metabolites are toxic to bacteria and the generally low concentrations of any specific metabolite within hemolymph.

Given the above evidence, it is plausible that a mirror *E. coli* or similar mirror generalist bacteria could replicate within many insects, which would likely be unable to mount an effective immune response. Systemic infections by *E. coli* and other typically non-pathogenic bacteria are often lethal

to immunocompromised *D. melanogaster* (Gottar *et al.*, 2002; Lemaitre *et al.*, 1996; Michel *et al.*, 2001; Rämet *et al.*, 2002; Rutschmann *et al.*, 2000; Schneider *et al.*, 2007), and bacterial loads during infection can reach into the hundreds of millions of cells, or roughly 1% of the total fly mass (Duneau *et al.*, 2017; Lemaitre *et al.*, 1996). Furthermore, bacterial infections in other insects often result in very high pathogen loads (Hu & Webster, 1998; Keshavarz *et al.*, 2023; S. Singh *et al.*, 2014). Therefore, despite a lack of toxins or deleterious immunological responses that hasten host death during natural infections (Shirasu-Hiza & Schneider, 2007), mirror bacteria may replicate to high titers and prove fatal by depleting critical nutrients, producing metabolic waste products, or creating physical obstructions in the hemolymphatic system.

## Insect infection with mirror bacteria could cause ecological damage and increase spread to vertebrates and plants

Insects are among the most abundant terrestrial animals, with a biomass estimated to be 20-fold higher than that of wild vertebrates (Bar-On *et al.*, 2018; Rosenberg *et al.*, 2023). They play diverse ecological roles, including pollination, seed dispersal, and nutrient recycling (Yang & Gratton, 2014; Noriega *et al.*, 2018), and modern declines in insect populations are already an important environmental concern (Wagner, 2020). Even if only a small fraction of insects were susceptible to a mirror bacterial infection, the ensuing ecological damage could be significant (see Chapter 8).

Insects may also spread mirror bacteria to and between other susceptible organisms. Mosquitoes, flies, and fleas, are known to act as vectors for bacterial infection in humans and other vertebrates (Berenger & Parola, 2017; Laroche *et al.*, 2018). Mirror bacteria might enter insects that feed on living vertebrates, cadavers, or fecal matter. The insect could then transmit mirror bacteria to other vertebrates through defecation or regurgitation onto food or into wounds and scratches on an animal's exterior, through consumption of the insect by a predator, or through direct injection while feeding off the vertebrate. All but the last mechanism—which requires invasion of the insect salivary glands (Mueller *et al.*, 2010)—are plausible even if the mirror bacteria were unable to infect the insect itself, though infection could increase the inoculum that is ultimately delivered to vertebrates. Whether infected by or merely carriers of mirror bacteria, insects could contribute substantially to the environmental dispersal of mirror bacteria, which we discuss further in Section 8.4.

Bacterial plant pathogens can also be spread by insects (Eigenbrode *et al.*, 2018; W. Huang *et al.*, 2020), especially hemipteran insects which have piercing mouthparts for extracting sap. Again, this requires invasion of the salivary glands, which would likely require the insect itself to be infected by mirror bacteria. We discuss the potential infection of plants by mirror bacteria in Chapter 7.

Mirror bacteria ingested by *C. elegans* could cause fatal infections if able to persist within the worm intestine

Free-living nematode worms are the most abundant animals in soil ecosystems (van den Hoogen *et al.*, 2019), where they play an essential ecological role as predators of microbes, including bacteria. Here we focus on the model nematode *C. elegans* because it is the most well-studied nematode.

Bacteria are the primary prey of *C. elegans* (Schulenburg & Félix, 2017), and most pathogenic bacteria infect the worm intestine following ingestion (Ermolaeva & Schumacher, 2014).

The first defense of *C. elegans* against pathogenic bacteria is to simply avoid eating them. *C. elegans* recognizes a large number of chemical cues as attractive or repellent, and these are mostly achiral, volatile molecules (Ferkey *et al.*, 2021; Worthy *et al.*, 2018). By contrast, most of the water-soluble attractants are chiral, including the amino acids lysine, histidine, cysteine, and methionine (Ward, 1973) as well as biotin (Bargmann & Horvitz, 1991), which could reduce the attractiveness of mirror bacteria as prey. To a nematode, mirror bacteria might "smell" similar to other bacteria, but they may not "taste" the same. Unless they emit a repellent pathogen-associated molecule such as phenazine-1-carboxamide (Meisel *et al.*, 2014) or dodecanoic acid (A. Tran *et al.*, 2017), mirror bacteria may be recognized through their volatile compounds as potential prey by *C. elegans*.

*C. elegans* feeds by drawing bacteria suspended in liquid into the pharynx and then trapping them while expelling the liquid (Fang-Yen *et al.*, 2009). The nematode has no difficulty ingesting bacterial-sized polystyrene beads, albeit at lower rates than prey (Kiyama *et al.*, 2012). This strongly suggests that filtration is primarily dependent on particle size and that nematodes would also ingest mirror bacteria, though plausibly at a reduced rate due to the absence of chiral signals.

Food particles are first processed mechanically by the pharyngeal grinder (Avery & You, 2012), then pass to the acid intestinal lumen where they are enzymatically digested (Dimov & Maduro, 2019). These mechanisms are not perfect, as many natural bacteria are capable of colonizing or infecting *C. elegans* (Garsin *et al.*, 2001; Merkx-Jacques *et al.*, 2013; F. Zhang *et al.*, 2017). This even includes laboratory *E. coli* MG1655, in which restoration of O antigen biosynthesis appears to provide mechanical resistance to pharyngeal grinding (Browning *et al.*, 2013). A mirror bacterium would be resistant to antibacterial and digestive enzymes, which would greatly reduce the efficacy of normal digestive mechanisms. The lysozyme ILYS-3, for example, appears to be required in the pharynx for the effective disruption of live bacteria (Gravato-Nobre *et al.*, 2016), suggesting that many mirror bacteria would survive the pharyngeal grinder.

While mirror bacterial resistance to digestive enzymes might allow them to survive nematode ingestion, it is not clear that this would suffice to cause infection. Mirror bacteria would likely lack adhesins capable of allowing tight adhesion to the intestinal tract, and, given that passage through the intestines can be very rapid (Ghafouri & McGhee, 2007), mirror bacteria may simply fail to adhere and therefore be expelled by the nematode. If, however, adhesion is not needed for infection, or if non-specific adhesion is sufficient to allow persistence within the gut, then it appears plausible that mirror bacterial infection would result, particularly given that the nematode is so readily colonized and infected by natural bacteria.

Unlike vertebrates or insects, *C. elegans* does not appear to recognize the highly conserved bacterial MAMPs detected by the PRRs of other animals; instead, infections appear to be recognized through a mixture of more specialized pathogen and host damage signals (T. D. Tran & Luallen, 2024). While the signals involved are not well understood, expression patterns for different lysozymes and antimicrobial peptides vary greatly depending on the bacteria that *C. elegans* is exposed to, with much higher expression during exposure to known pathogens as compared to non-pathogens such as

### Chapter 6: Animal Infection

*Bacillus megaterium* or *E. coli* (Dierking *et al.*, 2016), suggesting that the signals are highly specific. Such signals are indeed highly specific in general, and are either molecules associated with specific pathogens—for instance, the toxic metabolite phenazine-1-carboxamide produced by pathogenic strains of *P. aeruginosa* (Peterson *et al.*, 2023)—or the effects of toxins that cause specific forms of cellular inhibition (Dunbar *et al.*, 2012; Liu *et al.*, 2014; McEwan *et al.*, 2012; Melo & Ruvkun, 2012). Because mirror bacteria would, in general, produce neither pathogen-specific signals nor toxins, they are unlikely to trigger any of these innate immune responses.

Infection of *C. elegans* by O antigen restored *E. coli* MG1655, which likely lacks specialized means to cause host damage, may provide a reasonable model for a mirror bacterial infection. In this case, bacterial growth causes gut distension, leading to rupture and nematode death (Browning *et al.*, 2013). Similar overgrowth of bacteria has been observed in response to infection by *E. faecalis* (Garsin *et al.*, 2001), *Salmonella typhimurium* (Aballay *et al.*, 2000), and *S. marcescens* (Mallo *et al.*, 2002), and appears to be the typical end result of unchecked bacterial replication in the absence of other virulence factors.

Intestinal bloating and associated tissue damage eventually trigger an innate immune response (Radeke & Herman, 2021; J. Singh & Aballay, 2019), though only at a late stage of infection. Notably, the immune response does not appear sufficient to prevent nematode death during infection with O antigen restored *E. coli* MG1655, or in similar bacterial infections. A mirror bacterium would be additionally resistant to many of the downstream antibacterial effectors, including lysozymes (Dierking *et al.*, 2016) and C-type lectins (Pees *et al.*, 2021). Membrane-disrupting antimicrobial peptides such as caenopores (Roeder *et al.*, 2010) may still function—the D-enantiomers of similar peptides in other animals often retain efficacy against bacteria (Wei *et al.*, 2009)—but overall, it seems unlikely that *C. elegans* could successfully control a mirror bacterial infection.

*C. elegans* can learn to avoid both pathogens (J. Singh & Aballay, 2019; Y. Zhang *et al.*, 2005) and low-quality food (Shtonda & Avery, 2006). A nematode that encounters mirror bacteria and survives—either because the mirror bacteria cannot successfully colonize the gut or because the immune system successfully cleared the infection—could plausibly learn to avoid consuming mirror bacteria. Avoidance behavior is unlikely to be perfect, but it could greatly reduce ingestion of mirror bacteria.

In summary, *C. elegans* exposed to mirror bacteria could ingest them at similar or somewhat lower rates compared to natural bacteria. Mirror bacteria could be resistant to normal digestive processes and may simply pass through unharmed. If mirror bacteria are able to adhere to or otherwise persist within the intestines, they could probably cause a lethal infection. Avoidance behavior may be somewhat protective against infection if the infectious dose is high and *C. elegans* can learn that mirror bacteria are poor food items before infection occurs.

#### Substantial uncertainty remains regarding the immune responses of other invertebrates

Beyond insects and nematodes, relatively little is understood about the immune systems of most invertebrates. In many invertebrate species, innate immune receptors have expanded dramatically (Figure 6.2). Unfortunately, nearly all invertebrate innate immune receptors remain uncharacterized





#### Figure 6.2: The number of PRR genes varies widely across animal species

Waffle plot displaying the number of total known genes for Toll-like receptors (TLRs), nucleotide-binding domain and leucine-rich repeat-containing proteins (NLR), and scavenger receptor cysteine-rich receptors (SRCRs across select animal species. Each square represents one gene. The total number of genes is displayed to the right of each plot. Some PRR gene families have expanded dramatically in certain species. Although nearly all characterized PRRs recognize chiral ligands, most of the proteins encoded by these expanded gene families remain uncharacterized. TLR gene numbers are from (Saco *et al.*, 2023). SCSR gene numbers are from (Peng *et al.*, 2024). NLR gene numbers are from (Buckley & Rast, 2015), except for *Mus musculus* (Sundaram *et al.*, 2024), *Danio rerio* (Howe *et al.*, 2016), *Mytilus edulis* (Zhu *et al.*, 2022), and *Crassostrea gigas* (L. Zhang *et al.*, 2015).

at a molecular level. It is plausible that some invertebrates might have innate immune receptors that can recognize achiral components of pathogens. Moreover, the sheer number of immune receptors in some invertebrate species could increase the likelihood that one or more receptors would cross-react with mirror molecules, thus enabling those invertebrates to recognize mirror bacteria to some degree. Damage detection systems such as those found in *C. elegans* may also offer some protection. However, it seems likely that many invertebrates would have difficulty recognizing and responding to infecting mirror bacteria.
## **Chapter 7: Plant Infection**

Damon J. Binder<sup>1</sup>, Kevin M. Esvelt<sup>2</sup>, Jonathan D. G. Jones<sup>3</sup>, Nicholas J. Talbot<sup>3</sup>

Plants evolved sophisticated immune systems to defend themselves from bacterial pathogens while tolerating low concentrations of bacterial endophytes (Jones & Dangl, 2006; Jones *et al.*, 2024). Like animals, plants use a variety of pattern recognition receptors (PRRs) to detect the presence of foreign microbes and release effectors to keep them at low concentrations in the apoplast, the continuous extracellular space through which water and other molecules can move without crossing cell membranes. These receptors are unlikely to recognize mirror biomolecules, and so plants may be susceptible to mirror bacterial infection.

Since plants are poorly studied relative to mammals, and most experiments have focused on the model plant *Arabidopsis thaliana* or a handful of crops, our uncertainty regarding the potential consequences of mirror bacterial infection is considerably higher. Much remains unknown concerning the interactions between plant immunity and bacterial commensals, opportunists, and pathogens. Nevertheless, we have identified two pivotal questions, the answers to which may differ across plant species. First, could mirror bacteria capable of colonizing the apoplast spread beyond their initial site of entry, and second, could infecting mirror bacteria enter the phloem?

Section 7.1 discusses plant immune systems, and shows that a mirror bacterium would likely evade recognition as well as some of the downstream mechanisms used by plants to defend themselves from bacteria.

Section 7.2 discusses localized infections of leaves and roots. In both cases, mirror bacteria could plausibly enter and colonize the apoplast, establishing a local infection. It is unclear if a local infection would be particularly harmful to plants, though it is possible that widespread localized infections could cumulatively cause harm.

Section 7.3 analyzes the potential for infections to spread through vasculature. Mirror bacteria would, by default, lack the extracellular enzymes needed to break down plant physical barriers, which may limit their ability to spread through and colonize xylem. Bacterial entry into the phloem is not common, and primarily occurs via phloem-feeding insects. It is unclear if mirror bacteria could otherwise enter phloem during localized infection. Should it occur, mirror bacterial infection of phloem would likely prove highly damaging or lethal.

Section 7.4 describes countermeasures to protect agriculture. Conventional methods used to control bacterial pathogens would be of limited use against mirror bacteria. Plants could potentially be protected by engineering their existing immune receptors to detect mirror macromolecules, which

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup> Open Philanthropy, San Francisco, California, USA; <sup>2</sup> Media Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, USA; <sup>3</sup> The Sainsbury Laboratory, University of East Anglia, Norwich, UK

may offer scalable protection for key crop species. Protecting most crop varieties, or even a small fraction of wild plants, is unlikely to be feasible.

This chapter primarily focuses on generalist mirror bacteria such as mirror *Escherichia coli*, which would likely be of greatest scientific interest. It is plausible that deliberate engineering could generate mirror bacteria that were better equipped to colonize plants, or to cause harm to infected plants. Such details are beyond the scope of the chapter.

## 7.1 Mirror bacteria are likely to evade plant innate immunity

Unlike many animals, plants do not contain circulating professional phagocytes or other immune cells, and they also lack adaptive immunity. Instead, most plant cells rely on innate immune responses that are initiated upon the detection of pathogen molecules by cell surface or intracellular immune receptors. Pattern-triggered immunity (PTI) detects conserved microbial patterns such as bacterial flagellin, and induces a localized immune response featuring release of antimicrobial compounds and proteins, as well as signaling molecules that modify local plant physiology in order to combat and contain the infection (Jones & Dangl, 2006). As with animal immune systems (see Chapter 4 for a discussion on human immunity, and Chapter 6 for other vertebrate and invertebrate animals), cell surface PRRs in plants are expected to poorly recognize enantiomers of their usual targets, precluding direct pathogen targeting. Many downstream defense mechanisms, though not all, would be similarly impaired (Figure 7.1).

## Plant immune systems are unlikely to detect mirror bacteria

Plants recognize pathogens through two main pathways: PRRs that detect generic biomolecules found in microbes, and intracellular immune receptors that detect particular pathogen virulence factors ("effectors") tailored to suppress plant defenses or promote pathogen nutrition (Jones & Dangl, 2006; Katagiri & Tsuda, 2010; Ngou *et al.*, 2022). Effectors are protein virulence factors delivered into plant cells by specialized secretions systems, most commonly, the type III secretion system. Mirror bacteria are unlikely to have been created with specialized type III or other secretion systems to deliver intracellular effectors, and so intracellular receptors are unlikely to contribute useful resistance to mirror bacteria (Figure 7.1).

Plant PRRs detect conserved microbe- or pathogen-associated molecular patterns (MAMPs or PAMPs) (Chaudhry *et al.*, 2021; Dardick & Ronald, 2006). As in animals, these MAMPs are generally chiral (Boutrot & Zipfel, 2017; Schellenberger *et al.*, 2019); the only known achiral MAMP, the bacterial lipid cis-11-methyl-2-dodecenoic acid, is a quorum sensing molecule employed by a limited number of bacterial species (Kakkar *et al.*, 2015). Although novel receptors continue to be discovered in plants, and some degree of cross-reactivity between a mirror molecule and a normal PRR may be possible, it is likely that most plants would exhibit reduced or even entirely absent initial responses to mirror bacterial infections.

Poor MAMP detection reduces or abolishes downstream antibacterial responses, such as the closure of stomata to prevent pathogen entry (Zipfel *et al.*, 2004), the release of defense hormones, and the secretion of antimicrobials and reactive oxygen species (ROS) to kill pathogens. Deleting a single



#### Figure 7.1: Mirror bacteria are likely to evade detection by plant immunity

**A.** Natural-chirality extracellular<sup>12</sup> bacteria are recognized by two main pathways of plant immunity. Pattern-triggered immunity (PTI) relies on pattern recognition receptors (PRRs) to detect microbial-associated molecular patterns (MAMPs) and induce the release of antimicrobial compounds, reactive oxygen species (ROS), and signaling molecules that modify local plant physiology to combat and contain infection. Effector-triggered immunity (ETI) relies on intracellular immune receptors, such as nucleotide-binding domain and leucine-rich repeat containing receptors (NLRs), to detect bacterial virulence factors ("effectors") that suppress plant immune responses. Effectors are secreted inside of plant cells by specialist Gram-negative plant pathogens via type III secretion systems (T3SS). **B.** Mirror bacteria are unlikely to trigger PTI or ETI. Because nearly all MAMPs are chiral, mirror MAMPs are unlikely to be recognized by plant PRRs. ETI is unlikely to be relevant because mirror bacteria are unlikely to have T3SS or stereospecific effectors capable of interacting with natural-chirality plant proteins.

PRR can result in increased susceptibility to infection by natural phytopathogenic bacteria in experimental models (Ranf *et al.*, 2015; Willmann *et al.*, 2011), suggesting that impaired recognition by multiple receptors would render plants still more susceptible.

As in animals, it is possible that the later stages of mirror bacterial infection may damage the plant, leading to production of damage-associated molecular patterns (DAMPs). Some DAMPs, such as oligogalacturonides, trigger a significant plant immune response even in the absence of MAMPs (Ferrari *et al.*, 2013). However, unlike natural-chirality pathogens, mirror bacteria would not secrete polygalacturonases<sup>13</sup> that generate oligogalacturonides from the plant cell wall, so the pathway is unlikely to be triggered. It is possible that other DAMPs might activate innate immune responses in the absence of MAMPs once sufficient plant colonization has occurred, but how MAMP and DAMP signals are integrated to determine the plant immune response is not fully understood (Saijo & Loo,

<sup>&</sup>lt;sup>12</sup> Plant immune defenses of the intracellular phloem, discussed in Section 7.3, are poorly understood but may involve some analog of PTI (Jiang *et al.*, 2019).

<sup>&</sup>lt;sup>13</sup> Even if engineered to produce polygalacturonases, the mirror polygalacturonases produced by mirror bacteria are unlikely to target the (chiral) glycosidic bonds within plant cell walls.

2020). Moreover, it is unclear precisely how mirror bacterial infection would damage the plant beyond directly or indirectly depleting it of nutrients, which may not trigger the DAMP response until relatively late in the course of infection, if at all.

Effector-triggered immunity (ETI), typically encoded by disease resistance genes (R genes), recognizes specific pathogen effectors that are delivered into the plant to interfere with MAMP detection (Cui *et al.*, 2015; Pfeilmeier *et al.*, 2016). ETI powerfully restricts pathogen success: a pathogen that is normally capable of establishing a virulent infection loses that capability if it expresses a single effector recognized by an intracellular immune receptor. Mirror bacteria would not produce such virulence effectors by default and so would go unrecognized by effector-triggered immunity.

## Most antimicrobial secondary metabolites and peptides are unlikely to be effective against mirror bacteria

Plants synthesize and accumulate an arsenal of chemical defenses to protect against microbial infection (Tiku, 2020), including small molecules, antimicrobial peptides, and proteases. To be effective against mirror bacteria, these chemicals must be chirality-independent and either reliably released in the presence of the invader or constitutively present in a biologically active form. If immune recognition of mirror bacteria is substantially impaired, the only effective chemical defenses in plants would likely be those that are both active against mirror bacteria and constitutively present.

Such compounds appear to be rare. Antimicrobial proteases (Godson & van der Hoorn, 2021), as chiral enzymes that recognize chiral substrates, would likely be ineffective (see Section 4.2). Most small-molecule toxins generated by plants are phytoalexins, which are synthesized only after pathogens are detected. Phytoanticipins are produced constitutively, but they either exist in an inactive state and are activated by an immune signal or physical damage, or are stored intracellularly in an active form and released upon pathogen detection (Kliebenstein & Kvitko, 2023; Osbourn, 1996). If a mirror bacterial infection becomes sufficiently severe, damage to plant cells may result in the release of these compounds, but they are unlikely to prevent an initial infection.

Most secondary metabolites are also chiral, though there are exceptions, including allicin, catechol, and curcumin (Kocaçalişkan *et al.*, 2006; Marchese *et al.*, 2016; Zheng *et al.*, 2020). However, all of these are phytoalexins, so they would only be helpful if their release was triggered through DAMPs or other means. Cyanogenic glycosides and glucosinolates are activated upon cell damage (Gleadow & Møller, 2014) and the resulting isothiocyanates or hydrogen cyanide would likely act on mirror bacteria, but only if the plant was damaged. Some plants produce achiral metabolites with antibacterial activity that are known to be constitutively present in a biologically active form, including *p*-coumaric acid (González Moreno *et al.*, 2022; Lou *et al.*, 2012), cinnamic acid (Dorantes *et al.*, 2000), and quercetin (Wang *et al.*, 2018). Because they are concentrated at plant barriers, they are probably not sufficient to prevent survival of a sufficiently hardy mirror bacterium once it has already entered the plant, but they may help prevent an initial infection.

Plants also produce a diverse range of antimicrobial peptides (Nawrot *et al.*, 2014), which often have chiral mechanisms of action (Campos *et al.*, 2018) and so would typically not be effective against

## Chapter 7: Plant Infection

mirror bacteria. Thionins, which lyse cellular membranes and exhibit broad antimicrobial activity, provide a plausible exception (Tam *et al.*, 2015), although their effectiveness may depend upon potentially chiral interactions with phospholipids (Stec *et al.*, 2004). Some antimicrobial peptides are constitutively released by plant cells, but others are either stored intracellularly or synthesized only once a pathogen is recognized (Nawrot *et al.*, 2014). Overall, antimicrobial peptides appear analogous to secondary metabolites: most would be ineffective or inactive in the absence of an immune response, but plant immunity is poorly characterized, and there may be exceptions capable of protecting some species (Jones *et al.*, 2024).

## If triggered, ROS and localized desiccation would retain efficacy against mirror bacteria

While most chemical and enzymatic antibacterials are not expected to be effective against mirror bacteria, other defense mechanisms could still function if triggered. ROS are generally achiral and cause damage through a chirality-independent mechanism that would harm mirror bacteria. However, ROS are also damaging to the plant, so their concentrations within the apoplast are generally tightly controlled (Podgórska *et al.*, 2017). Following successful pathogen recognition (Wojtaszek, 1997) or other stressors (Sharma *et al.*, 2012), a plant cell may undergo an oxidative burst—a rapid, transient release of large quantities of ROS. Given the anticipated failure of pattern-triggered immunity to recognize mirror MAMPs, this reaction is unlikely to occur. Moreover, many generalist bacteria—like most plant pathogens—express catalase, providing them with some degree of resistance (Fones & Preston, 2012).

Plants also suppress bacterial growth by restricting water availability in the apoplast, much of which is dry and relatively inhospitable to endophytes (Beattie, 2011; Roussin-Léveillée *et al.*, 2024). The exact mechanisms are not fully understood, but appear to involve both pattern-triggered and effector-triggered immunity (Z. Liu *et al.*, 2022; Wright & Beattie, 2004), and may also be associated with plant cell death (Freeman & Beattie, 2009). Drying of the apoplast could prove effective at halting mirror bacterial growth, particularly since they are unlikely to have water-soaking effectors with which to release water from plant cells. However, local desiccation would only be effective if induced, which appears unlikely.

Finally, plants can reduce concentrations of sugars and other key nutrients in response to bacterial infection (Lemonnier *et al.*, 2014; Yamada *et al.*, 2016). This could again suppress mirror bacterial growth, but would also likely first require immune activation.

# 7.2 Mirror bacteria could plausibly establish chronic local infections within leaves and roots

In addition to molecular and chemical immune mechanisms, plants rely heavily on physical barriers to defend their internal spaces from microbial invasion. The external surfaces of plants are not very hospitable to bacteria and are mechanically tough. Only fungal pathogens have evolved mechanisms to traverse these barriers, and bacteria instead usually enter through natural openings, such as lateral root emergence sites, stomata on leaves, hydathodes at leaf margins, or through wounds. Despite these barriers, both specialized plant pathogens and non-specialist commensal bacteria are able to

enter the leaf and root apoplast, and at least occasional mirror bacterial entry into both spaces appears plausible. Given the anticipated failure of plant immunological defenses, it also appears plausible that mirror bacteria could colonize these apoplastic spaces, but the consequences of localized infection are difficult to predict.

This section focuses on whether plant exposure to environmental mirror bacteria could lead to infection. The potential survival and spread of mirror bacteria through the environment, which could result in widespread exposure of plants to mirror bacteria, is addressed more generally in Chapter 8. Different kinds of mirror bacteria could greatly vary in their ability to infect plants. As discussed below, it appears possible that immune evasion could allow a mirror *E. coli* or similar generalist bacteria to colonize many plants, particularly if equipped with the ability to catabolize common sugars. Ecologically invasive mirror bacteria would face strong selective pressures to increase robustness and adaptations could arise quickly (see Section 8.3), increasing both the probability of plant infection and an expansion in host range over time.

## Deficits in stomatal closure could allow mirror bacteria to enter leaves

In order to enter a leaf, mirror bacteria first need to attach to and survive on its surface. Leaf surfaces are harsh environments for bacteria, often hydrophobic with fluctuating temperatures and humidity, low nutrients, and exposure to ultraviolet radiation. Despite this, enteric mammalian pathogens such as *Salmonella* and *E. coli* O157:H7 are able to survive on leaf surfaces for long periods of time (Brandl, 2006; O'Brien & Lindow, 1989), causing outbreaks of food-borne illness when contaminating leafy vegetables or other produce (Heaton & Jones, 2008). A mirror bacterium would not necessarily be as hardy, and it would be unlikely to have specialized adhesins with which to attach to leaf surfaces—though this could potentially be (at least partially) remedied by curli or other non-specific adhesins (Jeter & Matthysse, 2005). Nonetheless, prolonged survival of generalist mirror bacteria on leaf surfaces remains plausible for sufficiently hardy strains.

Bacteria present on the surface of a leaf can enter the apoplast through stomata, microscopic pores on the surfaces of leaves that open and close to regulate gas exchange. To keep pathogens from entering, plants rapidly close their stomata when nearby PRRs are activated, a strategy known as "stomatal defense" (Melotto *et al.*, 2017). The importance of stomatal defense in preventing disease has been underscored by the vulnerability of model systems with even narrow defects in stomatal defense (Melotto *et al.*, 2006; Zeng & He, 2010; Zipfel *et al.*, 2004), and by the ubiquity of virulence factors from bacterial pathogens that specifically inhibit the pathways leading to stomatal closure (Kroupitski *et al.*, 2009; Melotto *et al.*, 2006).

All MAMPs currently known to trigger stomatal closure appear to be chiral (Melotto *et al.*, 2017), though many likely remain undiscovered. If most MAMPs that trigger stomatal closure do prove to be chiral, stomata may not reliably close in response to the presence of mirror bacteria, allowing invasion of the leaf interior (Figure 7.2). A chemotactic mirror bacterium may actively approach and enter open stomata due to the relatively greater concentration of nutrients inside the leaf (Matilla &

## Chapter 7: Plant Infection

Krell, 2018); a mirror *E. coli*, for instance, would be attracted by both achiral nutrients and by some natural-chirality hexoses and amino acids (Section 8.2).

Hydathodes, pore-like structures on leaf margins that secrete water and dissolved substances through guttation, provide another potential entry route for bacteria (Cerutti *et al.*, 2019). Like stomata, they are defended by PRRs (Paauw *et al.*, 2023), which are not expected to recognize mirror bacteria. Finally, mechanical damage—including wounds delivered by herbivores—may provide a third means of entry into the leaf apoplast (Savatin *et al.*, 2014).

## Generalist mirror bacteria could plausibly grow within the leaf apoplast

Abiotic conditions within the leaf apoplast are generally hospitable to bacterial growth, as reflected by the existence of communities of endophytic commensal bacteria in plant leaves (Chaudhry *et al.*, 2021; Wippel, 2023). Temperatures are similar to that of the external environment, and apoplastic fluids are typically mildly acidic (Sharp & Davies, 2009; Tarvainen *et al.*, 2023). Osmotic pressure is also unlikely to pose a barrier, given that the water potentials within leaves are typically above



### Figure 7.2: Mirror bacteria might enter leaves through open stomata and grow within the leaf apoplast

A single leaf is shown in cross-section. The cuticle and upper epidermis form a physical barrier that would likely prevent mirror bacteria from entering the leaf. The lower epidermis forms a similar barrier, but it is punctuated by microscopic pores called stomata that open and close to regulate gas exchange. Stomata typically close when pathogens are detected through PRRs, but would likely remain open because PRRs will not detect mirror bacteria. As a result, mirror bacteria could move into the apoplast, the continuous extracellular space in the leaf interior. Much of the apoplast is air, but the apoplastic fluid associated with the plant cell walls contains water and some nutrients and might be capable of supporting mirror bacterial growth. Although water and nutrients flow in and out of the leaf via xylem and phloem, it is unclear whether mirror bacteria in the apoplast would be able to pass through the physical barrier formed by bundle sheath cells and enter the vascular system.

-5 MPa (Bartlett *et al.*, 2012), equivalent to a water activity of 0.96. Roots and xylem must have even higher water activities in order for water to flow from the roots to the leaves, and activities for fruits and vegetables are also almost always above 0.95 (Schmidt & Fontana, 2008).

While much of the apoplast is filled with air, it contains liquid capable of supporting bacterial growth. Leaf apoplastic fluid contains all of the inorganic nutrients required for bacterial reproduction, as well as amino acids, sugars, and other organic compounds (Gabriel & Kesselmeier, 1999; Lohaus et al., 2001; López-Millán et al., 2000; O'Leary et al., 2016; Rico & Preston, 2008). Concentrations of individual metabolites and minerals within apoplastic fluids are likely highly variable between species. To our knowledge, there has been no systematic attempt to estimate metabolite concentrations across different types of plants. Nevertheless, available measurements suggest that the achiral compounds acetate,  $\gamma$ -aminobutyric acid, citrate, glycine, pyruvate, and succinate are often present at concentrations above 100 µM (Gabriel & Kesselmeier, 1999; O'Leary et al., 2016). All of these can be utilized by E. coli; acetate, pyruvate, and succinate can serve as sole carbon sources in E. coli K12 (see Table 1.1). Concentrations of common L-amino acids may also be around 100 µM (O'Leary et al., 2016; Rico & Preston, 2008) and many could provide nutrients for mirror bacteria; L-alanine in particular could be a sole carbon and nitrogen source for a mirror E. coli (see Chapter 1). L-malate, which could also be catabolized by a mirror E. coli, is often present at around 1 mM concentrations (López-Millán et al., 2000; O'Leary et al., 2016). In aggregate, it appears plausible that these nutrients could be sufficiently abundant to permit growth of a mirror E. coli.

As with other metabolites, apoplastic concentrations of sugars have not been systematically measured, and existing data is highly variable for individual sugars. Sucrose concentrations are typically between 0.1–4 mM (Aked & Hall, 1993; López-Millán *et al.*, 2000; Nadwodnik & Lohaus, 2008; O'Leary *et al.*, 2016; Oner-Sieben & Lohaus, 2014; Tetlow & Farrar, 1993), while concentrations of sorbitol or mannitol can be in the tens of millimolar (Nadwodnik & Lohaus, 2008). Concentrations of glucose and fructose were measured to be below 50 µM in *Arabidopsis* (O'Leary *et al.*, 2016), but 16 mM and 4.5 mM respectively in sugar beet (López-Millán *et al.*, 2000). In some circumstances host cell wall invertases could increase hexose concentration in the apoplast, such as in immature leaves (Ruan, 2014) or in response to pathogen infection (Biemelt & Sonnewald, 2006).

These sugars are chiral and so would not be accessible to an exact mirror-image of an *E. coli*. However, as discussed in Chapters 1 and 3, systems capable of L-glucose utilization do exist in a few natural bacteria and if implemented within mirror bacteria would allow D-glucose catabolism. Isomerases interconverting L-fructose and L-glucose are already known (Park *et al.*, 2007), as is a dehydrogenase capable of converting L-mannitol to L-fructose (Pail *et al.*, 2004), while L-sorbitol could likely be catabolized by incorporating a known pathway from *Stenotrophomonas maltophilia* (Brechtel *et al.*, 2002). There are no known enzymes capable of hydrolyzing L-sucrose, although it is possible that a suitable glycoside hydrolase could be discovered, evolved, or designed. Engineering mirror bacteria to consume common chiral sugars—which could be motivated by a number of scientific or practical considerations—might therefore increase the risk of apoplastic colonization.

## Chapter 7: Plant Infection

## The consequences of generalist mirror bacterial infection of leaves are unclear

Because mirror bacteria are expected to resist constitutively released plant antibacterials and avoid activating the PTI or ETI responses, their mortality rate within leaves could plausibly be low. Nutrients within apoplastic fluid could support growth, even for mirror bacteria that are unable to catabolize sucrose or other common sugars. Insofar as this growth rate is more rapid than the death rate, which is expected to be low due to impaired immune recognition, the mirror bacterial population would expand.

It is difficult to predict the possible consequences of a localized mirror bacterial infection as there are no obvious natural analogs. Leaves typically contain commensal bacteria, but these endophytes usually remain at low concentrations of  $10^3$  CFU/g (Hallmann, 2001). Given the availability of nutrients with apoplastic fluid, it appears likely that endophyte populations are directly or indirectly kept at low levels by the host immune response (Hacquard *et al.*, 2017). For instance, *Arabidopsis* plants in which multiple immunological pathways have been knocked out can host endophytic populations more than a hundred-fold greater than those in healthy plants (Chen *et al.*, 2020), and exhibit a density-dependent immunological response to inoculations with non-pathogenic bacteria (Miebach *et al.*, 2024).

By evading these responses, local mirror bacterial concentrations could potentially become multiple orders of magnitude greater than those of typical commensals. Indeed, pathogenic bacteria that release effectors capable of suppressing the immune response can likewise reach high concentrations in leaves (Hirano & Upper, 2000). However, such bacteria generally cause direct harm to the leaf through both intracellular effectors injected into plant cells and/or through extracellular enzymes that damage plant tissue, often in turn providing the bacteria with additional nutrients. They also trigger strong immune reactions from the plant itself, including localized cell death through the hypersensitive response (Balint-Kurti, 2019). Mirror bacteria could not cause harm through any of these mechanisms, and therefore the consequences of infection cannot be predicted on the basis of those caused by natural bacterial pathogens.

Instead, it appears most likely that mirror bacterial populations would expand until growth became limited by nutrients. Given the low concentrations of accessible organic compounds compared to the relatively high concentrations of inorganic nutrients such as iron (López-Millán *et al.*, 2000; O'Leary *et al.*, 2016) and phosphorus (Gabriel & Kesselmeier, 1999; Lohaus *et al.*, 2001; Mimura *et al.*, 1992; O'Leary *et al.*, 2016), organic carbon would likely prove limiting, causing accessible sugars and organic acids to become depleted in the apoplast. The extent to which this would be harmful to the leaf is unclear. While the leaf apoplast does contain some sugars and other organic nutrients, these comprise only a very small fraction of the total carbon metabolite pools within a leaf, and organic nutrients typically move between cells through symplasmic rather than apoplasmic pathways.

The primary exception is phloem-loading. In many plants sucrose is exported apoplastically within the minor leaf veins, then imported into neighboring phloem sieve elements and companion cells (Braun, 2022). The minor leaf veins are likely inaccessible to mirror bacteria, so even in apoplastic phloem loaders the consumption of photosynthetic production would be limited. There are a few known exceptions: in many deciduous trees, especially sugar maples, but also in plants such as grape

(Walker *et al.*, 2021), organic sugars are transported from starch reserves in the roots through xylem to support leaf growth in the spring. As a result, apoplastic sugars can reach concentrations as high as 30–150 mM sucrose (Holgate, 1950; Rapp *et al.*, 2019). If metabolically accessible to mirror bacteria, these levels could allow local sugar-limited populations to expand rapidly, potentially consuming enough of either organic or inorganic nutrients to trigger systemic changes in nutrient allocation that may stunt the growth of the plant. However, even this scenario would likely require large fractions of most leaves to be colonized.

Apoplastic sucrose and malate both play important roles in the metabolism and regulation of guard cells (Kang *et al.*, 2007; Lawson & Matthews, 2020; Santelia & Lawson, 2016), which unlike other plant cells lack plasmodesmata and so must import organic compounds from the apoplast. Organic compounds in the apoplast also provide nutrients to endophytes, and help regulate apoplastic pH and osmotic pressure. A localized mirror bacterial infection might therefore disrupt leaf homeostasis and reduce productivity, but this is highly speculative, and plants might substantially vary in their susceptibility to harm.

A key overarching question is the extent to which localized mirror bacterial infection could harm the plant as a whole. It is difficult to see how chronic local infections in a limited number of leaves would induce highly deleterious consequences for a plant. Plants routinely lose individual leaves to herbivores or other injuries; a localized mirror bacterial infection might not be very disruptive to an individual leaf, or simply result in the leaf being shed. However, the simultaneous and chronic infection of many leaves could cumulatively stunt plant growth by reducing the flow of sugars to roots and shoots, especially in species with seasonally higher concentrations of organic nutrients in the apoplast. Cumulative local leaf stresses induced by mirror bacterial infection could theoretically cause maladaptive plant behaviors by inducing global stress responses and systemic nutrient redirection, perhaps analogous to the formation of leaf galls that redirect resources by bacterial pathogens such as *Rhodococcus fascians* (Stes *et al.*, 2013), but this is highly speculative given that mirror bacteria could not similarly reprogram the plant cells. Further research on whether leaf-specific apoplastic disruptions can trigger systemic responses could help clarify the extent to which localized mirror bacterial infections might be harmful to the plant as a whole. Persistent localized plant infections, even if initially benign, would furthermore provide many opportunities for the evolution of more virulent mirror bacterial strains.

## Mirror generalist bacteria within soils might infect plant roots

As discussed in Chapter 8, mirror bacteria would resist most forms of predation and antibacterial weapons, greatly facilitating survival in, and colonization of, soils. The rhizosphere, the region surrounding the roots of a plant, features a distinctive microbiota that is strongly influenced by plant secretions (Baker *et al.*, 2024). Root exudates provide bacteria with sugars; predominantly glucose and fructose, organic acids, including citric, succinic, and malic acids (Kamilova et al., 2006); and antibiotic compounds to shape the community composition (Bais *et al.*, 2006). Rhizospheric bacteria also produce antibiotic compounds and lytic enzymes to harm competitors and can suppress plant infections by limiting the growth of pathogens (Beneduzi *et al.*, 2012; Doornbos *et al.*, 2012). Mirror bacterial resistance to most antibiotics and all lytic enzymes, along with resistance to most predatory

organisms (Section 8.1), could aid colonization of the rhizosphere despite competition for nutrients from the natural microbiota.

Plants host diverse endophytic communities that largely derive from the rhizosphere, some of which enter the plant (Figure 7.3). Passive penetration can occur through cracks at lateral root emergence sites, at root tips, or through damage created by other microorganisms (Hardoim *et al.*, 2008; Reinhold-Hurek & Hurek, 1998). Emergence sites are defended by root border cells, which produce a mucilage rich in antimicrobial proteins and phytoalexins to prevent bacterial entry (Driouich *et al.*, 2013). However, root border cell release and mucilage secretion are both inducible upon pathogen detection, and consequently are likely to prove less effective at blocking mirror bacteria.

Further evidence that bacteria can enter roots comes from studies of *E. coli* and *Salmonella*, which find they can be taken up from soils and hydroponic solutions into plant roots (Alegbeleye *et al.*, 2018; Hirneisen *et al.*, 2012; Perulli *et al.*, 2021). Studies of microplastic beads similarly suggest that, at least within wheat and lettuce, bacteria-sized beads can be internalized through roots (Li, Luo, Li *et al.*, 2020; Li, Luo, Peijnenburg *et al.*, 2020; Solomon & Matthews, 2005). While these studies generally focus on small, herbaceous plants under conditions that are not always reflective of those in the natural environment or in agricultural settings, they nevertheless suggest that if mirror bacteria were present in soil, they would likely be internalized into the roots of many plants at least occasionally, and perhaps frequently.



### Figure 7.3: The anatomy of a root could facilitate mirror bacterial colonization

An idealized root is shown in longitudinal cross-section (left) and horizontal cross-section (right). The root epidermis covers the outer surface of the root. Cracks in the epidermis at lateral root emergence sites, the root tip, or sites of damage could permit mirror bacteria in the rhizosphere (the microbe-rich soil zone near root systems) to enter the cortex. The cortex is composed of loosely packed cells with large intercellular spaces and apoplastic fluid and could plausibly be colonized by mirror bacteria. In order to reach the xylem and phloem, water and nutrients from the cortex must pass through the interiors of endodermal cells, a barrier that would likely prevent most mirror bacteria in the cortex from reaching the vascular tissue.

Endophytic and pathogenic bacteria are further able to penetrate the root endodermis and pericycle to reach the root xylem, but likely make use of cell wall degrading enzymes to do so (Compant *et al.*, 2010). Given that mirror bacteria would lack such enzymes, the extent to which they could further penetrate roots is less clear, although reports of plastic microbead and human pathogen transport again suggests that it may be possible in some plants (see Section 7.3).

Plant roots are typically home to a diverse array of endophytic bacteria (H. Liu *et al.*, 2017) and are generally hospitable to bacteria. There have been few studies of root apoplastic fluid (Dragišić Maksimović *et al.*, 2014), but like root exudate, xylem sap (discussed in the next section), and leaf apoplastic fluid, it contains variable concentrations of organic acids, simple sugars, and amino acids that could support mirror bacterial growth in many cases. Given that mirror bacteria suffer from reduced mortality from plant immunological responses as well as from antibacterials originating from root border cells and competing bacteria, it is plausible that mirror bacteria could colonize root apoplasts.

As with a localized leaf infection, it is unclear whether colonization of the rhizosphere or localized infections of the root cortex would prove harmful to plants. The consumption of organic nutrients could modify local redox potentials and pH. Metabolites released by plants play an important role in shaping endophytic and rhizospheric microbial communities (Badri & Vivanco, 2009; Shi *et al.*, 2011), which in turn can facilitate plant nutrient uptake (Singh *et al.*, 2022) and help protect against pathogens (Raaijmakers et al., 2009). Organic acids in particular play important roles in the acquisition of phosphate and micronutrients (Ryan *et al.*, 2001) as well as the tolerance of aluminum (Ma *et al.*, 2001). Whether the consumption of metabolites by mirror bacteria could disrupt these or other root processes, and if so, the extent to which such disruptions would harm plants, is unclear. Mirror bacteria might disrupt soil microbial communities more generally, causing soil degradation and thereby harming plant growth and productivity in both farmed and natural environments (see Section 8.5).

# 7.3 The extent to which mirror bacteria could spread through and colonize plant vascular tissue is unclear

The xylem and phloem are the two main components of the vascular system in plants (Figure 7.4). The xylem, which is part of the apoplast, primarily functions to transport water and dissolved minerals from the roots to the rest of the plant via dead, hollow cells, using transpiration and root pressure. The phloem is a bidirectional transport system composed of living cells that moves sugars, amino acids, and other organic compounds produced through photosynthesis from leaves to other parts of the plant, including roots, fruits, and developing tissues. Apoplastic pathogens that enter the xylem or phloem can sometimes spread widely through the plant. A key question is whether localized mirror bacterial infections might do the same, as such spread could have very negative consequences to the plant.



#### Figure 7.4: Plants transport water and nutrients via xylem and phloem

**A.** Roots absorb water from soil, which is transported via the xylem through the stem to the leaves and shoot apex. Sources (mature leaves) produce sugars and other photosynthates which are transported via the phloem to sinks, including shoots, young leaves, and roots. Unlike xylem, flow in the phloem can be bidirectional. **B.** Longitudinal cross-section of an idealized vascular bundle from an angiosperm. The xylem is composed of vessel elements and tracheids, which are dead, hollow cells. Vessel elements are stacked end-to-end to form vessels that permit both water conduction and the free movement of bacteria. Vessels are connected to each other and to tracheids by pit membranes that allow water conduction, but the pits are too small to allow free movement of bacteria. The phloem is composed of sieve tube elements and companion cells. Sieve tube elements are elongated, live cells that lack a nucleus and most organelles. They are arranged end-to-end to form sieve tubes and connected by sieve plates that allow fluid and bacteria to freely flow through the tube. Each sieve tube element is connected to a companion cell via plasmodesmata. Xylem and phloem are separated from each other by the cambium, a layer of actively dividing cells.

## Mirror bacteria may or may not spread from the roots through xylem

The xylem, a key component of the plant vascular system, is primarily composed of two types of specialized cells: tracheids and vessel elements. While tracheids are found in all vascular plants, vessel elements are unique to angiosperms. Both cell types undergo programmed cell death and lignification as part of their maturation process, resulting in hollow, reinforced structures optimized for water transport. In angiosperms, vessel elements align end-to-end to form vessels, which can range from less than a centimeter to several meters in length (Jacobsen *et al.*, 2012). Lateral connections between vessels, and between vessels and tracheids, are facilitated by pit membranes. These membranes contain pores that are typically less than 100 nm in diameter (Choat *et al.*, 2008),

which permit water movement while theoretically restricting the passage of larger particles, including microorganisms. The primary function of the xylem is to enable the transpiration-driven passive transport of water and dissolved inorganic nutrients from roots to aerial parts of the plant at velocities ranging from approximately 0.3–10 mm/s (Tyree & Zimmermann, 2011; Windt *et al.*, 2006).

Pathogenic bacteria can invade and spread through xylem. Being too large to readily pass through the pit membranes, it is generally assumed that xylem-colonizing bacteria utilize cell wall degrading enzymes to travel between these regions (Pérez-Donoso *et al.*, 2010; Reinhold-Hurek & Hurek, 2011; Roper *et al.*, 2007). The velocity of xylem sap is greater than typically achievable by flagellated bacteria (Milo & Phillips, 2015), so successful colonization requires adhesion to the xylem vessel walls. Movement upstream is thought to occur through twitching motility (De La Fuente *et al.*, 2022).

Mirror bacteria would lack cell wall degrading enzymes, which should render them unable to degrade pit membranes. They would also lack specialized adaptations to adhere to, and form biofilms within, xylem, impairing their ability to move against the flow of xylem sap.

Nevertheless, passage through xylem appears to be possible in many plant species (Ryser *et al.*, 2009). Studies of microplastic beads indicate that bacterial-sized beads can be internalized through roots and transported to leaves, at least in wheat and lettuce (Li, Luo, Li *et al.*, 2020; Li, Luo, Peijnenburg *et al.*, 2020; Solomon & Matthews, 2005). Other studies have found that the human enteric pathogens *E. coli* and *Salmonella* can be internalized through roots and transported to leaves in *Arabidopsis* (Cooley *et al.*, 2003), barley (Kutter *et al.*, 2006), lettuce (Franz *et al.*, 2007; Solomon & Matthews, 2005) and tomato (Guo *et al.*, 2002); at least in *Arabidopsis*, such movement appears to require flagellar motility (Cooley *et al.*, 2003). In rice, maize, and other monocots and dicots, rhizobial bacteria enter the roots and are transported to the leaves in large numbers, although they might express cell wall degrading enzymes (Chi *et al.*, 2005; Lamb *et al.*, 1996). Given that both microplastic beads and the human enteric pathogens definitely lack such enzymes, these findings suggest that a mirror bacteria could potentially travel through xylem in at least some cases.

It is unclear how such transport occurs. One possibility is that bacterial-sized "holes" in pit membranes form often enough to create the occasional unobstructed path from roots to leaves. It is similarly unclear whether such transport can occur in other plants; studies have found successful invasion of crop roots by *E. coli* and *Salmonella* without subsequent translocation to leaves, including in spinach (Hora *et al.*, 2005), cabbage (Wachtel *et al.*, 2002), and in apple and nectarine trees (Perulli *et al.*, 2021). Given the limited evidence available—which is almost entirely restricted to young herbaceous plants—it is difficult to say anything beyond that mirror bacterial passage through xylem could plausibly occur in at least some plants.

In addition to entry through roots, direct entry into the xylem could occur through wounds, particularly those created by insects; entry from leaves might be possible in some cases, particularly if damage is created by other pathogens. Movement of mirror bacteria against the flow of xylem sap, however, appears unlikely. This suggests that they would generally not be able to spread from an infected leaf into other branches or down to the roots absent the evolution of coordinated xylem-transiting capabilities. However, it must be noted that *E. coli* and *Salmonella* are not thought to colonize xylem. The intrinsic immune evasiveness of mirror bacteria could facilitate xylem colonization. Mirror bacteria that enter roots and become trapped at a pit membrane could potentially reside there for an extended period while increasing in number, affording more opportunities to pass through in the event of a breach.

## Generalist mirror bacteria could plausibly colonize accessible regions of xylem

While physical barriers and sap flow may make it difficult for a mirror bacterium to access xylem, any regions that are accessible would plausibly have favorable conditions for growth. Xylem sap is generally considered to be nutrient-poor (Fatima & Senthil-Kumar, 2015), but concentrations of nutrients can be highly variable. Measured glucose concentrations, for instance, vary from 20  $\mu$ M (Zuluaga *et al.*, 2013) to 20 mM (López-Millán *et al.*, 2000; Zuluaga *et al.*, 2013), while concentrations of organic acids such as malate and citrate can likewise vary from below 5  $\mu$ M (Anguita-Maeso *et al.*, 2021; Fernandez-Garcia *et al.*, 2011) to several millimolar (Kasper *et al.*, 2022; López-Millán *et al.*, 2000; Prima-Putra & Botton, 1998). Amino acid concentrations are also highly variable, with individual amino acids often in the 10–100  $\mu$ M range but sometimes over 10 mM (Anguita-Maeso *et al.*, 2021; Fernandez-Garcia *et al.*, 2011; Nakamura *et al.*, 2008; Prima-Putra & Botton, 1998; Zuluaga *et al.*, 2013). Given that a mirror generalist bacterium could likely catabolize many organic acids, some amino acids, and (with straightforward engineering) glucose and other simple sugars, it seems plausible that mirror bacteria would be broadly capable of replicating within xylem. In the absence of immune recognition and an effective immune response, this could allow colonization of the xylem.

Growing mirror bacterial colonies could plausibly create physical blockages in pit membranes, the likelihood and consequences of which are difficult to assess. A potentially relevant analog is *Xylella fastidiosa*, a xylem-restricted gamma-proteobacterium with a broad host range (European Food Safety Authority, 2020; Rapicavoli *et al.*, 2018). While a benign commensal in many plants, *X. fastidiosa* acts as a pathogen in others, causing symptoms that appear to result from vascular occlusions impeding water transport (Chatterjee *et al.*, 2008; Sun *et al.*, 2013). Similar phenotypes could result from mirror bacterial infections of the xylem. However, unlike *X. fastidiosa*—which can use cell wall degrading enzymes to travel through xylem (Roper *et al.*, 2007)—mirror bacteria may be restricted to a small fraction of xylem and consequently limited in their ability to harm the plant. If most of the xylem were accessible to mirror bacteria, whether due to physiological quirks of the plant, slow spread over a long time period, or—like *X. fastidiosa*—by being spread through insect vectors, it appears plausible that a systemic infection could eventually greatly reduce flow of water through xylem, damaging the plant.

## Entry into phloem appears difficult

The phloem consists of sieve tubes, elongated sieve element cells positioned end-to-end and connected both vertically and laterally by sieve pores 0.4–15 µm in width (Bussières, 2014; Esau & Cheadle, 1959; Mullendore *et al.*, 2010; Thompson & Wolniak, 2008). Phloem sap passively flows from leaves to sink tissues—roots, buds, seeds, and flowers—at around 0.05–0.4 mm/s (Fisher, 1990;

Liesche *et al.*, 2013; Windt *et al.*, 2006). Unlike xylem sap, the flow of phloem sap can be bidirectional. For instance, phloem sap flows into budding leaves to provide nutrients, but once the leaves mature they become phloem sources rather than sinks (Will *et al.*, 2013). A mirror bacterium able to enter and survive within the phloem would therefore likely be freely transported from mature leaves toward sink tissues, including budding shoots and leaves.

Entry into phloem is more challenging than xylem, and appears only likely to occur in rare or exceptional circumstances. Entry requires first breaching the protective bundle sheath cells and then companion cells or sieve elements. While generalized plant pathogens that have broadly overcome plant defenses occasionally enter phloem (Gao *et al.*, 2016; Mensi *et al.*, 2014; Roos & Hattingh, 1987), all known plant pathogens that primarily inhabit phloem rely on phloem-feeding insect hosts for entry (Jiang *et al.*, 2019). Infected insects could potentially deliver mirror bacteria into phloem, though only if the mirror bacteria can reach the insect salivary glands and grow to sufficiently high abundance prior to incapacitating the insect.

Entry into phloem could occur through other types of wounds (Savatin *et al.*, 2014). Phloem-localized proteins and callose deposition are believed to occlude sieve plates and thus restrict movement to the rest of the phloem (Bendix & Lewis, 2018), though this remains poorly understood.<sup>14</sup> Nevertheless, there have been reported observations of bacteria entering the phloem while colonizing other plant tissue, including both plant pathogens *Xanthomonas* (Mensi *et al.*, 2014; Wang *et al.*, 2018) and *Pseudomonas syringae* (Gao *et al.*, 2016) as well as human enteric pathogens *Salmonella* (Gu *et al.*, 2011) and *E. coli* (Deering *et al.*, 2011). If confirmed, especially for bacteria lacking cell wall degrading enzymes, mirror bacteria may be capable of similar entry.

Studies performing metagenomic sequencing of phloem exudate have reported finding nucleic acids corresponding to bacterial endophytes of the leaves and roots in stone fruit and citrus (Eichmeier *et al.*, 2019; Y. Wu *et al.*, 2020). This could be because living bacteria are occasionally translocated into the phloem before being killed by the immune response or by osmotic effects. Alternatively, these results may instead be explained by transport of nucleic acids from dead apoplastic bacteria into the phloem, or from sample contamination, a common occurrence due to the challenge of obtaining clean samples of phloem (Broussard *et al.*, 2023).

If endophytes do occasionally enter into the phloem of some plants without being vectored by insects, mirror bacteria may be capable of doing the same unless cell wall degrading enzymes are required, which is possible. Given how little is known about the presence of natural-chirality bacteria in the phloem outside of the phloem-limited pathogens, more research would be needed to determine whether entry into the phloem occurs commonly, infrequently, or is almost impossible.

Mirror bacteria could plausibly cause a lethal systemic infection if they were to enter the phloem

The immunological defenses of the phloem are poorly understood (Jiang *et al.*, 2019; Zhang *et al.*, 2020). As a result, it is difficult to assess the extent to which phloem defenses would be evaded by

<sup>&</sup>lt;sup>14</sup> The role of P-proteins in occluding phloem is particularly controversial (Knoblauch et al., 2014; Noll et al., 2022).

mirror bacteria. Given the extent to which chiral recognition is required for other plant immune responses, it appears plausible that mirror bacteria could evade phloem defenses. If so, then any mirror bacteria that enter the phloem could trigger unchecked growth in this compartment, with highly deleterious consequences for the plant.

Phloem sap is very nutrient-rich. Sucrose is the primary sugar in the phloem sap of most plants, with typical concentrations over several hundred millimolar (Broussard *et al.*, 2023; Dinant *et al.*, 2010). Reducing sugars such as glucose and sucrose are typically rare (D. D. Liu *et al.*, 2012), while polyols such as mannitol and sorbitol are the primary carbohydrates in some species (Nadwodnik & Lohaus, 2008). Amino acids are the primary source of nitrogen, with total concentrations over 100 mM, including 1–10 mM of alanine (Dinant *et al.*, 2010; Hijaz & Killiny, 2014). Organic acids, such as malate, succinate, and citrate, may also be present at millimolar concentrations (Broussard *et al.*, 2023; Hayashi & Chino, 1985; Lohaus *et al.*, 2000). Phloem thus contains plentiful nutrients for a mirror *E. coli* or a similarly robust mirror bacterium, though accessing the major sugars would require catabolic pathways for sucrose, mannitol, and sorbitol.

Even though the concentrations of organic compounds within phloem sap are high, the overall water potential of around -1 to -2 MPa (Thompson & Holbrook, 2003) should support the growth of most bacteria. *E. coli*, for instance, can grow at sucrose concentrations above 1 M, though rapid changes in concentration can cause most cells to die through plasmolysis (Kawarai *et al.*, 2009; Scheie & Rehberg, 1972). Phloem sap is also slightly alkaline (Dinant *et al.*, 2010), and temperatures are close to those of the external environment.

As discussed in Box 1.2 of Chapter 1, millimolar concentrations of individual D-amino acids can inhibit bacterial growth. Given the very high amino acid concentrations in phloem sap, it is plausible that amino acid toxicity could reduce or entirely preclude the growth of some kinds of mirror bacteria. However, resistance to these toxicities may be relatively accessible to evolution, such as through inactivation of the permease importing the toxic amino acid into the cytosol or upregulation of an existing detoxifying enzyme.

Given the high nutrient concentrations and likely failure of immunological defenses in phloem, it is plausible that a mirror bacterium able to enter and survive in phloem could replicate unchecked and reach very high concentrations. Mirror bacteria could accumulate in the phloem of roots, shoots, and other sinks, which could cause severe disruption to neighboring tissues by consuming nutrients, releasing metabolic waste products, or simply creating physical blockages.

In addition to the potential for deleterious effects from organic nutrient depletion, the introduction of an aerobic bacterial pathogen into the phloem could be quite harmful to the plant. Unlike animals, plants lack an active circulatory system specialized for distributing oxygen to tissue, and hypoxia within phloem tissue occurs within healthy plants (van Dongen *et al.*, 2003). Even if restricted to achiral nutrients, aerobic respiration by mirror bacteria would likely be limited by oxygen, not organic carbon, such that regions of phloem colonized by mirror bacteria could readily become anoxic. Anoxic plant cells must produce ATP through fermentation, which yields much less energy than aerobic respiration per mole of carbon and produces toxic byproducts such as ethanol and lactate (Geigenberger, 2003). Plants vary in their ability to tolerate temporarily anoxic conditions

(Bailey-Serres & Voesenek, 2008), but long-term anoxia would almost certainly disrupt plant cell functionality and lead to cell death (Loreti *et al.*, 2016).

Other disruptions are possible. The phloem is continuous with the cytoplasm of adjacent plant cells, so any changes to phloem acidity, redox potential, or composition could disrupt functionality of neighboring cells. Even mirror proteins and other mirror macromolecules from lysed mirror bacteria could be transported through plasmodesmata into neighboring cells (Stadler *et al.*, 2005), where they would remain undigested. Large mirror bacterial populations could themselves create physical blockages that prevent flow of phloem into roots, shoots, and other sink tissues.

Disruptions to phloem unloading caused by localized cell death or bacterial overgrowth would stunt or kill roots, shoots, fruits, and other sink tissues. Root death would, in turn, prevent the flow of water and inorganic nutrients back to the leaves, leading to wilting and plant death. Systemic stunting and necrosis are common symptoms of phloem-limited pathogens (Bendix & Lewis, 2018), which provide possible analogs to a mirror bacterial infection. Interestingly, phloem-limited pathogens are typically incapable of aerobic respiration (Christensen *et al.*, 2005; Duan *et al.*, 2009; Marcone, 2014; Oshima *et al.*, 2004) and may be poor analogies for mirror bacteria in other important ways.

Phloem immunological defenses are poorly understood and likely variable between plant species; any attempt to predict the consequences of a putative mirror bacterial phloem infection is therefore necessarily speculative. Nevertheless, if mirror bacteria are able to enter the phloem and establish an infection—which remains unclear—the consequences are likely to be highly deleterious for most plants.

## 7.4 Countermeasures for agricultural plants

The vast majority of human calories are produced through agriculture, with more than half coming from just four crops: wheat, rice, maize, and soybeans (D'Odorico *et al.*, 2014). Crop pests and pathogens, especially fungi, are responsible for yield losses of approximately 25% (Savary *et al.*, 2019). Given that fairly mild biotic and abiotic stressors can cause large productivity losses (Boyer, 1982), even relatively mild mirror bacterial infections could plausibly cause significant reductions in yield. Worse outcomes might be possible. Wheat reportedly transports bacteria-sized microplastic beads from its roots to the leaves (Li, Luo, Li *et al.*, 2020; Li, Luo, Peijnenburg *et al.*, 2020), while rice and maize transport rhizobial microbes in a similar manner (Chi *et al.*, 2005; Lamb *et al.*, 1996). If these organisms are susceptible to systemic infection, then invasive mirror bacteria could cause substantial losses in the major crops.

## Conventional methods of controlling bacterial pathogens would be of limited use against mirror bacteria

As discussed in Chapter 8, a sufficiently robust mirror bacterium could plausibly survive or even colonize many soils and spread widely through animal hosts and human activities. If such a strain were also broadly capable of infecting plants, it could cause unprecedented and widespread agricultural losses. Preventing crop exposure to invasive mirror bacteria through cultural methods

such as quarantines, crop rotations, elimination of alternative hosts, and removal of infected plants is unlikely to be effective. Soil treatments such as solarization, steam heating, fire, or chemical disinfestation could presumably reduce mirror bacteria concentration within soils, but are not feasible at scale, would severely disrupt natural microbiota, and/or may not be able to completely eliminate the bacteria. Even if mirror bacteria were to be eliminated in a field, they could be reintroduced by animals or from windborne dust and debris.

Antibiotics have seen limited use in agricultural contexts as their use is not economical (McManus *et al.*, 2002); they are generally only effective at preventing infection and remain active only for a few days (Stockwell & Duffy, 2012). Fungicides, which are far more common in agriculture, may be a better comparison. Many systemic fungicides, which are internalized into the plant and typically transported through the apoplast—phloem mobility is rare—are able to cure and eradicate, rather than merely prevent, fungal infection (Oliver & Hewitt, 2014). However, even if systemic antimirror compounds were identified, they would need to be constantly reapplied to crops to prevent reinfection from environmental reservoirs such as wild plants and untreated soils. The scale of production required to protect crops at scale may therefore prove infeasible.

Copper-based antimicrobial compounds are achiral and comparatively cheap. Due to their phytotoxicity, however, their application is limited to the preventative treatment of leaves: they have no curative or systemic activity (Lamichhane *et al.*, 2018).

Biological methods use natural enemies such as predators or microbial antagonists to control pathogens (van Lenteren *et al.*, 2018). As discussed in Section 8.1, mirror bacteria could be resistant to most forms of predation and many forms of microbial antagonism, and it is unlikely that existing biological controls would be very effective. More speculative biological controls, such as mirror phages or genetically engineered microbes, are described in Section 8.6. Using such methods to eliminate mirror bacteria from a local environment such as a field appears to be very difficult; containing and eliminating invasive mirror bacteria from the environment more broadly does not appear feasible by any method.

## Plants might be engineered to resist infection

While highly speculative, engineering plants to be resistant to mirror bacteria could be a feasible strategy to enable agricultural production at scale despite the presence of an invasive mirror bacterium. One promising strategy would involve engineering immune recognition of certain mirror MAMPs, thus allowing innate immune defenses to trigger in response to mirror bacteria.

Plant PRRs, which are responsible for monitoring the apoplast, are conserved and remain functional when expressed transgenically in a wide range of plants (Lacombe *et al.*, 2010). For example, the EF-Tu receptor from *Arabidopsis* protects against bacterial diseases in tomato (Kunwar *et al.*, 2018), potato (Boschi *et al.*, 2017), and citrus (Mitre *et al.*, 2021). If any naturally occurring PRRs have some cross-affinity to mirror bacterial cell wall components—which is plausible, given the diversity of PRRs found in nature (Boutrot & Zipfel, 2017)—then these could be enhanced and transferred to major crops.

Engineering the binding ectodomain of a PRR to recognize unique signatures of mirror bacteria would allow targeting of patterns such as p-peptide multimers that are absent from nature. Although there is not yet any equivalent to engineered mammalian SNIPR receptors in plants (Zhu et al., 2022), the anti-flagellin ectodomain of the FLS2 has been successfully evolved to recognize a different peptide (Helft et al., 2016), and similar leucine-rich repeat domains have been retargeted by phage and yeast display (Velásquez et al., 2017; Wezner-Ptasinska & Otlewski, 2015). Once a suitable ectodomain has been engineered, it could be used to create chimeric PRRs with native intracellular kinase domains. Chimeric PRRs utilizing natural ectodomains have already been engineered and found to work in planta (Albert & Felix, 2010; Schwessinger et al., 2015; J. Wu et al., 2019). Directed evolution of the ectodomain or design of a novel binder specific to mirror peptide multimers followed by engraftment onto an existing PTI receptor could potentially confer mirror-specific recognition and induction of the plant immune response. The XA21 receptor from rice, which provides ETI-level resistance despite being extracellular ETI (Ercoli et al., 2022), is a particularly promising candidate for engineered p-peptide multimer recognition. For example, promiscuous D-peptide minibinders that do not bind biomolecules found in plant tissues might be incorporated into the N-terminal half of the native leucine-rich repeat ectodomain scaffold (Yeh et al., 2023), likely preserving the signal-transducing interface between binding and transmembrane domains while reprogramming receptor specificity. Encoding several such engineered receptors recognizing a variety of p-peptide multimers could robustly confer immunity to mirror bacteria.

Another possible route might leverage the nucleotide-binding, leucine-rich repeat (NLR) proteins that recognize intracellular bacterial effectors (Kourelis & van der Hoorn, 2018), which have been subject to more extensive engineering than PRRs (Bentham *et al.*, 2023; Cesari *et al.*, 2022; De la Concepción *et al.*, 2019; Y. Liu *et al.*, 2021; Maidment *et al.*, 2023; Zdrzałek *et al.*, 2024). Most relevantly, nanobody-NLR fusions have been demonstrated to trigger ETI upon recognition by the nanobody, suggesting that this approach can be generalized to recognize any molecular trigger (Kourelis *et al.*, 2023). However, invasive mirror bacteria are not expected to enter plant cells, so traditional NLRs would not be useful: the relevant signaling domains would need to be fused to the transmembrane domain of an extracellular receptor, as with XA21.

To compensate for the reduced efficacy of many plant antimicrobials against mirror bacteria, new antimirror molecules might be engineered into plants, much as the transgenic expression of antimicrobial peptides can increase plant tolerance to existing bacterial pathogens (Balaji & Smart, 2012; Hao *et al.*, 2016; Zou *et al.*, 2017). For example, the constitutive overexpression of chirality-independent antimicrobial peptides could provide always-on resistance, while inducible effectors triggered by synthetic mirror-specific receptors could provide targeted immunity. Metabolic pathways for anti-mirror small molecules might also be introduced or (for achiral antimicrobials) upregulated. Modifications to limit the availability of key nutrients could provide an additional method to increase plant resistance to infection.

While many of the strategies for engineering heritable resistance to mirror bacteria may appear promising, it is important to note that all of these putative methods of preventing infection are necessarily speculative. They may have unintended and undesired consequences, such as reducing yields, and may not prove entirely effective at preventing mirror bacterial contamination of food.

## Chapter 7: Plant Infection

Developing microbial methods of food production or cultivating simple plants such as duckweeds that are likely to resist mirror bacterial infection might provide a more robust and generalizable defense against threats to traditional agriculture, but these alternatives are almost wholly undeveloped. If engineering heritable mirror resistance in crops is pursued, it would need to be performed separately in each crop species or cultivar. Protecting wild plants and their associated ecosystems would not be feasible.

## **Chapter 8: Environmental Survival and Spread**

Deepa Agashe<sup>1</sup>, Damon J. Binder<sup>2</sup>, Vaughn S. Cooper<sup>3</sup>, Kevin M. Esvelt<sup>4</sup>, Richard E. Lenski<sup>5,6</sup>, David A. Relman<sup>7,8,9</sup>

Populations of organisms expand when their reproductive rate exceeds their death rate. The availability of resources and efficiency of conversion into progeny determine rates of reproduction. Resource availability decreases in the presence of competing species. Death rates are governed by natural enemies such as predators and pathogens, as well as by environmental stresses such as desiccation, extreme temperatures, and starvation. When a species' reproductive rate exceeds its death rate, its abundance will increase until resources are exhausted, natural enemies proliferate, or the environment changes. For most species there is an approximate balance between births and deaths, such that populations remain approximately constant or exhibit a seasonal cycle influenced by nutrient availability and predators.

Occasionally, the population of a species introduced into a new environment with fewer natural enemies, including predators, can increase dramatically, sometimes causing great damage to natural and agricultural ecosystems. Even if the introduced species is somewhat less adapted to its abiotic conditions or local resources than native species, it can often expand and cause harm—becoming invasive—due to a low death rate resulting from the absence of natural enemies.

A major concern in this chapter is that mirror bacteria could, by virtue of their altered chirality, escape from many natural enemies and so have a greatly reduced death rate due to predation and parasitism. In addition, as discussed in Chapters 4, 6, and 7, mirror bacteria could plausibly infect and survive in a wide range of multicellular organisms. Nevertheless, mirror bacterial growth rates may be substantially lower than those of natural bacteria, as mirror bacteria would have a reduced capacity to obtain nutrients and would lack other specialized adaptations for most environments encountered. Should the advantages from reduced predation and any benefits from infecting multicellular organisms outweigh the disadvantages in a given environment, mirror bacteria could then colonize that ecosystem.

*Authors are listed in alphabetical order. Affiliations:* <sup>1</sup>National Centre for Biological Sciences, Tata Institute of Fundamental Research, Bengaluru, India; <sup>2</sup> Open Philanthropy, San Francisco, California, USA; <sup>3</sup> Department of Microbiology and Molecular Genetics, University of Pittsburgh, Pittsburgh, Pennsylvania, USA; <sup>4</sup> Media Laboratory, Massachusetts Institute of Technology, Cambridge, Massachusetts, USA; <sup>5</sup> Department of Microbiology, Genetics, and Immunology, Michigan State University, East Lansing, Michigan, USA; <sup>6</sup> Program in Ecology, Evolution, and Behavior, Michigan State University, East Lansing, Michigan, USA; <sup>7</sup> Department of Medicine, Stanford School of Medicine, Stanford University, California, USA; <sup>8</sup> Department of Microbiology and Immunology, Stanford School of Medicine, Stanford University, California, USA; <sup>9</sup> Infectious Diseases Section, Veterans Affairs Palo Alto Health Care System, Palo Alto, California, USA

Imagine, for example, an environment in which the natural-chirality bacteria, on average, divide once per day and die once per day, thereby maintaining roughly steady numbers. Now imagine a mirror bacterium is introduced that escapes predation and experiences only a quarter of that mortality. Even a two-thirds reduction in the mirror bacterium's growth rate would still exceed its death rate, allowing the mirror bacterium to invade and increase in numbers until the available nutrients are sufficiently depleted. Without effective predation and an accompanying increase in the number of predators, mirror bacterial populations could potentially reach very high densities, with extraordinarily damaging consequences for the environment, agriculture, and human wellbeing.

Section 8.1 discusses the effects of biotic processes on survival. Phage lysis, protist predation, and microbial warfare rely primarily on chiral mechanisms. Mirror bacteria would likely experience reduced rates of predation and would be completely resistant to natural phages, which may allow long-term survival outside of hosts and facilitate transmission between infected multicellular organisms.

Section 8.2 discusses the possibility that a mirror bacterial generalist capable of evading most forms of biological control could invade environments outside of multicellular hosts. Mirror heterotrophic bacteria could be capable of utilizing a number of common nutrients, and though their growth would be impaired compared to that of natural bacteria, escape from natural predators could still enable the colonization of some environments. Such invasive bacteria could plausibly grow until most nutrients are depleted, as most top-down biological controls would be absent.

Section 8.3 describes the possible spread of an invasive mirror bacterium through the environment by hitchhiking on human transportation and animal hosts, or by wind or water, allowing rapid dispersal.

Section 8.4 examines the possible evolution of invasive mirror bacteria. Absent human intervention, there would be little genetic diversity, and mirror bacteria would have no ability to obtain genetic material from natural chirality bacteria, leaving *de novo* mutation as the only source of variation and new capabilities. Nonetheless, evolutionary adaptation under conditions of population expansion into new environments can be rapid. By contrast, we expect there would be little potential for predators to evolve to consume mirror bacteria within any human-relevant timescale.

Section 8.5 describes potential ecological harms that could result from the spread of invasive mirror bacteria, including the extinction of susceptible animals and plants as well as changes to habitats and biogeochemical cycling.

In Section 8.6 we close with a brief discussion of possible countermeasures. With sufficient coordination and resources, certain countermeasures could mitigate some harmful effects, but they appear unlikely to prevent widespread and irreversible environmental harm.

# 8.1 Mirror bacteria would be inherently resistant to many biological controls

In natural environments, individual bacterial cells often perish from predation by protists and bacteriophages (viruses that infect bacteria, often referred to as "phages") or are killed by the microbial weapons of competitors (Kirchman, 2018; Pernthaler, 2005; Thingstad, 2000). Mirror

bacteria would likely be largely resistant to protist grazing, entirely resistant to natural phages, and resistant to most weapons of microbial competitors. This inherent resistance could substantially enhance the survival of mirror bacteria in most ecosystems.

## Mirror bacteria would be resistant to natural phages

Phages are responsible for a large fraction of bacterial deaths in many environments. There are 10 to 20 phages for every bacterial cell in both soils and natural waters (Cobián Güemes *et al.*, 2016), leading to frequent encounters (A. G. Murray & Jackson, 1992). However, not every cell is susceptible to every phage, and natural bacteria have evolved defenses against some of the phages that infected their ancestors. Despite these defenses, bacterial death rates from phages in the best-studied marine environments range from ~0.01 day<sup>-1</sup> to over 20 day<sup>-1</sup> (Carlson *et al.*, 2022; Mann, 2003; Weinbauer, 2004). Many bacterial populations grow rapidly while they are resistant or otherwise phage-free, then encounter a phage to which they are susceptible and are dramatically reduced in number. For example, cyanobacterial "blooms" created by nutrient influxes in aquatic environments are often ended by phages (Grasso *et al.*, 2022; T. D. Harris *et al.*, 2024; Mann, 2003). Mirror bacteria would be inherently and completely resistant to every natural phage, likely providing a major fitness advantage.

All viruses parasitize their host's cellular translation system and nucleoside triphosphate pool to produce proteins and new viral genomes. Mirror ribosomes could not produce natural-chirality proteins, nor could mirror nucleotide triphosphates be used to replicate natural-chirality viral genomes. Thus, no natural phage could successfully reproduce within a mirror bacterium. Even infection—the entry of the phage's genetic material into a mirror bacterium—is mediated by chiral phage proteins binding to chiral cellular receptors, and therefore is highly unlikely to occur (Labrie *et al.*, 2010).

In summary, natural phages would not be able to infect mirror bacteria, completely removing a major source of bacterial mortality. Only synthetic mirror phages, which do not currently exist, could kill and replicate within mirror bacteria; they are discussed as a potential medical countermeasure in Section 5.3 and as a potential environmental countermeasure in Section 8.6.

## Mirror bacteria would probably be resistant to other predators

In addition to phages, bacteria are preyed upon by phagocytic amoebae and other protists, other bacteria, and multicellular eukaryotes including nematodes, arthropods, and rotifers (T. D. Harris *et al.*, 2024; Morgan *et al.*, 2010; Potapov *et al.*, 2022). Mirror bacteria could be harder for predators to track because they would leave trails of mirror-image biomolecules that would not properly bind to the predator's receptors; they could be resistant to being killed even when consumed; and they would provide few nutrients of use to their predators. Animals that ingest mirror bacteria could instead become infected by them (Chapter 6).

We first discuss protists. In marine ecosystems, protists are often the leading cause of bacterial death and the primary control on bacterial abundance (Pernthaler, 2005). Grazing rates by microzooplankton on phytoplankton of ~40% per day are common, with approximately two-thirds of

primary production consumed by this protist-dominated group (Calbet & Landry, 2004); however, which species are primarily responsible for this consumption is much less certain (Frias-Lopez *et al.*, 2009; Wilken *et al.*, 2023). Although less well-studied than the marine variety (Geisen *et al.*, 2017), terrestrial protists are also major consumers of bacteria in soil (Geisen *et al.*, 2018; Trap *et al.*, 2016).

Bacterial consumption by protists usually occurs through phagocytosis. This is the same process used by the professional phagocytes of the human immune system (Desjardins *et al.*, 2005), and so protist phagocytosis would therefore likely suffer from analogous defects to those described in Section 4.2. Protists are highly diverse, and the molecular mechanisms involved in the tracking, recognition, engulfment, and killing of prey are not well-characterized in any organism. This makes it difficult to determine how a given protist would respond to mirror bacteria, particularly given that recognition and killing could still occur to some extent through non-specific or achiral mechanisms.

Box 8.1 discusses phagocytosis of mirror bacteria in the model amoeba *Dictyostelium discoideum*, which is perhaps the most well-studied of all protists. The evidence here suggests that mirror bacteria could be ingested at greatly reduced rates due to the absence of specific prey signals such as folate and peptidoglycan (Pan *et al.*, 2018). Whether ingested mirror bacteria would survive phagocytosis is unclear, but it is likely that the undigested mirror bacterium—whether living or dead—would be egested after several hours.

Outside of *D. discoideum*, the molecular mechanisms involved in phagocytosis are less characterized. Prey-seeking based on chiral mechanisms occurs in many protists, which can sense and actively swim toward bacterial exudates<sup>15</sup> such as sugars and amino acids (Spero, 1985; Martel, 2006; Rashidi & Ostrowski, 2019; Roberts *et al.*, 2011) and discriminate between prey through the recognition of specific surface compounds (Jürgens & Matz, 2002; Montagnes *et al.*, 2008; Shi *et al.*, 2021) such as mannose (Wootton *et al.*, 2007). Since most recognized surface molecules are thought to be chiral, it is likely that most protists would ingest mirror cells at substantially lower rates. However, protists ingest latex beads and microplastics (Rillig & Bonkowski, 2018) and some protists consume them at rates similar to their consumption of natural bacteria (Pace & Bailiff, 1987), so some phagocytosis of mirror bacteria would likely occur even if chemotaxis were to completely fail.

Once engulfed, bacteria are subjected to acidic conditions and lytic antimicrobial enzymes (Guerrier *et al.*, 2017; Shi *et al.*, 2021) and, at least in *D. discoideum*, ROS and transition metal poisons as well (Dunn *et al.*, 2017). Lytic enzymes are not expected to function, so efficient intracellular killing of mirror bacteria would be likely only if achiral mechanisms were adequate and triggered absent prey signals. It is not clear that this is the case for *D. discoideum* (Box 8.1). Regardless of whether it would be killed or not, the mirror bacterium would not be digested. Most likely either the mirror bacterium or its remains would be egested, but other outcomes are possible if egestion fails. In some cases it may even be possible for living mirror cells to persist within protist food vacuoles like certain

<sup>&</sup>lt;sup>15</sup> While most solute attractants are likely chiral, there are exceptions. The marine dinoflagellate *Oxyrrhis marina*, for example, is attracted to achiral dimethylsulphoniopropionate (Breckels *et al.*, 2011) as well as to chiral amino acids and sugars (Martel, 2006). Volatile compounds are small and often achiral, and so it is plausible that mirror bacteria might "smell" more normal to soil protists than they "taste". The extent to which volatile compounds produced by bacteria influence predatory protists is largely unknown, but one study of three protists and six bacterial species found effects that were strongly dependent on the specific pairing of predator and prey (Schulz-Bohm *et al.*, 2017).

## Box 8.1: *Dictyostelium discoideum* may not effectively phagocytose and kill mirror bacteria

*Dictyostelium discoideum* is a soil-dwelling amoeba (or "slime mold") used as a model organism to study phagocytosis, and has often served as a model for human phagocytic cells.

The mechanisms by which *D. discoideum* recognizes prey are not well understood (Dunn *et al.*, 2017), but different bacterial species are known to elicit specific transcriptional responses (Lamrabet *et al.*, 2020; Nasser *et al.*, 2013). The G-protein-coupled receptor fAR1 detects chiral folate and lipopolysaccharide produced by bacteria (Pan *et al.*, 2018); one fAR1 knockout strain showed a significant reduction in uptake of the prey bacterium *Klebsiella aerogenes* (Pan *et al.*, 2016). Other, as yet uncharacterized, pathways appear to be involved in uptake of *Bacillus subtilis* (Lima *et al.*, 2014). There is also evidence for the use of (as yet unidentified) lectin-like receptors in phagocytosis (Vogel *et al.*, 1980). It is unclear if any would detect achiral molecules. *D. discoideum* can also phagocytose latex beads, although mutants that can phagocytose beads but not bacteria are known (Vogel *et al.*, 1980), suggesting that latex bead uptake is driven by hydrophobic interactions that are absent for many natural bacteria and therefore may not be that relevant for their mirror counterparts. The extent to which *D. discoideum* would engulf mirror bacteria thus remains unclear, but it appears likely that engulfment would occur at a substantially lower frequency compared to natural bacteria.

Once the outer membrane engulfs a food particle, it becomes enclosed within a compartment called a phagosome. During phagosomal maturation the pH decreases to 3.5–4.5 (Gopaldass *et al.*, 2012; Marchetti *et al.*, 2009), and lytic enzymes, oxidases, and metal transporters are delivered, contributing to a bactericidal environment (Dunn *et al.*, 2017). During late maturation nutrients are absorbed and enzymes retrieved. Undigested material is then egested from the cell; the entire process occurs over a few hours. Phagosomal trafficking, acidification, the delivery of lysosomal hydrolases, and the final expulsion of undigested material all proceed normally during the phagocytosis of latex beads (Gotthardt *et al.*, 2002; Padh *et al.*, 1993) and so would likely also occur in response to the ingestion of a mirror bacterium.

*D. discoideum* deploys both chiral mechanisms, such as lytic enzymes; and achiral mechanisms, such as reactive oxygen species (ROS) and transition metal poisons (Barisch *et al.*, 2018; Hao *et al.*, 2016), to kill bacteria. Knockout experiments suggest that reducing the efficacy of lysosomal enzymes greatly impairs intracellular killing of *Escherichia coli* and other bacteria, while the effect of removing ROS generation or increasing phagosomal pH appears comparatively modest (Crespo-Yanez *et al.*, 2023; Jauslin *et al.*, 2021). Killing mechanisms vary with prey (Jauslin *et al.*, 2021; Nasser *et al.*, 2013) and the detection of folate is known to play an important role in intracellular killing (Bodinier *et al.*, 2020; Leiba

*et al.*, 2017). Whether other chiral signals also regulate intracellular killing is unknown. Given that lytic enzymes are unlikely to harm mirror bacteria, it appears that intracellular killing could be greatly impaired, but whether this would allow mirror *E. coli*, for example, to survive ingestions by *D. discoideum* is unclear.

In summary, *D. discoideum* would likely ingest mirror bacteria at lower rates than natural-chirality bacteria, and it is unclear how effectively mirror bacteria would be killed in the phagosome. After a few hours it is likely that either the mirror bacterium or its largely undigested remains would be egested from the cell, with the protist deriving few, if any, nutrients from it.

natural bacteria, including species which are not typically intracellular parasites (Barker *et al.*, 1999). Mirror metabolites released following the successful lysis of a mirror bacterium might also be imported into the protist cytosol or (if hydrophobic) incorporated into the protist lipid membrane, where they might accumulate and may or may not prove harmful to the protist.

To summarize, the mechanisms involved in protist predation are complex and not well understood. While protist species could exhibit diverse responses to a mirror bacterium, in most cases it is likely that key mechanisms would fail to function effectively, resulting in reduced rates of ingestion, killing, and digestion (Figure 8.1).

The nematode *Caenorhabditis elegans* illustrates some of the interactions between bacterivorous animals and mirror bacteria. As discussed in Chapter 6, *C. elegans* is a filter feeder that would ingest mirror bacteria, though the absence of chiral signals may somewhat reduce intake compared to the nematode's preferred prey. Lytic enzymes play key roles in both killing and digesting prey, both on their own and in synergy with other mechanisms such as pharyngeal grinding, which crushes most ingested bacteria (Seymour *et al.*, 1983) but is impaired in the absence of lysozyme (Gravato-Nobre *et al.*, 2016). Because mirror bacteria are expected to be resistant to these enzymes, it is plausible that those which survive ingestion would be expelled along with other waste. If able to persist within the intestine they could plausibly cause a lethal infection. If infection does not occur, the nematode might learn to avoid mirror bacteria as low-quality food items.

Bacteria are subject to many other forms of predation, including phages (discussed in the previous section), parasitic and predatory bacteria, and a diverse array of animals. A detailed analysis of all possibilities is beyond the scope of this report. Nevertheless, failures of recognition, killing, and digestion are expected to be common. Lytic enzymes, in particular, are used by almost all bacterial predators to kill and digest bacteria (Karasov & Douglas, 2013; Lancaster *et al.*, 2019; J. Pérez *et al.*, 2016), and while this is usually not the only killing mechanism, it often acts synergistically with others. This deficiency could partially, in some cases perhaps substantially, reduce the mortality rates experienced by mirror bacteria, increasing environmental persistence and potentially allowing net positive population growth in some environments.



### Figure 8.1: Mirror bacteria could be resistant to protists, a major cause of bacterial mortality

(A) Phagocytosis of natural-chirality bacteria by protists relies primarily on chiral mechanisms at each step: tracking, recognition, internalization, killing, and digestion. (B) Phagocytosis of mirror bacteria could fail because each step at least partially depends on the interactions of chiral molecules. Failure of any of the first three steps (tracking, recognition, or internalization) could result in failed predation. Should phagocytosis succeed but the mirror bacteria survive, multiple outcomes are possible, including ones in which the protist suffers harm. Even if mirror bacterial killing occurs, the protist would not be able to properly digest and extract nutrients.

Even predators that do successfully ingest and kill mirror bacteria would derive few nutrients from them. Natural enzymes would fail to decompose mirror proteins and other chiral macromolecules, which make up over 95% of the dry biomass of a typical bacterial cell (Milo & Phillips, 2015). Most small molecule metabolites in bacteria are also chiral (Bennett *et al.*, 2009), and so they too could probably not be catabolized. Under normal conditions, a growing bacterial population facilitates the

Chapter 8: Environmental	Survival	and Spread
--------------------------	----------	------------

Group	Examples	Target / Mechanism of action	Activity vs Mirror	References
Achiral antibiotics				
Dithiolopyrrolones	Holomycin, Thiolutin	RNA polymerase	Yes*	B. Li et al., 2014
Nitroimidazoles	Azomycin	Ribonucleotide reductase	Yes*	Saeki et al., 1974
Pigments	Prodigiosin, Pyocyanin, Violacein	Various	Yes*	Choi <i>et al.</i> , 2015; Danevčič <i>et al.</i> , 2016; Noto <i>et al.</i> , 2017
Membrane-disrupti	ng antibiotics			
Enniatins	Enniatin B	Lipid membranes	Yes*	Kamyar <i>et al.</i> , 2004; Shenyakin <i>et al.</i> , 1967
Lipopeptides	Daptomycin	Phosphatidylglycerol	No*	Moreira & Taylor, 2022
	Surfactin	Lipid membranes	Yes*	Zhen et al., 2023
Polyethers	Routiennocin	Ionophores	Yes*	Borcher Møller et al., 2024
	Monensin, Salinomycin	Ionophores	Yes*	Rutkowski & Brzezinski, 2013
Polypeptides	Bacitracin A	Bactoprenylpyrophosphate	e Yes*	McDougal & Griffin, 2003
	Gramicidin A	Ionophore	Yes*	Providence et al., 1995
	Polymyxin B	Lipid A	No*	Slingerland et al., 2022

**Table 8.1: Naturally occurring antibiotics that could retain efficacy against mirror bacteria** Most antibiotics are unlikely to retain bactericidal activity against mirror bacteria (see Section 5.1). Those listed in this table are exceptions, either because they are achiral or, although chiral, act by disrupting membrane function. For some of the chiral antibiotics, the efficacy of the enantiomer against bacteria has been tested, allowing the efficacy of the natural-chirality antibiotic against a mirror bacterium to be inferred. In those cases marked by \*, no such results are available, so the likely efficacy is inferred from what is known about the mechanism of action.

expansion of predator populations, which thereby increases the bacterial mortality rate and keeps the prey population in check (Kirchman, 2018; Levin *et al.*, 1977). A growing mirror bacterial population, however, would not support the growth of predators, and so it would likely expand until eventually constrained by resources.

## Mirror bacteria would be resistant to many natural antimicrobial toxins

Microbes have evolved a diverse array of toxins and other molecular weapons to harm and kill competitors (Granato *et al.*, 2019). Bacterial weapons are particularly well-documented in the gut microbiota of mammals (García-Bayona & Comstock, 2018) and the rhizosphere of plant roots (Doornbos *et al.*, 2012; Hassani *et al.*, 2018). Almost all known microbial weapons operate by chiral mechanisms and are unlikely to affect mirror bacteria.

Most antibiotics are chiral and typically function by inhibiting chiral enzymes and ribosomes required for key cellular processes such as translation or cell wall biosynthesis (Kohanski *et al.*, 2010). Thus, they are unlikely to be effective against a mirror bacterium (see Section 5.1). The only exceptions are those antibiotics that are achiral, act as ionophores, or otherwise disrupt membrane

	Examples	Mechanisms	Target / Receptor	Activity vs Mirror	References
Class I: Modifie	d bacteriocins				
Lantibiotics	Mersacidin	Membrane disruption	Lipid II	No	Hsu et al., 2003
	Nisin, Subtilin	Cell wall inhibition, membrane disruption	Lipid II	No	Hsu <i>et al.</i> , 2004; Parisot <i>et al.</i> , 2008
Lasso peptides	Capistruin, Microcin J25	Block transcription	RNA Pol III	No	Bayro <i>et al.</i> , 2003; Kuznedelov <i>et al.</i> , 2011
Thiopeptides	Thiostrepton	Block translation	50S ribosome	No	Harms et al., 2008
	GE2270A	Block translation	EF-Tu	No	Parmeggiani et al., 2006
Class IIa: Pedio	cin-like				
Pediocin-like	Pediocin, Leucocin A	Membrane disruption	Mannose-PTS	No	Gravesen <i>et al.</i> , 2002; Yan <i>et al.</i> , 2000; Zhu <i>et al.</i> , 2022
Class IIc: Cycli	c peptides				
Circular bacteriocins	AS-48, Carnocyclin, Gassericin	Membrane disruption	No receptor	Yes	Gálvez et al., 1991; Gong et al., 2009; Kawai et al., 2004
	Garvicin ML	Membrane disruption	Maltose transporter	r No	Gabrielsen et al., 2012
Class IId: Misco	ellaneous unmod	lified bacteriocins			
Saposin-like leaderless peptides	Aureocin A53, Lacticin Q, Enterocin L50	Membrane disruption	No receptor	Yes	Lander <i>et al.</i> , 2023; Netz, 2002; Tymoszewska <i>et al.</i> , 2021; Yoneyama <i>et al.</i> , 2009
Other	Garvicin Q	Membrane disruption	Mannose-PTS	No	Tymoszewska et al., 2017

### Chapter 8: Environmental Survival and Spread

Table 8.2: Efficacy of broad-spectrum bacteriocins against mirror bacteria

Examples of bacteriocins known to exhibit a relatively broad spectrum of antibacterial activity, listed according to a proposed classification system (Cotter *et al.*, 2013). Of these, only certain circular and leaderless bacteriocins act by disrupting membranes without a specific target on the cell surface and so are expected to remain harmful to mirror bacteria. For those examples in **bold**, testing of the enantiomer for antibacterial activity has been reported in the literature.

function (Table 8.1). There are very few naturally occurring achiral antibiotics—all of those on the WHO List of Medically Important Antimicrobials, for instance, are synthetic (World Health Organization, 2024)—but a number of non-ribosomal peptides and polyether ionophores would retain efficacy, primarily against the mirror versions of Gram-positive bacteria.

Bacteriocins are small ribosomally synthesized peptides with diverse mechanisms of action which are usually chiral (Table 8.2). Similar to antimicrobial peptides produced by humans (discussed in Section 4.2) and other eukaryotes, many bacteriocins form pores within membranes or otherwise disrupt membrane function (Acedo *et al.*, 2018). Although the basal mechanism is typically insensitive to membrane chirality, bacteriocins normally target the membrane through specific chiral receptors on the cell surface, and consequently are unlikely to retain efficacy against mirror bacteria. This inference has been verified experimentally in the case of leucocin A: its p-enantiomer showed

no antimicrobial activity at the minimum inhibitory concentration of the natural-chirality compound (Yan *et al.*, 2000).

The only bacteriocins likely to retain efficacy against mirror bacteria are those that do not require receptors to target membranes. To our knowledge, all such bacteriocins are either circular or leaderless peptides. They have shared structural features and broad-spectrum activity against Gram-positive bacteria, but are only effective against Gram-negative bacteria if the outer membrane is disrupted (Towle & Vederas, 2017). The D-enantiomers of the leaderless peptides aureocin A53 and lacticin Q were recently tested and found to have similar antimicrobial activity to their L-enantiomers (Lander *et al.*, 2023), confirming that the natural L-enantiomers should be effective against mirror Gram-positive bacteria.

Larger bacterial toxins, mechanical weapons, and prophages are almost always narrow-spectrum and generally require binding to specific receptors to properly function (Table 8.3). The type VI secretion system provides an exception (Acedo *et al.*, 2018), but only when equipped with a pore-forming protein such as Ssp6 or VasX. Because most bacterial weapons have not yet been characterized, additional achiral antibiotics, pore-forming peptides, and perhaps more sophisticated weapon systems that would retain efficacy may exist. Nevertheless, the available evidence suggests that mirror bacteria would be intrinsically resistant to the vast majority of weapons used for bacteria-on-bacteria warfare.

Unlike the specialized weapons considered so far, some metabolic byproducts such as ethanol (Ingram, 1990) and small organic acids (Pinhal *et al.*, 2019) act as broad inhibitors of bacterial growth when present at high concentrations. These metabolites are achiral or cause harm through general mechanisms, such as membrane uncoupling, that are not sensitive to chirality. More generally, microbial metabolic processes can create unfavorable growth environments, such as by modifying environmental pH or redox conditions in ways that could prevent a mirror cell that otherwise might colonize from doing so. Therefore, while many forms of interference-based competition between natural microbes and mirror bacteria would likely be impaired, some others could continue to function.

# 8.2 Mirror bacteria could colonize natural environments outside of multicellular hosts

No matter how predation-resistant they may be, organisms can only colonize new environments if capable of acquiring nutrients sufficient to replicate. Given that many generalist bacteria such as *E. coli* and *B. subtilis* can grow under diverse conditions and on achiral nutrients, there is no fundamental obstacle preventing mirrored versions of such organisms from surviving and replicating in the environment, though low nutrient concentrations or hostile abiotic conditions could create substantial barriers.

An introduced species can colonize an environment if it has a positive net growth rate. Population expansion will continue until one or more ecological feedbacks, such as decreases in nutrient concentrations or increases in predator populations, cause the growth rate to decline. If the

## Chapter 8: Environmental Survival and Spread

Mechanism	Examples	Target	Activity vs Mirror	References
Bacteriolysins				
Cleaves peptidoglycan	Enterolysin A	Peptidoglycan	No	Beukes et al., 2000
(Acedo et al., 2018)	Millericin B	Peptidoglycan	No	Khan et al., 2013
Pore-forming toxins				
Receptor-dependent pore formation	Colicin Ia	Colicin I receptor	No	Buchanan et al., 2007
in membranes (Dal Peraro & van der Goot, 2016)	Pyocin S5	TonB transporter	No	Behrens et al., 2020
Prophage				
Infection and lysis (XY. Li <i>et al.</i> , 2017)	Phage lambda	LamB protein	No	Wendling et al., 2021
Tailocin				
Membrane disruption through	R-type pyocins	Chiral receptors	No	Ghequire & De Mot, 2015;
binding to chiral receptors	F-type pyocins	Chiral receptors	No	Woudstra et al., 2024
Toxin delivery via CDI				
Toxin delivered via receptor-binding	Cdi_STEC031	Tsx	No	Ruhe et al., 2017
protein syringe (Pube et al. 2017)	EC536	OmpC	No	Ruhe et al., 2017
(Rune <i>et ut.</i> , 2017)	EC93	BamA	No	Aoki et al., 2010
Toxin delivery T6SS				
Toxin delivered by protein	ART-ARHs	FtsZ	No	Ting et al., 2018
syringe that does not require	RhsA	DNA	No	Koskiniemi et al., 2013
(Galán & Waksman, 2018)	Tle	Phospholipids	No	Russell et al., 2013
	Tne2	NAD(P)	No	J. Y. Tang et al., 2018
	Tre1	Peptidoglycan	No	Ting et al., 2018
	WapA	RNA	No	Koskiniemi et al., 2013
	Ssp6	Lipid membranes	Yes	Mariano et al., 2019
	VasX	Lipid membranes	Yes	Miyata et al., 2013

#### Table 8.3: Other examples of bacterial weapon systems

Bacteriolysins, pore-forming toxins, and tailocins act on chiral targets on the cell surface, while prophages and toxin-delivery via CDI requires chiral interactions at the cell surface in order to inject effectors into the cell. Only the delivery of lipid-membrane disrupting pore-forming proteins through a T6SS could plausibly retain efficacy against mirror bacteria.

introduced population is initially small or localized, it may suffer extinction through stochastic effects (Dallas *et al.*, 2021). However, so long as the expected growth rate is positive, then colonization will occur with some non-zero probability, and with repeated reintroductions successful colonization becomes increasingly likely.

The intrinsic resistance of mirror bacteria to many forms of predation could enable colonization of some environments despite their growth likely being slower than native bacteria. Even environments that cannot be stably colonized by mirror bacteria could act as temporary reservoirs, facilitating

spread between hosts or colonization of other environments, or provide time for adaptations that could allow persistence (Dennehy *et al.*, 2006).

## Mirror bacteria could obtain nutrients from the environment

All bacteria require nutrients to grow. Cyanobacteria and other photoautotrophs typically require only inorganic nutrients such as nitrogen, phosphorus, and sulfur, which are present in the environment as achiral ions such as ammonia, nitrate, phosphate, and sulfate. *Prochlorococcus, Synechococcus*, and *Synechocystis* strains can grow rapidly in fully synthetic media that contain no chiral compounds (L. R. Moore *et al.*, 2007; Yu *et al.*, 2015). Though some cyanobacteria opportunistically consume chiral organic compounds (Muñoz-Marín *et al.*, 2020) and some cyanobacterial strains have chiral vitamin requirements (A. A. Pérez *et al.*, 2016), mirror photoautotrophs could likely grow at similar rates to natural-chirality photoautotrophs.

Autotrophs generally do not require any chiral nutrients or chiral interactions with other living organisms to survive. Heterotrophs, by contrast, consume a wide range of low-molecular-weight organic compounds produced by other living organisms or by the breakdown of decaying biomass. Most of these organic compounds are chiral. Concentrations of these nutrients are typically low due to fierce microbial competition, but they suffice to support the slow growth of a diverse array of heterotrophic bacteria (Kirchman, 2016; Rousk & Bååth, 2011).

A mirror heterotroph would likely not be capable of utilizing most chiral nutrients in the environment. The catabolic capabilities of commonly studied heterotrophic bacteria are nevertheless diverse, and this breadth could allow their mirror counterparts to consume a range of common achiral and chiral compounds. For example, a mirrored version of *E. coli* K-12—the most commonly studied bacterial strain in microbiology and biotechnology—could grow on diverse achiral central metabolites, fatty acids, and fermentation products (see Table 1.1). It could also utilize a number of chiral organic compounds, including both enantiomers of alanine, cysteine, serine, xylose, lactate, malate, and tartrate (Chapter 1). These metabolites are often among the most abundant small organic compounds present in many environments. While a mirror *E. coli* would not be capable of catabolizing simple sugars such as D-glucose, it could be engineered to do so by encoding known metabolic pathways (see Table 1.3), which would expand the range of environments that could support mirror bacterial growth and substantially increase their population densities.

Environments are often highly heterogeneous, and many bacteria have evolved to detect and move toward nutrients through chemotaxis (Keegstra *et al.*, 2022). Chemotactic receptors are highly promiscuous, binding to a very wide array of biomolecules that might be catabolized. While many of the compounds recognized are chiral, it appears likely that mirrored versions of chemotactic bacteria could slowly move toward nutrients even absent modifications of their receptors. *E. coli*, for example, is attracted to diverse sugars and amino acids including both enantiomers of galactose, glucose, mannose, and xylose (Adler *et al.*, 1973), both enantiomers of aspartate and serine, and the achiral molecules ammonium, nitrate, fumarate, and succinate (Mesibov & Adler, 1972; Taylor *et al.*, 1979). While affinities for the enantiomers of common sugars are markedly lower, they do suffice for chemotaxis. It therefore appears likely that mirror *E. coli* cells would be able to recognize and find

chiral nutrients, and though they would not have an optimal foraging strategy, they could improve rapidly by natural selection (see Section 8.4). Other mirror bacteria would likely exhibit similar behaviors. Overall, mirror heterotrophs would presumably grow more slowly than native competitors, but some growth appears possible in many cases. And with little or no predation, even slow growth may allow mirror heterotrophs to colonize diverse environments.

The minimal carbon requirements for bacterial growth are very low. For instance, *E. coli*, a bacterium whose primary niche is within the nutrient-rich intestines of warm-blooded animals, can maintain a positive growth rate of ~0.2 hr<sup>-1</sup> with only 20  $\mu$ g/L glucose as its sole carbon source (Füchslin *et al.*, 2012; Shehata & Marr, 1971). While estimates of minimal required concentrations for other nutrients are not available, it is plausible that *E. coli* can slowly grow on a wide variety of substrates—including those in Table 1.1—at concentrations in the hundreds of micrograms per liter, and thus potentially proliferate due to their intrinsic resistance to predation.

In natural waters, many nutrients that could be accessible to mirror bacteria are typically present at concentrations of  $1-1000 \ \mu g/L$  (Thurman, 2014), including simple organic acids such as acetic, butyric, glycolic, lactic, and propionic acids; amino acids such as L-alanine and L-serine; and D-glucose, D-fructose, and other common sugars that could be available to a suitably engineered mirror bacterium. Though organic carbon might be too sparse to support mirror *E. coli* growth in seawater, this may not be true of all mirror heterotrophs. Moreover, growth in some rivers, eutrophic lakes, and marshes could be possible for many kinds of mirror heterotrophs.

Soils also contain diverse organic compounds. Sugars are the most important carbon and energy source for soil microorganisms (Gunina & Kuzyakov, 2015), while organic acids including acetic, fumaric, glycolic, lactic, and malic acid are also common due to exudation by plant roots (Kamilova *et al.*, 2006) and release by microbes (Adeleke *et al.*, 2017). Free amino acids are also significant contributors to soil organic matter (Hu *et al.*, 2018; Senwo & Tabatabai, 1998; Werdin-Pfisterer *et al.*, 2009), and p-amino acids are usually present, though at lower concentrations (Vranova *et al.*, 2012). While abiotic factors can limit the accessibility of these nutrients—particularly water activity (R. F. Harris, 1981) and redox potential (Husson, 2013)—some soils could plausibly support growth of a mirror *E. coli* strain or similar mirror heterotrophic bacteria, especially one engineered to metabolize p-glucose and other common sugars.

Bacterial growth in soils is typically limited by carbon (Demoling *et al.*, 2007), but phosphate and nitrogen concentrations can be limiting in natural waters (Paerl, 2018; Sterner, 2008). Marine phosphate concentrations, for instance, are typically much lower than 1  $\mu$ M (Paytan & McLaughlin, 2007), which is far too low to support *E. coli* growth (Shehata & Marr, 1971). In most environments, however, mirror heterotroph growth would probably not be prevented by a lack of inorganic nutrients.

Iron is an interesting case. The Fe<sup>3+</sup> ion is highly insoluble, and so in oxygenic environments iron is almost exclusively found complexed with organic molecules. A common strategy bacteria use to acquire iron is to produce siderophores, which are small iron-binding molecules released into the environment and subsequently reabsorbed. These siderophores are highly diverse and usually, though not always, chiral (Sandy & Butler, 2009). Mirror bacteria are therefore unlikely to be able to use

## Chapter 8: Environmental Survival and Spread

naturally occurring siderophores, and likewise natural-chirality bacteria could probably not use the siderophores produced by mirror bacteria. Because siderophores are released into the environment, they can act as "public goods" benefiting all strains able to utilize them, including "cheaters" that consume but do not produce siderophores (Kramer *et al.*, 2020). The extent to which producing and consuming reversed-chirality siderophores would prove harmful, or perhaps even beneficial, to mirror bacteria is unclear.

In the specific case of *E. coli*, it is known that mirror ferric enterobactin, the most commonly utilized siderophore, is imported into the cytoplasm but not cleaved, and therefore cannot be used to obtain iron (Abergel *et al.*, 2009). A mirror *E. coli* would produce mirror enterobactin, but would be unable to utilize natural-chirality enterobactin; thus, natural-chirality and mirror *E. coli* would likely interfere with each other's ability to acquire iron through enterobactin. *E. coli* can also obtain iron from iron(III)-citrate complexes and, in anoxic environments, by directly importing  $Fe^{2+}$  (Braun, 2003); these iron acquisition mechanisms would not be sensitive to chirality.

## Mirrored bacteria could be robust to abiotic stressors

In addition to nutrients, bacterial growth also requires tolerable environmental conditions. Abiotic stressors such as temperature, pH, oxidative stress, and osmotic pressure are achiral, and organismal responses to these stressors seldom involve chiral interactions with the external environment. A mirror bacterium would therefore have similar tolerances to abiotic stressors as its natural-chirality equivalent.

The first mirror bacteria, if created, might be fragile and functionally inflexible, akin to existing highly laboratory-adapted or engineered cells (Fredens *et al.*, 2019; Hutchison *et al.*, 2016). As described in Chapter 3, however, transforming these initial fragile cells into analogs of more robust bacterial species through gene addition would be feasible and required for most applications. A mirror *E. coli*, for instance, would be a natural target given the central role played by this model organism in both research and biotechnological applications. While the primary niche of *E. coli* is the mammalian gut, it can survive and replicate under diverse environmental conditions (Box 8.2).

We have focused on *E. coli* to this point because, as a model organism, a great deal is known about its resource utilization and environmental tolerances. Importantly, some bacterial species are more robust than *E. coli* in many environments. A mirror version of an environmental generalist such as *Pseudomonas aeruginosa* would likely exhibit better survival in many soils and natural waters. A mirror version of *Vibrio* would likely thrive in seawater if able to acquire nutrients, and a mirror cyanobacterium such as *Prochlorococcus* or *Synechococcus* would not even require organic carbon to grow.

While abiotic stressors are themselves achiral, some responses can involve chiral mechanisms that may be altered or impaired in mirror bacteria. For example, during osmotic stress *E. coli* preferentially imports L-proline as a compatible solute and cannot similarly use D-proline (Sasaki *et al.*, 2007). A mirror *E. coli* could not import L-proline, though it could still import the achiral betaine (Cayley *et al.*, 1992), and in the absence of either preferred osmolyte could simply import potassium ions.
## Box 8.2: *Escherichia coli* can survive and replicate under many environmental conditions

While primarily known as an intestinal commensal of warm-blooded animals, *E. coli* is capable of survival and growth under a wide range of conditions if nutrients are available. Many strains can sustain growth between ~8°C and as high as ~48°C (Baka *et al.*, 2013; Ross *et al.*, 2003), and they readily survive colder temperatures including even freezing (Sleight *et al.*, 2006). Strains can grow between pH of ~4 to ~9 (Baka *et al.*, 2013; Ross *et al.*, 2003), and many can survive more acidic conditions for hours (Gorden & Small, 1993). *E. coli* can even grow in seawater if nutrients are supplemented (Jannasch, 1968).

Such conditions are characteristic of many natural environments, implying that *E. coli* can survive for extended time periods in many soils and natural waters. Studies of pathogenic strain O157:H7, for instance, find that it can survive for many weeks or months in soil (Alegbeleye *et al.*, 2018; Mukherjee *et al.*, 2006; Ongeng *et al.*, 2011; Yao *et al.*, 2013; Zhang *et al.*, 2013), feces and manure (Fukushima *et al.*, 1999; Jiang *et al.*, 2002; Kudva *et al.*, 1998), and freshwaters (Ogden *et al.*, 2001; Wang & Doyle, 1998).

Predators appear to be a major cause of *E. coli* mortality in the environment, which is particularly important considering that a mirror *E. coli* strain would be resistant to many forms of predation. Studies that compare *E. coli* K-12 or O157:H7 growth in untreated vs. pre-sterilized soils, natural freshwaters, and seawater find greatly enhanced survival, and sometimes net growth, in the sterilized samples that lack phages, predators, and competitors (Bogosian *et al.*, 1996; W. L. Chao & Feng, 1990; Rozen & Belkin, 2001; Vital *et al.*, 2008). *E. coli* K-12 cells can persist for months in sterile soil and natural freshwaters (Bogosian *et al.*, 1996; Flint, 1987) and for over a week in sterile seawater (Byrd & Colwell, 1990; Gauthier & Le Rudulier, 1990; Sørensen, 1991). If seawater is enriched with nutrients, *E. coli* can even outcompete marine heterotrophs (Jannasch, 1968).

While neither K-12 nor O157:H7 are generally able to sustain population growth outside of their hosts, naturalized *E. coli* strains have been discovered in diverse environments (Ishii & Sadowsky, 2008; Jang *et al.*, 2017; Luo *et al.*, 2011), including in various soils (Byappanahalli & Fujioka, 2004; Ishii *et al.*, 2006; Solo-Gabriele *et al.*, 2000; Topp *et al.*, 2003), sands (Beversdorf *et al.*, 2007; R. L. Whitman & Nevers, 2003), sediments (Desmarais *et al.*, 2002), and natural freshwaters (Byappanahalli *et al.*, 2003; Carrill *et al.*, 1985). The fact that *E. coli* can become naturalized to such varied environments raises the possibility that a mirror *E. coli*, which would be resistant to most forms of predation, might also be capable of colonizing similar environments.

Biofilm production is another common response to abiotic stressors. Given that many bacteria can attach themselves to abiotic surfaces through often non-specific interactions (Beloin *et al.*, 2008; Berne *et al.*, 2018), the general ability for mirror bacteria to form biofilms on abiotic surfaces would not be impaired. However, the recognition of and attachment to biological surfaces may be impaired in some cases, as could participation in biofilms produced by natural bacteria. Reversed chirality

might also present some benefits: mirror bacteria would be resistant to many of the antagonistic interactions that are common in dense microbial communities (Rendueles & Ghigo, 2012; Stubbendieck & Straight, 2016), and they would excrete extracellular polymeric substances that would not easily be degraded by naturally occurring enzymes.

In addition to broad physical-chemical conditions such as temperature and osmotic pressure, bacterial growth can also be inhibited by the presence of specific toxic molecules. There is a possibility that toxicities of chiral molecules might pose a barrier to environmental colonization. As discussed in the Introduction, certain chiral compounds such as L-tryptophan or L-cysteine could be toxic to mirror bacteria, but the requisite concentrations, at least for mirror *E. coli*, are much greater than those of almost any organic molecule typically found in natural environments. Given the low concentrations of nutrients in the environment and the lack of evidence that any common metabolites would be particularly toxic to mirror bacteria, it appears unlikely that chiral incompatibilities would prevent growth of mirror bacteria in otherwise hospitable ecosystems.

In summary, a mirror *E. coli* or other generalist bacterium may be capable of abiotic survival in many environments, including most inhabited by humans, and could likely grow in the presence of sufficient nutrients. Some responses to abiotic stressors, such as biofilm production or osmolyte uptake, could be affected by their reversed chirality, but this appears unlikely to substantially reduce fitness in most environments. Moreover, any mirror bacteria that colonized a natural environment would experience evolutionary pressure to adapt to the local conditions, enabling them to grow to higher densities and expand their range. We return to this issue in Section 8.4.

### Colonization requires growth to outpace mortality

Even though bacteria can grow in a wide variety of environments, optimal growth is typically restricted to a narrow range of conditions: even bacterial "generalists" are typically comprised of many different strains that thrive in slightly different niches. It might therefore seem unlikely that a single mirror strain could compete successfully with natural bacteria across many different ecosystems. Unfortunately, even setting aside the evolution of the mirror strain (see Section 8.4), this intuition is incomplete, because mirror bacteria would likely evade predation to an unprecedented degree. If mirror bacteria are capable of net positive growth in an environment, then colonization is possible even if their replication rates are low.

While the first robust mirror bacteria are likely to be heterotrophs, such as *E. coli*, it is instructive to first consider the balance between growth and death of a mirror autotroph. *Prochlorococcus* strains are the most abundant cyanobacteria in the ocean, and are found in most tropical and subtropical waters (Flombaum *et al.*, 2013). Growth rates for *Prochlorococcus* are typically around 0.5 day<sup>-1</sup> (Kirchman, 2016), matched, on average, by an equal loss to predation by protists and phages. There are many different ecotypes and strains of *Prochlorococcus*, each of which grows optimally under a limited range of conditions (Biller *et al.*, 2015). The *Prochlorococcus* eMIT9312 strain, for instance, grows optimally at 25°C but can maintain growth between 16°C and 32°C (Johnson *et al.*, 2006). In natural environments, eMIT9312 will only dominate at higher temperatures, not because it cannot grow at 20°C, but because it is outcompeted by other *Prochlorococcus* strains that are better adapted

to cooler waters and grow faster than eMIT9312 at lower temperatures. At their equilibrium density, these cold-tolerant strains support populations of phages and protists that hold the cold-tolerant strains in check, but which would consume eMIT9312 faster than it can grow.<sup>16</sup> While eMIT9312 may be capable of replicating at 0.4 day<sup>-1</sup> in cooler waters, it will not persist if its mortality rate is 0.5 day<sup>-1</sup>.

A mirror eMIT9312 would grow at a similar rate to its natural-chirality counterpart, since it requires only sunlight and achiral inorganic nutrients.<sup>17</sup> But it would fall prey to phages and protists at substantially lower rates, greatly increasing the average lifespan of the mirror strain. While the mirror strain would still grow in 20°C water at a rate of 0.4 day<sup>-1</sup>, its death rate would be much lower than that, allowing it to invade areas such as cooler waters that its natural equivalent could not (Figure 8.2). The abiotic limits on growth would still hold—mirror *Prochloroccus* eMIT9312 could not colonize water below 16°C, although it might evolve to do so over time. It would also be unlikely to entirely outcompete native *Prochlorococcus* strains in the regions it colonized (see Section 8.5). Nevertheless, the mirror strain could probably colonize large fractions of the ocean even without evolution. Such a mirror bacterium would properly be considered an invasive species, and the consequences for marine ecosystems and even global biogeochemistry could be severe (Section 8.5).

There are several important differences between a mirror heterotroph and a mirror photoautotroph, but the core colonization dynamics are fundamentally similar. Natural heterotrophs exist in a quasi-steady state where slow grow rates—typically around 0.03–0.3 day<sup>-1</sup> in both soils (Rousk & Bååth, 2011) and marine environments (Kirchman, 2016)—are balanced by predation and other causes of mortality. A mirror heterotroph would likely have a substantially lower growth rate than natural heterotrophs due, in part, to a limited ability to acquire chiral nutrients. However, the mirror heterotroph would probably also have a much lower mortality rate because it could evade most forms of predation and microbial antagonism. Accessible carbon is often very limited in the environment, and in many cases this might preclude growth, and hence colonization, entirely. But if the mirror heterotroph can grow even slowly, and if its growth rate exceeds its mortality rate, then it would have net positive population growth. Thus, the mirror heterotroph could colonize the ecosystem, despite having slower growth rate than the native heterotrophic bacteria.

As described previously, there does not appear to be any fundamental obstacle to the growth of mirror heterotrophic bacteria in the environment. It is plausible that a mirror *E. coli*, for example, would be capable of growth in some soils and natural waters, particularly if able to catabolize D-glucose and other common sugars. Given that a mirror *E. coli* would likely be resistant to most forms of predation and microbial weapons, this advantage may allow it to colonize some environments even if it has other major disadvantages. Other mirror heterotrophs would differ in

<sup>&</sup>lt;sup>16</sup> Note that competition between *Prochlorococcus* strains is not perfect. Any given region of the ocean may contain hundreds of genetically distinct subpopulations (Kashtan *et al.*, 2014), though usually one ecotype is predominant (Farrant *et al.*, 2016). The mechanisms governing this coexistence are not well understood, though variable resistance to cyanophages is likely an important factor (Biller *et al.*, 2015; Rodriguez-Valera *et al.*, 2009).

<sup>&</sup>lt;sup>17</sup> *Prochlorococcus* can opportunistically consume chiral organic compounds (Muñoz-Marín *et al.*, 2020), so there could be some disadvantage, but in practice this would probably be small.



### Chapter 8: Environmental Survival and Spread

## Figure 8.2: Mirror phototrophic bacteria could evade predation, thereby colonizing large fractions of the ocean with severe consequences for marine ecosystems

**A.** Illustrative population growth and mortality rates of various *Prochlorococcus* strains show the potential consequences of mirror *Prochlorococcus* invasion. The precise rates would vary in real-world scenarios, but the values here are roughly in line with available data (Kirchman, 2016), though predation rates (including the split between protists and phages) are more uncertain. In this scenario, a native *Prochlorococcus* strain exhibits a growth rate (0.5 day<sup>-1</sup>) roughly matching predation rates by protists and phages (0.4 day<sup>-1</sup> and 0.1 day<sup>-1</sup>, respectively, for a total of 0.5 day<sup>-1</sup>), thereby resulting in a steady population. Non-native *Prochlorococcus* strains that are maladapted for this environment, here the eMIT9312 strain, will have slightly lower growth rates (0.4 day<sup>-1</sup>) but similar predation rates (0.5 day<sup>-1</sup>), and therefore cannot survive in conditions better suited to the native strain. A mirror eMIT9312 strain could exhibit similar growth rates to its natural-chirality, non-native counterpart but experience significantly decreased predation rates, speculatively shown here as a total of 0.1 day<sup>-1</sup>, thereby allowing invasive growth. **B.** Invasive growth could allow mirror *Prochlorococcus* strains to dominate niches normally occupied by multiple natural *Prochlorococcus* strains, thereby causing a decline in natural *Prochlorococcus* populations and the protists that feed on them. The resulting reduction in digestible biomass could drive many animal species that are higher in the food web to extinction (see Section 8.5).

their catabolic capabilities and robustness to abiotic stressors but could create similar, and in some cases even greater, concerns.

### Mirror bacteria capable of colonization could become highly abundant

Even at a modest net growth rate of 0.1 day<sup>-1</sup>, a single bacterium unchecked by nutrient limitations or predation could give rise to roughly  $5 \times 10^{31}$  descendants in just 2 years—more than the estimated total number of bacteria on the planet (Bar-On *et al.*, 2018; W. B. Whitman *et al.*, 1998). Of course, no population can grow forever. As populations grow, they deplete nutrient supplies and other

resources ("bottom-up control"), and they become prey to a growing population of predators and parasites ("top-down control"), eventually reducing the net growth rate and causing the population to stabilize or decline.

We have already discussed both nutrients and predation in the context of initial colonization—namely, could a small initial population of mirror bacteria sustain net positive growth? Here, the issue is slightly different—how would nutrient availability and predation pressure change as the mirror population grows? Reduced predation and parasitism is a common feature of many invasive species (the "enemy release hypothesis"; Keane & Crawley, 2002). Nevertheless, at least some predation and parasitism usually occurs, and the corresponding predator and parasite populations expand in response to growth of the invasive species.

As discussed in Section 8.1, a similar response to colonizing mirror bacteria is unlikely. Mirror bacteria would provide few if any nutrients to predators, precluding any commensurate expansion in predator or parasite populations as the mirror bacteria proliferate. Mirror bacterial populations would thus escape the normal mechanism of top-down control, allowing them to grow unchecked until limited by the available nutrients. Reduced predation would reduce the overall mortality rate and lower the minimum nutrient concentration necessary to sustain an equilibrium population, which in simple ecological models is what ultimately determines the winner in resource competition (Tilman, 1982). Mirror bacteria would essentially be accidental "defense-specialists" (Våge et al., 2014; Winter et al., 2010) with very low mortality due to predation, and this could allow them to reach population densities much larger than those typically achieved by individual bacterial strains in nature. The effect could be particularly strong in environments where bacterial mortality is dominated by predators. By contrast, mirror bacteria may struggle to colonize or outcompete native bacteria in environments where abiotic processes are the primary driver of bacterial mortality, as the advantages of reversed-chirality would be relatively small. Section 8.5 discusses potentially severe and irreversible harms that could result from mirror bacterial colonization and growth in natural ecosystems.

### Environmental reservoirs would facilitate the spread of infectious mirror bacteria

As discussed in Chapters 4, 6, and 7, mirror bacteria appear likely to enter animals and plants through a variety of mechanisms and may establish infections due to their anticipated immune evasion. Because mirror bacterial translocation through the animal gut appears plausible (see Sections 4.4 and 6.1–2), predators, parasites, and scavengers could be infected as well, then spread the mirror bacterium to other organisms through the food web. With many potential hosts and the presumed capacity to spread between them, mirror bacteria could potentially persist in some ecosystems even if they were largely incapable of surviving in the environment. The kinetics of infection and population expansion within each host species would be important for the rate of geographic spread (see Section 8.4), but any mirror bacterium capable of reaching a dense population within a host organism could plausibly spread to others that feed upon this host.

Infected animals would repeatedly introduce mirror bacteria into soils and natural waters. When the animal dies, it could release a relatively large population of mirror bacteria into the environment,

reducing the likelihood of local extinction. Any external environment susceptible to colonization by mirror bacteria might eventually be colonized, creating a permanent reservoir of mirror bacteria that could in turn infect new multicellular hosts.

Some environments could not be stably colonized by any given mirror bacterium. There might simply be too few nutrients available to support mirror bacterial growth, the abiotic conditions might not be suitable, or acute abiotic stresses might cause mortality that outpaces growth. Environments with low rates of bacterial mortality from predators or microbial warfare may be particularly unlikely to be colonized, as the ecological advantages of reverse chirality result from disruptions to these processes. Even so, such environments could act as temporary reservoirs for mirror bacteria. Persistent environmental phases are common for many bacterial commensals and pathogens. *E. coli* can persist for months in soils and natural waters (Box 8.2), which is unsurprising given its biphasic life cycle (Blount, 2015), and pathogens such as *Bacillus anthracis* (Hugh-Jones & Blackburn, 2009) and *Vibrio cholerae* (Faruque *et al.*, 1998) also persist for extended periods in the external environment between host infections.

Mirror bacteria could be made similarly robust to abiotic stressors while also avoiding predation and other biological processes that commonly remove pathogens from the environment. They could also potentially infect a greater range of animal and plant hosts. Mirror bacteria might therefore reach relatively high densities even in environments that they cannot colonize, as they would be repeatedly reintroduced by dying infected organisms. The presence of a large transient mirror bacterial population in soils and natural waters could in turn accelerate the evolution of adaptations that would facilitate true colonization of these environments (Kawecki, 2008).

# **8.3** Invasive mirror bacteria could rapidly disperse through the environment

A population of mobile organisms with a net positive growth rate can expand until it reaches the boundaries of the hospitable region. In the case of mirror bacteria, we expect that most movement would occur via multicellular host organisms or human transport. Mirror bacteria could plausibly infect and be carried by many different animal hosts, and they would presumably be as able as natural bacteria to hitchhike on vehicles. Therefore, the rates of spread of historical invasive species can provide an indication of how rapidly they would disperse.

Invasive species have been reported to spread at rates ranging from 1–100 km/yr for invasive mammals and insects (Shigesada & Kawasaki, 2001) to 50–10 000 km/yr for plant and animal pathogens (Kerr, 2012; McCallum *et al.*, 2003) (Table 8.4). Pandemic human viruses are the fastest of all: SARS-CoV-2 spread to every inhabited continent within four months of the first case (Ali *et al.*, 2020).

Invasive species	Origin	Spread rate (km/year)	References
Animals			
Earthworm, Lumbricus rubellus	Netherlands (20th C)	0.014	Marinissen & van den Bosch, 1992
Sea otter, Enhydra lutis	California (20th C)	1	Lubina & Levin, 1988
Muskrat, Ondatra zibethica	Czechoslovakia (1905)	13	Andow et al., 1990
Cereal leaf beetle, Oulema melanopus	Michigan (1958)	58	Andow et al., 1990
Rabbit, Oryctolagus cuniculus	Australia (19th C)	100	Alves et al., 2022
Pathogens			
Dutch elm disease, Ophiostoma ulmi	Britain (1960s)	50	Harwood et al., 2011
Rabies, Rabies lyssavirus	Europe (20th C)	50	Anderson et al., 1981
Mycoplasmal conjunctivitis, Mycoplasma gallisepticum	USA (1994)	510	Dhondt <i>et al.</i> , 1998
Coral bacterium, Sphingomonas	Florida (1995)	600	Richardson et al., 1998
Zika virus, ZKV	Brazil (2015)	2190	Lowe et al., 2018
Myxomatosis virus, Myxoma	Australia (1950)	7300	Kerr, 2012
Severe acute respiratory syndrome virus, SARS-CoV-2	World (2020)	48000	Ali <i>et al.</i> , 2020

### Chapter 8: Environmental Survival and Spread

Table 8.4: Rates of spread of animals and novel pathogens during historical invasions and epidemics

### Animal hosts could spread invasive mirror bacteria throughout the environment

The spread rate of an animal pathogen depends on how rapidly new hosts become infected and how far they travel before transmission. The simplest diffusion models of biological invasions predict a constant velocity of spread (Lewis *et al.*, 2016; J. D. Murray, 2003):

### $v = 2\sqrt{D\lambda}$

where  $\lambda$  is the growth rate of the infected population and *D* is the diffusion rate for the movement of infected individuals. These models give reasonable estimates for the spread rate of many invasive species and pathogens (Andow *et al.*, 1990; J. D. Murray *et al.*, 1986; Noble, 1974; van den Bosch *et al.*, 1992). In some cases, animal movements include small probabilities of long-distance movements ("jump dispersals"), which can cause much faster spread velocities (Clark *et al.*, 1998; Nathan, 2006; Neubert & Caswell, 2000; Pyšek & Hulme, 2005; Suarez *et al.*, 2001), a result readily derived from more general growth-dispersal models (Kot *et al.*, 1996; Lewis *et al.*, 2016).

A mirror bacterium would be a novel kind of pathogen, but if it were transmitted between animal hosts, it should spread approximately as fast as other diseases transmitted between similar hosts with similar growth rates. Since many different species may be susceptible to infection, the spread of an invasive mirror bacterium would likely be determined by those hosts that travel furthest and transmit most readily.

Flying insects are responsible for some of the fastest spreading terrestrial animal pathogens, including the myxoma and Zika viruses (Table 8.2). However, even mirror bacteria capable of infecting most insects are unlikely to spread as rapidly as these insect-vectored viruses, which tend to replicate faster within hosts than existing pathogenic bacteria. Moreover, mirror strains are unlikely to be efficiently transmitted via mosquito salivary glands, although a few bacteria (*Francisella tularensis*, *B. anthracis*, and *Rickettsia felis*) can be transmitted by mosquitoes via an unknown mechanism (Abdellahoum *et al.*, 2020).

Even mirror bacteria that are not very infectious or are limited in their host range could still spread reasonably quickly. Rabies has an incubation period of roughly 1 month (Toma & Andral, 1977) and is spread only by physical contact with an infected animal. As a result, the pathogen spread at a comparatively slow  $\sim$ 50 km/yr (Anderson *et al.*, 1981) through European foxes in the 20th century, which was still rapid enough to spread across the continent in a few decades. A mirror bacterium capable of infecting humans would likely also infect other terrestrial vertebrates, possibly including migratory birds, and rabies provides a conservative lower bound for the spread rate in most inhabited regions.

### Human activity could spread invasive mirror bacteria globally in just a few months

Modern humans travel much further and faster than any migratory animal; air travel allows an infected person to cross and move between continents within a day. Global travel explains why SARS-CoV-2 spread much faster than any animal virus.

Mirror bacteria could also be transported on contaminated items, within transported animals or plants or products made from them, or perhaps via hitchhiking insects (Gratz *et al.*, 2000; Tatem *et al.*, 2012). Container ships (Kaluza *et al.*, 2010; Lounibos, 2002; Ruiz *et al.*, 2000) transport goods across the planet and would also inadvertently spread mirror bacteria, as would trains, trucks, and cars traveling across continents (Ansong & Pickering, 2013; Gippet *et al.*, 2019). As noted in the previous section, such human-mediated "jump dispersals" can greatly increase the rate at which invasive species spread. It is difficult to envision any plausible way to avoid the inadvertent spread of an invasive mirror bacterium to most populated locations without ceasing nearly all global travel and cargo transport. Even if such extreme measures were feasible, they would not prevent more localized spread of mirror bacteria through natural animal populations.

Outside of animal hosts or human vehicles, mirror bacteria could still spread through terrestrial environments, albeit more slowly. Bacterial pathogens are often transported by water, including through surface runoff, by rivers and streams, and during flooding events (Boithias *et al.*, 2021; Bradford *et al.*, 2013; Cho *et al.*, 2016); mirror bacteria present in soil could be similarly transported. Bacterial plant pathogens are often dispersed through wind and rain, generally over short distances, but occasionally over several kilometers (West, 2014). Long-distance transport by wind is possible, though infrequent (Mayol *et al.*, 2017). Bacterial survival during atmospheric transport is challenging due to mortality from ultraviolet radiation and desiccation (Fernandez *et al.*, 2019; J. W. Tang, 2009), but some transport does occur, plausibly facilitated by co-transportation and shielding by dust or dead biomass (Griffin *et al.*, 2017; Yoo *et al.*, 2019). These and similar jump dispersals may be responsible

for the global mixing of prokaryotic species over many thousands of years (Louca, 2022), though anthropogenic transport likely dominates in the contemporary world.

### Marine mirror bacteria could spread through ocean currents

In marine environments the movement of water passively disperses organisms. Effective diffusion rates from turbulent mixing are  $\sim 1 \text{ km}^2/\text{day}$  over a few days (Okubo, 1971; Thorpe, 2012) and 5–50 km<sup>2</sup>/day over timescales greater than a couple of weeks (Zhurbas & Oh, 2004). Given a positive growth rate within the water column, an invasive mirror *Prochlorococcus* or other marine bacterium could spread through the ocean at velocities of a few kilometers each day due to turbulent diffusion.

A mirror *E. coli* may not be able to replicate in seawater due to the paucity of nutrients (particularly phosphorus). Nevertheless, *E. coli* can survive in seawater for several days (see Box 8.2). A mirror *E. coli* capable of infecting marine mammals or other marine organisms might therefore spread several kilometers each day by turbulence and currents when outside of hosts, much like the pathogenic coral bacterium *Sphingomonas* spread through Floridian reefs at 2 km/day on ocean currents in 1995 (Richardson *et al.*, 1998).

### 8.4 Invasive mirror bacteria could rapidly evolve and diversify

Natural bacteria evolve when errors in genome replication create mutations, some of which modify traits affecting survival or reproduction. Mirror bacteria would be no different. If invasive mirror bacteria were released into the environment, they could quickly adapt and diversify, further expanding the range of susceptible hosts and ecosystems.

This section focuses on the initial evolution of mirror bacterium subsequent to colonization of an environment. Existing organisms would also evolve in response to mirror bacteria. Evolution in response to mirror bacteria would, however, only occur once the mirror bacteria are sufficiently abundant to apply selective pressure on natural organisms. Such adaptations could have important consequences for the long-term ecological response to mirror bacteria but would have limited relevance during the initial expansion of an invasive mirror bacterium.

### Mirror bacterial evolution would depend on new mutations, which could arise quickly

A mirror bacterium that accidentally escapes containment would likely start with essentially no standing genetic variation. Mirror bacteria would be incapable of horizontal gene transfer with normal-chirality organisms, which would prevent the acquisition of novel genes and functions from other organisms. As a consequence, the genetic diversity that fuels adaptation by natural selection would have to be generated through *de novo* mutations arising after release. Even so, evolution that relies entirely on new mutations can fuel rapid and extensive adaptation, especially when bacteria encounter new environments.

Bacterial populations can replicate and expand quickly, with typical point mutation rates of  $10^{-10}$  to  $10^{-9}$  mutations per nucleotide per generation (Lynch *et al.*, 2016). The total number of bacteria in a typical human is about  $4 \times 10^{13}$  (Sender *et al.*, 2016). While the total number of mirror bacteria that

would arise during an infection is highly uncertain and would in part depend on the size of the infected animal, populations of 10<sup>10</sup> or more are certainly plausible. Assuming a genome size of several Mb (million bases), nearly every single possible point mutation could be generated and subjected to natural selection in the first large animal to become infected.

As the mirror bacterial population expands, lineages would experience different rates of adaptation due to variability in replication and mutation rates. Replication rates are highly variable because the environment is highly variable: doubling times within a host animal could be measured in hours or even minutes, whereas the strain might require many days to divide in soils and natural waters. Mutation rates can be dramatically increased by mutations to DNA replication and repair mechanisms. Such "mutator" phenotypes typically rise to high frequencies when bacteria encounter new selective conditions (Mao *et al.*, 1997; Oliver *et al.*, 2000; Sniegowski *et al.*, 1997; Taddei *et al.*, 1997; Tenaillon *et al.*, 2016; Wiser *et al.*, 2013), where they can accelerate the process of genetic adaptation. Mirror bacteria that are spreading invasively through new environments appear likely candidates to evolve hypermutability.

Lineages that reach higher population sizes would explore more mutations and thus may evolve more rapidly. These fast-adapting mirror lineages could experience hundreds or even thousands of generations each year, and with huge populations they would continuously generate new mutations. While deleterious changes would generally be eliminated by natural selection, these lineages could accumulate many neutral and, more consequentially, beneficial mutations each year (Tenaillon *et al.*, 2016).

### Natural selection could rapidly increase the fitness of an invasive mirror bacterium

Laboratory evolution experiments have demonstrated that bacterial populations can rapidly adapt to new conditions. Over the first 2 000 generations (about a year) of the long-term evolution experiment (LTEE), the fitness of the evolving *E. coli* populations increased by more than 20% relative to their ancestors in a benign environment without predators, competitors, or abiotic stresses. Similar studies in which *E. coli* were exposed to more substantial challenges have observed even faster and larger fitness improvements (Baym *et al.*, 2016; Tenaillon *et al.*, 2012). In the LTEE, the adaptation rate shows strongly diminishing returns, but fitness continues to increase. After 50 000 generations (about 22 years), relative fitness had increased by about 75% in those populations that evolved hypermutability, which accumulated 500–2 500 mutations, and by about 60% for those that retained the lower ancestral mutation rate, which had acquired an average of 60–100 mutations (Tenaillon *et al.*, 2016; Wiser *et al.*, 2013).

Other experiments have demonstrated that a wide variety of specific adaptations can evolve quickly in *E. coli*, including improved tolerance of acidity (Harden *et al.*, 2015), oxidative stress (Rodríguez-Rojas *et al.*, 2020), osmotic stress (Guo *et al.*, 2017), ionizing radiation (D. R. Harris *et al.*, 2009), starvation (Finkel & Kolter, 1999), high temperatures (Rudolph *et al.*, 2010), freeze-thaw cycles (Sleight *et al.*, 2008), and antibiotics (Baym *et al.*, 2016). *E. coli* can rapidly evolve to increase growth rates on non-preferred or even non-native carbon sources (Blount *et al.*, 2012; Fong *et al.*, 2003; Herring *et al.*, 2006; Izutsu & Lenski, 2022; Lee & Palsson, 2010), and they can quickly

overcome low growth rates from metabolic imbalances caused by genome reduction (Choe *et al.*, 2019; Hitomi *et al.*, 2024; Moger-Reischer *et al.*, 2023). Mutagenesis experiments show that mutants of *E. coli* K-12 can utilize numerous non-native nutrient sources (see Table 1.1–3), including the right-handed forms (i.e., reverse chirality) of eight of the 20 canonical amino acids (Celis, 1977; Kuhn & Somerville, 1971); a further seven of the 20 can be used natively by at least some *E. coli* strains (see Table 1.2).

An invasive mirror *E. coli* or similar mirror generalist would experience strong selection to increase its growth and survival, both within hosts and in the external environment. Mirror populations in a single large animal or patch of soil could greatly exceed the numbers typically used in laboratory evolution experiments, providing substantial opportunities for adaptive evolution. Spatiotemporal gradients and other environmental heterogeneities could further accelerate evolution by selecting for different capabilities, which could promote sustained adaptation by preventing populations from becoming trapped at local fitness peaks (Baym *et al.*, 2016; Kashtan *et al.*, 2007). Potential targets of adaptation in mirror bacteria would include improved chemotaxis toward accessible nutrients (Adler *et al.*, 1973), faster transport of nutrients across the cell envelope, improved regulation of catabolic pathways for switching between available nutrients, enhanced translocation into and infection of multicellular hosts, tolerance to acidity and reactive oxygen to survive in the phagosomes of protists and phagocytes of hosts, and tolerance to abiotic stressors such as temperature extremes and low water availability.

Evolution is typically slower to discover novel capabilities that are unrelated to existing genes. Despite their large populations, invasive mirror bacteria might take a long time to acquire some highly beneficial functions due to their initially low genetic diversity and lack of horizontal gene transfer with natural organisms. Enzymes able to catalyze novel reactions, such as a mirror nitrogenase in *E. coli*, might require millions of years to evolve, if ever, as no genes with related functions would be present in the initial mirror strain (unless introduced through human intervention). A key question concerns how long it would take for p-glucose catabolism to evolve if it was not engineered into the initial strain. A mirror *E. coli* would likely need at least four new enzymatic activities, as well as a transporter. However, the basic catalytic mechanisms would already be present in existing metabolic pathways targeting different sugar substrates. Moreover, the mirror bacterium would gain energy and fitness upon altering the substrate specificity of an existing enzyme to catalyze each of the first two steps (Shimizu *et al.*, 2012). The hydrolysis of di- and oligo-saccharides such as p-sucrose and p-trehalose could be important within plants and insects, though these capabilities might be able to evolve from mutations to existing glycoside hydrolases.

### An invasive mirror bacterium could diversify and fill available niches

If they spread through the environment, lineages of mirror bacteria would be subject to diverse selective pressures. Mutations that are beneficial under some conditions are often deleterious under others (Cooper & Lenski, 2000), favoring diversification into increasingly specialized strains. Even in extremely simple laboratory conditions, initially homogeneous populations of bacteria often evolve stable ecotypes that specialize on different physical, chemical, and biological aspects of their environment (Good *et al.*, 2017; Rainey & Travisano, 1998). Geographic distance would also cause

mirror bacterial populations to diverge over time, even without differences in selection pressures. Increasingly fit mirror bacteria could expand at the expense of their progenitors, although this displacement would usually be slower than the initial invasion of mirror bacteria.

As previously mentioned, horizontal gene transfer between natural-chirality bacteria and mirror bacteria would be impossible. Horizontal gene transfer between mirror bacteria would be theoretically possible, but in the absence of mirror phages or conjugative plasmids it would require the uptake of and transformation by L-DNA released from mirror cells. Also, during the initial evolution of invasive mirror bacteria, the importance of horizontal gene transfer for evolution would be limited by their lack of genetic diversity. However, as mirror lineages diverged, horizontal gene transfer between them could become a more important driver of their evolution, especially if mirror phages or plasmids were present.

In summary, an invasive mirror bacterium capable of spreading in the environment would likely produce a very large population, allowing it to generate diverse mutations upon which selection would act to increase both its local fitness and its ability to colonize previously inaccessible environments. However, novel enzymes with functions unrelated to those already present would be slow or unlikely to evolve. Even if a released mirror bacterium was initially restricted to a narrow set of environmental conditions, it could evolve rapidly to expand its range. The potential for mirror bacteria to evolve and diversify underscores the biosafety and biosecurity challenges raised by their creation (see Chapter 3).

### 8.5 Invasive mirror bacteria could cause irreversible ecological harm

Predicting whether a mirror bacterium could colonize any given environment is challenging, and forecasting the long-term consequences of a successful invasion is even more so. Invasive mirror bacteria that become common in the environment might well eliminate some of those animal and plant species that are susceptible to infection. Less visible but equally important impacts to microbial ecosystems could lead to major and irreversible changes to ecological and geochemical processes.

We emphasize severe outcomes because they are the most consequential and cannot be excluded based on what we know from the limited available data. Given the multipronged risks to human health (Chapter 4), animals (Chapter 6), plants (Chapter 7), and natural ecosystems (this chapter), we think it is important to emphasize the breadth and depth of the problems that could be caused by mirror bacteria if they invade and become established in nature.

### Invasive mirror bacteria could drive susceptible plant and animal species to extinction

Invasive pathogens are a major cause of ecological harm, and in some cases have driven species to extinction. The amphibian-killing fungus *Batrachochytrium dendrobatidis* spread from East Asia sometime in the early 20th century (O'Hanlon *et al.*, 2018), leading to global declines of amphibian populations and contributing to at least 90 presumed extinctions (Scheele *et al.*, 2019). In less than a decade, the invasive white-nose fungus has killed over 90% of several North American bat species (Cheng *et al.*, 2021) and may drive multiple species to extinction. The oomycete *Phytophthora cinnamomi* infects ~5 000 plant species, causing a form of root rot that devastates native ecosystems

in Australia and Europe and threatens hundreds of susceptible plants with extinction (Invasive Species Council, 2023). The invasive *Ceratocystis lukuohia* fungus is rapidly infecting and killing trees of the keystone 'ōhi'a tree (Barnes *et al.*, 2018), which accounts for ~80% of the biomass in Hawaiian forests and 50% of the surface area of all woody plants on the archipelago (Fortini *et al.*, 2019). Also in Hawaii, the introductions of avian malaria and bird pox were likely responsible for the extinction of a substantial proportion of the native avifauna in the late 19th century (van Riper *et al.*, 1986; Warner, 1968).

An important feature of the aforementioned pathogens is their ability to infect many host species. While there are numerous examples of infectious diseases causing large reductions in host populations, diseases do not usually cause the extinction of their host species (Smith *et al.*, 2006). Most pathogens require sufficient host density to survive and spread, and so pathogen frequencies generally decline once the host population falls below some threshold density (De Castro & Bolker, 2005; McCallum, 2012). For pathogens that can exploit multiple hosts or which replicate in the environment, this logic no longer applies. Persistence in the more resilient hosts or external reservoirs allows the pathogen to be repeatedly reintroduced into the more susceptible hosts.

An invasive mirror bacterium could plausibly infect a diverse set of host species (see Chapters 6 and 7), and might be able to colonize some external environments as well (this chapter). This breadth could make invasive mirror bacteria particularly ecologically devastating. Given the broad immunological deficiencies discussed in Chapters 4, 6, and 7—consequences of the complete naivety of all natural organisms to mirror bacteria—it is plausible that many susceptible populations of plants and animals would contain few or no individuals with intrinsic resistance to infection. Such populations could be rapidly driven extinct, with little opportunity for the evolution of a resistant phenotype.

Species in which preexisting resistance to mirror bacteria was fortuitously more common would likely see rapid evolution toward increased resistance to infection. While rapid evolution is observed in response to some pathogen outbreaks (Bonneaud *et al.*, 2018; Di Giallonardo & Holmes, 2015; Epstein *et al.*, 2016), there are important differences with the case of mirror bacteria. In particular, as previously noted, specialized pathogens typically cannot drive their host species to extinction, which allows time for the gradual evolution of traits that increase the host's resistance. For invasive mirror bacteria (or other diseases with environmental reservoirs or multiple hosts), species survival requires that some individuals are able to survive and reproduce in the face of repeated exposure from environmental reservoirs or spillover infections from other susceptible species. In vertebrates, this challenge may be further exacerbated by the likely impairment of adaptive immunity and immune memory formation (see Chapters 4 and 6); unlike most infectious diseases, an individual's fortuitous survival of a mirror bacterial infection may not provide robust protection against future infections.

Widespread infections leading to population crashes or extinction of susceptible animal and plant populations may be the most substantial cause of ecological harm from invasive mirror bacteria. We should emphasize that the degree to which different animal and plant species are susceptible to mirror bacterial infection is uncertain, and would likely depend on the specific mirror bacterium in question. Nevertheless, a mirror bacterium capable of infecting multiple species could be ecologically devastating to an extent that is potentially much greater than existing invasive pathogens.

# Disruptions to terrestrial ecosystems could eliminate key habitats and impact geochemical cycles

While the extinction of any species is unfortunate, the loss of many species simultaneously could cause severe disruptions to critically important ecosystems and the functions they provide. This is particularly true if certain "keystone" species or guilds (i.e., groups of species that perform similar functions) are susceptible to mirror bacterial infection. For example, while the invasive *C. lukuohia* fungus is unlikely to drive the 'ōhi'a to extinction (Luiz *et al.*, 2021), 'ōhi'a are so dominant in Hawaiian forests that the loss of most mature trees would eliminate the primary habitat of most other native plant and animal species (Fortini *et al.*, 2019). Coral reefs, mangrove forests, coastal salt marshes, and wetlands are likewise reliant on certain key species to stabilize the physical environment and provide habitats for other species. If these key species were susceptible to mirror bacterial infection, the result would not just be their potential loss, but also severe damage to the entire ecosystem.

While most forest ecosystems are more diverse and robust than Hawaiian ' $\bar{o}h$ i'a forests, a few major genera of trees dominate many global forests, particularly beech, eucalypt, pine, oak, and spruce trees. If members of these genera were susceptible to mirror bacterial infection, the results could severely damage local forest ecosystems. Forests are also important stores of carbon, containing about 450 GtC (Erb *et al.*, 2018), and so the death and decomposition of even a small fraction of these trees could greatly exceed the 11 GtC/yr emitted by human activities (Friedlingstein *et al.*, 2023). If only a moderate fraction of trees prove susceptible, the long-term effects on geochemical cycles might be moderated as resistant trees grow to replace susceptible ones. On the other hand, if mirror bacteria could infect most tree species and prevent their regrowth, the climate impacts could be dramatic, with atmospheric CO<sub>2</sub> potentially increasing on the order of 100 ppm over the course of a few years.

While not as visible as plants and animals, microbial communities within soils are also vital for many ecosystem functions. Soil microbiota are critical for the transformation and recycling of organic matter (Hoffland *et al.*, 2020), nitrogen fixation (Chen *et al.*, 2003), and making micronutrients available to plants (Singh *et al.*, 2022). Physical, chemical, and biological disruptions to soil ecosystems are known to degrade soil quality and thereby reduce agricultural or ecosystem productivity (Eswaran *et al.*, 2001), and can trigger desertification (D'Odorico *et al.*, 2013). Mirror bacteria could directly impact soil communities by competing with other microbes for organic matter (to the extent it is accessible to mirror metabolism) and key inorganic nutrients like phosphorus and nitrogen, as well as by potentially harming bacterial predators and other soil fauna. Because these types of disruptions would be unprecedented, predicting the consequences is exceptionally difficult.

Global soils contain around 2500 GtC (Jackson *et al.*, 2017), much greater than the 900 GtC currently present in the atmosphere and the 450 GtC in vegetation. Thus, the loss of substantial primary production combined with the potential release of stored carbon from degraded soils could

theoretically lead to a further increase in global temperatures, although the uncertainty surrounding this outcome is considerably greater than for direct impacts.

# Ecological disruptions would lead to strong selective pressures on natural organisms, but some responses could be severely constrained

If mirror bacteria were to become abundant, they would exert strong selection pressure on other species. As discussed earlier, organisms susceptible to mirror bacterial infection would experience intense selection for resistance. If most individuals in a population were susceptible to infection, a population crash could ensue, limiting the population's genetic diversity and potentially harming the associated ecosystem. If extinction does not occur, then rapid evolution driven by selection among the surviving variants is a likely consequence of the population crash.

More generally, other organisms could not preemptively adapt to the presence of mirror bacteria elsewhere in the world, or even to early colonists at low abundance. Strong selection pressure would require relatively abundant mirror bacteria in the local environment. Evolution is therefore unlikely to prevent either the initial spread of mirror bacteria or mitigate any irreversible ecological harms they might cause during the first wave of colonization.

Given enough time and selection pressure, other organisms could likely evolve to prey upon or otherwise harm the mirror bacteria. Natural bacteria might have a particularly rapid evolutionary response, given their own extremely large populations and high replication rates, as well as some preexisting weapons that may be partially effective against mirror bacteria and might be refined to increase their efficacy. It also seems likely that natural bacteria would evolve to consume some mirror bacterial detritus, eventually including the mirror-chirality macromolecules that initially could not be utilized; if so, some might evolve to actively kill and consume mirror bacteria. Horizontal gene transfer might also spread such antimirror adaptations between natural strains of bacteria, albeit only once populations of mirror organisms have become abundant.

The evolution of protists and animals that prey on mirror bacteria appears more challenging. Protists and animals lack the enzymes required to degrade mirror proteins, sugars, nucleic acids, and lipids. They likely have limited or no ability to catabolize most mirror metabolites (Friedman & Levin, 2012), and some p-amino acids are toxic to many organisms (Forsum *et al.*, 2008; Friedman & Levin, 2012; Yow *et al.*, 2006; see also Box 1.2). These deficiencies could not be remedied through a handful of mutations, but would require the much slower evolution of novel proteins and catabolic pathways.

To give an illustrative example, *C. elegans* relies on many dozens of distinct lysozymes, glycosidases, proteases, phospholipases, nucleases, and other lytic enzymes to digest the macromolecules present in their bacterial prey (McGhee *et al.*, 2007; Yilmaz & Walhout, 2016). Digesting even a simple macromolecule like bacterial-derived glycogen requires intestinal amylase and  $\alpha$ -glucosidase to decompose the glycogen into first maltose and then glucose, followed by transport into intestinal cells via the FGT-1 transporter (Kitaoka *et al.*, 2013), where glucose is catabolized into phosphorylated trioses through several additional enzymes. Evolving a similar catabolic pathway for mirror glycogen would almost certainly require a similar or greater number of

novel enzymes, and it seems questionable whether such adaptations could arise in nematodes even over millions of years.

Absent these adaptations, the consumption of mirror bacteria by many natural organisms would likely be futile or even harmful. Thus, there would be no selection pressure for increased consumption of the mirror bacteria, and indeed there might instead be selection to avoid them. It is certainly possible, perhaps even likely, that some protists and animals would eventually evolve the ability to consume mirror bacteria, but it is impossible to predict how long this would take or which taxa might acquire this ability.

It is unclear whether phages that target mirror bacteria would arise naturally. As discussed earlier, natural phages cannot infect mirror bacteria, as there is no plausible mechanism by which a natural-chirality viral genome could be replicated within a mirror bacterium. There is also no plausible natural mechanism by which a natural-chirality viral genome could be transcribed into mirror nucleic acids and translated into mirror proteins. It is therefore extremely unlikely that existing natural phages would evolve to infect mirror bacteria. However, if the mirror bacterium harbored a prophage in its genome, the prophage might be activated and evolve to become an effective predator. Also, the ultimate origins of viruses remain uncertain (Krupovic *et al.*, 2019), and so perhaps mirror phages could eventually evolve *de novo* from mirror bacteria.

While natural enemies to mirror bacteria might eventually arise and help limit mirror bacterial populations, they could not plausibly drive all mirror bacteria to extinction. Given the fundamental biochemical differences between mirror bacteria and natural-chirality organisms, predation on mirror bacteria would require specialized adaptations that were advantageous only if mirror bacteria were present in the environment. Once they become established in the environment, the complete elimination of mirror bacteria therefore appears unlikely, even over evolutionary timescales, and the final "equilibrium" state that is achieved may be very different from present-day ecosystems.

# Mirror cyanobacteria could profoundly alter marine ecosystems and the global carbon cycle

Photoautotrophs fix carbon directly from sunlight and carbon dioxide, and often do not require any chiral nutrients in order to grow. The production of organic matter by photoautotrophs in turn lays the foundation for almost all other life on the planet. This makes the prospect of a mirror photoautotroph particularly concerning from an ecological standpoint. For concreteness, this section focuses on the consequences of a mirror *Prochlorococcus*, though the effects of a mirror marine *Synechococcus* (a paraphyletic genus which contains *Prochlorococcus* as a subclade (Komárek *et al.*, 2020) could be similar. It appears less likely that a mirror phototroph would be created than a mirror heterotroph, given the potential biotechnological applications of a mirror heterotroph. Nonetheless, it is important to consider the possibility of a mirror prototroph, perhaps especially given interest in engineering solutions to climate change.

As discussed in Section 8.2, a mirror *Prochlorococcus* could plausibly replace native *Prochlorococcus* strains across much of the world's oceans. Natural *Prochlorococcus* growth rates under current marine conditions are typically  $\sim 0.5 \text{ day}^{-1}$  (Kirchman, 2016), and growth is typically

limited by nitrogen (Paerl, 2018), although iron (J. K. Moore *et al.*, 2001) and phosphate (Martiny *et al.*, 2019) can also be limiting. Because the protists and phages responsible for almost all *Prochlorococcus* mortality (Beckett *et al.*, 2024) would not efficiently target a mirror version, the mirror bacterial population would likely expand until the local concentrations of nitrogen, phosphorus, and/or iron were drawn down to levels too low to support further growth.

As nutrient concentrations decrease, so would the populations of existing phototrophs and the protists and phages that prey on them. While it is impossible to predict precisely how marine ecosystems would be affected, the consequences would likely be severe. It is extremely unlikely that natural-chirality phototrophs would be driven extinct; with the decline of many predators, some native *Prochlorococcus* strains and other phototrophs could still coexist with the mirror *Prochlorococcus* due to their greater genetic diversity and adaptation to particular niches. Nevertheless, it appears likely that a new equilibrium would be reached in which mirror *Prochlorococcus* dominated a nutrient-depleted ocean. The greatly reduced population of natural phototrophs would support a much less diverse marine food chain, as the reduction in digestible biomass would likely drive many animal species to extinction (see Figure 8.2).

The effects of mirror *Prochlorococcus* on the carbon cycle would be even more complicated. It has previously been speculated that mirror cyanobacteria would sequester vast quantities of  $CO_2$  from the atmosphere, causing terrestrial photosynthesis to fail over a few centuries and triggering an ice age (Bohannon, 2010). This scenario does not appear particularly likely to us because the production of mirror *Prochlorococcus* biomass would be limited by the availability of other nutrients; however, it cannot be ruled out if there were substantial carbon fixation beyond that needed to produce biomass. The next few paragraphs explore these complex dynamics in more detail.

In the tropical and subtropical waters where *Prochlorococcus* thrives, the limiting nutrient is often nitrogen. As the mirror *Prochlorococcus* population expanded, it would sequester nitrogen within its own biomass. Transferring nitrogen from natural-chirality biomass to mirror biomass would not greatly change the total fixed carbon in the surface oceans, as the C:N ratio of *Prochlorococcus* of about 7:1 is broadly comparable to other living organisms (Bertilsson *et al.*, 2003). Transferring nitrogen from dissolved organic matter to mirror biomass would meanwhile decrease the total fixed carbon, as the C:N of dissolved organic matter is typically closer to 15:1 (Sipler & Bronk, 2015). Transferring nitrogen from nitrate and other inorganic sources to mirror *Prochlorococcus* would also increase fixed carbon stockpiles. But because most nitrogen in the surface ocean is already sequestered in biomass and dissolved organic matter (Sipler & Bronk, 2015), it is not clear that the takeover of surface oceans by mirror *Prochlorococcus* would greatly increase the total organic matter in the surface oceans. Similar constraints would arise in regions where iron or phosphorus is limiting.

Not all carbon fixed by photoautotrophs goes into biomass. Since they are typically limited by nitrogen and managing photosynthetic fluxes is challenging, *Prochlorococcus* cells have been observed to release anywhere between 2% and 90% of the carbon that they fix back into the environment (Bertilsson *et al.*, 2005; López-Sandoval *et al.*, 2013; Roth-Rosenberg *et al.*, 2021). Some of this fixed carbon is released as glycolate, acetate, and other small metabolites, but much of it takes the form of complex and poorly characterized high-molecular-weight compounds (Iuculano *et al.*, 2017; Kujawinski, 2011). The mirror versions of these large compounds could probably not be

degraded by existing heterotrophs. Since these compounds lack nitrogen and could not support growth in a nitrogen-limited environment, there may not be substantial selection to acquire the capability to consume these compounds. Absent consumption, this leakage could therefore result in much greater concentrations of organic carbon in the surface ocean.

Predicting how much carbon overproduction would occur in the years and decades following the release of a mirror *Prochlorococcus* is challenging. On the one hand, the standing biomass of mirror *Prochlorococcus* could be large. A recent estimate suggests that the current standing biomass of planktonic autotrophs is roughly 0.6 GtC, of which perhaps a quarter is *Prochlorococcus*; the total biomass of marine organisms is closer to 6 GtC, or about ten-fold greater (Bar-On & Milo, 2019). While these estimates are very crude, they suggest that the standing biomass of a mirror *Prochlorococcus* could potentially even exceed that of all present-day phytoplankton. It is therefore at least possible that carbon overproduction by mirror *Prochlorococcus* could be comparable to or even greater than current marine net primary production (NPP), which is about 50 GtC/yr (Field *et al.*, 1998). On the other hand, given the wide uncertainty in the extent of carbon overproduction and the very slow biomass accumulation of the mirror *Prochlorococcus* population once nutrients are depleted, it is plausible that mirror *Prochlorococcus* would engage in very little carbon overproduction.

If carbon overproduction proved to be very large, then the climate consequences could be severe even over a few decades. The atmosphere contains about 870 GtC of carbon-up from the pre-industrial average of 571 GtC (IPCC, 2021)-and the removal of tens of GtC each year would lower atmospheric CO<sub>2</sub> to below pre-industrial levels in a matter of years or decades, and to levels below that required for terrestrial photosynthesis over hundreds of years, as speculated elsewhere (Bohannon, 2010). If carbon overproduction were instead minimal, then the initial climate effects would probably be small, though other ecological consequences would still be severe. Over longer timescales the climate would still likely be greatly altered, because biological processes play a crucial role in determining carbon storage in the deep ocean (Ridgwell & Arndt, 2015) and these processes could be profoundly disrupted by mirror Prochlorococcus. In particular, Prochlorococcus and heterotrophic bacteria are close to neutrally buoyant; the extent to which a mirror Prochlorococcus-dominated ecosystem would sequester carbon through the usual biological pump (Siegel et al., 2023) is therefore unclear. Careful modeling efforts might better discern the range of plausible climate outcomes over both decades and millenia. However, given the uncertainties in the extent of carbon overproduction and the longer-term ecological and evolutionary responses to mirror Prochlorococcus, substantial uncertainty will doubtless remain.

# 8.6 Countermeasures to invasive mirror bacteria might lessen but would not halt the ecological damage

Strategies to minimize the environmental harms caused by invasive mirror bacteria can be divided into two main categories: those that directly reduce populations of mirror bacteria, and those that reduce the adverse effects on other organisms. Neither approach appears likely to prevent widespread, irreversible environmental damage.

### Biological methods could reduce but not eliminate mirror bacterial populations

Given the immense size of natural ecosystems, the only realistic methods for reducing mirror bacterial populations at a global scale are self-replicating biological agents. Human manufacture of mirror antibiotic compounds might protect or remediate small areas, but even switching the entire global production of antibiotics to antimirror compounds may not provide all humans with prophylactic treatment. As noted in Chapter 7, protecting crops through antimirror compounds appears infeasible; scaling production to the level required to protect entire ecosystems would be impossible. Other conventional means for preventing the spread of agricultural pathogens could not be performed at the required scale (Section 7.4), and could entail significant environmental damage if applied to natural environments (rather than agricultural fields).

By contrast, biological agents that prey on mirror bacteria could self-replicate and multiply in the environment with no need for manufacture at scale. Mirror phages are the simplest and most promising such biological countermeasure. A mirror phage could function as an effective predator (Weitz, 2016), substantially reducing the local abundance of its prey. Like mirror bacteria themselves, mirror phages could be spread by the movement of water, wind, and animals; spread through marine environments would be particularly rapid due to passive movement by ocean currents and turbulent mixing (see Section 8.3). The extent to which a mirror phages could spread effectively through terrestrial environments is less clear, though mirror phages could be released in many locations to assist their spread.

Despite their ability to replicate, mirror phages suffer from inherent limitations. Sustained phage replication would require the presence of a sufficiently high density of mirror bacteria in the environment. As a result, mirror phages could never prevent the initial invasion and spread of mirror bacteria into a new environment, a fact more generally true of any attempt to use specialist predators as biological controls (Lewis *et al.*, 2016; Owen & Lewis, 2001). For similar reasons, mirror phages would not eliminate mirror bacteria from the environment, as the mirror phages require sufficient densities of mirror bacteria in order to replicate. At best, mirror phages would help control the equilibrium concentrations of mirror bacteria in the environment, thereby reducing, but not eliminating, infections of susceptible multicellular organisms. Such limitations are familiar from the analogous use of biological controls in the management of invasive species, where the intended aim is to maintain the density of the invasive species at a low homeostatic equilibrium (Murphy & Evans, 2009).

A further difficulty with mirror phages is that they would exert strong selection pressure on the mirror bacteria to evolve phage resistance. Inactivating the surface receptor used by a phage to enter the cell, or reducing its production, can render that cell entirely or partially resistant to that particular phage (Bohannan & Lenski, 2000; Meyer *et al.*, 2012). These receptors generally have important functions for the bacteria; many are involved in acquiring nutrients from the environment. In nature, resisting predation from any given phage may be less important than retaining the function of the receptor, although resistance can sometimes evolve by altering a receptor's structure without destroying its function. Moreover, an invasive mirror bacterium could plausibly lose multiple receptors and still thrive. Though the phages themselves may evolve to overcome host resistance, and humans could synthesize mirror versions of phage known to prey upon the natural equivalent of the

mirror strain, it is likely that the coevolving mirror bacteria would diverge into multiple lineages, each preyed upon by specialized phages (L. Chao *et al.*, 1977; Weitz *et al.*, 2005; Williams, 2013). A single mirror phage could effectively control the population of any given mirror bacterial strain. However, as the number of strains expands, the coevolving phage lineages may become less effective at controlling the total population of mirror bacteria (Thingstad, 2000; Waterbury & Valois, 1993).

As noted in Section 5.3, using mirror phages to control mirror bacteria would present some additional risks: they would provide a mechanism for horizontal gene transfer between mirror bacterial strains and introduce novel genes and proteins into the mirror bacterial population. If an invasive mirror bacterium were already causing major ecological damage, these considerations would likely be secondary to the need to reduce the mirror bacteria and their harmful effects by whatever means available. Importantly, mirror phage countermeasures would not need to be developed in advance, as they could be swiftly generated once the target species of mirror bacterium was identified.

Other biological means of controlling mirror bacterial populations appear even more challenging while suffering from similar limitations. Protists could, in principle, be reengineered to recognize mirror bacteria, equipped with enzymes capable of degrading mirror macromolecules, and provided with metabolic pathways to utilize mirror metabolites. Such engineering is, however, beyond current abilities. Because protist species have limited ecological ranges (Grossmann *et al.*, 2016), many different protist species may need to be reengineered in order to provide top-down biological control across the relevant ecosystems, which would further increase the challenge. The primary advantage of engineering protists to consume mirror bacteria is that, over the very long run, such protists might be more effective than mirror phages at controlling the total mirror bacterial population; protist predation tends to be less strain-specific and the evolution of resistance less straightforward. Engineering heterotrophic bacteria and/or fungi to produce antimirror compounds would likely be easier than reengineering protists, but also probably less effective.

In summary, if an invasive mirror bacterium were to be released into the environment, there appears to be no practical means to either prevent its spread or eliminate it from the environment. Predators including mirror phages and, more speculatively, engineered protists could restore some degree of top-down control on mirror bacteria and so help to reduce their overall numbers. However, they could neither eliminate mirror bacteria nor prevent potentially irreversible ecological changes. Indeed, the addition of human-engineered biological control agents might lead to further unexpected and irreversible ecological changes. While the production and establishment of artificial biological controls might prove to be necessary, they could neither prevent nor reverse severe ecological harms expected to result from the introduction of invasive mirror bacteria.

# Germline engineering might allow the creation of plants and animals resistant to mirror bacteria

Of the many potential ecological consequences of invasive mirror bacteria, perhaps the most concerning is that a mirror heterotroph could directly infect and kill many animal and plant species to the point of extinction. Given the rapidity with which invasive microbes can spread through the

environment and the absence of biological controls or other realistic countermeasures to block the spread of mirror bacteria, protecting susceptible wild populations appears implausible.

It might be possible to genetically engineer some animals and plants to resist mirror bacterial infection (as discussed for plants in Chapter 7). Receptors might be engineered to recognize common mirror macromolecules (Cesari *et al.*, 2022; June & Sadelain, 2018), synthetic mirror-responsive receptors could trigger the release of anti-mirror effectors (Weinberg *et al.*, 2024; Zhu *et al.*, 2022), and mirror-responsive opsonins (Banaszek *et al.*, 2019) could enable existing neutrophils to find and kill infecting mirror bacteria with ROS. Any natural plants and animals that proved resilient to infection might suggest other protective mechanisms. If successfully engineered into the germline, such measures could potentially prevent mirror bacteria from establishing an infection in the protected organisms and their descendants.

While heritable immunity to mirror infection appears feasible with sufficiently advanced biotechnology, logistical hurdles would likely preclude protecting more than a tiny fraction of the millions of different plants and animals in nature (Mora *et al.*, 2011). Thus, germline engineering would likely prove most effective if used to protect key crops and a few other highly valued species. Along with any naturally resistant survivors, these engineered species might be used to repopulate devastated ecosystems in order to create new functional ecosystems, akin to rewilding efforts that use functional analogs (Svenning *et al.*, 2016) but on a much larger scale. Engineered resistance might eventually mitigate some of the worst effects of invasive mirror bacteria, but the resulting artificial ecosystems would be greatly impoverished, and much biodiversity would be irreversibly lost.

## **Contributions and Acknowledgments**

#### **Author Contributions**

Authors are listed alphabetically by surname for each chapter; ordering does not indicate the level of individual contributions. The initial draft of the technical report was based on research conducted by KME, which was further developed through investigations by MLN, SBO, and DAR. All authors contributed additional research, writing, editing, or feedback for their respective chapters. DJB and JMS coordinated the compilation and review of the final report.

#### Acknowledgments

The authors of this report would like to thank the co-authors of the *Science* article for providing input at various stages of the technical report's creation. In addition, we are grateful to the following people who provided critical feedback on drafts of individual chapters, or, in some cases, the entire report: William J. Bradshaw, Salvadore Buse, Ivana Cvijović, John Donnelly, Seth Donoughe, Steven Erickson, Julius Fredens, Louise F.H. Funke, Jacob Glanville, Alex Kleinman, Michael McLaren, Alexander Mijalis, Sriharshita (Harshu) Musunuri, Marta Napiorkowska, Velislava Petrova, Georgia Ray, Daniel Rice, Adin Richards, Maximilian Schons, Jon Sanders, Elizabeth Shepherd, Chris Somerville, Christopher Stamper, Benjamin Stewart, Gina Suh, Joe Torres, Kyle Webster, James H.R. Wheeler, and Heather Youngs. In addition, we would like to thank Andrew Snyder-Beattie and James Wagstaff, who helped to initiate the writing of the report; and James A. Smith and Ivana Cvijović, who provided operational support in the production and review of the report. Finally, we would like to thank all of the reviewers (listed on pg iii), including those who chose to remain anonymous. Responsibility for the final content rests entirely with the authors.

Figures and illustrations presented in this report were based on concepts designed by the authors, and created in partnership with Alana McPherson and Kip Lyall at Science Journal Editors (SJE). The images are made available under a Creative Commons Attribution-Noncommercial-Sharealike License (CC BY-NC-SA 4.0).

The authors acknowledge Open Philanthropy for their support of the technical report through operational assistance, and funding to RKD, KME, GL, MLN, SBO, and JMS. DAR also acknowledges past support for work on the same topic. The authors also acknowledge operational support from the Mirror Biology Dialogues Fund, enabled by financial contributions from the Alfred P. Sloan Foundation, the David and Lucile Packard Foundation, the Gordon and Betty Moore Foundation, Open Philanthropy, and Patrick Collison. KPA and JIG acknowledge support from the National Science Foundation, Award #1901145 (RCN Build-a-Cell: An Open Community Considering & Advancing the Construction of Synthetic Cells). KME acknowledges support from NSF Award #1943141 (Towards Open and Responsive Ecological Editing). YC acknowledges support from UKRI Engineering and Physical Sciences Research Council EP/V05967X/1 (Open Plus Fellowship: Engineering and safeguarding synthetic genomes). Other authors acknowledge general support from a wide range of sources. The views expressed here are those of the individuals and not those of any organizations with which they are affiliated.

#### **Competing Interests**

VSC is a co-founder, Scientific Advisor, and equity holder in SeqCoast Genomics, LLC, and Middle Author Bioinformatics, LLC. VSC and JP are board members of the Mirror Biology Dialogues Fund. FJI is a co-founder of Pearl Bio. JDGJ is the founder and a shareholder of Norfolk Plant Sciences and Norfolk Healthy Produce. RM is a member of the advisory board for *Cell*, and the scientific advisory boards of Amgen, GlaxoSmithKline, and Janssen Pharmaceuticals. SBO is the founder and an executive at SynX Therapeutics Ltd. HS is listed as a co-inventor on patents pertaining to lactazole engineering, as well as patents pertaining to tRNA engineering to enable the incorporation of various D-amino acids. JMS is a former employee and shareholder of Teiko.bio. In the last 5 years, RKD, KME, TVI, GL, MLN, SBO, JP, DAR, JMS, and BW have, either as individuals or as project leaders, received funding from Open Philanthropy to support work that is not directly related to this report. KPA, DA, YC, TWH, JIG, REL, JWS, and NJT declare no conflicts of interest relevant to this research

## References

### **Chapter 1**

- Abdulrashid, N., & Clark, D. P. (1987). Isolation and genetic analysis of mutations allowing the degradation of furans and thiophenes by *Escherichia coli*. *Journal of Bacteriology*, 169(3), 1267–1271.
- Adler, J., Hazelbauer, G. L., & Dahl, M. M. (1973). Chemotaxis toward sugars in *Escherichia coli*. Journal of Bacteriology, 115(3), 824–847.
- Ambrogelly, A., Palioura, S., & Söll, D. (2007). Natural expansion of the genetic code. *Nature Chemical Biology*, *3*(1), 29–35.
- Baldomà, L., & Aguilar, J. (1987). Involvement of lactaldehyde dehydrogenase in several metabolic pathways of Escherichia coli K-12. The Journal of Biological Chemistry, 262(29), 13991–13996.
- Bentley, R. (2006). The nose as a stereochemist. Enantiomers and odor. Chemical Reviews, 106(9), 4099-4112.
- Cao, Y., Mu, H., Guo, J., Liu, H., Zhang, R., Liu, W., Xian, M., & Liu, H. (2020). Metabolic engineering of *Escherichia coli* for the utilization of ethanol. *Journal of Biological Research*, 27(1), 1–8.
- Celis, T. F. (1977). Independent regulation of transport and biosynthesis of arginine in *Escherichia coli* K-12. *Journal of Bacteriology*, *130*(3), 1244–1252.
- Chonoles Imlay, K. R., Korshunov, S., & Imlay, J. A. (2015). Physiological roles and adverse effects of the two cystine importers of *Escherichia coli*. Journal of Bacteriology, 197(23), 3629–3644.
- Chupp, T. E., Fierlinger, P., Ramsey-Musolf, M. J., & Singh, J. T. (2019). Electric dipole moments of atoms, molecules, nuclei, and particles. *Reviews of Modern Physics*, 91(1), 015001.
- Clark, D. P., & Cronan, J. E. (1996). Two-carbon compounds and fatty acids as carbon sources. In F. C. Neidhardt, R. Curtiss III, J. L. Ingraham, E. C. C. Lin, K. B. Low, B. Magasanik, W. S. Reznikoff, M. Riley, M. Schaechter, & H. E. Umbarger (Eds.), *Escherichia coli and Salmonella: Cellular and molecular biology* (2nd ed., pp. 343–357). ASM Press.
- Clark, D. P., & Rod, M. L. (1987). Regulatory mutations that allow the growth of *Escherichia coli* on butanol as carbon source. *Journal of Molecular Evolution*, 25(2), 151–158.
- Cotton, C. A. R., Bernhardsgrütter, I., He, H., Burgener, S., Schulz, L., Paczia, N., Dronsella, B., Erban, A., Toman, S., Dempfle, M., De Maria, A., Kopka, J., Lindner, S. N., Erb, T. J., & Bar-Even, A. (2020). Underground isoleucine biosynthesis pathways in *E. coli. eLife*, *9*, e54207.
- Crick, F. H. (1968). The origin of the genetic code. Journal of Molecular Biology, 38(3), 367-379.
- Cusa, E., Obradors, N., Baldomà, L., Badía, J., & Aguilar, J. (1999). Genetic analysis of a chromosomal region containing genes required for assimilation of allantoin nitrogen and linked glyoxylate metabolism in *Escherichia coli. Journal of Bacteriology*, 181(24), 7479–7484.

#### References

- Daussy, C., Marrel, T., Amy-Klein, A., Nguyen, C. T., Bordé, C. J., & Chardonnet, C. (1999). Limit on the parity nonconserving energy difference between the enantiomers of a chiral molecule by laser spectroscopy. *Physical Review Letters*, 83(8), 1554–1557.
- Díaz, E., Ferrández, A., Prieto, M. A., & García, J. L. (2001). Biodegradation of aromatic compounds by *Escherichia coli. Microbiology and Molecular Biology Reviews*, *65*(4), 523–569.
- Dong, J. M., Taylor, J. S., Latour, D. J., Iuchi, S., & Lin, E. C. (1993). Three overlapping lct genes involved in L-lactate utilization by *Escherichia coli*. *Journal of Bacteriology*, *175*(20), 6671–6678.
- Dougherty, T. J., Thanassi, J. A., & Pucci, M. J. (1993). The *Escherichia coli* mutant requiring D-glutamic acid is the result of mutations in two distinct genetic loci. *Journal of Bacteriology*, 175(1), 111–116.
- Dover, S., & Halpern, Y. S. (1972). Utilization of γ-aminobutyric acid as the sole carbon and nitrogen source by *Escherichia coli* K-12 mutants. *Journal of Bacteriology*, *109*(2), 835–843.
- Fung, V. K. Y., Tan, D. Z. J., & Zhou, K. (2023). Improving glycine utilization in *Escherichia coli*. Biochemical Engineering Journal, 193, 108834.
- Horný, L., & Quack, M. (2015). Computation of molecular parity violation using the coupled-cluster linear response approach. *Molecular Physics*, 113(13-14), 1768–1779.
- Hutchison, C. A., 3rd, Chuang, R.-Y., Noskov, V. N., Assad-Garcia, N., Deerinck, T. J., Ellisman, M. H., Gill, J., Kannan, K., Karas, B. J., Ma, L., Pelletier, J. F., Qi, Z.-Q., Richter, R. A., Strychalski, E. A., Sun, L., Suzuki, Y., Tsvetanova, B., Wise, K. S., Smith, H. O., ... Venter, J. C. (2016). Design and synthesis of a minimal bacterial genome. *Science*, 351(6280), aad6253.
- Ibañez, E., Gimenez, R., Pedraza, T., Baldoma, L., Aguilar, J., & Badia, J. (2000). Role of the yiaR and yiaS genes of *Escherichia coli* in metabolism of endogenously formed L-xylulose. *Journal of Bacteriology*, 182(16), 4625–4627.
- Iwadate, Y., & Kato, J.-I. (2019). Identification of a formate-dependent uric acid degradation pathway in *Escherichia* coli. Journal of Bacteriology, 201(11).
- Kadner, R. J. (1977). Transport and utilization of D-methionine and other methionine sources in *Escherichia coli*. *Journal of Bacteriology*, *129*(1), 207–216.
- Kakkar, A., Nizampatnam, N. R., Kondreddy, A., Pradhan, B. B., & Chatterjee, S. (2015). Xanthomonas campestris cell-cell signalling molecule DSF (diffusible signal factor) elicits innate immunity in plants and is suppressed by the exopolysaccharide xanthan. Journal of Experimental Botany, 66(21), 6697–6714.
- Kalyananda, M. K., Engel, R., & Tropp, B. E. (1987). Metabolism of L-glyceraldehyde 3-phosphate in *Escherichia coli*. Journal of Bacteriology, 169(6), 2488–2493.
- Kao, W. T. K., Frye, M., Gagnon, P., Vogel, J. P., & Chole, R. (2017). D-amino acids do not inhibit Pseudomonas aeruginosa biofilm formation. Laryngoscope Investigative Otolaryngology, 2(1), 4–9.
- Keller, P., Reiter, M. A., Kiefer, P., Gassler, T., Hemmerle, L., Christen, P., Noor, E., & Vorholt, J. A. (2022). Generation of an *Escherichia coli* strain growing on methanol via the ribulose monophosphate cycle. *Nature Communications*, 13(1), 5243.

- Koonin, E. V., & Novozhilov, A. S. (2017). Origin and evolution of the universal genetic code. *Annual Review of Genetics*, *51*(1), 45–62.
- Kuhn, J., & Somerville, R. L. (1971). Mutant strains of Escherichia coli K-12 that use D-amino acids. Proceedings of the National Academy of Sciences of the United States of America, 68(10), 2484–2487.
- Lander, A. J., Jin, Y., & Luk, L. Y. P. (2023). D-peptide and D-protein technology: Recent advances, challenges, and opportunities. *ChemBioChem*, 24(4), e202200537.
- LeBlanc, D. J., & Mortlock, R. P. (1971). Metabolism of D-arabinose: A new pathway in *Escherichia coli*. Journal of Bacteriology, 106(1), 90–96.
- Lin, E. C. C. (1996). Dissimilatory pathways for sugars, polyols, and carboxylates. In F. C. Neidhardt, R. Curtiss III, J. L. Ingraham, E. C. C. Lin, K. B. Low, B. Magasanik, W. S. Reznikoff, M. Riley, M. Schaechter, & H. E. Umbarger (Eds.), *Escherichia coli and Salmonella: Cellular and molecular biology* (2nd ed., pp. 307–342). ASM Press.
- Lukas, H., Reimann, J., Kim, O. B., Grimpo, J., & Unden, G. (2010). Regulation of aerobic and anaerobic D-malate metabolism of *Escherichia coli* by the LysR-type regulator DmlR (YeaT). *Journal of Bacteriology*, 192(10), 2503–2511.
- Miller, S. M., Simon, R. J., Ng, S., Zuckermann, R. N., Kerr, J. M., & Moos, W. H. (1995). Comparison of the proteolytic susceptibilities of homologous L-amino acid, D-amino acid, and N-substituted glycine peptide and peptoid oligomers. *Drug Development Research*, 35(1), 20–32.
- Milton, R. C., Milton, S. C., & Kent, S. B. (1992). Total chemical synthesis of a D-enzyme: The enantiomers of HIV-1 protease show reciprocal chiral substrate specificity [corrected]. *Science*, *256*(5062), 1445–1448.
- National Natural Science Foundation of China. (2022). *National Natural Science Foundation of China 2022 annual report*. https://www.nsfc.gov.cn/english/site\_1/report/C1/2023/09-12/340.html
- National Science Foundation. (2014). NSF award search: Award #1443228 ERASynBio: Establishment of a fully synthetic, mirror-image biological system. https://www.nsf.gov/awardsearch/showAward?AWD\_ID=1443228
- National Science Foundation. (2019a). NSF award search: Award #1935120 Collaborative research: Booting up a mirror cell. https://www.nsf.gov/awardsearch/showAward?AWD\_ID=1935120
- National Science Foundation. (2019b). NSF award search: Award #1935372 Collaborative research: Booting up a mirror cell. https://www.nsf.gov/awardsearch/showAward?AWD\_ID=1935372
- Park, H.-Y., Park, C.-S., Kim, H.-J., & Oh, D.-K. (2007). Substrate specificity of a galactose 6-phosphate isomerase from *Lactococcus lactis* that produces D-allose from D-psicose. *Journal of Biotechnology*, 132(1), 88–95.
- Prieto, M. I., Martin, J., Balaña-Fouce, R., & Garrido-Pertierra, A. (1987). Properties of γ-aminobutyraldehyde dehydrogenase from *Escherichia coli*. *Biochimie*, *69*(11-12), 1161–1168.
- Quack, M., Seyfang, G., & Wichmann, G. (2022). Perspectives on parity violation in chiral molecules: Theory, spectroscopic experiment and biomolecular homochirality. *Chemical Science*, *13*(36), 10598–10643.
- Reitzer, L. (2005). Catabolism of amino acids and related compounds. *EcoSal Plus*, 1(2). https://doi.org/10.1128/ecosalplus.3.4.7

- Rothschild, L. J., Averesch, N. J. H., Strychalski, E. A., Moser, F., Glass, J. I., Cruz Perez, R., Yekinni, I. O., Rothschild-Mancinelli, B., Roberts Kingman, G. A., Wu, F., Waeterschoot, J., Ioannou, I. A., Jewett, M. C., Liu, A. P., Noireaux, V., Sorenson, C., & Adamala, K. P. (2024). Building synthetic cells—From the technology infrastructure to cellular entities. *ACS Synthetic Biology*, *13*(4), 974–997.
- Rumbo, C., Vallejo, J. A., Cabral, M. P., Martínez-Guitián, M., Pérez, A., Beceiro, A., & Bou, G. (2016). Assessment of antivirulence activity of several D-amino acids against *Acinetobacter baumannii* and *Pseudomonas aeruginosa. The Journal of Antimicrobial Chemotherapy*, 71(12), 3473–3481.
- Safronova, M. S., Budker, D., DeMille, D., Jackson Kimball, D. F., Derevianko, A., & Clark, C. W. (2018). Search for new physics with atoms and molecules. *Reviews of Modern Physics*, *90*(2), 025008.
- Sallembien, Q., Bouteiller, L., Crassous, J., & Raynal, M. (2022). Possible chemical and physical scenarios towards biological homochirality. *Chemical Society Reviews*, 51(9), 3436–3476.
- Sarkar, S., & Pires, M. M. (2015). D-amino acids do not inhibit biofilm formation in *Staphylococcus aureus*. *PloS ONE*, 10(2), e0117613.
- Scangos, G. A., & Reiner, A. M. (1978). Ribitol and D-arabitol catabolism in *Escherichia coli*. Journal of Bacteriology, 134(2), 492–500.
- Shen, Q., Wang, L., Zhou, H., Jiang, H.-D., Yu, L.-S., & Zeng, S. (2013). Stereoselective binding of chiral drugs to plasma proteins. *Acta Pharmacologica Sinica*, 34(8), 998–1006.
- Shimizu, T., Takaya, N., & Nakamura, A. (2012). An L-glucose catabolic pathway in *Paracoccus* species 43P. *The Journal of Biological Chemistry*, 287(48), 40448–40456.
- Shiue, E., & Prather, K. L. J. (2014). Improving D-glucaric acid production from myo-inositol in *E. coli* by increasing MIOX stability and myo-inositol transport. *Metabolic Engineering*, *22*, 22–31.
- Smith, S. W. (2009). Chiral toxicology: It's the same thing...only different. Toxicological Sciences, 110(1), 4–30.
- Sørensen, M. A., & Pedersen, S. (1991). Cysteine, even in low concentrations, induces transient amino acid starvation in *Escherichia coli*. *Journal of Bacteriology*, 173(16), 5244–5246.
- Soutourina, J., Blanquet, S., & Plateau, P. (2001). Role of D-cysteine desulfhydrase in the adaptation of *Escherichia* coli to D-cysteine. *The Journal of Biological Chemistry*, 276(44), 40864–40872.
- Soutourina, J., Plateau, P., & Blanquet, S. (2000). Metabolism of D-aminoacyl-tRNAs in *Escherichia coli* and *Saccharomyces cerevisiae* cells. *The Journal of Biological Chemistry*, 275(42), 32535–32542.
- Soutourina, J., Plateau, P., Delort, F., Peirotes, A., & Blanquet, S. (1999). Functional characterization of the D-Tyr-tRNATyr deacylase from *Escherichia coli*. *The Journal of Biological Chemistry*, 274(27), 19109–19114.
- Spiegel, D. S., & Turner, E. L. (2012). Bayesian analysis of the astrobiological implications of life's early emergence on Earth. *Proceedings of the National Academy of Sciences of the United States of America*, 109(2), 395–400.
- Tang, C. T., Engel, R., & Tropp, B. E. (1977). L-Glyceraldehyde 3-phosphate, a bactericidal agent. Antimicrobial Agents and Chemotherapy, 11(1), 147–153.
- Tokunaga, E., Yamamoto, T., Ito, E., & Shibata, N. (2018). Understanding the thalidomide chirality in biological processes by the self-disproportionation of enantiomers. *Scientific Reports*, 8(1), 17131.

- Tong, M., French, S., El Zahed, S. S., Ong, W. K., Karp, P. D., & Brown, E. D. (2020). Gene dispensability in *Escherichia coli* grown in thirty different carbon environments. *mBio*, 11(5), e02259-20.
- Trimbur, D. E., & Mortlock, R. P. (1991). Isolation and characterization of *Escherichia coli* mutants able to utilize the novel pentose L-ribose. *Journal of Bacteriology*, *173*(8), 2459–2464.
- Tyler, B. (1978). Regulation of the assimilation of nitrogen compounds. *Annual Review of Biochemistry*, 47(1), 1127–1162.
- Uldrich, A. P., Rigau, M., & Godfrey, D. I. (2020). Immune recognition of phosphoantigen-butyrophilin molecular complexes by γδ T cells. *Immunological Reviews*, *298*(1), 74–83.
- Umbarger, H. E., & Brown, B. (1956). Threonine deamination in *Escherichia coli* I. *Journal of Bacteriology*, 71(4), 443–449.
- Unden, G., Strecker, A., Kleefeld, A., & Kim, O. B. (2016). C4-dicarboxylate utilization in aerobic and anaerobic growth. *EcoSal Plus*, 7(1). https://doi.org/10.1128/ecosalplus.ESP-0021-2015
- Usvalampi, A., Turunen, O., Valjakka, J., Pastinen, O., Leisola, M., & Nyyssölä, A. (2012). Production of L-xylose from L-xylulose using *Escherichia coli* L-fucose isomerase. *Enzyme and Microbial Technology*, 50(1), 71–76.
- Wydau, S., van der Rest, G., Aubard, C., Plateau, P., & Blanquet, S. (2009). Widespread distribution of cell defense against D-aminoacyl-tRNAs. *The Journal of Biological Chemistry*, 284(21), 14096–14104.
- Xi, H., Schneider, B. L., & Reitzer, L. (2000). Purine catabolism in *Escherichia coli* and function of xanthine dehydrogenase in purine salvage. *Journal of Bacteriology*, 182(19), 5332–5341.
- Xu, Y., & Zhu, T. F. (2022). Mirror-image T7 transcription of chirally inverted ribosomal and functional RNAs. *Science*, 378(6618), 405–412.
- Zawadzke, L. E., & Berg, J. M. (1993). The structure of a centrosymmetric protein crystal. Proteins, 16(3), 301-305.
- Zhou, Y., & Imlay, J. A. (2022). *Escherichia coli* uses a dedicated importer and desulfidase to ferment cysteine. *mBio*, *13*(2), e0296521.
- Zhu, Y., & Lin, E. C. (1986). An evolvant of *Escherichia coli* that employs the L-fucose pathway also for growth on L-galactose and D-arabinose. *Journal of Molecular Evolution*, *23*(3), 259–266.
- Zhu, Y., & Lin, E. C. (1987). Loss of aldehyde dehydrogenase in an *Escherichia coli* mutant selected for growth on the rare sugar L-galactose. *Journal of Bacteriology*, 169(2), 785–789.

### Chapter 2

- Abil, Z., Restrepo Sierra, A. M., & Danelon, C. (2023). Clonal amplification-enhanced gene expression in synthetic vesicles. ACS Synthetic Biology, 12(4), 1187–1203.
- Achenbach, J., Jahnz, M., Bethge, L., Paal, K., Jung, M., Schuster, M., Albrecht, R., Jarosch, F., Nierhaus, K. H., & Klussmann, S. (2015). Outwitting EF-Tu and the ribosome: Translation with D-amino acids. *Nucleic Acids Research*, 43(12), 5687–5698.

- Adamala, K. P., Dogterom, M., Elani, Y., Schwille, P., Takinoue, M., & Tang, T.-Y. D. (2024). Present and future of synthetic cell development. *Nature Reviews Molecular Cell Biology*, 25(3), 162–167.
- Agouridas, V., El Mahdi, O., Diemer, V., Cargoët, M., Monbaliu, J.-C. M., & Melnyk, O. (2019). Native chemical ligation and extended methods: Mechanisms, catalysis, scope, and limitations. *Chemical Reviews*, 119(12), 7328–7443.
- Aliashkevich, A., Alvarez, L., & Cava, F. (2018). New insights into the mechanisms and biological roles of D-amino acids in complex eco-systems. *Frontiers in Microbiology*, 9, 683.
- Amiram, M., Haimovich, A. D., Fan, C., Wang, Y.-S., Aerni, H.-R., Ntai, I., Moonan, D. W., Ma, N. J., Rovner, A. J., Hong, S. H., Kelleher, N. L., Goodman, A. L., Jewett, M. C., Söll, D., Rinehart, J., & Isaacs, F. J. (2015). Evolution of translation machinery in recoded bacteria enables multi-site incorporation of nonstandard amino acids. *Nature Biotechnology*, 33(12), 1272–1279.
- Aoyama, R., Masuda, K., Shimojo, M., Kanamori, T., Ueda, T., & Shimizu, Y. (2022). In vitro reconstitution of the Escherichia coli 70S ribosome with a full set of recombinant ribosomal proteins. The Journal of Biochemistry, 171(2), 227–237.
- Arranz-Gibert, P., Vanderschuren, K., & Isaacs, F. J. (2018). Next-generation genetic code expansion. *Current Opinion in Chemical Biology*, *46*, 203–211.
- Avital-Shmilovici, M., Mandal, K., Gates, Z. P., Phillips, N. B., Weiss, M. A., & Kent, S. B. H. (2013). Fully convergent chemical synthesis of ester insulin: Determination of the high resolution X-ray structure by racemic protein crystallography. *Journal of the American Chemical Society*, 135(8), 3173–3185.
- Baker, J. J., Shi, J., Wang, S., Mujica, E. M., Bianco, S., Capponi, S., & Dueber, J. E. (2024). ML-enhanced peroxisome capacity enables compartmentalization of multienzyme pathway. *Nature Chemical Biology*, 20(1), 1–9.
- Bangham, A. D., Standish, M. M., & Weissmann, G. (1965). The action of steroids and streptolysin S on the permeability of phospholipid structures to cations. *Journal of Molecular Biology*, *13*(1), 253–259.
- Banigan, J. R., Mandal, K., Sawaya, M. R., Thammavongsa, V., Hendrickx, A. P. A., Schneewind, O., Yeates, T. O., & Kent, S. B. H. (2010). Determination of the X-ray structure of the snake venom protein omwaprin by total chemical synthesis and racemic protein crystallography. *Protein Science*, 19(10), 1840–1849.
- Beattie, A. T., Dunkelmann, D. L., & Chin, J. W. (2023). Quintuply orthogonal pyrrolysyl-tRNA synthetase/tRNAPyl pairs. *Nature Chemistry*, 15(7), 948–959.
- Behrendt, R., White, P., & Offer, J. (2016). Advances in Fmoc solid-phase peptide synthesis. *Journal of Peptide Science: An Official Publication of the European Peptide Society*, 22(1), 4–27.

biomers.net. (n.d.). L-DNA. Retrieved from https://www.biomers.net/en/products/special\_nucleic\_acids/l\_dna.html

- Blair, H. A. (2016). Etelcalcetide: First global approval. Drugs, 76(18), 1787-1792.
- Caforio, A., Siliakus, M. F., Exterkate, M., Jain, S., Jumde, V. R., Andringa, R. L. H., Kengen, S. W. M., Minnaard, A. J., Driessen, A. J. M., & van der Oost, J. (2018). Converting *Escherichia coli* into an archaebacterium with a hybrid heterochiral membrane. *Proceedings of the National Academy of Sciences of the United States of America*, 115(14), 3704–3709.

#### References

- Callahan, A. J., Gandhesiri, S., Travaline, T. L., Reja, R. M., Lozano Salazar, L., Hanna, S., Lee, Y.-C., Li, K., Tokareva, O. S., Swiecicki, J.-M., Loas, A., Verdine, G. L., McGee, J. H., & Pentelute, B. L. (2024). Mirror-image ligand discovery enabled by single-shot fast-flow synthesis of D-proteins. *Nature Communications*, 15(1), 1813.
- Cervettini, D., Tang, S., Fried, S. D., Willis, J. C. W., Funke, L. F. H., Colwell, L. J., & Chin, J. W. (2020). Rapid discovery and evolution of orthogonal aminoacyl-tRNA synthetase-tRNA pairs. *Nature Biotechnology*, *38*(8), 989–999.
- Chang, T. M. (1964). Semipermeable microcapsules. Science, 146(3643), 524-525.
- Chatterjee, A., Lajoie, M. J., Xiao, H., Church, G. M., & Schultz, P. G. (2014). A bacterial strain with a unique quadruplet codon specifying non-native amino acids. *ChemBioChem*, 15(12), 1782–1786.
- Chin, J. W. (2017). Expanding and reprogramming the genetic code. Nature, 550(7674), 53-60.
- Costello, A. M., Peterson, A. A., Lanster, D. L., Li, Z., Carver, G. D., & Badran, A. H. (2024). Efficient genetic code expansion without host genome modifications. *Nature Biotechnology*, 42(9), 1121-1128.
- Cross, L. L., Paudyal, R., Kamisugi, Y., Berry, A., Cuming, A. C., Baker, A., & Warriner, S. L. (2017). Towards designer organelles by subverting the peroxisomal import pathway. *Nature Communications*, 8(1), 454.
- Cui, Z., Stein, V., Tnimov, Z., Mureev, S., & Alexandrov, K. (2015). Semisynthetic tRNA complement mediates in vitro protein synthesis. Journal of the American Chemical Society, 137(13), 4404–4413.
- Dang, B., Kubota, T., Mandal, K., Correa, A. M., Bezanilla, F., & Kent, S. B. H. (2016). Elucidation of the covalent and tertiary structures of biologically active Ts3 toxin. *Angewandte Chemie International Edition*, 55(30), 8639–8642.
- d'Aquino, A. E., Kim, D. S., & Jewett, M. C. (2018). Engineered ribosomes for basic science and synthetic biology. *Annual Review of Chemical and Biomolecular Engineering*, *9*, 311–340.
- de la Torre, D., & Chin, J. W. (2021). Reprogramming the genetic code. Nature Reviews Genetics, 22(3), 169-184.
- de Maddalena, L. L., Niederholtmeyer, H., Turtola, M., Swank, Z. N., Belogurov, G. A., & Maerkl, S. J. (2016). GreA and GreB enhance expression of *Escherichia coli* RNA polymerase promoters in a reconstituted transcription-translation system. ACS Synthetic Biology, 5(9), 929–935.
- Deyama, T., Matsui, Y., Chadani, Y., Sekine, Y., Doi, N., & Fujiwara, K. (2021). Transcription-translation of the *Escherichia coli* genome within artificial cells. *Chemical Communications*, *57*(80), 10367–10370.
- diCenzo, G. C., & Finan, T. M. (2017). The divided bacterial genome: Structure, function, and evolution. *Microbiology and Molecular Biology Reviews*, 81(3), e00019-17.
- Discher, B. M., Won, Y. Y., Ege, D. S., Lee, J. C., Bates, F. S., Discher, D. E., & Hammer, D. A. (1999). Polymersomes: Tough vesicles made from diblock copolymers. *Science*, 284(5417), 1143–1146.
- Dolberg, T. B., Meger, A. T., Boucher, J. D., Corcoran, W. K., Schauer, E. E., Prybutok, A. N., Raman, S., & Leonard, J. N. (2021). Computation-guided optimization of split protein systems. *Nature Chemical Biology*, 17(5), 531–539.

- Dunkelmann, D. L., Oehm, S. B., Beattie, A. T., & Chin, J. W. (2021). A 68-codon genetic code to incorporate four distinct non-canonical amino acids enabled by automated orthogonal mRNA design. *Nature Chemistry*, 13(11), 1110–1117.
- Dunkelmann, D. L., Piedrafita, C., Dickson, A., Liu, K. C., Elliott, T. S., Fiedler, M., Bellini, D., Zhou, A., Cervettini, D., & Chin, J. W. (2024). Adding α,α-disubstituted and β-linked monomers to the genetic code of an organism. *Nature*, 625(7995), 603–610.
- Fan, C., Deng, Q., & Zhu, T. F. (2021). Bioorthogonal information storage in L-DNA with a high-fidelity mirror-image Pfu DNA polymerase. *Nature Biotechnology*, 39(12), 1548–1555.
- Fredens, J., Wang, K., de la Torre, D., Funke, L. F. H., Robertson, W. E., Christova, Y., Chia, T., Schmied, W. H., Dunkelmann, D. L., Beránek, V., Uttamapinant, C., Llamazares, A. G., Elliott, T. S., & Chin, J. W. (2019). Total synthesis of *Escherichia coli* with a recoded genome. *Nature*, 569(7757), 514–518.
- Freschlin, C. R., Fahlberg, S. A., & Romero, P. A. (2022). Machine learning to navigate fitness landscapes for protein engineering. *Current Opinion in Biotechnology*, 75, 102713.
- Fujino, T., Goto, Y., Suga, H., & Murakami, H. (2013). Reevaluation of the D-amino acid compatibility with the elongation event in translation. *Journal of the American Chemical Society*, *135*(5), 1830–1837.
- Garenne, D., Thompson, S., Brisson, A., Khakimzhan, A., & Noireaux, V. (2021). The all-*E. coli* TXTL toolbox 3.0: New capabilities of a cell-free synthetic biology platform. *Synthetic Biology*, *6*(1), ysab017.
- Gaut, N. J., & Adamala, K. P. (2021). Reconstituting natural cell elements in synthetic cells. *Advanced Biology*, 5(3), e2000188.
- Gibson, D. G., Glass, J. I., Lartigue, C., Noskov, V. N., Chuang, R.-Y., Algire, M. A., Benders, G. A., Montague, M. G., Ma, L., Moodie, M. M., Merryman, C., Vashee, S., Krishnakumar, R., Assad-Garcia, N., Andrews-Pfannkoch, C., Denisova, E. A., Young, L., Qi, Z.-Q., Segall-Shapiro, T. H., ... Venter, J. C. (2010). Creation of a bacterial cell controlled by a chemically synthesized genome. *Science*, *329*(5987), 52–56.
- Gibson, D. G., Young, L., Chuang, R.-Y., Venter, J. C., Hutchison, C. A., 3rd, & Smith, H. O. (2009). Enzymatic assembly of DNA molecules up to several hundred kilobases. *Nature Methods*, 6(5), 343–345.
- Glen Research. (n.d.). *Beta L-DNA Phosphoramidites*. Retrieved from https://www.glenresearch.com/products/labels-and-modifiers/backbone-modification/beta-l-dna-phosphoramid ites.html
- Gregorio, N. E., Levine, M. Z., & Oza, J. P. (2019). A user's guide to cell-free protein synthesis. *Methods and Protocols*, *2*(1), 24.
- Hagino, K., & Ichihashi, N. (2023). In vitro transcription/translation-coupled DNA replication through partial regeneration of 20 aminoacyl-tRNA synthetases. ACS Synthetic Biology, 12(4), 1252–1263.
- Haimovich, A. D., Muir, P., & Isaacs, F. J. (2015). Genomes by design. Nature Reviews Genetics, 16(9), 501-516.
- Harrison, K., Mackay, A. S., Kambanis, L., Maxwell, J. W. C., & Payne, R. J. (2023). Synthesis and applications of mirror-image proteins. *Nature Reviews Chemistry*, 7(6), 383–404.

- Hartrampf, N., Saebi, A., Poskus, M., Gates, Z. P., Callahan, A. J., Cowfer, A. E., Hanna, S., Antilla, S., Schissel, C. K., Quartararo, A. J., Ye, X., Mijalis, A. J., Simon, M. D., Loas, A., Liu, S., Jessen, C., Nielsen, T. E., & Pentelute, B. L. (2020). Synthesis of proteins by automated flow chemistry. *Science*, *368*(6494), 980–987.
- Has, C., & Sunthar, P. (2020). A comprehensive review on recent preparation techniques of liposomes. *Journal of Liposome Research*, 30(4), 336–365.
- Holowka, E. P., Pochan, D. J., & Deming, T. J. (2005). Charged polypeptide vesicles with controllable diameter. *Journal of the American Chemical Society*, *127*(35), 12423–12428.
- Hoose, A., Vellacott, R., Storch, M., Freemont, P. S., & Ryadnov, M. G. (2023). DNA synthesis technologies to close the gene writing gap. *Nature Reviews Chemistry*, 7(3), 144–161.
- Huang, Y.-C., Chen, C.-C., Gao, S., Wang, Y.-H., Xiao, H., Wang, F., Tian, C.-L., & Li, Y.-M. (2016). Synthesis of L- and D-Ubiquitin by one-pot ligation and metal-free desulfurization. *Chemistry*, *22*(22), 7623–7628.
- Hung, L. W., Kohmura, M., Ariyoshi, Y., & Kim, S. H. (1999). Structural differences in D- and L-monellin in the crystals of racemic mixture. *Journal of Molecular Biology*, 285(1), 311–321.
- Hutchison, C. A., 3rd, Chuang, R.-Y., Noskov, V. N., Assad-Garcia, N., Deerinck, T. J., Ellisman, M. H., Gill, J., Kannan, K., Karas, B. J., Ma, L., Pelletier, J. F., Qi, Z.-Q., Richter, R. A., Strychalski, E. A., Sun, L., Suzuki, Y., Tsvetanova, B., Wise, K. S., Smith, H. O., ... Venter, J. C. (2016). Design and synthesis of a minimal bacterial genome. *Science*, 351(6280), aad6253.
- Imanishi, S., Katoh, T., Yin, Y., Yamada, M., Kawai, M., & Suga, H. (2021). *In vitro* selection of macrocyclic D/L-hybrid peptides against human EGFR. *Journal of the American Chemical Society*, *143*(15), 5680–5684.
- Itaya, M., & Tanaka, T. (1997). Experimental surgery to create subgenomes of *Bacillus subtilis* 168. *Proceedings of the National Academy of Sciences of the United States of America*, 94(10), 5378–5382.
- Iwane, Y., Hitomi, A., Murakami, H., Katoh, T., Goto, Y., & Suga, H. (2016). Expanding the amino acid repertoire of ribosomal polypeptide synthesis via the artificial division of codon boxes. *Nature Chemistry*, 8(4), 317–325.
- Iwane, Y., Katoh, T., Goto, Y., & Suga, H. (2018). Artificial division of codon boxes for expansion of the amino acid repertoire of ribosomal polypeptide synthesis. *Methods in Molecular Biology*, 1728, 17–47.
- Jewett, M. C., Fritz, B. R., Timmerman, L. E., & Church, G. M. (2013). *In vitro* integration of ribosomal RNA synthesis, ribosome assembly, and translation. *Molecular Systems Biology*, 9(1), 678.
- Jiang, W., Wu, Z., Gao, Z., Wan, M., Zhou, M., Mao, C., & Shen, J. (2022). Artificial cells: Past, present and future. ACS Nano, 16(10), 15705–15733.
- Joesaar, A., Yang, S., Bögels, B., van der Linden, A., Pieters, P., Kumar, B. V. V. S. P., Dalchau, N., Phillips, A., Mann, S., & de Greef, T. F. A. (2019). DNA-based communication in populations of synthetic protocells. *Nature Nanotechnology*, 14(4), 369–378.
- Joyce, G. F., & Szostak, J. W. (2018). Protocells and RNA self-replication. *Cold Spring Harbor Perspectives in Biology*, *10*(9), a034801.
- Katoh, T., Iwane, Y., & Suga, H. (2017). Logical engineering of D-arm and T-stem of tRNA that enhances D-amino acid incorporation. *Nucleic Acids Research*, *45*(22), 12601–12610.

- Katoh, T., & Suga, H. (2022). *In vitro* genetic code reprogramming for the expansion of usable noncanonical amino acids. *Annual Review of Biochemistry*, 91(1), 221–243.
- Katoh, T., Tajima, K., & Suga, H. (2017). Consecutive elongation of D-amino acids in translation. *Cell Chemical Biology*, 24(1), 46–54.
- Kaur, H., Bruno, J. G., Kumar, A., & Sharma, T. K. (2018). Aptamers in the therapeutics and diagnostics pipelines. *Theranostics*, *8*(15), 4016–4032.
- Kazuta, Y., Matsuura, T., Ichihashi, N., & Yomo, T. (2014). Synthesis of milligram quantities of proteins using a reconstituted *in vitro* protein synthesis system. *Journal of Bioscience and Bioengineering*, 118(5), 554–557.
- Kim, K.-R., Kim, H. Y., Lee, Y.-D., Ha, J. S., Kang, J. H., Jeong, H., Bang, D., Ko, Y. T., Kim, S., Lee, H., & Ahn, D.-R. (2016). Self-assembled mirror DNA nanostructures for tumor-specific delivery of anticancer drugs. *Journal of Controlled Release*, 243, 121–131.
- Kriebisch, C., Bantysh, O., Baranda, L., Belluati, A., Bertosin, E., Dai, K., de Roy, M., Fu, H., Galvanetto, N., Gibbs, J., Santhosh Gomez, S., Granatelli, G., Griffo, A., Guix, M., Onur Gurdap, C., Harth-Kitzerow, J., Haugerud, I., Häfner, G., Jaiswal, P., ... Boekhoven, J. (2024). A roadmap towards the synthesis of life. *ChemRxiv*. https://doi.org/10.26434/chemrxiv-2024-tnx83
- Krzyzosiak, W., Denman, R., Nurse, K., Hellmann, W., Boublik, M., Gehrke, C. W., Agris, P. F., & Ofengand, J. (1987). *In vitro* synthesis of 16S ribosomal RNA containing single base changes and assembly into a functional 30S ribosome. *Biochemistry*, 26(8), 2353–2364.
- Kulkarni, S. S., Sayers, J., Premdjee, B., & Payne, R. J. (2018). Rapid and efficient protein synthesis through expansion of the native chemical ligation concept. *Nature Reviews Chemistry*, 2(4), 0122.
- Kuruma, Y., & Ueda, T. (2015). The PURE system for the cell-free synthesis of membrane proteins. *Nature Protocols*, *10*(9), 1328–1344.
- Lajoie, M. J., Rovner, A. J., Goodman, D. B., Aerni, H.-R., Haimovich, A. D., Kuznetsov, G., Mercer, J. A., Wang, H. H., Carr, P. A., Mosberg, J. A., Rohland, N., Schultz, P. G., Jacobson, J. M., Rinehart, J., Church, G. M., & Isaacs, F. J. (2013). Genomically recoded organisms expand biological functions. *Science*, 342(6156), 357–360.
- Lavickova, B., Laohakunakorn, N., & Maerkl, S. J. (2020). A partially self-regenerating synthetic cell. *Nature Communications*, *11*(1), 6340.
- Leiman, S. A., May, J. M., Lebar, M. D., Kahne, D., Kolter, R., & Losick, R. (2013). D-amino acids indirectly inhibit biofilm formation in *Bacillus subtilis* by interfering with protein synthesis. *Journal of Bacteriology*, *195*(23), 5391–5395.
- Libicher, K., Hornberger, R., Heymann, M., & Mutschler, H. (2020). *In vitro* self-replication and multicistronic expression of large synthetic genomes. *Nature Communications*, 11(1), 904.
- Li, J., Zhang, C., Huang, P., Kuru, E., Forster-Benson, E. T. C., Li, T., & Church, G. M. (2017). Dissecting limiting factors of the Protein synthesis Using Recombinant Elements (PURE) system. *Translation*, 5(1), e1327006.
- Lin, C., Ke, Y., Li, Z., Wang, J. H., Liu, Y., & Yan, H. (2009). Mirror image DNA nanostructures for chiral supramolecular assemblies. *Nano Letters*, 9(1), 433–436.

#### References

- Lin, X., Nithun, R. V., Samanta, R., Harel, O., & Jbara, M. (2023). Enabling peptide ligation at aromatic junction mimics via native chemical ligation and palladium-mediated S-arylation. *Organic Letters*, 25(25), 4715–4719.
- Lovelock, S. L., Crawshaw, R., Basler, S., Levy, C., Baker, D., Hilvert, D., & Green, A. P. (2022). The road to fully programmable protein catalysis. *Nature*, 606(7912), 49–58.
- Luisier, S., Avital-Shmilovici, M., Weiss, M. A., & Kent, S. B. H. (2010). Total chemical synthesis of human proinsulin. *Chemical Communications*, 46(43), 8177–8179.
- Mackay, A. L. (1989). Crystal enigma. Nature, 342(6246), 133-133.
- Malyshev, D. A., Dhami, K., Lavergne, T., Chen, T., Dai, N., Foster, J. M., Corrêa, I. R., Jr., & Romesberg, F. E. (2014). A semi-synthetic organism with an expanded genetic alphabet. *Nature*, *509*(7500), 385–388.
- Mandal, K., Pentelute, B. L., Tereshko, V., Kossiakoff, A. A., & Kent, S. B. H. (2009). X-ray structure of native scorpion toxin BmBKTx1 by racemic protein crystallography using direct methods. *Journal of the American Chemical Society*, 131(4), 1362–1363.
- Mandal, K., Pentelute, B. L., Tereshko, V., Thammavongsa, V., Schneewind, O., Kossiakoff, A. A., & Kent, S. B. H. (2009). Racemic crystallography of synthetic protein enantiomers used to determine the X-ray structure of plectasin by direct methods. *Protein Science*, 18(6), 1146–1154.
- Milton, R. C., Milton, S. C., & Kent, S. B. (1992). Total chemical synthesis of a D-enzyme: The enantiomers of HIV-1 protease show reciprocal chiral substrate specificity [corrected]. *Science*, *256*(5062), 1445–1448.
- Mukai, T., Lajoie, M. J., Englert, M., & Söll, D. (2017). Rewriting the genetic code. *Annual Review of Microbiology*, 71, 557–577.
- Muttenthaler, M., King, G. F., Adams, D. J., & Alewood, P. F. (2021). Trends in peptide drug discovery. *Nature Reviews Drug Discovery*, 20(4), 309–325.
- Neumann, H., Wang, K., Davis, L., Garcia-Alai, M., & Chin, J. W. (2010). Encoding multiple unnatural amino acids via evolution of a quadruplet-decoding ribosome. *Nature*, 464(7287), 441–444.
- Noireaux, V., & Libchaber, A. (2004). A vesicle bioreactor as a step toward an artificial cell assembly. *Proceedings* of the National Academy of Sciences of the United States of America, 101(51), 17669–17674.
- Oakley, A. J. (2019). A structural view of bacterial DNA replication. Protein Science, 28(6), 990-1004.
- Oerlemans, R. A. J. F., Timmermans, S. B. P. E., & van Hest, J. C. M. (2021). Artificial organelles: Towards adding or restoring intracellular activity. *ChemBioChem*, 22(12), 2051–2078.
- Orelle, C., Carlson, E. D., Szal, T., Florin, T., Jewett, M. C., & Mankin, A. S. (2015). Protein synthesis by ribosomes with tethered subunits. *Nature*, *524*(7563), 119–124.
- Payne, C. D., Vadlamani, G., Hajiaghaalipour, F., Muhammad, T., Fisher, M. F., Andersson, H. S., Göransson, U., Clark, R. J., Bond, C. S., Mylne, J. S., & Rosengren, K. J. (2021). Solution NMR and racemic crystallography provide insights into a novel structural class of cyclic plant peptides. *RSC Chemical Biology*, 2(6), 1682–1691.
- Pech, A., Achenbach, J., Jahnz, M., Schülzchen, S., Jarosch, F., Bordusa, F., & Klussmann, S. (2017). A thermostable D-polymerase for mirror-image PCR. *Nucleic Acids Research*, 45(7), 3997–4005.
- Pentelute, B. L., Mandal, K., Gates, Z. P., Sawaya, M. R., Yeates, T. O., & Kent, S. B. H. (2010). Total chemical synthesis and X-ray structure of kaliotoxin by racemic protein crystallography. *Chemical Communications*, 46(43), 8174–8176.
- Petrovic Fabijan, A., Martinez-Martin, D., Venturini, C., Mickiewicz, K., Flores-Rodriguez, N., Errington, J., & Iredell, J. (2022). L-form switching in *Escherichia coli* as a common β-lactam resistance mechanism. *Microbiology Spectrum*, 10(5), e0241922.
- Pósfai, G., Plunkett, G., III, Fehér, T., Frisch, D., Keil, G. M., Umenhoffer, K., Kolisnychenko, V., Stahl, B., Sharma, S. S., de Arruda, M., Burland, V., Harcum, S. W., & Blattner, F. R. (2006). Emergent properties of reduced-genome *Escherichia coli*. *Science*, *312*(5776), 1044–1046.
- Rackham, O., & Chin, J. W. (2005). A network of orthogonal ribosome x mRNA pairs. *Nature Chemical Biology*, *1*(3), 159–166.
- Robertson, W. E., Funke, L. F. H., de la Torre, D., Fredens, J., Elliott, T. S., Spinck, M., Christova, Y., Cervettini, D., Böge, F. L., Liu, K. C., Buse, S., Maslen, S., Salmond, G. P. C., & Chin, J. W. (2021). Sense codon reassignment enables viral resistance and encoded polymer synthesis. *Science*, *372*(6546), 1057–1062.
- Rothschild, L. J., Averesch, N. J. H., Strychalski, E. A., Moser, F., Glass, J. I., Cruz Perez, R., Yekinni, I. O., Rothschild-Mancinelli, B., Roberts Kingman, G. A., Wu, F., Waeterschoot, J., Ioannou, I. A., Jewett, M. C., Liu, A. P., Noireaux, V., Sorenson, C., & Adamala, K. P. (2024). Building synthetic cells—From the technology infrastructure to cellular entities. *ACS Synthetic Biology*, *13*(4), 974–997.
- Rumbo, C., Vallejo, J. A., Cabral, M. P., Martínez-Guitián, M., Pérez, A., Beceiro, A., & Bou, G. (2016). Assessment of antivirulence activity of several D-amino acids against *Acinetobacter baumannii* and *Pseudomonas aeruginosa*. *The Journal of Antimicrobial Chemotherapy*, 71(12), 3473–3481.
- Schmied, W. H., Tnimov, Z., Uttamapinant, C., Rae, C. D., Fried, S. D., & Chin, J. W. (2018). Controlling orthogonal ribosome subunit interactions enables evolution of new function. *Nature*, 564(7736), 444–448.
- Seeman, N. C., & Sleiman, H. F. (2017). DNA nanotechnology. Nature Reviews Materials, 3(1), 17068.
- Semrad, K., & Green, R. (2002). Osmolytes stimulate the reconstitution of functional 50S ribosomes from *in vitro* transcripts of *Escherichia coli* 23S rRNA. *RNA*, 8(4), 401–411.
- Shang, L., Ye, F., Li, M., & Zhao, Y. (2022). Spatial confinement toward creating artificial living systems. *Chemical Society Reviews*, 51(10), 4075–4093.
- Shimizu, Y., Inoue, A., Tomari, Y., Suzuki, T., Yokogawa, T., Nishikawa, K., & Ueda, T. (2001). Cell-free translation reconstituted with purified components. *Nature Biotechnology*, 19(8), 751–755.
- Shin, J., Jardine, P., & Noireaux, V. (2012). Genome replication, synthesis, and assembly of the bacteriophage T7 in a single cell-free reaction. *ACS Synthetic Biology*, *1*(9), 408–413.
- Sidore, A. M., Plesa, C., Samson, J. A., Lubock, N. B., & Kosuri, S. (2020). DropSynth 2.0: High-fidelity multiplexed gene synthesis in emulsions. *Nucleic Acids Research*, 48(16), e95.
- Silverman, A. D., Karim, A. S., & Jewett, M. C. (2020). Cell-free gene expression: An expanded repertoire of applications. *Nature Reviews Genetics*, 21(3), 151–170.
- Simon, A. J., d'Oelsnitz, S., & Ellington, A. D. (2019). Synthetic evolution. Nature Biotechnology, 37(7), 730-743.

- Soutourina, J., Plateau, P., & Blanquet, S. (2000). Metabolism of D-aminoacyl-tRNAs in *Escherichia coli* and *Saccharomyces cerevisiae* cells. *The Journal of Biological Chemistry*, 275(42), 32535–32542.
- Tabata, K. V., Sogo, T., Moriizumi, Y., & Noji, H. (2019). Regeneration of *Escherichia coli* giant protoplasts to their original form. *Life*, 9(1), 24.
- Teng, P., Zheng, M., Cerrato, D. C., Shi, Y., Zhou, M., Xue, S., Jiang, W., Wojtas, L., Ming, L.-J., Hu, Y., & Cai, J. (2021). The folding propensity of α/sulfono-γ-AA peptidic foldamers with both left- and right-handedness. *Communications Chemistry*, 4(1), 58.
- Terasaka, N., Hayashi, G., Katoh, T., & Suga, H. (2014). An orthogonal ribosome-tRNA pair via engineering of the peptidyl transferase center. *Nature Chemical Biology*, *10*(7), 555–557.
- Tuerk, C., & Gold, L. (1990). Systematic evolution of ligands by exponential enrichment: RNA ligands to bacteriophage T4 DNA polymerase. *Science*, *249*(4968), 505–510.
- van Nies, P., Westerlaken, I., Blanken, D., Salas, M., Mencía, M., & Danelon, C. (2018). Self-replication of DNA by its encoded proteins in liposome-based synthetic cells. *Nature Communications*, 9(1), 1583.
- van Swaay, D., Tang, T.-Y. D., Mann, S., & de Mello, A. (2015). Microfluidic formation of membrane-free aqueous coacervate droplets in water. *Angewandte Chemie International Edition*, 54(29), 8398–8401.
- Vogelstein, B., & Kinzler, K. W. (1999). Digital PCR. Proceedings of the National Academy of Sciences of the United States of America, 96(16), 9236–9241.
- Volland, J.-M., Gonzalez-Rizzo, S., Gros, O., Tyml, T., Ivanova, N., Schulz, F., Goudeau, D., Elisabeth, N. H., Nath, N., Udwary, D., Malmstrom, R. R., Guidi-Rontani, C., Bolte-Kluge, S., Davies, K. M., Jean, M. R., Mansot, J.-L., Mouncey, N. J., Angert, E. R., Woyke, T., & Date, S. V. (2022). A centimeter-long bacterium with DNA contained in metabolically active, membrane-bound organelles. *Science*, *376*(6600), 1453–1458.
- Wang, H. H., Isaacs, F. J., Carr, P. A., Sun, Z. Z., Xu, G., Forest, C. R., & Church, G. M. (2009). Programming cells by multiplex genome engineering and accelerated evolution. *Nature*, 460(7257), 894–898.
- Wang, K., de la Torre, D., Robertson, W. E., & Chin, J. W. (2019). Programmed chromosome fission and fusion enable precise large-scale genome rearrangement and assembly. *Science*, *365*(6456), 922–926.
- Weidmann, J., Schnölzer, M., Dawson, P. E., & Hoheisel, J. D. (2019). Copying life: Synthesis of an enzymatically active mirror-image DNA-ligase made of D-amino acids. *Cell Chemical Biology*, 26(5), 645–651.e3.
- Wei, E., & Endy, D. (2021). Experimental tests of functional molecular regeneration via a standard framework for coordinating synthetic cell building. *bioRxiv*. https://www.biorxiv.org/content/10.1101/2021.03.03.433818v1
- Weinstock, M. T., Jacobsen, M. T., & Kay, M. S. (2014). Synthesis and folding of a mirror-image enzyme reveals ambidextrous chaperone activity. *Proceedings of the National Academy of Sciences of the United States of America*, 111(32), 11679–11684.
- Welch, B. D., Francis, J. N., Redman, J. S., Paul, S., Weinstock, M. T., Reeves, J. D., Lie, Y. S., Whitby, F. G., Eckert, D. M., Hill, C. P., Root, M. J., & Kay, M. S. (2010). Design of a potent D-peptide HIV-1 entry inhibitor with a strong barrier to resistance. *Journal of Virology*, 84(21), 11235–11244.
- Xu, W., Jiang, W., Wang, J., Yu, L., Chen, J., Liu, X., Liu, L., & Zhu, T. F. (2017). Total chemical synthesis of a thermostable enzyme capable of polymerase chain reaction. *Cell Discovery*, *3*(1), 17008.

- Xu, Y., & Zhu, T. F. (2022). Mirror-image T7 transcription of chirally inverted ribosomal and functional RNAs. *Science*, *378*(6618), 405–412.
- Yang, K. K., Wu, Z., & Arnold, F. H. (2019). Machine-learning-guided directed evolution for protein engineering. *Nature Methods*, 16(8), 687–694.
- Yeung, H., Squire, C. J., Yosaatmadja, Y., Panjikar, S., López, G., Molina, A., Baker, E. N., Harris, P. W. R., & Brimble, M. A. (2016). Radiation damage and racemic protein crystallography reveal the unique structure of the GASA/snakin protein superfamily. *Angewandte Chemie International Edition*, 55(28), 7930–7933.
- Yoneji, T., Fujita, H., Mukai, T., & Su'etsugu, M. (2021). Grand scale genome manipulation via chromosome swapping in *Escherichia coli* programmed by three one megabase chromosomes. *Nucleic Acids Research*, 49(15), 8407–8418.
- Young, D. D., & Schultz, P. G. (2018). Playing with the molecules of life. ACS Chemical Biology, 13(4), 854-870.
- Zhang, J., Wang, Y., Chai, B., Wang, J., Li, L., Liu, M., Zhao, G., Yao, L., Gao, X., Yin, Y., & Xu, J. (2020). Efficient and low-cost error removal in DNA synthesis by a high-durability MutS. ACS Synthetic Biology, 9(4), 940–952.
- Zhang, Y., Ptacin, J. L., Fischer, E. C., Aerni, H. R., Caffaro, C. E., San Jose, K., Feldman, A. W., Turner, C. R., & Romesberg, F. E. (2017). A semi-synthetic organism that stores and retrieves increased genetic information. *Nature*, 551(7682), 644–647.
- Zhan, P., Jahnke, K., Liu, N., & Göpfrich, K. (2022). Functional DNA-based cytoskeletons for synthetic cells. *Nature Chemistry*, 14(8), 958–963.

## Chapter 3

- Bahey-El-Din, M., Casey, P. G., Griffin, B. T., & Gahan, C. G. M. (2010). Efficacy of a Lactococcus lactis ΔpyrG vaccine delivery platform expressing chromosomally integrated hly from Listeria monocytogenes. Bioengineered Bugs, 1(1), 66–74.
- Bartke, K., Garoff, L., Huseby, D. L., Brandis, G., & Hughes, D. (2021). Genetic architecture and fitness of bacterial interspecies hybrids. *Molecular Biology and Evolution*, 38(4), 1472–1481.
- Brophy, J. A. N., & Voigt, C. A. (2014). Principles of genetic circuit design. Nature Methods, 11(5), 508-520.
- Butler, N. D., Sen, S., Brown, L. B., Lin, M., & Kunjapur, A. M. (2023). A platform for distributed production of synthetic nitrated proteins in live bacteria. *Nature Chemical Biology*, 19(7), 911–920.
- Byers, K. B., & Harding, A. L. (2017). Laboratory-associated infections. In D. P. Wooley & K. B. Byers (Eds.), *Biological safety: Principles and practices* (5th ed., pp. 59–92). ASM Press.
- Calles, J., Justice, I., Brinkley, D., Garcia, A., & Endy, D. (2019). Fail-safe genetic codes designed to intrinsically contain engineered organisms. *Nucleic Acids Research*, 47(19), 10439–10451.
- Chiang, A. J., & Hasty, J. (2023). Design of synthetic bacterial biosensors. *Current Opinion in Microbiology*, 76, 102380.

- Danzig, R., Sageman, M., Leighton, T., Hough, L., Yuki, H., Kotani, R., & Hosford, Z. M. (2011). Aum Shinrikyo: Insights into how terrorists develop biological and chemical weapons. Center for a New American Security. https://www.cnas.org/publications/reports/aum-shinrikyo-insights-into-how-terrorists-develop-biological-and-c hemical-weapons
- de la Torre, D., & Chin, J. W. (2021). Reprogramming the genetic code. Nature Reviews Genetics, 22(3), 169-184.
- Fong, S. S., Marciniak, J. Y., & Palsson, B. Ø. (2003). Description and interpretation of adaptive evolution of *Escherichia coli* K-12 MG1655 by using a genome-scale *in silico* metabolic model. *Journal of Bacteriology*, 185(21), 6400–6408.
- Gallagher, R. R., Patel, J. R., Interiano, A. L., Rovner, A. J., & Isaacs, F. J. (2015). Multilayered genetic safeguards limit growth of microorganisms to defined environments. *Nucleic Acids Research*, 43(3), 1945–1954.
- Genchi, G. (2017). An overview on D-amino acids. Amino Acids, 49(9), 1521-1533.
- Gerdes, S. Y., Scholle, M. D., D'Souza, M., Bernal, A., Baev, M. V., Farrell, M., Kurnasov, O. V., Daugherty, M. D., Mseeh, F., Polanuyer, B. M., Campbell, J. W., Anantha, S., Shatalin, K. Y., Chowdhury, S. A. K., Fonstein, M. Y., & Osterman, A. L. (2002). From genetic footprinting to antimicrobial drug targets: Examples in cofactor biosynthetic pathways. *Journal of Bacteriology*, *184*(16), 4555–4572.
- Gibson, D. G., Glass, J. I., Lartigue, C., Noskov, V. N., Chuang, R.-Y., Algire, M. A., Benders, G. A., Montague, M. G., Ma, L., Moodie, M. M., Merryman, C., Vashee, S., Krishnakumar, R., Assad-Garcia, N., Andrews-Pfannkoch, C., Denisova, E. A., Young, L., Qi, Z.-Q., Segall-Shapiro, T. H., ... Venter, J. C. (2010). Creation of a bacterial cell controlled by a chemically synthesized genome. *Science*, *329*(5987), 52–56.
- Gillum, D., Krishnan, P., & Byers, K. (2016). A searchable laboratory-acquired infection database. *Applied Biosafety: Journal of the American Biological Safety Association*, 21(4), 203–207.
- Goeddel, D. V., Kleid, D. G., Bolivar, F., Heyneker, H. L., Yansura, D. G., Crea, R., Hirose, T., Kraszewski, A., Itakura, K., & Riggs, A. D. (1979). Expression in *Escherichia coli* of chemically synthesized genes for human insulin. *Proceedings of the National Academy of Sciences of the United States of America*, 76(1), 106–110.
- Gould, C., & Folb, P. (2002). *Project Coast: Apartheid's chemical and biological warfare programme*. United Nations Institute for Disarmament Research.
- Gronvall, G. K. (2016). *Synthetic biology: Safety, security, and promise*. CreateSpace Independent Publishing Platform.
- Harrison, K., Mackay, A. S., Kambanis, L., Maxwell, J. W. C., & Payne, R. J. (2023). Synthesis and applications of mirror-image proteins. *Nature Reviews Chemistry*, 7(6), 383–404.
- Hemez, C., Mohler, K., Radford, F., Moen, J., Rinehart, J., & Isaacs, F. J. (2024). Genomically recoded *Escherichia coli* with optimized functional phenotypes. *bioRxiv*. https://doi.org/10.1101/2024.08.29.610322
- Herring, C. D., Raghunathan, A., Honisch, C., Patel, T., Applebee, M. K., Joyce, A. R., Albert, T. J., Blattner, F. R., van den Boom, D., Cantor, C. R., & Palsson, B. Ø. (2006). Comparative genome sequencing of *Escherichia coli* allows observation of bacterial evolution on a laboratory timescale. *Nature Genetics*, 38(12), 1406–1412.
- Hughes, R. A., & Ellington, A. D. (2017). Synthetic DNA synthesis and assembly: Putting the synthetic in synthetic biology. *Cold Spring Harbor Perspectives in Biology*, *9*(1), a023812.

- Itaya, M., Tsuge, K., Koizumi, M., & Fujita, K. (2005). Combining two genomes in one cell: Stable cloning of the Synechocystis PCC6803 genome in the Bacillus subtilis 168 genome. Proceedings of the National Academy of Sciences of the United States of America, 102(44), 15971–15976.
- Jackson, S. S., Sumner, L. E., Garnier, C. H., Basham, C., Sun, L. T., Simone, P. L., Gardner, D. S., & Casagrande, R. J. (2019). The accelerating pace of biotech democratization. *Nature Biotechnology*, 37(12), 1403–1408.
- Labroussaa, F., Lebaudy, A., Baby, V., Gourgues, G., Matteau, D., Vashee, S., Sirand-Pugnet, P., Rodrigue, S., & Lartigue, C. (2016). Impact of donor-recipient phylogenetic distance on bacterial genome transplantation. *Nucleic Acids Research*, 44(17), 8501–8511.
- LeBlanc, D. J., & Mortlock, R. P. (1971). Metabolism of D-arabinose: A new pathway in *Escherichia coli*. Journal of Bacteriology, 106(1), 90–96.
- Lee, D.-H., & Palsson, B. Ø. (2010). Adaptive evolution of *Escherichia coli* K-12 MG1655 during growth on a nonnative carbon source, L-1,2-propanediol. *Applied and Environmental Microbiology*, 76(13), 4158–4168.
- Leitenberg, M., & Zilinskas, R. A. (2012). *The soviet biological weapons program: A history*. Harvard University Press.
- Mandell, D. J., Lajoie, M. J., Mee, M. T., Takeuchi, R., Kuznetsov, G., Norville, J. E., Gregg, C. J., Stoddard, B. L., & Church, G. M. (2015). Biocontainment of genetically modified organisms by synthetic protein design. *Nature*, 518(7537), 55–60.
- Manheim, D., & Lewis, G. (2022). High-risk human-caused pathogen exposure events from 1975-2016. *F1000Research*, 10, 752.
- Miyamoto, T., & Homma, H. (2021). D-Amino acid metabolism in bacteria. *The Journal of Biochemistry*, *170*(1), 5–13.
- Moger-Reischer, R. Z., Glass, J. I., Wise, K. S., Sun, L., Bittencourt, D. M. C., Lehmkuhl, B. K., Schoolmaster, D. R., Jr, Lynch, M., & Lennon, J. T. (2023). Evolution of a minimal cell. *Nature*, *620*(7972), 122–127.
- Nielsen, J., & Keasling, J. D. (2016). Engineering cellular metabolism. Cell, 164(6), 1185–1197.
- Overton, T. W. (2014). Recombinant protein production in bacterial hosts. Drug Discovery Today, 19(5), 590-601.
- Parachini, J. V., & Gunaratna, R. K. (2022). Implications of the pandemic for terrorist interest in biological weapons. RAND.
- Park, H.-Y., Park, C.-S., Kim, H.-J., & Oh, D.-K. (2007). Substrate specificity of a galactose 6-phosphate isomerase from *Lactococcus lactis* that produces D-allose from D-psicose. *Journal of Biotechnology*, 132(1), 88–95.
- Riglar, D. T., & Silver, P. A. (2018). Engineering bacteria for diagnostic and therapeutic applications. *Nature Reviews Microbiology*, 16(4), 214–225.
- Ro, D.-K., Paradise, E. M., Ouellet, M., Fisher, K. J., Newman, K. L., Ndungu, J. M., Ho, K. A., Eachus, R. A., Ham, T. S., Kirby, J., Chang, M. C. Y., Withers, S. T., Shiba, Y., Sarpong, R., & Keasling, J. D. (2006). Production of the antimalarial drug precursor artemisinic acid in engineered yeast. *Nature*, 440(7086), 940–943.

- Ronchel, M. C., & Ramos, J. L. (2001). Dual system to reinforce biological containment of recombinant bacteria designed for rhizoremediation. *Applied and Environmental Microbiology*, 67(6), 2649–2656.
- Rovner, A. J., Haimovich, A. D., Katz, S. R., Li, Z., Grome, M. W., Gassaway, B. M., Amiram, M., Patel, J. R., Gallagher, R. R., Rinehart, J., & Isaacs, F. J. (2015). Recoded organisms engineered to depend on synthetic amino acids. *Nature*, 518(7537), 89–93.
- Sandberg, T. E., Salazar, M. J., Weng, L. L., Palsson, B. O., & Feist, A. M. (2019). The emergence of adaptive laboratory evolution as an efficient tool for biological discovery and industrial biotechnology. *Metabolic Engineering*, 56, 1–16.
- Schindler, D., Dai, J., & Cai, Y. (2018). Synthetic genomics: A new venture to dissect genome fundamentals and engineer new functions. *Current Opinion in Chemical Biology*, *46*, 56–62.
- Sharan, S. K., Thomason, L. C., Kuznetsov, S. G., & Court, D. L. (2009). Recombineering: A homologous recombination-based method of genetic engineering. *Nature Protocols*, 4(2), 206–223.
- Shimizu, T., Takaya, N., & Nakamura, A. (2012). An L-glucose catabolic pathway in *Paracoccus* species 43P. *The Journal of Biological Chemistry*, 287(48), 40448–40456.
- Spratt, B. G. (2007). *Independent review of the safety of UK facilities handling foot-and-mouth disease virus*. Department for Environment, Food & Rural Affairs
- Steidler, L., Neirynck, S., Huyghebaert, N., Snoeck, V., Vermeire, A., Goddeeris, B., Cox, E., Remon, J. P., & Remaut, E. (2003). Biological containment of genetically modified *Lactococcus lactis* for intestinal delivery of human interleukin 10. *Nature Biotechnology*, 21(7), 785–789.
- Sun, K., Li, S., Zheng, B., Zhu, Y., Wang, T., Liang, M., Yao, Y., Zhang, K., Zhang, J., Li, H., Han, D., Zheng, J., Coventry, B., Cao, L., Baker, D., Liu, L., & Lu, P. (2024). Accurate *de novo* design of heterochiral protein-protein interactions. *Cell Research*. https://doi.org/10.1038/s41422-024-01014-2.
- Tan, Y., Wu, H., Wei, T., & Li, X. (2020). Chemical protein synthesis: Advances, challenges, and outlooks. *Journal of the American Chemical Society*, 142(48), 20288–20298.
- Teng, Y., Jiang, T., & Yan, Y. (2024). The expanded CRISPR toolbox for constructing microbial cell factories. *Trends in Biotechnology*, 42(1), 104–118.
- Trimbur, D. E., & Mortlock, R. P. (1991). Isolation and characterization of *Escherichia coli* mutants able to utilize the novel pentose L-ribose. *Journal of Bacteriology*, *173*(8), 2459–2464.
- Shooter, R. A. (Chairman). (1980). *Report of the investigation into the cause of the 1978 Birmingham smallpox occurrence*. Her Majesty's Stationery Office.

United Nations. (1972). Convention on the prohibition of the development, production and stockpiling of bacteriological (biological) and toxin weapons and on their destruction.

- United Nations Monitoring, Verification and Inspection Commission. (2007). *Compendium of Iraq's proscribed weapons programmes in the chemical, biological and missile areas*. United Nations.
- Usvalampi, A., Turunen, O., Valjakka, J., Pastinen, O., Leisola, M., & Nyyssölä, A. (2012). Production of L-xylose from L-xylulose using *Escherichia coli* L-fucose isomerase. *Enzyme and Microbial Technology*, 50(1), 71–76.

- Wang, H. H., Isaacs, F. J., Carr, P. A., Sun, Z. Z., Xu, G., Forest, C. R., & Church, G. M. (2009). Programming cells by multiplex genome engineering and accelerated evolution. *Nature*, 460(7257), 894–898.
- Weidmann, J., Schnölzer, M., Dawson, P. E., & Hoheisel, J. D. (2019). Copying life: Synthesis of an enzymatically active mirror-image DNA-ligase made of D-amino acids. *Cell Chemical Biology*, 26(5), 645–651.e3.
- World Health Organization. (1980). Thirty-third World Health Assembly, Geneva, 5–23 May 1980: Resolutions and decisions, annexes (WHA33/1980/REC/3). World Health Organization. https://iris.who.int/handle/10665/154893
- World Health Organization. (1996). Executive Board, ninety-seventh session, Geneva, 15–24 January 1996: Resolutions and decisions, annexes (EB97/1996/REC/1). World Health Organization. https://iris.who.int/handle/10665/172804
- World Health Organization. (2010). *Scientific review of variola virus research 1999–2010* (WHO/HSE/GAR/BDP/2010.3). World Health Organization. https://www.who.int/publications/i/item/WHO-HSE-GAR-BDP-2010-3
- Yachida, M., & Nakamura, A. (2024). Complete genome sequence of *Luteolibacter* sp. strain LG18, an L-glucose-utilizing bacterium isolated from soil. *Microbiology Resource Announcements*, 13(3), e0088823.
- Zhu, Y., & Lin, E. C. (1987). Loss of aldehyde dehydrogenase in an *Escherichia coli* mutant selected for growth on the rare sugar L-galactose. *Journal of Bacteriology*, *169*(2), 785–789.

# **Chapter 4**

- Abdulkadieva, M. M., Sysolyatina, E. V., Vasilieva, E. V., Gusarov, A. I., Domnin, P. A., Slonova, D. A., Stanishevskiy, Y. M., Vasiliev, M. M., Petrov, O. F., & Ermolaeva, S. A. (2022). Strain specific motility patterns and surface adhesion of virulent and probiotic *Escherichia coli*. *Scientific Reports*, 12(1), 614.
- Abergel, R. J., Wilson, M. K., Arceneaux, J. E. L., Hoette, T. M., Strong, R. K., Byers, B. R., & Raymond, K. N. (2006). Anthrax pathogen evades the mammalian immune system through stealth siderophore production. *Proceedings of the National Academy of Sciences of the United States of America*, 103(49), 18499–18503.
- Absolom, D. R. (1988). The role of bacterial hydrophobicity in infection: Bacterial adhesion and phagocytic ingestion. *Canadian Journal of Microbiology*, *34*(3), 287–298.
- Achinas, S., Charalampogiannis, N., & Euverink, G. J. W. (2019). A brief recap of microbial adhesion and biofilms. *Applied Sciences*, 9(14), 2801.
- Alloatti, A., Kotsias, F., Pauwels, A.-M., Carpier, J.-M., Jouve, M., Timmerman, E., Pace, L., Vargas, P., Maurin, M., Gehrmann, U., Joannas, L., Vivar, O. I., Lennon-Duménil, A.-M., Savina, A., Gevaert, K., Beyaert, R., Hoffmann, E., & Amigorena, S. (2015). Toll-like receptor 4 engagement on dendritic cells restrains Phago-lysosome fusion and promotes cross-presentation of antigens. *Immunity*, 43(6), 1087–1100.
- Alquraini, A., & El Khoury, J. (2020). Scavenger receptors. Current Biology: CB, 30(14), R790-R795.
- Andersen-Nissen, E., Smith, K. D., Strobe, K. L., Barrett, S. L. R., Cookson, B. T., Logan, S. M., & Aderem, A. (2005). Evasion of Toll-like receptor 5 by flagellated bacteria. *Proceedings of the National Academy of Sciences of the United States of America*, 102(26), 9247–9252.

- Anderson, B. N., Ding, A. M., Nilsson, L. M., Kusuma, K., Tchesnokova, V., Vogel, V., Sokurenko, E. V., & Thomas, W. E. (2007). Weak rolling adhesion enhances bacterial surface colonization. *Journal of Bacteriology*, 189(5), 1794–1802.
- Anderson, G. J., & Frazer, D. M. (2017). Current understanding of iron homeostasis. The American Journal of Clinical Nutrition, 106(Suppl 6), 1559S–1566S.
- Arpaia, N., & Barton, G. M. (2013). The impact of Toll-like receptors on bacterial virulence strategies. *Current Opinion in Microbiology*, 16(1), 17–22.
- Arranz-Gibert, P., Ciudad, S., Seco, J., García, J., Giralt, E., & Teixidó, M. (2018). Immunosilencing peptides by stereochemical inversion and sequence reversal: Retro-D-peptides. *Scientific Reports*, 8(1), 6446.
- Baka, M., Van Derlinden, E., Boons, K., Mertens, L., & Van Impe, J. F. (2013). Impact of pH on the cardinal temperatures of *E. coli* K12: Evaluation of the gamma hypothesis. *Food Control*, 29(2), 328–335.
- Bartnes, K., Hannestad, K., Guichard, G., & Briand, J. P. (1997). A retro-inverso analog mimics the cognate peptide epitope of a CD4+ T cell clone. *European Journal of Immunology*, 27(6), 1387–1391.
- Beloin, C., Roux, A., & Ghigo, J. M. (2008). Escherichia coli biofilms. Current Topics in Microbiology and Immunology, 322, 249–289.
- Bemark, M., Pitcher, M. J., Dionisi, C., & Spencer, J. (2024). Gut-associated lymphoid tissue: A microbiota-driven hub of B cell immunity. *Trends in Immunology*, 45(3), 211–223.
- Benachour, A., Ladjouzi, R., Le Jeune, A., Hébert, L., Thorpe, S., Courtin, P., Chapot-Chartier, M.-P., Prajsnar, T. K., Foster, S. J., & Mesnage, S. (2012). The lysozyme-induced peptidoglycan N-acetylglucosamine deacetylase PgdA (EF1843) is required for *Enterococcus faecalis* virulence. *Journal of Bacteriology*, 194(22), 6066–6073.
- Benkirane, N., Friede, M., Guichard, G., Briand, J. P., Van Regenmortel, M. H., & Muller, S. (1993). Antigenicity and immunogenicity of modified synthetic peptides containing D-amino acid residues. Antibodies to a D-enantiomer do recognize the parent L-hexapeptide and reciprocally. *The Journal of Biological Chemistry*, 268(35), 26279–26285.
- Ben-Mustapha, I., Ben-Farhat, K., Guirat-Dhouib, N., Dhemaied, E., Larguèche, B., Ben-Ali, M., Chemli, J., Bouguila, J., Ben-Mansour, L., Mellouli, F., Khemiri, M., Béjaoui, M., & Barbouche, M.-R. (2013). Clinical, immunological and genetic findings of a large Tunisian series of major histocompatibility complex class II deficiency patients. *Journal of Clinical Immunology*, 33(4), 865–870.
- Berends, E. T. M., Dekkers, J. F., Nijland, R., Kuipers, A., Soppe, J. A., van Strijp, J. A. G., & Rooijakkers, S. H. M. (2013). Distinct localization of the complement C5b-9 complex on *Gram*-positive bacteria: Localized C5b-9 deposition on *Gram*-positive bacteria. *Cellular Microbiology*, 15(12), 1955–1968.
- Bhan, U., Trujillo, G., Lyn-Kew, K., Newstead, M. W., Zeng, X., Hogaboam, C. M., Krieg, A. M., & Standiford, T. J. (2008). Toll-like receptor 9 regulates the lung macrophage phenotype and host immunity in murine pneumonia caused by *Legionella pneumophila*. *Infection and Immunity*, 76(7), 2895–2904.
- Bock, A. M., Cao, Q., Ferrieri, P., Young, J.-A. H., & Weisdorf, D. J. (2013). Bacteremia in blood or marrow transplantation patients: Clinical risk factors for infection and emerging antibiotic resistance. *Biology of Blood* and Marrow Transplantation: Journal of the American Society for Blood and Marrow Transplantation, 19(1), 102–108.

- Boes, M., Prodeus, A. P., Schmidt, T., Carroll, M. C., & Chen, J. (1998). A critical role of natural immunoglobulin M in immediate defense against systemic bacterial infection. *The Journal of Experimental Medicine*, 188(12), 2381–2386.
- Booty, L. M., & Bryant, C. E. (2022). Gasdermin D and beyond Gasdermin-mediated pyroptosis in bacterial infections. *Journal of Molecular Biology*, 434(4), 167409.
- Borek, F., Stupp, Y., Fuchs, S., & Sela, M. (1965). Relation between optical configuration and immunogenicity of synthetic polypeptides. *The Biochemical Journal*, *96*(3), 577–582.
- Boucher, Y., & Doolittle, W. F. (2000). The role of lateral gene transfer in the evolution of isoprenoid biosynthesis pathways. *Molecular Microbiology*, *37*(4), 703–716.
- Brecher, M. E., & Hay, S. N. (2005). Bacterial contamination of blood components. *Clinical Microbiology Reviews*, 18(1), 195–204.
- Brodell, L. A., & Rosenthal, K. S. (2008). Skin structure and function: The body's primary defense against infection. *Infectious Diseases in Clinical Practice*, *16*(2), 113–117.
- Brogden, K. A. (2005). Antimicrobial peptides: pore formers or metabolic inhibitors in bacteria? *Nature Reviews Microbiology*, *3*(3), 238–250.
- Brown, G. C. (2024). Cell death by phagocytosis. Nature Reviews Immunology, 24(2), 91-102.
- Burn, G. L., Foti, A., Marsman, G., Patel, D. F., & Zychlinsky, A. (2021). The neutrophil. *Immunity*, 54(7), 1377–1391.
- Callewaert, L., & Michiels, C. W. (2010). Lysozymes in the animal kingdom. *Journal of Biosciences*, 35(1), 127–160.
- Cecconi, M., Evans, L., Levy, M., & Rhodes, A. (2018). Sepsis and septic shock. Lancet, 392(10141), 75-87.
- Chakraborty, R. K., & Burns, B. (2024). Systemic Inflammatory Response Syndrome. StatPearls.
- Chandok, M. R., & Farber, D. L. (2004). Signaling control of memory T cell generation and function. *Seminars in Immunology*, *16*(5), 285–293.
- Chandra, P., Grigsby, S. J., & Philips, J. A. (2022). Immune evasion and provocation by *Mycobacterium tuberculosis*. *Nature Reviews Microbiology*, *20*(12), 750–766.
- Chaudhary, N. S., Donnelly, J. P., Moore, J. X., Baddley, J. W., Safford, M. M., & Wang, H. E. (2017). Association of baseline steroid use with long-term rates of infection and sepsis in the REGARDS cohort. *Critical Care*, 21(1), 185.
- Chawla, R. K., Lewis, F. W., Kutner, M. H., Bate, D. M., Roy, R. G., & Rudman, D. (1984). Plasma cysteine, cystine, and glutathione in cirrhosis. *Gastroenterology*, 87(4), 770–776.
- Chen, J. Y., Cortes, C., & Ferreira, V. P. (2018). Properdin: A multifaceted molecule involved in inflammation and diseases. *Molecular Immunology*, *102*, 58–72.
- Chen, T., Nomura, K., Wang, X., Sohrabi, R., Xu, J., Yao, L., Paasch, B. C., Ma, L., Kremer, J., Cheng, Y., Zhang, L., Wang, N., Wang, E., Xin, X.-F., & He, S. Y. (2020). A plant genetic network for preventing dysbiosis in the phyllosphere. *Nature*, 580(7805), 653–657.

- Chicz, R. M., Urban, R. G., Lane, W. S., Gorga, J. C., Stern, L. J., Vignali, D. A., & Strominger, J. L. (1992). Predominant naturally processed peptides bound to HLA-DR1 are derived from MHC-related molecules and are heterogeneous in size. *Nature*, 358(6389), 764–768.
- Chiu, M. L., Goulet, D. R., Teplyakov, A., & Gilliland, G. L. (2019). Antibody structure and function: The basis for engineering therapeutics. *Antibodies*, 8(4), 55.
- Chmiela, M., Miszczyk, E., & Rudnicka, K. (2014). Structural modifications of *Helicobacter pylori* lipopolysaccharide: an idea for how to live in peace. *World Journal of Gastroenterology: WJG*, 20(29), 9882–9897.
- Chong, P., Sia, C., Tripet, B., James, O., & Klein, M. (1996). Comparative immunological properties of enantiomeric peptides. *Letters in Peptide Science*, *3*(2), 99–106.
- Chu, D. S. H., Johnson, R. N., & Pun, S. H. (2012). Cathepsin B-sensitive polymers for compartment-specific degradation and nucleic acid release. *Journal of Controlled Release*, 157(3), 445–454.
- Chun, E., Lavoie, S., Fonseca-Pereira, D., Bae, S., Michaud, M., Hoveyda, H. R., Fraser, G. L., Gallini Comeau, C. A., Glickman, J. N., Fuller, M. H., Layden, B. T., & Garrett, W. S. (2019). Metabolite-sensing receptor Ffar2 regulates colonic group 3 innate lymphoid cells and gut immunity. *Immunity*, 51(5), 871–884.e6.
- Cicchinelli, S., Pignataro, G., Gemma, S., Piccioni, A., Picozzi, D., Ojetti, V., Franceschi, F., & Candelli, M. (2024). PAMPs and DAMPs in sepsis: A review of their molecular features and potential clinical implications. *International Journal of Molecular Sciences*, *25*(2), 962.
- Clarke, C. A., Scheurwater, E. M., & Clarke, A. J. (2010). The vertebrate lysozyme inhibitor Ivy functions to inhibit the activity of lytic transglycosylase. *The Journal of Biological Chemistry*, 285(20), 14843–14847.
- Clasen, S. J., Bell, M. E. W., Borbón, A., Lee, D.-H., Henseler, Z. M., de la Cuesta-Zuluaga, J., Parys, K., Zou, J., Wang, Y., Altmannova, V., Youngblut, N. D., Weir, J. R., Gewirtz, A. T., Belkhadir, Y., & Ley, R. E. (2023). Silent recognition of flagellins from human gut commensal bacteria by Toll-like receptor 5. *Science Immunology*, 8(79), eabq7001.
- Clifton, M. C., Rupert, P. B., Hoette, T. M., Raymond, K. N., Abergel, R. J., & Strong, R. K. (2019). Parsing the functional specificity of Siderocalin/Lipocalin 2/NGAL for siderophores and related small-molecule ligands. *Journal of Structural Biology: X, 2*, 100008.
- Cobe, H. M. (1954). Transitory bacteremia. Oral Surgery, Oral Medicine, and Oral Pathology, 7(6), 609-615.
- Cole, A. M., Thapa, D. R., Gabayan, V., Liao, H.-I., Liu, L., & Ganz, T. (2005). Decreased clearance of *Pseudomonas aeruginosa* from airways of mice deficient in lysozyme M. *Journal of Leukocyte Biology*, 78(5), 1081–1085.
- Collin-Faure, V., Vitipon, M., Torres, A., Tanyeres, O., Dalzon, B., & Rabilloud, T. (2023). The internal dose makes the poison: Higher internalization of polystyrene particles induce increased perturbation of macrophages. *Frontiers in Immunology*, *14*, 1092743.
- Contreras, H., Chim, N., Credali, A., & Goulding, C. W. (2014). Heme uptake in bacterial pathogens. *Current Opinion in Chemical Biology*, *19*, 34–41.

- Corbett, A. J., Eckle, S. B. G., Birkinshaw, R. W., Liu, L., Patel, O., Mahony, J., Chen, Z., Reantragoon, R., Meehan, B., Cao, H., Williamson, N. A., Strugnell, R. A., Van Sinderen, D., Mak, J. Y. W., Fairlie, D. P., Kjer-Nielsen, L., Rossjohn, J., & McCluskey, J. (2014). T-cell activation by transitory neo-antigens derived from distinct microbial pathways. *Nature*, 509(7500), 361–365.
- Coutinho, A., Kazatchkine, M. D., & Avrameas, S. (1995). Natural autoantibodies. *Current Opinion in Immunology*, 7(6), 812–818.
- Coverdale, J. P. C., Khazaipoul, S., Arya, S., Stewart, A. J., & Blindauer, C. A. (2019). Crosstalk between zinc and free fatty acids in plasma. *Biochimica et Biophysica Acta Molecular and Cellular Biology of Lipids*, *1864*(4), 532–542.
- Cowland, J. B., & Borregaard, N. (2016). Granulopoiesis and granules of human neutrophils. *Immunological Reviews*, 273(1), 11–28.
- Crowe, H. M. (1987). Invasive *Haemophilus influenzae* disease in adults. *Archives of Internal Medicine*, 147(2), 241.
- Dabral, D., & van den Bogaart, G. (2021). The roles of phospholipase A2 in phagocytes. *Frontiers in Cell and Developmental Biology*, *9*, 673502.
- Davidsson, L., Lönnerdal, B., Sandström, B., Kunz, C., & Keen, C. L. (1989). Identification of transferrin as the major plasma carrier protein for manganese introduced orally or intravenously or after *in vitro* addition in the rat. *Journal of Nutrition*, *119*(10), 1461–1464.
- Davis, K. M., Nakamura, S., & Weiser, J. N. (2011). Nod2 sensing of lysozyme-digested peptidoglycan promotes macrophage recruitment and clearance of *S. pneumoniae* colonization in mice. *The Journal of Clinical Investigation*, 121(9), 3666–3676.
- Dean, S. N., Bishop, B. M., & van Hoek, M. L. (2011a). Natural and synthetic cathelicidin peptides with anti-microbial and anti-biofilm activity against *Staphylococcus aureus*. *BMC Microbiology*, *11*(1), 114.
- Dean, S. N., Bishop, B. M., & van Hoek, M. L. (2011b). Susceptibility of *Pseudomonas aeruginosa* Biofilm to Alpha-Helical Peptides: D-enantiomer of LL-37. *Frontiers in Microbiology*, *2*, 128.
- Deitch, E. A. (2002). Bacterial translocation or lymphatic drainage of toxic products from the gut: What is important in human beings? *Surgery*, *131*(3), 241–244.
- Delano, M. J., & Ward, P. A. (2016). The immune system's role in sepsis progression, resolution, and long-term outcome. *Immunological Reviews*, 274(1), 330–353.
- de Leeuw, E., Li, C., Zeng, P., Li, C., Diepeveen-de Buin, M., Lu, W.-Y., Breukink, E., & Lu, W. (2010). Functional interaction of human neutrophil peptide-1 with the cell wall precursor lipid II. *FEBS Letters*, *584*(8), 1543–1548.
- Del Giudice, M., & Gangestad, S. W. (2018). Rethinking IL-6 and CRP: Why they are more than inflammatory biomarkers, and why it matters. *Brain, Behavior, and Immunity*, *70*, 61–75.
- Demeure, C. E., Dussurget, O., Mas Fiol, G., Le Guern, A.-S., Savin, C., & Pizarro-Cerdá, J. (2019). Yersinia pestis and plague: An updated view on evolution, virulence determinants, immune subversion, vaccination, and diagnostics. Genes and Immunity, 20(5), 357–370.

- Derde, M., Lechevalier, V., Guérin-Dubiard, C., Cochet, M.-F., Jan, S., Baron, F., Gautier, M., Vié, V., & Nau, F. (2013). Hen egg white lysozyme permeabilizes *Escherichia coli* outer and inner membranes. *Journal of Agricultural and Food Chemistry*, 61(41), 9922–9929.
- Derde, M., Nau, F., Guérin-Dubiard, C., Lechevalier, V., Paboeuf, G., Jan, S., Baron, F., Gautier, M., & Vié, V. (2015). Native and dry-heated lysozyme interactions with membrane lipid monolayers: Lipid packing modifications of a phospholipid mixture, model of the *Escherichia coli* cytoplasmic membrane. *Biochimica et Biophysica Acta*, 1848(4), 1065–1073.
- Deshmukh, H. S., Hamburger, J. B., Ahn, S. H., McCafferty, D. G., Yang, S. R., & Fowler, V. G., Jr. (2009). Critical role of NOD2 in regulating the immune response to *Staphylococcus aureus*. *Infection and Immunity*, 77(4), 1376-1382.
- Dill, B. D., Gierlinski, M., Härtlova, A., Arandilla, A. G., Guo, M., Clarke, R. G., & Trost, M. (2015). Quantitative proteome analysis of temporally resolved phagosomes following uptake via key phagocytic receptors. *Molecular & Cellular Proteomics: MCP*, 14(5), 1334–1349.
- Dintzis, H. M., Symer, D. E., Dintzis, R. Z., Zawadzke, L. E., & Berg, J. M. (1993). A comparison of the immunogenicity of a pair of enantiomeric proteins. *Proteins*, 16(3), 306–308.
- Divangahi, M., Mostowy, S., Coulombe, F., Kozak, R., Guillot, L., Veyrier, F., Kobayashi, K. S., Flavell, R. A., Gros, P., & Behr, M. A. (2008). NOD2-deficient mice have impaired resistance to *Mycobacterium tuberculosis* infection through defective innate and adaptive immunity. *The Journal of Immunology*, 181(10), 7157–7165.
- Doorduijn, D. J., Rooijakkers, S. H. M., & Heesterbeek, D. A. C. (2019). How the membrane attack complex damages the bacterial cell envelope and kills *Gram*-negative bacteria. *BioEssays*, *41*(10), e1900074.
- Dufrêne, Y. F. (2015). Sticky microbes: Forces in microbial cell adhesion. Trends in Microbiology, 23(6), 376-382.
- Dunkelberger, J. R., & Song, W.-C. (2010). Complement and its role in innate and adaptive immune responses. *Cell Research*, 20(1), 34–50.
- Dunne, M. W. (2002). Bacterial adhesion: Seen any good biofilms lately? *Clinical Microbiology Reviews*, 15(2), 155.
- EBI Web Services. (n.d.). *IPD-IMGT/HLA*. Retrieved October 17, 2024, from https://www.ebi.ac.uk/ipd/imgt/hla/about/statistics/
- Eichner, M., Protze, J., Piontek, A., Krause, G., & Piontek, J. (2017). Targeting and alteration of tight junctions by bacteria and their virulence factors such as *Clostridium perfringens* enterotoxin. *Pflügers Archiv: European Journal of Physiology*, 469(1), 77–90.
- Fagarasan, S., Kawamoto, S., Kanagawa, O., & Suzuki, K. (2010). Adaptive immune regulation in the gut: T cell-dependent and T cell-independent IgA synthesis. *Annual Review of Immunology*, *28*(1), 243–273.
- Feng, C. G., Scanga, C. A., Collazo-Custodio, C. M., Cheever, A. W., Hieny, S., Caspar, P., & Sher, A. (2003). Mice lacking myeloid differentiation factor 88 display profound defects in host resistance and immune responses to *Mycobacterium avium* infection not exhibited by Toll-like receptor 2 (TLR2)- and TLR4-deficient animals. *The Journal of Immunology*, *171*(9), 4758–4764.
- Fine, R. L., Manfredo Vieira, S., Gilmore, M. S., & Kriegel, M. A. (2020). Mechanisms and consequences of gut commensal translocation in chronic diseases. *Gut Microbes*, 11(2), 217–230.

- Firon, N., Ofek, I., & Sharon, N. (1983). Carbohydrate specificity of the surface lectins of *Escherichia coli*, *Klebsiella pneumoniae*, and *Salmonella typhimurium*. *Carbohydrate Research*, 120, 235–249.
- Fischbach, M. A., Lin, H., Liu, D. R., & Walsh, C. T. (2006). How pathogenic bacteria evade mammalian sabotage in the battle for iron. *Nature Chemical Biology*, *2*(3), 132–138.
- Fleischmann-Struzek, C., Goldfarb, D. M., Schlattmann, P., Schlapbach, L. J., Reinhart, K., & Kissoon, N. (2018). The global burden of paediatric and neonatal sepsis: A systematic review. *The Lancet Respiratory Medicine*, 6(3), 223–230.
- Flo, T. H., Smith, K. D., Sato, S., Rodriguez, D. J., Holmes, M. A., Strong, R. K., Akira, S., & Aderem, A. (2004). Lipocalin 2 mediates an innate immune response to bacterial infection by sequestrating iron. *Nature*, 432(7019), 917–921.
- Füchslin, H. P., Schneider, C., & Egli, T. (2012). In glucose-limited continuous culture the minimum substrate concentration for growth, Smin, is crucial in the competition between the enterobacterium *Escherichia coli* and *Chelatobacter heintzii*, an environmentally abundant bacterium. *The ISME Journal*, 6(4), 777–789.
- Furman, D., Campisi, J., Verdin, E., Carrera-Bastos, P., Targ, S., Franceschi, C., Ferrucci, L., Gilroy, D. W., Fasano, A., Miller, G. W., Miller, A. H., Mantovani, A., Weyand, C. M., Barzilai, N., Goronzy, J. J., Rando, T. A., Effros, R. B., Lucia, A., Kleinstreuer, N., & Slavich, G. M. (2019). Chronic inflammation in the etiology of disease across the life span. *Nature Medicine*, 25(12), 1822–1832.
- Fusco, W., Lorenzo, M. B., Cintoni, M., Porcari, S., Rinninella, E., Kaitsas, F., Lener, E., Mele, M. C., Gasbarrini, A., Collado, M. C., Cammarota, G., & Ianiro, G. (2023). Short-chain fatty-acid-producing bacteria: Key components of the human gut microbiota. *Nutrients*, 15(9), 2211.
- Fu, Y. L., & Harrison, R. E. (2021). Microbial phagocytic receptors and their potential involvement in cytokine induction in macrophages. *Frontiers in Immunology*, 12, 662063.
- Gade, M., Chaudhary, P. M., Thulasiram, H. V., & Kikkeri, R. (2017). Engineering cell surface glycans with carbohydrate enantiomers to alter bacterial binding and adhesion. *ChemistrySelect*, 2(28), 8865–8869.
- Gallo, R. L., & Hooper, L. V. (2012). Epithelial antimicrobial defence of the skin and intestine. Nature Reviews Immunology, 12(7), 503–516.
- Ganz, T., Gabayan, V., Liao, H.-I., Liu, L., Oren, A., Graf, T., & Cole, A. M. (2003). Increased inflammation in lysozyme M-deficient mice in response to *Micrococcus luteus* and its peptidoglycan. *Blood*, 101(6), 2388–2392.
- Gao, J. L., Lee, E. J., & Murphy, P. M. (1999). Impaired antibacterial host defense in mice lacking the N-formylpeptide receptor. *The Journal of Experimental Medicine*, *189*(4), 657–662.
- Gawrysiak, M., Gajewski, A., Szewczyk, R., Likońska, A., Michlewska, S., Chmiela, M., Kowalski, M. L., & Chałubiński, M. (2021). Human rhinovirus HRV16 impairs barrier functions and regeneration of human lung vascular endothelium. *Allergy*, 76(6), 1872–1875.
- Gebert, A., Rothkötter, H. J., & Pabst, R. (1996). M cells in Peyer's patches of the intestine. *International Review of Cytology*, *167*, 91–159.
- Geiger, O., López-Lara, I. M., & Sohlenkamp, C. (2013). Phosphatidylcholine biosynthesis and function in bacteria. *Biochimica et Biophysica Acta*, 1831(3), 503–513.

- Gerlini, A., Colomba, L., Furi, L., Braccini, T., Manso, A. S., Pammolli, A., Wang, B., Vivi, A., Tassini, M., van Rooijen, N., Pozzi, G., Ricci, S., Andrew, P. W., Koedel, U., Moxon, E. R., & Oggioni, M. R. (2014). The role of host and microbial factors in the pathogenesis of pneumococcal bacteraemia arising from a single bacterial cell bottleneck. *PLoS Pathogens*, 10(3), e1004026.
- Gill, T. J., III, Gould, H. J., & Doty, P. (1963). Role of optical isomers in determining the antigenicity of synthetic polypeptides. *Nature*, *197*(4869), 746–747.
- Gill, T. J., III, Kunz, H. W., Gould, H. J., & Doty, P. (1964). Studies on synthetic polypeptide antigens. Xi. The antigenicity of optically isomeric synthetic polypeptides. *The Journal of Biological Chemistry*, 239, 1107–1113.
- Gill, T. J., III, Kunz, H. W., & Papermaster, D. S. (1967). Studies on synthetic polypeptide antigens. 18. The role of composition, charge, and optical isomerism in the immunogenicity of synthetic polypeptides. *The Journal of Biological Chemistry*, 242(14), 3308–3318.
- Goetz, D. H., Holmes, M. A., Borregaard, N., Bluhm, M. E., Raymond, K. N., & Strong, R. K. (2002). The neutrophil lipocalin NGAL is a bacteriostatic agent that interferes with siderophore-mediated iron acquisition. *Molecular Cell*, 10(5), 1033–1043.
- Gong, T., Liu, L., Jiang, W., & Zhou, R. (2020). DAMP-sensing receptors in sterile inflammation and inflammatory diseases. *Nature Reviews Immunology*, 20(2), 95–112.
- Gorden, J., & Small, P. L. (1993). Acid resistance in enteric bacteria. Infection and Immunity, 61(1), 364–367.
- Gordon, S. (2016). Phagocytosis: An immunobiologic process. Immunity, 44(3), 463-475.
- Gotts, J. E., & Matthay, M. A. (2016). Sepsis: Pathophysiology and clinical management. BMJ, 353, i1585.
- Grabenstein, J. D., & Manoff, S. B. (2012). Pneumococcal polysaccharide 23-valent vaccine: long-term persistence of circulating antibody and immunogenicity and safety after revaccination in adults. *Vaccine*, 30(30), 4435–4444.
- Gras, S., Van Rhijn, I., Shahine, A., & Le Nours, J. (2018). Molecular recognition of microbial lipid-based antigens by T cells. *Cellular and Molecular Life Sciences: CMLS*, 75(9), 1623–1639.
- Greenberg, J. A., Hohmann, S. F., James, B. D., Shah, R. C., Hall, J. B., Kress, J. P., & David, M. Z. (2018). Hospital volume of immunosuppressed patients with sepsis and sepsis mortality. *Annals of the American Thoracic Society*, 15(8), 962–969.
- Guan, Q., Jiang, J., Huang, Y., Wang, Q., Liu, Z., Ma, X., Yang, X., Li, Y., Wang, S., Cui, W., Tang, J., Wan, H., Xu, Q., Tu, Y., Wu, D., & Xia, Y. (2023). The landscape of micron-scale particles including microplastics in human enclosed body fluids. *Journal of Hazardous Materials*, 442, 130138.
- Gudmand, M., Rocha, S., Hatzakis, N. S., Peneva, K., Müllen, K., Stamou, D., Uji-I, H., Hofkens, J., Bjørnholm, T., & Heimburg, T. (2010). Influence of lipid heterogeneity and phase behavior on phospholipase A2 action at the single molecule level. *Biophysical Journal*, *98*(9), 1873–1882.
- Guerry, P. (2007). Campylobacter flagella: Not just for motility. Trends in Microbiology, 15(10), 456-461.

- Guichard, G., Benkirane, N., Zeder-Lutz, G., van Regenmortel, M. H., Briand, J. P., & Muller, S. (1994). Antigenic mimicry of natural L-peptides with retro-inverso-peptidomimetics. *Proceedings of the National Academy of Sciences of the United States of America*, 91(21), 9765–9769.
- Guo, R.-F., & Ward, P. A. (2005). Role of C5a in inflammatory responses. *Annual Review of Immunology*, 23(1), 821–852.
- Hanna, S., & Etzioni, A. (2014). MHC class I and II deficiencies. *The Journal of Allergy and Clinical Immunology*, 134(2), 269–275.
- Hardy, R. R., & Hayakawa, K. (1994). CD5 B cells, a fetal B cell lineage. Advances in Immunology, 55, 297-339.
- Heesterbeek, D. A. C., Angelier, M. L., Harrison, R. A., & Rooijakkers, S. H. M. (2018). Complement and bacterial infections: From molecular mechanisms to therapeutic applications. *Journal of Innate Immunity*, 10(5-6), 455–464.
- He, H.-Q., & Ye, R. D. (2017). The formyl peptide receptors: Diversity of ligands and mechanism for recognition. *Molecules*, 22(3), 455.
- Heijmans, C. M. C., de Groot, N. G., & Bontrop, R. E. (2020). Comparative genetics of the major histocompatibility complex in humans and nonhuman primates. *International Journal of Immunogenetics*, 47(3), 243–260.
- Hellings, P. W., & Steelant, B. (2020). Epithelial barriers in allergy and asthma. *The Journal of Allergy and Clinical Immunology*, 145(6), 1499–1509.
- Herb, M., & Schramm, M. (2021). Functions of ROS in macrophages and antimicrobial immunity. *Antioxidants*, *10*(2), 313.
- Hervé, M., Maillere, B., Mourier, G., Texier, C., Leroy, S., & Ménez, A. (1997). On the immunogenic properties of retro-inverso peptides. Total retro-inversion of T-cell epitopes causes a loss of binding to MHC II molecules. *Molecular Immunology*, 34(2), 157–163.
- Heuston, S., Begley, M., Gahan, C. G. M., & Hill, C. (2012). Isoprenoid biosynthesis in bacterial pathogens. *Microbiology (Reading, England)*, 158(Pt 6), 1389–1401.
- Hill, C. M., Liu, A., Marshall, K. W., Mayer, J., Jorgensen, B., Yuan, B., Cubbon, R. M., Nichols, E. A., Wicker, L. S., & Rothbard, J. B. (1994). Exploration of requirements for peptide binding to HLA DRB1\*0101 and DRB1\*0401. *The Journal of Immunology*, 152(6), 2890–2898.
- Hoffmann, E., Kotsias, F., Visentin, G., Bruhns, P., Savina, A., & Amigorena, S. (2012). Autonomous phagosomal degradation and antigen presentation in dendritic cells. *Proceedings of the National Academy of Sciences of the United States of America*, 109(36), 14556–14561.
- Hollands, A., Pence, M. A., Timmer, A. M., Osvath, S. R., Turnbull, L., Whitchurch, C. B., Walker, M. J., & Nizet, V. (2010). Genetic switch to hypervirulence reduces colonization phenotypes of the globally disseminated group A streptococcus M1T1 clone. *The Journal of Infectious Diseases*, 202(1), 11–19.
- Hotchkiss, R. S., Monneret, G., & Payen, D. (2013). Sepsis-induced immunosuppression: From cellular dysfunctions to immunotherapy. *Nature Reviews Immunology*, 13(12), 862–874.
- Ibrahim, H. R., Matsuzaki, T., & Aoki, T. (2001). Genetic evidence that antibacterial activity of lysozyme is independent of its catalytic function. *FEBS Letters*, *506*(1), 27–32.

- Ishida, T., Fujihara, N., Nishimura, T., Funabashi, H., Hirota, R., Ikeda, T., & Kuroda, A. (2019). Live-cell imaging of macrophage phagocytosis of asbestos fibers under fluorescence microscopy. *Genes and Environment*, 41(1), 14.
- Islam, A., Actis, L. A., & Wilson, T. J. (2023). Natural antibodies mediate protection against Acinetobacter baumannii respiratory infections. The Journal of Infectious Diseases, 228(3), 353–363.
- Iwasaki, A., & Medzhitov, R. (2015). Control of adaptive immunity by the innate immune system. *Nature Immunology*, 16(4), 343–353.
- Jakubzick, C. V., Randolph, G. J., & Henson, P. M. (2017). Monocyte differentiation and antigen-presenting functions. *Nature Reviews Immunology*, 17(6), 349–362.
- Janeway, C. A., Jr. (1989). Approaching the asymptote? Evolution and revolution in immunology. *Cold Spring Harbor Symposia on Quantitative Biology*, 54 Pt 1, 1–13.
- Janeway, C. A., Jr, & Sela, M. (1967). Synthetic antigens composed exclusively of L- or D-amino acids. I. Effect of optical configuration on the immunogenicity of synthetic polypeptides in mice. *Immunology*, 13(1), 29–38.
- Jarczak, D., Kluge, S., & Nierhaus, A. (2021). Sepsis-pathophysiology and therapeutic concepts. *Frontiers in Medicine*, *8*, 628302.
- Jin, M. S., Kim, S. E., Heo, J. Y., Lee, M. E., Kim, H. M., Paik, S.-G., Lee, H., & Lee, J.-O. (2007). Crystal structure of the TLR1-TLR2 heterodimer induced by binding of a tri-acylated lipopeptide. *Cell*, 130(6), 1071–1082.
- Jin, S., Wetzel, D., & Schirmer, M. (2022). Deciphering mechanisms and implications of bacterial translocation in human health and disease. *Current Opinion in Microbiology*, 67, 102147.
- Ji, S.-R., Zhang, S.-H., Chang, Y., Li, H.-Y., Wang, M.-Y., Lv, J.-M., Zhu, L., Tang, P. M. K., & Wu, Y. (2023). C-reactive protein: The most familiar stranger. *The Journal of Immunology*, 210(6), 699–707.
- Johnson, D. R., Stevens, D. L., & Kaplan, E. L. (1992). Epidemiologic analysis of group A streptococcal serotypes associated with severe systemic infections, rheumatic fever, or uncomplicated pharyngitis. *The Journal of Infectious Diseases*, 166(2), 374–382.
- Joiner, K. A., Brown, E. J., & Frank, M. M. (1984). Complement and bacteria: Chemistry and biology in host defense. Annual Review of Immunology, 2(1), 461–491.
- Jorgensen, I., Rayamajhi, M., & Miao, E. A. (2017). Programmed cell death as a defence against infection. *Nature Reviews Immunology*, *17*(3), 151–164.
- Jung, G. (1992). Proteins from the D-chiral world. *Angewandte Chemie International Edition in English*, 31(11), 1457–1459.
- Kantor, A. B., & Herzenberg, L. A. (1993). Origin of murine B cell lineages. *Annual Review of Immunology*, *11*(1), 501–538.
- Kappler, K., & Hennet, T. (2020). Emergence and significance of carbohydrate-specific antibodies. *Genes and Immunity*, 21(4), 224–239.
- Kawai, T., & Akira, S. (2010). The role of pattern-recognition receptors in innate immunity: Update on Toll-like receptors. *Nature Immunology*, 11(5), 373–384.

- Kawasaki, K., Ernst, R. K., & Miller, S. I. (2004). 3-O-deacylation of lipid A by PagL, a PhoP/PhoQ-regulated deacylase of *Salmonella typhimurium*, modulates signaling through Toll-like receptor 4. *The Journal of Biological Chemistry*, 279(19), 20044–20048.
- Kellum, J. A., & Ronco, C. (2023). The role of endotoxin in septic shock. Critical Care, 27(1), 400.
- Khademian, M., & Imlay, J. A. (2021). How microbes evolved to tolerate oxygen. *Trends in Microbiology*, 29(5), 428–440.
- Khan, A., & Jia, Z. (2023). Recent insights into uptake, toxicity, and molecular targets of microplastics and nanoplastics relevant to human health impacts. *iScience*, *26*(2), 106061.
- Kim, B. E., & Leung, D. Y. M. (2018). Significance of skin barrier dysfunction in atopic dermatitis. Allergy, Asthma & Immunology Research, 10(3), 207–215.
- Kim, M., Ashida, H., Ogawa, M., Yoshikawa, Y., Mimuro, H., & Sasakawa, C. (2010). Bacterial interactions with the host epithelium. *Cell Host & Microbe*, 8(1), 20–35.
- Kim, S.-H., Cho, B.-H., Kiyono, H., & Jang, Y.-S. (2017). Microbiota-derived butyrate suppresses group 3 innate lymphoid cells in terminal ileal Peyer's patches. *Scientific Reports*, 7(1), 3980.
- Kim, Y.-G., Park, J.-H., Shaw, M. H., Franchi, L., Inohara, N., & Núñez, G. (2008). The cytosolic sensors Nod1 and Nod2 are critical for bacterial recognition and host defense after exposure to Toll-like receptor ligands. *Immunity*, 28(2), 246–257.
- King, T. P., Wade, D., Coscia, M. R., Mitchell, S., Kochoumian, L., & Merrifield, B. (1994). Structure-immunogenicity relationship of melittin, its transposed analogues, and D-melittin. *The Journal of Immunology*, 153(3), 1124–1131.
- Kirsipuu, T., Zadorožnaja, A., Smirnova, J., Friedemann, M., Plitz, T., Tõugu, V., & Palumaa, P. (2020). Copper(II)-binding equilibria in human blood. *Scientific Reports*, *10*(1), 5686.
- Kleetz, J., Vasilopoulos, G., Czolkoss, S., Aktas, M., & Narberhaus, F. (2021). Recombinant and endogenous ways to produce methylated phospholipids in *Escherichia coli*. *Applied Microbiology and Biotechnology*, 105(23), 8837–8851.
- König, J., Wells, J., Cani, P. D., García-Ródenas, C. L., MacDonald, T., Mercenier, A., Whyte, J., Troost, F., & Brummer, R.-J. (2016). Human intestinal barrier function in health and disease. *Clinical and Translational Gastroenterology*, 7(10), e196.
- Korn, T., Bettelli, E., Oukka, M., & Kuchroo, V. K. (2009). IL-17 and Th17 cells. *Annual Review of Immunology*, 27(1), 485–517.
- Kreve, S., & Dos Reis, A. C. D. (2021). Bacterial adhesion to biomaterials: What regulates this attachment? A review. *The Japanese Dental Science Review*, 57, 85–96.
- Kristinsson, S. Y., Gridley, G., Hoover, R. N., Check, D., & Landgren, O. (2014). Long-term risks after splenectomy among 8,149 cancer-free American veterans: a cohort study with up to 27 years follow-up. *Haematologica*, 99(2), 392–398.
- Krueger, J., Ray, A., Tamm, I., & Sehgal, P. B. (1991). Expression and function of interleukin-6 in epithelial cells. *Journal of Cellular Biochemistry*, 45(4), 327–334.

- Kubagawa, H., Oka, S., Kubagawa, Y., Torii, I., Takayama, E., Kang, D.-W., Gartland, G. L., Bertoli, L. F., Mori, H., Takatsu, H., Kitamura, T., Ohno, H., & Wang, J.-Y. (2009). Identity of the elusive IgM Fc receptor (FcmuR) in humans. *The Journal of Experimental Medicine*, 206(12), 2779–2793.
- Ku, C.-L., von Bernuth, H., Picard, C., Zhang, S.-Y., Chang, H.-H., Yang, K., Chrabieh, M., Issekutz, A. C., Cunningham, C. K., Gallin, J., Holland, S. M., Roifman, C., Ehl, S., Smart, J., Tang, M., Barrat, F. J., Levy, O., McDonald, D., Day-Good, N. K., ... Casanova, J.-L. (2007). Selective predisposition to bacterial infections in IRAK-4-deficient children: IRAK-4-dependent TLRs are otherwise redundant in protective immunity. *The Journal of Experimental Medicine*, 204(10), 2407–2422.
- Kumar, G., Ahmad, S., Taneja, A., Patel, J., Guddati, A. K., Nanchal, R., & Milwaukee Initiative in Critical Care Outcomes Research Group of Investigators. (2015). Severe sepsis in hematopoietic stem cell transplant recipients. *Critical Care Medicine*, 43(2), 411–421.
- Kwaan, H. C., & Bongu, A. (1999). The hyperviscosity syndromes. *Seminars in Thrombosis and Hemostasis*, 25(2), 199–208.
- Lacey, C. A., & Miao, E. A. (2020). Programmed cell death in the evolutionary race against bacterial virulence factors. *Cold Spring Harbor Perspectives in Biology*, 12(2), a036459.
- Lalbiaktluangi, C., Yadav, M. K., Singh, P. K., Singh, A., Iyer, M., Vellingiri, B., Zomuansangi, R., Zothanpuia, & Ram, H. (2023). A cooperativity between virus and bacteria during respiratory infections. *Frontiers in Microbiology*, 14, 1279159.
- Lam, N., Lee, Y., & Farber, D. L. (2024). A guide to adaptive immune memory. *Nature Reviews Immunology*, 24(11), 810–829.
- Lander, A. J., Jin, Y., & Luk, L. Y. P. (2023). D-peptide and D-protein technology: Recent advances, challenges, and opportunities. *Chembiochem: A European Journal of Chemical Biology*, 24(4), e202200537.
- Laube, D. M., Yim, S., Ryan, L. K., Kisich, K. O., & Diamond, G. (2006). Antimicrobial peptides in the airway. *Current Topics in Microbiology and Immunology*, 306, 153–182.
- Law, S. K., & Dodds, A. W. (1997). The internal thioester and the covalent binding properties of the complement proteins C3 and C4. *Protein Science*, *6*(2), 263–274.
- Leslie, H. A., van Velzen, M. J. M., Brandsma, S. H., Vethaak, A. D., Garcia-Vallejo, J. J., & Lamoree, M. H. (2022). Discovery and quantification of plastic particle pollution in human blood. *Environment International*, 163, 107199.
- Levin, B. R., & Antia, R. (2001). Why we don't get sick: The within-wost population dynamics of bacterial infections. *Science*, *292*(5519), 1112–1115.
- Lewinsohn, D. A., Gold, M. C., & Lewinsohn, D. M. (2011). Views of immunology: effector T cells. *Immunological Reviews*, 240(1), 25–39.
- Lin, Q., Lin, S., Fan, Z., Liu, J., Ye, D., & Guo, P. (2024). A review of the mechanisms of bacterial colonization of the mammal gut. *Microorganisms*, 12(5), 1026.
- Liu, M., Chen, K., Yoshimura, T., Liu, Y., Gong, W., Wang, A., Gao, J.-L., Murphy, P. M., & Wang, J. M. (2012). Formylpeptide receptors are critical for rapid neutrophil mobilization in host defense against *Listeria monocytogenes*. *Scientific Reports*, 2(1), 786.

- Lombardi, A., Concepcion, E., Hou, H., Arib, H., Mezei, M., Osman, R., & Tomer, Y. (2020). Retro-inverso D-peptides as a novel targeted immunotherapy for Type 1 diabetes. *Journal of Autoimmunity*, *115*, 102543.
- Lovett, A., & Duncan, J. A. (2018). Human immune responses and the natural history of *Neisseria gonorrhoeae* infection. *Frontiers in Immunology*, *9*, 3187.
- Lu, L. L., Suscovich, T. J., Fortune, S. M., & Alter, G. (2018). Beyond binding: antibody effector functions in infectious diseases. *Nature Reviews Immunology*, 18(1), 46–61.
- Malhotra, R., Dhawan, B., Garg, B., Shankar, V., & Nag, T. C. (2019). A comparison of bacterial adhesion and biofilm formation on commonly used orthopaedic metal implant materials: An *in vitro* study. *Indian Journal of Orthopaedics*, *53*(1), 148–153.
- Marfella, R., Prattichizzo, F., Sardu, C., Fulgenzi, G., Graciotti, L., Spadoni, T., D'Onofrio, N., Scisciola, L., La Grotta, R., Frigé, C., Pellegrini, V., Municinò, M., Siniscalchi, M., Spinetti, F., Vigliotti, G., Vecchione, C., Carrizzo, A., Accarino, G., Squillante, A., ... Paolisso, G. (2024). Microplastics and nanoplastics in atheromas and cardiovascular events. *The New England Journal of Medicine*, 390(10), 900–910.
- Marinec, P. S., Landgraf, K. E., Uppalapati, M., Chen, G., Xie, D., Jiang, Q., Zhao, Y., Petriello, A., Deshayes, K., Kent, S. B. H., Ault-Riche, D., & Sidhu, S. S. (2021). A non-immunogenic bivalent d-protein potently inhibits retinal vascularization and tumor growth. ACS Chemical Biology, 16(3), 548–556.
- Markart, P., Korfhagen, T. R., Weaver, T. E., & Akinbi, H. T. (2004). Mouse lysozyme M is important in pulmonary host defense against *Klebsiella pneumoniae* infection. *American Journal of Respiratory and Critical Care Medicine*, 169(4), 454–458.
- Marks, C., & Deane, C. M. (2020). How repertoire data are changing antibody science. *The Journal of Biological Chemistry*, 295(29), 9823–9837.
- Marshall, J. S., Warrington, R., Watson, W., & Kim, H. L. (2018). An introduction to immunology and immunopathology. Allergy, Asthma, and Clinical Immunology: Official Journal of the Canadian Society of Allergy and Clinical Immunology, 14(Suppl 2), 49.
- Martínez-Meléndez, A., Morfin-Otero, R., Villarreal-Treviño, L., Baines, S. D., Camacho-Ortíz, A., & Garza-González, E. (2022). Analysis of biofilm production and expression of adhesion structures of circulating *Clostridioides difficile* strains from Mexico. *Enfermedades Infecciosas y Microbiología Clínica*, 40(8), 445–448.
- Mastellos, D. C., Hajishengallis, G., & Lambris, J. D. (2024). A guide to complement biology, pathology and therapeutic opportunity. *Nature Reviews Immunology*, 24(2), 118–141.
- Matter, K., & Balda, M. S. (2003). Functional analysis of tight junctions. *Methods*, 30(3), 228-234.
- Maurer, P. H. (1963). Antigenicity of polypeptides (poly alpha amino acids). X. Studies with polymers of D amino acids. *Proceedings of the Society for Experimental Biology and Medicine. Society for Experimental Biology and Medicine*, 113, 553–557.
- Maurer, P. H. (1965). Antigenicity of polypeptides (poly alpha amino acids). 13. Immunological studies with synthetic polymers containing only D- or D- and L-alpha-amino acids. *The Journal of Experimental Medicine*, *121*, 339–349.

- McGuckin, M. A., Lindén, S. K., Sutton, P., & Florin, T. H. (2011). Mucin dynamics and enteric pathogens. *Nature Reviews Microbiology*, 9(4), 265–278.
- McWilliam, H. E. G., & Villadangos, J. A. (2024). MR1 antigen presentation to MAIT cells and other MR1-restricted T cells. *Nature Reviews Immunology*, 24(3), 178–192.
- Meldal, M., Svendsen, I. B., Juliano, L., Juliano, M. A., Nery, E. D., & Scharfstein, J. (1998). Inhibition of cruzipain visualized in a fluorescence quenched solid-phase inhibitor library assay. D-amino acid inhibitors for cruzipain, cathepsin B and cathepsin L. *Journal of Peptide Science*, 4(2), 83–91.
- Merle, N. S., Church, S. E., Fremeaux-Bacchi, V., & Roumenina, L. T. (2015). Complement system part I—Molecular mechanisms of activation and regulation. *Frontiers in Immunology*, 6, 262.
- Metzger, J., Jung, G., Bessler, W. G., Hoffmann, P., Strecker, M., Lieberknecht, A., & Schmidt, U. (1991). Lipopeptides containing 2-(palmitoylamino)-6,7-bis(palmitoyloxy) heptanoic acid: synthesis, stereospecific stimulation of B-lymphocytes and macrophages, and adjuvanticity *in vivo* and *in vitro*. *Journal of Medicinal Chemistry*, 34(7), 1969–1974.
- Mézière, C., Viguier, M., Dumortier, H., Lo-Man, R., Leclerc, C., Guillet, J. G., Briand, J. P., & Muller, S. (1997). *In vivo* T helper cell response to retro-inverso peptidomimetics. *The Journal of Immunology*, *159*(7), 3230–3237.
- Miethke, M., & Marahiel, M. A. (2007). Siderophore-based iron acquisition and pathogen control. *Microbiology and Molecular Biology Reviews*, *71*(3), 413–451.
- Miller, S. M., Simon, R. J., Ng, S., Zuckermann, R. N., Kerr, J. M., & Moos, W. H. (1995). Comparison of the proteolytic susceptibilities of homologous L-amino acid, D-amino acid, and N-substituted glycine peptide and peptoid oligomers. *Drug Development Research*, 35(1), 20–32.
- Mills, K. H. G. (2023). IL-17 and IL-17-producing cells in protection versus pathology. *Nature Reviews Immunology*, 23(1), 38–54.
- Milo, R., & Phillips, R. (2015). Cell Biology by the Numbers. Garland Science.
- Mold, C., Rodic-Polic, B., & Du Clos, T. W. (2002). Protection from *Streptococcus pneumoniae* infection by C-reactive protein and natural antibody requires complement but not Fc gamma receptors. *The Journal of Immunology*, 168(12), 6375–6381.
- Monack, D. M., Mueller, A., & Falkow, S. (2004). Persistent bacterial infections: The interface of the pathogen and the host immune system. *Nature Reviews Microbiology*, *2*(9), 747–765.
- Mori, L., Lepore, M., & De Libero, G. (2016). The immunology of CD1- and MR1-restricted T cells. *Annual Review of Immunology*, *34*(1), 479–510.
- Morris, D. E., Cleary, D. W., & Clarke, S. C. (2017). Secondary bacterial infections associated with influenza pandemics. *Frontiers in Microbiology*, 8, 1041.
- Mosser, D. M., & Edwards, J. P. (2008). Exploring the full spectrum of macrophage activation. *Nature Reviews Immunology*, 8(12), 958–969.
- Moxon, E. R., & Kroll, J. S. (1990). The role of bacterial polysaccharide capsules as virulence factors. *Current Topics in Microbiology and Immunology*, 150, 65–85.

- Mozes, E., Sela, M., & Taussig, M. J. (1974). Tolerance to thymus-independent antigens. Characteristics of induction of tolerance to thymus-independent synthetic polypeptides. *Immunology*, 27(4), 641–646.
- Muniz, L. R., Knosp, C., & Yeretssian, G. (2012). Intestinal antimicrobial peptides during homeostasis, infection, and disease. *Frontiers in Immunology*, *3*, 310.
- Muñoz-Wolf, N., & Lavelle, E. C. (2016). Innate immune receptors. Methods in Molecular Biology, 1417, 1-43.
- Murakami, M., Yamamoto, K., Miki, Y., Murase, R., Sato, H., & Taketomi, Y. (2016). The roles of the secreted phospholipase A2 gene family in immunology. *Advances in Immunology*, *132*, 91–134.
- Mylvaganam, S., & Freeman, S. A. (2023). The resolution of phagosomes. Immunological Reviews, 319(1), 45-64.
- Nair, D. T., Kaur, K. J., Singh, K., Mukherjee, P., Rajagopal, D., George, A., Bal, V., Rath, S., Rao, K. V. S., & Salunke, D. M. (2003). Mimicry of native peptide antigens by the corresponding retro-inverso analogs is dependent on their intrinsic structure and interaction propensities. *The Journal of Immunology*, 170(3), 1362–1373.
- Nash, J. A., Ballard, T. N. S., Weaver, T. E., & Akinbi, H. T. (2006). The peptidoglycan-degrading property of lysozyme is not required for bactericidal activity *in vivo*. *The Journal of Immunology*, *177*(1), 519–526.
- Niess, J. H., Brand, S., Gu, X., Landsman, L., Jung, S., McCormick, B. A., Vyas, J. M., Boes, M., Ploegh, H. L., Fox, J. G., Littman, D. R., & Reinecker, H.-C. (2005). CX3CR1-mediated dendritic cell access to the intestinal lumen and bacterial clearance. *Science*, 307(5707), 254–258.
- Nish, S., & Medzhitov, R. (2011). Host defense pathways: Role of redundancy and compensation in infectious disease phenotypes. *Immunity*, *34*(5), 629–636.
- Oliver, S. E., Rubis, A. B., Soeters, H. M., Reingold, A., Barnes, M., Petit, S., Moore, A. E., Harrison, L. H., Lynfield, R., Angeles, K. M., Burzlaff, K. E., Thomas, A., Schaffner, W., Marjuki, H., Wang, X., & Hariri, S. (2023). Secondary cases of invasive disease caused by encapsulated and nontypeable *Haemophilus influenzae* 10 U.S. jurisdictions, 2011-2018. *Morbidity and Mortality Weekly Report*, 72(15), 386–390.
- Ostaff, M. J., Stange, E. F., & Wehkamp, J. (2013). Antimicrobial peptides and gut microbiota in homeostasis and pathology. *EMBO Molecular Medicine*, *5*(10), 1465–1483.
- Ottemann, K. M., & Lowenthal, A. C. (2002). *Helicobacter pylori* uses motility for initial colonization and to attain robust infection. *Infection and Immunity*, 70(4), 1984–1990.
- Ott, L. W., Resing, K. A., Sizemore, A. W., Heyen, J. W., Cocklin, R. R., Pedrick, N. M., Woods, H. C., Chen, J. Y., Goebl, M. G., Witzmann, F. A., & Harrington, M. A. (2007). Tumor Necrosis Factor-alpha- and interleukin-1-induced cellular responses: coupling proteomic and genomic information. *Journal of Proteome Research*, 6(6), 2176–2185.
- Ouederni, M., Vincent, Q. B., Frange, P., Touzot, F., Scerra, S., Bejaoui, M., Bousfiha, A., Levy, Y.,
  Lisowska-Grospierre, B., Canioni, D., Bruneau, J., Debré, M., Blanche, S., Abel, L., Casanova, J.-L., Fischer,
  A., & Picard, C. (2011). Major histocompatibility complex class II expression deficiency caused by a
  RFXANK founder mutation: A survey of 35 patients. *Blood*, *118*(19), 5108–5118.
- Paczosa, M. K., Fisher, M. L., Maldonado-Arocho, F. J., & Mecsas, J. (2014). Yersinia pseudotuberculosis uses Ail and YadA to circumvent neutrophils by directing Yop translocation during lung infection: Yersinia utilizes Ail and YadA to block killing by PMNs. Cellular Microbiology, 16(2), 247–268.

- Palm, A.-K. E., & Henry, C. (2019). Remembrance of things past: Long-term B cell memory after infection and vaccination. *Frontiers in Immunology*, 10, 1787.
- Park, B. S., Song, D. H., Kim, H. M., Choi, B.-S., Lee, H., & Lee, J.-O. (2009). The structural basis of lipopolysaccharide recognition by the TLR4-MD-2 complex. *Nature*, 458(7242), 1191–1195.
- Park, M. S., Kim, J. I., Lee, I., Park, S., Bae, J.-Y., & Park, M.-S. (2018). Towards the application of human defensins as antivirals. *Biomolecules & Therapeutics*, 26(3), 242–254.
- Pat, Y., Yazici, D., D'Avino, P., Li, M., Ardicli, S., Ardicli, O., Mitamura, Y., Akdis, M., Dhir, R., Nadeau, K., Agache, I., Ogulur, I., & Akdis, C. A. (2024). Recent advances in the epithelial barrier theory. *International Immunology*, 36(5), 211–222.
- Pauwels, A.-M., Trost, M., Beyaert, R., & Hoffmann, E. (2017). Patterns, receptors, and signals: Regulation of phagosome maturation. *Trends in Immunology*, 38(6), 407–422.
- Payne, J. M., Sansom, B. F., Garner, R. J., Thomson, A. R., & Miles, B. J. (1960). Uptake of small resin particles (1-5 microns diameter) by the alimentary canal of the calf. *Nature*, 188(4750), 586–587.
- Pennock, N. D., White, J. T., Cross, E. W., Cheney, E. E., Tamburini, B. A., & Kedl, R. M. (2013). T cell responses: Naïve to memory and everything in between. *Advances in Physiology Education*, *37*(4), 273–283.
- Pettigrew, H. D., Teuber, S. S., & Gershwin, M. E. (2009). Clinical significance of complement deficiencies. Annals of the New York Academy of Sciences, 1173(1), 108–123.
- Picard, C., Casanova, J.-L., & Puel, A. (2011). Infectious diseases in patients with IRAK-4, MyD88, NEMO, or IκBα deficiency. *Clinical Microbiology Reviews*, *24*(3), 490–497.
- Picard, C., Puel, A., Bonnet, M., Ku, C.-L., Bustamante, J., Yang, K., Soudais, C., Dupuis, S., Feinberg, J., Fieschi, C., Elbim, C., Hitchcock, R., Lammas, D., Davies, G., Al-Ghonaium, A., Al-Rayes, H., Al-Jumaah, S., Al-Hajjar, S., Al-Mohsen, I. Z., ... Casanova, J.-L. (2003). Pyogenic bacterial infections in humans with IRAK-4 deficiency. *Science*, 299(5615), 2076–2079.
- Picard, C., von Bernuth, H., Ghandil, P., Chrabieh, M., Levy, O., Arkwright, P. D., McDonald, D., Geha, R. S., Takada, H., Krause, J. C., Creech, C. B., Ku, C.-L., Ehl, S., Maródi, L., Al-Muhsen, S., Al-Hajjar, S., Al-Ghonaium, A., Day-Good, N. K., Holland, S. M., ... Casanova, J.-L. (2010). Clinical features and outcome of patients with IRAK-4 and MyD88 deficiency. *Medicine*, *89*(6), 403–425.
- Pone, E. J., Zhang, J., Mai, T., White, C. A., Li, G., Sakakura, J. K., Patel, P. J., Al-Qahtani, A., Zan, H., Xu, Z., & Casali, P. (2012). BCR-signalling synergizes with TLR-signalling for induction of AID and immunoglobulin class-switching through the non-canonical NF-κB pathway. *Nature Communications*, *3*(1), 767.
- Portincasa, P., Bonfrate, L., Khalil, M., De Angelis, M., Calabrese, F. M., D'Amato, M., Wang, D. Q.-H., & Di Ciaula, A. (2022). Intestinal barrier and permeability in health, obesity and NAFLD. *Biomedicines*, 10(1), 83.
- Prudent, E., La Scola, B., Drancourt, M., Angelakis, E., & Raoult, D. (2018). Molecular strategy for the diagnosis of infectious lymphadenitis. *European Journal of Clinical Microbiology & Infectious Diseases: Official Publication of the European Society of Clinical Microbiology*, 37(6), 1179–1186.
- Psychogios, N., Hau, D. D., Peng, J., Guo, A. C., Mandal, R., Bouatra, S., Sinelnikov, I., Krishnamurthy, R., Eisner, R., Gautam, B., Young, N., Xia, J., Knox, C., Dong, E., Huang, P., Hollander, Z., Pedersen, T. L., Smith, S. R., Bamforth, F., ... Wishart, D. S. (2011). The human serum metabolome. *PloS ONE*, 6(2), e16957.

- Ragland, S. A., & Criss, A. K. (2017). From bacterial killing to immune modulation: Recent insights into the functions of lysozyme. *PLoS Pathogens*, 13(9), e1006512.
- Rampoldi, F., Ullrich, L., & Prinz, I. (2020). Revisiting the interaction of γδ T-cells and B-cells. Cells, 9(3), 743.
- Raposo, C. D., Canelas, A. B., & Barros, M. T. (2021). Human lectins, their carbohydrate affinities and where to find them. *Biomolecules*, 11(2), 188.
- Raymond, K. N., Dertz, E. A., & Kim, S. S. (2003). Enterobactin: An archetype for microbial iron transport. *Proceedings of the National Academy of Sciences of the United States of America*, 100(7), 3584–3588.
- Reith, W., Picard, C., & Fischer, A. (2013). Molecular basis of major histocompatibility complex class II deficiency. In H. D. Ochs, C. I. Edvard Smith, & J. M. Puck (Eds.), *Primary immunodeficiency diseases: A molecular and genetic approach* (3rd ed., pp. 241–257). Oxford University Press.
- Renaud, B., Brun-Buisson, C., & ICU-Bacteremia Study Group. (2001). Outcomes of primary and catheter-related bacteremia. A cohort and case-control study in critically ill patients. *American Journal of Respiratory and Critical Care Medicine*, 163(7), 1584–1590.
- Riazanski, V., Sui, Z., & Nelson, D. J. (2020). Kinetic separation of oxidative and non-oxidative metabolism in single phagosomes from alveolar macrophages: Impact on bacterial killing. *iScience*, 23(11), 101759.
- Ribet, D., & Cossart, P. (2015). How bacterial pathogens colonize their hosts and invade deeper tissues. *Microbes and Infection*, 17(3), 173–183.
- Roberts, M. B., & Fishman, J. A. (2021). Immunosuppressive agents and infectious risk in transplantation: Managing the "net state of immunosuppression". *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America*, 73(7), e1302–e1317.
- Rossjohn, J., Gras, S., Miles, J. J., Turner, S. J., Godfrey, D. I., & McCluskey, J. (2015). T cell antigen receptor recognition of antigen-presenting molecules. *Annual Review of Immunology*, 33(1), 169–200.
- Ross, T., Ratkowsky, D. A., Mellefont, L. A., & McMeekin, T. A. (2003). Modelling the effects of temperature, water activity, pH and lactic acid concentration on the growth rate of *Escherichia coli*. *International Journal of Food Microbiology*, 82(1), 33–43.
- Rozemuller, E., Eckle, S. B. G., McLaughlin, I., Penning, M., Mulder, W., de Bruin, H., & van Wageningen, S. (2021). MR1 encompasses at least six allele groups with coding region alterations. *HLA*, 98(6), 509–516.
- Safdar, A., Bodey, G., & Armstrong, D. (2011). Infections in patients with cancer: Overview. In A. Safdar (Ed.), *Principles and practice of cancer infectious diseases* (pp. 3–15). Humana Press.
- Sakamoto, H., Naito, H., Ohta, Y., Tanakna, R., Maeda, N., Sasaki, J., & Nord, C. E. (1999). Isolation of bacteria from cervical lymph nodes in patients with oral cancer. *Archives of Oral Biology*, 44(10), 789–793.
- Salio, M., Silk, J. D., Jones, E. Y., & Cerundolo, V. (2014). Biology of CD1- and MR1-restricted T cells. Annual Review of Immunology, 32, 323–366.
- Sanjuan, M. A., Dillon, C. P., Tait, S. W. G., Moshiach, S., Dorsey, F., Connell, S., Komatsu, M., Tanaka, K., Cleveland, J. L., Withoff, S., & Green, D. R. (2007). Toll-like receptor signalling in macrophages links the autophagy pathway to phagocytosis. *Nature*, 450(7173), 1253–1257.

- Sass, V., Schneider, T., Wilmes, M., Körner, C., Tossi, A., Novikova, N., Shamova, O., & Sahl, H.-G. (2010). Human beta-defensin 3 inhibits cell wall biosynthesis in *Staphylococci. Infection and Immunity*, 78(6), 2793–2800.
- Schembri, M. A., Dalsgaard, D., & Klemm, P. (2004). Capsule shields the function of short bacterial adhesins. *Journal of Bacteriology*, 186(5), 1249–1257.
- Schnell, A., Littman, D. R., & Kuchroo, V. K. (2023). TH17 cell heterogeneity and its role in tissue inflammation. *Nature Immunology*, 24(1), 19–29.
- Sedman, P. C., Macfie, J., Sagar, P., Mitchell, C. J., May, J., Mancey-Jones, B., & Johnstone, D. (1994). The prevalence of gut translocation in humans. *Gastroenterology*, 107(3), 643–649.
- Sela, M., Mozes, E., & Shearer, G. M. (1972). Thymus-independence of slowly metabolized immunogens. Proceedings of the National Academy of Sciences of the United States of America, 69(9), 2696–2700.
- Sela, M., & Zisman, E. (1997). Different roles of D-amino acids in immune phenomena. FASEB Journal: Official Publication of the Federation of American Societies for Experimental Biology, 11(6), 449–456.
- Serruto, D., Rappuoli, R., Scarselli, M., Gros, P., & van Strijp, J. A. G. (2010). Molecular mechanisms of complement evasion: learning from *staphylococci* and *meningococci*. *Nature Reviews Microbiology*, 8(6), 393–399.
- Sethna, Z., Elhanati, Y., Callan, C. G., Walczak, A. M., & Mora, T. (2019). OLGA: fast computation of generation probabilities of B- and T-cell receptor amino acid sequences and motifs. *Bioinformatics (Oxford, England)*, 35(17), 2974–2981.
- Shehata, T. E., & Marr, A. G. (1971). Effect of nutrient concentration on the growth of *Escherichia coli*. Journal of *Bacteriology*, 107(1), 210–216.
- Sheng, Y. H., & Hasnain, S. Z. (2022). Mucus and mucins: The underappreciated host defence system. Frontiers in Cellular and Infection Microbiology, 12, 856962.
- Shimada, J., Moon, S. K., Lee, H.-Y., Takeshita, T., Pan, H., Woo, J.-I., Gellibolian, R., Yamanaka, N., & Lim, D. J. (2008). Lysozyme M deficiency leads to an increased susceptibility to *Streptococcus pneumoniae*-induced otitis media. *BMC Infectious Diseases*, 8(1), 134.
- Shimada, K., Chen, S., Dempsey, P. W., Sorrentino, R., Alsabeh, R., Slepenkin, A. V., Peterson, E., Doherty, T. M., Underhill, D., Crother, T. R., & Arditi, M. (2009). The NOD/RIP2 pathway is essential for host defenses against *Chlamydophila pneumoniae* lung infection. *PLoS Pathogens*, 5(4), e1000379.
- Siggins, M. K., & Sriskandan, S. (2021). Bacterial lymphatic metastasis in infection and immunity. Cells, 11(1), 33.
- Siggins, M. K., Lynskey, N. N., Lamb, L. E., Johnson, L. A., Huse, K. K., Pearson, M., Banerji, S., Turner, C. E., Woollard, K., Jackson, D. G., & Sriskandan, S. (2020). Extracellular bacterial lymphatic metastasis drives *Streptococcus pyogenes* systemic infection. *Nature Communications*, 11(1), 4697.
- Simonian, S. J., Gill, T. J., 3rd, & Gershoff, S. N. (1968). Studies on synthetic polypeptide antigens. XX. Genetic control of the antibody response in the rat to structurally different synthetic polypeptide antigens. *The Journal* of Immunology, 101(4), 730–742.

- Singer, M., Deutschman, C. S., Seymour, C. W., Shankar-Hari, M., Annane, D., Bauer, M., Bellomo, R., Bernard, G. R., Chiche, J.-D., Coopersmith, C. M., Hotchkiss, R. S., Levy, M. M., Marshall, J. C., Martin, G. S., Opal, S. M., Rubenfeld, G. D., van der Poll, T., Vincent, J.-L., & Angus, D. C. (2016). The third international consensus definitions for sepsis and septic shock (sepsis-3). *The Journal of the American Medical Association*, 315(8), 801–810.
- Singh, J., Ranganathan, R., & Hajdu, J. (2010). Surface dilution kinetics using substrate analog enantiomers as diluents: enzymatic lipolysis by bee venom phospholipase A2. *Analytical Biochemistry*, 407(2), 253–260.
- Sivaloganathan, D. M., & Brynildsen, M. P. (2021). Phagosome-bacteria interactions from the bottom up. *Annual Review of Chemical and Biomolecular Engineering*, *12*(1), 309–331.
- Sivick, K. E., Arpaia, N., Reiner, G. L., Lee, B. L., Russell, B. R., & Barton, G. M. (2014). Toll-like receptor-deficient mice reveal how innate immune signaling influences *Salmonella* virulence strategies. *Cell Host & Microbe*, 15(2), 203–213.
- Sjölinder, H., Mogensen, T. H., Kilian, M., Jonsson, A.-B., & Paludan, S. R. (2008). Important role for Toll-like receptor 9 in host defense against meningococcal sepsis. *Infection and Immunity*, 76(11), 5421–5428.
- Snapper, C. M. (2016). Differential regulation of polysaccharide-specific antibody responses to isolated polysaccharides, conjugate vaccines, and intact *Gram*-positive versus *Gram*-negative extracellular bacteria. *Vaccine*, 34(30), 3542–3548.
- Sonnenberg, G. F., & Hepworth, M. R. (2019). Functional interactions between innate lymphoid cells and adaptive immunity. *Nature Reviews Immunology*, *19*(10), 599–613.
- Soutourina, J., Plateau, P., & Blanquet, S. (2000). Metabolism of D-aminoacyl-tRNAs in *Escherichia coli* and Saccharomyces cerevisiae cells. *The Journal of Biological Chemistry*, 275(42), 32535–32542.
- Stones, D. H., & Krachler, A. M. (2016). Against the tide: The role of bacterial adhesion in host colonization. *Biochemical Society Transactions*, 44(6), 1571–1580.
- Stupp, Y., & Sela, M. (1967). Further studies on the immunogenicity of D-amino acid polymers. *Biochimica et Biophysica Acta*, *140*(2), 349–359.
- Sultana, S., Crompton, M. E., Meurer, K., Jankiewicz, O., Morales, G. H., Johnson, C., Horbach, E., Hoffmann, K. P., Kr, P., Shah, R., Anderson, G. M., Mortimer, N. T., Schmitz, J. E., Hadjifrangiskou, M., Foti, A., & Dahl, J.-U. (2022). Redox-mediated inactivation of the transcriptional repressor RcrR is responsible for uropathogenic *Escherichia coli*'s increased resistance to reactive chlorine species. *mBio*, *13*(5), e0192622.
- Swain, S., Selmi, C., Gershwin, M. E., & Teuber, S. S. (2019). The clinical implications of selective IgA deficiency. *Journal of Translational Autoimmunity*, *2*, 100025.
- Takeuchi, O., Hoshino, K., & Akira, S. (2000). Cutting edge: TLR2-deficient and MyD88-deficient mice are highly susceptible to *Staphylococcus aureus* infection. *The Journal of Immunology*, *165*(10), 5392–5396.
- Tan, C. C. S., Ko, K. K. K., Chen, H., Liu, J., Loh, M., SG10K\_Health Consortium, Chia, M., & Nagarajan, N. (2023). No evidence for a common blood microbiome based on a population study of 9,770 healthy humans. *Nature Microbiology*, 8(5), 973–985.

- Telepnev, M. V., Klimpel, G. R., Haithcoat, J., Knirel, Y. A., Anisimov, A. P., & Motin, V. L. (2009). Tetraacylated lipopolysaccharide of *Yersinia pestis* can inhibit multiple Toll-like receptor-mediated signaling pathways in human dendritic cells. *The Journal of Infectious Diseases*, 200(11), 1694–1702.
- Thompson, D., Pepys, M. B., & Wood, S. P. (1999). The physiological structure of human C-reactive protein and its complex with phosphocholine. *Structure*, 7(2), 169–177.
- Tolsma, V., Schwebel, C., Azoulay, E., Darmon, M., Souweine, B., Vesin, A., Goldgran-Toledano, D., Lugosi, M., Jamali, S., Cheval, C., Adrie, C., Kallel, H., Descorps-Declere, A., Garrouste-Orgeas, M., Bouadma, L., & Timsit, J.-F. (2014). Sepsis severe or septic shock: Outcome according to immune status and immunodeficiency profile. *Chest*, 146(5), 1205–1213.
- Tomás, I., Diz, P., Tobías, A., Scully, C., & Donos, N. (2012). Periodontal health status and bacteraemia from daily oral activities: Systematic review/meta-analysis. *Journal of Clinical Periodontology*, *39*(3), 213–228.
- Torres, D., Barrier, M., Bihl, F., Quesniaux, V. J. F., Maillet, I., Akira, S., Ryffel, B., & Erard, F. (2004). Toll-like receptor 2 is required for optimal control of *listeria monocytogenes* infection. *Infection and Immunity*, 72(4), 2131–2139.
- Tran, A. X., Stead, C. M., & Trent, M. S. (2005). Remodeling of *Helicobacter pylori* lipopolysaccharide. *Journal of Endotoxin Research*, 11(3), 161–166.
- Turner, J. S., Benet, Z. L., & Grigorova, I. L. (2020). Signals 1, 2 and B cell fate or: Where, when and for how long? *Immunological Reviews*, 296(1), 9–23.
- Tu, Y., Deng, S., Wang, Y., Lin, X., & Yang, Z. (2022). Adhesive ability of different oral pathogens to various dental materials: An *in vitro* study. *The Canadian Journal of Infectious Diseases & Medical Microbiology*, 2022, 9595067.
- Uldrich, A. P., Rigau, M., & Godfrey, D. I. (2020). Immune recognition of phosphoantigen-butyrophilin molecular complexes by γδ T cells. *Immunological Reviews*, *298*(1), 74–83.
- Uppalapati, M., Lee, D. J., Mandal, K., Li, H., Miranda, L. P., Lowitz, J., Kenney, J., Adams, J. J., Ault-Riché, D., Kent, S. B. H., & Sidhu, S. S. (2016). A potent d-protein antagonist of VEGF-A is nonimmunogenic, metabolically stable, and longer-circulating *in vivo*. ACS Chemical Biology, 11(4), 1058–1065.
- Uribe-Querol, E., & Rosales, C. (2020). Phagocytosis: Our current understanding of a universal biological process. *Frontiers in Immunology*, 11, 1066.
- Vaissière, A., Aldrian, G., Konate, K., Lindberg, M. F., Jourdan, C., Telmar, A., Seisel, Q., Fernandez, F., Viguier, V., Genevois, C., Couillaud, F., Boisguerin, P., & Deshayes, S. (2017). A retro-inverso cell-penetrating peptide for siRNA delivery. *Journal of Nanobiotechnology*, 15(1), 34.
- Valentini, M., Gonzalez, D., Mavridou, D. A., & Filloux, A. (2018). Lifestyle transitions and adaptive pathogenesis of *Pseudomonas aeruginosa*. *Current Opinion in Microbiology*, *41*, 15–20.
- van der Poll, T., van de Veerdonk, F. L., Scicluna, B. P., & Netea, M. G. (2017). The immunopathology of sepsis and potential therapeutic targets. *Nature Reviews Immunology*, *17*(7), 407–420.
- Van Regenmortel, M. H., & Muller, S. (1998). D-peptides as immunogens and diagnostic reagents. Current Opinion in Biotechnology, 9(4), 377–382.

- Van Rhijn, I., Godfrey, D. I., Rossjohn, J., & Moody, D. B. (2015). Lipid and small-molecule display by CD1 and MR1. *Nature Reviews Immunology*, 15(10), 643–654.
- Verbrugh, H. A., Van Dijk, W. C., Peters, R., Van Der Tol, M. E., Peterson, P. K., & Verhoef, J. (1979). *Staphylococcus aureus* opsonization mediated via the classical and alternative complement pathways. A kinetic study using MgEGTA chelated serum and human sera deficient in IgG and complement factors C1s and C2. *Immunology*, 36(3), 391–397.
- Victora, G. D., & Nussenzweig, M. C. (2022). Germinal centers. Annual Review of Immunology, 40(1), 413-442.
- Vivier, E., Artis, D., Colonna, M., Diefenbach, A., Di Santo, J. P., Eberl, G., Koyasu, S., Locksley, R. M., McKenzie, A. N. J., Mebius, R. E., Powrie, F., & Spits, H. (2018). Innate lymphoid cells: 10 years on. *Cell*, 174(5), 1054–1066.
- Vollmer, W., & Tomasz, A. (2000). The pgdA gene encodes for a peptidoglycan N-acetylglucosamine deacetylase in *streptococcus pneumoniae. The Journal of Biological Chemistry*, 275(27), 20496–20501.
- von Bernuth, H., Picard, C., Jin, Z., Pankla, R., Xiao, H., Ku, C.-L., Chrabieh, M., Mustapha, I. B., Ghandil, P., Camcioglu, Y., Vasconcelos, J., Sirvent, N., Guedes, M., Vitor, A. B., Herrero-Mata, M. J., Aróstegui, J. I., Rodrigo, C., Alsina, L., Ruiz-Ortiz, E., ... Casanova, J.-L. (2008). Pyogenic bacterial infections in humans with MyD88 deficiency. *Science*, *321*(5889), 691–696.
- von Burg, N., Chappaz, S., Baerenwaldt, A., Horvath, E., Bose Dasgupta, S., Ashok, D., Pieters, J., Tacchini-Cottier, F., Rolink, A., Acha-Orbea, H., & Finke, D. (2014). Activated group 3 innate lymphoid cells promote T-cell-mediated immune responses. *Proceedings of the National Academy of Sciences of the United States of America*, 111(35), 12835–12840.
- Vos, Q., Lees, A., Wu, Z. Q., Snapper, C. M., & Mond, J. J. (2000). B-cell activation by T-cell-independent type 2 antigens as an integral part of the humoral immune response to pathogenic microorganisms. *Immunological Reviews*, 176(1), 154–170.
- Walker, W. A. (2017). The importance of appropriate initial bacterial colonization of the intestine in newborn, child, and adult health. *Pediatric Research*, 82(3), 387–395.
- Wang, C., Zada, B., Wei, G., & Kim, S.-W. (2017). Metabolic engineering and synthetic biology approaches driving isoprenoid production in *Escherichia coli*. *Bioresource Technology*, 241, 430–438.
- Wang, G. (2014). Human antimicrobial peptides and proteins. *Pharmaceuticals*, 7(5), 545–594.
- Wang, G., Maier, S. E., Lo, L. F., Maier, G., Dosi, S., & Maier, R. J. (2010). Peptidoglycan deacetylation in *Helicobacter pylori* contributes to bacterial survival by mitigating host immune responses. *Infection and Immunity*, 78(11), 4660–4666.
- Wang, Y., Liu, J., Burrows, P. D., & Wang, J.-Y. (2020). B cell development and maturation. Advances in Experimental Medicine and Biology, 1254, 1–22.
- Wan, Y. Y., & Flavell, R. A. (2009). How diverse--CD4 effector T cells and their functions. *Journal of Molecular Cell Biology*, 1(1), 20–36.
- Ward, P. A. (2010). The harmful role of c5a on innate immunity in sepsis. *Journal of Innate Immunity*, 2(5), 439–445.

- Wei, G., de Leeuw, E., Pazgier, M., Yuan, W., Zou, G., Wang, J., Ericksen, B., Lu, W.-Y., Lehrer, R. I., & Lu, W. (2009). Through the looking glass, mechanistic insights from enantiomeric human defensins. *The Journal of Biological Chemistry*, 284(42), 29180–29192.
- Weinstein, M. P., Towns, M. L., Quartey, S. M., Mirrett, S., Reimer, L. G., Parmigiani, G., & Reller, L. B. (1997). The clinical significance of positive blood cultures in the 1990s: A prospective comprehensive evaluation of the microbiology, epidemiology, and outcome of bacteremia and fungemia in adults. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America*, 24(4), 584–602.
- Wells, C. L., Jechorek, R. P., Twiggs, L. B., & Brooker, D. C. (1990). Recovery of viable bacteria from pelvic lymph nodes of patients with gynecologic tumors. *The Journal of Infectious Diseases*, 162(5), 1216–1218.
- Wiersinga, W. J., Leopold, S. J., Cranendonk, D. R., & van der Poll, T. (2014). Host innate immune responses to sepsis. *Virulence*, *5*(1), 36–44.
- Wilkinson, I. B., Raine, T., Wiles, K., Hateley, P., Kelly, D., & McGurgan, I. (2024). Oxford Handbook of Clinical Medicine (11th ed.). Oxford University Press.
- Wilson, B. R., Bogdan, A. R., Miyazawa, M., Hashimoto, K., & Tsuji, Y. (2016). Siderophores in iron metabolism: From mechanism to therapy potential. *Trends in Molecular Medicine*, *22*(12), 1077–1090.
- Yang, C., Chen-Liaw, A., Spindler, M. P., Tortorella, D., Moran, T. M., Cerutti, A., & Faith, J. J. (2022). Immunoglobulin A antibody composition is sculpted to bind the self gut microbiome. *Science Immunology*, 7(73), eabg3208.
- Yang, Y., Kong, X., Niu, B., Yang, J., & Chen, Q. (2024). Differences in biofilm formation of *Listeria* monocytogenes and their effects on virulence and drug resistance of different strains. Foods, 13(7), 1076.
- Yarbrough, V. L., Winkle, S., & Herbst-Kralovetz, M. M. (2015). Antimicrobial peptides in the female reproductive tract: a critical component of the mucosal immune barrier with physiological and clinical implications. *Human Reproduction Update*, 21(3), 353–377.
- Yarkoni, Y., Getahun, A., & Cambier, J. C. (2010). Molecular underpinning of B-cell anergy. *Immunological Reviews*, 237(1), 249–263.
- Yoshida, M., Thiriet-Rupert, S., Mayer, L., Beloin, C., & Ghigo, J.-M. (2022). Selection for nonspecific adhesion is a driver of FimH evolution increasing *Escherichia coli* biofilm capacity. *microLife*, *3*, uqac001.
- Yoshida, R., Yoshimura, T., & Hemmi, H. (2020). Reconstruction of the "Archaeal" Mevalonate Pathway from the Methanogenic Archaeon Methanosarcina mazei in Escherichia coli Cells. Applied and Environmental Microbiology, 86(6), e02889-19.
- Yu, C.-H., & Sczepanski, J. T. (2023). The influence of chirality on the behavior of oligonucleotides inside cells: revealing the potent cytotoxicity of G-rich l-RNA. *Chemical Science*, *14*(5), 1145–1154.
- Yu, X., Yuan, J., Shi, L., Dai, S., Yue, L., & Yan, M. (2024). Necroptosis in bacterial infections. Frontiers in Immunology, 15, 1394857.
- Zewde, N., Gorham, R. D., Jr, Dorado, A., & Morikis, D. (2016). Quantitative modeling of the alternative pathway of the complement system. *PloS ONE*, *11*(3), e0152337.

- Zhang, Y.-Y., & Ning, B.-T. (2021). Signaling pathways and intervention therapies in sepsis. *Signal Transduction* and Targeted Therapy, 6(1), 407.
- Zhou, W., & Sonnenberg, G. F. (2020). Activation and suppression of group 3 innate lymphoid cells in the gut. *Trends in Immunology*, 41(8), 721–733.
- Zhou, Z.-H., Zhang, Y., Hu, Y.-F., Wahl, L. M., Cisar, J. O., & Notkins, A. L. (2007). The broad antibacterial activity of the natural antibody repertoire is due to polyreactive antibodies. *Cell Host & Microbe*, 1(1), 51–61.
- Zisman, E., Dayan, M., Sela, M., & Mozes, E. (1993). Ia-antigen-T-cell interactions for a thymus-independent antigen composed of D amino acids. *Proceedings of the National Academy of Sciences of the United States of America*, 90(3), 994–998.

# **Chapter 5**

- Abd-Aziz, N., & Poh, C. L. (2022). Development of peptide-based vaccines for cancer. *Journal of Oncology*, 2022, 9749363.
- Adenowo, A. F., Oyinloye, B. E., Ogunyinka, B. I., & Kappo, A. P. (2015). Impact of human schistosomiasis in sub-Saharan Africa. *The Brazilian Journal of Infectious Diseases*, 19(2), 196–205.
- Anderson, M., Panteli, D., van Kessel, R., Ljungqvist, G., Colombo, F., & Mossialos, E. (2023). Challenges and opportunities for incentivising antibiotic research and development in Europe. *The Lancet Regional Health -Europe*, 33, 100705.
- Anderson, R., Groundwater, P. W., Todd, A., & Worsley, A. (2012). Antibacterial agents: Chemistry, mode of action, mechanisms of resistance and clinical applications. John Wiley & Sons.
- Brewer, N. S., & Hellinger, W. C. (1991). The monobactams. Mayo Clinic Proceedings, 66(11), 1152-1157.
- Brickner, S. J. (1996). Oxazolidinone antibacterial agents. Current Pharmaceutical Design, 2(2), 175-194.
- Browne, A. J., Chipeta, M. G., Haines-Woodhouse, G., Kumaran, E. P. A., Hamadani, B. H. K., Zaraa, S., Henry, N. J., Deshpande, A., Reiner, R. C., Jr., Day, N. P. J., Lopez, A. D., Dunachie, S., Moore, C. E., Stergachis, A., Hay, S. I., & Dolecek, C. (2021). Global antibiotic consumption and usage in humans, 2000-18: A spatial modelling study. *The Lancet Planetary Health*, 5(12), e893–e904.
- Cao, Y., Peng, Q., Li, S., Deng, Z., & Gao, J. (2019). The intriguing biology and chemistry of fosfomycin: The only marketed phosphonate antibiotic. *RSC Advances*, 9(72), 42204–42218.
- Center for Drug Evaluation and Research. (1992, May). *Development of New Stereoisomeric Drugs*. U.S. Food and Drug Administration. https://www.fda.gov/regulatory-information/search-fda-guidance-documents/development-new-stereoisomeric-drugs
- Cesta, D. (2021). Synthesis and biological evaluation of the unnatural enantiomer of penicillin and isosteres of cephalosporin (Doctoral dissertation, University of North Carolina at Chapel Hill). Carolina Digital Repository. https://cdr.lib.unc.edu/concern/dissertations/zs25xj41n.
- Chan, L. Y., Kosuri, S., & Endy, D. (2005). Refactoring bacteriophage T7. *Molecular Systems Biology*, 1(1), 2005.0018.

- Chu, D. T., Nordeen, C. W., Hardy, D. J., Swanson, R. N., Giardina, W. J., Pernet, A. G., & Plattner, J. J. (1991). Synthesis, antibacterial activities, and pharmacological properties of enantiomers of temafloxacin hydrochloride. *Journal of Medicinal Chemistry*, 34(1), 168–174.
- Church, J. A., Fitzgerald, F., Walker, A. S., Gibb, D. M., & Prendergast, A. J. (2015). The expanding role of co-trimoxazole in developing countries. *The Lancet Infectious Diseases*, *15*(3), 327–339.
- De Jonghe, S., & Herdewijn, P. (2022). An overview of marketed nucleoside and nucleotide analogs. *Current Protocols*, 2(3), e376.
- Eskandari, S., Good, M. F., & Pandey, M. (2021). Peptide-protein conjugation and characterization to develop vaccines for group A *Streptococcus*. *Methods in Molecular Biology*, 2355, 17–33.
- European Medicines Agency. (1993, January 10). Investigation of chiral active substances (human) Scientific guideline. https://www.ema.europa.eu/en/investigation-chiral-active-substances-human-scientific-guideline
- Fedorenko, V., Genilloud, O., Horbal, L., Marcone, G. L., Marinelli, F., Paitan, Y., & Ron, E. Z. (2015). Antibacterial discovery and development: From gene to product and back. *BioMed Research International*, 2015, 591349.
- Fenton, C., Scott, L. J., & Plosker, G. L. (2004). Palivizumab: A review of its use as prophylaxis for serious respiratory syncytial virus infection. *Paediatric Drugs*, 6(3), 177–197.
- Forrat, R., Dumas, R., Seiberling, M., Merz, M., Lutsch, C., & Lang, J. (1998). Evaluation of the safety and pharmacokinetic profile of a new, pasteurized, human tetanus immunoglobulin administered as sham, postexposure prophylaxis of tetanus. *Antimicrobial Agents and Chemotherapy*, 42(2), 298–305.
- Fortuna, C. G., Berardozzi, R., Bonaccorso, C., Caltabiano, G., Di Bari, L., Goracci, L., Guarcello, A., Pace, A., Palumbo Piccionello, A., Pescitelli, G., Pierro, P., Lonati, E., Bulbarelli, A., Cocuzza, C. E. A., Musumarra, G., & Musumeci, R. (2014). New potent antibacterials against Gram-positive multiresistant pathogens: Effects of side chain modification and chirality in linezolid-like 1,2,4-oxadiazoles. *Bioorganic & Medicinal Chemistry*, 22(24), 6814–6825.
- Foti, C., Piperno, A., Scala, A., & Giuffrè, O. (2021). Oxazolidinone antibiotics: Chemical, biological and analytical aspects. *Molecules*, 26(14), 4280.
- Frost, I., Sati, H., Garcia-Vello, P., Hasso-Agopsowicz, M., Lienhardt, C., Gigante, V., & Beyer, P. (2023). The role of bacterial vaccines in the fight against antimicrobial resistance: An analysis of the preclinical and clinical development pipeline. *The Lancet Microbe*, 4(2), e113–e125.
- Gigante, V., Sati, H., & Beyer, P. (2022). Recent advances and challenges in antibacterial drug development. *ADMET & DMPK*, 10(2), 147–151.
- Hahn, F. E. (1967). Chloramphenicol. In D. Gottlieb & P. D. Shaw (Eds.), *Antibiotics* (pp. 308–330). Springer Berlin Heidelberg.
- Hanna, S., & Etzioni, A. (2014). MHC class I and II deficiencies. *The Journal of Allergy and Clinical Immunology*, 134(2), 269–275.
- Haradanhalli, R. S., Fotedar, N., Kumari, N., & Narayana, D. H. A. (2022). Safety and clinical efficacy of human rabies immunoglobulin in post exposure prophylaxis for category III animal exposures. *Human Vaccines & Immunotherapeutics*, 18(5), 2081024.

- Hirota, K., Murakami, K., Nemoto, K., Ono, T., Matsuo, T., Kumon, H., & Miyake, Y. (1998). Fosfomycin reduces CD15s-related antigen expression of *Streptococcus pyogenes*. *Antimicrobial Agents and Chemotherapy*, 42(5), 1083–1087.
- Hitchcock, N. M., Devequi Gomes Nunes, D., Shiach, J., Valeria Saraiva Hodel, K., Dantas Viana Barbosa, J., Alencar Pereira Rodrigues, L., Coler, B. S., Botelho Pereira Soares, M., & Badaró, R. (2023). Current clinical landscape and global potential of bacteriophage therapy. *Viruses*, 15(4), 1020.
- Hutchings, M. I., Truman, A. W., & Wilkinson, B. (2019). Antibiotics: past, present and future. Current Opinion in Microbiology, 51, 72–80.
- Kay, E., Cuccui, J., & Wren, B. W. (2019). Recent advances in the production of recombinant glycoconjugate vaccines. NPJ Vaccines, 4(1), 16.
- Kohanski, M. A., Dwyer, D. J., & Collins, J. J. (2010). How antibiotics kill bacteria: From targets to networks. *Nature Reviews Microbiology*, 8(6), 423–435.
- Korst, J. J., Johnston, J. D., Butler, K., Bianco, E. J., Conover, L. H., & Woodward, R. B. (1968). The total synthesis of dl-6-demethyl-6-deoxytetracycline. *Journal of the American Chemical Society*, 90(2), 439–457.
- Lin, D. M., Koskella, B., & Lin, H. C. (2017). Phage therapy: An alternative to antibiotics in the age of multi-drug resistance. World Journal of Gastrointestinal Pharmacology and Therapeutics, 8(3), 162–173.
- Liu, B., Yang, C.-H., Xu, G.-Y., Zhu, Y.-H., Cui, J.-R., Wu, X.-H., & Xie, Y.-Y. (2005). Syntheses of quinolone hydrochloride enantiomers from synthons (R)- and (S)-2-methylpiperazine. *Bioorganic & Medicinal Chemistry*, 13(7), 2451–2458.
- Lu, L. L., Suscovich, T. J., Fortune, S. M., & Alter, G. (2018). Beyond binding: Antibody effector functions in infectious diseases. *Nature Reviews Immunology*, 18(1), 46–61.
- Lu, R.-M., Hwang, Y.-C., Liu, I.-J., Lee, C.-C., Tsai, H.-Z., Li, H.-J., & Wu, H.-C. (2020). Development of therapeutic antibodies for the treatment of diseases. *Journal of Biomedical Science*, 27(1), 1.
- Malonis, R. J., Lai, J. R., & Vergnolle, O. (2020). Peptide-based vaccines: Current progress and future challenges. *Chemical Reviews*, 120(6), 3210–3229.
- Maxwell, R. E., & Nickel, V. S. (1954). The antibacterial activity of the isomers of chloramphenicol. *Antibiotics & Chemotherapy*, 4(3), 289–295.
- Mba, I. E., Sharndama, H. C., Anyaegbunam, Z. K. G., Anekpo, C. C., Amadi, B. C., Morumda, D., Doowuese, Y., Ihezuo, U. J., Chukwukelu, J. U., & Okeke, O. P. (2023). Vaccine development for bacterial pathogens: Advances, challenges and prospects. *Tropical Medicine & International Health*, 28(4), 275–299.
- McCallin, S., Sacher, J. C., Zheng, J., & Chan, B. K. (2019). Current state of compassionate phage therapy. *Viruses*, *11*(4), 343.
- McHugh, K. J. (2020). Controlled vaccine delivery. In Z. Li & Y. Ge (Eds.), *Controlled drug delivery systems* (1st ed., pp. 77–89). Springer.
- Mitsunaka, S., Yamazaki, K., Pramono, A. K., Ikeuchi, M., Kitao, T., Ohara, N., Kubori, T., Nagai, H., & Ando, H. (2022). Synthetic engineering and biological containment of bacteriophages. *Proceedings of the National Academy of Sciences of the United States of America*, 119(48), e2206739119.

- Moreira, R., & Taylor, S. D. (2022). The chiral target of daptomycin is the 2R,2'S stereoisomer of phosphatidylglycerol. *Angewandte Chemie International Edition*, 61(4), e202114858.
- Morrissey, I., Hoshino, K., Sato, K., Yoshida, A., Hayakawa, I., Bures, M. G., & Shen, L. L. (1996). Mechanism of differential activities of ofloxacin enantiomers. *Antimicrobial Agents and Chemotherapy*, 40(8), 1775–1784.
- Muxfeldt, H., Haas, G., Hardtmann, G., Kathawala, F., Mooberry, J. B., & Vedejs, E. (1979). Tetracyclines. 9. Total synthesis of dl-terramycin. *Journal of the American Chemical Society*, *101*(3), 689–701.
- Nguyen, R., Khanna, N. R., Safadi, A. O., Patel, P., & Sun, Y. (2024). Bacitracin topical. In *StatPearls*. StatPearls Publishing
- Niu, G., & Tan, H. (2015). Nucleoside antibiotics: biosynthesis, regulation, and biotechnology. *Trends in Microbiology*, 23(2), 110–119.
- Orange, J. S., Hossny, E. M., Weiler, C. R., Ballow, M., Berger, M., Bonilla, F. A., Buckley, R., Chinen, J., El-Gamal, Y., Mazer, B. D., Nelson, R. P., Jr, Patel, D. D., Secord, E., Sorensen, R. U., Wasserman, R. L., Cunningham-Rundles, C., & Primary Immunodeficiency Committee of the American Academy of Allergy, Asthma and Immunology. (2006). Use of intravenous immunoglobulin in human disease: A review of evidence by members of the Primary Immunodeficiency Committee of the American Academy of Allergy, Asthma and Immunology. *The Journal of Allergy and Clinical Immunology*, 117, S525–S553.
- Oromí-Bosch, A., Antani, J. D., & Turner, P. E. (2023). Developing phage therapy that overcomes the evolution of bacterial resistance. *Annual Review of Virology*, *10*(1), 503–524.
- Osterloh, A. (2022). Vaccination against bacterial infections: Challenges, progress, and new approaches with a focus on intracellular bacteria. *Vaccines*, *10*(5), 751.
- Ou, L., Kong, W.-P., Chuang, G.-Y., Ghosh, M., Gulla, K., O'Dell, S., Varriale, J., Barefoot, N., Changela, A., Chao, C. W., Cheng, C., Druz, A., Kong, R., McKee, K., Rawi, R., Sarfo, E. K., Schön, A., Shaddeau, A., Tsybovsky, Y., ... Kwong, P. D. (2020). Preclinical development of a fusion peptide conjugate as an HIV vaccine immunogen. *Scientific Reports*, 10(1), 3032.
- Papp-Wallace, K. M., Endimiani, A., Taracila, M. A., & Bonomo, R. A. (2011). Carbapenems: Past, present, and future. Antimicrobial Agents and Chemotherapy, 55(11), 4943–4960.
- Pavithrra, G., & Rajasekaran, R. (2020). Gramicidin peptide to combat antibiotic resistance: A review. *International Journal of Peptide Research and Therapeutics*, 26(1), 191–199.
- Picard, C., Casanova, J.-L., & Puel, A. (2011). Infectious diseases in patients with IRAK-4, MyD88, NEMO, or IκBα deficiency. *Clinical Microbiology Reviews*, 24(3), 490–497.
- Picard, C., von Bernuth, H., Ghandil, P., Chrabieh, M., Levy, O., Arkwright, P. D., McDonald, D., Geha, R. S., Takada, H., Krause, J. C., Creech, C. B., Ku, C.-L., Ehl, S., Maródi, L., Al-Muhsen, S., Al-Hajjar, S., Al-Ghonaium, A., Day-Good, N. K., Holland, S. M., Casanova, J.-L. (2010). Clinical features and outcome of patients with IRAK-4 and MyD88 deficiency. *Medicine*, *89*(6), 403–425.
- Pollard, A. J., Perrett, K. P., & Beverley, P. C. (2009). Maintaining protection against invasive bacteria with protein-polysaccharide conjugate vaccines. *Nature Reviews Immunology*, 9(3), 213–220.
- Pronker, E. S., Weenen, T. C., Commandeur, H., Claassen, E. H. J. H. M., & Osterhaus, A. D. M. E. (2013). Risk in vaccine research and development quantified. *PloS ONE*, *8*(3), e57755.

- Quiambao, B. P., Bavdekar, A., Dubey, A. P., Jain, H., Kolhe, D., Bianco, V., Miller, J. M., & Van der Wielen, M. (2017). Antibody persistence up to 5 y after vaccination with a quadrivalent meningococcal ACWY-tetanus toxoid conjugate vaccine in adolescents. *Human Vaccines & Immunotherapeutics*, 13(3), 636–644.
- Rappuoli, R., De Gregorio, E., & Costantino, P. (2019). On the mechanisms of conjugate vaccines. *Proceedings of the National Academy of Sciences of the United States of America*, *116*(1), 14–16.
- Romero-Steiner, S., Frasch, C. E., Carlone, G., Fleck, R. A., Goldblatt, D., & Nahm, M. H. (2006). Use of opsonophagocytosis for serological evaluation of pneumococcal vaccines. *Clinical and Vaccine Immunology*, *13*(2), 165–169.
- Seddon, J., & Bhagani, S. (2011). Antimicrobial therapy for the treatment of opportunistic infections in HIV/AIDS patients: a critical appraisal. *HIV/AIDS*, *3*, 19–33.
- Shepherd, F. R., & McLaren, J. E. (2020). T cell immunity to bacterial pathogens: Mechanisms of immune control and bacterial evasion. *International Journal of Molecular Sciences*, 21(17), 6144.
- Slingerland, C. J., Kotsogianni, I., Wesseling, C. M. J., & Martin, N. I. (2022). Polymyxin stereochemistry and its role in antibacterial activity and outer membrane disruption. *ACS Infectious Diseases*, *8*(12), 2396–2404.
- Smith, H. O., Hutchison, C. A., 3rd, Pfannkoch, C., & Venter, J. C. (2003). Generating a synthetic genome by whole genome assembly: phiX174 bacteriophage from synthetic oligonucleotides. *Proceedings of the National Academy of Sciences of the United States of America*, 100(26), 15440–15445.
- Stark, J. C., Jaroentomeechai, T., Moeller, T. D., Hershewe, J. M., Warfel, K. F., Moricz, B. S., Martini, A. M., Dubner, R. S., Hsu, K. J., Stevenson, T. C., Jones, B. D., DeLisa, M. P., & Jewett, M. C. (2021). On-demand biomanufacturing of protective conjugate vaccines. *Science Advances*, 7(6), eabe9444.
- Taplitz, R. A., Kennedy, E. B., Bow, E. J., Crews, J., Gleason, C., Hawley, D. K., Langston, A. A., Nastoupil, L. J., Rajotte, M., Rolston, K. V., Strasfeld, L., & Flowers, C. R. (2018). Antimicrobial prophylaxis for adult patients with cancer-related immunosuppression: ASCO and IDSA clinical practice guideline update. *Journal of Clinical Oncology*, 36(30), 3043–3054.
- UNICEF. (2024, March). Levels and trends in child mortality. https://data.unicef.org/resources/levels-and-trends-in-child-mortality-2024/
- Verez-Bencomo, V., Fernández-Santana, V., Hardy, E., Toledo, M. E., Rodríguez, M. C., Heynngnezz, L.,
  Rodriguez, A., Baly, A., Herrera, L., Izquierdo, M., Villar, A., Valdés, Y., Cosme, K., Deler, M. L., Montane,
  M., Garcia, E., Ramos, A., Aguilar, A., Medina, E., ... Roy, R. (2004). A synthetic conjugate polysaccharide
  vaccine against Haemophilus influenzae type b. *Science*, *305*(5683), 522–525.
- Wasserman, M., Chapman, R., Lapidot, R., Sutton, K., Dillon-Murphy, D., Patel, S., Chilson, E., Snow, V., Farkouh, R., & Pelton, S. (2021). Twenty-year public health impact of 7- and 13-Valent pneumococcal conjugate vaccines in US children. *Emerging Infectious Diseases*, 27(6), 1627–1636.
- World Health Organization. (2023). *WHO Model List of Essential Medicines: 23rd list, 2023*. https://www.who.int/publications/i/item/WHO-MHP-HPS-EML-2023.02
- Zhou, S., Wang, F., Wong, E. T., Fonkem, E., Hsieh, T.-C., Wu, J. M., & Wu, E. (2013). Salinomycin: A novel anti-cancer agent with known anti-coccidial activities. *Current Medicinal Chemistry*, 20(33), 4095–4101.

# **Chapter 6**

- Aballay, A., Yorgey, P., & Ausubel, F. M. (2000). *Salmonella typhimurium* proliferates and establishes a persistent infection in the intestine of *Caenorhabditis elegans*. *Current Biology: CB*, *10*(23), 1539–1542.
- Akat, E., Yenmiş, M., Pombal, M. A., Molist, P., Megías, M., Arman, S., Veselỳ, M., Anderson, R., & Ayaz, D. (2022). Comparison of vertebrate skin structure at class level: A review. *The Anatomical Record*, 305(12), 3543–3608.
- Apidianakis, Y., & Rahme, L. G. (2009). Drosophila melanogaster as a model host for studying Pseudomonas aeruginosa infection. Nature Protocols, 4(9), 1285–1294.
- Avery, L., & You, Y.-J. (2012). C. elegans feeding. WormBook: The Online Review of C. Elegans Biology, 1-23.
- Baka, M., Van Derlinden, E., Boons, K., Mertens, L., & Van Impe, J. F. (2013). Impact of pH on the cardinal temperatures of *E. coli* K12: Evaluation of the gamma hypothesis. *Food Control*, 29(2), 328–335.
- Bargmann, C. I., & Horvitz, H. R. (1991). Chemosensory neurons with overlapping functions direct chemotaxis to multiple chemicals in C. elegans. Neuron, 7(5), 729–742.
- Bar-On, Y. M., Phillips, R., & Milo, R. (2018). The biomass distribution on Earth. *Proceedings of the National Academy of Sciences of the United States of America*, *115*(25), 6506–6511.
- Barreaux, A. M. G., Barreaux, P., Thomas, M. B., & Koella, J. C. (2017). Inoculating Anopheles gambiae mosquitoes with beads to induce and measure the melanization immune response. Journal of Visualized Experiments, 119, e55013.
- Bartlett, A., Padfield, D., Lear, L., Bendall, R., & Vos, M. (2022). A comprehensive list of bacterial pathogens infecting humans. *Microbiology*, 168(12), 001269.
- Beasley, D. E., Koltz, A. M., Lambert, J. E., Fierer, N., & Dunn, R. R. (2015). The evolution of stomach acidity and its relevance to the human microbiome. *PLOS ONE*, *10*(7), Article e0134116.
- Berenger, J.-M., & Parola, P. (2017). Arthropod vectors of medical importance. In J. Cohen, W. G. Powderly, & S. M. Opal (Eds.), *Infectious diseases* (4th ed., pp. 104–112). Elsevier.
- Berg, R. D. (1999). Bacterial translocation from the gastrointestinal tract. Advances in Experimental Medicine and Biology, 473, 11–30.
- Bischoff, V., Vignal, C., Boneca, I. G., Michel, T., Hoffmann, J. A., & Royet, J. (2004). Function of the *drosophila* pattern-recognition receptor PGRP-SD in the detection of Gram-positive bacteria. *Nature Immunology*, *5*(11), 1175–1180.
- Bland, J. M., De Lucca, A. J., Jacks, T. J., & Vigo, C. B. (2001). All-D-cecropin B: Synthesis, conformation, lipopolysaccharide binding, and antibacterial activity. *Molecular and Cellular Biochemistry*, 218(1-2), 105–111.
- Boehm, T., Hirano, M., Holland, S. J., Das, S., Schorpp, M., & Cooper, M. D. (2018). Evolution of alternative adaptive immune systems in vertebrates. *Annual Review of Immunology*, 36, 19–42.
- Brennan, J. J., & Gilmore, T. D. (2018). Evolutionary origins of toll-like receptor signaling. *Molecular Biology and Evolution*, 35(7), 1576–1587.

- Browne, N., Heelan, M., & Kavanagh, K. (2013). An analysis of the structural and functional similarities of insect hemocytes and mammalian phagocytes. *Virulence*, *4*(7), 597–603.
- Browning, D. F., Wells, T. J., França, F. L. S., Morris, F. C., Sevastsyanovich, Y. R., Bryant, J. A., Johnson, M. D., Lund, P. A., Cunningham, A. F., Hobman, J. L., May, R. C., Webber, M. A., & Henderson, I. R. (2013). Laboratory adapted *Escherichia coli* K-12 becomes a pathogen of *Caenorhabditis elegans* upon restoration of O antigen biosynthesis. *Molecular Microbiology*, 87(5), 939–950.
- Brune, A., & Kühl, M. (1996). pH profiles of the extremely alkaline hindguts of soil-feeding termites (Isoptera: Termitidae) determined with microelectrodes. *Journal of Insect Physiology*, 42(11-12), 1121–1127.
- Buchmann, K. (2014). Evolution of innate immunity: Clues from invertebrates via fish to mammals. *Frontiers in Immunology*, *5*, 459.
- Buchon, N., Broderick, N. A., & Lemaitre, B. (2013). Gut homeostasis in a microbial world: Insights from Drosophila melanogaster. Nature Reviews Microbiology, 11(9), 615–626.
- Buchon, N., Silverman, N., & Cherry, S. (2014). Immunity in Drosophila melanogaster From microbial recognition to whole-organism physiology. Nature Reviews Immunology, 14(12), 796–810.
- Buckley, K. M., & Rast, J. P. (2015). Diversity of animal immune receptors and the origins of recognition complexity in the deuterostomes. *Developmental and Comparative Immunology*, 49(1), 179–189.
- Bulet, P., Urge, L., Ohresser, S., Hetru, C., & Otvos, L., Jr. (1996). Enlarged scale chemical synthesis and range of activity of drosocin, an O-glycosylated antibacterial peptide of *Drosophila*. *European Journal of Biochemistry*, 238(1), 64–69.
- Cabrera, K., Hoard, D. S., Gibson, O., Martinez, D. I., & Wunderlich, Z. (2023). *Drosophila* immune priming to *Enterococcus faecalis* relies on immune tolerance rather than resistance. *PLoS Pathogens*, 19(8), e1011567.
- Carter, D. O., Yellowlees, D., & Tibbett, M. (2007). Cadaver decomposition in terrestrial ecosystems. *Naturwissenschaften*, 94(1), 12–24.
- Casteels, P., & Tempst, P. (1994). Apidaecin-type peptide antibiotics function through a non-poreforming mechanism involving stereospecificity. *Biochemical and Biophysical Research Communications*, 199(1), 339–345.
- Chang, C.-I., Chelliah, Y., Borek, D., Mengin-Lecreulx, D., & Deisenhofer, J. (2006). Structure of tracheal cytotoxin in complex with a heterodimeric pattern-recognition receptor. *Science*, *311*(5768), 1761–1764.
- Chang, C.-I., Pili-Floury, S., Hervé, M., Parquet, C., Chelliah, Y., Lemaitre, B., Mengin-Lecreulx, D., & Deisenhofer, J. (2004). A *Drosophila* pattern recognition receptor contains a peptidoglycan docking groove and unusual L,D-carboxypeptidase activity. *PLoS Biology*, 2(9), E277.
- Cherry, S., & Buchon, N. (2022). Insect Immunity. In M. Flajnik, N. Singh, & S. Holland (Eds.), *Paul's Fundamental Immunology* (pp. 52–78). Wolters Kluwer Health.
- Ciesielczuk, H., Betts, J., Phee, L., Doumith, M., Hope, R., Woodford, N., & Wareham, D. W. (2015). Comparative virulence of urinary and bloodstream isolates of extra-intestinal pathogenic *Escherichia coli* in a *Galleria mellonella* model. *Virulence*, 6(2), 145–151.

- Clarke, A., & Rothery, P. (2008). Scaling of body temperature in mammals and birds. *Functional Ecology*, 22(1), 58–67.
- Denis, C., Jeuniaux, C., Gerebtzoff, M. A., & Goffart, M. (1966). La digestion stomacale chez un paresseux: L'unau *Choloepus hoffmanni* Peters. *Annales de la Société Royale Malacologique de Belgique*, 97(1), 9–29.
- Dierking, K., Yang, W., & Schulenburg, H. (2016). Antimicrobial effectors in the nematode *Caenorhabditis elegans*: An outgroup to the Arthropoda. *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences*, 371(1695), 20150299.
- Dimov, I., & Maduro, M. F. (2019). The *C. elegans* intestine: Organogenesis, digestion, and physiology. *Cell and Tissue Research*, 377(3), 383–396.
- Dressman, J. B., Berardi, R. R., Dermentzoglou, L. C., Russell, T. L., Schmaltz, S. P., Barnett, J. L., & Jarvenpaa, K. M. (1990). Upper gastrointestinal (GI) pH in young, healthy men and women. *Pharmaceutical Research*, 7(7), 756–761.
- Dunbar, T. L., Yan, Z., Balla, K. M., Smelkinson, M. G., & Troemel, E. R. (2012). C. elegans detects pathogen-induced translational inhibition to activate immune signaling. Cell Host & Microbe, 11(4), 375–386.
- Duneau, D., Ferdy, J.-B., Revah, J., Kondolf, H., Ortiz, G. A., Lazzaro, B. P., & Buchon, N. (2017). Stochastic variation in the initial phase of bacterial infection predicts the probability of survival in *D. melanogaster. eLife*, 6.
- Eigenbrode, S. D., Bosque-Pérez, N. A., & Davis, T. S. (2018). Insect-borne plant pathogens and their vectors: Ecology, evolution, and complex interactions. *Annual Review of Entomology*, *63*, 169–191.
- Eleftherianos, I., Heryanto, C., Bassal, T., Zhang, W., Tettamanti, G., & Mohamed, A. (2021). Haemocyte-mediated immunity in insects: Cells, processes and associated components in the fight against pathogens and parasites. *Immunology*, 164(3), 401–432.
- Eleftherianos, I., Zhang, W., Heryanto, C., Mohamed, A., Contreras, G., Tettamanti, G., Wink, M., & Bassal, T. (2021). Diversity of insect antimicrobial peptides and proteins - A functional perspective: A review. *International Journal of Biological Macromolecules*, 191, 277–287.
- Engel, P., & Moran, N. A. (2013). The gut microbiota of insects Diversity in structure and function. FEMS Microbiology Reviews, 37(5), 699–735.
- Ermolaeva, M. A., & Schumacher, B. (2014). Insights from the worm: the *C. elegans* model for innate immunity. *Seminars in Immunology*, *26*(4), 303–309.
- Evans, J. D., Aronstein, K., Chen, Y. P., Hetru, C., Imler, J.-L, Jiang, H., Kanost, M., Thompson, G. J., Zou, Z., & Hultmark, D. (2006). Immune pathways and defence mechanisms in honey bees *Apis mellifera*. *Insect Molecular Biology*, 15(5), 645–656.
- Fang-Yen, C., Avery, L., & Samuel, A. D. T. (2009). Two size-selective mechanisms specifically trap bacteria-sized food particles in *Caenorhabditis elegans*. Proceedings of the National Academy of Sciences of the United States of America, 106(47), 20093–20096.
- Fehlbaum, P., Bulet, P., Chernysh, S., Briand, J. P., Roussel, J. P., Letellier, L., Hetru, C., & Hoffmann, J. A. (1996). Structure-activity analysis of thanatin, a 21-residue inducible insect defense peptide with sequence homology to frog skin antimicrobial peptides. *Proceedings of the National Academy of Sciences of the United States of America*, 93(3), 1221–1225.
- Ferkey, D. M., Sengupta, P., & L'Etoile, N. D. (2021). Chemosensory signal transduction in *Caenorhabditis elegans*. *Genetics*, 217(3), iyab004.
- Filipe, S. R., Tomasz, A., & Ligoxygakis, P. (2005). Requirements of peptidoglycan structure that allow detection by the *Drosophila* Toll pathway. *EMBO Reports*, 6(4), 327–333.
- Flajnik, M. F. (2018). A cold-blooded view of adaptive immunity. Nature Reviews Immunology, 18(7), 438-453.
- Fung, V. K. Y., Tan, D. Z. J., & Zhou, K. (2023). Improving glycine utilization in *Escherichia coli*. Biochemical Engineering Journal, 193(108834), 108834.
- Garsin, D. A., Sifri, C. D., Mylonakis, E., Qin, X., Singh, K. V., Murray, B. E., Calderwood, S. B., & Ausubel, F. M. (2001). A simple model host for identifying Gram-positive virulence factors. *Proceedings of the National Academy of Sciences of the United States of America*, 98(19), 10892–10897.
- Gauthier, D. T. (2015). Bacterial zoonoses of fishes: A review and appraisal of evidence for linkages between fish and human infections. *The Veterinary Journal*, 203(1), 27–35.
- Ghafouri, S., & McGhee, J. D. (2007). Bacterial residence time in the intestine of *Caenorhabditis elegans*. Nematology: International Journal of Fundamental and Applied Nematological Research, 9(1), 87–91.
- Ghelardi, E., Celandroni, F., Salvetti, S., Fiscarelli, E., & Senesi, S. (2007). Bacillus thuringiensis pulmonary infection: Critical role for bacterial membrane-damaging toxins and host neutrophils. Microbes and Infection, 9(5), 591–598.
- Gobert, V., Gottar, M., Matskevich, A. A., Rutschmann, S., Royet, J., Belvin, M., Hoffmann, J. A., & Ferrandon, D. (2003). Dual activation of the *Drosophila* toll pathway by two pattern recognition receptors. *Science*, *302*(5653), 2126–2130.
- Gottar, M., Gobert, V., Michel, T., Belvin, M., Duyk, G., Hoffmann, J. A., Ferrandon, D., & Royet, J. (2002). The Drosophila immune response against Gram-negative bacteria is mediated by a peptidoglycan recognition protein. Nature, 416(6881), 640–644.
- Gravato-Nobre, M. J., Vaz, F., Filipe, S., Chalmers, R., & Hodgkin, J. (2016). The invertebrate lysozyme effector ILYS-3 is systemically activated in response to danger signals and confers antimicrobial protection in *C. elegans. PLoS Pathogens*, 12(8), e1005826.
- Hammer, T. J., Janzen, D. H., Hallwachs, W., Jaffe, S. P., & Fierer, N. (2017). Caterpillars lack a resident gut microbiome. *Proceedings of the National Academy of Sciences of the United States of America*, 114(36), 9641–9646.
- Hartenstein, V. (2006). Blood cells and blood cell development in the animal kingdom. *Annual Review of Cell and Developmental Biology*, 22(1), 677–712.
- Hashimoto, Y., Tabuchi, Y., Sakurai, K., Kutsuna, M., Kurokawa, K., Awasaki, T., Sekimizu, K., Nakanishi, Y., & Shiratsuchi, A. (2009). Identification of lipoteichoic acid as a ligand for draper in the phagocytosis of *Staphylococcus aureus* by *Drosophila* hemocytes. *The Journal of Immunology*, 183(11), 7451–7460.

Hejazi, A., & Falkiner, F. R. (1997). Serratia marcescens. Journal of Medical Microbiology, 46(11), 903–912.

- Herczeg, D., Ujszegi, J., Kásler, A., Holly, D., & Hettyey, A. (2021). Host-multiparasite interactions in amphibians: A review. *Parasites & Vectors*, 14(1), 296.
- Herpol, C. (1967). Etude de l'activité protéolytique des divers organes du système digestif de quelques espèces d'oiseaux en rapport avec leur régime alimentaire. *Zeitschrift für vergleichende Physiologie*, *57*(3), 209–217.
- Hillyer, J. F. (2016). Insect immunology and hematopoiesis. *Developmental and Comparative Immunology*, 58, 102–118.
- Hirsch, R., Wiesner, J., Marker, A., Pfeifer, Y., Bauer, A., Hammann, P. E., & Vilcinskas, A. (2019). Profiling antimicrobial peptides from the medical maggot *Lucilia sericata* as potential antibiotics for MDR Gram-negative bacteria. *The Journal of Antimicrobial Chemotherapy*, 74(1), 96–107.
- Hoffmann, J. A., Kafatos, F. C., Janeway, C. A., & Ezekowitz, R. A. (1999). Phylogenetic perspectives in innate immunity. *Science*, 284(5418), 1313–1318.
- Howe, K., Schiffer, P. H., Zielinski, J., Wiehe, T., Laird, G. K., Marioni, J. C., Soylemez, O., Kondrashov, F., & Leptin, M. (2016). Structure and evolutionary history of a large family of NLR proteins in the zebrafish. *Open Biology*, 6(4), 160009.
- Huang, G., Qu, Q., Wang, M., Huang, M., Zhou, W., & Wei, F. (2022). Global landscape of gut microbiome diversity and antibiotic resistomes across vertebrates. *The Science of the Total Environment*, 838(Part 2), 156178.
- Huang, W., Reyes-Caldas, P., Mann, M., Seifbarghi, S., Kahn, A., Almeida, R. P. P., Béven, L., Heck, M., Hogenhout, S. A., & Coaker, G. (2020). Bacterial vector-borne plant diseases: Unanswered questions and future directions. *Molecular Plant*, 13(10), 1379–1393.
- Hu, K., & Webster, J. M. (1998). In vitro and in vivo characterization of a small-colony variant of the primary form of *Photorhabdus luminescens* MD (Enterobacteriaceae). *Applied and Environmental Microbiology*, 64(9), 3214–3219.
- Iatsenko, I., Kondo, S., Mengin-Lecreulx, D., & Lemaitre, B. (2016). PGRP-SD, an extracellular pattern-recognition receptor, enhances peptidoglycan-mediated activation of the *Drosophila* imd pathway. *Immunity*, 45(5), 1013–1023.
- Jander, G., Rahme, L. G., & Ausubel, F. M. (2000). Positive correlation between virulence of *Pseudomonas* aeruginosa mutants in mice and insects. *Journal of Bacteriology*, 182(13), 3843–3845.
- Janeway, C. A., Jr, & Medzhitov, R. (2002). Innate immune recognition. *Annual Review of Immunology*, 20(1), 197–216.
- Kaspers, B., Schat, K. A., Göbel, T., & Vervelde, L. (Eds.). (2021). Avian immunology (3rd ed.). Academic Press.
- Keshavarz, M., Zanchi, C., & Rolff, J. (2023). The effect of combined knockdowns of Attacins on survival and bacterial load in *Tenebrio molitor*. Frontiers in Immunology, 14, 1140627.
- Kim, E.-K., Lee, K.-A., Hyeon, D. Y., Kyung, M., Jun, K.-Y., Seo, S. H., Hwang, D., Kwon, Y., & Lee, W.-J. (2020). Bacterial nucleoside catabolism controls quorum sensing and commensal-to-pathogen transition in the *Drosophila* gut. *Cell Host & Microbe*, 27(3), 345–357.e6.

- Kim, S.-H., & Lee, W.-J. (2014). Role of DUOX in gut inflammation: Lessons from *Drosophila* model of gut-microbiota interactions. *Frontiers in Cellular and Infection Microbiology*, *3*, 116.
- Kirchman, D. L., Morán, X. A. G., & Ducklow, H. (2009). Microbial growth in the polar oceans Role of temperature and potential impact of climate change. *Nature Reviews Microbiology*, 7(6), 451–459.
- Kiyama, Y., Miyahara, K., & Ohshima, Y. (2012). Active uptake of artificial particles in the nematode *Caenorhabditis elegans. The Journal of Experimental Biology*, *215*(Pt 7), 1178–1183.
- Kobayashi, M., Sasaki, T., Saito, N., Tamura, K., Suzuki, K., Watanabe, H., & Agui, N. (1999). Houseflies: Not simple mechanical vectors of enterohemorrhagic *Escherichia coli* O157:H7. *The American Journal of Tropical Medicine and Hygiene*, 61(4), 625–629.
- Koga, R., Moriyama, M., Onodera-Tanifuji, N., Ishii, Y., Takai, H., Mizutani, M., Oguchi, K., Okura, R., Suzuki, S., Gotoh, Y., Hayashi, T., Seki, M., Suzuki, Y., Nishide, Y., Hosokawa, T., Wakamoto, Y., Furusawa, C., & Fukatsu, T. (2022). Single mutation makes *Escherichia coli* an insect mutualist. *Nature Microbiology*, 7(8), 1141–1150.
- Kragol, G., Lovas, S., Varadi, G., Condie, B. A., Hoffmann, R., & Otvos, L., Jr. (2001). The antibacterial peptide pyrrhocoricin inhibits the ATPase actions of DnaK and prevents chaperone-assisted protein folding. *Biochemistry*, 40(10), 3016–3026.
- Laroche, M., Raoult, D., & Parola, P. (2018). Insects and the transmission of bacterial agents. *Microbiology Spectrum*, 6(5), MTBP-0017-2016.
- Lee, K.-A., Kim, B., Bhin, J., Kim, D. H., You, H., Kim, E.-K., Kim, S.-H., Ryu, J.-H., Hwang, D., & Lee, W.-J. (2015). Bacterial uracil modulates *Drosophila* DUOX-dependent gut immunity via Hedgehog-induced signaling endosomes. *Cell Host & Microbe*, 17(2), 191–204.
- Lee, K.-A., Kim, S.-H., Kim, E.-K., Ha, E.-M., You, H., Kim, B., Kim, M.-J., Kwon, Y., Ryu, J.-H., & Lee, W.-J. (2013). Bacterial-derived uracil as a modulator of mucosal immunity and gut-microbe homeostasis in *Drosophila. Cell*, 153(4), 797–811.
- Lemaitre, B., Nicolas, E., Michaut, L., Reichhart, J. M., & Hoffmann, J. A. (1996). The dorsoventral regulatory gene cassette spätzle/toll/cactus controls the potent antifungal response in *Drosophila* adults. *Cell*, *86*(6), 973–983.
- Leone, P., Bischoff, V., Kellenberger, C., Hetru, C., Royet, J., & Roussel, A. (2008). Crystal structure of *Drosophila* PGRP-SD suggests binding to DAP-type but not lysine-type peptidoglycan. *Molecular Immunology*, 45(9), 2521–2530.
- Liegeois, S., & Ferrandon, D. (2022). Sensing microbial infections in the *Drosophila melanogaster* genetic model organism. *Immunogenetics*, 74(1), 35–62.
- Lim, J.-H., Kim, M.-S., Kim, H.-E., Yano, T., Oshima, Y., Aggarwal, K., Goldman, W. E., Silverman, N., Kurata, S., & Oh, B.-H. (2006). Structural basis for preferential recognition of diaminopimelic acid-type peptidoglycan by a subset of peptidoglycan recognition proteins. *The Journal of Biological Chemistry*, 281(12), 8286–8295.
- Lin, Z., Wang, J.-L., Cheng, Y., Wang, J.-X., & Zou, Z. (2020). Pattern recognition receptors from lepidopteran insects and their biological functions. *Developmental and Comparative Immunology*, *108*, 103688.
- Liu, Y., Samuel, B. S., Breen, P. C., & Ruvkun, G. (2014). *Caenorhabditis elegans* pathways that surveil and defend mitochondria. *Nature*, 508(7496), 406–410.

- Liu, Y., Ye, N., Chen, M., Zhao, H., & An, J. (2020). Structural and functional analysis of PGRP-LC indicates exclusive Dap-type PGN binding in bumblebees. *International Journal of Molecular Sciences*, 21(7), 2441.
- Liu, Y., Zhao, X., Huang, J., Chen, M., & An, J. (2019). Structural insights into the preferential binding of PGRP-SAs from bumblebees and honeybees to Dap-type peptidoglycans rather than Lys-type peptidoglycans. *The Journal of Immunology*, 202(1), 249–259.
- Magnadóttir, B. (2006). Innate immunity of fish (overview). Fish & Shellfish Immunology, 20(2), 137-151.
- Mallo, G. V., Kurz, C. L., Couillault, C., Pujol, N., Granjeaud, S., Kohara, Y., & Ewbank, J. J. (2002). Inducible antibacterial defense system in *C. elegans. Current Biology: CB*, 12(14), 1209–1214.
- McEwan, D. L., Kirienko, N. V., & Ausubel, F. M. (2012). Host translational inhibition by *Pseudomonas aeruginosa* Exotoxin A triggers an immune response in *Caenorhabditis elegans*. *Cell Host & Microbe*, *11*(4), 364–374.
- Meisel, J. D., Panda, O., Mahanti, P., Schroeder, F. C., & Kim, D. H. (2014). Chemosensation of bacterial secondary metabolites modulates neuroendocrine signaling and behavior of *C. elegans*. *Cell*, *159*(2), 267–280.
- Melcarne, C., Lemaitre, B., & Kurant, E. (2019). Phagocytosis in *Drosophila*: From molecules and cellular machinery to physiology. *Insect Biochemistry and Molecular Biology*, 109, 1–12.
- Melo, J. A., & Ruvkun, G. (2012). Inactivation of conserved *C. elegans* genes engages pathogen- and xenobiotic-associated defenses. *Cell*, *149*(2), 452–466.
- Merkx-Jacques, A., Coors, A., Brousseau, R., Masson, L., Mazza, A., Tien, Y.-C., & Topp, E. (2013). Evaluating the pathogenic potential of environmental *Escherichia coli* by using the *Caenorhabditis elegans* infection model. *Applied and Environmental Microbiology*, 79(7), 2435–2445.
- Mermin, J., Hutwagner, L., Vugia, D., Shallow, S., Daily, P., Bender, J., Koehler, J., Marcus, R., Angulo, F. J., & Emerging Infections Program FoodNet Working Group. (2004). Reptiles, amphibians, and human Salmonella infection: A population-based, case-control study. *Clinical Infectious Diseases*, 38(Suppl 3), S253–S261.
- Metcalf, J. L., Wegener Parfrey, L., Gonzalez, A., Lauber, C. L., Knights, D., Ackermann, G., Humphrey, G. C., Gebert, M. J., Van Treuren, W., Berg-Lyons, D., Keepers, K., Guo, Y., Bullard, J., Fierer, N., Carter, D. O., & Knight, R. (2013). A microbial clock provides an accurate estimate of the postmortem interval in a mouse model system. *eLife*, 2, e01104.
- Michel, T., Reichhart, J. M., Hoffmann, J. A., & Royet, J. (2001). *Drosophila* Toll is activated by Gram-positive bacteria through a circulating peptidoglycan recognition protein. *Nature*, *414*(6865), 756–759.
- Misof, B., Liu, S., Meusemann, K., Peters, R. S., Donath, A., Mayer, C., Frandsen, P. B., Ware, J., Flouri, T., Beutel, R. G., Niehuis, O., Petersen, M., Izquierdo-Carrasco, F., Wappler, T., Rust, J., Aberer, A. J., Aspöck, U., Aspöck, H., Bartel, D., ... Zhou, X. (2014). Phylogenomics resolves the timing and pattern of insect evolution. *Science*, 346(6210), 763–767.
- Mueller, A.-K., Kohlhepp, F., Hammerschmidt, C., & Michel, K. (2010). Invasion of mosquito salivary glands by malaria parasites: Prerequisites and defense strategies. *International Journal for Parasitology*, 40(11), 1229–1235.
- Mulcahy, H., Sibley, C. D., Surette, M. G., & Lewenza, S. (2011). *Drosophila melanogaster* as an animal model for the study of *Pseudomonas aeruginosa* biofilm infections *in vivo*. *PLoS Pathogens*, 7(10), e1002299.

- Neerukonda, S. N., & Katneni, U. (2020). Avian pattern recognition receptor sensing and signaling. *Veterinary Sciences*, 7(1), 14.
- Nehme, N. T., Liégeois, S., Kele, B., Giammarinaro, P., Pradel, E., Hoffmann, J. A., Ewbank, J. J., & Ferrandon, D. (2007). A model of bacterial intestinal infections in *Drosophila melanogaster*. *PLoS Pathogens*, *3*(11), e173.
- Noriega, J. A., Hortal, J., Azcárate, F. M., Berg, M. P., Bonada, N., Briones, M. J. I., Del Toro, I., Goulson, D., Ibanez, S., Landis, D. A., Moretti, M., Potts, S. G., Slade, E. M., Stout, J. C., Ulyshen, M. D., Wackers, F. L., Woodcock, B. A., & Santos, A. M. C. (2018). Research trends in ecosystem services provided by insects. *Basic* and Applied Ecology, 26, 8–23.
- Ohsugi, T., Kurosawa, T., & Maejima, K. (1992). Effect of age on bacterial translocation from the gastrointestinal tract in mice. *Experimental Animals*, 41(1), 55-60.
- Overend, G., Luo, Y., Henderson, L., Douglas, A. E., Davies, S. A., & Dow, J. A. T. (2016). Molecular mechanism and functional significance of acid generation in the *Drosophila* midgut. *Scientific Reports*, *6*, 27242.
- Pasmans, F., Martel, A., & Jacobson, E. R. (2020). Bacterial diseases of reptiles. In E. R. Jacobson & M. M. Garner (Eds.), *Infectious diseases and pathology of reptiles* (2nd ed., pp. 461–550). CRC Press.
- Pazos, M., & Peters, K. (2019). Peptidoglycan. Sub-Cellular Biochemistry, 92, 127-168.
- Pees, B., Yang, W., Kloock, A., Petersen, C., Peters, L., Fan, L., Friedrichsen, M., Butze, S., Zárate-Potes, A., Schulenburg, H., & Dierking, K. (2021). Effector and regulator: Diverse functions of *C. elegans* C-type lectin-like domain proteins. *PLoS Pathogens*, 17(4), e1009454.
- Peng, Z., Zhang, W., Fu, H., Li, Y., Zhang, C., Li, J., Chan, J., & Zhang, L. (2024). Genome-wide comparative analysis of SRCR gene superfamily in invertebrates reveals massive and independent gene expansions in the sponge and sea urchin. *International Journal of Molecular Sciences*, 25(3), 1515.
- Peterson, N. D., Tse, S. Y., Huang, Q. J., Wani, K. A., Schiffer, C. A., & Pukkila-Worley, R. (2023). Non-canonical pattern recognition of a pathogen-derived metabolite by a nuclear hormone receptor identifies virulent bacteria in *C. elegans. Immunity*, 56(4), 768–782.e9.
- Piyankarage, S. C., Augustin, H., Grosjean, Y., Featherstone, D. E., & Shippy, S. A. (2008). Hemolymph amino acid analysis of individual *Drosophila* larvae. *Analytical Chemistry*, 80(4), 1201–1207.
- Pradeu, T., Thomma, B. P. H. J., Girardin, S. E., & Lemaitre, B. (2024). The conceptual foundations of innate immunity: Taking stock 30 years later. *Immunity*, 57(4), 613–631.
- Prescott, J. F., Rycroft, A. N., Boyce, J. D., MacInnes, J. I., Van Immerseel, F., & Vázquez-Boland, J. A. (Eds.). (2022). *Pathogenesis of bacterial infections in animals*. Wiley.
- Radeke, L. J., & Herman, M. A. (2021). Take a walk to the wild side of *Caenorhabditis elegans*-pathogen interactions. *Microbiology and Molecular Biology Reviews*, 85(2).
- Rahman, M. T., Sobur, M. A., Islam, M. S., Ievy, S., Hossain, M. J., El Zowalaty, M. E., Rahman, A. T., & Ashour, H. M. (2020). Zoonotic diseases: Etiology, impact, and control. *Microorganisms*, 8(9), 1405.
- Rämet, M., Manfruelli, P., Pearson, A., Mathey-Prevot, B., & Ezekowitz, R. A. B. (2002). Functional genomic analysis of phagocytosis and identification of a *Drosophila* receptor for *E. coli. Nature*, 416(6881), 644–648.

- Rauta, P. R., Nayak, B., & Das, S. (2012). Immune system and immune responses in fish and their role in comparative immunity study: A model for higher organisms. *Immunology Letters*, 148(1), 23–33.
- Raymann, K., Coon, K. L., Shaffer, Z., Salisbury, S., & Moran, N. A. (2018). Pathogenicity of Serratia marcescens strains in honey bees. mBio, 9(5), e01649-18.
- Raymann, K., & Moran, N. A. (2018). The role of the gut microbiome in health and disease of adult honey bee workers. *Current Opinion in Insect Science*, 26, 97–104.
- Reiser, J.-B., Teyton, L., & Wilson, I. A. (2004). Crystal structure of the *Drosophila* peptidoglycan recognition protein (PGRP)-SA at 1.56 Å resolution. *Journal of Molecular Biology*, 340(4), 909–917.
- Riera-Romo, M., Pérez-Martínez, D., & Castillo Ferrer, C. (2016). Innate immunity in vertebrates: An overview. *Immunology*, 148(2), 125–139.
- Rinas, U., Hellmuth, K., Kang, R., Seeger, A., & Schlieker, H. (1995). Entry of *Escherichia coli* into stationary phase is indicated by endogenous and exogenous accumulation of nucleobases. *Applied and Environmental Microbiology*, 61(12), 4147–4151.
- Rochon, K., Lysyk, T. J., & Selinger, L. B. (2004). Persistence of *Escherichia coli* in immature house fly and stable fly (Diptera: Muscidae) in relation to larval growth and survival. *Journal of Medical Entomology*, 41(6), 1082–1089.
- Rodgers, F. H., Cai, J. A., Pitaluga, A. N., Mengin-Lecreulx, D., Gendrin, M., & Christophides, G. K. (2020). Functional analysis of the three major PGRPLC isoforms in the midgut of the malaria mosquito *Anopheles coluzzii*. *Insect Biochemistry and Molecular Biology*, 118, 103288.
- Roeder, T., Stanisak, M., Gelhaus, C., Bruchhaus, I., Grötzinger, J., & Leippe, M. (2010). Caenopores are antimicrobial peptides in the nematode *Caenorhabditis elegans* instrumental in nutrition and immunity. *Developmental and Comparative Immunology*, 34(2), 203–209.
- Rosenberg, Y., Bar-On, Y. M., Fromm, A., Ostikar, M., Shoshany, A., Giz, O., & Milo, R. (2023). The global biomass and number of terrestrial arthropods. *Science Advances*, 9(5), eabq4049.
- Ross, A. A., Rodrigues Hoffmann, A., & Neufeld, J. D. (2019). The skin microbiome of vertebrates. *Microbiome*, 7(1), 79.
- Ross, T., Ratkowsky, D. A., Mellefont, L. A., & McMeekin, T. A. (2003). Modelling the effects of temperature, water activity, pH and lactic acid concentration on the growth rate of *Escherichia coli*. *International Journal of Food Microbiology*, 82(1), 33–43.
- Ruiz, V. L., & Robert, J. (2023). The amphibian immune system. *Philosophical Transactions of the Royal Society of* London. Series B, Biological Sciences, 378(1882), 20220123.
- Russell, T. L., Berardi, R. R., Barnett, J. L., Dermentzoglou, L. C., Jarvenpaa, K. M., Schmaltz, S. P., & Dressman, J. B. (1993). Upper gastrointestinal pH in seventy-nine healthy, elderly, North American men and women. *Pharmaceutical Research*, 10(2), 187–196.
- Rutschmann, S., Jung, A. C., Zhou, R., Silverman, N., Hoffmann, J. A., & Ferrandon, D. (2000). Role of *Drosophila* IKK gamma in a toll-independent antibacterial immune response. *Nature Immunology*, *1*(4), 342–347.

- Ryu, J.-H., Ha, E.-M., Oh, C.-T., Seol, J.-H., Brey, P. T., Jin, I., Lee, D. G., Kim, J., Lee, D., & Lee, W.-J. (2006). An essential complementary role of NF-kappaB pathway to microbicidal oxidants in *Drosophila* gut immunity. *The EMBO Journal*, *25*(15), 3693–3701.
- Saco, A., Novoa, B., Greco, S., Gerdol, M., & Figueras, A. (2023). Bivalves present the largest and most diversified repertoire of toll-like receptors in the animal kingdom, suggesting broad-spectrum pathogen recognition in marine waters. *Molecular Biology and Evolution*, 40(6), msad133.
- Schell, M. A., Lipscomb, L., & DeShazer, D. (2008). Comparative genomics and an insect model rapidly identify novel virulence genes of *Burkholderia mallei*. *Journal of Bacteriology*, 190(7), 2306–2313.
- Schikora, A., Garcia, A. V., & Hirt, H. (2012). Plants as alternative hosts for Salmonella. Trends in Plant Science, 17(5), 245–249.
- Schmidt, K., & Engel, P. (2021). Mechanisms underlying gut microbiota-host interactions in insects. *The Journal of Experimental Biology*, 224(Pt 2), jeb207696.
- Schneider, D. S., Ayres, J. S., Brandt, S. M., Costa, A., Dionne, M. S., Gordon, M. D., Mabery, E. M., Moule, M. G., Pham, L. N., & Shirasu-Hiza, M. M. (2007). *Drosophila* eiger mutants are sensitive to extracellular pathogens. *PLoS Pathogens*, 3(3), e41.
- Schulenburg, H., & Félix, M.-A. (2017). The natural biotic environment of *Caenorhabditis elegans*. *Genetics*, 206(1), 55–86.
- Schwartz, A., & Koella, J. C. (2004). The cost of immunity in the yellow fever mosquito, *Aedes aegypti* depends on immune activation. *Journal of Evolutionary Biology*, 17(4), 834–840.
- Shao, Y., Mason, C. J., & Felton, G. W. (2024). Toward an integrated understanding of the Lepidoptera microbiome. *Annual Review of Entomology*, 69, 117–137.
- Shaw, L. P., Wang, A. D., Dylus, D., Meier, M., Pogacnik, G., Dessimoz, C., & Balloux, F. (2020). The phylogenetic range of bacterial and viral pathogens of vertebrates. *Molecular Ecology*, 29(17), 3361–3379.
- Shirasu-Hiza, M. M., & Schneider, D. S. (2007). Confronting physiology: How do infected flies die? *Cellular Microbiology*, 9(12), 2775–2783.
- Shtonda, B. B., & Avery, L. (2006). Dietary choice behavior in Caenorhabditis elegans. The Journal of Experimental Biology, 209(Pt 1), 89–102.
- Thompson, S.N. (2003). Trehalose the insect 'blood' sugar. Advances in Insect Physiology, 31, 205-285.
- Singh, J., & Aballay, A. (2019). Microbial colonization activates an immune fight-and-flight response via neuroendocrine signaling. *Developmental Cell*, 49(1), 89–99.e4.
- Singh, S., Reese, J. M., Casanova-Torres, A. M., Goodrich-Blair, H., & Forst, S. (2014). Microbial population dynamics in the hemolymph of *Manduca sexta* infected with *Xenorhabdus nematophila* and the entomopathogenic nematode *Steinernema carpocapsae*. *Applied and Environmental Microbiology*, 80(14), 4277–4285.
- Solbakken, M. H., Tørresen, O. K., Nederbragt, A. J., Seppola, M., Gregers, T. F., Jakobsen, K. S., & Jentoft, S. (2016). Evolutionary redesign of the Atlantic cod (*Gadus morhua* L.) Toll-like receptor repertoire by gene losses and expansions. *Scientific Reports*, 6, 25211.

- Southern, P. M., Jr. (1996). Bacteremia due to *Agrobacterium tumefaciens* (*radiobacter*). Report of infection in a pregnant woman and her stillborn fetus. *Diagnostic Microbiology and Infectious Disease*, 24(1), 43–45.
- Soutourina, J., Blanquet, S., & Plateau, P. (2001). Role of D-cysteine desulfhydrase in the adaptation of *Escherichia* coli to D-cysteine. *The Journal of Biological Chemistry*, 276(44), 40864–40872.
- Star, B., Nederbragt, A. J., Jentoft, S., Grimholt, U., Malmstrøm, M., Gregers, T. F., Rounge, T. B., Paulsen, J., Solbakken, M. H., Sharma, A., Wetten, O. F., Lanzén, A., Winer, R., Knight, J., Vogel, J.-H., Aken, B., Andersen, O., Lagesen, K., Tooming-Klunderud, A., ... Jakobsen, K. S. (2011). The genome sequence of Atlantic cod reveals a unique immune system. *Nature*, 477(7363), 207–210.
- Subasi, B. S., Grabe, V., Kaltenpoth, M., Rolff, J., & Armitage, S. A. O. (2024). How frequently are insects wounded in the wild? A case study using *Drosophila melanogaster*. *Royal Society Open Science*, 11(6), 240256.
- Sundaram, B., Tweedell, R. E., Prasanth Kumar, S., & Kanneganti, T.-D. (2024). The NLR family of innate immune and cell death sensors. *Immunity*, *57*(4), 674–699.
- Takehana, A., Yano, T., Mita, S., Kotani, A., Oshima, Y., & Kurata, S. (2004). Peptidoglycan recognition protein (PGRP)-LE and PGRP-LC act synergistically in *Drosophila* immunity. *The EMBO Journal*, 23(23), 4690–4700.
- Terra, W. R., & Ferreira, C. (1994). Insect digestive enzymes: Properties, compartmentalization and function. *Comparative Biochemistry and Physiology Part B: Comparative Biochemistry*, 109(1), 1–62.
- Tizard, I. R. (2023). Comparative mammalian immunology: The evolution and diversity of the immune systems of mammals. Academic Press.
- Tran, A., Tang, A., O'Loughlin, C. T., Balistreri, A., Chang, E., Coto Villa, D., Li, J., Varshney, A., Jimenez, V., Pyle, J., Tsujimoto, B., Wellbrook, C., Vargas, C., Duong, A., Ali, N., Matthews, S. Y., Levinson, S., Woldemariam, S., Khuri, S., ... VanHoven, M. K. (2017). *C. elegans* avoids toxin-producing Streptomyces using a seven transmembrane domain chemosensory receptor. *eLife*, *6*, e23770.
- Tran, T. D., & Luallen, R. J. (2024). An organismal understanding of *C. elegans* innate immune responses, from pathogen recognition to multigenerational resistance. *Seminars in Cell & Developmental Biology*, 154, 77–84.
- Troha, K., Im, J. H., Revah, J., Lazzaro, B. P., & Buchon, N. (2018). Comparative transcriptomics reveals CrebA as a novel regulator of infection tolerance in *D. melanogaster*. *PLoS Pathogens*, *14*(2), e1006847.
- Tsai, C. J.-Y., Loh, J. M. S., & Proft, T. (2016). *Galleria mellonella* infection models for the study of bacterial diseases and for antimicrobial drug testing. *Virulence*, 7(3), 214–229.
- Uchida, K., Ohmori, D., Yamakura, F., & Suzuki, K. (1990). Changes in free amino acid concentration in the hemolymph of the female *Culex pipiens pallens* (Diptera: Culicidae), after a blood meal. *Journal of Medical Entomology*, 27(3), 302–308.
- Van den Bergh, B. (2022). Bugs on drugs: A Drosophila melanogaster gut model to study in vivo antibiotic tolerance of E. coli. Microorganisms, 10(1), 119.

- van den Hoogen, J., Geisen, S., Routh, D., Ferris, H., Traunspurger, W., Wardle, D. A., de Goede, R. G. M., Adams, B. J., Ahmad, W., Andriuzzi, W. S., Bardgett, R. D., Bonkowski, M., Campos-Herrera, R., Cares, J. E., Caruso, T., de Brito Caixeta, L., Chen, X., Costa, S. R., Creamer, R., ... Crowther, T. W. (2019). Soil nematode abundance and functional group composition at a global scale. *Nature*, *572*(7768), 194–198.
- Van Herreweghe, J. M., & Michiels, C. W. (2012). Invertebrate lysozymes: Diversity and distribution, molecular mechanism and *in vivo* function. *Journal of Biosciences*, 37(2), 327–348.
- Vilcinskas, A. (2013). Evolutionary plasticity of insect immunity. Journal of Insect Physiology, 59(2), 123-129.
- Wade, D., Boman, A., Wåhlin, B., Drain, C. M., Andreu, D., Boman, H. G., & Merrifield, R. B. (1990). All-D amino acid-containing channel-forming antibiotic peptides. *Proceedings of the National Academy of Sciences of the United States of America*, 87(12), 4761–4765.
- Wagner, D. L. (2020). Insect declines in the anthropocene. Annual Review of Entomology, 65(1), 457-480.
- Wang, L., Weber, A. N. R., Atilano, M. L., Filipe, S. R., Gay, N. J., & Ligoxygakis, P. (2006). Sensing of Gram-positive bacteria in *Drosophila*: GNBP1 is needed to process and present peptidoglycan to PGRP-SA. *EMBO Journal*, 25(20), 5005–5014.
- Wang, Q., Ren, M., Liu, X., Xia, H., & Chen, K. (2019). Peptidoglycan recognition proteins in insect immunity. *Molecular Immunology*, 106, 69–76.
- Wang, S., & Beerntsen, B. T. (2015). Functional implications of the peptidoglycan recognition proteins in the immunity of the yellow fever mosquito, *Aedes aegypti. Insect Molecular Biology*, 24(3), 293–310.
- Ward, S. (1973). Chemotaxis by the nematode *Caenorhabditis elegans*: Identification of attractants and analysis of the response by use of mutants. *Proceedings of the National Academy of Sciences of the United States of America*, 70(3), 817–821.
- Wei, G., de Leeuw, E., Pazgier, M., Yuan, W., Zou, G., Wang, J., Ericksen, B., Lu, W.-Y., Lehrer, R. I., & Lu, W. (2009). Through the looking glass, mechanistic insights from enantiomeric human defensins. *Journal of Biological Chemistry*, 284(42), 29180–29192.
- Wells, C. L., Jechorek, R. P., & Maddaus, M. A. (1988). The translocation of intestinal facultative and anaerobic bacteria in defined flora mice. *Microbial Ecology in Health and Disease*, 1(4), 227–235.
- Wiesner, A. (1992). Characteristics of inert beads provoking humoral immune responses in *Galleria mellonella* larvae. *Journal of Insect Physiology*, *38*(7), 533–541.
- Worthy, S. E., Haynes, L., Chambers, M., Bethune, D., Kan, E., Chung, K., Ota, R., Taylor, C. J., & Glater, E. E. (2018). Identification of attractive odorants released by preferred bacterial food found in the natural habitats of *C. elegans. PloS ONE*, 13(7), e0201158.
- Wu, Q., Patočka, J., & Kuča, K. (2018). Insect antimicrobial peptides, a mini review. Toxins, 10(11), 461.
- Xu, Y., Zhao, Z., Tong, W., Ding, Y., Liu, B., Shi, Y., Wang, J., Sun, S., Liu, M., Wang, Y., Qi, Q., Xian, M., & Zhao, G. (2020). An acid-tolerance response system protecting exponentially growing *Escherichia coli*. *Nature Communications*, 11(1), 1496.
- Yang, L. H., & Gratton, C. (2014). Insects as drivers of ecosystem processes. *Current Opinion in Insect Science*, 2, 26–32.

- Yasuhara, M., & Danovaro, R. (2016). Temperature impacts on deep-sea biodiversity: Temperature impacts on deep-sea biodiversity. *Biological Reviews of the Cambridge Philosophical Society*, 91(2), 275–287.
- Zanotto, F. P., Raubenheimer, D., & Simpson, S. J. (1996). Haemolymph amino acid and sugar levels in locusts fed nutritionally unbalanced diets. *Journal of Comparative Physiology B*, 166(3), 223–229.
- Zasloff, M. (2002). Antimicrobial peptides of multicellular organisms. Nature, 415(6870), 389-395.
- Zhang, F., Berg, M., Dierking, K., Félix, M.-A., Shapira, M., Samuel, B. S., & Schulenburg, H. (2017). Caenorhabditis elegans as a model for microbiome research. Frontiers in Microbiology, 8, 485.
- Zhang, L., Li, L., Guo, X., Litman, G. W., Dishaw, L. J., & Zhang, G. (2015). Massive expansion and functional divergence of innate immune genes in a protostome. *Scientific Reports*, 5(1), 8693.
- Zhang, Y., Lu, H., & Bargmann, C. I. (2005). Pathogenic bacteria induce aversive olfactory learning in *Caenorhabditis elegans*. *Nature*, *438*(7065), 179–184.
- Zhu, X., Mu, K., Wan, Y., & Zhang, L. (2022). Evolutionary history of the NLR gene families across lophotrochozoans. *Gene*, 843, 146807.
- Ziarati, M., Zorriehzahra, M. J., Hassantabar, F., Mehrabi, Z., Dhawan, M., Sharun, K., Emran, T. B., Dhama, K., Chaicumpa, W., & Shamsi, S. (2022). Zoonotic diseases of fish and their prevention and control. *The Veterinary Quarterly*, 42(1), 95–118.
- Zimmerman, L. M. (2020). The reptilian perspective on vertebrate immunity: 10 years of progress. Journal of Experimental Biology, 223(Pt 21), jeb214171.

# **Chapter 7**

- Aked, J., & Hall, J. L. (1993). Effect of powdery mildew infection on concentrations of apoplastic sugars in pea leaves. *New Phytologist*, *123*(2), 283–288.
- Albert, M., & Felix, G. (2010). Chimeric receptors of the *Arabidopsis thaliana* pattern recognition receptors EFR and FLS2. *Plant Signaling & Behavior*, 5(11), 1430–1432.
- Alegbeleye, O. O., Singleton, I., & Sant'Ana, A. S. (2018). Sources and contamination routes of microbial pathogens to fresh produce during field cultivation: A review. *Food Microbiology*, 73, 177–208.
- Anguita-Maeso, M., Haro, C., Montes-Borrego, M., De La Fuente, L., Navas-Cortés, J. A., & Landa, B. B. (2021). Metabolomic, ionomic and microbial characterization of olive xylem sap reveals differences according to plant age and genotype. *Agronomy*, 11(6), 1179.
- Badri, D. V., & Vivanco, J. M. (2009). Regulation and function of root exudates. *Plant, Cell & Environment*, 32(6), 666–681.
- Bailey-Serres, J., & Voesenek, L. A. C. J. (2008). Flooding stress: Acclimations and genetic diversity. Annual Review of Plant Biology, 59(1), 313–339.
- Bais, H. P., Weir, T. L., Perry, L. G., Gilroy, S., & Vivanco, J. M. (2006). The role of root exudates in rhizosphere interactions with plants and other organisms. *Annual Review of Plant Biology*, 57(1), 233–266.

- Baker, N. R., Zhalnina, K., Yuan, M., Herman, D., Ceja-Navarro, J. A., Sasse, J., Jordan, J. S., Bowen, B. P., Wu, L., Fossum, C., Chew, A., Fu, Y., Saha, M., Zhou, J., Pett-Ridge, J., Northen, T. R., & Firestone, M. K. (2024). Nutrient and moisture limitations reveal keystone metabolites linking rhizosphere metabolomes and microbiomes. *Proceedings of the National Academy of Sciences of the United States of America*, *121*(32), e2303439121.
- Balaji, V., & Smart, C. D. (2012). Over-expression of snakin-2 and extensin-like protein genes restricts pathogen invasiveness and enhances tolerance to *Clavibacter michiganensis* subsp. *michiganensis* in transgenic tomato (*Solanum lycopersicum*). *Transgenic Research*, 21(1), 23–37.
- Balint-Kurti, P. (2019). The plant hypersensitive response: Concepts, control and consequences. *Molecular Plant Pathology*, *20*(8), 1163–1178.
- Bartlett, M. K., Scoffoni, C., & Sack, L. (2012). The determinants of leaf turgor loss point and prediction of drought tolerance of species and biomes: A global meta-analysis. *Ecology Letters*, *15*(5), 393–405.
- Beattie, G. A. (2011). Water relations in the interaction of foliar bacterial pathogens with plants. *Annual Review of Phytopathology*, *49*(1), 533–555.
- Bendix, C., & Lewis, J. D. (2018). The enemy within: Phloem-limited pathogens. *Molecular Plant Pathology*, 19(1), 238–254.
- Beneduzi, A., Ambrosini, A., & Passaglia, L. M. P. (2012). Plant growth-promoting rhizobacteria (PGPR): Their potential as antagonists and biocontrol agents. *Genetics and Molecular Biology*, 35(4), 1044–1051.
- Bentham, A. R., De la Concepción, J. C., Benjumea, J. V., Kourelis, J., Jones, S., Mendel, M., Stubbs, J., Stevenson, C. E. M., Maidment, J. H. R., Youles, M., Zdrzałek, R., Kamoun, S., & Banfield, M. J. (2023). Allelic compatibility in plant immune receptors facilitates engineering of new effector recognition specificities. *The Plant Cell*, 35(10), 3809–3827.
- Biemelt, S., & Sonnewald, U. (2006). Plant-microbe interactions to probe regulation of plant carbon metabolism. *Journal of Plant Physiology*, *163*(3), 307–318.
- Boschi, F., Schvartzman, C., Murchio, S., Ferreira, V., Siri, M. I., Galván, G. A., Smoker, M., Stransfeld, L., Zipfel, C., Vilaró, F. L., & Dalla-Rizza, M. (2017). Enhanced bacterial wilt resistance in potato through expression of *Arabidopsis* EFR and introgression of quantitative resistance from *Solanum commersonii*. *Frontiers in Plant Science*, 8, 1642.
- Boutrot, F., & Zipfel, C. (2017). Function, discovery, and exploitation of plant pattern recognition receptors for broad-spectrum disease resistance. *Annual Review of Phytopathology*, 55, 257–286.
- Boyer, J. S. (1982). Plant productivity and environment. Science, 218(4571), 443-448.
- Brandl, M. T. (2006). Fitness of human enteric pathogens on plants and implications for food safety. *Annual Review* of *Phytopathology*, 44(1), 367–392.
- Braun, D. M. (2022). Phloem loading and unloading of sucrose: What a long, strange trip from source to sink. *Annual Review of Plant Biology*, 73(1), 553–584.
- Brechtel, E., Huwig, A., & Giffhorn, F. (2002). L-glucitol catabolism in *Stenotrophomonas maltophilia* Ac. *Applied and Environmental Microbiology*, 68(2), 582–587.

- Broussard, L., Abadie, C., Lalande, J., Limami, A. M., Lothier, J., & Tcherkez, G. (2023). Phloem sap composition: What have we learnt from metabolomics? *International Journal of Molecular Sciences*, 24(8), 6917.
- Bussières, P. (2014). Estimating the number and size of phloem sieve plate pores using longitudinal views and geometric reconstruction. *Scientific Reports*, *4*, 4929.
- Campos, M. L., de Souza, C. M., de Oliveira, K. B. S., Dias, S. C., & Franco, O. L. (2018). The role of antimicrobial peptides in plant immunity. *Journal of Experimental Botany*, 69(21), 4997–5011.
- Cerutti, A., Jauneau, A., Laufs, P., Leonhardt, N., Schattat, M. H., Berthomé, R., Routaboul, J.-M., & Noël, L. D. (2019). Mangroves in the leaves: Anatomy, physiology, and immunity of epithemal hydathodes. *Annual Review of Phytopathology*, 57(1), 91–116.
- Cesari, S., Xi, Y., Declerck, N., Chalvon, V., Mammri, L., Pugnière, M., Henriquet, C., de Guillen, K., Chochois, V., Padilla, A., & Kroj, T. (2022). New recognition specificity in a plant immune receptor by molecular engineering of its integrated domain. *Nature Communications*, 13, 1524.
- Chatterjee, S., Almeida, R. P. P., & Lindow, S. (2008). Living in two worlds: the plant and insect lifestyles of *Xylella fastidiosa*. *Annual Review of Phytopathology*, *46*(1), 243–271.
- Chaudhry, V., Runge, P., Sengupta, P., Doehlemann, G., Parker, J. E., & Kemen, E. (2021). Shaping the leaf microbiota: Plant-microbe-microbe interactions. *Journal of Experimental Botany*, 72(1), 36–56.
- Chen, K., Magri, G., Grasset, E. K., & Cerutti, A. (2020). Rethinking mucosal antibody responses: IgM, IgG and IgD join IgA. *Nature Reviews Immunology*, 20(7), 427–441.
- Chi, F., Shen, S.-H., Cheng, H.-P., Jing, Y.-X., Yanni, Y. G., & Dazzo, F. B. (2005). Ascending migration of endophytic rhizobia, from roots to leaves, inside rice plants and assessment of benefits to rice growth physiology. *Applied and Environmental Microbiology*, 71(11), 7271–7278.
- Choat, B., Cobb, A. R., & Jansen, S. (2008). Structure and function of bordered pits: New discoveries and impacts on whole-plant hydraulic function. *The New Phytologist*, 177(3), 608–626.
- Christensen, N. M., Axelsen, K. B., Nicolaisen, M., & Schulz, A. (2005). Phytoplasmas and their interactions with hosts. *Trends in Plant Science*, *10*(11), 526–535.
- Compant, S., Clément, C., & Sessitsch, A. (2010). Plant growth-promoting bacteria in the rhizo- and endosphere of plants: Their role, colonization, mechanisms involved and prospects for utilization. *Soil Biology & Biochemistry*, 42(5), 669–678.
- Cooley, M. B., Miller, W. G., & Mandrell, R. E. (2003). Colonization of Arabidopsis thaliana with Salmonella enterica and enterohemorrhagic Escherichia coli O157:H7 and competition by Enterobacter asburiae. Applied and Environmental Microbiology, 69(8), 4915–4926.
- Cui, H., Tsuda, K., & Parker, J. E. (2015). Effector-triggered immunity: From pathogen perception to robust defense. *Annual Review of Plant Biology*, 66(1), 487–511.
- Dardick, C., & Ronald, P. (2006). Plant and animal pathogen recognition receptors signal through non-RD kinases. *PLoS Pathogens*, 2(1), e2.

- Deering, A. J., Pruitt, R. E., Mauer, L. J., & Reuhs, B. L. (2011). Identification of the cellular location of internalized *Escherichia coli* O157:H7 in mung bean, *Vigna radiata*, by immunocytochemical techniques. *Journal of Food Protection*, 74(8), 1224–1230.
- De la Concepción, J. C., Franceschetti, M., MacLean, D., Terauchi, R., Kamoun, S., & Banfield, M. J. (2019). Protein engineering expands the effector recognition profile of a rice NLR immune receptor. *Elife*, *8*, e47713.
- De La Fuente, L., Merfa, M. V., Cobine, P. A., & Coleman, J. J. (2022). Pathogen adaptation to the xylem environment. *Annual Review of Phytopathology*, 60(1), 163–186.
- Dinant, S., Bonnemain, J.-L., Girousse, C., & Kehr, J. (2010). Phloem sap intricacy and interplay with aphid feeding. *Comptes Rendus Biologies*, *333*(6-7), 504–515.
- D'Odorico, P., Carr, J. A., Laio, F., Ridolfi, L., & Vandoni, S. (2014). Feeding humanity through global food trade. *Earth's Future*, *2*(9), 458–469.
- Doornbos, R. F., van Loon, L. C., & Bakker, P. A. H. M. (2012). Impact of root exudates and plant defense signaling on bacterial communities in the rhizosphere. *Agronomy for Sustainable Development*, 32(1), 227–243.
- Dorantes, L., Colmenero, R., Hernandez, H., Mota, L., Jaramillo, M. E., Fernandez, E., & Solano, C. (2000). Inhibition of growth of some foodborne pathogenic bacteria by *Capsicum annum* extracts. *International Journal of Food Microbiology*, 57(1-2), 125–128.
- Dragišić Maksimović, J. J., Zivanović, B. D., Maksimović, V. M., Mojović, M. D., Nikolic, M. T., & Vučinić, Z. B. (2014). Filter strip as a method of choice for apoplastic fluid extraction from maize roots. *Plant Science*, *223*, 49–58.
- Driouich, A., Follet-Gueye, M.-L., Vicré-Gibouin, M., & Hawes, M. (2013). Root border cells and secretions as critical elements in plant host defense. *Current Opinion in Plant Biology*, *16*(4), 489–495.
- Duan, Y., Zhou, L., Hall, D. G., Li, W., Doddapaneni, H., Lin, H., Liu, L., Vahling, C. M., Gabriel, D. W., Williams, K. P., Dickerman, A., Sun, Y., & Gottwald, T. (2009). Complete genome sequence of citrus huanglongbing bacterium, "*Candidatus Liberibacter asiaticus*" obtained through metagenomics. *Molecular Plant-Microbe Interactions*, 22(8), 1011–1020.
- Eichmeier, A., Kiss, T., Necas, T., Penazova, E., Tekielska, D., Bohunicka, M., Valentova, L., Cmejla, R., Morais, D., & Baldrian, P. (2019). High-throughput sequencing analysis of the bacterial community in stone fruit phloem tissues infected by "*Candidatus Phytoplasma prunorum*". *Microbial Ecology*, 77(3), 664–675.
- Ercoli, M. F., Luu, D. D., Rim, E. Y., Shigenaga, A., Teixeira de Araujo, A., Jr., Chern, M., Jain, R., Ruan, R., Joe, A., Stewart, V., & Ronald, P. (2022). Plant immunity: Rice XA21-mediated resistance to bacterial infection. *Proceedings of the National Academy of Sciences of the United States of America*, 119(8).
- Esau, K., & Cheadle, V. I. (1959). Size of pores and their contents in sieve elements of dicotyledons. *Proceedings of the National Academy of Sciences of the United States of America*, 45(2), 156–162.
- European Food Safety Authority (EFSA). (2020). Update of the *Xylella* spp. host plant database Systematic literature search up to 30 June 2019. *EFSA Journal*, 18(4), e06114.
- Fatima, U., & Senthil-Kumar, M. (2015). Plant and pathogen nutrient acquisition strategies. *Frontiers in Plant Science*, *6*, 750.

- Fernandez-Garcia, N., Hernandez, M., Casado-Vela, J., Bru, R., Elortza, F., Hedden, P., & Olmos, E. (2011).
  Changes to the proteome and targeted metabolites of xylem sap in *Brassica oleracea* in response to salt stress:
  Proteome and metabolite changes under salt stress. *Plant, Cell & Environment*, 34(5), 821–836.
- Ferrari, S., Savatin, D. V., Sicilia, F., Gramegna, G., Cervone, F., & Lorenzo, G. D. (2013). Oligogalacturonides: Plant damage-associated molecular patterns and regulators of growth and development. *Frontiers in Plant Science*, 4, 49.
- Fisher, D. B. (1990). Measurement of Phloem transport rates by an indicator-dilution technique. *Plant Physiology*, *94*(2), 455–462.
- Fones, H., & Preston, G. M. (2012). Reactive oxygen and oxidative stress tolerance in plant pathogenic Pseudomonas. FEMS Microbiology Letters, 327(1), 1–8.
- Franz, E., Visser, A. A., Van Diepeningen, A. D., Klerks, M. M., Termorshuizen, A. J., & van Bruggen, A. H. C. (2007). Quantification of contamination of lettuce by GFP-expressing *Escherichia coli* O157:H7 and *Salmonella enterica* serovar Typhimurium. *Food Microbiology*, 24(1), 106–112.
- Freeman, B. C., & Beattie, G. A. (2009). Bacterial growth restriction during host resistance to *Pseudomonas* syringae is associated with leaf water loss and localized cessation of vascular activity in *Arabidopsis thaliana*. *Molecular Plant-Microbe Interactions: MPMI*, 22(7), 857–867.
- Gabriel, R., & Kesselmeier, J. (1999). Apoplastic solute concentrations of organic acids and mineral nutrients in the leaves of several *Fagaceae*. *Plant & Cell Physiology*, *40*(6), 604–612.
- Gao, X., Huang, Q., Zhao, Z., Han, Q., Ke, X., Qin, H., & Huang, L. (2016). Studies on the infection, colonization, and movement of *Pseudomonas syringae* pv. actinidiae in kiwifruit tissues using a GFPuv-labeled strain. *PloS* ONE, 11(3), e0151169.
- Geigenberger, P. (2003). Response of plant metabolism to too little oxygen. *Current Opinion in Plant Biology*, *6*(3), 247–256.
- Gleadow, R. M., & Møller, B. L. (2014). Cyanogenic glycosides: Synthesis, physiology, and phenotypic plasticity. *Annual Review of Plant Biology*, 65(1), 155–185.
- Godson, A., & van der Hoorn, R. A. L. (2021). The front line of defence: A meta-analysis of apoplastic proteases in plant immunity. *Journal of Experimental Botany*, 72(9), 3381–3394.
- González Moreno, A., de Cózar, A., Prieto, P., Domínguez, E., & Heredia, A. (2022). Radiationless mechanism of UV deactivation by cuticle phenolics in plants. *Nature Communications*, *13*(1), 1786.
- Gu, G., Hu, J., Cevallos-Cevallos, J. M., Richardson, S. M., Bartz, J. A., & van Bruggen, A. H. C. (2011). Internal colonization of *Salmonella enterica* serovar Typhimurium in tomato plants. *PloS ONE*, *6*(11), e27340.
- Guo, X., van Iersel, M. W., Chen, J., Brackett, R. E., & Beuchat, L. R. (2002). Evidence of association of salmonellae with tomato plants grown hydroponically in inoculated nutrient solution. Applied and Environmental Microbiology, 68(7), 3639–3643.
- Hacquard, S., Spaepen, S., Garrido-Oter, R., & Schulze-Lefert, P. (2017). Interplay between innate immunity and the plant Microbiota. *Annual Review of Phytopathology*, *55*, 565–589.

- Hallmann, J. (2001). Plant interactions with endophytic bacteria. In M. J. Jeger & N. J. Spence (Eds.), *Biotic interactions in plant-pathogen associations* (pp. 87–119). CABI Publishing.
- Hao, G., Stover, E., & Gupta, G. (2016). Overexpression of a modified plant thionin enhances disease resistance to citrus canker and huanglongbing (HLB). *Frontiers in Plant Science*, 7, 1078.
- Hardoim, P. R., van Overbeek, L. S., & van Elsas, J. D. (2008). Properties of bacterial endophytes and their proposed role in plant growth. *Trends in Microbiology*, 16(10), 463–471.
- Hayashi, H., & Chino, M. (1985). Nitrate and other anions in the rice phloem sap. *Plant & Cell Physiology*, 26(2), 325–330.
- Heaton, J. C., & Jones, K. (2008). Microbial contamination of fruit and vegetables and the behaviour of enteropathogens in the phyllosphere: A review. *Journal of Applied Microbiology*, *104*(3), 613–626.
- Helft, L., Thompson, M., & Bent, A. F. (2016). Directed evolution of FLS2 towards novel flagellin peptide recognition. *PloS ONE*, 11(6), e0157155.
- Hijaz, F., & Killiny, N. (2014). Collection and chemical composition of phloem sap from *Citrus sinensis* L. Osbeck (sweet orange). *PloS ONE*, 9(7), e101830.
- Hirano, S. S., & Upper, C. D. (2000). Bacteria in the leaf ecosystem with emphasis on *Pseudomonas syringae*—a pathogen, ice nucleus, and epiphyte. *Microbiology and Molecular Biology Reviews*, 64(3), 624–653.
- Hirneisen, K. A., Sharma, M., & Kniel, K. E. (2012). Human enteric pathogen internalization by root uptake into food crops. *Foodborne Pathogens and Disease*, *9*(5), 396–405.
- Holgate, K. C. (1950). *Changes in the composition of maple sap during the tapping season* (Bulletin No. 742). New York State Agricultural Experiment Station.
- Hora, R., Warriner, K., Shelp, B. J., & Griffiths, M. W. (2005). Internalization of *Escherichia coli* O157:H7 following biological and mechanical disruption of growing spinach plants. *Journal of Food Protection*, 68(12), 2506–2509.
- Jacobsen, A. L., Pratt, R. B., Tobin, M. F., Hacke, U. G., & Ewers, F. W. (2012). A global analysis of xylem vessel length in woody plants. *American Journal of Botany*, 99(10), 1583–1591.
- Jeter, C., & Matthysse, A. G. (2005). Characterization of the binding of diarrheagenic strains of *E. coli* to plant surfaces and the role of curli in the interaction of the bacteria with alfalfa sprouts. *Molecular Plant-Microbe Interactions: MPMI*, *18*(11), 1235–1242.
- Jiang, Y., Zhang, C.-X., Chen, R., & He, S. Y. (2019). Challenging battles of plants with phloem-feeding insects and prokaryotic pathogens. *Proceedings of the National Academy of Sciences of the United States of America*, 116(47), 23390–23397.
- Jones, J. D. G., & Dangl, J. L. (2006). The plant immune system. Nature, 444(7117), 323–329.
- Jones, J. D. G., Staskawicz, B. J., & Dangl, J. L. (2024). The plant immune system: From discovery to deployment. *Cell*, 187(9), 2095–2116.

- Kakkar, A., Nizampatnam, N. R., Kondreddy, A., Pradhan, B. B., & Chatterjee, S. (2015). Xanthomonas campestris cell-cell signalling molecule DSF (diffusible signal factor) elicits innate immunity in plants and is suppressed by the exopolysaccharide xanthan. Journal of Experimental Botany, 66(21), 6697–6714.
- Kamilova, F., Kravchenko, L. V., Shaposhnikov, A. I., Azarova, T., Makarova, N., & Lugtenberg, B. (2006). Organic acids, sugars, and L-tryptophane in exudates of vegetables growing on stonewool and their effects on activities of rhizosphere bacteria. *Molecular Plant-Microbe Interactions: MPMI*, 19(3), 250–256.
- Kang, Y., Outlaw, W. H., Jr, Fiore, G. B., & Riddle, K. A. (2007). Guard cell apoplastic photosynthate accumulation corresponds to a phloem-loading mechanism. *Journal of Experimental Botany*, 58(15-16), 4061–4070.
- Kasper, K., Abreu, I. N., Feussner, K., Zienkiewicz, K., Herrfurth, C., Ischebeck, T., Janz, D., Majcherczyk, A., Schmitt, K., Valerius, O., Braus, G. H., Feussner, I., & Polle, A. (2022). Multi-omics analysis of xylem sap uncovers dynamic modulation of poplar defenses by ammonium and nitrate. *The Plant Journal*, 111(1), 282–303.
- Katagiri, F., & Tsuda, K. (2010). Understanding the plant immune system. *Molecular Plant-Microbe Interactions: MPMI*, *23*(12), 1531–1536.
- Kawarai, T., Furukawa, S., Narisawa, N., Hagiwara, C., Ogihara, H., & Yamasaki, M. (2009). Biofilm formation by *Escherichia coli* in hypertonic sucrose media. *Journal of Bioscience and Bioengineering*, *107*(6), 630–635.
- Kliebenstein, D. J., & Kvitko, B. H. (2023). Better living through phytochemistry: "Phytoavengins" and reappraising the production-focused dichotomy for defensive phytochemicals. *Physiological and Molecular Plant Pathology*, 125, 101978.
- Knoblauch, M., Froelich, D. R., Pickard, W. F., & Peters, W. S. (2014). SEORious business: Structural proteins in sieve tubes and their involvement in sieve element occlusion. *Journal of Experimental Botany*, 65(7), 1879–1893.
- Kocaçalişkan, I., Talan, I., & Terzi, I. (2006). Antimicrobial activity of catechol and pyrogallol as allelochemicals. *Z. Naturforsch. C*, *61*(9-10), 639–642.
- Kourelis, J., Marchal, C., Posbeyikian, A., Harant, A., & Kamoun, S. (2023). NLR immune receptor-nanobody fusions confer plant disease resistance. *Science*, 379(6635), 934–939.
- Kourelis, J., & van der Hoorn, R. A. L. (2018). Defended to the nines: 25 years of resistance gene cloning identifies nine mechanisms for R protein function. *The Plant Cell*, *30*(2), 285–299.
- Kroupitski, Y., Golberg, D., Belausov, E., Pinto, R., Swartzberg, D., Granot, D., & Sela, S. (2009). Internalization of *Salmonella enterica* in leaves is induced by light and involves chemotaxis and penetration through open stomata. *Applied and Environmental Microbiology*, 75(19), 6076–6086.
- Kunwar, S., Iriarte, F., Fan, Q., Evaristo da Silva, E., Ritchie, L., Nguyen, N. S., Freeman, J. H., Stall, R. E., Jones, J. B., Minsavage, G. V., Colee, J., Scott, J. W., Vallad, G. E., Zipfel, C., Horvath, D., Westwood, J., Hutton, S. F., & Paret, M. L. (2018). Transgenic expression of EFR and Bs2 genes for field management of bacterial wilt and bacterial spot of tomato. *Phytopathology*, *108*(12), 1402–1411.
- Kutter, S., Hartmann, A., & Schmid, M. (2006). Colonization of barley (*Hordeum vulgare*) with Salmonella enterica and Listeria spp: Colonization of barley with food-borne pathogens. FEMS Microbiology Ecology, 56(2), 262–271.

- Lacombe, S., Rougon-Cardoso, A., Sherwood, E., Peeters, N., Dahlbeck, D., van Esse, H. P., Smoker, M., Rallapalli, G., Thomma, B. P. H. J., Staskawicz, B., Jones, J. D. G., & Zipfel, C. (2010). Interfamily transfer of a plant pattern-recognition receptor confers broad-spectrum bacterial resistance. *Nature Biotechnology*, 28(4), 365–369.
- Lamb, T. G., Tonkyn, D. W., & Kluepfel, D. A. (1996). Movement of *Pseudomonas aureofaciens* from the rhizosphere to aerial plant tissue. *Canadian Journal of Microbiology*, *42*(11), 1112–1120.
- Lamichhane, J. R., Osdaghi, E., Behlau, F., Köhl, J., Jones, J. B., & Aubertot, J.-N. (2018). Thirteen decades of antimicrobial copper compounds applied in agriculture. A review. *Agronomy for Sustainable Development*, 38(3), 1–18.
- Lapidot, A., & Yaron, S. (2009). Transfer of *Salmonella enterica* serovar Typhimurium from contaminated irrigation water to parsley is dependent on curli and cellulose, the biofilm matrix components. *Journal of Food Protection*, 72(3), 618–623.
- Lawson, T., & Matthews, J. (2020). Guard cell metabolism and stomatal function. *Annual Review of Plant Biology*, 71(1), 273–302.
- Lemonnier, P., Gaillard, C., Veillet, F., Verbeke, J., Lemoine, R., Coutos-Thévenot, P., & La Camera, S. (2014). Expression of *Arabidopsis* sugar transport protein STP13 differentially affects glucose transport activity and basal resistance to *Botrytis cinerea*. *Plant Molecular Biology*, 85(4-5), 473–484.
- Liesche, J., Schulz, A., Jensen, K. H., Minchin, P., & Bohr, T. (2013). Theoretical and experimental determination of phloem translocation speeds in gymnosperm and angiosperm trees. *Acta Horticulturae*, *991*, 45–52.
- Li, L., Luo, Y., Li, R., Zhou, Q., Peijnenburg, W. J. G. M., Yin, N., Yang, J., Tu, C., & Zhang, Y. (2020). Effective uptake of submicrometre plastics by crop plants via a crack-entry mode. *Nature Sustainability*, *3*(11), 929–937.
- Li, L., Luo, Y., Peijnenburg, W. J. G. M., Li, R., Yang, J., & Zhou, Q. (2020). Confocal measurement of microplastics uptake by plants. *MethodsX*, 7, 100750.
- Liu, D. D., Chao, W. M., & Turgeon, R. (2012). Transport of sucrose, not hexose, in the phloem. Journal of Experimental Botany, 63(11), 4315–4320.
- Liu, H., Carvalhais, L. C., Crawford, M., Singh, E., Dennis, P. G., Pieterse, C. M. J., & Schenk, P. M. (2017). Inner plant values: Diversity, colonization and benefits from endophytic bacteria. *Frontiers in Microbiology*, 8, 2552.
- Liu, Y., Zhang, X., Yuan, G., Wang, D., Zheng, Y., Ma, M., Guo, L., Bhadauria, V., Peng, Y.-L., & Liu, J. (2021). A designer rice NLR immune receptor confers resistance to the rice blast fungus carrying noncorresponding avirulence effectors. *Proceedings of the National Academy of Sciences of the United States of America*, 118(44), e2110751118.
- Liu, Z., Hou, S., Rodrigues, O., Wang, P., Luo, D., Munemasa, S., Lei, J., Liu, J., Ortiz-Morea, F. A., Wang, X., Nomura, K., Yin, C., Wang, H., Zhang, W., Zhu-Salzman, K., He, S. Y., He, P., & Shan, L. (2022). Phytocytokine signalling reopens stomata in plant immunity and water loss. *Nature*, 605(7909), 332–339.
- Lohaus, G., Hussmann, M., Pennewiss, K., Schneider, H., Zhu, J. J., & Sattelmacher, B. (2000). Solute balance of a maize (*Zea mays* L.) source leaf as affected by salt treatment with special emphasis on phloem retranslocation and ion leaching. *Journal of Experimental Botany*, 51(351), 1721–1732.

- Lohaus, G., Pennewiss, K., Sattelmacher, B., Hussmann, M., & Muehling, K. H. (2001). Is the infiltration-centrifugation technique appropriate for the isolation of apoplastic fluid? A critical evaluation with different plant species. *Physiologia Plantarum*, 111(4), 457–465.
- López-Millán, A. F., Morales, F., Abadía, A., & Abadía, J. (2000). Effects of iron deficiency on the composition of the leaf apoplastic fluid and xylem sap in sugar beet. Implications for iron and carbon transport. *Plant Physiology*, 124(2), 873–884.
- Loreti, E., van Veen, H., & Perata, P. (2016). Plant responses to flooding stress. *Current Opinion in Plant Biology*, 33, 64–71.
- Lou, Z., Wang, H., Rao, S., Sun, J., Ma, C., & Li, J. (2012). p-Coumaric acid kills bacteria through dual damage mechanisms. *Food Control*, 25(2), 550–554.
- Maidment, J. H. R., Shimizu, M., Bentham, A. R., Vera, S., Franceschetti, M., Longya, A., Stevenson, C. E. M., De la Concepción, J. C., Białas, A., Kamoun, S., Terauchi, R., & Banfield, M. J. (2023). Effector target-guided engineering of an integrated domain expands the disease resistance profile of a rice NLR immune receptor. *eLife*, 12.
- Ma, J. F., Ryan, P. R., & Delhaize, E. (2001). Aluminium tolerance in plants and the complexing role of organic acids. *Trends in Plant Science*, 6(6), 273–278.
- Marchese, A., Barbieri, R., Sanches-Silva, A., Daglia, M., Nabavi, S. F., Jafari, N. J., Izadi, M., Ajami, M., & Nabavi, S. M. (2016). Antifungal and antibacterial activities of allicin: A review. *Trends in Food Science & Technology*, 52, 49–56.
- Marcone, C. (2014). Molecular biology and pathogenicity of phytoplasmas. *The Annals of Applied Biology*, *165*(2), 199–221.
- Matilla, M. A., & Krell, T. (2018). The effect of bacterial chemotaxis on host infection and pathogenicity. *FEMS Microbiology Reviews*, 42(1), 40–67.
- McManus, P. S., Stockwell, V. O., Sundin, G. W., & Jones, A. L. (2002). Antibiotic use in plant agriculture. *Annual Review of Phytopathology*, *40*(1), 443–465.
- Melotto, M., Underwood, W., Koczan, J., Nomura, K., & He, S. Y. (2006). Plant stomata function in innate immunity against bacterial invasion. *Cell*, *126*(5), 969–980.
- Melotto, M., Zhang, L., Oblessuc, P. R., & He, S. Y. (2017). Stomatal defense a decade later. *Plant Physiology*, *174*(2), 561–571.
- Mensi, I., Vernerey, M.-S., Gargani, D., Nicole, M., & Rott, P. (2014). Breaking dogmas: The plant vascular pathogen *Xanthomonas albilineans* is able to invade non-vascular tissues despite its reduced genome. *Open Biology*, 4(2), 130116.
- Miebach, M., Faivre, L., Schubert, D., Jameson, P., & Remus-Emsermann, M. (2024). Nonpathogenic leaf-colonizing bacteria elicit pathogen-like responses in a colonization density-dependent manner. *Plant-Environment Interactions*, 5, e10137.
- Milo, R., & Phillips, R. (2015). Cell biology by the numbers. Garland Science.

- Mimura, T., Yin, Z.-H., Wirth, E., & Dietz, K.-J. (1992). Phosphate transport and apoplastic phosphate homeostasis in barley leaves. *Plant & Cell Physiology*, *33*(5), 563–568.
- Mitre, L. K., Teixeira-Silva, N. S., Rybak, K., Magalhães, D. M., de Souza-Neto, R. R., Robatzek, S., Zipfel, C., & de Souza, A. A. (2021). The Arabidopsis immune receptor EFR increases resistance to the bacterial pathogens *Xanthomonas* and *Xylella* in transgenic sweet orange. *Plant Biotechnology Journal*, *19*(7), 1294–1296.
- Mullendore, D. L., Windt, C. W., Van As, H., & Knoblauch, M. (2010). Sieve tube geometry in relation to phloem flow. *The Plant Cell*, 22(3), 579–593.
- Nadwodnik, J., & Lohaus, G. (2008). Subcellular concentrations of sugar alcohols and sugars in relation to phloem translocation in *Plantago major*, *Plantago maritima*, *Prunus persica*, and *Apium graveolens*. *Planta*, 227(5), 1079–1089.
- Nakamura, S.-I., Akiyama, C., Sasaki, T., Hattori, H., & Chino, M. (2008). Effect of cadmium on the chemical composition of xylem exudate from oilseed rape plants (*Brassica napus* L.). Soil Science and Plant Nutrition, 54(1), 118–127.
- Nawrot, R., Barylski, J., Nowicki, G., Broniarczyk, J., Buchwald, W., & Goździcka-Józefiak, A. (2014). Plant antimicrobial peptides. *Folia Microbiologica*, 59(3), 181–196.
- Ngou, B. P. M., Ding, P., & Jones, J. D. G. (2022). Thirty years of resistance: Zig-zag through the plant immune system. *The Plant Cell*, *34*(5), 1447–1478.
- Noll, G. A., Furch, A. C. U., Rose, J., Visser, F., & Prüfer, D. (2022). Guardians of the phloem forisomes and beyond. *The New Phytologist*, 236(4), 1245–1260.
- O'Brien, R. D., & Lindow, S. E. (1989). Effect of plant species and environmental conditions on epiphytic population sizes of *Pseudomonas syringae* and other bacteria. *Phytopathology*, *79*(5), 619-627.
- O'Leary, B. M., Neale, H. C., Geilfus, C.-M., Jackson, R. W., Arnold, D. L., & Preston, G. M. (2016). Early changes in apoplast composition associated with defence and disease in interactions between *Phaseolus vulgaris* and the halo blight pathogen *Pseudomonas syringae* Pv. phaseolicola. *Plant, Cell & Environment, 39*(10), 2172–2184.
- Oliver, R., & Hewitt, H. G. (2014). Fungicides in crop protection (2nd ed.). CABI Publishing.
- Oner-Sieben, S., & Lohaus, G. (2014). Apoplastic and symplastic phloem loading in *Quercus robur* and *Fraxinus* excelsior. Journal of Experimental Botany, 65(7), 1905–1916.
- Osbourn, A. E. (1996). Preformed antimicrobial compounds and plant defense against fungal attack. *The Plant Cell*, *8*(10), 1821–1831.
- Oshima, K., Kakizawa, S., Nishigawa, H., Jung, H.-Y., Wei, W., Suzuki, S., Arashida, R., Nakata, D., Miyata, S.-I., Ugaki, M., & Namba, S. (2004). Reductive evolution suggested from the complete genome sequence of a plant-pathogenic phytoplasma. *Nature Genetics*, 36(1), 27–29.
- Paauw, M., van Hulten, M., Chatterjee, S., Berg, J. A., Taks, N. W., Giesbers, M., Richard, M. M. S., & van den Burg, H. A. (2023). Hydathode immunity protects the *Arabidopsis* leaf vasculature against colonization by bacterial pathogens. *Current Biology*, 33(4), 697–710.e6.

- Pail, M., Peterbauer, T., Seiboth, B., Hametner, C., Druzhinina, I., & Kubicek, C. P. (2004). The metabolic role and evolution of L-arabinitol 4-dehydrogenase of *Hypocrea jecorina*. *European Journal of Biochemistry*, 271(10), 1864–1872.
- Park, H.-Y., Park, C.-S., Kim, H.-J., & Oh, D.-K. (2007). Substrate specificity of a galactose 6-phosphate isomerase from *Lactococcus lactis* that produces d-allose from d-psicose. *Journal of Biotechnology*, 132(1), 88–95.
- Pérez-Donoso, A. G., Sun, Q., Roper, M. C., Greve, L. C., Kirkpatrick, B., & Labavitch, J. M. (2010). Cell wall-degrading enzymes enlarge the pore size of intervessel pit membranes in healthy and *Xylella fastidiosa*-infected grapevines. *Plant Physiology*, 152(3), 1748–1759.
- Perulli, G. D., Gaggia, F., Sorrenti, G., Donati, I., Boini, A., Bresilla, K., Manfrini, L., Baffoni, L., Di Gioia, D., Grappadelli, L. C., Spinelli, F., & Morandi, B. (2021). Treated wastewater as irrigation source: A microbiological and chemical evaluation in apple and nectarine trees. *Agricultural Water Management*, 244, 106403.
- Pfeilmeier, S., Caly, D. L., & Malone, J. G. (2016). Bacterial pathogenesis of plants: Future challenges from a microbial perspective: Challenges in Bacterial Molecular Plant Pathology. *Molecular Plant Pathology*, 17(8), 1298–1313.
- Podgórska, A., Burian, M., & Szal, B. (2017). Extra-cellular but extra-ordinarily important for cells: Apoplastic reactive oxygen species metabolism. *Frontiers in Plant Science*, *8*, 1353.
- Prima-Putra, D., & Botton, B. (1998). Organic and inorganic compounds of xylem exudates from five woody plants at the stage of bud breaking. *Journal of Plant Physiology*, *153*(5-6), 670–676.
- Raaijmakers, J. M., Paulitz, T. C., Steinberg, C., Alabouvette, C., & Moënne-Loccoz, Y. (2009). The rhizosphere: A playground and battlefield for soilborne pathogens and beneficial microorganisms. *Plant and Soil*, 321(1-2), 341–361.
- Ranf, S., Gisch, N., Schäffer, M., Illig, T., Westphal, L., Knirel, Y. A., Sánchez-Carballo, P. M., Zähringer, U., Hückelhoven, R., Lee, J., & Scheel, D. (2015). A lectin S-domain receptor kinase mediates lipopolysaccharide sensing in *Arabidopsis thaliana*. *Nature Immunology*, 16(4), 426–433.
- Rapicavoli, J., Ingel, B., Blanco-Ulate, B., Cantu, D., & Roper, C. (2018). *Xylella fastidiosa*: An examination of a re-emerging plant pathogen. *Molecular Plant Pathology*, 19(4), 786–800.
- Rapp, J. M., Lutz, D. A., Huish, R. D., Dufour, B., Ahmed, S., Morelli, T. L., & Stinson, K. A. (2019). Finding the sweet spot: Shifting optimal climate for maple syrup production in North America. *Forest Ecology and Management*, 448, 187–197.
- Reinhold-Hurek, B., & Hurek, T. (1998). Life in grasses: Diazotrophic endophytes. *Trends in Microbiology*, *6*(4), 139–144.
- Reinhold-Hurek, B., & Hurek, T. (2011). Living inside plants: Bacterial endophytes. *Current Opinion in Plant Biology*, 14(4), 435–443.
- Rico, A., & Preston, G. M. (2008). Pseudomonas syringae pv. tomato DC3000 uses constitutive and apoplast-induced nutrient assimilation pathways to catabolize nutrients that are abundant in the tomato apoplast. Molecular Plant-Microbe Interactions: MPMI, 21(2), 269–282.

- Roos, I. M. M., & Hattingh, M. J. (1987). Systemic invasion of plum leaves and shoots by *Pseudomonas syringae* pv. *Syringae* introduced into petiole. *Phytopathology*, 77(9), 1253–1257.
- Roper, M. C., Greve, L. C., Warren, J. G., Labavitch, J. M., & Kirkpatrick, B. C. (2007). *Xylella fastidiosa* requires polygalacturonase for colonization and pathogenicity in *Vitis vinifera* grapevines. *Molecular Plant-Microbe Interactions: MPMI*, 20(4), 411–419.
- Roussin-Léveillée, C., Mackey, D., Ekanayake, G., Gohmann, R., & Moffett, P. (2024). Extracellular niche establishment by plant pathogens. *Nature Reviews Microbiology*, 22(6), 360–372.
- Ruan, Y.-L. (2014). Sucrose metabolism: Gateway to diverse carbon use and sugar signaling. *Annual Review of Plant Biology*, 65(1), 33–67.
- Ryan, P. R., Delhaize, E., & Jones, D. L. (2001). Function and mechanism of organic anion exudation from plant roots. *Annual Review of Plant Physiology and Plant Molecular Biology*, *52*(1), 527–560.
- Ryser, E. T., Hao, J., & Yan, Z. (2009). Internalization of pathogens in produce. In *Microbial Safety of Fresh Produce* (pp. 55–80). Wiley-Blackwell.
- Saijo, Y., & Loo, E. P.-I. (2020). Plant immunity in signal integration between biotic and abiotic stress responses. *The New Phytologist*, 225(1), 87–104.
- Santelia, D., & Lawson, T. (2016). Rethinking guard cell metabolism. Plant Physiology, 172(3), 1371–1392.
- Savary, S., Willocquet, L., Pethybridge, S. J., Esker, P., McRoberts, N., & Nelson, A. (2019). The global burden of pathogens and pests on major food crops. *Nature Ecology & Evolution*, 3(3), 430–439.
- Savatin, D. V., Gramegna, G., Modesti, V., & Cervone, F. (2014). Wounding in the plant tissue: the defense of a dangerous passage. *Frontiers in Plant Science*, 5, 470.
- Scheie, P. O., & Rehberg, R. (1972). Response of *Escherichia coli* B-r to high concentrations of sucrose in a nutrient medium. *Journal of Bacteriology*, 109(1), 229–235.
- Schellenberger, R., Touchard, M., Clément, C., Baillieul, F., Cordelier, S., Crouzet, J., & Dorey, S. (2019). Apoplastic invasion patterns triggering plant immunity: Plasma membrane sensing at the frontline. *Molecular Plant Pathology*, 20(11), 1602–1616.
- Schmidt, S. J., & Fontana, A. J., Jr. (2008). Appendix E: Water activity values of select food ingredients and products. In G. V. Barbosa-Cánovas, A. J. Fontana, Jr., S. J. Schmidt, & T. P. Labuza (Eds.), *Water activity in foods: Fundamentals and applications* (pp. 407–420). Blackwell Publishing.
- Schwessinger, B., Bahar, O., Thomas, N., Holton, N., Nekrasov, V., Ruan, D., Canlas, P. E., Daudi, A., Petzold, C.
  J., Singan, V. R., Kuo, R., Chovatia, M., Daum, C., Heazlewood, J. L., Zipfel, C., & Ronald, P. C. (2015).
  Transgenic expression of the dicotyledonous pattern recognition receptor EFR in rice leads to ligand-dependent activation of defense responses. *PLoS Pathogens*, 11(3), e1004809.
- Sharma, P., Jha, A. B., Dubey, R. S., & Pessarakli, M. (2012). Reactive oxygen species, oxidative damage, and antioxidative defense mechanism in plants under stressful conditions. *Journal of Botany*, 2012(1), 1–26.
- Sharp, R. G., & Davies, W. J. (2009). Variability among species in the apoplastic pH signalling response to drying soils. *Journal of Experimental Botany*, 60(15), 4363–4370.

- Shi, S., Richardson, A. E., O'Callaghan, M., DeAngelis, K. M., Jones, E. E., Stewart, A., Firestone, M. K., & Condron, L. M. (2011). Effects of selected root exudate components on soil bacterial communities: Root exudate components and soil microbial communities. *FEMS Microbiology Ecology*, 77(3), 600–610.
- Singh, S. K., Wu, X., Shao, C., & Zhang, H. (2022). Microbial enhancement of plant nutrient acquisition. Stress Biology, 2(1), 3.
- Solomon, E. B., & Matthews, K. R. (2005). Use of fluorescent microspheres as a tool to investigate bacterial interactions with growing plants. *Journal of Food Protection*, 68(4), 870–873.
- Solomon, E. B., Yaron, S., & Matthews, K. R. (2002). Transmission of *Escherichia coli* O157:H7 from contaminated manure and irrigation water to lettuce plant tissue and its subsequent internalization. *Applied and Environmental Microbiology*, 68(1), 397–400.
- Stadler, R., Wright, K. M., Lauterbach, C., Amon, G., Gahrtz, M., Feuerstein, A., Oparka, K. J., & Sauer, N. (2005). Expression of GFP-fusions in *Arabidopsis* companion cells reveals non-specific protein trafficking into sieve elements and identifies a novel post-phloem domain in roots: Protein trafficking in the phloem. *The Plant Journal*, 41(2), 319–331.
- Stec, B., Markman, O., Rao, U., Heffron, G., Henderson, S., Vernon, L. P., Brumfeld, V., & Teeter, M. M. (2004). Proposal for molecular mechanism of thionins deduced from physico-chemical studies of plant toxins. *The Journal of Peptide Research: Official Journal of the American Peptide Society*, 64(6), 210–224.
- Stes, E., Francis, I., Pertry, I., Dolzblasz, A., Depuydt, S., & Vereecke, D. (2013). The leafy gall syndrome induced by *Rhodococcus fascians*. *FEMS Microbiology Letters*, 342(2), 187–194.
- Stockwell, V. O., & Duffy, B. (2012). Use of antibiotics in plant agriculture. *Revue Scientifique et Technique (International Office of Epizootics)*, 31(1), 199–210.
- Sun, Q., Sun, Y., Walker, M. A., & Labavitch, J. M. (2013). Vascular occlusions in grapevines with Pierce's disease make disease symptom development worse. *Plant Physiology*, 161(3), 1529–1541.
- Tam, J. P., Wang, S., Wong, K. H., & Tan, W. L. (2015). Antimicrobial peptides from plants. *Pharmaceuticals*, 8(4), 711–757.
- Tarvainen, L., Henriksson, N., Näsholm, T., & Marshall, J. D. (2023). Among-species variation in sap pH affects the xylem CO2 transport potential in trees. *The New Phytologist*, 238(3), 926–931.
- Tetlow, I. J., & Farrar, J. F. (1993). Apoplastic sugar concentration and pH in barley leaves infected with brown rust. *Journal of Experimental Botany*, 44(5), 929–936.
- Thompson, M. V., & Holbrook, N. M. (2003). Scaling phloem transport: Water potential equilibrium and osmoregulatory flow. *Plant, Cell & Environment, 26*(9), 1561–1577.
- Thompson, M. V., & Wolniak, S. M. (2008). A plasma membrane-anchored fluorescent protein fusion illuminates sieve element plasma membranes in *Arabidopsis* and tobacco. *Plant Physiology*, *146*(4), 1599–1610.
- Tiku, A. R. (2020). Antimicrobial compounds (phytoanticipins and phytoalexins) and their role in plant defense. In J.-M. Mérillon & K. G. Ramawat (Eds.), *Co-evolution of secondary metabolites* (pp. 845–868). Springer.
- Tyree, M. T., & Zimmermann, M. H. (2011). Xylem structure and the ascent of sap. Springer.

- van Dongen, J. T., Schurr, U., Pfister, M., & Geigenberger, P. (2003). Phloem metabolism and function have to cope with low internal oxygen. *Plant Physiology*, *131*(4), 1529–1543.
- van Lenteren, J. C., Bolckmans, K., Köhl, J., Ravensberg, W. J., & Urbaneja, A. (2018). Biological control using invertebrates and microorganisms: Plenty of new opportunities. *BioControl*, 63(1), 39–59.
- Velásquez, A. C., Nomura, K., Cooper, M. D., Herrin, B. R., & He, S. Y. (2017). Leucine-rich-repeat-containing variable lymphocyte receptors as modules to target plant-expressed proteins. *Plant Methods*, 13(1), 29.
- Wachtel, M. R., Whitehand, L. C., & Mandrell, R. E. (2002). Prevalence of *Escherichia coli* associated with a cabbage crop inadvertently irrigated with partially treated sewage wastewater. *Journal of Food Protection*, 65(3), 471–475.
- Walker, R. P., Bonghi, C., Varotto, S., Battistelli, A., Burbidge, C. A., Castellarin, S. D., Chen, Z.-H., Darriet, P., Moscatello, S., Rienth, M., Sweetman, C., & Famiani, F. (2021). Sucrose metabolism and transport in grapevines, with emphasis on berries and leaves, and insights gained from a cross-species comparison. *International Journal of Molecular Sciences*, 22(15), 7794.
- Wang, H., McTavish, C., & Turechek, W. W. (2018). Colonization and Movement of *Xanthomonas fragariae* in Strawberry Tissues. *Phytopathology*, 108(6), 681–690.
- Wezner-Ptasinska, M., & Otlewski, J. (2015). Selection of specific interactors from phage display library based on sea lamprey variable lymphocyte receptor sequences. *Biochimica et Biophysica Acta*, *1854*(12), 1833–1841.
- Willmann, R., Lajunen, H. M., Erbs, G., Newman, M.-A., Kolb, D., Tsuda, K., Katagiri, F., Fliegmann, J., Bono, J.-J., Cullimore, J. V., Jehle, A. K., Götz, F., Kulik, A., Molinaro, A., Lipka, V., Gust, A. A., & Nürnberger, T. (2011). *Arabidopsis* lysin-motif proteins LYM1 LYM3 CERK1 mediate bacterial peptidoglycan sensing and immunity to bacterial infection. *Proceedings of the National Academy of Sciences of the United States of America*, 108(49), 19824–19829.
- Will, T., Furch, A. C. U., & Zimmermann, M. R. (2013). How phloem-feeding insects face the challenge of phloem-located defenses. *Frontiers in Plant Science*, 4, 336.
- Windt, C. W., Vergeldt, F. J., de Jager, P. A., & van As, H. (2006). MRI of long-distance water transport: A comparison of the phloem and xylem flow characteristics and dynamics in poplar, castor bean, tomato and tobacco. *Plant, Cell & Environment*, 29(9), 1715–1729.
- Wippel, K. (2023). Plant and microbial features governing an endophytic lifestyle. *Current Opinion in Plant Biology*, 76, 102483.
- Wojtaszek, P. (1997). Oxidative burst: An early plant response to pathogen infection. *Biochemical Journal*, 322(3), 681–692.
- Wright, C. A., & Beattie, G. A. (2004). Pseudomonas syringae pv. tomato cells encounter inhibitory levels of water stress during the hypersensitive response of Arabidopsis thaliana. Proceedings of the National Academy of Sciences of the United States of America, 101(9), 3269–3274.
- Wu, J., Reca, I.-B., Spinelli, F., Lironi, D., De Lorenzo, G., Poltronieri, P., Cervone, F., Joosten, M. H. A. J., Ferrari, S., & Brutus, A. (2019). An EFR-Cf-9 chimera confers enhanced resistance to bacterial pathogens by SOBIR1and BAK1-dependent recognition of elf18. *Molecular Plant Pathology*, 20(6), 751–764.

- Wu, Y., Qu, M., Pu, X., Lin, J., & Shu, B. (2020). Distinct microbial communities among different tissues of citrus tree *Citrus reticulata* cv. Chachiensis. *Scientific Reports*, 10(1), 6068.
- Yamada, K., Saijo, Y., Nakagami, H., & Takano, Y. (2016). Regulation of sugar transporter activity for antibacterial defense in *Arabidopsis. Science*, 354(6318), 1427–1430.
- Yeh, A. H.-W., Norn, C., Kipnis, Y., Tischer, D., Pellock, S. J., Evans, D., Ma, P., Lee, G. R., Zhang, J. Z., Anishchenko, I., Coventry, B., Cao, L., Dauparas, J., Halabiya, S., DeWitt, M., Carter, L., Houk, K. N., & Baker, D. (2023). *De novo* design of luciferases using deep learning. *Nature*, 614(7949), 774–780.
- Zdrzałek, R., Xi, Y., Langner, T., Bentham, A. R., Petit-Houdenot, Y., De la Concepción, J. C., Harant, A., Shimizu, M., Were, V., Talbot, N. J., Terauchi, R., Kamoun, S., & Banfield, M. J. (2024). Bioengineering a plant NLR immune receptor with a robust binding interface toward a conserved fungal pathogen effector. *Proceedings of the National Academy of Sciences of the United States of America*, 121(28), e2402872121.
- Zeng, W., & He, S. Y. (2010). A prominent role of the flagellin receptor FLAGELLIN-SENSING2 in mediating stomatal response to *Pseudomonas syringae* pv tomato DC3000 in *Arabidopsis*. *Plant Physiology*, 153(3), 1188–1198.
- Zhang, J., Wang, Y., Chai, B., Wang, J., Li, L., Liu, M., Zhao, G., Yao, L., Gao, X., Yin, Y., & Xu, J. (2020). Efficient and low-cost error removal in DNA synthesis by a high-durability MutS. ACS Synthetic Biology, 9(4), 940–952.
- Zheng, D., Huang, C., Huang, H., Zhao, Y., Khan, M. R. U., Zhao, H., & Huang, L. (2020). Antibacterial mechanism of curcumin: A review. *Chemistry & Biodiversity*, 17(8), e2000171.
- Zhu, L., Zeng, J., Wang, C., & Wang, J. (2022). Structural basis of pore formation in the mannose phosphotransferase system by pediocin PA-1. *Applied and Environmental Microbiology*, 88(3), e0199221.
- Zipfel, C., Robatzek, S., Navarro, L., Oakeley, E. J., Jones, J. D. G., Felix, G., & Boller, T. (2004). Bacterial disease resistance in *Arabidopsis* through flagellin perception. *Nature*, *428*(6984), 764–767.
- Zou, X., Jiang, X., Xu, L., Lei, T., Peng, A., He, Y., Yao, L., & Chen, S. (2017). Transgenic citrus expressing synthesized cecropin B genes in the phloem exhibits decreased susceptibility to Huanglongbing. *Plant Molecular Biology*, 93(4-5), 341–353.
- Zuluaga, A. P., Puigvert, M., & Valls, M. (2013). Novel plant inputs influencing *Ralstonia solanacearum* during infection. *Frontiers in Microbiology*, *4*, 349.

# **Chapter 8**

Abdellahoum, Z., Maurin, M., & Bitam, I. (2021). Tularemia as a mosquito-borne disease. *Microorganisms*, 9(1), 26.

- Abergel, R. J., Zawadzka, A. M., Hoette, T. M., & Raymond, K. N. (2009). Enzymatic hydrolysis of trilactone siderophores: where chiral recognition occurs in enterobactin and bacillibactin iron transport. *Journal of the American Chemical Society*, *131*(35), 12682–12692.
- Acedo, J. Z., Chiorean, S., Vederas, J. C., & van Belkum, M. J. (2018). The expanding structural variety among bacteriocins from Gram-positive bacteria. *FEMS Microbiology Reviews*, 42(6), 805–828.

- Adeleke, R., Nwangburuka, C., & Oboirien, B. (2017). Origins, roles and fate of organic acids in soils: A review. *South African Journal of Botany*, *108*, 393–406.
- Adler, J., Hazelbauer, G. L., & Dahl, M. M. (1973). Chemotaxis toward sugars in *Escherichia coli*. Journal of Bacteriology, 115(3), 824–847.
- Alegbeleye, O. O., Singleton, I., & Sant'Ana, A. S. (2018). Sources and contamination routes of microbial pathogens to fresh produce during field cultivation: A review. *Food Microbiology*, *73*, 177–208.
- Ali, H., Hossain, M. F., Hasan, M. M., & Abujar, S. (2020). COVID-19 Dataset: Worldwide spread log including countries first case and first death. *Data in Brief*, 32, 106173.
- Alves, J. M., Carneiro, M., Day, J. P., Welch, J. J., Duckworth, J. A., Cox, T. E., Letnic, M., Strive, T., Ferrand, N., & Jiggins, F. M. (2022). A single introduction of wild rabbits triggered the biological invasion of Australia. *Proceedings of the National Academy of Sciences of the United States of America*, 119(35), e2122734119.
- Anderson, R. M., Jackson, H. C., May, R. M., & Smith, A. M. (1981). Population dynamics of fox rabies in Europe. *Nature*, 289(5800), 765–771.
- Andow, D. A., Kareiva, P. M., Levin, S. A., & Okubo, A. (1990). Spread of invading organisms. *Landscape Ecology*, 4(2-3), 177–188.
- Ansong, M., & Pickering, C. (2013). Are weeds hitchhiking a ride on your car? A systematic review of seed dispersal on cars. *PloS ONE*, 8(11), e80275.
- Aoki, S. K., Diner, E. J., de Roodenbeke, C. T., Burgess, B. R., Poole, S. J., Braaten, B. A., Jones, A. M., Webb, J. S., Hayes, C. S., Cotter, P. A., & Low, D. A. (2010). A widespread family of polymorphic contact-dependent toxin delivery systems in bacteria. *Nature*, 468(7322), 439–442.
- Baka, M., Van Derlinden, E., Boons, K., Mertens, L., & Van Impe, J. F. (2013). Impact of pH on the cardinal temperatures of *E. coli* K12: Evaluation of the gamma hypothesis. *Food Control*, 29(2), 328–335.
- Banaszek, A., Bumm, T. G. P., Nowotny, B., Geis, M., Jacob, K., Wölfl, M., Trebing, J., Kucka, K., Kouhestani, D., Gogishvili, T., Krenz, B., Lutz, J., Rasche, L., Hönemann, D., Neuweiler, H., Heiby, J. C., Bargou, R. C., Wajant, H., Einsele, H., Riethmüller, G., ... Stuhler, G. (2019). On-target restoration of a split T cell-engaging antibody for precision immunotherapy. *Nature Communications*, 10(1), 5387.
- Barisch, C., Kalinina, V., Lefrançois, L. H., Appiah, J., López-Jiménez, A. T., & Soldati, T. (2018). Localization of all four ZnT zinc transporters in *Dictyostelium* and impact of ZntA and ZntB knockout on bacteria killing. *Journal of Cell Science*, 131(23), jcs222000.
- Barker, J., Humphrey, T. J., & Brown, M. W. R. (1999). Survival of *Escherichia coli* O157 in a soil protozoan: Implications for disease. *FEMS Microbiology Letters*, *173*(2), 291–295.
- Barnes, I., Fourie, A., Wingfield, M. J., Harrington, T. C., McNew, D. L., Sugiyama, L. S., Luiz, B. C., Heller, W. P., & Keith, L. M. (2018). New *Ceratocystis* species associated with rapid death of *Metrosideros polymorpha* in Hawai'i. *Persoonia: Molecular Phylogeny and Evolution of Fungi*, 40, 154–181.
- Bar-On, Y. M., & Milo, R. (2019). The biomass composition of the oceans: A blueprint of our blue planet. *Cell*, *179*(7), 1451–1454.

- Bar-On, Y. M., Phillips, R., & Milo, R. (2018). The biomass distribution on Earth. Proceedings of the National Academy of Sciences of the United States of America, 115(25), 6506–6511.
- Baym, M., Lieberman, T. D., Kelsic, E. D., Chait, R., Gross, R., Yelin, I., & Kishony, R. (2016). Spatiotemporal microbial evolution on antibiotic landscapes. *Science*, 353(6304), 1147–1151.
- Bayro, M. J., Mukhopadhyay, J., Swapna, G. V. T., Huang, J. Y., Ma, L.-C., Sineva, E., Dawson, P. E., Montelione, G. T., & Ebright, R. H. (2003). Structure of antibacterial peptide microcin J25: A 21-residue lariat protoknot. *Journal of the American Chemical Society*, 125(41), 12382–12383.
- Beckett, S. J., Demory, D., Coenen, A. R., Casey, J. R., Dugenne, M., Follett, C. L., Connell, P., Carlson, M. C. G., Hu, S. K., Wilson, S. T., Muratore, D., Rodriguez-Gonzalez, R. A., Peng, S., Becker, K. W., Mende, D. R., Armbrust, E. V., Caron, D. A., Lindell, D., White, A. E., ... Weitz, J. S. (2024). Disentangling top-down drivers of mortality underlying diel population dynamics of *Prochlorococcus* in the North Pacific Subtropical Gyre. *Nature Communications*, 15(1), 2105.
- Behrens, H. M., Lowe, E. D., Gault, J., Housden, N. G., Kaminska, R., Weber, T. M., Thompson, C. M. A., Mislin, G. L. A., Schalk, I. J., Walker, D., Robinson, C. V., & Kleanthous, C. (2020). Pyocin S5 import into *Pseudomonas aeruginosa* reveals a generic mode of bacteriocin transport. *mBio*, 11(2), e03230-19.
- Beloin, C., Roux, A., & Ghigo, J. M. (2008). Escherichia coli biofilms. Current Topics in Microbiology and Immunology, 322, 249–289.
- Bennett, B. D., Kimball, E. H., Gao, M., Osterhout, R., Van Dien, S. J., & Rabinowitz, J. D. (2009). Absolute metabolite concentrations and implied enzyme active site occupancy in *Escherichia coli*. *Nature Chemical Biology*, 5(8), 593–599.
- Berne, C., Ellison, C. K., Ducret, A., & Brun, Y. V. (2018). Bacterial adhesion at the single-cell level. *Nature Reviews Microbiology*, 16(10), 616–627.
- Bertilsson, S., Berglund, O., Chisholm, S. W., & Pullin, M. J. (2005). Release of dissolved organic matter by *Prochlorococcus. Vie Et Milieu-Life and Environment*, 55(3-4), 225-231.
- Bertilsson, S., Berglund, O., Karl, D. M., & Chisholm, S. W. (2003). Elemental composition of marine *Prochlorococcus* and *Synechococcus*: Implications for the ecological stoichiometry of the sea. *Limnology and Oceanography*, 48(5), 1721–1731.
- Beukes, M., Bierbaum, G., Sahl, H. G., & Hastings, J. W. (2000). Purification and partial characterization of a murein hydrolase, millericin B, produced by *Streptococcus milleri* NMSCC 061. *Applied and Environmental Microbiology*, 66(1), 23–28.
- Beversdorf, L. J., Bornstein-Forst, S. M., & McLellan, S. L. (2007). The potential for beach sand to serve as a reservoir for *Escherichia coli* and the physical influences on cell die-off. *Journal of Applied Microbiology*, *102*(5), 1372–1381.
- Biller, S. J., Berube, P. M., Lindell, D., & Chisholm, S. W. (2015). Prochlorococcus: The structure and function of collective diversity. Nature Reviews Microbiology, 13(1), 13–27.
- Blount, Z. D. (2015). The unexhausted potential of E. coli. eLife, 4, e05826.
- Blount, Z. D., Barrick, J. E., Davidson, C. J., & Lenski, R. E. (2012). Genomic analysis of a key innovation in an experimental *Escherichia coli* population. *Nature*, 489(7417), 513–518.

- Bodinier, R., Leiba, J., Sabra, A., Jauslin, T. N., Lamrabet, O., Guilhen, C., Marchetti, A., Iwade, Y., Kawata, T., Lima, W. C., & Cosson, P. (2020). LrrkA, a kinase with leucine-rich repeats, links folate sensing with Kil2 activity and intracellular killing. *Cellular Microbiology*, 22(1), e13129.
- Bogosian, G., Sammons, L. E., Morris, P. J., O'Neil, J. P., Heitkamp, M. A., & Weber, D. B. (1996). Death of the *Escherichia coli* K-12 strain W3110 in soil and water. *Applied and Environmental Microbiology*, 62(11), 4114–4120.
- Bohannan, B. J. M., & Lenski, R. E. (2000). Linking genetic change to community evolution: insights from studies of bacteria and bacteriophage. *Ecology Letters*, 3(4), 362–377.
- Bohannon, J. (2010, December 14). Building a parallel universe. *WIRED*. https://www.wired.com/story/building-a-parallel-universe/
- Boithias, L., Ribolzi, O., Lacombe, G., Thammahacksa, C., Silvera, N., Latsachack, K., Soulileuth, B., Viguier, M., Auda, Y., Robert, E., Evrard, O., Huon, S., Pommier, T., Zouiten, C., Sengtaheuanghoung, O., & Rochelle-Newall, E. (2021). Quantifying the effect of overland flow on *Escherichia coli* pulses during floods: Use of a tracer-based approach in an erosion-prone tropical catchment. *Journal of Hydrology, 594*, 125935.
- Bonneaud, C., Giraudeau, M., Tardy, L., Staley, M., Hill, G. E., & McGraw, K. J. (2018). Rapid antagonistic coevolution in an emerging pathogen and its vertebrate host. *Current Biology: CB*, 28(18), 2978–2983.e5.
- Borcher Møller, S. L., Rasmussen, M. H., Li, J., Svenningsen, E. B., Wørmer, G. J., Tørring, T., & Poulsen, T. B. (2024). The biological activities of polyether ionophore antibiotic routiennocin is independent of absolute stereochemistry. *ChemBioChem*, 25(7), e202400013.
- Bradford, S. A., Morales, V. L., Zhang, W., Harvey, R. W., Packman, A. I., Mohanram, A., & Welty, C. (2013). Transport and fate of microbial pathogens in agricultural settings. *Critical Reviews in Environmental Science* and Technology, 43(8), 775–893.
- Braun, V. (2003). Iron uptake by Escherichia Coli. Frontiers in Bioscience, 8(6), s1409-s1421.
- Breckels, M. N., Roberts, E. C., Archer, S. D., Malin, G., & Steinke, M. (2011). The role of dissolved infochemicals in mediating predator-prey interactions in the heterotrophic dinoflagellate *Oxyrrhis marina*. *Journal of Plankton Research*, *33*(4), 629–639.
- Buchanan, S. K., Lukacik, P., Grizot, S., Ghirlando, R., Ali, M. M. U., Barnard, T. J., Jakes, K. S., Kienker, P. K., & Esser, L. (2007). Structure of colicin I receptor bound to the R-domain of colicin Ia: Implications for protein import. *The EMBO Journal*, 26(10), 2594–2604.
- Byappanahalli, M., Fowler, M., Shively, D., & Whitman, R. (2003). Ubiquity and persistence of *Escherichia coli* in a midwestern coastal stream. *Applied and Environmental Microbiology*, 69(8), 4549–4555.
- Byappanahalli, M., & Fujioka, R. (2004). Indigenous soil bacteria and low moisture may limit but allow faecal bacteria to multiply and become a minor population in tropical soils. *Water Science and Technology*, *50*(1), 27–32.
- Byrd, J. J., & Colwell, R. R. (1990). Maintenance of plasmids pBR322 and pUC8 in nonculturable *Escherichia coli* in the marine environment. *Applied and Environmental Microbiology*, *56*(7), 2104–2107.
- Calbet, A., & Landry, M. R. (2004). Phytoplankton growth, microzooplankton grazing, and carbon cycling in marine systems. *Limnology and Oceanography*, 49(1), 51–57.

- Carlson, M. C. G., Ribalet, F., Maidanik, I., Durham, B. P., Hulata, Y., Ferrón, S., Weissenbach, J., Shamir, N., Goldin, S., Baran, N., Cael, B. B., Karl, D. M., White, A. E., Armbrust, E. V., & Lindell, D. (2022). Viruses affect picocyanobacterial abundance and biogeography in the North Pacific Ocean. *Nature Microbiology*, 7(4), 570–580.
- Carrillo, M., Estrada, E., & Hazen, T. C. (1985). Survival and enumeration of the fecal indicators *Bifidobacterium adolescentis* and *Escherichia coli* in a tropical rain forest watershed. *Applied and Environmental Microbiology*, 50(2), 468–476.
- Cayley, S., Lewis, B. A., & Record, M. T., Jr. (1992). Origins of the osmoprotective properties of betaine and proline in *Escherichia coli* K-12. *Journal of Bacteriology*, 174(5), 1586–1595.
- Celis, T. F. (1977). Independent regulation of transport and biosynthesis of arginine in *Escherichia coli* K-12. *Journal of Bacteriology*, *130*(3), 1244–1252.
- Cesari, S., Xi, Y., Declerck, N., Chalvon, V., Mammri, L., Pugnière, M., Henriquet, C., de Guillen, K., Chochois, V., Padilla, A., & Kroi, T. (2022). New recognition specificity in a plant immune receptor by molecular engineering of its integrated domain. *Nature Communications*, *13*, 1524.
- Chao, L., Levin, B. R., & Stewart, F. M. (1977). A complex community in a simple habitat: An experimental study with bacteria and phage. *Ecology*, *58*(2), 369–378.
- Chao, W. L., & Feng, R. L. (1990). Survival of genetically engineered *Escherichia coli* in natural soil and river water. *Journal of Applied Bacteriology*, 68(4), 319–325.
- Cheng, T. L., Reichard, J. D., Coleman, J. T. H., Weller, T. J., Thogmartin, W. E., Reichert, B. E., Bennett, A. B., Broders, H. G., Campbell, J., Etchison, K., Feller, D. J., Geboy, R., Hemberger, T., Herzog, C., Hicks, A. C., Houghton, S., Humber, J., Kath, J. A., King, R. A., ... Frick, W. F. (2021). The scope and severity of white-nose syndrome on hibernating bats in North America. *Conservation Biology*, 35(5), 1586–1597.
- Chen, G., Zhu, H., & Zhang, Y. (2003). Soil microbial activities and carbon and nitrogen fixation. *Research in Microbiology*, 154(6), 393–398.
- Choe, D., Lee, J. H., Yoo, M., Hwang, S., Sung, B. H., Cho, S., Palsson, B., Kim, S. C., & Cho, B.-K. (2019). Adaptive laboratory evolution of a genome-reduced *Escherichia coli*. *Nature Communications*, 10(1), 935.
- Choi, S. Y., Yoon, K.-H., Lee, J. I., & Mitchell, R. J. (2015). Violacein: Properties and production of a versatile bacterial pigment. *BioMed Research International*, 2015, 465056.
- Cho, K. H., Pachepsky, Y. A., Oliver, D. M., Muirhead, R. W., Park, Y., Quilliam, R. S., & Shelton, D. R. (2016). Modeling fate and transport of fecally-derived microorganisms at the watershed scale: State of the science and future opportunities. *Water Research*, 100, 38–56.
- Clark, J. S., Fastie, C., Hurtt, G., Jackson, S. T., Johnson, C., King, G. A., Lewis, M., Lynch, J., Pacala, S., Prentice, C., Schupp, E. W., Webb, T., & Wyckoff, P. (1998). Reid's paradox of rapid plant migration. *BioScience*, 48(1), 13–24.
- Murphy, S. T., & Evans, H. C. (2009). Biological control of invasive species. In M. N. Clout & P. A. Williams (Eds.), *Invasive species management: A handbook of principles and techniques* (pp. 77–92). Oxford University Press.

- Cobián Güemes, A. G., Youle, M., Cantú, V. A., Felts, B., Nulton, J., & Rohwer, F. (2016). Viruses as winners in the game of life. *Annual Review of Virology*, *3*(1), 197–214.
- Cooper, V. S., & Lenski, R. E. (2000). The population genetics of ecological specialization in evolving *Escherichia coli* populations. *Nature*, 407(6805), 736–739.
- Cotter, P. D., Ross, R. P., & Hill, C. (2013). Bacteriocins A viable alternative to antibiotics? *Nature Reviews Microbiology*, 11(2), 95–105.
- Crespo-Yanez, X., Oddy, J., Lamrabet, O., Jauslin, T., Marchetti, A., & Cosson, P. (2023). Sequential action of antibacterial effectors in *Dictyostelium discoideum* phagosomes. *Molecular Microbiology*, 119(1), 74–85.
- Dallas, T., Melbourne, B. A., Legault, G., & Hastings, A. (2021). Initial abundance and stochasticity influence competitive outcome in communities. *Journal of Animal Ecology*, 90(7), 1691–1700.
- Dal Peraro, M., & van der Goot, F. G. (2016). Pore-forming toxins: Ancient, but never really out of fashion. *Nature Reviews Microbiology*, 14(2), 77–92.
- Danevčič, T., Borić Vezjak, M., Zorec, M., & Stopar, D. (2016). Prodigiosin A multifaceted *Escherichia coli* antimicrobial agent. *PLoS ONE*, 11(9), e0162412.
- De Castro, F., & Bolker, B. (2005). Mechanisms of disease-induced extinction. Ecology Letters, 8(1), 117-126.
- Demoling, F., Figueroa, D., & Bååth, E. (2007). Comparison of factors limiting bacterial growth in different soils. *Soil Biology and Biochemistry*, *39*(10), 2485–2495.
- Dennehy, J. J., Friedenberg, N. A., Holt, R. D., & Turner, P. E. (2006). Viral ecology and the maintenance of novel host use. *The American Naturalist*, 167(3), 429–439.
- Desjardins, M., Houde, M., & Gagnon, E. (2005). Phagocytosis: The convoluted way from nutrition to adaptive immunity. *Immunological Reviews*, 207(1), 158–165.
- Desmarais, T. R., Solo-Gabriele, H. M., & Palmer, C. J. (2002). Influence of soil on fecal indicator organisms in a tidally influenced subtropical environment. *Applied and Environmental Microbiology*, 68(3), 1165–1172.
- Dhondt, A. A., Tessaglia, D. L., & Slothower, R. L. (1998). Epidemic mycoplasmal conjunctivitis in house finches from eastern North America. *Journal of Wildlife Diseases*, *34*(2), 265–280.
- Di Giallonardo, F., & Holmes, E. C. (2015). Viral biocontrol: Grand experiments in disease emergence and evolution. *Trends in Microbiology*, 23(2), 83–90.
- D'Odorico, P., Bhattachan, A., Davis, K. F., Ravi, S., & Runyan, C. W. (2013). Global desertification: Drivers and feedbacks. *Advances in Water Resources*, *51*, 326–344.
- Doornbos, R. F., van Loon, L. C., & Bakker, P. A. H. M. (2012). Impact of root exudates and plant defense signaling on bacterial communities in the rhizosphere. *Agronomy for Sustainable Development*, 32(1), 227–243.
- Dunn, J. D., Bosmani, C., Barisch, C., Raykov, L., Lefrançois, L. H., Cardenal-Muñoz, E., López-Jiménez, A. T., & Soldati, T. (2017). Eat prey, live: *Dictyostelium discoideum* as a model for cell-autonomous defenses. *Frontiers in Immunology*, 8, 1906.

- Epstein, B., Jones, M., Hamede, R., Hendricks, S., McCallum, H., Murchison, E. P., Schönfeld, B., Wiench, C., Hohenlohe, P., & Storfer, A. (2016). Rapid evolutionary response to a transmissible cancer in Tasmanian devils. *Nature Communications*, 7(1), 12684.
- Erb, K.-H., Kastner, T., Plutzar, C., Bais, A. L. S., Carvalhais, N., Fetzel, T., Gingrich, S., Haberl, H., Lauk, C., Niedertscheider, M., Pongratz, J., Thurner, M., & Luyssaert, S. (2018). Unexpectedly large impact of forest management and grazing on global vegetation biomass. *Nature*, 553(7686), 73–76.
- Eswaran, H., Lal, R., & Reich, P. F. (2001). Land degradation: An overview. In E. M. Bridges, I. D. Hannam, L. R. Oldeman, F. W. T. P. de Vries, S. J. Scherr, & S. Sompatpanit (Eds.), *Response to land degradation* (pp. 20–35). CRC Press.
- Farrant, G. K., Doré, H., Cornejo-Castillo, F. M., Partensky, F., Ratin, M., Ostrowski, M., Pitt, F. D., Wincker, P., Scanlan, D. J., Iudicone, D., Acinas, S. G., & Garczarek, L. (2016). Delineating ecologically significant taxonomic units from global patterns of marine picocyanobacteria. *Proceedings of the National Academy of Sciences of the United States of America*, 113(24), E3365–E3374.
- Faruque, S. M., Albert, M. J., & Mekalanos, J. J. (1998). Epidemiology, genetics, and ecology of toxigenic Vibrio cholerae. Microbiology and Molecular Biology Reviews, 62(4), 1301–1314.
- Fernandez, M. O., Thomas, R. J., Garton, N. J., Hudson, A., Haddrell, A., & Reid, J. P. (2019). Assessing the airborne survival of bacteria in populations of aerosol droplets with a novel technology. *Journal of the Royal Society Interface*, 16(150), 20180779.
- Field, C. B., Behrenfeld, M. J., Randerson, J. T., & Falkowski, P. (1998). Primary production of the biosphere: integrating terrestrial and oceanic components. *Science*, *281*(5374), 237–240.
- Finkel, S. E., & Kolter, R. (1999). Evolution of microbial diversity during prolonged starvation. *Proceedings of the National Academy of Sciences of the United States of America*, *96*(7), 4023–4027.
- Flint, K. P. (1987). The long-term survival of *Escherichia coli* in river water. *Journal of Applied Bacteriology*, 63(3), 261–270.
- Flombaum, P., Gallegos, J. L., Gordillo, R. A., Rincón, J., Zabala, L. L., Jiao, N., Karl, D. M., Li, W. K. W., Lomas, M. W., Veneziano, D., Vera, C. S., Vrugt, J. A., & Martiny, A. C. (2013). Present and future global distributions of the marine cyanobacteria *Prochlorococcus* and *Synechococcus*. *Proceedings of the National Academy of Sciences of the United States of America*, 110(24), 9824–9829.
- Fong, S. S., Marciniak, J. Y., & Palsson, B. Ø. (2003). Description and interpretation of adaptive evolution of *Escherichia coli* K-12 MG1655 by using a genome-scale *in silico* metabolic model. *Journal of Bacteriology*, 185(21), 6400–6408.
- Forsum, O., Svennerstam, H., Ganeteg, U., & Näsholm, T. (2008). Capacities and constraints of amino acid utilization in *Arabidopsis. New Phytologist*, 179(4), 1058–1069.
- Fortini, L. B., Kaiser, L. R., Keith, L. M., Price, J., Hughes, R. F., Jacobi, J. D., & Friday, J. B. (2019). The evolving threat of Rapid 'Ohi'a Death (ROD) to Hawai'i's native ecosystems and rare plant species. *Forest Ecology and Management*, 448, 376–385.
- Fredens, J., Wang, K., de la Torre, D., Funke, L. F. H., Robertson, W. E., Christova, Y., Chia, T., Schmied, W. H., Dunkelmann, D. L., Beránek, V., Uttamapinant, C., Llamazares, A. G., Elliott, T. S., & Chin, J. W. (2019). Total synthesis of *Escherichia coli* with a recoded genome. *Nature*, 569(7757), 514–518.

- Frias-Lopez, J., Thompson, A., Waldbauer, J., & Chisholm, S. W. (2009). Use of stable isotope-labelled cells to identify active grazers of picocyanobacteria in ocean surface waters. *Environmental Microbiology*, 11(2), 512–525.
- Friedlingstein, P., O'Sullivan, M., Jones, M. W., Andrew, R. M., Bakker, D. C. E., Hauck, J., Landschützer, P., Le Quéré, C., Luijkx, I. T., Peters, G. P., Peters, W., Pongratz, J., Schwingshackl, C., Sitch, S., Canadell, J. G., Ciais, P., Jackson, R. B., Alin, S. R., Anthoni, P., ... Zheng, B. (2023). Global carbon budget 2023. *Earth System Science Data*, 15(12), 5301–5369.
- Friedman, M., & Levin, C. E. (2012). Nutritional and medicinal aspects of D-amino acids. *Amino Acids*, 42(5), 1553–1582.
- Füchslin, H. P., Schneider, C., & Egli, T. (2012). In glucose-limited continuous culture the minimum substrate concentration for growth, Smin, is crucial in the competition between the enterobacterium *Escherichia coli* and *Chelatobacter heintzii*, an environmentally abundant bacterium. *ISME Journal*, 6(4), 777–789.
- Fukushima, H., Hoshina, K., & Gomyoda, M. (1999). Long-term survival of shiga toxin-producing *Escherichia coli* O26, O111, and O157 in bovine feces. *Applied and Environmental Microbiology*, 65(11), 5177–5181.
- Gabrielsen, C., Brede, D. A., Hernández, P. E., Nes, I. F., & Diep, D. B. (2012). The maltose ABC transporter in *Lactococcus lactis* facilitates high-level sensitivity to the circular bacteriocin garvicin ML. *Antimicrobial Agents and Chemotherapy*, 56(6), 2908–2915.
- Galán, J. E., & Waksman, G. (2018). Protein-injection machines in bacteria. Cell, 172(6), 1306–1318.
- Gálvez, A., Maqueda, M., Martínez-Bueno, M., & Valdivia, E. (1991). Permeation of bacterial cells, permeation of cytoplasmic and artificial membrane vesicles, and channel formation on lipid bilayers by peptide antibiotic AS-48. *Journal of Bacteriology*, 173(2), 886–892.
- García-Bayona, L., & Comstock, L. E. (2018). Bacterial antagonism in host-associated microbial communities. *Science*, *361*(6408), eaat2456.
- Gauthier, M. J., & Le Rudulier, D. (1990). Survival in seawater of *Escherichia coli* cells grown in marine sediments containing glycine betaine. *Applied and Environmental Microbiology*, *56*(9), 2915–2918.
- Geisen, S., Mitchell, E. A. D., Adl, S., Bonkowski, M., Dunthorn, M., Ekelund, F., Fernández, L. D., Jousset, A., Krashevska, V., Singer, D., Spiegel, F. W., Walochnik, J., & Lara, E. (2018). Soil protists: A fertile frontier in soil biology research. *FEMS Microbiology Reviews*, 42(3), 293–323.
- Geisen, S., Mitchell, E. A. D., Wilkinson, D. M., Adl, S., Bonkowski, M., Brown, M. W., Fiore-Donno, A. M., Heger, T. J., Jassey, V. E. J., Krashevska, V., Lahr, D. J. G., Marcisz, K., Mulot, M., Payne, R., Singer, D., Anderson, O. R., Charman, D. J., Ekelund, F., Griffiths, B. S., ... Lara, E. (2017). Soil protistology rebooted: 30 fundamental questions to start with. *Soil Biology & Biochemistry*, *111*, 94–103.
- Ghequire, M. G. K., & De Mot, R. (2015). The tailocin tale: Peeling off phage tails. *Trends in Microbiology*, 23(10), 587–590.
- Gippet, J. M., Liebhold, A. M., Fenn-Moltu, G., & Bertelsmeier, C. (2019). Human-mediated dispersal in insects. *Current Opinion in Insect Science*, 35, 96–102.

- Gong, X., Martin-Visscher, L. A., Nahirney, D., Vederas, J. C., & Duszyk, M. (2009). The circular bacteriocin, carnocyclin A, forms anion-selective channels in lipid bilayers. *Biochimica et Biophysica Acta -Biomembranes*, 1788(9), 1797–1803.
- Good, B. H., McDonald, M. J., Barrick, J. E., Lenski, R. E., & Desai, M. M. (2017). The dynamics of molecular evolution over 60,000 generations. *Nature*, *551*(7678), 45–50.
- Gopaldass, N., Patel, D., Kratzke, R., Dieckmann, R., Hausherr, S., Hagedorn, M., Monroy, R., Krüger, J., Neuhaus, E. M., Hoffmann, E., Hille, K., Kuznetsov, S. A., & Soldati, T. (2012). Dynamin A, Myosin IB and Abp1 couple phagosome maturation to F-actin binding. *Traffic*, *13*(1), 120–130.
- Gorden, J., & Small, P. L. (1993). Acid resistance in enteric bacteria. Infection and Immunity, 61(1), 364-367.
- Gotthardt, D., Warnatz, H. J., Henschel, O., Brückert, F., Schleicher, M., & Soldati, T. (2002). High-resolution dissection of phagosome maturation reveals distinct membrane trafficking phases. *Molecular Biology of the Cell*, *13*(10), 3508–3520.
- Granato, E. T., Meiller-Legrand, T. A., & Foster, K. R. (2019). The evolution and ecology of bacterial warfare. *Current Biology*, 29(11), R521–R537.
- Grasso, C. R., Pokrzywinski, K. L., Waechter, C., Rycroft, T., Zhang, Y., Aligata, A., Kramer, M., & Lamsal, A. (2022). A review of cyanophage-host relationships: Highlighting cyanophages as a potential cyanobacteria control strategy. *Toxins*, 14(6), 385.
- Gratz, N. G., Steffen, R., & Cocksedge, W. (2000). Why aircraft disinsection? *Bulletin of the World Health Organization*, 78(8), 995–1004.
- Gravato-Nobre, M. J., Vaz, F., Filipe, S., Chalmers, R., & Hodgkin, J. (2016). The invertebrate lysozyme effector ILYS-3 is systemically activated in response to danger signals and confers antimicrobial protection in *C. elegans*. *PLoS Pathogens*, *12*(8), e1005826.
- Gravesen, A., Ramnath, M., Rechinger, K. B., Andersen, N., Jänsch, L., Héchard, Y., Hastings, J. W., & Knøchel, S. (2002). High-level resistance to class IIa bacteriocins is associated with one general mechanism in *Listeria monocytogenes*. *Microbiology*, 148(8), 2361–2369.
- Griffin, D. W., Gonzalez-Martin, C., Hoose, C., & Smith, D. J. (2017). Global-scale atmospheric dispersion of microorganisms. In *Microbiology of Aerosols* (pp. 155–194). Wiley.
- Grossmann, L., Jensen, M., Heider, D., Jost, S., Glücksman, E., Hartikainen, H., Mahamdallie, S. S., Gardner, M., Hoffmann, D., Bass, D., & Boenigk, J. (2016). Protistan community analysis: Key findings of a large-scale molecular sampling. *ISME Journal*, 10(9), 2269–2279.
- Guerrier, S., Plattner, H., Richardson, E., Dacks, J. B., & Turkewitz, A. P. (2017). An evolutionary balance: conservation vs innovation in ciliate membrane trafficking. *Traffic*, *18*(1), 18–28.
- Gunina, A., & Kuzyakov, Y. (2015). Sugars in soil and sweets for microorganisms: Review of origin, content, composition and fate. *Soil Biology and Biochemistry*, *90*, 87–100.
- Guo, Y., Winkler, J., & Kao, K. C. (2017). Insights on osmotic tolerance mechanisms in *Escherichia coli* gained from an rpoC mutation. *Bioengineering*, 4(3), 61.

- Hao, G., Stover, E., & Gupta, G. (2016). Overexpression of a modified plant thionin enhances disease resistance to citrus canker and huanglongbing (HLB). *Frontiers in Plant Science*, *7*, 1078.
- Harden, M. M., He, A., Creamer, K., Clark, M. W., Hamdallah, I., Martinez, K. A. II, Kresslein, R. L., Bush, S. P., & Slonczewski, J. L. (2015). Acid-adapted strains of *Escherichia coli* K-12 obtained by experimental evolution. *Applied and Environmental Microbiology*, 81(6), 1932–1941.
- Harms, J. M., Wilson, D. N., Schluenzen, F., Connell, S. R., Stachelhaus, T., Zaborowska, Z., Spahn, C. M. T., & Fucini, P. (2008). Translational regulation via L11: Molecular switches on the ribosome turned on and off by thiostrepton and micrococcin. *Molecular Cell*, 30(1), 26–38.
- Harris, D. R., Pollock, S. V., Wood, E. A., Goiffon, R. J., Klingele, A. J., Cabot, E. L., Schackwitz, W., Martin, J., Eggington, J., Durfee, T. J., Middle, C. M., Norton, J. E., Popelars, M. C., Li, H., Klugman, S. A., Hamilton, L. L., Bane, L. B., Pennacchio, L. A., Albert, T. J., ... Battista, J. R. (2009). Directed evolution of ionizing radiation resistance in *Escherichia coli*. *Journal of Bacteriology*, *191*(16), 5240–5252.
- Harris, R. F. (1981). Effect of water potential on microbial growth and activity. In D. M. Kral, M. H. Frere, & D. R. Nielsen (Eds.), *Water potential relations in soil microbiology* (pp. 23–95). Soil Science Society of America.
- Harris, T. D., Reinl, K. L., Azarderakhsh, M., Berger, S. A., Berman, M. C., Bizic, M., Bhattacharya, R., Burnet, S. H., Cianci-Gaskill, J. A., de Senerpont Domis, L. N., Elfferich, I., Ger, K. A., Grossart, H. F., Ibelings, B. W., Ionescu, D., Mazaheri Kouhanestani, Z., Mauch, J., McElarney, Y. R., Nava, V., ... Zhan, Q. (2024). What makes a cyanobacterial bloom disappear? A review of the abiotic and biotic cyanobacterial bloom loss factors. *Harmful Algae*, *133*, 102599.
- Harwood, T. D., Tomlinson, I., Potter, C. A., & Knight, J. D. (2011). Dutch elm disease revisited: Past, present and future management in Great Britain. *Plant Pathology*, *60*(3), 545–555.
- Hassani, M. A., Durán, P., & Hacquard, S. (2018). Microbial interactions within the plant holobiont. *Microbiome*, *6*(1), 58.
- Herring, C. D., Raghunathan, A., Honisch, C., Patel, T., Applebee, M. K., Joyce, A. R., Albert, T. J., Blattner, F. R., van den Boom, D., Cantor, C. R., & Palsson, B. Ø. (2006). Comparative genome sequencing of *Escherichia coli* allows observation of bacterial evolution on a laboratory timescale. *Nature Genetics*, 38(12), 1406–1412.
- Hitomi, K., Ishii, Y., & Ying, B.-W. (2024). Experimental evolution for the recovery of growth loss due to genome reduction. *eLife*, *13*, e93520.
- Hoffland, E., Kuyper, T. W., Comans, R. N. J., & Creamer, R. E. (2020). Eco-functionality of organic matter in soils. *Plant and Soil*, 455(1-2), 1–22.
- Hsu, S.-T. D., Breukink, E., Bierbaum, G., Sahl, H.-G., de Kruijff, B., Kaptein, R., van Nuland, N. A. J., & Bonvin, A. M. J. J. (2003). NMR study of mersacidin and lipid II interaction in dodecylphosphocholine micelles. *Journal of Biological Chemistry*, 278(15), 13110–13117.
- Hsu, S.-T. D., Breukink, E., Tischenko, E., Lutters, M. A. G., de Kruijff, B., Kaptein, R., Bonvin, A. M. J. J., & van Nuland, N. A. J. (2004). The nisin–lipid II complex reveals a pyrophosphate cage that provides a blueprint for novel antibiotics. *Nature Structural & Molecular Biology*, 11(10), 963–967.
- Hugh-Jones, M., & Blackburn, J. (2009). The ecology of *Bacillus anthracis*. *Molecular Aspects of Medicine*, 30(6), 356–367.

- Husson, O. (2013). Redox potential (Eh) and pH as drivers of soil/plant/microorganism systems: A transdisciplinary overview pointing to integrative opportunities for agronomy. *Plant and Soil*, *362*(1-2), 389–417.
- Hutchison, C. A., III, Chuang, R.-Y., Noskov, V. N., Assad-Garcia, N., Deerinck, T. J., Ellisman, M. H., Gill, J., Kannan, K., Karas, B. J., Ma, L., Pelletier, J. F., Qi, Z.-Q., Richter, R. A., Strychalski, E. A., Sun, L., Suzuki, Y., Tsvetanova, B., Wise, K. S., Smith, H. O., Glass, J. I., Merryman, C., Gibson, D. G., & Venter, J. C. (2016). Design and synthesis of a minimal bacterial genome. *Science*, *351*(6280), aad6253.
- Hu, Y., Zheng, Q., Zhang, S., Noll, L., & Wanek, W. (2018). Significant release and microbial utilization of amino sugars and D-amino acid enantiomers from microbial cell wall decomposition in soils. *Soil Biology and Biochemistry*, 123, 115–125.
- Ingram, L. O. (1990). Ethanol tolerance in bacteria. Critical Reviews in Biotechnology, 9(4), 305-319.
- Invasive Species Council. (2023). *Threats to nature field guide: Phytophthora dieback.* https://invasives.org.au/wp-content/uploads/2023/01/Threats-to-Nature-Field-Guide-Phytophthora-Rebranded. pdf
- IPCC. (2021). Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press.
- Ishii, S., Ksoll, W. B., Hicks, R. E., & Sadowsky, M. J. (2006). Presence and growth of naturalized *Escherichia coli* in temperate soils from Lake Superior watersheds. *Applied and Environmental Microbiology*, 72(1), 612–621.
- Ishii, S., & Sadowsky, M. J. (2008). *Escherichia coli* in the environment: Implications for water quality and human health. *Microbes and Environments*, 23(2), 101–108.
- Iuculano, F., Mazuecos, I. P., Reche, I., & Agustí, S. (2017). Prochlorococcus as a possible source for transparent exopolymer particles (TEP). Frontiers in Microbiology, 8, 709.
- Izutsu, M., & Lenski, R. E. (2022). Experimental test of the contributions of initial variation and new mutations to adaptive evolution in a novel environment. *Frontiers in Ecology and Evolution*, *10*, 958406.
- Jackson, R. B., Lajtha, K., Crow, S. E., Hugelius, G., Kramer, M. G., & Piñeiro, G. (2017). The ecology of soil carbon: Pools, vulnerabilities, and biotic and abiotic controls. *Annual Review of Ecology, Evolution, and Systematics*, 48(1), 419–445.
- Jang, J., Hur, H.-G., Sadowsky, M. J., Byappanahalli, M. N., Yan, T., & Ishii, S. (2017). Environmental Escherichia coli: Ecology and public health implications. Journal of Applied Microbiology, 123(3), 570–581.
- Jannasch, H. W. (1968). Competitive elimination of *Enterobacteriaceae* from seawater. *Applied Microbiology*, *16*(10), 1616–1618.
- Jauslin, T., Lamrabet, O., Crespo-Yañez, X., Marchetti, A., Ayadi, I., Ifrid, E., Guilhen, C., Leippe, M., & Cosson, P. (2021). How phagocytic cells kill different bacteria: A quantitative analysis using *Dictyostelium discoideum*. *mBio*, 12(1), e03169-20.
- Jiang, X., Morgan, J., & Doyle, M. P. (2002). Fate of *Escherichia coli* O157:H7 in Manure-Amended Soil. *Applied and Environmental Microbiology*, 68(5), 2605–2609.

- Johnson, Z. I., Zinser, E. R., Coe, A., McNulty, N. P., Woodward, E. M. S., & Chisholm, S. W. (2006). Niche partitioning among *Prochlorococcus* ecotypes along ocean-scale environmental gradients. *Science*, 311(5768), 1737–1740.
- June, C. H., & Sadelain, M. (2018). Chimeric antigen receptor therapy. *New England Journal of Medicine*, 379(1), 64–73.
- Jürgens, K., & Matz, C. (2002). Predation as a shaping force for the phenotypic and genotypic composition of planktonic bacteria. *Antonie van Leeuwenhoek*, *81*(1/4), 413–434.
- Kaluza, P., Kölzsch, A., Gastner, M. T., & Blasius, B. (2010). The complex network of global cargo ship movements. *Journal of the Royal Society Interface*, 7(48), 1093–1103.
- Kamilova, F., Kravchenko, L. V., Shaposhnikov, A. I., Azarova, T., Makarova, N., & Lugtenberg, B. (2006). Organic acids, sugars, and L-tryptophane in exudates of vegetables growing on stonewool and their effects on activities of rhizosphere bacteria. *Molecular Plant-Microbe Interactions*, 19(3), 250–256.
- Kamyar, M., Rawnduzi, P., Studenik, C. R., Kouri, K., & Lemmens-Gruber, R. (2004). Investigation of the electrophysiological properties of enniatins. *Archives of Biochemistry and Biophysics*, 429(2), 215–223.
- Karasov, W. H., & Douglas, A. E. (2013). Comparative digestive physiology. *Comprehensive Physiology*, *3*(2), 741–783.
- Kashtan, N., Noor, E., & Alon, U. (2007). Varying environments can speed up evolution. *Proceedings of the National Academy of Sciences of the United States of America*, 104(34), 13711–13716.
- Kashtan, N., Roggensack, S. E., Rodrigue, S., Thompson, J. W., Biller, S. J., Coe, A., Ding, H., Marttinen, P., Malmstrom, R. R., Stocker, R., Follows, M. J., Stepanauskas, R., & Chisholm, S. W. (2014). Single-cell genomics reveals hundreds of coexisting subpopulations in wild *Prochlorococcus. Science*, 344(6182), 416–420.
- Kawai, Y., Ishii, Y., Arakawa, K., Uemura, K., Saitoh, B., Nishimura, J., Kitazawa, H., Yamazaki, Y., Tateno, Y., Itoh, T., & Saito, T. (2004). Structural and functional differences in two cyclic bacteriocins with the same sequences produced by *lactobacilli*. *Applied and Environmental Microbiology*, 70(5), 2906–2911.
- Kawecki, T. J. (2008). Adaptation to marginal habitats. *Annual Review of Ecology, Evolution, and Systematics*, 39(1), 321–342.
- Keane, R. & Crawly, M. J. (2002). Exotic plant invasions and the enemy release hypothesis. *Trends in Ecology and Evolution*, *17*(4), 164–170.
- Keegstra, J. M., Carrara, F., & Stocker, R. (2022). The ecological roles of bacterial chemotaxis. *Nature Reviews Microbiology*, 20(8), 491–504.
- Kerr, P. J. (2012). Myxomatosis in Australia and Europe: a model for emerging infectious diseases. *Antiviral Research*, 93(3), 387–415.
- Khan, H., Flint, S. H., & Yu, P.-L. (2013). Determination of the mode of action of enterolysin A, produced by *Enterococcus faecalis* B9510. *Journal of Applied Microbiology*, *115*(2), 484–494.

Kirchman, D. L. (2016). Growth rates of microbes in the oceans. Annual Review of Marine Science, 8(1), 285-309.

Kirchman, D. L. (2018). Processes in microbial ecology (2nd ed.). Oxford University Press.

- Kitaoka, S., Morielli, A. D., & Zhao, F.-Q. (2013). FGT-1 is a mammalian GLUT2-like facilitative glucose transporter in *Caenorhabditis elegans* whose malfunction induces fat accumulation in intestinal cells. *PLoS* ONE, 8(6), e68475.
- Kohanski, M. A., Dwyer, D. J., & Collins, J. J. (2010). How antibiotics kill bacteria: from targets to networks. *Nature Reviews Microbiology*, 8(6), 423–435.
- Komárek, J., Johansen, J. R., Šmarda, J., & Strunecký, O. (2020). Phylogeny and taxonomy of *Synechococcus*-like cyanobacteria. *Fottea*, 20(2), 171–191.
- Koskiniemi, S., Lamoureux, J. G., Nikolakakis, K. C., t'Kint de Roodenbeke, C., Kaplan, M. D., Low, D. A., & Hayes, C. S. (2013). Rhs proteins from diverse bacteria mediate intercellular competition. *Proceedings of the National Academy of Sciences of the United States of America*, 110(17), 7032–7037.
- Kot, M., Lewis, M. A., & van den Driessche, P. (1996). Dispersal data and the spread of invading organisms. *Ecology*, 77(7), 2027–2042.
- Kramer, J., Özkaya, Ö., & Kümmerli, R. (2020). Bacterial siderophores in community and host interactions. *Nature Reviews Microbiology*, 18(3), 152–163.
- Krupovic, M., Dolja, V. V., & Koonin, E. V. (2019). Origin of viruses: primordial replicators recruiting capsids from hosts. *Nature Reviews Microbiology*, 17(7), 449–458.
- Kudva, I. T., Blanch, K., & Hovde, C. J. (1998). Analysis of *Escherichia coli* O157:H7 survival in ovine or bovine manure and manure slurry. *Applied and Environmental Microbiology*, 64(9), 3166–3174.
- Kuhn, J., & Somerville, R. L. (1971). Mutant strains of Escherichia coli K12 that use D-amino acids. Proceedings of the National Academy of Sciences of the United States of America, 68(10), 2484–2487.
- Kujawinski, E. B. (2011). The impact of microbial metabolism on marine dissolved organic matter. *Annual Review* of Marine Science, 3(1), 567–599.
- Kuznedelov, K., Semenova, E., Knappe, T. A., Mukhamedyarov, D., Srivastava, A., Chatterjee, S., Ebright, R. H., Marahiel, M. A., & Severinov, K. (2011). The antibacterial threaded-lasso peptide capistruin inhibits bacterial RNA polymerase. *Journal of Molecular Biology*, 412(5), 842–848.
- Labrie, S. J., Samson, J. E., & Moineau, S. (2010). Bacteriophage resistance mechanisms. *Nature Reviews Microbiology*, 8(5), 317–327.
- Lamrabet, O., Melotti, A., Burdet, F., Hanna, N., Perrin, J., Nitschke, J., Pagni, M., Hilbi, H., Soldati, T., & Cosson, P. (2020). Transcriptional responses of *Dictyostelium discoideum* exposed to different classes of bacteria. *Frontiers in Microbiology*, 11, 410.
- Lancaster, C. E., Ho, C. Y., Hipolito, V. E. B., Botelho, R. J., & Terebiznik, M. R. (2019). Phagocytosis: what's on the menu? *Biochemistry and Cell Biology*, 97(1), 21–29.
- Lander, A. J., Domínguez Mercado, L., Li, X., Taily, I. M., Findlay, B. L., Jin, Y., & Luk, L. Y. P. (2023). Roles of inter- and intramolecular tryptophan interactions in membrane-active proteins revealed by racemic protein crystallography. *Communications Chemistry*, 6, 154.
Lee, D.-H., & Palsson, B. Ø. (2010). Adaptive evolution of *Escherichia coli* K-12 MG1655 during growth on a nonnative carbon source, L-1,2-propanediol. *Applied and Environmental Microbiology*, 76(13), 4158–4168.

Leiba, J., Sabra, A., Bodinier, R., Marchetti, A., Lima, W. C., Melotti, A., Perrin, J., Burdet, F., Pagni, M., Soldati, T., Lelong, E., & Cosson, P. (2017). Vps13F links bacterial recognition and intracellular killing in *Dictyostelium*. *Cellular Microbiology*, *19*(7), e12722.

- Levin, B. R., Stewart, F. M., & Chao, L. (1977). Resource-limited growth, competition, and predation: A model and experimental studies with bacteria and bacteriophage. *The American Naturalist*, 111(977), 3–24.
- Lewis, M. A., Petrovskii, S. V., & Potts, J. R. (2016). The mathematics behind biological invasions. Springer.
- Li, B., Wever, W. J., Walsh, C. T., & Bowers, A. A. (2014). Dithiolopyrrolones: Biosynthesis, synthesis, and activity of a unique class of disulfide-containing antibiotics. *Natural Product Reports*, *31*(7), 905–923.
- Lima, W. C., Balestrino, D., Forestier, C., & Cosson, P. (2014). Two distinct sensing pathways allow recognition of *Klebsiella pneumoniae* by *Dictyostelium* amoebae. *Cellular Microbiology*, 16(3), 311–323.
- Li, X.-Y., Lachnit, T., Fraune, S., Bosch, T. C. G., Traulsen, A., & Sieber, M. (2017). Temperate phages as self-replicating weapons in bacterial competition. *Journal of the Royal Society, Interface*, 14(137), 20170563.
- López-Sandoval, D. C., Rodríguez-Ramos, T., Cermeño, P., & Marañón, E. (2013). Exudation of organic carbon by marine phytoplankton: Dependence on taxon and cell size. *Marine Ecology Progress Series*, 477, 53–60.
- Louca, S. (2022). The rates of global bacterial and archaeal dispersal. ISME Journal, 16(1), 159-167.
- Lounibos, L. P. (2002). Invasions by insect vectors of human disease. *Annual Review of Entomology*, 47(1), 233–266.
- Lowe, R., Barcellos, C., Brasil, P., Cruz, O. G., Honório, N. A., Kuper, H., & Carvalho, M. S. (2018). The Zika virus epidemic in Brazil: From discovery to future implications. *International Journal of Environmental Research* and Public Health, 15(1), 96.
- Lubina, J. A., & Levin, S. A. (1988). The spread of a reinvading species: Range expansion in the California sea otter. *The American Naturalist*, 131(4), 526–543.
- Luiz, B., Stacy, E. A., & Keith, L. M. (2021). Screening of *Metrosideros polymorpha* ('ōhi'a) varieties for resistance to *Ceratocystis lukuohia*. Forest Pathology, 51(1), e12656.
- Luo, C., Walk, S. T., Gordon, D. M., Feldgarden, M., Tiedje, J. M., & Konstantinidis, K. T. (2011). Genome sequencing of environmental *Escherichia coli* expands understanding of the ecology and speciation of the model bacterial species. *Proceedings of the National Academy of Sciences of the United States of America*, 108(17), 7200–7205.
- Lynch, M., Ackerman, M. S., Gout, J.-F., Long, H., Sung, W., Thomas, W. K., & Foster, P. L. (2016). Genetic drift, selection and the evolution of the mutation rate. *Nature Reviews Genetics*, *17*(11), 704–714.
- Mann, N. H. (2003). Phages of the marine cyanobacterial picophytoplankton. *FEMS Microbiology Reviews*, 27(1), 17–34.
- Mao, E. F., Lane, L., Lee, J., & Miller, J. H. (1997). Proliferation of mutators in a cell population. *Journal of Bacteriology*, 179(2), 417–422.

- Marchetti, A., Lelong, E., & Cosson, P. (2009). A measure of endosomal pH by flow cytometry in *Dictyostelium*. *BMC Research Notes*, 2(1), 7.
- Mariano, G., Trunk, K., Williams, D. J., Monlezun, L., Strahl, H., Pitt, S. J., & Coulthurst, S. J. (2019). A family of Type VI secretion system effector proteins that form ion-selective pores. *Nature Communications*, *10*(1), 5484.
- Marinissen, J. C. Y., & van den Bosch, F. (1992). Colonization of new habitats by earthworms. *Oecologia*, 91(3), 371–376.
- Martel, C. M. (2006). Prey location, recognition and ingestion by the phagotrophic marine dinoflagellate Oxyrrhis marina. Journal of Experimental Marine Biology and Ecology, 335(2), 210–220.
- Martiny, A. C., Lomas, M. W., Fu, W., Boyd, P. W., Chen, Y.-L. L., Cutter, G. A., Ellwood, M. J., Furuya, K., Hashihama, F., Kanda, J., Karl, D. M., Kodama, T., Li, Q. P., Ma, J., Moutin, T., Woodward, E. M. S., & Moore, J. K. (2019). Biogeochemical controls of surface ocean phosphate. *Science Advances*, 5(8), eaax0341.
- Mayol, E., Arrieta, J. M., Jiménez, M. A., Martínez-Asensio, A., Garcias-Bonet, N., Dachs, J., González-Gaya, B., Royer, S.-J., Benítez-Barrios, V. M., Fraile-Nuez, E., & Duarte, C. M. (2017). Long-range transport of airborne microbes over the global tropical and subtropical ocean. *Nature Communications*, 8(1), 201.
- McCallum, H. (2012). Disease and the dynamics of extinction. *Philosophical Transactions of the Royal Society B*, 367(1604), 2828–2839.
- McCallum, H., Harvell, D., & Dobson, A. (2003). Rates of spread of marine pathogens. *Ecology Letters*, 6(12), 1062–1067.
- McDougal, P. G., & Griffin, J. H. (2003). Enantiotracin. *Bioorganic and Medicinal Chemistry Letters*, 13(13), 2239–2240.
- McGhee, J. D., Sleumer, M. C., Bilenky, M., Wong, K., McKay, S. J., Goszczynski, B., Tian, H., Krich, N. D., Khattra, J., Holt, R. A., Baillie, D. L., Kohara, Y., Marra, M. A., Jones, S. J. M., Moerman, D. G., Robertson, A. G. (2007). The ELT-2 GATA-factor and the global regulation of transcription in the *C. elegans* intestine. *Developmental Biology*, 302(2), 627–645.
- Mesibov, R., & Adler, J. (1972). Chemotaxis toward amino acids in *Escherichia coli*. Journal of Bacteriology, 112(1), 315–326.
- Meyer, J. R., Dobias, D. T., Weitz, J. S., Barrick, J. E., Quick, R. T., & Lenski, R. E. (2012). Repeatability and contingency in the evolution of a key innovation in phage lambda. *Science*, *335*(6067), 428–432.
- Milo, R., & Phillips, R. (2015). Cell biology by the numbers. Garland Science.
- Miyata, S. T., Unterweger, D., Rudko, S. P., & Pukatzki, S. (2013). Dual expression profile of type VI secretion system immunity genes protects pandemic *Vibrio cholerae*. *PLoS Pathogens*, *9*(12), e1003752.
- Moger-Reischer, R. Z., Glass, J. I., Wise, K. S., Sun, L., Bittencourt, D. M. C., Lehmkuhl, B. K., Schoolmaster, D. R., Jr., Lynch, M., Lennon, J. T. (2023). Evolution of a minimal cell. *Nature*, 620(7972), 122–127.
- Montagnes, D. J. S., Barbosa, A. B., Boenigk, J., Davidson, D., Jürgens, K., Macek, M., Parry, J. D., Roberts, E. C., & Šimek, K. (2008). Selective feeding behaviour of key free-living protists: Avenues for continued study. *Aquatic Microbial Ecology*, 53(1), 83–98.

- Moore, J. K., Doney, S. C., Glover, D. M., & Fung, I. Y. (2001). Iron cycling and nutrient-limitation patterns in surface waters of the World Ocean. *Deep-Sea Research II*, 49(1-3), 463–507.
- Moore, L. R., Coe, A., Zinser, E. R., Saito, M. A., Sullivan, M. B., Lindell, D., Frois-Moniz, K., Waterbury, J., & Chisholm, S. W. (2007). Culturing the marine cyanobacterium *Prochlorococcus*. *Limnology and Oceanography: Methods*, 5(10), 353–362.
- Mora, C., Tittensor, D. P., Adl, S., Simpson, A. G. B., & Worm, B. (2011). How many species are there on Earth and in the ocean? *PLoS Biology*, *9*(8), e1001127.
- Moreira, R., & Taylor, S. D. (2022). The chiral target of daptomycin is the 2R,2'S stereoisomer of phosphatidylglycerol. *Angewandte Chemie*, *134*(4), e202114858.
- Morgan, A. D., MacLean, R. C., Hillesland, K. L., & Velicer, G. J. (2010). Comparative analysis of myxococcus predation on soil bacteria. Applied and Environmental Microbiology, 76(20), 6920–6927.
- Mukherjee, A., Cho, S., Scheftel, J., Jawahir, S., Smith, K., & Diez-Gonzalez, F. (2006). Soil survival of *Escherichia* coli O157:H7 acquired by a child from garden soil recently fertilized with cattle manure. *Journal of Applied Microbiology*, 101(2), 429–436.
- Muñoz-Marín, M. C., Gómez-Baena, G., López-Lozano, A., Moreno-Cabezuelo, J. A., Díez, J., & García-Fernández, J. M. (2020). Mixotrophy in marine picocyanobacteria: Use of organic compounds by *Prochlorococcus* and *Synechococcus*. *ISME Journal*, 14(5), 1065–1073.
- Murray, A. G., & Jackson, G. A. (1992). Viral dynamics: a model of the effects of size, shape, motion and abundance of single-celled planktonic organisms and other particles. *Marine Ecology Progress Series*, *89*, 103–116.
- Murray, J. D. (2003). Mathematical Biology II: Spatial Models and Biomedical Applications. Springer.
- Murray, J. D., Stanley, E. A., & Brown, D. L. (1986). On the spatial spread of rabies among foxes. Proceedings of the Royal Society B, 229(1255), 111–150.
- Nasser, W., Santhanam, B., Miranda, E. R., Parikh, A., Juneja, K., Rot, G., Dinh, C., Chen, R., Zupan, B., Shaulsky, G., & Kuspa, A. (2013). Bacterial discrimination by *Dictyostelid* amoebae reveals the complexity of ancient interspecies interactions. *Current Biology*, 23(10), 862–872.
- Nathan, R. (2006). Long-distance dispersal of plants. Science, 313(5788), 786-788.
- Netz, D. J. A., Bastos, M. do C. de F., & Sahl, H.-G. (2002). Mode of action of the antimicrobial peptide aureocin A53 from *Staphylococcus aureus*. *Applied and Environmental Microbiology*, *68*(11), 5274–5280.
- Neubert, M. G., & Caswell, H. (2000). Demography and dispersal: Calculation and sensitivity analysis of invasion speed for structured populations. *Ecology*, *81*(6), 1613–1628.
- Noble, J. V. (1974). Geographic and temporal development of plagues. Nature, 250(5469), 726-729.
- Noto, M. J., Burns, W. J., Beavers, W. N., & Skaar, E. P. (2017). Mechanisms of pyocyanin toxicity and genetic determinants of resistance in *Staphylococcus aureus*. *Journal of Bacteriology*, *199*(17), e00221-17.
- Ogden, L. D., Fenlon, D. R., Vinten, A. J., & Lewis, D. (2001). The fate of *Escherichia coli* O157 in soil and its potential to contaminate drinking water. *International Journal of Food Microbiology*, 66(1-2), 111–117.

- O'Hanlon, S. J., Rieux, A., Farrer, R. A., Rosa, G. M., Waldman, B., Bataille, A., Kosch, T. A., Murray, K. A., Brankovics, B., Fumagalli, M., Martin, M. D., Wales, N., Alvarado-Rybak, M., Bates, K. A., Berger, L., Böll, S., Brookes, L., Clare, F., Courtois, E. A., ... Fisher, M. C. (2018). Recent Asian origin of chytrid fungi causing global amphibian declines. *Science*, *360*(6389), 621–627.
- Okubo, A. (1971). Oceanic diffusion diagrams. Deep Sea Research, 18(8), 789-802.
- Oliver, A., Cantón, R., Campo, P., Baquero, F., & Blázquez, J. (2000). High frequency of hypermutable *Pseudomonas aeruginosa* in cystic fibrosis lung infection. *Science*, *288*(5469), 1251–1254.
- Ongeng, D., Muyanja, C., Ryckeboer, J., Geeraerd, A. H., & Springael, D. (2011). Rhizosphere effect on survival of *Escherichia coli* O157:H7 and *Salmonella enterica* serovar Typhimurium in manure-amended soil during cabbage (*Brassica oleracea*) cultivation under tropical field conditions in Sub-Saharan Africa. *International Journal of Food Microbiology*, 149(2), 133–142.
- Owen, M. R., & Lewis, M. A. (2001). How predation can slow, stop or reverse a prey invasion. Bulletin of Mathematical Biology, 63(4), 655–684.
- Pace, M. L., & Bailiff, M. D. (1987). Evaluation of a fluorescent microsphere technique for measuring grazing rates of phagotrophic microorganisms. *Marine Ecology Progress Series*, 40, 185–193.
- Padh, H., Ha, J., Lavasa, M., & Steck, T. L. (1993). A post-lysosomal compartment in *Dictyostelium discoideum*. Journal of Biological Chemistry, 268(9), 6742–6747.
- Paerl, H. W. (2018). Why does N-limitation persist in the world's marine waters? Marine Chemistry, 206, 1-6.
- Pan, M., Neilson, M. P., Grunfeld, A. M., Cruz, P., Wen, X., Insall, R. H., & Jin, T. (2018). A G-protein-coupled chemoattractant receptor recognizes lipopolysaccharide for bacterial phagocytosis. *PLoS Biology*, 16(5), e2005754.
- Pan, M., Xu, X., Chen, Y., & Jin, T. (2016). Identification of a chemoattractant G-protein-coupled receptor for folic acid that controls both chemotaxis and phagocytosis. *Developmental Cell*, 36(4), 428–439.
- Parisot, J., Carey, S., Breukink, E., Chan, W. C., Narbad, A., & Bonev, B. (2008). Molecular mechanism of target recognition by subtilin, a class I lanthionine antibiotic. *Antimicrobial Agents and Chemotherapy*, 52(2), 612–618.
- Parmeggiani, A., Krab, I. M., Okamura, S., Nielsen, R. C., Nyborg, J., & Nissen, P. (2006). Structural basis of the action of pulvomycin and GE2270 A on elongation factor Tu. *Biochemistry*, 45(22), 6846–6857.
- Paytan, A., & McLaughlin, K. (2007). The oceanic phosphorus cycle. Chemical Reviews, 107(2), 563-576.
- Pérez, A. A., Rodionov, D. A., & Bryant, D. A. (2016). Identification and regulation of genes for cobalamin transport in the cyanobacterium *Synechococcus* sp. strain PCC 7002. *Journal of Bacteriology*, 198(19), 2753–2761.
- Pérez, J., Moraleda-Muñoz, A., Marcos-Torres, F. J., & Muñoz-Dorado, J. (2016). Bacterial predation: 75 years and counting! *Environmental Microbiology*, 18(3), 766–779.
- Pernthaler, J. (2005). Predation on prokaryotes in the water column and its ecological implications. *Nature Reviews Microbiology*, *3*(7), 537–546.

- Pinhal, S., Ropers, D., Geiselmann, J., & de Jong, H. (2019). Acetate metabolism and the inhibition of bacterial growth by acetate. *Journal of Bacteriology*, 201(13), e00147-19.
- Potapov, A. M., Beaulieu, F., Birkhofer, K., Bluhm, S. L., Degtyarev, M. I., Devetter, M., Goncharov, A. A., Gongalsky, K. B., Klarner, B., Korobushkin, D. I., Liebke, D. F., Maraun, M., Mc Donnell, R. J., Pollierer, M. M., Schaefer, I., Shrubovych, J., Semenyuk, I. I., Sendra, A., ... Scheu, S. (2022). Feeding habits and multifunctional classification of soil-associated consumers from protists to vertebrates. *Biological Reviews of the Cambridge Philosophical Society*, *97*(3), 1057–1117.
- Providence, L. L., Andersen, O. S., Greathouse, D. V., Koeppe, R. E., II, & Bittman, R. (1995). Gramicidin channel function does not depend on phospholipid chirality. *Biochemistry*, 34(50), 16404–16411.
- Pyšek, P., & Hulme, P. E. (2005). Spatio-temporal dynamics of plant invasions: Linking pattern to process. *Écoscience*, *12*(3), 302–315.
- Rainey, P. B., & Travisano, M. (1998). Adaptive radiation in a heterogeneous environment. *Nature*, *394*(6688), 69–72.
- Rashidi, G., & Ostrowski, E. A. (2019). Phagocyte chase behaviours: Discrimination between Gram-negative and Gram-positive bacteria by amoebae. *Biology Letters*, *15*(1), 20180607.
- Rendueles, O., & Ghigo, J.-M. (2012). Multi-species biofilms: How to avoid unfriendly neighbors. FEMS Microbiology Reviews, 36(5), 972–989.
- Richardson, L. L., Goldberg, W. M., Kuta, K. G., Aronson, R. B., Smith, G. W., Ritchie, K. B., Halas, J. C., Feingold, J. S., & Miller, S. L. (1998). Florida's mystery coral-killer identified. *Nature*, 392(6676), 557–558.
- Ridgwell, A., & Arndt, S. (2015). Why dissolved organics matter. In *Biogeochemistry of Marine Dissolved Organic Matter* (pp. 1–20). Elsevier.
- Rillig, M. C., & Bonkowski, M. (2018). Microplastic and soil protists: A call for research. *Environmental Pollution*, 241, 1128–1131.
- Roberts, E. C., Wootton, E. C., Davidson, K., Jeong, H. J., Lowe, C. D., & Montagnes, D. J. S. (2011). Feeding in the dinoflagellate Oxyrrhis marina: Linking behaviour with mechanisms. Journal of Plankton Research, 33(4), 603–614.
- Rodríguez-Rojas, A., Kim, J. J., Johnston, P. R., Makarova, O., Eravci, M., Weise, C., Hengge, R., & Rolff, J. (2020). Non-lethal exposure to H2O2 boosts bacterial survival and evolvability against oxidative stress. *PLoS Genetics*, 16(3), e1008649.
- Rodriguez-Valera, F., Martín-Cuadrado, A.-B., Rodriguez-Brito, B., Pašić, L., Thingstad, T. F., Rohwer, F., & Mira, A. (2009). Explaining microbial population genomics through phage predation. *Nature Reviews Microbiology*, 7(11), 828–836.
- Ross, T., Ratkowsky, D. A., Mellefont, L. A., & McMeekin, T. A. (2003). Modelling the effects of temperature, water activity, pH and lactic acid concentration on the growth rate of *Escherichia coli*. *International Journal of Food Microbiology*, 82(1), 33–43.
- Roth-Rosenberg, D., Aharonovich, D., Omta, A. W., Follows, M. J., and Sher, D. (2021). Dynamic macromolecular composition and high exudation rates in *Prochlorococcus. Limnology and Oceanography*, 66(5), 1759–1773.

- Rousk, J., & Bååth, E. (2011). Growth of saprotrophic fungi and bacteria in soil. *FEMS Microbiology Ecology*, 78(1), 17–30.
- Rozen, Y., & Belkin, S. (2001). Survival of enteric bacteria in seawater. *FEMS Microbiology Reviews*, 25(5), 513–529.
- Rudolph, B., Gebendorfer, K. M., Buchner, J., & Winter, J. (2010). Evolution of *Escherichia coli* for growth at high temperatures. *Journal of Biological Chemistry*, 285(25), 19029–19034.
- Ruhe, Z. C., Nguyen, J. Y., Xiong, J., Koskiniemi, S., Beck, C. M., Perkins, B. R., Low, D. A., & Hayes, C. S. (2017). CdiA effectors use modular receptor-binding domains to recognize target bacteria. *mBio*, 8(2), e00290-17.
- Ruiz, G. M., Rawlings, T. K., Dobbs, F. C., Drake, L. A., Mullady, T., Huq, A., & Colwell, R. R. (2000). Global spread of microorganisms by ships. *Nature*, 408(6808), 49–50.
- Russell, A. B., LeRoux, M., Hathazi, K., Agnello, D. M., Ishikawa, T., Wiggins, P. A., Wai, S. N., & Mougous, J. D. (2013). Diverse type VI secretion phospholipases are functionally plastic antibacterial effectors. *Nature*, 496(7446), 508–512.
- Rutkowski, J., & Brzezinski, B. (2013). Structures and properties of naturally occurring polyether antibiotics. *BioMed Research International*, 2013, 1–31.
- Saeki, T., Umezawa, H., Tokieda-Fujishige, T., & Hori, M. (1974). Biological effects of azomycin (2-Nitroimidazole): Inhibition of ribonucleotide reductase. *Journal of Antibiotics*, 27(3), 225–227.
- Sandy, M., & Butler, A. (2009). Microbial iron acquisition: Marine and terrestrial siderophores. *Chemical Reviews*, *109*(10), 4580–4595.
- Sasaki, H., Takaki, A., Oshima, A., Ishida, A., & Nagata, S. (2007). Comparison of the function of L- and D-proline as compatible solute in *Escherichia coli* K-12 under high osmolarity. *Annals of Microbiology*, 57(2), 265–268.
- Scheele, B. C., Pasmans, F., Skerratt, L. F., Berger, L., Martel, A., Beukema, W., Acevedo, A. A., Burrowes, P. A., Carvalho, T., Catenazzi, A., De la Riva, I., Fisher, M. C., Flechas, S. V., Foster, C. N., Frías-Álvarez, P., Garner, T. W. J., Gratwicke, B., Guayasamin, J. M., Hirschfeld, M., ... Canessa, S. (2019). Amphibian fungal panzootic causes catastrophic and ongoing loss of biodiversity. *Science*, *363*(6434), 1459–1463.
- Schulz-Bohm, K., Geisen, S., Wubs, E. R. J., Song, C., de Boer, W., & Garbeva, P. (2017). The prey's scent -Volatile organic compound mediated interactions between soil bacteria and their protist predators. *ISME Journal*, 11(3), 817–820.
- Sender, R., Fuchs, S., & Milo, R. (2016). Revised estimates for the number of human and bacteria cells in the body. *PLoS Biology*, *14*(8), e1002533.
- Senwo, Z. N., & Tabatabai, M. A. (1998). Amino acid composition of soil organic matter. *Biology and Fertility of Soils*, 26(3), 235–242.
- Seymour, M. K., Wright, K. A., & Doncaster, C. C. (1983). The action of the anterior feeding apparatus of *Caenorhabditis elegans* (Nematoda: Rhabditida). *Journal of Zoology*, 201(4), 527–539.
- Shehata, T. E., & Marr, A. G. (1971). Effect of nutrient concentration on the growth of *Escherichia coli*. Journal of *Bacteriology*, 107(1), 210–216.

- Shenyakin, M. M., Ovchinnikov, Y. A., Ivanov, V. T., & Evstratov, A. V. (1967). Topochemical approach in studies of the structure-activity relation: Enantio-enniatin B. *Nature*, *213*(5074), 412–413.
- Shigesada, N., & Kawasaki, K. (2001). Biological invasions: Theory and practice. Oxford University Press.
- Shimizu, T., Takaya, N., & Nakamura, A. (2012). An L-glucose catabolic pathway in *Paracoccus* species 43P. *Journal of Biological Chemistry*, 287(48), 40448–40456.
- Shi, Y., Queller, D. C., Tian, Y., Zhang, S., Yan, Q., He, Z., He, Z., Wu, C., Wang, C., & Shu, L. (2021). The ecology and evolution of amoeba-bacterium interactions. *Applied and Environmental Microbiology*, 87(2), e01866-20.
- Siegel, D. A., DeVries, T., Cetinić, I., & Bisson, K. M. (2023). Quantifying the ocean's biological pump and its carbon cycle impacts on global scales. *Annual Review of Marine Science*, 15(1), 329–356.
- Singh, S. K., Wu, X., Shao, C., & Zhang, H. (2022). Microbial enhancement of plant nutrient acquisition. *Stress Biology*, *2*(1), 3.
- Sipler, R. E., & Bronk, D. A. (2015). Dynamics of dissolved organic nitrogen. In *Biogeochemistry of Marine Dissolved Organic Matter* (pp. 127–232). Elsevier.
- Sleight, S. C., Orlic, C., Schneider, D., & Lenski, R. E. (2008). Genetic basis of evolutionary adaptation by *Escherichia coli* to stressful cycles of freezing, thawing and growth. *Genetics*, 180(1), 431–443.
- Sleight, S. C., Wigginton, N. S., & Lenski, R. E. (2006). Increased susceptibility to repeated freeze-thaw cycles in Escherichia coli following long-term evolution in a benign environment. BMC Evolutionary Biology, 6, 104.
- Slingerland, C. J., Kotsogianni, I., Wesseling, C. M. J., & Martin, N. I. (2022). Polymyxin stereochemistry and its role in antibacterial activity and outer membrane disruption. *ACS Infectious Diseases*, *8*(12), 2396–2404.
- Smith, K. F., Sax, D. F., & Lafferty, K. D. (2006). Evidence for the role of infectious disease in species extinction and endangerment. *Conservation Biology*, 20(5), 1349–1357.
- Sniegowski, P. D., Gerrish, P. J., & Lenski, R. E. (1997). Evolution of high mutation rates in experimental populations of *E. coli. Nature*, 387(6634), 703–705.
- Solo-Gabriele, H. M., Wolfert, M. A., Desmarais, T. R., & Palmer, C. J. (2000). Sources of *Escherichia coli* in a coastal subtropical environment. *Applied and Environmental Microbiology*, 66(1), 230–237.
- Sørensen, S. J. (1991). Survival of Escherichia coli K12 in seawater. FEMS Microbiology Ecology, 8(2), 161–167.
- Spero, H. J. (1985). Chemosensory capabilities in the phagotrophic dinoflagellate *Gymnodinium fungiforme*. Journal of Phycology, 21(2), 181–184.
- Sterner, R. W. (2008). On the phosphorus limitation paradigm for lakes. *International Review of Hydrobiology*, 93(4-5), 433–445.
- Stubbendieck, R. M., & Straight, P. D. (2016). Multifaceted interfaces of bacterial competition. Journal of Bacteriology, 198(16), 2145–2155.
- Suarez, A. V., Holway, D. A., & Case, T. J. (2001). Patterns of spread in biological invasions dominated by long-distance jump dispersal: Insights from Argentine ants. *Proceedings of the National Academy of Sciences* of the United States of America, 98(3), 1095–1100.

- Svenning, J.-C., Pedersen, P. B. M., Donlan, C. J., Ejrnæs, R., Faurby, S., Galetti, M., Hansen, D. M., Sandel, B., Sandom, C. J., Terborgh, J. W., & Vera, F. W. M. (2016). Science for a wilder Anthropocene: Synthesis and future directions for trophic rewilding research. *Proceedings of the National Academy of Sciences of the United States of America*, 113(4), 898–906.
- Taddei, F., Radman, M., Maynard-Smith, J., Toupance, B., Gouyon, P. H., & Godelle, B. (1997). Role of mutator alleles in adaptive evolution. *Nature*, *387*(6634), 700–702.
- Tang, J. W. (2009). The effect of environmental parameters on the survival of airborne infectious agents. *Journal of the Royal Society, Interface*, 6(S6), S737–S746.
- Tang, J. Y., Bullen, N. P., Ahmad, S., & Whitney, J. C. (2018). Diverse NADase effector families mediate interbacterial antagonism via the type VI secretion system. *Journal of Biological Chemistry*, 293(5), 1504–1514.
- Tatem, A. J., Huang, Z., Das, A., Qi, Q., Roth, J., & Qiu, Y. (2012). Air travel and vector-borne disease movement. *Parasitology*, 139(14), 1816–1830.
- Taylor, B. L., Miller, J. B., Warrick, H. M., & Koshland, D. E., Jr. (1979). Electron acceptor taxis and blue light effect on bacterial chemotaxis. *Journal of Bacteriology*, *140*(2), 567–573.
- Tenaillon, O., Barrick, J. E., Ribeck, N., Deatherage, D. E., Blanchard, J. L., Dasgupta, A., Wu, G. C., Wielgoss, S., Cruveiller, S., Médigue, C., Schneider, D., & Lenski, R. E. (2016). Tempo and mode of genome evolution in a 50,000-generation experiment. *Nature*, 536(7615), 165–170.
- Tenaillon, O., Rodríguez-Verdugo, A., Gaut, R. L., McDonald, P., Bennett, A. F., Long, A. D., & Gaut, B. S. (2012). The molecular diversity of adaptive convergence. *Science*, *335*(6067), 457–461.
- Thingstad, T. F. (2000). Elements of a theory for the mechanisms controlling abundance, diversity, and biogeochemical role of lytic bacterial viruses in aquatic systems. *Limnology and Oceanography*, 45(6), 1320–1328.
- Thorpe, S. A. (2012). An introduction to ocean turbulence. Cambridge University Press.
- Thurman, E. M. (2014). Organic geochemistry of natural waters. Springer.
- Tilman, D. (1982). Resource competition and community structure. Monographs in Population Biology, 17, 1–296.
- Ting, S.-Y., Bosch, D. E., Mangiameli, S. M., Radey, M. C., Huang, S., Park, Y.-J., Kelly, K. A., Filip, S. K., Goo, Y. A., Eng, J. K., Allaire, M., Veesler, D., Wiggins, P. A., Peterson, S. B., Mougous, J. D. (2018). Bifunctional immunity proteins protect bacteria against FtsZ-targeting ADP-ribosylating toxins. *Cell*, 175(5), 1380–1392.e14.
- Toma, B., & Andral, L. (1977). Epidemiology of fox rabies. Advances in Virus Research, 21, 1-36.
- Topp, E., Welsh, M., Tien, Y.-C., Dang, A., Lazarovits, G., Conn, K., & Zhu, H. (2003). Strain-dependent variability in growth and survival of *Escherichia coli* in agricultural soil. *FEMS Microbiology Ecology*, 44(3), 303–308.
- Towle, K. M., & Vederas, J. C. (2017). Structural features of many circular and leaderless bacteriocins are similar to those in saposins and saposin-like peptides. *MedChemComm*, 8(2), 276–285.

- Trap, J., Bonkowski, M., Plassard, C., Villenave, C., & Blanchart, E. (2016). Ecological importance of soil bacterivores for ecosystem functions. *Plant and Soil*, 398(1-2), 1–24.
- Tymoszewska, A., Diep, D. B., Wirtek, P., & Aleksandrzak-Piekarczyk, T. (2017). The non-lantibiotic bacteriocin garvicin Q targets Man-PTS in a broad spectrum of sensitive bacterial genera. *Scientific Reports*, 7(1), 8359.
- Tymoszewska, A., Ovchinnikov, K. V., Diep, D. B., Słodownik, M., Maron, E., Martínez, B., & Aleksandrzak-Piekarczyk, T. (2021). *Lactococcus lactis* resistance to aureocin A53- and enterocin L50-like bacteriocins and membrane-targeting peptide antibiotics relies on the YsaCB-KinG-LlrG four-component system. *Antimicrobial Agents and Chemotherapy*, 65(12), e00921-21.
- Våge, S., Storesund, J. E., Giske, J., & Thingstad, T. F. (2014). Optimal defense strategies in an idealized microbial food web under trade-off between competition and defense. *PLOS ONE*, 9(7), e101415.
- van den Bosch, F., Hengeveld, R., & Metz, J. A. J. (1992). Analysing the velocity of animal range expansion. *Journal of Biogeography*, 19(2), 135–150.
- van Riper, C., III, van Riper, S. G., Goff, M. L., & Laird, M. (1986). The epizootiology and ecological significance of malaria in Hawaiian land birds. *Ecological Monographs*, *56*(4), 327–344.
- Vital, M., Hammes, F., & Egli, T. (2008). Escherichia coli O157 can grow in natural freshwater at low carbon concentrations. Environmental Microbiology, 10(9), 2387–2396.
- Vogel, G., Thilo, L., Schwarz, H., & Steinhart, R. (1980). Mechanism of phagocytosis in *Dictyostelium discoideum*. *Journal of Cell Biology*, 86(2), 456–465.
- Vranova, V., Zahradnickova, H., Janous, D., Skene, K. R., Matharu, A. S., Rejsek, K., & Formanek, P. (2012). The significance of D-amino acids in soil, fate and utilization by microbes and plants: Review and identification of knowledge gaps. *Plant and Soil*, 354(1-2), 21–39.
- Wang, G., & Doyle, M. P. (1998). Survival of enterohemorrhagic *Escherichia coli* O157:H7 in water. *Journal of Food Protection*, 61(6), 662–667.
- Warner, R. E. (1968). The role of introduced diseases in the extinction of the endemic Hawaiian avifauna. *Condor*, 70(2), 101–120.
- Waterbury, J. B., & Valois, F. W. (1993). Resistance to co-occurring phages enables marine Synechococcus communities to coexist with cyanophages abundant in seawater. *Applied and Environmental Microbiology*, 59(10), 3393–3399.
- Weinbauer, M. G. (2004). Ecology of prokaryotic viruses. FEMS Microbiology Reviews, 28(2), 127-181.
- Weinberg, Z. Y., Soliman, S. S., Kim, M. S., Chen, I. P., Ott, M., & El-Samad, H. (2024). De novo-designed minibinders expand the synthetic biology sensing repertoire. eLife, 13, e96154.
- Weitz, J. S. (2016). *Quantitative viral ecology: Dynamics of viruses and their microbial hosts*. Princeton University Press.
- Weitz, J. S., Hartman, H., & Levin, S. A. (2005). Coevolutionary arms races between bacteria and bacteriophage. *Proceedings of the National Academy of Sciences of the United States of America*, 102(27), 9535–9540.

- Wendling, C. C., Refardt, D., & Hall, A. R. (2021). Fitness benefits to bacteria of carrying prophages and prophage-encoded antibiotic-resistance genes peak in different environments. *Evolution*, 75(2), 515–528.
- Werdin-Pfisterer, N. R., Kielland, K., & Boone, R. D. (2009). Soil amino acid composition across a boreal forest successional sequence. *Soil Biology and Biochemistry*, 41(6), 1210–1220.
- West, J. S. (2014). Plant pathogen dispersal. In eLS. John Wiley & Sons.
- Whitman, R. L., & Nevers, M. B. (2003). Foreshore sand as a source of *Escherichia coli* in nearshore water of a Lake Michigan beach. *Applied and Environmental Microbiology*, 69(9), 5555–5562.
- Whitman, W. B., Coleman, D. C., & Wiebe, W. J. (1998). Prokaryotes: The unseen majority. *Proceedings of the National Academy of Sciences of the United States of America*, 95(12), 6578–6583.
- Wilken, S., Yung, C. C. M., Poirier, C., Massana, R., Jimenez, V., & Worden, A. Z. (2023). Choanoflagellates alongside diverse uncultured predatory protists consume the abundant open-ocean cyanobacterium *Prochlorococcus. Proceedings of the National Academy of Sciences of the United States of America*, 120(27), e2302388120.
- Williams, H. T. P. (2013). Phage-induced diversification improves host evolvability. *BMC Evolutionary Biology*, 13(1), 17.
- Winter, C., Bouvier, T., Weinbauer, M. G., & Thingstad, T. F. (2010). Trade-offs between competition and defense specialists among unicellular planktonic organisms: The "killing the winner" hypothesis revisited. *Microbiology and Molecular Biology Reviews*, 74(1), 42–57.
- Wiser, M. J., Ribeck, N., & Lenski, R. E. (2013). Long-term dynamics of adaptation in asexual populations. *Science*, *342*(6164), 1364–1367.
- Wootton, E. C., Zubkov, M. V., Jones, D. H., Jones, R. H., Martel, C. M., Thornton, C. A., & Roberts, E. C. (2007). Biochemical prey recognition by planktonic protozoa. *Environmental Microbiology*, 9(1), 216–222.
- World Health Organization. (2024). WHO's list of medically important antimicrobials: A risk management tool for mitigating antimicrobial resistance due to non-human use. https://cdn.who.int/media/docs/default-source/gcp/who-mia-list-2024-lv.pdf
- Woudstra, C., Sørensen, A. N., Sørensen, M. C. H., & Brøndsted, L. (2024). Strategies for developing phages into novel antimicrobial tailocins. *Trends in Microbiology*, 32(10), 996–1006.
- Yan, L. Z., Gibbs, A. C., Stiles, M. E., Wishart, D. S., & Vederas, J. C. (2000). Analogues of bacteriocins: Antimicrobial specificity and interactions of leucocin A with its enantiomer, carnobacteriocin B2, and truncated derivatives. *Journal of Medicinal Chemistry*, 43(24), 4579–4581.
- Yao, Z., Wei, G., Wang, H., Wu, L., Wu, J., & Xu, J. (2013). Survival of *Escherichia coli* O157:H7 in soils from vegetable fields with different cultivation patterns. *Applied and Environmental Microbiology*, 79(5), 1755–1756.
- Yilmaz, L. S., & Walhout, A. J. M. (2016). A *Caenorhabditis elegans* genome-scale metabolic network model. *Cell Systems*, 2(5), 297–311.

- Yoneyama, F., Imura, Y., Ichimasa, S., Fujita, K., Zendo, T., Nakayama, J., Matsuzaki, K., & Sonomoto, K. (2009). Lacticin Q, a lactococcal bacteriocin, causes high-level membrane permeability in the absence of specific receptors. *Applied and Environmental Microbiology*, 75(2), 538–541.
- Yoo, K., Han, I., Ko, K. S., Lee, T. K., Yoo, H., Khan, M. I., Tiedje, J. M., & Park, J. (2019). *Bacillus*-dominant airborne bacterial communities identified during Asian dust events. *Microbial Ecology*, 78(3), 677–687.
- Yow, G.-Y., Uo, T., Yoshimura, T., & Esaki, N. (2006). Physiological role of D-amino acid-N-acetyltransferase of Saccharomyces cerevisiae: Detoxification of D-amino acids. Archives of Microbiology, 185(1), 39–46.

Yu, J., Liberton, M., Cliften, P. F., Head, R. D., Jacobs, J. M., Smith, R. D., Koppenaal, D. W., Brand, J. J., & Pakrasi, H. B. (2015). *Synechococcus elongatus* UTEX 2973, a fast growing cyanobacterial chassis for biosynthesis using light and CO<sub>2</sub>. *Scientific Reports*, *5*(1), 8132.

- Zhang, T., Wang, H., Wu, L., Lou, J., Wu, J., Brookes, P. C., & Xu, J. (2013). Survival of *Escherichia coli* O157:H7 in soils from Jiangsu Province, China. *PLoS ONE*, *8*(12), e81178.
- Zhen, C., Ge, X.-F., Lu, Y.-T., & Liu, W.-Z. (2023). Chemical structure, properties and potential applications of surfactin, as well as advanced strategies for improving its microbial production. *AIMS Microbiology*, 9(2), 195–217.
- Zhu, L., Zeng, J., Wang, C., & Wang, J. (2022). Structural basis of pore formation in the mannose phosphotransferase system by pediocin PA-1. *Applied and Environmental Microbiology*, 88(3), e01992-21.
- Zhurbas, V., & Oh, I. S. (2004). Drifter-derived maps of lateral diffusivity in the Pacific and Atlantic Oceans in relation to surface circulation patterns. *Journal of Geophysical Research: Oceans, 109*(C5), C05015.